This is to confirm that the student has carefully and thoroughly gone through the Examiners comments and reports and has carried out the corrections and comments by the examiners to my satisfaction as his supervision. I therefore approve the submission of the corrected and final version of her thesis. Thanks

O.T. Mewomo (PhD) Supervisor 25-01-2023



Fixed Point Approach for Solving Optimization Problems in Hilbert, Banach and Convex Metric Spaces

by

Grace Nnennaya Ogwo

Thesis submitted in fulfilment of the requirements for the degree of Doctor of Philosophy $\rm (PhD)$

The University of KwaZulu-Natal



School of Mathematics, Statistics and Computer Science University of KwaZulu-Natal, South Africa.

January, 2023.

Fixed point approach for solving optimization problems in Hilbert, Banach and convex metric spaces

by

Grace Nnennaya Ogwo B.Sc (Hons)(MOUAU). M.Sc (UCT). M.Sc (UKZN)

As the candidate's supervisor, I have approved this thesis for submission.

Prof. O. T. Mewomo

.....

Dedication

This thesis is dedicated to my father Mr. Ogwo Chukwu Okoroafor (of blessed memory) for all his sacrifices, support and words of encouragement.

Acknowledgments

I return all glory to Almighty God whose mercies and grace were sufficient for me all through my PhD program. I thank God specially for the protection and provision He granted me during the period of my study. I appreciate Him a lot for not giving up on me.

I sincerely appreciate my supervisor and mentor, Prof. Oluwatosin T. Mewomo for his passion for excellence, understanding, guidance, kindness, patience, counseling and all the contributions he made towards the success of my program. I thank him for granting me the privilege of working with him. The advice and the push motivated me and contributed to the success of this program. I am also grateful for his timely and careful proof-reading of this thesis which greatly improved the quality of this thesis.

My profound gratitude goes to my co-supervisor, Dr. Timilrhin O. Alakoya for his time, support and contributions towards the success of this program. I really appreciate it.

I specially thank Dr. G. C. Ugwunnadi for all the advice and support he has given to me all through my academic journey. Thank you for believing in me more than I ever believed in myself.

I am deeply indebted to Dr. Chinedu Izuchukwu for being a friend and mostly a teacher. Thank you so much for your patience, advice, time, financial support and contributions towards the success of my program.

I acknowledge with many thanks the scholarship and financial support from the University of KwaZulu-Natal (UKZN) Doctoral scholarship for my study.

My sincere gratitude goes to the University of KwaZulu-Natal (UKZN), South Africa, College of Agriculture, Engineering and Science, and School of Mathematics, Statistics and Computer Science for her financial support (fee remission), access to learning resources and conducive learning environment all through the program. I also appreciate the administrator, Howard College, School of Mathematics, Statistics and Computer Science, UKZN in the person of Princess Bavuyile Nhlangulela for her care and hospitality.

My heartfelt gratitude goes to Dr. Adeolu Taiwo, Dr. Kazeem Aremu, Dr. Chibueze Okeke, Dr. Hammed Abass, Dr. Olawale Kazeem Oyewole, Dr. Lateef Jolaoso and Dr. Akindele Mebawondu for all their help and support. My profound gratitude goes to my colleagues; Emeka Godwin, Abd-semii- Owolabi Oluwatosin, Uzor Victor and Regina Nwokoye.

I sincerely appreciate Mrs. Harriet Mbah (my granny) for her support, encouragement and financial support. I thank my uncle Mr. Elijah Mbah for his support financially and other ways. Thank you for all you do.

My special appreciation goes to my best friends; Daniel Igwe, Eunice Mbah and Chiamaka Obi. Thank you guys for all the times you lent me a shoulder. To my friends Mercy Eke Stephen and Justina Oriaku thank you for been such good friends. To my family; Mr. Ogwo Chukwu Okoroafor (of blessed memory), Mrs. Love Ogwo, Mr. Jennifer Onyedikachi, Mrs. Ebere Chukwu, Alfred Ogwo, Clinton Ogwo, Godswill Ogwo, my sweetest sister Chisom Ogwo, my aunties, uncles and cousins. Your words of encouragements are highly appreciated.

Abstract

In this thesis, we study the fixed point approach for solving optimization problems in real Hilbert, Banach and Hadamard spaces. These optimization problems include the variational inequality problem, split variational inequality problem, generalized variational inequality problem, split equality problem, monotone inclusion problem, split monotone inclusion problem, minimization problem, split equilibrium problem, among others. We consider some interesting classes of mappings such as the nonexpansive semigroup in real Hilbert spaces, strict pseudo-contractive mapping in real Hilbert spaces and 2-uniformly convex real Banach spaces, nonexpansive mapping between a Hilbert space and a Banach space, and quasi-pseudocontractive mapping in Hilbert spaces and Hadamard spaces. We introduce several iterative schemes for approximating the solutions of the various aforementioned optimization problems and fixed point problems and prove their convergence results. We adopt and implement several inertial methods such as the inertial-viscosity-type algorithm, relaxed inertial subgradient extragradient, modified inertial forward-backward splitting algorithm viscosity method, among others. Furthermore, we present several novel and practical applications of our results to solve other optimization problems, image restoration problem, among others. Finally we present several numerical examples in comparison with some results in the literature to illustrate the applicability of our proposed methods.

Contents

	Title	e page		ii			
	Ded	ication		iii			
	Ack	nowledg	gements	iv			
	Abs	tract .		vi			
	Decl	laration		xi			
	Con	tributic	ons from the thesis	xii			
1	General Introduction 1						
	1.1	Backg	round of study	1			
	1.2	Resear	rch problems and motivation	4			
		1.2.1	Research problems	4			
		1.2.2	Motivation	7			
	1.3	Objec	tives	10			
	1.4	Organ	ization of the thesis	11			
2	Pre	limina	ries and Literature Review	14			
	2.1	Geom	etric properties of Hilbert spaces	14			
		2.1.1	Some inequalities that characterize Hilbert spaces	15			
		2.1.2	Some nonlinear single-valued mappings in Hilbert spaces	15			
	2.2	Geom	etric properties of Banach spaces	19			
		2.2.1	Some inequalities that characterize Banach spaces	21			
		2.2.2	Some nonlinear single-valued mappings in Banach spaces	22			

	2.3	Geometric properties of geodesic metric spaces					
		2.3.1	Metric midpoints	23			
		2.3.2	Geodesic triangles	24			
		2.3.3	Geometric properties of Hadamard spaces	24			
		2.3.4	Examples of Hadamard spaces	26			
		2.3.5	Quasilinearization mapping and dual space	26			
		2.3.6	Some inequalities that characterize Hadamard spaces	28			
		2.3.7	Some nonlinear single-valued mappings in Hadamard spaces \ldots .	28			
		2.3.8	Monotone operators and its resolvents in Hadamard spaces	29			
	2.4	Metrie	c and generalized projections	30			
		2.4.1	Convex functions	32			
	2.5	Revie	w on some optimization problems	35			
		2.5.1	Variational inequality and fixed point problems	35			
		2.5.2	Split variational inequality problem	39			
		2.5.3	Split equalities of equilibrium, variational inequality and fixed point problems	42			
		2.5.4	Variational inclusion and split equilibrium problems	44			
		2.5.5	Split monotone variational inclusion and fixed point problems $\ . \ .$	46			
		2.5.6	Minimization and fixed point problems	48			
		2.5.7	Some important lemmas	51			
3	Res	Results on Variational Inequality and Fixed Point Problems					
	3.1	Introduction					
	3.2	Preliminaries					
	3.3	Comm	non solution of the variational inequality and the fixed point problems	59			
		3.3.1	Proposed method	60			
		3.3.2	Convergence analysis	62			
		3.3.3	Applications	73			
		3.3.4	Numerical experiments	75			
	3.4	On qu	asimonotone variational inequality problems	80			
		3.4.1	Proposed methods	81			
		3.4.2	Convergence analysis	85			
		3.4.3	Numerical experiments	101			

	3.5	On minimum-norm solutions of quasimonotone variational inequalities with fixed point constraint					
		3.5.1	Proposed method	107			
		3.5.2	Convergence analysis	109			
		3.5.3	Numerical experiments	117			
4	\mathbf{Res}	ults or	n Split Variational Inequality Problems	128			
	4.1	Introd	luction	128			
	4.2	On sp	lit variational inequality problems without product space formulation	128			
		4.2.1	Proposed methods	129			
		4.2.2	Convergence analysis	135			
		4.2.3	Numerical experiments	148			
	4.3	On sp	lit variational inequality problem beyond monotonicity	157			
		4.3.1	Proposed methods	158			
		4.3.2	Convergence analysis	163			
		4.3.3	Numerical experiments	175			
	4.4	Projec	tion and contraction methods for split variational inequality problem	185			
		4.4.1	Proposed methods	185			
		4.4.2	Convergence analysis	188			
		4.4.3	Numerical experiments	206			
	4.5	On a o	class of generalized variational inequality problem	210			
		4.5.1	Main results	212			
		4.5.2	Numerical experiments	221			
5	6 Results on Split Equalities Variational Inclusion and Split Equilibrium						
	Pro	blems		230			
	5.1	Introd	luction	230			
	5.2	On sp	lit equalities of some nonlinear optimization problems	230			
		5.2.1	Proposed method	231			
		5.2.2	Convergence analysis	234			
		5.2.3	Numerical experiments	247			
	5.3	On mi	inimum-norm solutions of inclusion and split equilibrium problems	252			
		5.3.1	Proposed method	252			
		5.3.2	Convergence analysis	255			

		5.3.3	Applications	265		
		5.3.4	Numerical experiments	267		
6	Results on Variational Inequality, Split Monotone Variational Inclusion and Fixed Point Problems in Banach Spaces 27					
	6.1	Introd	luction	274		
	6.2	On ps	eudomonotone variational inequalities with non-Lipschitz operators .	274		
		6.2.1	Proposed method	275		
		6.2.2	Convergence analysis	276		
		6.2.3	Numerical experiments	287		
6.3 On finite family of split monotone variational inclusion and problems		nite family of split monotone variational inclusion and fixed point ems	293			
		6.3.1	Proposed method	293		
		6.3.2	Convergence analysis	294		
		6.3.3	Application	306		
		6.3.4	Numerical experiments	309		
7	Results on Minimization Problems in Hadamard Spaces 313					
	7.1	Introd	luction	313		
	7.2	Prelin	ninaries	313		
7.3 Modified proximal point methods		Modif	ied proximal point methods	316		
		7.3.1	Main results	317		
		7.3.2	Extension to monotone inclusion problems	321		
		7.3.3	Numerical experiments	326		
	7.4	On ge	neralized viscosity implicit rule for quasi- pseudocontractive mappings	<mark>s</mark> 330		
		7.4.1	Main result	333		
8	Cor	nclusio	n, Contributions to Knowledge and Future research	341		
	8.1	Concl	usion	341		
	8.2	Contributions to Knowledge				
	8.3	Future	e research	343		

Declaration

This thesis in its entirety or in part, has not been submitted to this or any other institution in support of an application for the award of a degree. It represents the author's own work and where the work of others has been used, proper reference has been made.

Grace Nnennaya Ogwo

.....

Contributed papers from the thesis

Papers published/accepted from the thesis.

- G. N. Ogwo, T. O. Alakoya, O. T. Mewomo, Inertial forward-backward method with self-adaptive step sizes for finding minimum-norm solutions of inclusion and split equilibrium problems, Applied Set-Valued Analysis and Optimization, 4 (2) (2022), 185-206.
- (2) G. N. Ogwo, C. Izuchukwu, Y. Shehu, O. T. Mewomo, Convergence of Relaxed Inertial Subgradient Extragradient Methods for Quasimonotone Variational Inequality Problems, Journal of Scientific Computing, 90 (1) (2022), Paper No. 10, 35 pp.
- (3) G. N. Ogwo, C. Izuchukwu, O.T. Mewomo, Relaxed inertial methods for solving split variational inequality problem without product space formulation, Acta Mathematica Scientia, 42 (5) (2022), 1701-1733.
- (4) G. N. Ogwo, T.O. Alakoya, O. T. Mewomo, Inertial iterative method with selfadaptive step size for finite family of split monotone variational inclusion and fixed point problems in Banach spaces, Demonstratio Mathematica, 59 (1) (2021), 198-216.
- (5) G. N. Ogwo, T.O. Alakoya, O. T. Mewomo, Iterative algorithm with self-adaptive size for approximating the common solution of variational inequality and fixed point problems, Optimization, (2021), https://doi.org/10.1080/02331934.2021.1981897.
- (6) G. N. Ogwo, C. Izuchukwu, O. T. Mewomo, Inertial Extrapolation Method for a class of Generalized Variational Inequality Problem in real Hilbert Spaces, Periodica Mathematica Hungarica, (2022), https://doi.org/10.1007/s10998-022-00470-w
- (7) G. N. Ogwo, C. Izuchukwu, O. T. Mewomo, Inertial methods for finding minimumnorm solutions of the split variational inequality problem beyond monotonicity, Numerical Algorithms, 88 (3) (2021), 1419-1456.
- (8) G. N. Ogwo, H. A. Abass, C. Izuchukwu, O. T. Mewomo, Modified proximal point methods involving quasi-psedocontractive mappings in Hadamard spaces, Acta Mathematica Vietnamica, 47 (2022), 847-873.
- (9) G. N. Ogwo, T. O. Alakoya, O. T. Mewomo, Inertial subgradient extragradient method with adaptive step-size for pseudomonotone variational inequalities with non-Lipschitz operators in Banach spaces, Journal of Industrial and Management Optimization, (2022), Doi: 10.3934/jimo.2022239.

Selected papers under review in peer reviewed journals

(1) O. T. Mewomo, **G. N. Ogwo**, T. O. Alakoya, Inertial iterative algorithm for approximating common solution of split equalities of some nonlinear optimization problems, Numerical Algebra, Control and Optimization.

- (2) G. N. Ogwo, T. O. Alakoya, O. T. Mewomo, Strong convergent algorithm for finding minimum-norm solutions of quasimonotone variational inequalities with fixed point constraint and application, Journal of Scientific Computing.
- (3) G. N. Ogwo, T. O. Alakoya, C. Izuchukwu, O. T. Mewomo, Strong convergent inertial projection and contraction methods for split variational inequality problem, Acta Mathematica Scientia.
- (4) G. N. Ogwo, O. T. Mewomo, A generalized viscosity implict rule for quasi- pseudocontractive mappings in Hadamard spaces, Topological Algebra and Its Applications.

CHAPTER 1

General Introduction

1.1 Background of study

Optimization theory has shown to be a very important research area with numerous applications in various fields. This is because many problems arising from these fields can be modeled as optimization problems. Optimization problems include; the variational inequality problem (VIP), split variational inequality problem (SVIP), generalized variational inequality problem (GVIP), variational inclusion problem (V_q IP), split monotone variational inclusion problem (SMV_qIP), split equality problem (SE_qp), split equilibrium problem (SEP), minimization problem (MP), among others. These optimization problems can be transformed into a fixed point problem (FPP) of a suitable nonlinear mapping. The fixed point problem is defined as follows:

Find
$$x^* \in X$$
 such that $Tx^* = x^*$, (1.1.1)

where X is a metric space and $T: X \to X$ is a nonlinear mapping. We denote by F(T) the fixed point set of T, that is

$$F(T) = \{x^* \in X : Tx^* = x^*\}.$$
(1.1.2)

The fixed point method is one of the most effective methods for solving optimization problems. Fixed point theory is a crucial area of research in nonlinear functional analysis which has continuously attracted the interest of numerous researchers due to its applications in various branches of mathematics and other related fields such as economics, physics, chemistry, biology, dynamical system theory, data science, among others (see [28, 87, 134, 172, 173, 212, 219, 236, 260] and other references therein).

Poincaré [207] introduced the study of the fixed point theory in 1886. Brouwer [46] in 1912 proved that a continuous mapping on a closed unit ball in finite dimensional space has

a fixed point. Birkoff and Kellog [42] extended the result of Brouwer [46] to the infinite dimensional space.

The fixed points of nonlinear mappings are difficult and sometimes impossible to obtain. Hence, there is a need to study and develop iterative schemes to approximate the fixed points of nonlinear mappings. In 1922, Banach [38] proved the famous Banach contraction principle which is also known as the Banach fixed point theorem. He proved that a contraction mapping T defined on a complete metric space X has a unique fixed point. Furthermore, he proved that for any starting point $x_0 \in X$, the sequence defined by the Picard iteration

$$x_{n+1} = Tx_n, \qquad n \ge 1, \tag{1.1.3}$$

converges to the unique fixed point. The Banach fixed point theorem guarantees the existence and uniqueness of fixed points of certain self mappings. This has made it one of the main foundations of the theory of metric fixed points. However, for classes of mappings that are more general than the class of contraction mappings, the Banach contraction principle may fail to hold. Several examples in literature have shown that for a nonexpansive mapping T, the sequence generated by the Picard iteration (1.1.3) may fail to converge even when the fixed point exists. To overcome this limitation, several authors have studied and proposed iterative methods for approximating fixed points of mappings that are more general that the contraction mapping. The Mann iterative method proposed by Mann [171] is a more general iterative formula for approximating fixed points of nonlinear mappings. The Mann iteration generates a sequence $\{x_n\}$ as follows:

$$x_{n+1} = (1 - a_n)x_n + a_n T x_n, \quad n \in \mathbb{N} \text{ and } x_1 \in \mathcal{H},$$

$$(1.1.4)$$

where $\{a_n\}_{n=1}^{\infty}$ is a sequence in (0,1) and \mathcal{H} is a real Hilbert space. The Mann iterative method converges weakly if T is nonexpansive, $\lim_{n\to\infty} a_n = 0$ and $\sum_{n=0}^{\infty} a_n = \infty$. However, when T is not nonexpansive, the Mann iterative method may fail to converge to a fixed point of T even when the algorithm converges. If $\{a_n\} = \{\lambda\}$ where λ is a positive real number, then (1.1.4) reduces to the Kransnoselskii iterative method proposed by Kransnoselskii [157], and if $\{a_n\} = 1$, then (1.1.4) reduces to (1.1.3). In 1974, Ishikawa [127] introduced a generalization of the Mann iterative method for approximating fixed points of pseudocontractive mappings in Hilbert spaces. Ishikawa called this generalization the Ishikawa method and defined it as follows:

$$x_{n+1} = (1 - a_n)x_n + a_n T[(1 - b_n)x_n + b_n T x_n], \ n \in \mathbb{N} \text{ and } x_1 \in \mathcal{H},$$
(1.1.5)

where $\{a_n\}_{n=1}^{\infty}$ and $\{b_n\}_{n=1}^{\infty}$ are sequences in [0, 1]. Writing (1.1.5) in a system form, we have

$$\begin{cases} y_n = (1 - b_n)x_n + b_n T x_n \\ x_{n+1} = (1 - a_n)x_n + a_n T y_n, \quad n \in \mathbb{N}, \end{cases}$$
(1.1.6)

which implies that the Ishikawa iteration can be seen as a double Mann iteration and it reduces to Mann iteration when $b_n = 0$.

Agarwal *et al.* [2] introduced and studied the following S-iteration method defined as follows:

$$x_{n+1} = (1 - a_n)Tx_n + a_nT[(1 - b_n)x_n + b_nTx_n], \quad n \in \mathbb{N} \text{ and } x_1 \in \mathcal{H},$$
(1.1.7)

where $\{a_n\}_{n=1}^{\infty}$ and $\{b_n\}_{n=1}^{\infty}$ are sequences in (0, 1). It is well known that (1.1.7) has better convergence properties than (1.1.4) and (1.1.5) (see [3]).

We note that (1.1.4), (1.1.5) and (1.1.7) only generates sequences that converge weakly. However, strong convergence are more desirable than the weak convergence in infinite dimensional space. In an attempt to obtain strong convergence results, Halpern [117] introduced the Halpern iterative method which converges strongly to a fixed point of a nonexpansive mapping in real Hilbert spaces. Halpern [117] defined it as follows;

$$x_{n+1} = \alpha_n u + (1 - \alpha_n) T x_n, \quad \text{for} \quad n \in \mathbb{N},$$
(1.1.8)

where $\{\alpha_n\} \subset [0, 1]$ and $u, x_1 \in \mathcal{H}$. The Halpern iterative method was generalized to the viscosity iterative method by Moudafi [181]. The viscosity iterative method is defined as follows:

$$x_{n+1} = \alpha_n f(x_n) + (1 - \alpha_n) T x_n, \quad \text{for} \quad n \in \mathbb{N},$$
(1.1.9)

where $\{\alpha_n\} \subset [0,1]$ and f is a contraction mapping on \mathcal{H} .

The implicit midpoint rule is an important method for solving Ordinary Differential Equations (ODEs) because of its ability to eliminate stability errors of systems of ODEs (see [32, 203, 226] and other references therein). For an initial value problem of the ODE

$$x'(t) = f(x(t)), \quad x(0) = x_0,$$
 (1.1.10)

the implicit midpoint rule is a recursion procedure that generates the sequence $\{x_n\}$ by

$$x_{n+1} = x_n + hf\left(\frac{x_n + x_{n+1}}{2}\right), \quad n \ge 0,$$
 (1.1.11)

where h > 0 is a stepsize and $f : \mathbb{R}^N \to \mathbb{R}^N$ is a continuous function. It is known that if f is Lipschitz continuous and sufficiently smooth, then the sequence $\{x_n\}$ converges to an exact solution of (1.1.10) as $h \to 0$ uniformly over $u \in [0, u]$ for any fixed point u. In 2015, Xu et. al. [263] combined (1.1.9) and (1.1.11) for nonexpansive mappings in Hilbert spaces and proposed a viscosity implicit midpoint rule defined as follows:

$$x_{n+1} = \alpha_n f(x_n) + (1 - \alpha_n) T\left(\frac{x_n + x_{n+1}}{2}\right), \quad n \ge 0,$$
 (1.1.12)

where x_0 is an arbitrary point in the domain of T, f is a contraction on \mathcal{H} and $\{\alpha_n\} \in [0, 1]$. They proved that the sequence generated converge strongly under some suitable conditions to a fixed point of the nonexpansive mapping which is also a solution of a variational inequality. It is well known that (1.1.9) and (1.1.12) converge faster than (1.1.8) (see [263]). Over the years, several researchers have studied these iterative methods (1.1.4)-(1.1.12) and their modifications for approximating fixed points of several nonlinear mappings and solutions of optimization problems in Hilbert, Banach and Hadamard spaces (see [5, 6, 23, 24, 51, 74, 105, 110, 152, 191, 194] and other references therein).

In this thesis, we study the VIP, V_q IP, SE_q P, SEP, MP, FPP and other related optimization problems in the frameworks of Hilbert, Banach and Hadamard spaces. We present several effective iterative methods for approximating the solutions of these problems. Furthermore, we analyze the convergence results of our proposed methods. Finally, we give where necessary some applications of our results and present numerical examples to show the applicability of our methods.

1.2 Research problems and motivation

In this section, we discuss the research problems and motivation for the study.

1.2.1 Research problems

In this thesis, we consider the following optimization problems; VIP, SVIP, GVIP, V_qIP , SMV_qIP, SE_qP, SEP, MP and FFP of certain nonlinear mappings in the frameworks of Hilbert, Banach and Hadamard spaces. A detailed review of these optimization problems and the spaces under consideration will be given in Chapter 2.

Let \mathcal{C} be a nonempty, closed and convex subset of a real Hilbert space \mathcal{H} and $A : \mathcal{H} \to \mathcal{H}$ be an operator. The VIP is defined as follows: Find $x \in \mathcal{C}$ such that

$$\langle Ax, y - x \rangle \ge 0, \ \forall y \in \mathcal{C}.$$
 (1.2.1)

We denote the solution set of the problem (1.2.1) by $VI(\mathcal{C}, A)$. Many authors have proposed and analyzed several iterative algorithms for solving Problem (1.2.1) (see [9, 135], and references therein). To accelerate the rate of convergence of these iterative algorithms, authors often employ the inertial technique. Polyak [208] studied the convergence of the following inertial extrapolation algorithm

$$x_{n+1} = x_n + \beta_n (x_n - x_{n-1}) - \alpha_n A x_n, \ \forall n \ge 0,$$

where $\{\alpha_n\}$ and $\{\beta_n\}$ are two real sequences. Recently, there has been an increased interest in studying inertial type algorithm (see [5, 7, 8, 27, 40, 96, 246] and other references therein). On the other hand, the relaxation technique has also proven to be an essential ingredient in the resolution of optimization problems due to the improved convergence rate that it contributes to iterative schemes.

In this work, we propose and study new inertial viscosity Tseng's extragradient algorithms with self-adaptive step size for solving the VIP and the FPP in Hilbert spaces. Our main aim is to find a common solution of the VIP (1.2.1) and the FPP (1.1.1) for an infinite family of strict pseudo-contraction mappings T. That is, find a point $x^* \in C$ such that

$$x^* \in \Gamma = VI(\mathcal{C}, A) \bigcap \bigcap_{i=1}^{\infty} F(T_i).$$
(1.2.2)

Furthermore, we present and study two new relaxed inertial subgradient extragradient methods for solving quasimonotone VIPs in a real Hilbert space. Also, we present a new inertial Tseng's extragradient method with self-adaptive step size for approximating the minimum-norm solutions of the quasimonotone VIP and FPP of a quasi-pseudocontractive mapping T in the framework of Hilbert space. That is, the problem of finding a point $x^* \in \mathcal{C}$ such that

$$x^* \in \Gamma = VI(\mathcal{C}, A) \bigcap F(T).$$
(1.2.3)

The VIP (1.2.1) was later generalized to the following SVIP: Find $x \in \mathcal{C}$ such that

$$\langle Ax, y - x \rangle \ge 0, \quad \forall \ y \in \mathcal{C},$$
 (1.2.4)

and $z = \mathcal{T}x \in \mathcal{Q}$ solves

$$\langle Fz, u-z \rangle \ge 0, \quad \forall \ u \in \mathcal{Q},$$
 (1.2.5)

where \mathcal{C} and \mathcal{Q} are nonempty, closed and convex subsets of real Hilbert spaces \mathcal{H}_1 and \mathcal{H}_2 respectively, $A : \mathcal{H}_1 \to \mathcal{H}_1, F : \mathcal{H}_2 \to \mathcal{H}_2$ are two operators and $\mathcal{T} : \mathcal{H}_1 \to \mathcal{H}_2$ is a bounded linear operator.

To study the SVIP, we present two new relaxed methods with inertial steps and two new inertial methods for solving the SVIP in real Hilbert spaces without any product space transformation, and for which the underlying operators are freed from the restrictive cocoercive assumption. Furthermore, we propose two new inertial projection and contraction methods for solving the SVIPs in real Hilbert spaces without the co-coercive condition and without the product space formulation, which does not fully exploit the attractive splitting structure of the SVIP. Also, we study an inertial viscosity-type method for solving a more generalized VIP (GVIP) in Hilbert spaces which is defined as follows: Find $x \in C$ such that

$$\langle Ax, y - x \rangle \ge 0, \quad \forall y \in \mathcal{C} \text{ and } \mathcal{T}x \in F(T),$$

$$(1.2.6)$$

where \mathcal{T} is a bounded linear operator and F(T) is as defined in (1.1.2).

Furthermore, we consider the split equalities of EP, VIP and FPP.

Let \mathcal{C} be a nonempty, closed and convex subset of a real Hilbert space \mathcal{H} and $\Phi : \mathcal{C} \times \mathcal{C} \to \mathbb{R}$ be a bifunction. The EP is defined as follows: Find $x \in \mathcal{C}$ such that

$$\Phi(x,y) \ge 0, \ \forall \ y \in \mathcal{C}.$$
(1.2.7)

The solution set of the Problem (1.2.7) is denoted by $EP(\Phi)$.

Let $\mathcal{H}_1, \mathcal{H}_2$ and \mathcal{H}_3 be real Hilbert spaces. Let \mathcal{C}, \mathcal{Q} be nonempty, closed and convex subsets of \mathcal{H}_1 and \mathcal{H}_2 , respectively. Let $\mathcal{F}_1 : \mathcal{H}_1 \to \mathcal{H}_3$ and $\mathcal{F}_2 : \mathcal{H}_2 \to \mathcal{H}_3$ be bounded linear operators. The SE_qP is defined as follows:

Find
$$x \in \mathcal{C}$$
 and $y \in \mathcal{Q}$ such that $\mathcal{F}_1 x = \mathcal{F}_2 y$. (1.2.8)

We denote the solution set of (1.2.8) by

$$\Omega_{SEP} := \left\{ (x, y) \in \mathcal{C} \times \mathcal{Q} \mid \mathcal{F}_1 x = \mathcal{F}_2 y \right\}.$$

In this study, we present an inertial Tseng's algorithm for approximating the common solution of split equalities of EP, VIP and FPP in Hilbert spaces. That is, find a point $(x, y) \in \Gamma = \{x \in EP(\Phi_1) \cap VI(\mathcal{C}, A) \cap F(\mathcal{T}_a), y \in EP(\Phi_2) \cap VI(\mathcal{Q}, B) \cap F(\mathcal{T}_b) : \mathcal{F}_1 x = \mathcal{F}_2 y\}$, where \mathcal{T}_a and \mathcal{T}_b are one-parameter nonexpansive semigroups and $F(\mathcal{T}_a), F(\mathcal{T}_b)$ are as defined in (1.1.2).

The next optimization problems considered in this thesis are the V_q IP and the SEP.

Let \mathcal{H} be a real Hilbert space and $B: \mathcal{H} \to \mathcal{H}$ be an operator. The V_q IP is defined as: Find $x^* \in \mathcal{H}$ such that

$$0 \in (B+D)x^*,$$
 (1.2.9)

where $D: \mathcal{H} \to 2^{\mathcal{H}}$ is a multivalued operator. The solution set of the Problem (1.2.9) is called the set of zero points of (B+D) and it is denoted by $(B+D)^{-1}(0)$.

Let \mathcal{C}, \mathcal{Q} be nonempty subsets of $\mathcal{H}_1, \mathcal{H}_2$, respectively, and $F_1 : \mathcal{C} \times \mathcal{C} \to \mathbb{R}, F_2 : \mathcal{Q} \times \mathcal{Q} \to \mathbb{R}$ be bifunctions. Let $\mathcal{T} : \mathcal{H}_1 \to \mathcal{H}_2$ be a bounded linear operator. The SEP is defined as follows: Find $x \in \mathcal{C}$ such that

$$F_1(x, x^*) \ge 0, \ \forall \ x \in \mathcal{C} \tag{1.2.10}$$

and such that

$$y = \mathcal{T}x \in \mathcal{Q}$$
 solves $F_2(y, y^*), \forall y^* \in \mathcal{Q}.$ (1.2.11)

The solution set of SEP (1.2.10)-(1.2.11) is denoted by $\Omega = \{z \in EP(F_1) : \mathcal{T}z \in EP(F_2)\}$. In this study, we propose a modified inertial forward-backward splitting algorithm with self-adaptive step size for approximating the common solution of the V_qIP (1.2.9) and SEP (1.2.10)-(1.2.11) in Hilbert spaces. That is, find a point $x^* \in \mathcal{C}$ such that

$$x^* \in \Gamma = (B+D)^{-1}(0) \cap \Omega.$$
 (1.2.12)

Next, we extend the study of the VIP (1.2.1) from the framework of Hilbert spaces to the framework of Banach spaces. We propose an inertial subgradient extragradient algorithm with Armijo type step-size for solving the VIP in a 2-uniformly convex real Banach space.

The $V_q IP$ was later generalized to the $SMV_q IP$ formulated as follows:

$$\begin{cases} \text{Find} \quad x^* \in \mathcal{H}_1, \quad \text{such that} \quad 0 \in B_1(x^*) + D_1(x^*), \\ \text{and} \\ y^* = \mathcal{T}x^* \in \mathcal{H}_2 \quad \text{such that} \quad 0 \in B_2(y^*) + D_2(y^*), \end{cases}$$
(1.2.13)

where $D_1: \mathcal{H}_1 \to 2^{\mathcal{H}_1}$ and $D_2: \mathcal{H}_2 \to 2^{\mathcal{H}_2}$ are multivalued maximal monotone mappings, $B_1: \mathcal{H}_1 \to \mathcal{H}_1, B_2: \mathcal{H}_1 \to \mathcal{H}_2$ are operators and $\mathcal{T}: \mathcal{H}_1 \to \mathcal{H}_2$ is a bounded linear operator. In this study, we propose a new inertial iterative algorithm with self-adaptive step size for approximating a common solution of finite family of SMV_qIPs (1.2.13) and FPP (1.1.1) of a nonexpansive mapping between a Banach space and Hilbert space. That is,

Find
$$x^* \in \mathcal{H}$$
 such that $0 \in \bigcap_{i=1}^{m} (B_i + D_i) x^*$, $Tx^* = x^*$; (1.2.14)

and

$$y^* = \mathcal{T}x^*$$
 such that $0 \in \bigcap_{i=1}^m C_i y^*$, (1.2.15)

where $B_i : \mathcal{H} \to \mathcal{H}$ is a finite family of α_i -inversely strongly monotone operators, $D_i : \mathcal{H} \to 2^{\mathcal{H}}$ and $C_i : \mathcal{X} \to 2^{\mathcal{X}^*}$ are finite families of maximal monotone operators for each $i = 1, 2, \ldots, m, T : \mathcal{H} \to \mathcal{H}$ is a nonexpansive mapping and $\mathcal{T} : \mathcal{H} \to \mathcal{X}$ is a bounded linear operator.

We denote the solution set of Problem (1.2.14)-(1.2.15) by

$$\Gamma := \bigcap_{i=1}^{m} (B_i + D_i)^{-1}(0) \cap \bigcap_{i=1}^{m} \mathcal{T}^{-1}(C_i^{-1}0) \cap F(T).$$

Finally, we consider the MP in the framework of Hadamard spaces.

Let X be a metric space and $f: X \to (-\infty, \infty]$ be a proper and convex function. One of the major and interesting problems in optimization is the problem of finding minimizers of f. That is, find $x \in X$ such that

$$f(x) = \min_{y \in X} f(y).$$
 (1.2.16)

We study two new proximal point methods. Using the first method, we find a common solution of a finite family of MPs and FPP for a finite family of quasi-pseudocontractive mappings and using the second method, we find a common zero of a finite family of multivalued monotone operators which is also a common fixed point of a finite family of quasi-pseudocontractive mappings in an Hadamard space. Furthermore, we propose a generalized viscosity implicit rule for finding a common solution of a finite family of MPs and FPP for a finite family of quasi-pseudocontractive mappings in an Hadamard space.

1.2.2 Motivation

The motivation for our study will be discussed under the following headings:

(a) Variational inequality and fixed point problems in real Hilbert spaces

The VIP (1.2.1) has received a lot of research attention due to its wide range of applications in diverse fields (see, for example [5, 14, 104, 110, 123, 130, 135, 149, 235] and the references therein). Several authors have proposed and studied iterative methods for solving the VIPs (see [5, 9, 105, 106, 135, 136, 220, 221] and other references therein). The problem of finding the common solution of a VIP and FPP is

motivated by the application it has in mathematical models whose constraints can be expressed as VIPs and FPPs. In particular, such a common solution problem has application in image recovery, signal processing, among others (see [126, 168] and other references therein). Recently, Nadeshkina and Takahashi [185] introduced an algorithm for finding a common solution of the FPP for a nonexpansive mapping and the VIP. They proved that the sequence generated by their proposed algorithm converges weakly to an element in the solution set of the problem under consideration. Since the strong convergence are more desirable than the weak convergence in infinite dimensional space, there is need to develop iterative methods that guarantees strong convergence. Also, since the pseudo-contractive mappings are more general than the nonexpansive mappings, we study the monotone VIP and FPP for a pseudo-contractive mapping in the real Hilbert space and introduce inertial iterative methods for approximating a common solution of the problem. We prove that our proposed method converges strongly to a point in the solution set. Considering the fact that the quasimonotone VIPs are more general than the monotone VIPs, we study the quasimonotone VIP in a real Hilbert space and introduce inertial and relaxed subgradient extragradient methods for solving the problem. We prove that the sequence generated by our proposed method converges weakly to the solution of the problem. Furthermore, we introduce an effective iterative method for finding a common solution of the quasimonotone VIP and FPP of quasi-pseudocontractive mappings in a real Hilbert space and prove that the sequence generated by our proposed method converge strongly to a point in the solution set of the aforementioned problem. Our results extend and generalize some existing results in the literature.

(b) Split variational inequality problems in real Hilbert spaces

The SVIP which can also be viewed as a combination of the classical VIP and the split feasibility problem (SFP) (introduced and studied by Censor *et al.* [57]) was introduced and studied by Censor *et al.* [61]. The first known results for SVIP is due to Censor *et al.* [63] (see also [61]). They studied the SVIP when the operators are monotone and Lipschitz continuous, by first transforming the problem into an equivalent Constrained VIP (CVIP) in the product space $\mathcal{H}_1 \times \mathcal{H}_2$ (see [63, Section 4) and then employed the well-known subgradient extragradient method to solve it. To improve this result, Tian and Jiang [243] introduced a new class of SVIP which generalizes the class of SVIP considered by Censor *et al.* [63]. They proposed an algorithm for approximating a solution of the new class of SVIP and also proved the weak convergence of this algorithm when the mapping is nonexpansive and the cost operator is monotone and Lipschitz continuous. Motivated by the works of Censor *et al.* [63], Tian and Jiang [243] and other works in this direction, we propose two new relaxed inertial methods for solving the SVIP without any product space transformation where the underlying operator is monotone and Lipschitz continuous. Furthermore, we present two new methods with inertial steps for solving the SVIP in real Hilbert spaces without any product space formulation when the cost operators are pseudomonotone and Lipschitz continuous. We prove that the sequences generated by these methods converge strongly to a minimum-norm solution of the problem. We further propose two new inertial projection and contraction methods for solving the SVIPs in real Hilbert spaces without the co-coercive condition and without the product space formulation. Also, we propose a viscosity-type iterative method and prove its strong convergence to a solution of the GVIP in real Hilbert spaces. Our results extend and generalize some existing results in the literature.

(c) Split equalities variational inclusion, split equilibrium and fixed point problems in real Hilbert spaces

The problem of finding a common solution of the split equalities of EP, VIP and FPP of nonexpansive semigroups has applications to many nonlinear problems. Recently, Latif and Eslamian [159] studied and introduced a new algorithm for finding a common element of split equalities of EP, monotone VIP with Lipschitz operator and fixed point problem of nonexpansive semigroups satisfying the uniformly asymptotically regularity (u.a.r) condition in Hilbert spaces. Motivated by this result and other results in the literature, we propose and study an inertial Tseng's extragradient algorithm for the SE_qP for finding a common element of solution of the EP, non-Lipschitz pseudomonotone VIP and common fixed point of nonexpansive semigroups without the u.a.r condition. The problem of finding a common solution of the monotone V_q IP and SEP has possible applications to many nonlinear problems and mathematical models whose constraints can be expressed as V_q IPs and SEPs. Very recently, Cholamjiak et al. [81] proposed a modified inertial forward-backward splitting method for solving the SEP and the inverse strongly monotone V_q IP. Motivated by this, we propose a modified inertial forward-backward splitting algorithm with self-adaptive step sizes for approximating the solution of the aforementioned problem in Hilbert spaces without the inverse strongly monotone condition. Our result extend and generalize some existing results in the literature.

(d) Variational inequality, variational inclusion and fixed point problems in Banach spaces

It is well-known that many real life problems are generally defined in Banach spaces than Hilbert spaces, therefore there is need to extend the study of these optimization problems from Hilbert spaces to Banach spaces. Recently, Tan *et al.* [240] proposed an inertial iterative method for solving the VIP (1.2.1) in a real Hilbert space. Motivated by the work of Tan *et al.* [240], we propose and study an efficient method for solving the VIP (1.2.1) in a 2-uniformly convex real Banach space. The problem of finding the common solution of the SMV_q IP and FPP has possible applications to many nonlinear problems and mathematical models whose constraints can be expressed as SMV_q IPs and FPPs. Motivated by this and the ongoing research activities in this direction, we propose and study a new inertial iterative algorithm with self-adaptive step size for approximating a common solution of finite family of SMV_q IP and FPP for a nonexpansive mapping between a Banach space and Hilbert space.

(e) Minimization and fixed point problems in Hadamard spaces

The Hadamard space is an analogue of the Hilbert space with related geometry in the setting of the nonlinear space. Many results related to optimization problems have more applications in Hadamard spaces than in Hilbert spaces. This is due to the fact that the classical results on optimization and FPP in Hadamard spaces occur naturally in Hilbert spaces but the converse may not be true. This motivates the need to extend the results on optimization and FPP from the Hilbert and Banach spaces to Hadamard spaces. Just recently, Chang *et al.* [70] proposed an iterative algorithm for approximating a common solution of a finite family of MP and FPP for two demicontractive mappings in Hadamard spaces. Motivated by this, we propose two new proximal point methods involving quasi-pseudocontractive mappings in Hadamard spaces. Inspired by the viscosity technique, we propose a generalized viscosity implicit rule for approximating the common solution of a finite family of MPs and FPP for a finite family of quasi-pseudocontractive mapping in Hadamard spaces.

1.3 Objectives

The objectives we aim to achieve at the end of this study are:

- (i) to introduce and study iterative algorithms for approximating the common solutions of VIPs and FPPs in Hilbert spaces;
- (ii) to introduce efficient inertial and relaxed algorithms for approximating the solution of the VIP without any form of monotonicity in a real Hilbert space;
- (iii) to introduce iterative algorithms for approximating the solution of the SVIP without any product space formation in real Hilbert spaces;
- (iv) to introduce an efficient algorithm for approximating the solution of the GVIP in a real Hilbert space;
- (v) to introduce an effective iterative algorithm for approximating the common solution of the split equalities of VIP, EP and FPP in real Hilbert spaces;
- (vi) to introduce an iterative algorithm with self-adaptive step-size for approximating the common solution of the V_q IPs and SEPs in real Hilbert spaces;
- (vii) to generalize the study of the VIPs from the framework of Hilbert space to the framework of Banach space and also introduce an iterative algorithm for approximating the solution of the VIPs in Banach spaces;
- (viii) to introduce an iterative algorithm with self-adaptive step-size for approximating a common solution of finite family of SMV_qIPs and FPPs of a nonexpansive mapping between a Banach space and Hilbert space;
- (ix) to introduce the concepts of MPs and FPPs in Hadamard spaces;
- (x) to apply the obtained results to solve some optimization problems;
- (xi) to give some numerical experiments to illustrate the applicability of our proposed methods and also compare them with some existing methods in the literature.

1.4 Organization of the thesis

The thesis is organized as follows;

Chapter 1 (General Introduction): In this chapter, we give a brief background study of our research interest. We discuss the research problems and motivations. Finally, we state the objectives of the study and give a detailed organization of the thesis.

Chapter 2 (Preliminaries and Literature Review): In this chapter, we recall some basic definitions, concepts, terms and results that will be needed in our study. Also, we give a detailed literature review of some important past and recent works that are related to our study.

Chapter 3 (Results on Variational Inequality and Fixed Point Problems in real Hilbert Spaces): The main results of this thesis starts in this chapter. The chapter consists of four (4) sections.

In Section 3.1, we give a brief introduction of the study of the chapter.

In Section 3.2, we propose and study new inertial Tseng's extragradient algorithms with self-adaptive step size to solve the monotone VIP and FPP in a real Hilbert space. We prove that the sequences generated by our proposed algorithms converge strongly to the solution set. We apply our result to solve the zero point problem (ZPP) in Hilbert spaces. Finally, we present numerical experiments to illustrate the applicability of our proposed methods and also compare them with some existing methods in the literature.

In Section 3.3, we introduce two new relaxed inertial subgradient extragradient methods for solving quasimonotone VIPs in a real Hilbert space. We prove that the sequences generated by our proposed methods converge weakly to a point in the solution set. Finally, we present numerical experiments to illustrate the applicability of our proposed methods and also compare them with some existing methods in the literature.

In Section 3.4, we study the quasimonotone VIP with constraint of FPP of a quasipseudocontractive mapping. We propose and study a new inertial Tseng's extragradient method with self-adaptive step size for approximating the minimum-norm solutions of aforementioned problem. Furthermore, we present some numerical experiments for our proposed method and compare it with other existing methods in the literature. Finally, we apply our result to image restoration problem.

Chapter 4 (Results on Split Variational Inequality in real Hilbert Spaces): This chapter consists of four (4) sections.

In Section 4.1, we give a brief introduction of the study of the chapter.

In Section 4.2, we propose two new relaxed inertial methods for solving the SVIP in real Hilbert spaces without any product space transformation when the cost operator is not co-coercive. We prove that the sequences generated by our proposed algorithms converge strongly to a point in the solution set. Finally, we present numerical experiments to illustrate the applicability of our proposed methods and also compare them with some existing methods in the literature.

In Section 4.3, we propose two new methods with inertial steps for solving the SVIP beyond monotonicity in real Hilbert spaces without any product space formulation. We prove that the sequences generated by these methods converge strongly to a minimum-norm solution of the aforementioned problem. Finally, we provide several numerical experiments of the

proposed methods in comparison with other related methods in literature.

In Section 4.4, we propose two new inertial projection and contraction methods for solving the SVIP in real Hilbert spaces. The sequences generated by these methods converge strongly to a point in the solution set of the SVIP in real Hilbert spaces. Furthermore, we present several numerical experiments for the proposed methods and compare their performances with other related methods in the literature.

In Section 4.5, we propose a viscosity-type method for solving a GVIP when the cost operator is pseudomonotone in a real Hilbert space. We prove that the sequence generated by our proposed algorithm converges strongly to the solution set of the GVIP. Finally, we present numerical experiments to illustrate the applicability of our proposed method and also compare it with some existing methods in the literature.

Chapter 5 (Results on Split equalities Variational Inclusion and Split Equilibrium Problems in real Hilbert Spaces): This chapter consists of three (3) sections. In Section 5.1, we give a brief introduction of the study of the chapter.

In Section 5.2, we propose an inertial Tseng's extragradient algorithm with self adaptive step size for approximating a common solution of the split equalities of EP, non-Lipschitz pseudomonotone VIP and FPP of nonexpansive semigroups in real Hilbert spaces. We prove that the sequence generated by our proposed method converges strongly to a point in the solution set of the problem. Finally, we provide some numerical experiments for the proposed method in comparison with related methods in literature.

In Section, 5.3, we propose a modified inertial forward-backward splitting algorithm with self-adaptive step size for approximating a solution of the V_q IP and SEP in real Hilbert spaces. We prove that the sequence generated by our proposed method converges strongly to a minimum-norm solution of the aforementioned problem. Furthermore, we apply our result to study certain optimization problems. Finally, we provide some numerical experiments of our proposed method in comparison with other existing methods in the literature.

Chapter 6 (Results on Variational Inequality, Variational Inclusion and Fixed Point Problems in real Banach Spaces): This chapter consists of three (3) sections. In Section 6.1, we give a brief introduction of the study of this chapter.

In Section 6.2, we study an inertial subgradient extragraident algorithm with Armijo-type step size for solving the VIP in a real Banach space. We prove that the sequence generated by our proposed method converges strongly to a solution of the VIP in a real Banach space. Finally, we provide some numerical experiments of the proposed method in comparison with other existing method in literature.

In Section 6.3, we propose and study a new inertial iterative algorithm with self-adaptive step size for approximating a common solution of finite family of SMV_qIP and FPP between a Hilbert and Banach space. Furthermore, we apply our result to study some optimization problems. Finally, we provide some numerical experiments to demonstrate the efficiency of our method in comparison with some well-known methods in the literature.

Chapter 7 (Results on Minimization Problems in Hadamard Spaces): This chapter consists of four (4) sections.

In Section 7.1, we give a brief introduction of the study of the chapter.

In Section 7.2, we prove some lemmas that will be required to establish our main result in this chapter.

In Section 7.3, we propose two new proximal point methods involving quasi-pseudocontractive mappings in an Hadamard space. We prove that the first method converges strongly to a common solution of a finite family of MP and FPP for a finite family of quasi-pseudocontractive mappings in an Hadamard space. We extend this method to monotone inclusion problems (MIPs) and prove that it converges strongly to a common zero of a finite family of multivalued monotone operator in an Hadamard space. Furthermore, we provide various nontrival numerical implementations of our methods in Hadamard spaces and compare them with some other recent methods in the literature.

In Section 7.4, we propose a generalized viscosity implicit rule involving quasi-pseudocontractive mappings in Hadamard spaces. We obtain a strong convergence result of our algorithm to the solution set.

Chapter 8 (Conclusion, Contributions to Knowledge and Future Research): In this chapter, we give a conclusion of our study and also highlight our contributions to existing knowledge. Furthermore, we discuss possible problems for future research.

CHAPTER 2

Preliminaries and Literature Review

In this section, we provide some basic definitions, terms, notations and concepts that will be relevant throughout our study. We also give a comprehensive literature review of some important past and recent works on optimization and fixed point problem. Furthermore, we state some important lemmas and results that will be required in establishing the proofs of our main results.

2.1 Geometric properties of Hilbert spaces

The Hilbert space which is an extension of the concept of Euclidean spaces to infinite dimensional spaces is known to have the most simplest and clearly discernible geometric properties. Let \mathcal{C} be a nonempty, closed and convex subset of a real Hilbert space \mathcal{H} with norm $\|\cdot\|$ and inner product $\langle \cdot, \cdot \rangle$. We will denote the weak and strong convergence of a sequence $\{x_n\}$ to a point x by $x_n \rightharpoonup x$ and $x_n \rightarrow x$, respectively. Also, we denote the set of weak limits of $\{x_n\}$ by $w_{\omega}(x_n)$, that is

$$w_{\omega}(x_n) := \{ x \in \mathcal{H} : x_{n_i} \rightharpoonup x \text{ for some subsequence } \{x_{n_i}\} \text{ of } \{x_n\} \}.$$

Some of the geometric properties that characterize Hilbert spaces include: the availability of the inner product, the fact that the nearest point map of a real Hilbert space \mathcal{H} onto a closed convex subset \mathcal{C} of \mathcal{H} is Lipschitzian with constant 1 and the following identities:

$$\|x+y\|^{2} = \|x\|^{2} + \|y\|^{2} + 2\langle x, y \rangle, \qquad (2.1.1)$$

$$\|\lambda x + (1-\lambda)y\|^2 = \lambda \|x\|^2 + (1-\lambda)\|y\|^2 - \lambda(1-\lambda)\|x-y\|^2, \qquad (2.1.2)$$

which hold for $x, y \in \mathcal{H}$ and $\lambda \in [0, 1]$. These geometric characteristics of Hilbert spaces makes certain problems posed in Hilbert spaces more manageable than those in general Banach spaces [74].

2.1.1 Some inequalities that characterize Hilbert spaces

Lemma 2.1.1. [198, 201] Let \mathcal{H} be a real Hilbert space, then the following assertions hold:

(1)
$$2\langle x, y \rangle = ||x||^2 + ||y||^2 - ||x - y||^2 = ||x + y||^2 - ||x||^2 - ||y||^2, \quad \forall x, y \in \mathcal{H};$$

(2)
$$\|\alpha x + (1-\alpha)y\|^2 = \alpha \|x\|^2 + (1-\alpha)\|y\|^2 - \alpha(1-\alpha)\|x-y\|^2, \quad \forall x, y \in \mathcal{H}, \ \alpha \in \mathbb{R};$$

- (3) $||x-y||^2 \le ||x||^2 + 2\langle y, x-y \rangle, \quad \forall x, y \in \mathcal{H}.$
- (4) $||x+y||^2 \le ||x||^2 + 2\langle y, x+y \rangle, \forall x, y \in \mathcal{H}.$

Lemma 2.1.2. [154] For $x, y, v, z \in \mathcal{H}$, we have

$$2\langle x-y,v-z\rangle = \|x-z\|^2 + \|y-v\|^2 - \|x-v\|^2 - \|y-z\|^2$$

Also,

$$\begin{split} \|x - y + v - z\|^2 &= \|x - y\|^2 + \|v - z\|^2 + 2\langle x - y, v - z\rangle \\ &= \|x - y\|^2 + \|v - z\|^2 + \|x - z\|^2 + \|y - v\|^2 - \|x - v\|^2 - \|y - z\|^2. \end{split}$$

2.1.2 Some nonlinear single-valued mappings in Hilbert spaces

Definition 2.1.3. Let \mathcal{H} be a real Hilbert space and \mathcal{C} be a nonempty closed and convex subset of \mathcal{H} . A mapping $T : \mathcal{H} \to \mathcal{H}$ is said to be

(i) L-Lipschitz if there exists L > 0 such that

$$||Tx - Ty|| \le L||x - y||, \quad \forall \quad x, y \in \mathcal{H};$$

if L = 1, then T is called nonexpansive while T is called a contraction if $L \in [0, 1)$,

(ii) uniformly continuous, if for every $\epsilon > 0$, there exists $\delta = \delta(\epsilon) > 0$, such that

$$||Tx - Ty|| < \epsilon \quad whenever \quad ||x - y|| < \delta, \quad \forall x, y \in \mathcal{H};$$

(iii) quasi-nonexpansive, if $F(T) \neq \emptyset$ and

$$||Tx - y|| \le ||x - y||, \quad \forall x \in \mathcal{H}, \quad and \quad y \in F(T);$$

(iv) firmly nonexpansive, if

$$||Tx - Ty||^2 \le ||x - y||^2 - ||(I - T)x - (I - T)y||^2, \quad \forall x, y \in \mathcal{H};$$

(v) β -demicontractive with $0 \leq \beta < 1$, if $F(T) \neq \emptyset$ and

$$||Tx - y||^2 \le ||x - y||^2 + \beta ||(I - T)x||^2, \quad \forall \quad x \in \mathcal{H}, \quad y \in F(T);$$

(vi) k-strictly pseudocontractive, if for $0 \le k < 1$

$$||Tx - Ty||^{2} \le ||x - y||^{2} + k||(I - T)x - (I - T)y||^{2} \quad \forall \ x, y \in \mathcal{H}.$$
 (2.1.3)

(vii) quasi-pseudocontractive, if $F(T) \neq \emptyset$ and

$$||Tx - y||^2 \le ||x - y||^2 + ||Tx - x||^2, \quad \forall \ x \in \mathcal{H} \quad and \quad x \in F(T);$$

(viii) η -strongly monotone, if there exists $\eta > 0$ such that

$$\langle Tx - Ty, x - y \rangle \ge \eta \|x - y\|^2 \quad \forall \ x, y \in \mathcal{H};$$

(ix) η -inverse strongly monotone (η -ism), if there exists $\eta > 0$ such that

$$\langle Tx - Ty, x - y \rangle \ge \eta \|Tx - Ty\|^2 \quad \forall \ x, y \in \mathcal{H};$$

if $\eta = 1$, then T is called firmly nonexpansive,

(x) monotone, if

$$\langle Tx - Ty, x - y \rangle \ge 0 \quad \forall \ x, y \in \mathcal{H};$$

(xi) pseudo-monotone, if

$$\langle Tx, y - x \rangle \ge 0 \implies \langle Ty, y - x \rangle \ge 0 \quad \forall \ x, y \in \mathcal{H};$$

(xii) quasimonotone, if

$$\langle Ty, x - y \rangle > 0 \implies \langle Tx, x - y \rangle \ge 0, \ \forall x, y \in \mathcal{H},$$

- (xiii) sequentially weakly continuous on \mathcal{H} , if for each sequence $\{x_n\} \subset \mathcal{H}$ we have that $\{x_n\}$ converges weakly to $x \in \mathcal{H}$ implies that $\{Tx_n\}$ converges weakly to Tx;
- (xiv) sequentially weakly-strongly continuous, if for every sequence $\{x_n\}$ that converges weakly to a point x, the sequence $\{Tx_n\}$ converges strongly to Tx,
- (xv) α -averaged, if $f = (1-\alpha)I + \alpha T$, where $\alpha \in (0,1)$ and $T : \mathcal{H} \to \mathcal{H}$ is nonexpansive,

From the above definition, we can deduce that firmly nonexpansive mappings are $\frac{1}{2}$ -averaged while averaged mappings are nonexpansive. This implies that every firmly nonexpansive mapping is nonexpansive. Clearly, the class of quasi-pseudocontractive mappings is more general and properly contains all the other classes of mappings defined in (iii) – (vi) with nonempty fixed points set. It is also known that every η -inverse strongly monotone mapping is $\frac{1}{\eta}$ -Lipschitz continuous. Also, if T is η -strongly monotone and L-Lipschitz continuous, then T is η/L^2 -ism. Furthermore, both η -strongly monotone and η -inverse strongly monotone mappings are monotone while monotone mappings are pseudo-monotone. We also observe that monotone implies pseudo-monotone which implies quasi-monotone but the converses are not always true. Also, observe that uniform continuity is a weaker notion than Lipschitz continuity. **Remark 2.1.4.** It is well known that if D is a convex subset of \mathcal{H} , then $T: D \to \mathcal{H}$ is uniformly continuous if and only if, for every $\epsilon > 0$, there exists a constant $M < \infty$ such that

$$||Tx - Ty|| \le M ||x - y|| + \epsilon \quad \forall x, y \in D.$$

$$(2.1.4)$$

Example 2.1.5. Let $\mathcal{H} = l_2(\mathbb{R})$, where $l_2(\mathbb{R}) := \{x = (x_1, x_2, x_3, \dots), x_i \in \mathbb{R} : \sum_{i=1}^{\infty} |x_i|^2 < \infty\}$. Then, the operator $A : \mathcal{H} \to \mathcal{H}$ defined by

$$A(x_1, x_2, x_3, \dots) = (x_1 e^{-x_1^2}, 0, 0, \dots)$$

is pseudomonotone, Lipschitz continuous and sequentially weakly continuous but not monotone.

Example 2.1.6. [145] Let \mathbb{R} denote the real number with the usual norm and $\mathcal{C} = [0, 2]$. Let $T : \mathcal{C} \to \mathcal{C}$ be defined by

$$Tx = \frac{1}{3}\left(x^2 + 2\right)$$

for all $x \in C$, where $x + 2 \leq 3$, $x \in [0, 1]$ and x + 2 = 3, x > 1. Clearly $F(T) = \{1, 2\}$. Then T is a quasi-nonexpansive mapping but not a nonexpansive mapping.

Example 2.1.7. Let X be a real line with the usual norm and $\mathbb{C} = \mathbb{R}$. Define $T : \mathcal{C} \to \mathcal{C}$ by

$$Tx = 7x$$

T is k-strictly pseudo-contractive but not quasi-nonexpansive or nonexpansive.

Proof. It is clear that $F(T) = \{0\}$, thus for $x \in \mathbb{R}$, we have

$$|7x - 0|^2 = 49|x - 0|^2 > |x - 0|^2$$

which implies that T is not quasi-nonexpansive and hence, not nonexpansive.

Next we show that T is k- strictly pseudo-contractive.

$$|Tx - Ty|^2 = |7x - 7y|^2 = 49|x - y|^2.$$

Also,

$$|x - y - (Tx - Ty)|^{2} = |x - y - (7x - 7y)|^{2}$$

= 36|x - y|².

$$|Tx - Ty|^{2} = |x - y|^{2} + 48|x - y|^{2}$$
$$= |x - y|^{2} + \frac{4}{3}|x - y - (Tx - Ty)|^{2}.$$

Hence, T is $\frac{4}{3}$ -strictly pseudo-contractive mapping.

Lemma 2.1.8. [276] Let \mathcal{H} be a real Hilbert space and $S : \mathcal{H} \to \mathcal{H}$ be a κ -strictly pseudocontractive mapping with $\kappa \in [0, 1)$. Let $S_{\beta} := \beta I + (1 - \beta)S$, where $\beta \in [\kappa, 1)$, then

(i) $F(S) = F(S_{\beta}),$

(ii) S_{β} is a nonexpansive mapping.

Lemma 2.1.9. [238] Let \mathcal{H}_1 and \mathcal{H}_2 be real Hilbert spaces. Let $T : \mathcal{H}_1 \to \mathcal{H}_2$ be a bounded linear operator with $T \neq 0$, and $S : \mathcal{H}_2 \to \mathcal{H}_2$ be a nonexpansive mapping. Then $T^*(I-S)T$ is $\frac{1}{2||\mathcal{A}||^2}$ -inverse strongly monotone.

Lemma 2.1.10. [261] Let \mathcal{H} be a real Hilbert space and $A : \mathcal{H} \to \mathcal{H}$ be a nonlinear mapping, then the following hold.

- (i) A is η -inverse strongly monotone and $\gamma > 0$, then γA is $\frac{\eta}{\gamma}$ -inverse strongly monotone.
- (ii) A is averaged if and only if the complement I A is η -inverse strongly monotone for some $\eta > \frac{1}{2}$. Indeed, for $\beta \in (0, 1)$, A is β -averaged if and only if I A is $\frac{1}{2\beta}$ -inverse strongly monotone.

Definition 2.1.11. A one-parameter family mapping $\mathcal{T} = \{T(s) : 0 \leq s < +\infty\}$ from \mathcal{H}_1 into itself is said to be a nonexpansive semigroup if it satisfies the following conditions:

(i) $T(0)x = x, \forall x \in \mathcal{H}_1;$

(ii)
$$T(s+u) = T(s)T(u)$$
, for all $s, u \ge 0$;

- (iii) for each $x \in \mathcal{H}_1$, the mapping T(s)x is continuous;
- (*iv*) $||T(s)x T(s)y|| \le ||x y||$, for all $x, y \in \mathcal{H}_1$ and $s \ge 0$.

We denote the common fixed point set of the semigroup \mathcal{T} by $F(\mathcal{T}) = \{x \in \mathcal{C} : T(s)x = x, \forall s \geq 0\}$. It is well known that $F(\mathcal{T})$ is closed and convex [47].

Lemma 2.1.12. [240] Let C be a nonempty bounded closed and convex subset of a real Hilbert space \mathcal{H} . Let $\mathcal{T} = \{T(s) : s \ge 0\}$ from C be a nonexpansive semigroup on C; then for all $h \ge 0$,

$$\lim_{t \to \infty, \ x \in \mathcal{C}} \left\| \frac{1}{t} \int_0^t T(s)x - T(h) \left(\frac{1}{t} \int_0^t T(s)x dx \right) \right\| = 0.$$

Lemma 2.1.13. [240] Let C be a nonempty bounded closed and convex subset of a real Hilbert space \mathcal{H} . Let $\{x_n\}$ be a sequence and let $\mathcal{T} = \{T(s) : s \ge 0\}$ from C be a nonexpansive semigroup on C, if the following conditions are satisfied

- (i) $x_n \rightharpoonup x;$
- (*ii*) $\limsup_{s \to \infty} \limsup_{n \to \infty} \left\| T(s) x_n x_n \right\| = 0,$

then, $x \in F(\mathcal{T})$.

Lemma 2.1.14. [112] Let \mathcal{H} be a real Hilbert space and let $S : \mathcal{H} \to \mathcal{H}$ be a nonexpansive mapping with $F(S) \neq \emptyset$. Then, the mapping I - S is demiclosed at zero, that is, for any sequence $\{x_n\}$ in \mathcal{H} such that $x_n \rightharpoonup x \in \mathcal{H}$ and $||x_n - Sx_n|| \to 0$ implies $x \in F(S)$.

Lemma 2.1.15. [199] Each Hilbert space \mathcal{H} satisfies the Opial condition, that is, for any sequence $\{x_n\}$ with $x_n \rightharpoonup x$, the inequality $\liminf_{n \to \infty} ||x_n - x|| < \liminf_{n \to \infty} ||x_n - y||$ holds for every $y \in \mathcal{H}$ with $y \neq x$.

2.2 Geometric properties of Banach spaces

In this section, we discuss some of the geometric properties of Banach spaces that will be required in this study.

Let \mathcal{X} be a real Banach space and \mathcal{C} a nonempty, closed and convex subset of \mathcal{X} . We denote by \mathcal{X}^* the dual of \mathcal{X} and $\|\cdot\|$ the norm of \mathcal{X} or \mathcal{X}^* .

Definition 2.2.1. A Banach space \mathcal{X} is said to be strictly convex if for all $x, y \in \mathcal{X}, x \neq y$, ||x|| = ||y|| = 1, we have $||\lambda x + (1 - \lambda)y|| < 1$, $\forall \lambda \in (0, 1)$.

Definition 2.2.2. Let $S(\mathcal{X}) := \{x \in \mathcal{X} : ||x|| = 1\}$. A Banach space \mathcal{X} is said to be smooth if

$$\lim_{\sigma \to 0} \frac{\|x + \sigma y\| - \|x\|}{\sigma}$$

exists for $x, y \in S(\mathcal{X})$.

Definition 2.2.3. Let E be a normed space with dim $E \ge 2$. The modulus of convexity $\delta_E : (0,2] \rightarrow [0,1]$ is defined by

$$\delta_E(\epsilon) = \inf\left\{1 - \frac{\|x+y\|}{2} : \|x\| = 1 = \|y\|, \|x-y\| = \epsilon\right\}.$$

Definition 2.2.4. Let E be a normed space with dim $E \ge 2$. The modulus of smoothness $\rho_{\mathcal{X}}(\sigma) : [0, \infty) \to [0, \infty)$ is defined by

$$\rho_{\mathcal{X}}(\sigma) = \sup\left\{\frac{\|x + \sigma y\| + \|x - \sigma y\|}{2} - 1 : \|x\| = \|y\| = 1\right\}.$$

Let $1 < q \leq 2 \leq p < \infty$ with $\frac{1}{p} + \frac{1}{q} = 1$. A Banach space \mathcal{X} is called uniformly convex if $\delta_{\mathcal{X}}(\epsilon) > 0$ for any $\epsilon \in (0, 2]$, and *p*-uniformly convex if there exists a constant $c_p > 0$ such that $\delta_{\mathcal{X}}(\epsilon) > c_p \epsilon^p$ for any $\epsilon \in (0, 2]$. \mathcal{X} is called *uniformly smooth* if $\lim_{\sigma \to 0} \left(\frac{\rho_{\mathcal{X}}(\sigma)}{\sigma}\right) = 0$ and *q*-uniformly smooth if there is a constant q > 0 such that $\rho_{\mathcal{X}}(\sigma) \leq q\sigma^2$ for some $\sigma > 0$.

Theorem 2.2.5. [74] Every uniformly smooth space is reflexive.

Lemma 2.2.6. [37] The space \mathcal{X} is 2-uniformly convex if and only if there exists $\mu_{\mathcal{X}} \geq 1$ such that

$$\frac{\|u+v\|^2 + \|u-v\|^2}{2} \ge \|u\|^2 + \|\mu_{\mathcal{X}}^{-1}v\|^2, \ \forall \ u, v \in \mathcal{X}.$$
(2.2.1)

The minimum value of the set of all $\mu_{\mathcal{X}} \geq 1$ satisfying (2.2.1) $\forall u, v \in \mathcal{X}$ is denoted by μ and is called the 2-uniform convexity constant of \mathcal{X} . It is easy to see that $\mu = 1$ whenever \mathcal{X} is a Hilbert space. We denote the minimum value of the set of all $\mu_{\mathcal{X}} \geq 1$ by μ such that (2.2.1) holds for any $u, v \in \mathcal{X}$.

Lemma 2.2.7. [262] Let \mathcal{X} be a 2-uniformly smooth Banach space with the best smoothness constant $\mu > 0$. Then, the following inequality holds:

$$||x + y||^2 \le ||x||^2 + 2\langle y, Jx \rangle + 2||\mu y||^2, \quad \forall x, y \in \mathcal{X}.$$

Remark 2.2.8. In Hilbert spaces the best smoothness constant is $\mu = \frac{1}{\sqrt{2}}$.

Remark 2.2.9. Every 2-uniformly convex Banach space is uniformly convex and every Hilbert space is uniformly smooth and 2-uniformly convex.

Definition 2.2.10. The generalized duality mapping $J_p^{\mathcal{X}} : \mathcal{X} \to 2^{\mathcal{X}^*}$ is defined by

$$J_p^{\mathcal{X}}(x) = \{ x^* \in \mathcal{X}^* : \langle x, x^* \rangle = \|x\|^p, \|x^*\| = \|x\|^{p-1}, \quad \forall x \in \mathcal{X} \}.$$

In particular, when p = 2 the duality map $J = J_2^{\mathcal{X}}$ is called the normalized duality map. The Hilbert spaces, $L_p(\text{or } \ell_p)$ spaces and the Sobolev spaces $W^{k,p}$, (1 are $uniformly convex and uniformly smooth Banach spaces. Furthermore, the <math>L_p, \ell_p$ and $W^{k,p}$ spaces are *p*-uniformly smooth if 1 and 2-uniformly smooth if <math>p > 2. If $\mathcal{X} = \mathcal{H}$, where \mathcal{H} is a real Hilbert space, we have J = I.

Now, we state the relationship between uniformly convex and uniformly smooth space.

Theorem 2.2.11. [74] Let \mathcal{X} be a Banach space;

- (i) \mathcal{X} is uniformly smooth if and only if \mathcal{X}^* is uniformly convex.
- (ii) \mathcal{X} is uniformly convex if and only if \mathcal{X}^* is uniformly smooth.

We give an example of a generalized duality mapping in Banach spaces.

Example 2.2.12. [3] Let $\mathcal{X} := L_p([\alpha, \beta])$ $(1 , where <math>\alpha, \beta \in \mathbb{R}$ and let $f \in \mathcal{X}$. Then the generalized duality mapping $J_p^{\mathcal{X}}$ is given by

$$J_p^{\mathcal{X}}(f)(t) = |f(t)|^{p-1} \operatorname{sgn}(f(t)).$$

The normalized duality mapping is known to have the following properties [85]:

(i) If \mathcal{X} is smooth, then J is single valued denoted by j.

- (ii) Jx is nonempty closed, convex and bounded sub set of \mathcal{X}^* for all $x \in \mathcal{X}$.
- (iii) If \mathcal{X} is reflexive, then J is surjective.
- (iv) If \mathcal{X} is strictly convex, then J is one-to-one and strictly monotone, ie $\langle x y, Jx Jy \rangle \ge 0, \forall x, y \in \mathcal{X}$ such that $x \neq y$.
- (v) If \mathcal{X} is reflexive and strictly convex, then J^{-1} is norm-to-weak continuous.
- (vi) If \mathcal{X} is uniformly smooth and uniformly convex, then J is norm-to-norm continuous on bounded subset of \mathcal{X} and $J^{-1} = J^*$. Also the normalized duality mapping on \mathcal{X}^* is also uniformly norm-to-norm continuous on bounded subsets of \mathcal{X}^* .
- (vii) If \mathcal{X}^* is uniformly convex, then J is single-valued, one-to-one and uniformly continuous on bounded subsets of \mathcal{X} .

If \mathcal{X} is smooth, $J : \mathcal{X} \to \mathcal{X}^*$ is said to be weak-to-weak continuous if for every $y \in \mathcal{X}, \langle y, Jx_n \rangle \to \langle y, Jx \rangle$ as $x_n \rightharpoonup x$. It is known that the $l_p(p > 1)$ space has the weak-to-weak continuous property, but the $L_p(p > 2)$ space does not possess this property.

The Lyapunov functional $\phi : \mathcal{X} \times \mathcal{X} \to \mathbb{R}$ [16] is defined by

$$\phi(u,v) := \|u\|^2 - 2\langle u, Jv \rangle + \|v\|^2, \ \forall \ u, v \in \mathcal{X}.$$

We can easily see that

$$\phi(u, v) \ge \left(\|u\| - \|v\| \right)^2 \ge 0.$$

Moreover, we observe that $\phi(u, u) = 0$ for each $u \in \mathcal{X}$. If \mathcal{X} is strictly convex, then $\phi(u, v) = 0 \iff u = v$. If \mathcal{X} is a Hilbert space, then $\phi(u, v) = ||u - v||^2$ for all $u, v \in \mathcal{X}$.

2.2.1 Some inequalities that characterize Banach spaces

Lemma 2.2.13. [16, 20] Suppose that \mathcal{X} is a real, uniformly convex, smooth Banach space. Then, the Lyapunov functional satisfies the following properties for all $x, y, z \in \mathcal{X}$ and $\alpha \in (0, 1)$:

(1) $\phi(x,y) = \phi(x,z) + \phi(z,y) + 2\langle x-z, Jz - Jy \rangle;$

(2)
$$\phi(x,y) + \phi(y,x) = 2\langle x-y, Jx - Jy \rangle;$$

(3)
$$\phi(x, y) = ||x|| ||Jx - Jy|| + ||y|| ||x - y||;$$

(4) $0 \le (||x|| - ||y||)^2 \le \phi(x, y) \le (||x|| + ||y||)^2;$

(5)
$$\phi(x, J^{-1}(\alpha Jz + (1 - \alpha)Jy)) \le \alpha \phi(x, z) + (1 - \alpha)\phi(x, y).$$

Alber [16] also studied the following functional $V: \mathcal{X} \times \mathcal{X}^* \to \mathbb{R}$ defined by

$$V(x, x^*) = ||x||^2 - 2\langle x, x^* \rangle + ||x^*||^2$$

for all $x \in \mathcal{X}$ and $x^* \in \mathcal{X}^*$. It is easy to see from the definition of the Lyapunov functional that

$$V(x, x^*) = \phi(x, J^{-1}x^*),$$

for all $x \in \mathcal{X}$ and $x^* \in \mathcal{X}^*$. For each $x \in \mathcal{X}$, the mapping $f : \mathcal{X}^* \to \mathcal{R}$ defined by $f(x^*) = V(x, x^*)$ for all $x^* \in \mathcal{X}^*$ is a continuous and convex function. Moreover, the functional V is known to satisfy the following inequality [15]:

$$V(x,x^*) + 2\langle J^{-1}x^* - x, y^* \rangle \le V(x,x^* + y^*), \ \forall \ x \in \mathcal{X}, \ x^*, y^* \in \mathcal{X}^*.$$
(2.2.2)

2.2.2 Some nonlinear single-valued mappings in Banach spaces

Definition 2.2.14. Let \mathcal{X} be a real Banach space and \mathcal{C} be a nonempty closed and convex subset of \mathcal{X} . A mapping $T : \mathcal{X} \to \mathcal{X}$ is said to be

- (i) closed if for any sequence $\{x_n\} \subset \mathcal{C}$ with $x_n \to x$ and $Tx_n \to y$, then Tx = y;
- (ii) semi-compact if for any bounded sequence $\{x_n\}$ in C with $x_n Tx_n \to 0$, there exists a subsequence $\{x_{n_k}\}$ of $\{x_n\}$ which converges strongly to some $x \in \mathcal{X}$;
- (iii) ϕ -nonexpansive, if

$$\phi(Tx, Ty) \le \phi(x, y), \quad \forall \ x, y \in \mathcal{X};$$

(iv) quasi ϕ -nonexpansive, if $F(T) \neq \emptyset$ and

$$\phi(p, Tx) \le \phi(p, x), \quad \forall \ x \in \mathcal{X}, \ p \in F(T);$$

(v) firmly nonexpansive, if for all $x, y \in \mathcal{X}$

$$\phi(Tx,Ty) + \phi(Ty,Tx) + \phi(Tx,x) + \phi(Ty,y) \le \phi(Tx,y) + \phi(Ty,x)$$

or equivalently,

$$\left\langle Tx - Ty, Jx - JTx - (Jy - JTy) \right\rangle \ge 0.$$

Remark 2.2.15. Every firmly nonexpansive mapping is nonexpansive.

Example 2.2.16. [178] Consider \mathbb{R} with the usual norm and let $\mathbb{R} \to \mathbb{R}$ be a function defined by

$$T(x,y) = (-y,x), \ \forall (x,y) \in \mathbb{R}^2.$$

Then, T is nonexpansive but not firmly nonexpansive.

2.3 Geometric properties of geodesic metric spaces

It is well known that the geometric structure of Hilbert spaces makes problems that occur in Hilbert spaces more manageable and easier to solve. However, most real life problems naturally occur in nonlinear spaces (for instance, in metric spaces). Therefore, there is the need to extend our study to nonlinear spaces. Sometimes, extending some known results to this space may be difficult because of the nonlinear structure of nonlinear spaces. In order to extend these results to nonlinear spaces, one needs to introduce some kind of convex properties which provide sufficient information that ensures the applications of such existing results. One of these properties is the existence of distance-preserving mapping, which provides the metric space (nonlinear space) with a structure that is similar to the linear structure of a normed linear space (in particular, Hilbert space). Metric spaces with this distance-preserving mappings are called geodesic metric spaces.

Definition 2.3.1. [33] Let (X, d) be a metric space. A continuous mapping from the interval [0, 1] to X is called a path.

Definition 2.3.2. Let $x, y \in X$ and I = [0, d(x, y)], a geodesic path joining x to y is a distance-preserving mapping (isometry) $c : I \to X$, such that c(0) = x, c(d(x, y)) = yand d(c(t), c(t')) = |t - t'| for all $t, t' \in [0, d(x, y)]$.

The image of a geodesic path is a geodesic segment and it is denoted by [x, y] whenever it is unique. A metric space (X, d) is said to be a geodesic space if every pair of points xand y in X are connected by a geodesic. A subset C of a geodesic space X is said to be convex, if for all $x, y \in C$, the segment [x, y] remains in C. For $x, y \in X$ and $t \in [0, 1]$ we write $tx \oplus (1-t)y$ for the unique point z in the geodesic segment joining x to y such that

$$d(x, z) = (1 - t)d(x, y)$$
 and $d(z, y) = td(x, y).$ (2.3.1)

2.3.1 Metric midpoints

Definition 2.3.3. Let X be a metric space and $x, y \in X$. A point $z \in X$ is a metric midpoint of x and y if $d(x, z) = \frac{1}{2}d(x, y)$.

Proposition 2.3.1. Let X be a complete metric space. Then the following are equivalent;

- (a) The space X is a geodesic space.
- (b) For every $x, y \in X$, there exists a point $z \in X$ such that

$$d^{2}(x,z) + d^{2}(y,z) = \frac{1}{2}d^{2}(x,y).$$

(c) Every pair of points in X has a metric midpoint.

Remark 2.3.4. If X is a geodesic metric space, then every pair of points in X has at least one midpoint which is also a metric midpoint.
Definition 2.3.5. Let X be a geometric metric space. A subset C of X is said to be convex in C if it includes every geodesic segment joining two of its point. This means that C is convex if $x, y \in C$, we have that $tx \oplus (1-t)y \in C$.

Remark 2.3.6. [45]

A geodesic segment in a space that is not uniquely geodesic may not necessarily be convex.

(b) A subset of a uniquely geodesic metric space which is endowed with the induced metric is geodesic if and only if it is convex.

2.3.2 Geodesic triangles

Definition 2.3.7. A geodesic triangle $\triangle(x_1, x_2, x_3)$ in a geodesic space X consists of three points $x_1, x_2, x_3 \in X$ (which are also called the vertices of \triangle) and a geodesic segment between each pair of vertices (which are also known as edges of \triangle).

Definition 2.3.8. A comparison triangle for the geodesic triangle $\triangle(x_1, x_2, x_3) \in X$ is a triangle $\overline{\triangle}(x_1, x_2, x_3) := \triangle(\overline{x_1}, \overline{x_2}, \overline{x_3})$ in Euclidean space \mathbb{R}^2 such that $d(x_i, x_j) = d_{\mathbb{R}^2}(\overline{x_i}, \overline{x_j}) \quad \forall i, j \in \{1, 2, 3\}.$

Definition 2.3.9. Let $\{x_n\}$ be a bounded sequence in a geodesic metric space X. Then, the asymptotic center $A(\{x_n\})$ of $\{x_n\}$ is defined by $A(\{x_n\}) = \{\bar{v} \in X : \limsup d(\bar{v}, x_n) =$

 $\inf_{v \in X} \limsup_{n \to \infty} d(v, x_n) \}.$

The sequence $\{x_n\}$ in X is said to be Δ -convergent to a point $\bar{v} \in X$ if $A(\{x_{n_k}\}) = \{\bar{v}\}$ for every subsequence $\{x_{n_k}\}$ of $\{x_n\}$. In this case, we write Δ - $\lim_{n\to\infty} x_n = \bar{v}$ and we say that \bar{v} is the Δ -limit of $\{x_n\}$.

2.3.3 Geometric properties of Hadamard spaces

The Hadamard spaces which were named after Jacques Hadamard are known as complete uniquely geodesic metric spaces of nonpositive curvatures and they include Hilbert spaces, Euclidean spaces (\mathbb{R}^n), \mathbb{R} -trees, hyperbolic spaces, Hilbert ball, among others. The geometry of Hadamard spaces can be seen as the nonlinearization of the geometry of Hilbert spaces. Karl Menger [176] introduced the notion of a geodesic in metric spaces and he generalized classical results in geometry to his new metric space with geodesic. To improve the work of Menger [176], Wald [258] introduced the notion of two-dimensional curvature in metric spaces. Alexandror [17] also contributed by discovering some interesting characteristics of the spaces. Over the years, the Hadamard spaces has shown to be an appropriate framework for the study of optimization problems which has applications in diverse fields such as economics, engineering and science.

Now we are in the position to present some characterizations and conditions required for geodesic spaces to be CAT(0) spaces.

Characterizations of CAT(0) spaces

Gromov [114] coined the term CAT(0) from the initials of three mathematicians where C stands for Cartan, A stands for Alexandrov and T stands for Toponogov. Precisely, CAT(0) spaces are spaces of non-positive curvature bounded above by 0.

Definition 2.3.10. A geodesic space is called a CAT(0) space if all geodesic triangles satisfy the comparison axiom.

Definition 2.3.11. Let \triangle be a geodesic triangle in X and let \triangle be its comparison triangle in \mathbb{R}^2 . Then, \triangle is said to satisfy the CAT(0) inequality, if for all $x, y \in \triangle$ and all comparison points $\bar{x}, \bar{y} \in \bar{\triangle}$,

$$d(x,y) \le d_{\mathbb{R}^2}(\bar{x},\bar{y}).$$

Let x, y, z be points in X and y_0 be the midpoint of the segment [y, z], then the CAT(0) inequality implies

$$d^{2}(x, y_{0}) \leq \frac{1}{2}d^{2}(x, y) + \frac{1}{2}d^{2}(x, z) - \frac{1}{4}d^{2}(y, z).$$
(2.3.2)

Thus, a geodesic space is a CAT(0) space if and only if it satisfies (2.3.2). It is generally known that a CAT(0) space is a uniquely geodesic space. Inequality (2.3.2) is known as the CN inequality of Bruhat and Titis [48].

Definition 2.3.12. A geodesic metric space X is called a CAT(0) space if all geodesic triangles satisfy the CAT(0) inequality. Equivalently, X is called a CAT(0) space if and only if it satisfies the CN inequality.

Theorem 2.3.13. (see [33, Theorem 1.3.2]). Let X be a complete metric space. Then the following are equivalent.

- (i) The space X is a CAT(0) space.
- (ii) For every pair of points $x, y \in X$, there exists $m \in X$ such that for each $z \in X$, we have that

$$d^{2}(m,z) \leq \frac{1}{2}d^{2}(x,z) + \frac{1}{2}d^{2}(y,z) - \frac{1}{4}d^{2}(x,y).$$

(iii) For every pair of points $x, y \in X$ and $\epsilon > 0$, there exists $m \in X$ such that for each $z \in X$, we have that

$$d^{2}(m,z) \leq \frac{1}{2}d^{2}(x,z) + \frac{1}{2}d^{2}(y,z) - \frac{1}{4}d^{2}(x,y) + \epsilon$$

Next, we present the following Theorem which gives equivalent conditions for a geodesic space to be a CAT(0) space.

Theorem 2.3.14. (see [33, Theorem 1.3.3]). Let X be a geodesic metric space. Then the following are equivalent:

(i) The space X is a CAT(0) space.

(ii) For every pair of points $x, y, z \in X$, we have

$$d^{2}(m,x) \leq \frac{1}{2}d^{2}(x,y) + \frac{1}{2}d^{2}(x,z) - \frac{1}{4}d^{2}(x,z),$$

where m is the midpoint of [y, z].

(iii) For every geodesic $x : [0,1] \to X$ and every point $p \in X$, we have

$$d^{2}(p, x_{1}) \leq (1-t)d^{2}(p, x_{0}) + td^{2}(p, x_{1}) - t(1-t)d^{2}(x_{0}, x_{1}).$$

(iv) For every $x, y, u, v \in X$, we have

$$d^{2}(x, u) + d^{2}(y, v) \leq d^{2}(x, y) + d^{2}(u, v) + 2d(x, v)d(y, u).$$

(v) For every $x, y, u, v \in X$, we have

$$d^{2}(x, u) + d^{2}(y, v) \leq d^{2}(x, y) + d^{2}(y, u) + d^{2}(u, v) + d^{2}(v, x).$$

Definition 2.3.15. A complete CAT(0) space is an Hadamard space.

2.3.4 Examples of Hadamard spaces

Now, we present some examples of Hadamard spaces;

Example 2.3.16. (Hilbert space)[33]. Hilbert spaces are Hadamard. The geodesics are the line segments. It is also known that a Banach space is CAT(0) if and only if it is Hilbert.

Example 2.3.17. (\mathbb{R} - trees)[33]. A metric space (X, d) is an \mathbb{R} -tree if it is uniquely geodesic and for every $x, y, z \in X$, we have $[x, z] = [x, y] \cup [y, z]$ whenever $[x, y] \cap [y, z] = \{y\}$. Also, all triangles in an \mathbb{R} -tree are trivial.

Example 2.3.18. All simply connected Riemannian manifold with non-positive sectional curvature induced with the Riemannian metric.

2.3.5 Quasilinearization mapping and dual space

The concept of quasilinearization in Hadamard spaces was introduced by Berg and Nikolaev [41]. They denoted a pair $(a, b) \in X \times X$ by \overrightarrow{ab} and called it a vector. Using this concept, they defined the quasilinearization as a map $\langle \cdot, \cdot \rangle : (X \times X) \times (X \times X) \to \mathbb{R}$ defined by

$$\langle \overrightarrow{ab}, \overrightarrow{cd} \rangle = \frac{1}{2} \left(d^2(a, d) + d^2(b, c) - d^2(a, c) - d^2(b, d) \right), \quad \forall a, b, c, d \in X.$$
(2.3.3)

It is easy to verify that $\langle \overrightarrow{ab}, \overrightarrow{ab} \rangle = d^2(a, b), \quad \langle \overrightarrow{ba}, \overrightarrow{cd} \rangle = -\langle \overrightarrow{ab}, \overrightarrow{cd} \rangle, \quad \langle \overrightarrow{ab}, \overrightarrow{cd} \rangle = \langle \overrightarrow{ae}, \overrightarrow{cd} \rangle + \langle \overrightarrow{eb}, \overrightarrow{cd} \rangle \text{ and } \langle \overrightarrow{ab}, \overrightarrow{cd} \rangle = \langle \overrightarrow{cd}, \overrightarrow{ab} \rangle, \text{ for all } a, b, c, d, e \in X.$

Definition 2.3.19. The space X is said to satisfy the Cauchy Schwartz inequality, if

$$\langle \overrightarrow{ab}, \overrightarrow{cd} \rangle \le d(a, b)d(c, d) \ \forall a, b, c, d \in X.$$

Moreover, a geodesic space is a CAT(0) space if and only if it satisfies the Cauchy-Schwartz inequality (see [138]).

Using the idea of quasilinearization mapping, Kakavandi and Amini [138] introduced the concept of dual space of an Hadamard space X as follows: Consider the map $\theta : \mathbb{R} \times X \times X \to C(X, \mathbb{R})$ defined by

$$\theta(t, a, b)(x) = t \langle \overrightarrow{ab}, \overrightarrow{ax} \rangle \ (t \in \mathbb{R}, a, b, x \in X),$$

where $C(X, \mathbb{R})$ denotes the space of all continuous real valued functions on X. The Cauchy-Schwartz inequality implies that $\theta(t, a, b)$ is a Lipschitz function with Lipschitz semi-norm $L(\theta(t, a, b)) = |t|d(a, b) \quad (t \in \mathbb{R}, a, b \in X)$, where

$$L(\varphi) = \sup\left\{\frac{\varphi(x) - \varphi(y)}{d(x, y)} : x, y \in X, x \neq y\right\}$$

is the Lipschitz semi-norm for any function $\varphi: X \to \mathbb{R}$.

Definition 2.3.20. A pseudometric D on $\mathbb{R} \times X \times X$ is defined by

$$D((t,a,b),(s,c,d)) = L(\theta(t,a,b) - \theta(s,c,d)), \quad (t,s \in \mathbb{R}, a, b, c, d \in X).$$

In an Hadamard space (X, d), the pseudometric space $(\mathbb{R} \times X \times X, D)$ can be considered as a subset of the pseudometric space of all real valued Lipschitz functions $(\text{Lip}(X, \mathbb{R}), L)$ (see [93, 211, 255]).

It is shown in [138] that D((t, a, b), (s, c, d)) = 0 if and only if $t\langle \overrightarrow{ab}, \overrightarrow{xy} \rangle = s\langle \overrightarrow{cd}, \overrightarrow{xy} \rangle$ for all $x, y \in X$. Thus, D induces an equivalence relation on $\mathbb{R} \times X \times X$, where the equivalence class of (t, a, b) is defined as

$$[\overrightarrow{tab}] := \{\overrightarrow{scd} : D((t, a, b), (s, c, d)) = 0\}.$$

Thus, the set $X^* = \{[\overrightarrow{tab}] : (t, a, b) \in \mathbb{R} \times X \times X\}$ is a metric space with metric $D([\overrightarrow{tab}], [\overrightarrow{scd}]) := D((t, a, b), (s, c, d)).$

Definition 2.3.21. Let (X, d) be an Hadamard space. Then, the pair (X^*, D) is called the dual space of (X, d).

Throughout this dissertation, we shall write X^* for the dual space of an Hadamard space X.

Remark 2.3.22. [138] The dual of a closed and convex subset of a Hilbert space H with nonempty interior is an Hadamard space and $t(b-a) \equiv [\overrightarrow{tab}]$ for all $t \in \mathbb{R}$, $a, b \in H$. We also note that X^* acts on $X \times X$ by

$$\langle x^*, \overrightarrow{xy} \rangle = t \langle \overrightarrow{ab}, \overrightarrow{xy} \rangle, \quad (x^* = [\overrightarrow{tab}] \in X^*, \ x, y \in X \text{ and } t \in \mathbb{R}).$$

2.3.6 Some inequalities that characterize Hadamard spaces

Lemma 2.3.23. [194] Let X be a CAT(0) space, $x, y, z, \in X$ and $t \in [0, 1]$. Then

(a)
$$d(tx \oplus (1-t)y, z) \le td(x, z) + (1-t)d(y, z).$$

(b)
$$d^2(tx \oplus (1-t)y, z) \le td^2(y, z) + (1-t)d^2(y, z) - t(1-t)d^2(x, y).$$

$$(c) \ d^2(tx \oplus (1-t)y, z) \le t^2 d^2(x, z) + (1-t)^2 d^2(y, z) + 2t(1-t) \langle \overrightarrow{xz}, \overrightarrow{yz} \rangle.$$

2.3.7 Some nonlinear single-valued mappings in Hadamard spaces

Let X be an Hadamard space and C be a nonempty closed and convex subset of X. A mapping $T: C \to C$ is said to be

(i) L-Lipschitzian, if there exists L > 0 such that

$$d(Tx, Ty) \le Ld(x, y), \ \forall \ x, y \in C,$$

(ii) nonexpansive, if

$$d(Tx, Ty) \le d(x, y) \quad \forall \ x, y \in C,$$

(iii) quasi-nonexpansive, if $F(T) \neq \emptyset$ and

$$d(Tx, y) \le d(x, y), \quad \forall x \in C \quad \text{and} \quad y \in F(T),$$

(iv) demicontractive, if $F(T) \neq \emptyset$ and there exists $k \in (0, 1)$ such that

$$d^{2}(Tx,y) \leq d^{2}(x,y) + kd^{2}(x,Tx), \quad \forall x \in C, \quad \forall y \in F(T).$$

(v) quasi-pseudocontractive if $F(T) \neq \emptyset$ and

$$d^{2}(Tx, y) \leq d^{2}(x, y) + d^{2}(x, Tx), \quad \forall \ x \in C, \quad \forall \ y \in F(T).$$
(2.3.4)

Remark 2.3.24. The class of quasi-pseudocontractive mappings includes some nonlinear mappings like nonexpansive mappings (with nonempty fixed points set), quasi-nonexpansive mappings and demicontractive mappings.

Example 2.3.25. [186] Let C be the closed interval [0, 1] with the absolute value as norm. Define $T: C \to C$ by

$$Tx = \begin{cases} \frac{1}{2}, & \text{if } x \in [0, \frac{1}{2}]\\ 0, & \text{if } x \in (\frac{1}{2}, 1]. \end{cases}$$

 ${\cal T}$ is quasi-pseudocontractive but not demicontractive.

2.3.8 Monotone operators and its resolvents in Hadamard spaces

One of the most important area of nonlinear and convex analysis is the monotone operator theory. This is due to the role it plays in optimization theory and related mathematical problems. In this section, we study the concept of monotone operators and its resolvent in Hadamard spaces.

Definition 2.3.26. Let X be an Hadamard space and X^* be its dual space. A multivalued operator $A: X \to 2^{X^*}$ is monotone if and only if for all $x, y \in D(A)$, $x^* \in Ax$, $y^* \in Ay$, we have

$$\langle x^* - y^*, \overline{yx} \rangle \ge 0.$$

A monotone operator A is called a maximal monotone operator if the graph G(A) of A defined by

$$G(A) := \{ (x, x^*) \in X \times X^* : x^* \in A(x) \},\$$

is not properly contained in the graph of any other monotone operator.

Definition 2.3.27. The resolvent of a monotone operator A of order $\lambda > 0$ is the multivalued mapping $J_{\lambda}^{A}: X \to 2^{X^{*}}$ defined by

$$J_{\lambda}^{A}x := \left\{ z \in X \mid \left[\frac{1}{\lambda} \overrightarrow{zx}\right] \in Az \right\}.$$
 (2.3.5)

The operator A satisfies the range condition if for every $\lambda > 0$, $D(J_{\lambda}^{A}) = X$.

The resolvent of monotone operators plays a crucial role in the approximation of solutions of MIPs. We present some lemmas that relate the fixed points of a resolvent of a monotone operator and the set of solutions of the MIP (1.2.16).

Lemma 2.3.28. [150] Let X be a CAT(0) space and J_{λ}^{A} be the resolvent of the operator A of order λ . Then we have that

- (a) For any $\lambda > 0$, $R(J_{\lambda}^{A}) \subset D(A)$ and $F(J_{\lambda}^{A}) = A^{-1}(0)$, where $R(J_{\lambda}^{A})$ is the range of J_{λ}^{A} .
- (b) If A is monotone, then J^A_{λ} is a single-valued and firmly nonexpansive mapping.
- (c) If A is monotone and $0 < \lambda \leq \mu$, then $d^2(J^A_\lambda x, J^A_\mu x) \leq \frac{\mu \lambda}{\mu + \lambda} d^2(x, J^A_\mu x)$, which implies that $d(x, J^A_\lambda x) \leq 2d(x, J^A_\mu x)$.

Lemma 2.3.29. [255] Let X be an Hadamard space and $A : X \to 2^{X^*}$ be a monotone mapping. Then,

$$d^{2}(u, J_{\lambda}^{A}x) + d^{2}(J_{\lambda}^{A}x, x) \le d^{2}(u, x), \qquad (2.3.6)$$

for all $u \in F(J^A_{\lambda}), x \in X$ and $\lambda > 0$.

The Moreau-Yosida resolvent $J_{\lambda}^f: X \to X$ of a proper convex and lower semicontinuous function f in X is defined by

$$J_{\lambda}^{f}(x) = \arg\min_{y \in X} \left[f(y) + \frac{1}{2\lambda} d^{2}(y, x) \right] \quad \forall x \in X, \ \lambda > 0.$$

$$(2.3.7)$$

Definition 2.3.30. Let C be a nonempty closed and convex subset of an Hadamard space X. A mapping $T: C \to C$ is said to be Δ -demiclosed, if for any bounded sequence $\{x_n\}$ in C such that $\Delta -\lim_{n\to\infty} x_n = x$ and $\lim_{n\to\infty} d(x_n, Tx_n) = 0$, then x = Tx.

2.4 Metric and generalized projections

In this section, we introduce and discuss some basic results on metric projection and present some examples.

Definition 2.4.1. Let C be a nonempty, closed and convex subset of a real Hilbert space \mathcal{H} . For every point $x \in \mathcal{H}$, there exists a unique nearest point in C denoted by $P_{C}x$ such that

 $||x - P_{\mathcal{C}}x|| \le ||x - y|| \quad \forall \ x \in \mathcal{H} \quad and \quad y \in \mathcal{C}.$

The operator $P_{\mathcal{C}} : \mathcal{H} \to \mathcal{C}$ is called the metric projection of \mathcal{H} onto \mathcal{C} .

Lemma 2.4.2. [16] The operator $\Pi_{\mathcal{C}} : \mathcal{X} \to \mathcal{C}$ is called the generalized projection operator if it is the solution to the minimization problem

$$\Pi_{\mathcal{C}} x = x^* \iff \phi(x^*, x) = \inf_{u \in \mathcal{C}} \phi(u, x).$$

If \mathcal{X} is a real Hilbert space then $\Pi_{\mathcal{C}} = P_{\mathcal{C}}$.

The metric projection $P_{\mathcal{C}}x$ is nonexpansive with the following properties;

Proposition 2.4.1. The metric projection is characterized by

$$\langle x - P_{\mathcal{C}}x, y - P_{\mathcal{C}}x \rangle \le 0 \quad \forall y \in \mathcal{C} \quad and \quad x \in \mathcal{H}.$$
 (2.4.1)

The consequences of Proposition 2.4.1 are

(i) the metric projection is firmly nonexpansive, that is

$$\|P_{\mathcal{C}}x - P_{\mathcal{C}}y\|^2 \le \langle x - y, P_{\mathcal{C}}x - P_{\mathcal{C}}y \rangle \quad \forall \quad x, y \in \mathcal{H},$$

(ii)

$$\|x - P_{\mathcal{C}}x\|^2 + \|y - P_{\mathcal{C}}x\|^2 \le \|x - y\|^2 \quad \forall \quad x \in \mathcal{H} \quad \text{and} \quad y \in \mathcal{C}.$$

(iii) If C is a closed subspace, then P_C coincides with the orthogonal projection from \mathcal{H} onto C, that is, $x - P_C x$ is orthogonal to C. Hence, for any $y \in C$,

$$\langle x - P_{\mathcal{C}}x, y \rangle = 0.$$

We know that the metric projections in Banach and Hadamard spaces have the same property.

Lemma 2.4.3. [16] Let C be a nonempty, closed and convex subset of a reflexive, strictly convex and smooth Banach space \mathcal{X} . If $x \in \mathcal{X}$ and $q \in C$, then

$$q = \Pi_{\mathcal{C}} x \iff \langle y - q, \ Jx - Jq \rangle \le 0, \quad \forall \ y \in \mathcal{C}$$

$$(2.4.2)$$

and

$$\phi(y, \ \Pi_{\mathcal{C}} x) + \phi(\Pi_{\mathcal{C}} x, x) \le \phi(y, \ x), \quad \forall \ y \in \mathcal{C}, \ x \in \mathcal{X}.$$
(2.4.3)

We are now in the position to present some examples of the metric projection.

Example 2.4.4. If $C = \{y \in \mathcal{H} : ||y - s|| \le \alpha\}$ is a closed ball centered at $s \in \mathcal{H}$ with radius $\alpha \ge 0$,

$$P_{\mathcal{C}}x = \begin{cases} s + \frac{\alpha(x-s)}{\|x-s\|}, & \text{if } x \notin \mathcal{C} \\ x, & \text{if } x \in \mathcal{C}. \end{cases}$$

Example 2.4.5. Let C = [a, b] be a closed rectangle in \mathbb{R}^n , where $a = (a_1, a_2, \dots, a_n)^T$ and $b = (b_1, b_2, \dots, b_n)^T$, then for $1 \le i \le n$

$$(P_{\mathcal{C}}x)_i = \begin{cases} a_i, & x_i < a_i, \\ x_i, & x_i \in [a_i, b_i], \\ b_i, & x_i > b_i \end{cases}$$
(2.4.4)

is the metric projection with the i^{th} coordinate.

Example 2.4.6. Let $C = \{y \in \mathcal{H} : \langle s, y \rangle \leq \alpha\}$ be a closed half space, with $s \neq 0$ and $\alpha \in \mathbb{R}$, then

$$P_{\mathcal{C}}x = \begin{cases} x - \frac{\langle s, x \rangle - \alpha}{\|s\|^2} s, & \text{if } \langle s, x \rangle > \alpha, \\ x, & \text{if } \langle s, x \rangle \le \alpha \end{cases}$$
(2.4.5)

is the metric projection onto \mathcal{C} .

Example 2.4.7. Let $C = \{y \in \mathcal{H} : \langle s, y \rangle = \alpha\}$ be a hyperplane, with $s \neq 0$ and $\alpha \in \mathbb{R}$, then

$$P_{\mathcal{C}}x = x - \frac{\langle s, x \rangle - \alpha}{\|s\|^2}s \tag{2.4.6}$$

is the metric projection onto \mathcal{C} .

Example 2.4.8. If C is the range of a $m \times n$ matrix A with full column rank, then $P_{\mathcal{C}}x = A(A^*A)^{-1}A^*X$, where A^* is the adjoint of A.

2.4.1 Convex functions

Definition 2.4.9. A function $c : \mathcal{H} \to \mathbb{R}$ is said to be Gâteaux differential at $x \in \mathcal{H}$, if there exists an element denoted by $c'(x) \in \mathcal{H}$ such that

$$\lim_{h \to 0} \frac{c(x+hv) - c(x)}{h} = \langle v, c'(x) \rangle, \quad \forall v \in \mathcal{H},$$

where c'(x) is called the Gâteaux differential of c at x. We say that if for each $x \in \mathcal{H}$, c is Gâteaux differentiable at x, then c is Gâteaux differentiable on \mathcal{H} .

Definition 2.4.10. A function $c : \mathcal{H} \to \mathbb{R}$ is called convex, if for all $t \in [0, 1]$ and $x, y \in \mathcal{H}$,

$$c(tx + (1 - t)y) \le tc(x) + (1 - t)c(y).$$

Remark 2.4.11. We note that in an Hadamard space X, the convex function is defined by

$$c(tx \oplus (1-t)y) \le tc(x) + (1-t)c(y) \ \forall x, y \in X, t \in (0,1);$$

Definition 2.4.12. Let X be a geodesic metric space. The function $f : D(f) \subseteq X \rightarrow \mathbb{R} \cup \{+\infty\}$ is said to be uniformly convex (see [77]), if there exists a strictly increasing function $\phi : \mathbb{R}_+ \rightarrow \mathbb{R}_+$ such that

$$f\left(\frac{1}{2}x\oplus\frac{1}{2}y\right) \le \frac{1}{2}f(x) + \frac{1}{2}f(y) - \phi(d(x,y)).$$

Now, we present an example of a convex function in an Hadamard space.

Example 2.4.13. [35] Let X be an Hadamard space. For a finite number of points a_1, a_2, \ldots, a_N and $(w_1, w_2, \ldots, w_N) \in S$ (where S is the convex hull of the canonical basis $e_1, e_2, \ldots, e_N \in \mathbb{R}^N$), the function $f: X \to \mathbb{R}$ defined by $f(x) = \sum_{n=1}^N w_n d^2(x, a_n)$ is convex and continuous.

Definition 2.4.14. A convex function $c : \mathcal{H} \to \mathbb{R}$ is said to be subdifferentiable at a point $x \in \mathcal{H}$ if the set

$$\partial c(x) = \{ u \in \mathcal{H} \mid c(y) \ge c(x) + \langle u, y - x \rangle, \ \forall y \in \mathcal{H} \}$$
(2.4.7)

is nonempty, where each element in $\partial c(x)$ is called a subgradient of c at x, $\partial c(x)$ is called the subdifferential of c at x and the inequality in (2.4.7) is called the subdifferential inequality of c at x. We say that c is subdifferentiable on \mathcal{H} if c is subdifferentiable at each $x \in \mathcal{H}$.

We also note that if c is Gâteaux differentiable at x, then c is subdifferentiable at x, and $\partial c(x) = \{c'(x)\}.$

Remark 2.4.15. We note that in a Banach space \mathcal{X} with a dual space \mathcal{X}^* , the subdifferential inequality (2.4.7) is given by

$$\partial c(x) = \{ u \in \mathcal{X}^* \mid c(y) \ge c(x) + \langle u, y - x \rangle, \ \forall y \in \mathcal{X} \}.$$
(2.4.8)

Definition 2.4.16. Let \mathcal{H} be a Hilbert space. The domain of a function $f : \mathcal{H} \to \mathbb{R} \cup \{+\infty\}$ is defined by $D(f) = \{x \in \mathcal{H} : f(x) < +\infty\}$. The function $f : D(f) \subseteq \mathcal{H} \to \mathbb{R} \cup \{+\infty\}$ is said to be

- (a) proper, if $D(f) \neq \emptyset$;
- (b) lower semicontinuous at a point $x \in D(f)$, if

$$f(x) \le \liminf_{x_n \to x} f(x_n),$$

(c) weakly lower semicontinuous at a point $x \in D(f)$, if

$$\liminf_{n \to \infty} f(x_n) \ge f(x),$$

holds for an arbitrary sequence $\{x_n\}_{n=0}^{\infty}$ in \mathcal{H} satisfying $x_n \rightharpoonup x$;

(d) weakly upper semicontinuous at a point $x \in D(f)$, if

$$f(x) \ge \limsup_{n \to \infty} f(x_n),$$

holds for an arbitrary sequence $\{x_n\}_{n=0}^{\infty}$ in \mathcal{H} satisfying $x_n \rightharpoonup x$;

Definition 2.4.17. A bifunction $f : \mathcal{C} \times \mathcal{C} \to \mathbb{R}$ is called

(i) strongly monotone on C if there exists a constant $\beta > 0$ such that

$$f(x,y) + f(y,x) \le -\beta ||x - y||^2, \ \forall x, y \in \mathcal{C};$$

(ii) monotone on C if

$$f(x,y) + f(y,x) \le 0, \quad \forall x, y \in \mathcal{C};$$

(iii) pseudomonotone on C if

$$f(x,y) \ge 0 \implies f(y,x) \le 0 \ \forall \ x,y \in \mathcal{C}$$

It is easy to see that (i) \implies (ii) \implies (iii) but the converses are not generally true.

Definition 2.4.18. A function $f : \mathcal{C} \to \mathbb{R}$ is said to be hemicontinuous at $y \in \mathcal{C}$, if and only if

$$\lim_{t \to 0^+} f(tx + (1-t)y) = f(y), \text{ for all } x \in \mathcal{C}$$

Nonlinear Multivalued mappings

In this section, we shall denote by $P(\mathcal{C}), CB(\mathcal{C})$ and $2^{\mathcal{C}}$, the family of all nonempty proximinal subsets of \mathcal{C} , the family of all nonempty, closed and bounded subsets of \mathcal{C} and the family of all nonempty subsets of \mathcal{C} respectively. Let **H** denote the Hausdorff metric on $CB(\mathcal{C})$, then for all $A, B \in CB(\mathcal{C})$,

$$\mathbf{H}(A,B) = \max\{\sup_{a \in A} d(a,B), \ \sup_{b \in B} d(b,A)\},$$
(2.4.9)

where $d(a, B) = \inf_{b \in B} ||a-b||$ is the distance from the point *a* to the subset *B*. Let $T : \mathcal{C} \to 2^{\mathcal{C}}$ be a multivalued mapping. A point $x \in \mathcal{C}$ is called a fixed point of *T*, if $x \in Tx$ while $x \in \mathcal{C}$ is called a strict fixed point of *T*, if $Tx = \{x\}$.

Definition 2.4.19. The mapping $T : \mathcal{C} \to 2^{\mathcal{C}}$ is called

• L-Lipschitz, if there exists L > 0 such that

$$\mathbf{H}(Tx, Ty) \le L \|x - y\| \ \forall \ x, y \in \mathcal{C},$$

if L = 1, then T is called nonexpansive, while T is called a contraction if $L \in [0, 1)$;

• quasinonexpansive, if $F(T) \neq \emptyset$ and

$$\mathbf{H}(Tx,p) \le \|x-p\| \ \forall \ x \in \mathcal{C} \ and \ p \in F(T).$$

Definition 2.4.20. Let \mathcal{X} be a Banach space and B a mapping of \mathcal{X} into $2^{\mathcal{X}^*}$. The effective domain of B denoted by dom(B) is given as dom $(B) = \{x \in \mathcal{X} : Bx \neq \emptyset\}$. Let $B : \mathcal{X} \to 2^{\mathcal{X}^*}$ be a multivalued operator on \mathcal{X} . Then

(i) The graph G(B) is defined by

$$G(B) := \{ (x, u) \in \mathcal{X} \times \mathcal{X} : u \in B(x) \},\$$

- (ii) the operator B is said to be monotone if $\langle x-y, u^*-v^* \rangle \ge 0$ for all $x, y \in dom(A), u^* \in Ax$ and $v^* \in Ay$.
- (iii) A monotone operator B on \mathcal{X} is said to be maximal if its graph is not properly contained in the graph of any other monotone operator on \mathcal{X} .

Let \mathcal{X} be a uniformly convex and smooth Banach space with a Gâteaux differential norm and let $B: \mathcal{X} \to 2^{\mathcal{X}^*}$ be a maximal monotone operator. We consider the metric resolvent of B,

$$Q^B_{\mu} = \left(I + \mu J^{-1}_{\mathcal{X}}B\right)^{-1}, \ \mu > 0.$$

It is known that the operator Q^B_{μ} is firmly nonexpansive and the fixed points of the operator Q^B_{μ} are the null points of *B* (see [152, 153]). The resolvent plays an important role in the approximation theory for zero points of maximal monotone operators in Banach spaces.

The following are the properties of the resolvent (see [21])

$$\langle Q^B_\mu x - x^*, J(x - Q^B_\mu x) \rangle \ge 0, \ x \in \mathcal{X}, x^* \in B^{-1}(0),$$
 (2.4.10)

in particular, if \mathcal{X} is a real Hilbert space, then

$$\langle J^B_{\mu}x - x^*, x - J^B_{\mu}x \rangle \ge 0, \ x \in \mathcal{X}, x^* \in B^{-1}(0),$$

where $J^B_{\mu} = (I + \mu B)^{-1}$ is the general resolvent, $B^{-1}(0) = \{z \in \mathcal{X} : 0 \in Bz\}$ is the set of null points of B. Also, we know that $B^{-1}(0)$ is closed and convex (see [237]).

Lemma 2.4.21. [44] Let $B : \mathcal{H} \to 2^{\mathcal{H}}$ be a maximal monotone mapping, and $A : \mathcal{H} \to \mathcal{H}$ be a Lipschitz continuous and monotone mapping. Then the mapping A + B is a maximal monotone.

2.5 Review on some optimization problems

2.5.1 Variational inequality and fixed point problems

The VIP which was first introduced independently by Fichera [104] and Stampacchia [228] is a central problem in nonlinear analysis. It provides an effective tool for studying many interesting problems across several fields of study such as economics, physics, mathematical programming, among others (see [118, 197] and other references therein). Many authors have proposed and analyzed several iterative algorithms for solving the VIP (1.2.1) and other related optimization problems, (see [9, 59, 130, 132, 135, 191, 193, 233, 234, 245, 257], and references therein).

Definition 2.5.1. [160, 175] Let $A : \mathcal{C} \to \mathcal{X}^*$ be an operator. The Minty variational inequalities (MVI) is a problem of finding a point $x^* \in \mathcal{C}$ such that

$$\langle Ay, y - x^* \rangle \ge 0, \forall y \in \mathcal{C}.$$
 (2.5.1)

We denote the set of solution of (2.5.1) by $M(\mathcal{C}, A)$. It is known that pseudomonotonicity implies $M(\mathcal{C}, A) \neq \emptyset$ but the converse is not true.

We have the following result showing the relationship between the VIP (1.2.1) and MVI (2.5.1).

Lemma 2.5.2. [175] Consider the VIP (1.2.1). Suppose the mapping $g : [0,1] \to \mathcal{X}^*$ defined by g(t) = A(tx + (1-t)y) for all $x, y \in \mathcal{C}$ and $t \in [0,1]$ (i.e., g is hemicontinuous), then $M(\mathcal{C}, A) \subset VI(\mathcal{C}, A)$. However, if A is pseudomonotone, then $VI(\mathcal{C}, A)$ is well defined and $M(\mathcal{C}, A) = VI(\mathcal{C}, A)$.

In recent years, authors have paid a lot of attention to developing efficient iterative algorithms for solving VIPs and other related optimization problems in Hilbert, Banach and Hadamard spaces. When approximating the solution of the VIP (1.2.1) under certain conditions, two common methods are used namely; the projection method and the regularized method. To use these methods, a certain level of monotonicity is required from the cost operator. In this study, we focus on approximating the solution of the VIP (1.2.1)using the projection method. Several authors have proposed and studied the projection type algorithms for approximating the solutions of VIPs (1.2.1) (see [9, 156] and other references therein).

In a real Hilbert space \mathcal{H} , the following fixed point theory characterizes the solution set of the VIP: For $\lambda > 0$, a point x^* is a solution of the VIP if and only if

$$x^* = P_{\mathcal{C}}(x^* - \lambda A x^*),$$

where $P_{\mathcal{C}}$ is the metric projection of \mathcal{H} onto \mathcal{C} . The simplest algorithm for solving the VIP (1.2.1) is the gradient projection method given by

Algorithm 2.5.1.

$$x_{n+1} = P_{\mathcal{C}}(x_n - \lambda A x_n), \ n \ge 1,$$

where $\lambda > 0$. This method involves only one projection onto the feasible set C per iteration. The weak convergence result of this method was obtained under certain strict conditions. This method is only effective for solving VIP (1.2.1) when A is either strongly monotone or inverse strongly monotone. To circumvent this limitation, Korpelevich [155] proposed an extragradient method for solving the VIP (1.2.1) in Euclidean spaces when A is monotone and L-Lipschitz continuous. The extragradient method is defined as follows:

Algorithm 2.5.2.

$$\begin{cases} y_n = P_{\mathcal{C}}(x_n - \lambda A x_n), \ n \ge 1\\ x_{n+1} = P_{\mathcal{C}}(x_n - \lambda A y_n), \end{cases}$$
(2.5.2)

where $\lambda \in (0, \frac{1}{L})$ and $P_{\mathcal{C}}$ is the metric projection from \mathcal{H} onto \mathcal{C} . If the solution set $VI(\mathcal{C}, A)$ is nonempty, then the sequence $\{x_n\}$ generated by (2.5.2) converges to an element in $VI(\mathcal{C}, A)$. The extragradient method involves two projections onto the feasible set \mathcal{C} per iteration. Computing projection onto an arbitrary closed convex set might be computationally expensive, which could be a barrier to the implementation of the extragradient method and its variants. To improve on the extragradient method, authors try to minimize the number of evaluations of the projection map $P_{\mathcal{C}}$ per iteration. Censor et. al [58] initiated a study in this direction and proposed a new method called subgradient extragradient method. The authors replaced the second projection onto \mathcal{C} in Algorithm (2.5.2) by a projection onto a specific constructible half space to come up with the following algorithm:

Algorithm 2.5.3.

$$\begin{cases} x_1 \in \mathcal{H}, \\ y_n = P_{\mathcal{C}}(x_n - \lambda_n A x_n), \\ T_n = \{ w \in \mathcal{H} : \langle x_n - \lambda_n A x_n - y_n, w - y_n \rangle \le 0 \}, \\ x_{n+1} = P_{T_n}(x_n - \lambda_n A y_n), n \ge 1, \end{cases}$$

$$(2.5.3)$$

where \mathcal{H} is a real Hilbert space. Algorithm (2.5.3) is less computationally expensive and more efficient than Algorithm (2.5.2). The authors proved that Algorithm (2.5.3) converges weakly to an element in $VI(\mathcal{C}, A) \neq \emptyset$. To improve on Algorithm (2.5.3), Tseng [253] proposed and studied a forward-backward method (also known as Tseng's extragradient method) which requires only one projection per iteration. The proposed method is presented as follow:

Algorithm 2.5.4.

$$\begin{cases} y_n = P_{\mathcal{C}}(x_n - \lambda A x_n) \\ x_{n+1} = y_n - \lambda (A y_n - A x_n), \quad \forall \ n \ge 0, \end{cases}$$
(2.5.4)

where A is monotone, L-Lipschitz continuous and $\lambda \in (0, \frac{2}{L})$. The author proved that the sequence $\{x_n\}$ generated by Algorithm (2.5.4) converges weakly to the solution set of VIP (1.2.1) under the assumption that $VI(\mathcal{C}, A) \neq \emptyset$.

The step size of all the algorithms above require prior knowledge of the Lipschitz constant of the monotone operator, which is often difficult to calculate or estimate. Recently, Yang and Liu [266] proposed and studied a Tseng's extragradient method combined with the Moudafi viscosity scheme, which does not require prior knowledge of the Lipschitz constant of the monotone operator. The authors proved strong convergence result for the proposed algorithm. Very recently, Shehu and Iyiola [223] proposed an algorithm which combines the viscosity method and the subgradient extragradient method for solving the VIP. They proved that the sequence generated by their algorithm converges strongly to a point in the solution set under appropriate conditions.

Motivated by the Tseng's method and the importance of studying the VIPs and FPPs, Yin et.al [270] proposed and studied a Tseng's type algorithm where $A : \mathcal{H} \to \mathcal{H}$ is quasimonotone, Lipschitz continuous and sequentially weakly continuous, and the mapping T is pseudocontractive. The authors were only able to obtain a weak convergence result of their method (see Appendix 3.5.17) to some point $x^* \in VI(\mathcal{C}, A) \cap F(T)$.

Chidume and Nnakwe [75] extended the study of the subgradient extragradient method from the framework of a real Hilbert space to the framework of a 2-uniformly convex and uniformly smooth Banach space. The authors proposed and studied the following method for solving the VIP (1.2.1):

Algorithm 2.5.5.

$$\begin{cases} x_{0} \in \mathcal{X}, \\ y_{n} = \Pi_{\mathcal{C}} J^{-1} (Jx_{n} - \tau A x_{n}), \\ T_{n} = \{ w \in \mathcal{X} : \langle w - y_{n}, Jx_{n} - \tau A x_{n} - Jy_{n} \rangle \leq 0 \}, \\ x_{n+1} = \Pi_{T_{n}} J^{-1} (Jx_{n} - \tau A y_{n}), \end{cases}$$
(2.5.5)

where $\Pi_{\mathcal{C}} : \mathcal{X} \to \mathcal{C}$ is the generalized projection of the Banach space, $J : \mathcal{X} \to 2^{\mathcal{X}^*}$ the normalized duality mapping and $\tau > 0$. Using this proposed method, the authors obtained a weak convergence result.

Recently, Cai *et al.* [53] proposed an algorithm for solving the variational inequalities involving monotone and Lipschitz continuous mapping in Banach spaces. The proposed algorithm by the authors is presented as follows:

Algorithm 2.5.6.

Step 0: Let $u \in \mathcal{X}$ be a given starting point . Set n = 1.

Step 1: Given the current iterate x_n , compute

$$y_n = \prod_{\mathcal{C}} (Jx_n - \lambda_n Ax_n).$$

If $x_n = y_n$, Stop. Else, construct the set

$$T_n := \{ z \in \mathcal{X} : \langle Jx_n - \lambda_n Ax_n - Jy_n, \ z - y_n \rangle \le 0 \}$$

and compute

$$z_n = \prod_{T_n} (Jx_n - \lambda_n Ay_n)$$

and update the next iterate via

$$x_{n+1} = J^{-1}(\alpha_n J u + (1 - \alpha_n) J z_n).$$
(2.5.6)

Step 2: Set n := n + 1 and go to Step 1.

Where \mathcal{X} is a 2-uniformly convex Banach space and \mathcal{X}^* the dual of \mathcal{X} . Under certain conditions, the authors obtained a strong convergence result for the proposed algorithm.

Recently, Tan *et al.* [240] proposed an inertial subgradient extragradient method with a new Armijo type step-size strategy to solve VIP (1.2.1) in real Hilbert spaces when the underlying operator is pseudomonotone and uniformly continuous. The proposed algorithm is presented as follows:

Algorithm 2.5.7.

Initialization: Given $\lambda_1 > 0$, $\theta > 0$, $\delta > 0, \ell \in (0, 1)$ and $\eta \in (0, 1)$. Let $x_0, x_1 \in \mathcal{H}$ be arbitrary. Set n := 1.

Iterative Steps: Given the iterates x_{n-1} and x_n for each $n \ge 1$, calculate x_{n+1} as follows;

Step 1: Compute

$$w_n = x_n + \theta_n (x_n - x_{n-1}),$$

where

$$\theta_n := \begin{cases} \min\left\{\theta, \frac{\tau_n}{\|x_n - x_{n-1}\|}\right\}, & \text{if } x_n \neq x_{n-1} \\ \theta, & \text{otherwise.} \end{cases}$$
(2.5.7)

Step 2: Compute

$$y_n = P_{\mathcal{C}}(w_n - \lambda_n A w_n).$$

If $w_n = y_n$ or $Ay_n = 0$, then **Stop** and y_n is a solution of the VIP. Otherwise, go to **Step** 3.

Step 3: Compute

$$z_n = P_{T_n}(w_n - \lambda_n A y_n)$$

where

$$T_n := \{ x \in \mathcal{H} \mid \langle w_n - \lambda_n A w_n - y_n, x - y_n \rangle \le 0 \},\$$

 $\lambda_n := \delta \ell^{m_n}$ and m_n is the smallest nonnegative integer m satisfying

$$\delta \ell^m \langle Ay_n - Aw_n, \quad y_n - z_n \rangle \le \frac{\eta}{2} \Big[\|w_n - y_n\|^2 + \|y_n - z_n\|^2 \Big].$$

Step 4: Calculate

$$x_{n+1} = \alpha_n f(x_n) + (1 - \alpha_n) z_n.$$

Set n := n + 1 and go to **Step 1**.

Under some mild conditions, the authors obtained strong convergence result for the proposed algorithm.

2.5.2 Split variational inequality problem

The SVIP which was introduced by Censor *et al.* [61] can be seen as a pair of VIPs in which a solution of one VIP occurs in the first space \mathcal{H}_1 whose image under a given bounded linear operator A is a solution of the second VIP in the second space \mathcal{H}_2 . Hence, the SVIP (1.2.4)-(1.2.5) is an interesting combination of the classical VIP and the following SFP introduced and studied by Censor and Elfving [57]:

Find
$$x \in \mathcal{C}$$
 such that $z = Tx \in \mathcal{Q}$, (2.5.8)

where C, Q are nonempty, closed and convex subsets of real Hilbert spaces \mathcal{H}_1 and \mathcal{H}_2 respectively and $T : \mathcal{H}_1 \to \mathcal{H}_2$ is a bounded linear operator. Thus, it has wide applications in many fields such as phase retrieval, medical image reconstruction, signal processing, radiation therapy treatment planning, among others (for example, see [52, 57, 61–63, 121, 132]).

Censor *et al.* [63] proposed an iterative algorithm to solve the SVIP when the cost operators A and F are monotone and Lipschitz continuous. They transformed the SVIP into an equivalent constrained VIP (CVIP) in the product space $\mathcal{H}_1 \times \mathcal{H}_2$ (see [63, Section 4]) and then solved the problem using the well-known subgradient extragradient method [59]. This product space formation has some limitations which include: the difficulty encountered when computing the projection onto some new product subspace formulations, the difficulty encountered when translating the method back to the original spaces \mathcal{H}_1 and \mathcal{H}_2 , and the fact that it does not fully exploit the splitting structure of the SVIP (1.2.4)-(1.2.5) (see, for example [63, Page 12]). To circumvent these limitations, Censor *et al.* [63] proposed a projection-based method that does not require any product space formulation. This makes the projection-based method easier to implement. The proposed projection-based method is presented as follows: For $x_1 \in \mathcal{H}_1$, the sequence $\{x_n\}$ is generated by

$$x_{n+1} = P_{\mathcal{C}}(I - \lambda A)(x_n + \eta T^*(P_{\mathcal{Q}}(I - \lambda F) - I)Tx_n), \quad n \ge 1,$$
(2.5.9)

where $\eta \in (0, \frac{1}{L})$ with L being the spectral radius of T^*T and T^* is the adjoint of T. The identity operator is denoted by I and $P_{\mathcal{C}}, P_{\mathcal{Q}}$ are metric projections onto \mathcal{C}, \mathcal{Q} , respectively. They obtained a weak convergence of the sequence $\{x_n\}$ generated by (2.5.9) to a solution of (1.2.4)-(1.2.5) under the condition that the solution set of problem (1.2.4)-(1.2.5) is nonempty, A, F are L_1, L_2 -co-coercive operators respectively, $\lambda \in [0, 2\alpha]$, where $\alpha := \min\{L_1, L_2\}$, and for all x which are solutions of (1.2.4),

$$\langle Ay, P_{\mathcal{C}}(I - \lambda A)(y) - x \rangle \ge 0, \quad \forall \ y \in \mathcal{H}.$$
 (2.5.10)

Observe that Algorithm (2.5.9) does not require the product space formation, thus it fully exploits the attractive splitting structure of the SVIP (1.2.4)-(1.2.5). However, the authors obtained a weak convergence of this method under some strong assumptions which are the fact that both mappings are required to be co-coercive and (2.5.10). Many authors have studied several methods which do not rely on assumption (2.5.10) for solving SVIP and other related problems (see for example [131, 145, 183]), but their methods also relied on the co-coercivity of the cost operators.

In a quest to overcome these limitations, Tian and Jiang [243] proposed an iterative method and they defined it as follows:

Algorithm 2.5.8.

$$\begin{cases} y_n = P_{\mathcal{C}}(x_n - \tau_n A^* (I - S') T x_n) \\ v_n = P_{\mathcal{C}}(y_n - \lambda_n A y_n) \\ x_{n+1} = P_{\mathcal{C}}(y_n - \lambda_n A v_n) \end{cases}$$
(2.5.11)

where $\{\tau_n\} \subset [a,b], \{\lambda_n\} \subset [c,d]$ for some $c, d \in (0,\frac{1}{L}), T : \mathcal{H}_1 \to \mathcal{H}_2$ is a bounded linear operator, $S' : \mathcal{H}_2 \to \mathcal{H}_2$ is a nonexpansive mapping and $A : \mathcal{C} \to \mathcal{H}_1$ is a monotone and Lipschitz continuous mapping. They obtained a weak convergence result of the sequence generated by Algorithm (2.5.11) to the following problem; Find

$$x^* \in \mathcal{C}$$
 such that $\langle Ax^*, x - x^* \rangle$, $\forall x \in \mathcal{C}$ such that $Tx^* \in F(S')$ (2.5.12)

where F(S') is the set of fixed points of S'. Since strong convergence results are more desirable and more applicable than the weak convergence results in infinite dimensional spaces, there is need to develop algorithms that generate strong convergence results.

Tian and Jiang [244] modified Algorithm (2.5.11) into the following viscosity method and they defined it as follows:

Algorithm 2.5.9.

$$\begin{cases} y_n = P_{\mathcal{C}}(x_n - \tau_n A^* (I - S') T x_n) \\ v_n = P_{\mathcal{C}}(y_n - \lambda_n A y_n) \\ t_n = P_{\mathcal{C}}(y_n - \lambda_n A v_n) \\ x_{n+1} = \alpha_n h(x_n) + (1 - \alpha_n) t_n \end{cases}$$
(2.5.13)

where $\{\tau_n\} \subset [a, b], \{\lambda_n\} \subset [c, d]$ for some $c, d \in (0, \frac{1}{L}), \{\alpha_n\} \subset (0, 1), T: \mathcal{H}_1 \to \mathcal{H}_2$ is a bounded linear operator, $S' : \mathcal{H}_2 \to \mathcal{H}_2$ is a nonexpansive mapping, h is a contraction mapping and $A: \mathcal{C} \to \mathcal{H}_1$ is a monotone and Lipschitz continuous mapping. We observe that the conditions on the underlying operators in Algorithms (2.5.11)-(2.5.13)do not require the strong co-coercive assumption but involve computation of many projections which makes them computationally expensive and may affect the efficiency of Algorithms (2.5.11)-(2.5.13). Algorithms (2.5.11)-(2.5.13) can be used to solve the SVIP (1.2.4)-(1.2.5) if we set $S' = P_Q(I - \lambda A)$ and let A be co-coercive. This implies that when solving the SVIP (1.2.4)-(1.2.5), these methods (Algorithm (2.5.11)-(2.5.13)) still relies on the co-coercive assumption on the underlying operator A. To weaken the condition on the underlying operators, Pham et al. [206] combined the Halpern method with the subgradient extragradient method for solving the SVIP (1.2.4)-(1.2.5) in real Hilbert spaces when the underlying operators A and F are pseudomonotone and Lipschitz continuous. The authors obtained a strong convergence result of their proposed method (see Appendix (4.3.16) to a solution of the SVIP (1.2.4)-(1.2.5) under the following conditions: $\limsup \langle A(x_n), y - y_n \rangle \leq \langle A(\bar{x}), y - \bar{y} \rangle$, for every sequences $\{x_n\}$ and $\{y_n\}$ in \mathcal{H}_1 converging weakly to \bar{x} and \bar{y} respectively, and $\limsup \langle F(c_n), d - d_n \rangle \leq \langle F(\bar{c}), d - \bar{d} \rangle$ for every sequences $\{c_n\}$ and $\{d_n\}$ in \mathcal{H}_2 converging weakly to \bar{c} and \bar{d} respectively.

To accelerate the convergence of iterative methods for solving the SVIP (1.2.4)-(1.2.5), the inertial and relaxation techniques were employed. This dynamical approach leads to the following Relaxed Inertial Tseng's Forward-Backward-Forward (RITFBF) with parameter $\phi_n = \theta h_n^2$, $\forall n \ge 1$ (see, for example [25]):

Algorithm 2.5.10.

$$\begin{cases} w_n = x_n + \alpha_n (x_n - x_{n-1}) \\ y_n = P_{\mathcal{C}} (w_n - \lambda_n A w_n) \\ x_{n+1} = (1 - \phi_n) w_n + \phi_n (y_n + \lambda_n (A w_n - A y_n)) . \end{cases}$$
(2.5.14)

When $\phi_n = 1$ and $\alpha_n = 0$ for all $n \ge 1$, Algorithm (2.5.14) reduces to the well-known Tseng's forward-backward-forward method [253], which is known to converge weakly to a solution of the classical VIP.

Given the importance of these two techniques (inertial and relaxation), it is of interest to consider their combination for solving optimization problems. Attouch and Cabot [26, 27] employed both techniques into proximal algorithms (resulting to Relaxed Inertial Proximal Algorithms (RIPA)) for solving convex minimization and null point problems. Also, Iutzeler and Hendrickx [129] studied the influence of inertial and relaxation techniques on the numerical performance of iterative schemes.

Remark 2.5.3. Several of the existing methods for solving the SVIP require that the problem be transformed into a product space problem which does not fully exploit the splitting structure of the SVIP. Therefore, there is need to develop efficient algorithms for solving the SVIP which does not require the product space formulation and converges to the minimum-norm solution of the SVIP. Note that in many practical problems with physical and engineering backgrounds, it is very important if the minimum-norm solutions of such problems can be found (see for example, [121, Section 5] and [54, 261]).

2.5.3 Split equalities of equilibrium, variational inequality and fixed point problems

The EP (1.2.7) has received a lot of attention from several authors due to its application to problems arising in the field of optimization, economics, physics, variational inequalities, among others (see for example [194, 200] and other references therein). Several authors have analyzed and proposed various iterative algorithms for approximating the solution of the EP and other related optimization problems, (see for example [191, 205, 209] and other references therein).

The SEP was first proposed by Moudafi [180]. It is used in numerous practical problems such as game theory, medical image reconstruction, partial differential equation, decomposition method, among others (see [141, 224, 241]).

If $\mathcal{H}_2 = \mathcal{H}_3$ and $\mathcal{F}_2 = I$ (*I* is the identity operator), the equation (1.2.8) reduces to the SFP proposed by Censor et al. [64] and defined as follows:

Find
$$x \in \mathcal{C}$$
 such that $\mathcal{F}_1 x \in \mathcal{Q}$,

where $\mathcal{F}_1 : \mathcal{H}_1 \to \mathcal{H}_2$ is a bounded linear operator. One of the most common method for solving (1.2.8) is the CQ projection method proposed and studied by Bryne et al. [49]. They defined it as follows:

Algorithm 2.5.11.

$$\begin{cases} x_{n+1} = P_{\mathcal{C}}(x_n - \eta_n \mathcal{F}_1^* (\mathcal{F}_1 x_n - \mathcal{F}_2 y_n)) \\ y_{n+1} = P_{\mathcal{Q}}(y_n + \eta_n \mathcal{F}_2^* (\mathcal{F}_1 x_n - \mathcal{F}_2 y_n)), \end{cases}$$
(2.5.15)

where $\eta_n \in \left(\epsilon, \frac{2}{\lambda_{\mathcal{F}_1} + \lambda_{\mathcal{F}_2}} - \epsilon\right)$, and $\lambda_{\mathcal{F}_1}$ and $\lambda_{\mathcal{F}_2}$ are the matrix operator norms $\|\mathcal{F}_1\|$ and $\|\mathcal{F}_2\|$, respectively. Note that the step size η_n in Algorithm (2.5.15) is dependent on the operator norms, which are difficult and sometimes impossible to compute.

If C and Q are the sets of fixed points of some nonlinear operators, the SEP (1.2.8) becomes the split equality common fixed point problem (SECFPP) which is defined as:

Find
$$x \in F(T_1)$$
 and $y \in F(T_2)$ such that $\mathcal{F}_1 x = \mathcal{F}_2 y$, (2.5.16)

where $F(T_1) \neq \emptyset$ and $F(T_2) \neq \emptyset$ are the sets of fixed points of T_1 and T_2 , respectively, $T_1 : \mathcal{H}_1 \to \mathcal{H}_1$ and $T_2 : \mathcal{H}_2 \to \mathcal{H}_2$ are nonlinear mappings and $\mathcal{F}_1 : \mathcal{H}_1 \to \mathcal{H}_3$, $\mathcal{F}_2 : \mathcal{H}_2 \to \mathcal{H}_3$ are bounded linear operators.

If $\mathcal{H}_2 = \mathcal{H}_3$ and $\mathcal{F}_2 = I$, then the SECFPP (2.5.16) reduces to the following split common fixed point problem (SCFPP) introduced by Censor et al. [66]

Find
$$x \in F(T_1)$$
 such that $\mathcal{F}_1 x \in F(T_2)$.

The SECFPP was first studied by Moudafi et al. [180]. They introduced the following simultaneous iterative method for solving the SECFPP

Algorithm 2.5.12.

$$\begin{cases} x_{n+1} = T_1(x_n - \eta_n \mathcal{F}_1^*(\mathcal{F}_1 x_n - \mathcal{F}_2 y_n)) \\ y_{n+1} = T_2(y_n + \eta_n \mathcal{F}_2^*(\mathcal{F}_1 x_n - \mathcal{F}_2 y_n)), \end{cases}$$
(2.5.17)

where $\eta_n \in \left(\epsilon, \frac{2}{\lambda_{\mathcal{F}_1} + \lambda_{\mathcal{F}_2}} - \epsilon\right), \lambda_{\mathcal{F}_1}$ and $\lambda_{\mathcal{F}_2}$ are the spectral radius of $\mathcal{F}_1^* \mathcal{F}_1$ and $\mathcal{F}_2^* \mathcal{F}_2$, respectively, and T_1 and T_2 are firmly quasi-nonexpansive mappings. We also observe that the step size of Algorithm (2.5.17) depends on the operator norms. Hence to implement Algorithm (2.5.17), one has to compute the operator norms of \mathcal{F}_1 and \mathcal{F}_2 which are difficult to compute. Several authors have studied and proposed modifications of Algorithm (2.5.17) for better implementation (see [177, 180, 272, 273] and other references there in).

Recently, Lopez et al. [115] studied and proposed a method for estimating the step size which does not require prior knowledge of the operator norms for solving the SFP. Dong et al. [119] and J. Zhao [271] also proposed new choices of step size which do not require prior knowledge of the operator norm for solving SECFPP. Zhao [271] studied the SEP and presented the following step size which guarantees convergence of the iterative method without prior knowledge of the operator norm of \mathcal{F}_1 and \mathcal{F}_2 :

Algorithm 2.5.13.

$$\eta_n \in \left(0, \frac{2\|\mathcal{F}_1 x_n - \mathcal{F}_2 y_n\|^2}{\|\mathcal{F}_1^* (\mathcal{F}_1 x_n - \mathcal{F}_2 y_n)\|^2 + \|\mathcal{F}_2^* (\mathcal{F}_1 x_n - \mathcal{F}_2 y_n)\|^2}\right)$$

Recently, Latif and Eslamian [159] studied and introduced a new algorithm for finding a common element of split equalities of EP, monotone VIP with Lipschitz operator and fixed point problem of nonexpansive semigroups satisfying the uniformly asymptotically regularity (u.a.r) condition in Hilbert spaces. The authors obtained strong convergence result for the proposed algorithm. However, their proposed algorithm has certain drawbacks. For instance, their method requires computing two projections each per iteration onto C

and Q, which makes it computationally expensive to implement. Moreover, the associated cost operators for the VIP are required to be monotone and Lipschitz continuous and the step size of the algorithm depends on the Lipschitz constants of these operators. In addition, the authors needed to impose the uniformly asymptotically regularity condition on the nonexpansive semigroups to obtain their result. All of these drawbacks limit the scope of application of their proposed method. Therefore, there is need to develop and study effective iterative algorithms which does not depend on all the above mentioned weaknesses.

2.5.4 Variational inclusion and split equilibrium problems

The V_q IP has attracted the interest of many researchers due to its application to various problems arising in optimal control, mathematical economics, convex programming and other related optimization problems. Specifically, some problems in machine learning, image processing and linear inverse problems can be modeled mathematically as V_q IP (1.2.9) [1, 10, 65, 86, 90, 111, 195, 256]. Many authors have studied and proposed iterative algorithms for solving the V_q IP (1.2.9). An efficient method for solving the V_q IP is the forward-backward splitting method considered in [11, 13, 39, 82, 86, 132, 161, 165, 203] and is formulated as follows:

Algorithm 2.5.14.

$$x_{n+1} = (I + rD)^{-1}(I - rB)x_n, \quad n \ge 1,$$

where r > 0, (I - rB) is the forward operator and $(I + rD)^{-1}$ is the resolvent operator introduced in [179] and it is also known as the backward operator. This forward-backward splitting method includes as special cases, the proximal point algorithm [215] and the gradient method. Several authors have studied and extended the forward-backward splitting methods. Lion and Mercier [161] introduced the Peaceman-Rachford splitting method [204]

Algorithm 2.5.15.

$$x_{n+1} = (2J_r^{D_1} - I)(2J_r^{D_2} - I)x_n, \quad n \ge 1$$

and the Douglas-Rachford splitting method [98]

$$x_{n+1} = J_r^{D_1} (2J_r^{D_1}) x_n + (I - J_r^{D_1}) x_n, \quad n \ge 1,$$

where $J_r^{D_i} = (I + rD_i)^{-1}$ i = 1, 2, with r > 0. They obtained weak convergence of these methods. Gibali *et al.* [108] introduced and studied two modifications of the forward-backward splitting method with a new step size rule for solving V_q IPs in real Hilbert spaces where the operators are Lipschitz continuous and monotone and maximal monotone. They obtained strong convergence results for the two methods.

Alvarez and Attouch [18] employed the heavy ball method which was studied by Polyak in [208] for maximal monotone operators using the proximal point algorithm. The proposed algorithm combines the inertial technique and the proximal point algorithm. The algorithm is called the inertial proximal point algorithm and it is defined as follows:

Algorithm 2.5.16.

$$\begin{cases} y_n = x_n + \theta_n (x_n - x_{n-1}) \\ x_{n+1} = (I + r_n D)^{-1} y_n, \quad n \ge 1. \end{cases}$$
(2.5.18)

They obtained a weak convergence of Algorithm 2.5.18 to a zero of B when $\{r_n\}$ is nondecreasing and $\{\theta_n\} \subset [0, 1)$ with

$$\sum_{n=1}^{\infty} \theta_n \|x_n - x_{n-1}\|^2 < \infty.$$
(2.5.19)

Moudafi and Oliny [184] proposed an inertial proximal point algorithm for solving the zero problem of the sum of two monotone operators, which is presented as follows:

Algorithm 2.5.17.

$$\begin{cases} y_n = x_n + \theta_n (x_n - x_{n-1}) \\ x_{n+1} = (I + r_n D)^{-1} (y_n - r_n B x_n), \quad n \ge 1 \end{cases}$$
(2.5.20)

They obtained a weak convergence of Algorithm 2.5.20 to the zero of (B+D) when $r_n < \frac{2}{L}$, L is the Lipschitz constant of B and condition (2.5.19) is satisfied. Note that Algorithm 2.5.20 does not take the form of the forward-backward splitting algorithm if $\theta_n > 0$ because B is still evaluated at the point x_n .

Recently, Lorenz and Pock [165] proposed the following inertial forward-backward algorithm for monotone operators:

Algorithm 2.5.18.

$$\begin{cases} y_n = x_n + \theta_n (x_n - x_{n-1}) \\ x_{n+1} = (I + r_n D)^{-1} (y_n - r_n B y_n), \quad n \ge 1 \end{cases}$$
(2.5.21)

where $r_n > 0$. They obtained a weak convergence analysis for this method. In 2018, Cholamjiak *et al.* [83] introduced the following inertial forward-backward splitting algorithm which combines Halpern and Mann iteration methods for solving V_q IPs in Hilbert spaces

Algorithm 2.5.19.

$$\begin{cases} y_n = x_n + \theta_n (x_n - x_{n-1}) \\ x_{n+1} = \beta_n u + \epsilon_n y_n + \mu_n J^D_{\lambda_n} (y_n - \lambda_n B y_n), & n \ge 1, \end{cases}$$
(2.5.22)

where $B: \mathcal{H} \to \mathcal{H}$ is k-inverse strongly monotone operator and $D: H \to 2^{\mathcal{H}}$ is a maximal monotone operator, $J_{\lambda_n}^D = (I + \lambda_n D)^{-1}$, $0 < \lambda_n \leq 2k$, $\{\alpha_n\} \subset [0, \alpha]$ with $\alpha \in [0, 1)$ and $\{\beta_n\}, \{\epsilon_n\}$ and $\{u_n\}$ are sequences in [0, 1] with $\beta_n + \epsilon_n + \mu_n = 1$ satisfying some certain conditions. They obtained a strong convergence of Algorithm 2.5.22 to an element in the solution set.

Kazmi and Rizvi [144] introduced and studied the SEP (1.2.10)-(1.2.11). Observe that problem (1.2.10) is the classical EP. When viewed separately, (1.2.10)-(1.2.11) are classical EPs. Thus, SEP constitutes a pair of EPs which have to be solved so that the image $y = \mathcal{T}x$ under a given bounded linear operator \mathcal{T} , of the solution of the EP in \mathcal{H}_1 , is a solution of the other EP in \mathcal{H}_2 .

Many authors have studied the problem of finding a common solution of the V_q IP and SEP due to its importance in solving real world problems whose constraints can be written as V_q IP and SEP. Very recently, Cholamjiak *et al.* [81] proposed a modified inertial forward-backward splitting method for solving the SEP and the V_q IP, which is presented as follows:

Algorithm 2.5.20.

$$\begin{cases} y_n = x_n + \delta_n (x_n - x_{n-1}) \\ z_n = \alpha_n y_n + (1 - \alpha_n) T_{r_n}^{F_1} (I - \gamma A^* (I - T_{r_n}^{F_2}) A) y_n, \\ x_{n+1} = \beta_n z_n + (1 - \beta_n) J_{t_n}^D (I - t_n B) z_n, \ n \ge 1, \end{cases}$$

$$(2.5.23)$$

where $J_{t_n}^D = (I + t_n D)^{-1}$, $\{t_n\} \subset (0, 2\alpha)$, $\{\delta_n\} \subset [0, \delta]$, $\delta \in [0, 1)$, $\{r_n\} \subset (0, \infty)$ with $\gamma \in (0, \frac{1}{L})$ such that L is the spectral radius of A^*A and $\{\alpha_n\}, \{\beta_n\}$ are sequences in [0, 1]. Under some certain conditions they obtained a weak convergence of Algorithm 2.5.23 to an element in the solution set (1.2.12). However, it has been pointed out by authors that the summability condition used in Algorithm 2.5.22 and Algorithm 2.5.23 is a drawback in their implementation. Another computational weakness of Algorithm 2.5.23 is the fact that its step size γ depends on the norm of the operator ||A||, which in most cases is unknown or very difficult to calculate or even estimate. Moreover, it is required that the operator B in Algorithm 2.5.23 and the other algorithms above is co-coercive (inverse strongly monotone) which is a very stringent condition. These are major drawbacks with the above algorithms and several existing algorithms in the literature. Therefore, there is need to develop and study effective iterative algorithms which does not depend on all the above mentioned weaknesses.

2.5.5 Split monotone variational inclusion and fixed point problems

In 2011, Moudafi [183] introduced the SMV_qIP (1.2.13). If $B_1 = B_2 = 0$ then the problem (1.2.13) reduces to the split variational inclusion problem (SV_qIP) defined as follows:

Algorithm 2.5.21.

$$\begin{cases} \text{Find} \quad x^* \in \mathcal{H}_1, \quad \text{such that} \quad 0 \in D_1(x^*), \\ \text{and} \\ y^* = \mathcal{T}x^* \in \mathcal{H}_2 \quad \text{such that} \quad 0 \in D_2(y^*), \end{cases}$$
(2.5.24)

where 0 is the zero vector, \mathcal{H}_1 and \mathcal{H}_2 are real Hilbert spaces, $D_1, D_2 : \mathcal{H}_2 \to 2^{\mathcal{H}_2}$ are multi-valued mappings, $\mathcal{T} : H_1 \to H_2$ is a bounded linear operator. The SV_qIP is also known as the split common null point problem (SCNPP) or the split zero problem (SZP). It includes as special cases, the split common fixed point problem (SCFFP), the SVIP and SFP (see [169, 182, 213] and other references therein), and it has wide applications in different fields such as medical treatment of the intensity-modulated radiation therapy (IMRT), data compression, medical image reconstruction, signal processing, phase retrieval, among others (for example, see [52, 57, 63]). The SV_qIP is considered to be a central problem in optimization and nonlinear analysis and has attracted the attention of several researchers because the theory provides a simple, natural and unified framework for a general treatment of many important mathematical problems such as MPs, network equilibrium problems, complementary problems, systems of nonlinear equations and others (see [31, 36, 63, 86, 105, 107, 147, 151, 210, 243] and other references therein).

Byrne *et al.* [51] proposed and studied the following algorithms for solving the SV_qIP for two maximal monotone operators A and D in Hilbert spaces. Take $x_0 \in \mathcal{H}_1$ such that

Algorithm 2.5.22.

$$x_{n+1} = J^A_\mu(x_n + \lambda T^*(J^D_\mu - I)Tx_n)$$
(2.5.25)

and

$$x_{n+1} = \alpha_n x_0 + (1 - \alpha_n) J^A_\mu (x_n + \lambda T^* (J^D D_\mu - I) T x_n)$$
(2.5.26)

for $\mu > 0, T^*$ the adjoint of $T, \lambda \in \left(0, \frac{2}{L}\right), L = ||T^*T||$ and $J^A_{\mu} := (I + \mu A)^{-1}, J^D_{\mu} := (I + \mu D)^{-1}$ are the resolvent operators of A and D, respectively. Under certain conditions, the authors obtained a weak convergence result for Algorithm 2.5.25 and a strong convergence result for Algorithm 2.5.26.

Moudafi [181] first introduced the viscosity approximation method which is defined as : Choose $x_0 \in \mathcal{H}$ such that the sequence $\{x_n\}$ is generated by

$$x_{n+1} = \alpha_n f(x_n) + (1 - \alpha_n) S x_n, \quad \forall n \ge 1,$$
(2.5.27)

where $\{\alpha_n\} \subset (0,1)$ and f is a contraction mapping. He proved that the sequence $\{x_n\}$ generated by (2.5.27) converges strongly to a fixed point of a nonexpansive mapping S under some suitable control conditions.

Very recently, Suantai *et al.* [229] proposed the following viscosity iterative scheme to approximate the solution of SV_qIP between a Banach space \mathcal{X} and a Hilbert space \mathcal{H} :

$$x_{n+1} = \alpha_n f(x_n) + \beta_n x_n + \gamma_n J^A_{\lambda_n}(x_n + \lambda_n T^* J_{\mathcal{X}}(Q^D_\mu - I)Tx_n), \quad \forall n \ge 1,$$
(2.5.28)

where $\{\mu_n\}, \{\lambda_n\} \subset (0, \infty), \{\alpha_n\}, \{\beta_n\}, \{\gamma_n\} \subset (0, 1)$ and $\alpha_n + \beta_n + \gamma_n = 1, f : \mathcal{H} \to \mathcal{H}$ is a contraction, $J_{\mathcal{X}}$ is the duality mapping on $\mathcal{X}, J_{\lambda}^A$ is the resolvent of A for $\lambda > 0, Q_{\mu}^D$ is the metric resolvent of D for $\mu > 0$. They proved that the sequence $\{x_n\}$ defined by (2.5.28) converges strongly to a solution of the SV_qIP under some certain conditions.

Many problems in sciences and engineering can be formulated as a problem of finding the solution of a FPP of a nonlinear mapping.

Motivated by the work of Byrne *et al.* [51], Kazmi and Rizvi [141] proposed the following algorithm for approximating a solution of SV_qIP which is also a fixed point of nonexpansive mapping.

Theorem 2.5.4. [141] Let \mathcal{H}_1 and \mathcal{H}_2 be two real Hilbert spaces and $T : \mathcal{H}_1 \to \mathcal{H}_2$ be a bounded linear operator. Let $f : \mathcal{H}_1 \to \mathcal{H}_1$ be a contraction mapping with constant $\rho \in (0,1)$ and $S : \mathcal{H}_1 \to \mathcal{H}_1$ be a nonexpansive mapping. For $x_0 \in \mathcal{H}_1$, let the sequences $\{u_n\}$ and $\{x_n\}$ be generated by

Algorithm 2.5.24.

$$\begin{cases} u_n = J_r^B(x_n + \lambda T^*(J_r^C - I)Tx_n), \\ x_{n+1} = \alpha_n f(x_n) + (1 - \alpha_n)Su_n, \ n \ge 0, \end{cases}$$
(2.5.29)

where r > 0 and $\lambda \in \left(0, \frac{1}{L}\right)$, L is the spectral radius of the operator T^*T and T^* is the adjoint of T, $\{\alpha_n\}$ is a sequence in (0,1) such that $\lim_{n\to\infty} \alpha_n = 0$, $\sum_{n=0}^{\infty} \alpha_n = \infty$ and $\sum_{n=0}^{\infty} |\alpha_n - \alpha_{n-1}| < \infty$. Then the sequence $\{u_n\}$ and $\{x_n\}$ both converge strongly to an element in the solution set.

Remark 2.5.5. All the above algorithms for solving SV_qIP have a common feature, which is also their computational weakness. It is the fact that their step size λ (or λ_n) depends on the norm of the operator ||T||, which in most cases is unknown or very difficult to calculate or even estimate. This is a major drawback of the algorithms and several existing algorithms in the literature. Therefore, there is need to develop efficient iterative algorithms (where the step size does not depend on the operator norm) for approximating the common solution of the SV_qIP and FPP.

2.5.6 Minimization and fixed point problems

A well-known and powerful method for finding solutions of MPs is the proximal point algorithm (PPA), which was introduced in Hilbert spaces by Martinet [174] and further

developed by Rockafellar [215] who proved that the PPA converges weakly to a minimizer of f. Rockafellar [215] then posed a question as to whether the PPA converges strongly or not without additional assumptions, and this question was resolved in the negative by $G\ddot{u}$ ler's counter example (see [113]). To obtain strong convergence of the PPA, Kamimura and Takahashi [140] incorporated the Halpern iterative method into the PPA and proved a strong convergence of the resulting sequence to a minimizer of f.

The study of the PPA in spaces with nonlinearity plays crucial role in the branch of analysis and geometry. For instance, the PPA in Hadamard spaces (nonlinear version of Hilbert spaces) are known to have applications in computational phylogenetics, consensus algorithms, modeling of airway systems in human lungs and blood vessels, computing of medians and means, among others (see [33–35, 103] and the references therein). The PPA was first studied in Hadamard spaces by Bačák [34] in 2013, as an extension from the classical linear spaces (for example, Euclidean and Hilbert spaces): For a starting point $x_1 \in X$, define $\{x_n\}$ by

Algorithm 2.5.25.

$$x_{n+1} = \arg\min_{y \in X} \left(f(y) + \frac{1}{2\lambda_n} d^2(y, x_n) \right), \qquad (2.5.30)$$

for each $n \in \mathbb{N}$, where $\lambda_n > 0$ for all $n \in \mathbb{N}$. Under the conditions that f has a minimizer in X and $\sum_{n=1}^{\infty} \lambda_n = \infty$, Bačák [34] proved that $\{x_n\}$ Δ -converges to a minimizer of f. Later in 2015, Cholamjiak [79] modified the PPA (2.5.30) into the following Halpern-type PPA and proved that it converges strongly to a minimizer of f in an Hadamard space:

Algorithm 2.5.26.

$$\begin{cases} y_n = \arg\min_{y \in X} (f(y) + \frac{1}{2\lambda} d^2(y, x_n)), \\ x_{n+1} = \alpha_n u \oplus (1 - \alpha_n) y_n, n \ge 1, \end{cases}$$
(2.5.31)

where $\lambda > 0, \{\alpha_n\} \subset (0, 1), \lim_{n \to \infty} \alpha_n = 0$ and $\sum_{n=1}^{\infty} \alpha_n = \infty$. In the same year, Cholamjiak *et al.* [80] proposed the following modified PPA by using the *S*-type iteration process for solving MP (2.4.4) and FPP for two nonexpansive mappings in CAT(0) spaces:

Algorithm 2.5.27.

$$\begin{cases} z_n = \arg \min_{y \in X} [f(y) + \frac{1}{2\lambda_n} d^2(y, x_n)], \\ y_n = (1 - \beta_n) x_n \oplus \beta_n T_1 z_n, \\ x_{n+1} = (1 - \eta_n) T_1 x_n \oplus \eta_n T_2 y_n, \end{cases}$$
(2.5.32)

where T_1 and T_2 are nonexpansive mappings on X, $\{\eta_n\}, \{\beta_n\}$ are sequences in (0, 1)and $\{\lambda_n\} \subset (0, \infty)$. They obtained a weak convergence result of Algorithm (2.5.32), and under some compactness conditions, they obtained some strong convergence results. By combining the PPA with the Noor iteration process [189], Thounthong *et al.* [252] proposed the following method for finding a common element of the set of fixed points of three nonexpansive mappings and the set of minimizers of lower semicontinuous and convex function in Hadamard spaces:

Algorithm 2.5.28.

$$\begin{cases} z_n = \arg\min_{y \in X} [f(y) + \frac{1}{2\lambda_n} d^2(y, x_n)], \\ w_n = (1 - \eta_n) x_n \oplus \eta_n T_1 z_n, \\ y_n = (1 - \beta_n) x_n \oplus \beta_n T_2 w_n, \\ x_{n+1} = (1 - \xi_n) x_n \oplus \xi_n T_3 y_n, \end{cases}$$
(2.5.33)

where $\lambda_n > 0$ for all $n \in \mathbb{N}, \{\eta_n\}, \{\beta_n\}, \{\xi_n\}$ are sequences in (0, 1) and $T_i (i = 1, 2, 3)$ are nonexpansive mappings. They also obtained a weak convergence result and some strong convergence results under some compactness conditions.

In 2017, Suparatulatorn *et al.* [232] proposed the following Halpern-type proximal point algorithm for approximating a common solution of MP and FFP for a nonexpansive mapping in an Hadamard space: For arbitrary $u, x_1 \in X$, define the sequence $\{x_n\}$ iteratively by

Algorithm 2.5.29.

$$\begin{cases} y_n = \arg\min_{y \in X} [f(y) + \frac{1}{2\lambda_n} d^2(y, x_n)], \\ x_{n+1} = \alpha_n u \oplus (1 - \alpha_n) T y_n \ n \ge 1, \end{cases}$$
(2.5.34)

where $\{\alpha_n\} \subset (0,1)$ and $\lambda_n \geq \lambda > 0$. They obtained a strong convergence result without the compactness condition. Later in 2018, Cuntavepanita and Phuengrattana [91] extended the results of Suparatulatorn *et al.* [232] from one nonexpansive mapping to a finite family of nonexpansive mappings. They obtained a weak convergence result and under some compactness assumption, they obtained some strong convergence results.

Just recently, Chang *et al.* [70] proposed the following iterative algorithm for approximating a common solution of a finite family of MPs and FFP for two demicontractive mappings in Hadamard spaces:

Algorithm 2.5.30.

$$\begin{cases} u_n = S_{\lambda}^m(x_n), \\ y_n = (1 - \beta_n) u_n \oplus \beta_n K_1 u_n, \\ x_{n+1} = (1 - \eta_n) u_n \oplus \eta_n K_2 y_n, \quad n \ge 1, \end{cases}$$
(2.5.35)

where $K_i(x) := \delta x \oplus (1-\delta)T_i x$, $x \in C$, i = 1, 2, $S_{\lambda}^j := J_{\lambda}^{f_j} \circ J_{\lambda}^{f_{j-1}} \circ, \cdots, \circ J_{\lambda}^{f_2} \circ J_{\lambda}^{f_1}$, $j = 1, 2, \cdots, m$. $\lambda > 0$, $\{\eta_n\}$ and $\{\beta_n\}$ are sequences in (0, 1) with $0 < a \le \eta_n, \beta_n < b < 0$

1. Chang *et al.* [70] obtained a weak convergence result and under some compactness assumption, they obtained some strong convergence results.

Another effective iterative method for finding solutions of MPs is the viscosity implicit method (VIM). Many authors have obtained convergence results using the VIM in more general spaces (see [137, 264] and other references therein). This study has also been extended to the Hadamard space (see [265]) and like every other type of iterative method, the PPA type of VIM has also been studied in this setting. In 2015, Xu et. al. [263] applied the VIM to the implict midpoint rule (IMR) for nonexpansive mappings in Hilbert spaces and proposed a viscosity implicit midpoint rule (VIMR). They proved that the sequence generated converges strongly under suitable conditions to a fixed point of the nonexpansive mapping which is also a solution of the variational inequality.

Recently, Ahmad and Ahmad [4] proposed a VIM of IMR in Hadamard space and they defined it as follows;

Algorithm 2.5.31.

$$\begin{cases} w_n = \frac{x_n \oplus x_{n+1}}{2}, \\ y_n = \alpha_n(w_n) \oplus \beta_n g(w_n) \oplus \tau_n T(w_n), \\ x_{n+1} = T(y_n), \end{cases}$$
(2.5.36)

where $\{\alpha_n\}, \{\beta_n\}$ and $\{\tau_n\}$ are sequences in (0, 1), g is a contraction with a coefficient $\theta \in [0, 1)$ and T is a nonexpansive mapping on D. They also obtained a strong convergence of (2.5.36) to a fixed point of the nonexpansive mapping.

Remark 2.5.6. Besides the above mentioned works which mainly motivated our study of MPs in Hadamard spaces, there are very few other results on MPs in Hadamard spaces. Thus, it is very important to further develop and generalize this study in Hadamard spaces.

2.5.7 Some important lemmas

Lemma 2.5.7. [234] For each $x_1, \dots, x_m \in \mathcal{H}$ and $\alpha_1, \dots, \alpha_m \in [0, 1]$ with $\sum_{i=1}^m \alpha_i = 1$, the equality

$$\|\alpha_1 x_1 + \dots + \alpha_m x_m\|^2 = \sum_{i=1}^m \alpha_i \|x_i\|^2 - \sum_{1 \le i < j \le m} \alpha_i \alpha_j \|x_i - x_j\|^2$$

holds.

Lemma 2.5.8. [89] Assume that $A : \mathcal{H} \to \mathcal{H}$ is a continuous and pseudomonotone operator. Then, x is a solution of (1.2.1) if and only if $\langle Ay, y - x \rangle \ge 0$, $\forall y \in \mathcal{C}$.

Theorem 2.5.9. [124, Theorem 2.3]. Let $p \in [1, \infty)$ be a rational number except for p = 1, 2. Unless P = NP, there is no algorithm which computes the p-norm of a matrix with entries in $\{-1, 0, 1\}$ to relative error with running time polynomial in the dimensions.

Lemma 2.5.10. [261, Theorem 4.1] Assume that the solution set VI(A, C) of the classical VIP (1.2.1) is nonempty and C is defined as $C := \{x \in \mathcal{H} \mid c(x) \leq 0\}$, where $c : \mathcal{H} \to \mathbb{R}$ is a continuously differential convex function. Let $p \in C$. Then, $p \in VI(A, C)$ if and only if either

- (*i*) Ap = 0, or
- (ii) $p \in \partial C$ and there exists $\rho > 0$ such that $Ap = -\rho c'(p)$, where ∂C denotes the boundary of C.

Following Attouch and Cabot [29, pages 5, 10], we note that if $x_{n+1} = x_n + \theta_n(x_n - x_{n-1})$, then for all $n \ge 1$, we have that

$$x_{n+1} - x_n = \left(\prod_{j=1}^n \theta_j\right) (x_1 - x_0),$$

which implies that

$$x_n = x_1 + \left(\sum_{j=1}^{n-1} \prod_{j=1}^{l} \theta_j\right) (x_1 - x_0).$$

Thus, $\{x_n\}$ converges if and only if $x_1 = x_0$ or if $\sum_{l=1}^{\infty} \prod_{j=1}^{l} \theta_j < \infty$. Therefore, we assume henceforth that

$$\sum_{l=i}^{\infty} \left(\prod_{j=i}^{l} \theta_{j}\right) < \infty \quad \forall i \ge 1.$$
(2.5.37)

Hence, we can define the sequence $\{t_i\}$ in \mathbb{R} by

$$t_i := \sum_{l=i-1}^{\infty} \left(\prod_{j=i}^l \theta_j \right) = 1 + \sum_{l=i}^{\infty} \left(\prod_{j=i}^l \theta_j \right), \qquad (2.5.38)$$

with the convention $\prod_{j=i}^{i-1} \theta_j = 1 \ \forall i \ge 1.$

Remark 2.5.11. (See also [29]).

Assumption (2.5.37) ensures that the sequence $\{t_i\}$ given by (2.5.38) is well-defined, and

$$t_i = 1 + \theta_i t_{i+1}, \ \forall i \ge 1.$$
 (2.5.39)

The following proposition provides a criterion for ensuring assumption (2.5.37).

Proposition 2.5.32. [29, Proposition 3.1] Let $\{\theta_n\}$ be a sequence such that $\theta_n \in [0, 1)$ for every $n \ge 1$. Assume that

$$\lim_{n \to \infty} \left(\frac{1}{1 - \theta_{n+1}} - \frac{1}{1 - \theta_n} \right) = c,$$

for some $c \in [0, 1)$. Then,

- (i) assumption (2.5.37) holds, and $t_{n+1} \sim \frac{1}{(1-c)(1-\theta_n)}$ as $n \to \infty$,
- (ii) the equivalence $1 \theta_n \sim 1 \theta_{n+1}$ holds true as $n \to \infty$. Hence, $t_{n+1} \sim t_{n+2}$ as $n \to \infty$.

Remark 2.5.12. Using Proposition 2.5.32, we can see that $\theta_n = 1 - \frac{\bar{\theta}}{n}$, $\bar{\theta} > 1$, is a typical example of a sequence satisfying assumption (2.5.37). Indeed, we have that

$$\lim_{n \to \infty} \left(\frac{1}{1 - \theta_{n+1}} - \frac{1}{1 - \theta_n} \right) = \lim_{n \to \infty} \left(\frac{1}{\overline{\theta}} (n+1) - \frac{1}{\overline{\theta}} n \right) = \frac{1}{\overline{\theta}} \in [0, 1),$$

which satisfies the assumption of Proposition 2.5.32. Hence by Proposition 2.5.32(i), assumption (2.5.37) holds.

It is worthy of note that the example $\theta_n = 1 - \frac{\bar{\theta}}{n}$, $\bar{\theta} > 1$, falls within the setting of Nesterov's extrapolation methods. In fact, many practical choices for θ_n satisfy assumption (2.5.37) (for instance, see [29, 40, 68, 187]).

The corresponding finite sum expression for $\{t_i\}$ is defined for $i, n \ge 1$, by

$$t_{i,n} := \begin{cases} \sum_{l=i-1}^{n-1} \left(\prod_{j=i}^{l} \theta_j\right) = 1 + \sum_{l=i}^{n-1} \left(\prod_{j=i}^{l} \theta_j\right), & i \le n, \\ 0, & \text{otherwise.} \end{cases}$$
(2.5.40)

In the same manner, we have that $\{t_{i,n}\}$ is well-defined and

$$t_{i,n} = 1 + \theta_i t_{i+1,n} \ \forall i \ge 1, \ n \ge i+1.$$
(2.5.41)

The sequences $\{t_i\}$ and $\{t_{i,n}\}$ are very crucial to our convergence analysis. In fact, their effect can be seen in the following lemma which also plays a crucial role in establishing our convergence results.

Lemma 2.5.13. [29, page 42, Lemma B.1]. Let $\{a_n\}, \{\theta_n\}$ and $\{b_n\}$ be sequences of real numbers satisfying

$$a_{n+1} \leq \theta_n a_n + b_n$$
 for every $n \geq 1$.

Assume that $\theta_n \geq 0$ for every $n \geq 1$.

(a) For every $n \ge 1$, we have

$$\sum_{i=1}^{n} a_i \le t_{1,n} a_1 + \sum_{i=1}^{n-1} t_{i+1,n} b_i,$$

where the double sequence $\{t_{i,n}\}$ is defined by (2.5.40).

(b) Under (2.5.37), assume that the sequence $\{t_i\}$ defined by (2.5.38) satisfies $\sum_{i=1}^{\infty} t_{i+1}[b_i]_+ < \infty$. Then, the series $\sum_{i>1} [a_i]_+$ is convergent, and

$$\sum_{i=1}^{\infty} [a_i]_+ \le t_1[a_1]_+ + \sum_{i=1}^{\infty} t_{i+1}[b_i]_+ ,$$

where $[t]_+ := \max\{t, 0\}$ for any $t \in \mathbb{R}$.

Lemma 2.5.14. [29, page 7, Lemma 2.1]. Let $\{x_n\}$ be a sequence in \mathcal{H} and $\{\theta_n\}$ be a sequence of real numbers. Given $z \in \mathcal{H}$, define the sequence $\{\Gamma_n\}$ by $\Gamma_n := \frac{1}{2} ||x_n - z||^2$. Then

$$\Gamma_{n+1} - \Gamma_n - \theta_n (\Gamma_n - \Gamma_{n-1}) = \frac{1}{2} (\theta_n + \theta_n^2) \|x_n - x_{n-1}\|^2 + \langle x_{n+1} - w_n, w_n - z \rangle + \frac{1}{2} \|x_{n+1} - w_n\|^2, \qquad (2.5.42)$$

where $w_n = x_n + \theta_n (x_n - x_{n-1})$.

The following lemmas are well-known.

Lemma 2.5.15. Let $\{x_n\}$ be any sequence in \mathcal{H} such that $x_n \rightharpoonup x$. Then,

$$\liminf_{n \to \infty} \|x_n - x\| < \liminf_{n \to \infty} \|x_n - y\|, \forall y \neq x.$$

Lemma 2.5.16. [122] Let C be a closed convex set in H, f be a real-valued function on H and define $K := \{x \in C : h(x) \leq 0\}$. If K is nonempty and h is Lipschitz continuous on C with modulus L > 0, then

$$dist(x, K) \ge L^{-1} \max \{f(x), 0\}, \quad \forall x \in \mathcal{C},$$

where dist(x, K) denote the distance function from x to K.

Lemma 2.5.17. [264] Let \mathcal{H} be a real Hilbert space and $S : \mathcal{H} \to \mathcal{H}$ be a nonexpansive mapping with $F(S) \neq \emptyset$. If $\{x_n\}$ is a sequence in \mathcal{H} converging weakly to x^* and $\{(I-S)x_n\}$ converges strongly to y, then $(I-S)x^* = y$.

Lemma 2.5.18. [221] Let $C \subseteq H$ be a nonempty, closed and convex subset of a real Hilbert space H. Let $u \in H$ be arbitrarily given, $z := P_{\mathcal{C}}u$, and $\Omega := \{x \in H : \langle x - u, x - z \rangle \leq 0\}$. Then $\Omega \cap \mathcal{C} = \{z\}$.

Lemma 2.5.19. Let \mathcal{X} be a smooth, strictly convex and reflexive Banach space. Let \mathcal{C} be a nonempty, closed and convex subset of \mathcal{X} , and let $x_1 \in \mathcal{X}$ and $z \in \mathcal{C}$. Then, $z = P_{\mathcal{C}} x_1$ if and only if

$$\langle z - y, J_{\mathcal{X}}(x_1 - z) \rangle \ge 0, \ \forall y \in \mathcal{C}.$$

Lemma 2.5.20. [164] Let \mathcal{X} be a real Banach space. Let $B : \mathcal{X} \to 2^{\mathcal{X}}$ be a maximal monotone operator and $A : \mathcal{X} \to \mathcal{X}$ be a k-inverse strongly monotone mapping on \mathcal{X} . Define $T_{\lambda} = (I + \lambda B)^{-1}(I - \lambda A), \ \lambda > 0$. Then we have

- (i) $F(T_{\lambda}) = (A+B)^{-1}(0);$
- (*ii*) for $0 < s \le \lambda$ and $x \in \mathcal{X}$, $||x T_s x|| \le 2||x T_\lambda x||$.

Lemma 2.5.21. [128] Let \mathcal{X} be a 2-uniformly convex Banach space and \mathcal{X}^* the dual space of \mathcal{X} . Suppose $A : \mathcal{X} \to \mathcal{X}^*$ is uniformly continuous on bounded subsets of \mathcal{X} and B is a bounded subset of \mathcal{X} . Then A(B) is bounded.

Lemma 2.5.22. [20] Suppose \mathcal{X} is a 2- uniformly convex Banach space. Then there exists $\mu \geq 1$ such that

$$\phi(x,y) \ge \frac{1}{\mu} \|x - y\|^2, \quad \forall \ x, y \in \mathcal{X},$$

where μ is the 2-uniform convexity constant of \mathcal{X} . If \mathcal{X} is a Hilbert space, then $\mu = 1$.

Lemma 2.5.23. [95] Let $x \in \mathcal{X}$ and $\psi \geq \sigma > 0$. The following inequality holds;

$$\frac{\|x - \Pi_{\mathcal{C}} J^{-1}(x - \psi A x)\|}{\psi} \le \frac{\|x - \Pi_{\mathcal{C}} J^{-1}(x - \sigma A x)\|}{\sigma}$$

Lemma 2.5.24. [139] Let \mathcal{X} be a smooth and uniformly convex real Banach space. Let $\{x_n\}$ and $\{y_n\}$ be two sequences in \mathcal{X} . If either $\{x_n\}$ or $\{y_n\}$ is bounded and $\phi(x_n, y_n) \to 0$, as $n \to \infty$, then $||x_n - y_n|| \to 0$, as $n \to \infty$.

Lemma 2.5.25. [196] Let X be an Hadamard space and $f: X \to (-\infty, \infty]$ be a proper convex and lower semi-continuous function. Then, $d^2(J^f_{\lambda}x, x) \leq d^2(J^f_{\mu}x, x)$ for $0 < \lambda < \mu$ and $x \in X$.

Lemma 2.5.26. [196] Let X be an Hadamard space and $f_j : X \to (-\infty, \infty], \quad j = 1, 2, \cdots, m$ be a finite family of proper, convex and lower semicontinuous functions. If $0 < \lambda < \mu$ and $\left(\bigcap_{j=1}^m F\left(J_{\mu}^{(j)}\right) \right) \neq \emptyset$. Then,

$$F\left(\Pi_{j=1}^{m}J_{\mu}^{(j)}\right)\subseteq\left(\cap_{j=1}^{m}F\left(J_{\lambda}^{(j)}\right)\right),$$

where, $\Pi_{j=1}^{m} J_{\mu}^{(j)} = J_{\mu}^{(1)} \circ J_{\mu}^{(2)} \circ \cdots \circ J_{\mu}^{(m)}$.

Lemma 2.5.27. [158] Let X be an Hadamard space and $f : X \to (-\infty, \infty]$ be a proper, convex and lower semicontinuous function. Then, for all $x, y \in X$ and $\lambda > 0$, we have

$$\frac{1}{2\lambda}d^2(J^f_{\lambda}x,y) - \frac{1}{2\lambda}d^2(x,y) + \frac{1}{2\lambda}d^2(x,J^f_{\lambda}x) + f(J^f_{\lambda}x) \le f(y).$$

Lemma 2.5.28. [93] Let X be a CAT(0) space, $\{v_1, v_2, \cdots, v_N\} \subset X$ and $\{\lambda_1, \lambda_2, \cdots, \lambda_N\} \subset (0,1)$ with $\sum_{i=1}^N \lambda_i = 1$. Then,

$$d\left(\sum_{i=1}^{N} \oplus \lambda_{i} v_{i}, x\right) \leq \sum_{i=1}^{N} \lambda_{i} d(v_{i}, x) \quad for \quad each \quad x \in X.$$

Remark 2.5.29. [93] For a CAT(0) space X, if $\{x_i, i = 1, 2, ..., N\} \subset X$, and $\alpha_i \in (0, 1), i = 1, 2, ..., N$. Then by induction, we can write

$$\sum_{i=1}^{N} \oplus \alpha_i x_i := (1 - \alpha_N) \sum_{i=1}^{N-1} \oplus \frac{\alpha_i}{1 - \alpha_N} x_i \oplus \alpha_N x_N.$$
(2.5.43)

Lemma 2.5.30. Let X be a CAT(0) space, $\{v_1, v_2, \cdots, v_N\} \subset X$ and $\{\lambda_1, \lambda_2, \cdots, \lambda_N\} \subset (0, 1)$ with $\sum_{i=1}^N \lambda_i = 1$. Then,

$$d^2\left(\sum_{i=1}^N \oplus \lambda_i v_i, x\right) \le \sum_{i=1}^N \lambda_i d^2(v_i, x) - \sum_{i,j=1, i \ne j}^N \lambda_i \lambda_j d^2(v_i, v_j).$$

Lemma 2.5.31. [194] Every bounded sequence in an Hadamard space has a \triangle -convergence subsequence.

Lemma 2.5.32. [94] Let C be a nonempty convex subset of a CAT(0) space X and $x \in X$. Then, $u = P_C x$ if and only if $\langle \overrightarrow{ux}, \overrightarrow{uy} \rangle \leq 0 \quad \forall y \in C$, where P_C is the metric projection of X onto C.

Lemma 2.5.33. [255] Let X be an Hadamard space and $T : X \to X$ be a nonexpansive mapping. Then T is a \triangle - demiclosed.

Lemma 2.5.34. [242] Suppose $\{\lambda_n\}$ and $\{\theta_n\}$ are two nonnegative real sequences such that

$$\lambda_{n+1} \le \lambda_n + \phi_n, \quad \forall n \ge 1.$$

If $\sum_{n=1}^{\infty} \phi_n < \infty$, then $\lim_{n \to \infty} \lambda_n$ exists.

Lemma 2.5.35. [166] Let $\{a_n\}$ be a sequence of nonnegative real numbers satisfying the following

$$a_{n+1} \le (1 - \beta_n)a_n + \tau_n + \gamma_n, \quad n \ge 1,$$

where $\{\beta_n\}$ is a sequence in (0,1) and $\{\tau_n\}$ is a real sequence. Suppose that $\sum_{n=1}^{\infty} \gamma_n < \infty$ and $\tau_n \leq \beta_n M$ for some M > 0. Then, $\{a_n\}$ is a bounded sequence.

Lemma 2.5.36. [9] Let $\{a_n\}$ be a sequence of non-negative real numbers, $\{\gamma_n\}$ be a sequence of real numbers in (0, 1) with conditions $\sum_{n=1}^{\infty} \gamma_n = \infty$ and $\{d_n\}$ be a sequence of real numbers. Assume that

$$a_{n+1} \le (1 - \gamma_n)a_n + \gamma_n d_n, \quad n \ge 1.$$

If $\limsup_{k \to \infty} d_{n_k} \leq 0$ for every subsequence $\{a_{n_k}\}$ of $\{a_n\}$ satisfying $\liminf_{k \to \infty} (a_{n_k+1} - a_{n_k}) \geq 0$, then $\lim_{n \to \infty} a_n = 0$.

Chapter 3

Results on Variational Inequality and Fixed Point Problems

3.1 Introduction

In this chapter, we present our results on monotone VIP and FPP of an infinite family of strict pseudo-contractive mappings in the real Hilbert space and apply these results to solve other nonlinear problems. Furthermore, using the projection technique, we present our results on VIP for a quasimonotone and Lipschitz operator in a real Hilbert space. Also, we present our result on quasimonotone VIP and FPP of a quasi pseudocontractive mapping in the real Hilbert space. We provide some numerical experiments of our proposed methods in comparison to other existing methods in the literature.

3.2 Preliminaries

Lemma 3.2.1. [276] Let C be a nonempty closed convex subset of a real Hilbert space \mathcal{H} and $S: C \to C$ be a k-strict pseudo-contractive mapping. Define a mapping $T: C \to C$ by $Tx = \alpha x + (1 - \alpha)Sx$ for all $x \in C$ and $\alpha \in [k, 1)$. Then T is a nonexpansive mapping such that F(T) = F(S).

Lemma 3.2.2. [239] Assume that $A : \mathcal{H} \to \mathcal{H}$ is a continuous and monotone operator. Then, x is a solution of the classical VIP (1.2.1) if and only if x is a solution of the following problem:

Find $x \in \mathcal{C}$ such that

$$\langle Ay, y - x \rangle \ge 0, \quad \forall y \in \mathcal{C}.$$

Definition 3.2.3. [259] Let $\{S_n\}$ be a sequence of k_n -strict pseudo-contractions. Define $S'_n = t_n I + (1 - t_n) S_n, t_n \in [k_n, 1)$. Then, by Lemma 3.2.1, S'_n is nonexpansive. In this

work, we consider the mapping W_n defined by

$$\begin{cases} U_{n,n+1} = I, \\ U_{n,n} = \zeta_n S'_n U_{n,n+1} + (1 - \zeta_n) I, \\ U_{n,n-1} = \zeta_{n-1} S'_{n-1} U_{n,n} + (1 - \zeta_{n-1}) I, \\ \cdots, \\ U_{n,k} = \zeta_k S'_k U_{n,k+1} + (1 - \zeta_k) I, \\ U_{n,k-1} = \zeta_{k-1} S'_{k-1} U_{n,k} + (1 - \zeta_{k-1}) I, \\ \cdots, \\ U_{n,2} = \zeta_2 S'_2 U_{n,3} + (1 - \zeta_2) I, \\ W_n = U_{n,1} = \zeta_1 S'_1 U_{n,2} + (1 - \zeta_1) I. \end{cases}$$
(3.2.1)

where $\{\zeta_i\}$ is a sequence of real numbers such that $0 \leq \zeta_i \leq 1$ for all $i \geq 1$. For each $n \geq 1$, such a mapping W_n is nonexpansive.

We have the following lemmas relating to the mapping W_n , which are needed in proving our main results.

Lemma 3.2.4. [225] Let C be a nonempty closed convex subset of a strictly convex Banach space \mathcal{X} . Let $\{S'_i\}$ be an infinite family of nonexpansive mappings of C into itself such that $\bigcap_{i=1}^{\infty} F(S'_i) \neq \emptyset$ and $\{\zeta_i\}$ be a real sequence such that $0 < \zeta_i \leq b < 1$ for all $i \geq 1$. Then we have the following:

- (1) W_n is nonexpansive and $F(W_n) = \bigcap_{i=1}^n F(S'_i)$ for each $n \ge 1$;
- (2) for each $x \in \mathcal{C}$ and for each positive integer k, then $\lim_{n \to \infty} U_{n,k}x$ exists;
- (3) the mapping W of C into itself defined by

$$Wx := \lim_{n \to \infty} W_n x = \lim_{n \to \infty} U_{n,1} x \quad \text{for all } x \in \mathcal{C}$$
(3.2.2)

is a nonexpansive mapping satisfying $F(W) = \bigcap_{i=1}^{\infty} F(S'_i)$, which is called the modified W-mapping generated by $S'_1, S'_2, \cdots, \zeta_1, \zeta_2, \cdots$ and t_1, t_2, \cdots .

By combining Lemma 3.2.1 and Lemma 3.2.4, it follows that $F(W) = \bigcap_{i=1}^{\infty} F(S'_i) = \bigcap_{i=1}^{\infty} F(S_i)$.

Lemma 3.2.5. [69] Let C be a nonempty closed convex subset of a strictly convex Banach space \mathcal{X} . Let $\{S'_i\}$ be an infinite family of nonexpansive mappings of C into itself such that $\bigcap_{i=1}^{\infty} F(S'_i) \neq \emptyset$ and $\{\zeta_i\}$ be a real sequence such that $0 < \zeta_i \leq b < 1$ for all $i \geq 1$, where b is a positive real number. If K is any bounded subset of C, then

$$\lim_{n \to \infty} \sup_{x \in K} ||Wx - W_n x|| = 0.$$

Lemma 3.2.6. [276] If S is a k-strict pseudo-contraction on closed convex subset C of a real Hilbert space \mathcal{H} , then I - S is demiclosed at any point $y \in \mathcal{H}$.

Let $VI(\mathcal{C}, A)$ be the solution set of problem (1.2.1) and Γ be the solution set of the following problem:

Find
$$x \in \mathcal{C}$$
 such that $\langle Ay, y - x \rangle \ge 0, \ \forall y \in \mathcal{C}.$ (3.2.3)

Then, Γ is a closed and convex subset of C, and since C is convex and A is continuous, we have the following relation

$$\Gamma \subseteq VI(\mathcal{C}, A). \tag{3.2.4}$$

Lemma 3.2.7. [269] Let C be a nonempty closed and convex subset of H. If either

- (i) A is pseudomonotone on \mathcal{C} and $VI(\mathcal{C}, A) \neq \emptyset$,
- (ii) A is the gradient of G, where G is a differential quasiconvex function on an open set $\mathcal{K} \supset \mathcal{C}$ and attains its global minimum on \mathcal{C} ,
- (iii) A is quasimonotone on $\mathcal{C}, A \neq 0$ on \mathcal{C} and \mathcal{C} is bounded,
- (iv) A is quasimonotone on $\mathcal{C}, A \neq 0$ on \mathcal{C} and there exists a positive number r such that, for every $x \in \mathcal{C}$ with $||x|| \geq r$, there exists $y \in \mathcal{C}$ such that $||y|| \leq r$ and $\langle Ax, y x \rangle \leq 0$,
- (v) A is quasimonotone on C, int C is nonempty and there exists $x^* \in VI(C, A)$ such that $Ax^* \neq 0$.

Then, Γ is nonempty.

Lemma 3.2.8. [71] Let \mathcal{H} be a real Hilbert space and $T : \mathcal{H} \to \mathcal{H}$ be an L-Lipschitz mapping with $L \geq 1$ such that for all $x \in \mathcal{H}$,

$$K := (1 - \zeta)I + \zeta T((1 - \eta)I + \eta T).$$
(3.2.5)

If $0 < \zeta < \eta < \frac{1}{1+\sqrt{1+L^2}}$, then the following conditions hold:

- (a) $F(T) = F(T((1 \eta)I + \eta T)) = F(K);$
- (b) if I T is demiclosed at 0, then I K is also demiclosed at 0;
- (c) if T is a quasi-pseudocontractive mapping, then K is a quasi-nonexpansive mapping.

3.3 Common solution of the variational inequality and the fixed point problems

In this section, we propose and study new inertial viscosity Tseng's extragradient algorithms with self-adaptive step size to solve the VIP and the FPP in Hilbert spaces. Our proposed methods involves a projection onto a halfspace and self-adaptive step size.
We prove that the sequence generated by our proposed methods converges strongly to a common solution of the VIP and FPP of an infinite family of strict pseudo-contractive mappings in Hilbert spaces under some mild assumptions when the underlying operator is monotone and Lipschitz continuous. Furthermore, we apply our results to find a common solution of VIP and ZPP for an infinite family of maximal monotone operators. Finally, we provide some numerical experiments of the proposed methods in comparison with other existing methods in the literature.

3.3.1 Proposed method

In this section, we present the following method with self-adaptive step size for solving the VIP (1.2.1) and the FPP (1.1.1).

```
Assumption 3.3.1. Suppose that the following conditions hold:
Condition A:
```

(A1) The feasible set C is given by

$$\mathcal{C} = \{ x \in \mathcal{H} : h(x) \le 0 \},\$$

where $h : \mathcal{H} \to \mathbb{R}$ is a convex and subdifferentiable function on \mathcal{C} .

- (A2) h is weakly lower semicontinuous.
- (A3) For any $x \in \mathcal{H}$, at least one subgradient $\xi \in \partial h(x)$ can be calculated, where $\partial h(x)$ is defined as follows

$$\partial h(x) = \{ z \in \mathcal{H} : h(u) \ge h(x) + \langle u - x, z \rangle, \quad \forall u \in \mathcal{H} \}.$$

In addition, $\partial h(x)$ is bounded on bounded sets.

(A4) Define the set C_n by the following half-space

$$\mathcal{C}_n = \{ x \in \mathcal{H} : h(w_n) + \langle \xi_n, x - w_n \rangle \le 0 \},\$$

where $\xi_n \in \partial h(w_n)$. By the definition of the subgradient, it is clear that $\mathcal{C} \subseteq \mathcal{C}_n$.

Condition B:

- (B1) $A: \mathcal{H} \to \mathcal{H}$ is monotone and Lipschitz continuous with Lipschitz constant L > 0.
- (B2) The solution set $\Gamma := \{z \in VI(A, \mathcal{C}) \bigcap \bigcap_{i=1}^{\infty} F(\mathcal{S}_i)\}$ is nonempty, where $\mathcal{S}_i : \mathcal{H} \to \mathcal{H}$ is an infinite family of k_i -strict pseudo-contractions.
- (B3) $D: \mathcal{H} \to \mathcal{H}$ is a strongly positive bounded linear operator with the coefficient γ_2 .
- (B4) $f: \mathcal{H} \to \mathcal{H}$ is a contraction with the coefficient $\rho \in (0,1)$ and $0 < \gamma_1 < \frac{\gamma_2}{\rho}$.

 $\begin{array}{ll} (B5) \ \{\beta_n\}_{n=1}^{\infty}, \{\theta_n\}_{n=1}^{\infty} and \ \{\tau_n\}_{n=1}^{\infty} \ are \ positive \ sequences \ satisfying \ the \ following \ conditions: \\ \{\beta_n\}, \{\theta_n\} \subset (0,1), \ 0 < c_1 \leq \beta_n, \ \lim_{n \to \infty} \theta_n = 0, \quad \sum_{n=1}^{\infty} \theta_n = \infty, \ \lim_{n \to \infty} \frac{\tau_n}{\theta_n} = 0. \end{array}$

Algorithm 3.3.2. Inertial method with adaptive step size strategy.

Step 0: Choose sequences $\{\theta_n\}_{n=1}^{\infty}$ and $\{\tau_n\}_{n=1}^{\infty}$ such that condition (B5) holds and let $\lambda_1 > 0, \mu \in (0, 1), \alpha \ge 3$ and $x_0, x_1 \in \mathcal{H}$ be given arbitrarily. Set n := 1.

Step 1: Given the iterates x_{n-1} and x_n for each $n \ge 1$, choose α_n such that $0 \le \alpha_n \le \overline{\alpha}_n$, where

$$\bar{\alpha}_n := \begin{cases} \min\left\{\frac{n-1}{n+\alpha-1}, \frac{\tau_n}{\|x_n - x_{n-1}\|}\right\}, & \text{if } x_n \neq x_{n-1} \\ \frac{n-1}{n+\alpha-1}, & \text{otherwise.} \end{cases}$$
(3.3.1)

Step 2: Compute

$$w_n = x_n + \alpha_n (x_n - x_{n-1}).$$

Step 3: Construct the halfspace

$$\mathcal{C}_n = \{ w \in \mathcal{H} \ | c(w_n) + \langle \xi_n, w - w_n \rangle \le 0 \}$$

and compute

$$y_n = P_{\mathcal{C}_n}(w_n - \lambda_n A w_n).$$

If $y_n = w_n (Aw_n = 0)$, then set $w_n = z_n$ and go to Step 5. Else go to Step 4.

Step 4: compute

$$z_n = y_n - \lambda_n (Ay_n - Aw_n).$$

Step 5: Compute

$$x_{n+1} = \theta_n \gamma_1 f x_n + (I - \theta_n D) T_n z_n,$$

where

$$T_n = (1 - \beta_n)I + \beta_n W_n$$
 and W_n is the mapping defined by (3.2.1).

Step 6: Compute

$$\lambda_{n+1} = \begin{cases} \min\left\{\frac{\mu||w_n - y_n||}{||Aw_n - Ay_n||}, \lambda_n\right\}, & \text{if } Aw_n \neq Ay_n\\ \lambda_n, & \text{otherwise.} \end{cases}$$
(3.3.2)

Set n := n + 1 and go back to Step 1.

Remark 3.3.1. Unlike several other existing methods for solving the VIP (1.2.1), our methods involves a projection onto a half-space without any projection onto the feasible set. Note that while projection onto the feasible set may be difficult to execute, projections onto half-spaces have closed formulas, and hence, they are easy to execute. Thus, our methods are more easily implementable than several existing ones in the literature. We prove that the sequence $\{x_n\}$ generated by our proposed methods converges strongly to (1.2.2) (a common solution of the VIP (1.2.1) and the FPP (1.1.1) for an infinite family of strict pseudo-contractive mappings) in Hilbert spaces under some mild conditions when the cost operator is monotone and Lipschitz continuous.

Remark 3.3.2. From (3.3.2) in Algorithm 3.3.2, it is clear that $\lambda_{n+1} \leq \lambda_n \forall n \geq 1$. Also, since A is L-Lipschitz continuous, we get in the case of $Aw_n \neq Ay_n$ in Algorithm 3.3.2, that

$$\lambda_{n+1} = \min\left\{\frac{\mu||w_n - y_n||}{||Aw_n - Ay_n||}, \ \lambda_n\right\} \ge \min\left\{\frac{\mu}{L}, \ \lambda_n\right\},$$

which by induction, implies that $\{\lambda_n\}$ is bounded below by $\min\{\frac{\mu}{L}, \lambda_1\}$. Since $\{\lambda_n\}$ is also monotone nonincreasing, we have that the limit exists, and $\lim_{n\to\infty} \lambda_n \geq \min\{\frac{\mu}{L}, \lambda_1\} > 0$.

Remark 3.3.3. By condition (f), one can easily verify from (3.3.1) that

$$\lim_{n \to \infty} \alpha_n ||x_n - x_{n-1}|| = 0 \text{ and } \lim_{n \to \infty} \frac{\alpha_n}{\theta_n} ||x_n - x_{n-1}|| = 0.$$
 (3.3.3)

Remark 3.3.4. By the definition of the subgradient, it is clear that $\mathcal{C} \subseteq \mathcal{C}_n$.

3.3.2 Convergence analysis

Lemma 3.3.5. Let $\{x_n\}$ be a sequence generated by Algorithm 3.3.2 under Assumption 3.3.1. Then

$$||z_n - p||^2 \le ||w_n - p||^2 - \left(1 - \frac{\lambda_n^2 \mu^2}{\lambda_{n+1}^2}\right) ||y_n - w_n||^2.$$

Proof. Let $p \in \Gamma$. Since $y_n = P_{\mathcal{C}_n}(w_n - \lambda_n w_n)$, we obtain by the characteristic property of $P_{\mathcal{C}_n}$ that $\langle y_n - w_n + \lambda_n A w_n, y_n - p \rangle \leq 0$ and this implies that

$$\langle y_n - w_n, y_n - p \rangle \le -\lambda_n \langle Aw_n, y_n - p \rangle.$$
 (3.3.4)

Also, from the definition of z_n in Step 4, and Lemma 2.1.1, we have

$$\begin{aligned} \|z_{n} - p\|^{2} &\leq \|y_{n} - \lambda_{n}(Ay_{n} - Aw_{n}) - p\|^{2} \\ &= \|y_{n} - p\|^{2} + \lambda_{n}^{2} \|Ay_{n} - Aw_{n}\|^{2} - 2\lambda_{n} \langle Ay_{n} - Aw_{n}, y_{n} - p \rangle \\ &= \|w_{n} - p\|^{2} + \|y_{n} - w_{n}\|^{2} + 2\langle y_{n} - w_{n}, w_{n} - p \rangle + \lambda_{n}^{2} \|Ay_{n} - Aw_{n}\|^{2} \\ &- 2\lambda_{n} \langle Ay_{n} - Aw_{n}, y_{n} - p \rangle \\ &= \|w_{n} - p\|^{2} + \|y_{n} - w_{n}\|^{2} + \lambda_{n}^{2} \|Ay_{n} - Aw_{n}\|^{2} - 2\langle y_{n} - w_{n}, y_{n} - w_{n} \rangle \\ &+ 2\langle y_{n} - w_{n}, y_{n} - p \rangle \\ &- 2\lambda_{n} \langle Ay_{n} - Aw_{n}, y_{n} - p \rangle \\ &= \|w_{n} - p\|^{2} - \|y_{n} - w_{n}\|^{2} + \lambda_{n}^{2} \|Ay_{n} - Aw_{n}\|^{2} + 2\langle y_{n} - w_{n}, y_{n} - p \rangle \\ &- 2\lambda_{n} \langle Ay_{n} - Aw_{n}, y_{n} - p \rangle \end{aligned}$$

$$(3.3.5)$$

From (3.3.4) and (3.3.5), we obtain

$$\begin{aligned} \|z_{n} - p\|^{2} &\leq \|w_{n} - p\|^{2} - \|y_{n} - w_{n}\|^{2} + \lambda_{n}^{2} \|Ay_{n} - Aw_{n}\|^{2} - 2\lambda_{n} \langle Aw_{n}, y_{n} - p \rangle \\ &- 2\lambda_{n} \langle Ay_{n} - Aw_{n}, y_{n} - p \rangle \\ &= \|w_{n} - p\|^{2} - \|y_{n} - w_{n}\|^{2} + \lambda_{n}^{2} \|Ay_{n} - Aw_{n}\|^{2} - 2\lambda_{n} \langle Ay_{n}, y_{n} - p \rangle \\ &\leq \|w_{n} - p\|^{2} - \|y_{n} - w_{n}\|^{2} + \lambda_{n}^{2} \frac{\mu^{2}}{\lambda_{n+1}^{2}} \|y_{n} - w_{n}\|^{2} - 2\lambda_{n} \langle Ay_{n}, y_{n} - p \rangle \\ &\leq \|w_{n} - p\|^{2} - \left(1 - \frac{\lambda_{n}^{2} \mu^{2}}{\lambda_{n+1}^{2}}\right) \|y_{n} - w_{n}\|^{2}. \end{aligned}$$

$$(3.3.6)$$

Consider the limit

$$\lim_{n \to \infty} \left(1 - \frac{\lambda_n^2 \mu^2}{\lambda_{n+1}^2} \right) = 1 - \mu^2 > 0.$$

Hence there exists $n_0 \ge 0$ such that for all $n \ge n_0$, we have that $\left(1 - \frac{\lambda_n^2 \mu^2}{\lambda_{n+1}^2}\right) > 0$. Thus, from (3.3.6), we have that

$$||z_n - p|| \le ||w_n - p||.$$

Lemma 3.3.6. Let $\{x_n\}$ be a sequence generated by Algorithm 3.3.2 under Assumption 3.3.1. Then, $\{x_n\}$ is bounded.

Proof. First, we show that $P_{\Gamma}(I - D + \gamma_1 f)$ is a contraction of \mathcal{H} . For all $x, y \in \mathcal{H}$, we have

$$\begin{aligned} \|P_{\Gamma}(I - D + \gamma_1 f)x - P_{\Gamma}(I - D + \gamma_1 f)y\| &\leq \|(I - D + \gamma_1 f)x - (I - D + \gamma_1 f)y\| \\ &\leq \|(I - D)x - (I - D)y\| + \gamma_1 \|fx - fy\| \\ &\leq (1 - \gamma_2)\|x - y\| + \gamma_1 \rho\|x - y\| \\ &= (1 - (\gamma_2 - \gamma_1 \rho))\|x - y\|. \end{aligned}$$

Hence, $P_{\Gamma}(I - D + \gamma_1 f)$ is a contraction. Let $p \in \Gamma$. Then from the definition of w_n in **Step 2**, we have

$$|w_n - p|| = ||x_n + \alpha_n(x_n - x_{n-1}) - p||$$

= ||x_n - p|| + \alpha_n ||x_n - x_{n-1}||

Also, from **Step 2**, we observe that $\alpha_n ||x_n - x_{n-1}|| \leq \tau_n, \forall n \geq 1$, which implies that

$$\frac{\alpha_n}{\theta_n} \|x_n - x_{n-1}\| \le \frac{\tau_n}{\theta_n} \to 0, \quad \text{as} \quad n \to \infty.$$
(3.3.7)

Hence, there exists $M_1 > 0$ such that

$$\frac{\alpha_n}{\theta_n} \|x_n - x_{n-1}\| \le M_1, \quad \forall n \ge 1.$$
(3.3.8)

This implies that $||w_n - p|| \le ||x_n - p|| + \theta_n M_1, \forall n \ge 1$ and hence

$$||z_n - p|| \le ||w_n - p|| \le ||x_n - p|| + \theta_n M_1.$$
(3.3.9)

From **Step 5**, we have

$$\|T_n z_n - p\| \le \|(1 - \beta_n) z_n + \beta_n W_n z_n - p\|$$

$$\le (1 - \beta_n) \|z_n - p\| + \beta_n \|W_n z_n - p\|$$

$$\le (1 - \beta_n) \|z_n - p\| + \beta_n \|z_n - p\|$$

$$= \|z_n - p\|$$
(3.3.10)

From (3.3.9) and (3.3.10), we obtain for all $n \ge n_0$,

$$\begin{aligned} \|x_{n+1} - p\| &\leq \|\theta_n \gamma_1 f x_n + (1 - \theta_n D) T_n z_n - p\| \\ &= \|\theta_n (\gamma_1 f x_n - Dp) + (1 - \theta_n D) (T_n z_n - p)\| \\ &\leq \theta_n \|\gamma_1 f x_n - Dp\| + (1 - \theta_n \gamma_2) \|T_n z_n - p\| \\ &\leq \theta_n \|\gamma_1 (f x_n - fp) + (\gamma_1 f p - Dp)\| + (1 - \theta_n \gamma_2) (\|x_n - p\| + \theta_n M_1) \\ &\leq (1 - \theta_n (\gamma_2 - \gamma_1 \rho)) \|x_n - p\| + \theta_n \|\gamma_1 f p - Dp\| + \theta_n M_1 \\ &= (1 - \theta_n (\gamma_2 - \gamma_1 \rho)) \|x_n - p\| + \theta_n (\gamma_2 - \gamma_1 \rho) \frac{\|\gamma_1 f p - Dp\| + M_1}{\gamma_2 - \gamma_1 \rho} \\ &\leq \max \Big\{ \|x_n - p\|, \frac{\|\gamma_1 f p - Dp\| + M_1}{\gamma_2 - \gamma_1 \rho} \Big\} \\ &\leq \max \Big\{ \|x_{n_0} - p\|, \frac{\|\gamma_1 f p - Dp\| + M_1}{\gamma_2 - \gamma_1 \rho} \Big\} \end{aligned}$$

Hence, the sequence $\{x_n\}$ is bounded. Consequently, $\{w_n\}, \{y_n\}$ and $\{z_n\}$ are also bounded.

Lemma 3.3.7. Assume that $\{w_n\}$ and $\{y_n\}$ are sequences generated by Algorithm 3.3.2 such that $\lim_{n\to\infty} ||w_n - y_n|| = 0. \text{ If } \{w_{n_j}\} \text{ converges weakly to some } \hat{x} \in \mathcal{H} \text{ as } j \to \infty, \text{ then } \hat{x} \in VI(\mathcal{C}, A).$

Proof. Since $w_{n_j} \rightharpoonup \hat{x}$, then by the hypothesis of the lemma it follows that $y_{n_j} \rightharpoonup \hat{x}$ as $j \rightarrow \infty$. Also, since $y_{n_j} \in \mathcal{C}_{n_j}$, then by the definition of \mathcal{C}_n we get

$$h(w_{n_j}) + \langle \xi_{n_j}, y_{n_j} - w_{n_j} \rangle \le 0.$$
 (3.3.11)

By the boundedness of $\{w_n\}$ and by condition (A3), there exists a constant M > 0 such that $||\xi_{n_j}|| \leq M$ for all $j \geq 0$. Then, from (3.3.11) we obtain $h(w_{n_j}) \leq M ||w_{n_j} - y_{n_j}|| \rightarrow 0$ as $j \to \infty$, and this in turn implies that $\liminf_{j \to \infty} h(w_{n_j}) \leq 0$. Applying condition (A2), we have $h(\hat{x}) \leq \liminf_{j \to \infty} h(w_{n_j}) \leq 0$. This implies that $\hat{x} \in \mathcal{C}$. From Lemma 2.4.1, we obtain

$$\langle y_{n_j} - w_{n_j} + \lambda_{n_j} A w_{n_j}, \ z - y_{n_j} \rangle \ge 0, \quad \forall \ z \in \mathcal{C} \subseteq \mathcal{C}_{n_j}.$$

Since A is monotone, we have

$$\begin{split} 0 &\leq \langle y_{n_j} - w_{n_j}, \ z - y_{n_j} \rangle + \lambda_{n_j} \langle A w_{n_j}, \ z - y_{n_j} \rangle \\ &= \langle y_{n_j} - w_{n_j}, \ z - y_{n_j} \rangle + \lambda_{n_j} \langle A w_{n_j}, \ z - w_{n_j} \rangle + \lambda_{n_j} \langle A w_{n_j}, \ w_{n_j} - y_{n_j} \rangle \\ &\leq \langle y_{n_j} - w_{n_j}, \ z - y_{n_j} \rangle + \lambda_{n_j} \langle A z, \ z - w_{n_j} \rangle + \lambda_{n_j} \langle A w_{n_j}, \ w_{n_j} - y_{n_j} \rangle. \end{split}$$

Letting $j \to \infty$, and since $\lim_{j \to \infty} ||y_{n_j} - w_{n_j}|| = 0$, we have

$$\langle Az, z - \hat{x} \rangle \ge 0, \quad \forall z \in \mathcal{C}.$$

Applying Lemma 3.2.2, we have that $\hat{x} \in VI(\mathcal{C}, A)$.

Lemma 3.3.8. Let $\{x_n\}$ be a sequence generated by Algorithm 3.3.2 under Assumption 3.3.1. Then,

$$\begin{split} \|x_{n+1} - p\|^2 &\leq (1 - \eta_n) \|x_n - p\|^2 + \eta_n \Big[\frac{\theta_n \gamma_2^2}{2(\gamma_2 - \gamma_1 \rho)} M_3 + \frac{3M_2(1 - \theta_n \gamma_2)^2}{2(\gamma_2 - \gamma_1 \rho)} \frac{\alpha_n}{\theta_n} ||x_n - x_{n-1}|| \\ &+ \frac{1}{(\gamma_2 - \gamma_1 \rho)} \langle \gamma_1 f p - D p, x_{n+1} - p \rangle \Big] - \frac{(1 - \theta_n \gamma_2)^2}{(1 - \theta_n \gamma_1 \rho)} \Big[\left(1 - \frac{\lambda_n^2 \mu^2}{\lambda_{n+1}^2} \right) \|y_n - w_n\|^2 \\ &+ \beta_n (1 - \beta_n) \|W_n z_n - z_n\|^2 \Big], \end{split}$$
where $\eta_n = \frac{2\theta_n (\gamma_2 - \gamma_1 \rho)}{(1 - \theta_n \gamma_1 \rho)}.$

Proof. Let $p \in \Gamma$. Then from **Step 2**, by applying the Cauchy-Schwartz inequality and Lemma 2.1.1, we obtain

$$||w_{n} - p||^{2} = ||x_{n} + \alpha_{n}(x_{n} - x_{n-1}) - p||^{2}$$

$$= ||x_{n} - p||^{2} + \alpha_{n}^{2}||x_{n} - x_{n-1}||^{2} + 2\alpha_{n}\langle x_{n} - p, x_{n} - x_{n-1}\rangle$$

$$\leq ||x_{n} - p||^{2} + \alpha_{n}^{2}||x_{n} - x_{n-1}||^{2} + 2\alpha_{n}||x_{n} - x_{n-1}||||x_{n} - p||$$

$$= ||x_{n} - p||^{2} + \alpha_{n}||x_{n} - x_{n-1}||(\alpha_{n}||x_{n} - x_{n-1}|| + 2||x_{n} - p||)$$

$$\leq ||x_{n} - p||^{2} + 3M_{2}\alpha_{n}||x_{n} - x_{n-1}||$$

$$= ||x_{n} - p||^{2} + 3M_{2}\theta_{n}\frac{\alpha_{n}}{\theta_{n}}||x_{n} - x_{n-1}||, \qquad (3.3.12)$$

where $M_2 := \sup_{n \in \mathbb{N}} \{ ||x_n - p||, \alpha_n ||x_n - x_{n-1}|| \} > 0.$

Now, applying the last inequality Lemma 2.1.1 and (3.3.6), we have

$$\begin{aligned} \|T_{n}z_{n}-p\|^{2} &= \|(1-\beta_{n})(z_{n}-p)+\beta_{n}(W_{n}z_{n}-p)\|^{2} \\ &\leq (1-\beta_{n})\|z_{n}-p\|^{2}+\beta_{n}\|W_{n}z_{n}-p\|^{2}-\beta_{n}(1-\beta_{n})\|W_{n}z_{n}-z_{n}\|^{2} \\ &\leq (1-\beta_{n})\|z_{n}-p\|^{2}+\beta_{n}\|z_{n}-p\|^{2}-\beta_{n}(1-\beta_{n})\|W_{n}z_{n}-z_{n}\|^{2} \\ &= \|z_{n}-p\|^{2}-\beta_{n}(1-\beta_{n})\|W_{n}z_{n}-z_{n}\|^{2} \\ &\leq \|w_{n}-p\|^{2}-\left(1-\frac{\lambda_{n}^{2}\mu^{2}}{\lambda_{n+1}^{2}}\right)\|y_{n}-w_{n}\|^{2}-\beta_{n}(1-\beta_{n})\|W_{n}z_{n}-z_{n}\|^{2} \\ &\leq \||x_{n}-p\||^{2}+3M_{2}\theta_{n}\frac{\alpha_{n}}{\theta_{n}}\||x_{n}-x_{n-1}\||-\left(1-\frac{\lambda_{n}^{2}\mu^{2}}{\lambda_{n+1}^{2}}\right)\|y_{n}-w_{n}\|^{2} \\ &-\beta_{n}(1-\beta_{n})\|W_{n}z_{n}-z_{n}\|^{2}. \end{aligned}$$

$$(3.3.13)$$

Also, by applying Lemma 2.1.1 and (3.3.13), we have

$$\begin{split} \|x_{n+1} - p\|^2 &= \|\theta_n \gamma_1 f x_n + (1 - \theta_n D) T_n z_n - p\|^2 \\ &= \|\theta_n (\gamma_1 f x_n - Dp) + (1 - \theta_n D) (T_n z_n - p)\|^2 \\ &\leq (1 - \theta_n \gamma_2)^2 \|T_n z_n - p\|^2 + 2\theta_n \gamma_1 f x_n - Dp, x_{n+1} - p) \\ &\leq (1 - \theta_n \gamma_2)^2 \|T_n z_n - p\|^2 + 2\theta_n \gamma_1 \langle f x_n - fp, x_{n+1} - p \rangle \\ &+ 2\theta_n \langle \gamma_1 fp - Dp, x_{n+1} - p \rangle \\ &\leq (1 - \theta_n \gamma_2)^2 \left[\|x_n - p\|^2 + 3M_2 \theta_n \frac{\alpha_n}{\theta_n} \|x_n - x_{n-1}\| \right] \\ &- \left(1 - \frac{\lambda_n^2 \mu^2}{\lambda_{n+1}^2} \right) \|y_n - w_n\|^2 - \beta_n (1 - \beta_n) \|W_n z_n - z_n\|^2 \right] \\ &+ 2\theta_n \langle \gamma_1 f x_n - fp, x_{n+1} - p \rangle + 2\theta_n \langle \gamma_1 fp - Dp, x_{n+1} - p \rangle \\ &\leq (1 - \theta_n \gamma_2)^2 \|x_n - p\|^2 + 3M_2 (1 - \theta_n \gamma_2)^2 \theta_n \frac{\alpha_n}{\theta_n} \|x_n - x_{n-1}\| \\ &- (1 - \theta_n \gamma_2)^2 \left(1 - \frac{\lambda_n^2 \mu^2}{\lambda_{n+1}^2} \right) \|y_n - w_n\|^2 - (1 - \theta_n \gamma_2)^2 \beta_n (1 - \beta_n) \|W_n z_n - z_n\|^2 \\ &+ 2\theta_n \gamma_1 \langle f x_n - fp, x_{n+1} - p \rangle + 2\theta_n \langle \gamma_1 fp - Dp, x_{n+1} - p \rangle \\ &\leq (1 - \theta_n \gamma_2)^2 \left(1 - \frac{\lambda_n^2 \mu^2}{\lambda_{n+1}^2} \right) \|y_n - w_n\|^2 \\ &- (1 - \theta_n \gamma_2)^2 (1 - \frac{\lambda_n^2 \mu^2}{\lambda_{n+1}^2} \right) \|y_n - w_n\|^2 \\ &- (1 - \theta_n \gamma_2)^2 \beta_n (1 - \beta_n) \|W_n z_n - z_n\|^2 + 2\theta_n \gamma_1 \rho \|x_n - p\| \|x_{n+1} - p\| \\ &+ 2\theta_n \langle \gamma_1 fp - Dp, x_{n+1} - p \rangle \\ &\leq (1 - \theta_n \gamma_2)^2 \|x_n - p\|^2 + 3M_2 (1 - \theta_n \gamma_2)^2 \theta_n \frac{\alpha_n}{\theta_n} \|x_n - x_{n-1}\| \\ &- (1 - \theta_n \gamma_2)^2 \|x_n - p\|^2 + 3M_2 (1 - \theta_n \gamma_2)^2 \theta_n \frac{\alpha_n}{\theta_n} \|x_n - x_{n-1}\| \\ &- (1 - \theta_n \gamma_2)^2 \|x_n - p\|^2 + 3M_2 (1 - \theta_n \gamma_2)^2 \theta_n \frac{\alpha_n}{\theta_n} \|x_n - x_{n-1}\| \\ &- (1 - \theta_n \gamma_2)^2 \|x_n - p\|^2 + 3M_2 (1 - \theta_n \gamma_2)^2 \theta_n \frac{\alpha_n}{\theta_n} \|x_n - x_{n-1}\| \\ &- (1 - \theta_n \gamma_2)^2 \left[\left(1 - \frac{\lambda_n^2 \mu^2}{\lambda_{n+1}^2} \right) \|y_n - w_n\|^2 + \beta_n (1 - \beta_n) \|W_n z_n - z_n\|^2 \right] \\ &+ \theta_n \gamma_1 \rho \Big(\|x_n - p\|^2 + \|x_{n+1} - p\|^2 \Big) + 2\theta_n \langle \gamma_1 fp - Dp, x_{n+1} - p \rangle \end{aligned}$$

$$= ((1 - \theta_n \gamma_2)^2 + \theta_n \gamma_1 \rho) \|x_n - p\|^2 + \theta_n \gamma_1 \rho \|x_{n+1} - p\|^2 + 3M_2 (1 - \theta_n \gamma_2)^2 \theta_n \frac{\alpha_n}{\theta_n} \|x_n - x_{n-1}\| + 2\theta_n \langle \gamma_1 f p - Dp, x_{n+1} - p \rangle - (1 - \theta_n \gamma_2)^2 \Big[\left(1 - \frac{\lambda_n^2 \mu^2}{\lambda_{n+1}^2}\right) \|y_n - w_n\|^2 + \beta_n (1 - \beta_n) \|W_n z_n - z_n\|^2 \Big].$$

From this, we obtain

$$\begin{split} \|x_{n+1} - p\|^2 &\leq \frac{(1 - 2\theta_n \gamma_2 + (\theta_n \gamma_2)^2 + \theta_n \gamma_1 \rho)}{(1 - \theta_n \gamma_1 \rho)} \|x_n - p\|^2 \\ &+ \frac{\theta_n}{(1 - \theta_n \gamma_1 \rho)} \Big[3M_2 (1 - \theta_n \gamma_2)^2 \frac{\alpha_n}{\theta_n} \|x_n - x_{n-1}\| + 2\langle \gamma_1 fp - Dp, x_{n+1} - p \rangle \Big] \\ &- \frac{(1 - \theta_n \gamma_2)^2}{(1 - \theta_n \gamma_1 \rho)} \Big[\left(1 - \frac{\lambda_n^2 \mu^2}{\lambda_{n+1}^2} \right) \|y_n - w_n\|^2 + \beta_n (1 - \beta_n) \|W_n z_n - z_n\|^2 \Big] \\ &= \frac{(1 - 2\theta_n \gamma_2 + \theta_n \gamma_1 \rho)}{(1 - \theta_n \gamma_1 \rho)} \Big[3M_2 (1 - \theta_n \gamma_2)^2 \frac{\alpha_n}{\theta_n} \|x_n - x_{n-1}\| + 2\langle \gamma_1 fp - Dp, x_{n+1} - p \rangle \Big] \\ &+ \frac{\theta_n}{(1 - \theta_n \gamma_1 \rho)} \Big[3M_2 (1 - \theta_n \gamma_2)^2 \frac{\alpha_n}{\theta_n} \|x_n - x_{n-1}\| + 2\langle \gamma_1 fp - Dp, x_{n+1} - p \rangle \Big] \\ &- \frac{(1 - \theta_n \gamma_2)^2}{(1 - \theta_n \gamma_1 \rho)} \Big[\left(1 - \frac{\lambda_n^2 \mu^2}{\lambda_{n+1}^2} \right) \|y_n - w_n\|^2 + \beta_n (1 - \beta_n) \|W_n z_n - z_n\|^2 \Big] \\ &\leq \Big(1 - \frac{2\theta_n (\gamma_2 - \gamma_1 \rho)}{(1 - \theta_n \gamma_1 \rho)} \Big) \Big\|x_n - p\|^2 \\ &+ \frac{2\theta_n (\gamma_2 - \gamma_1 \rho)}{(1 - \theta_n \gamma_1 \rho)} \Big[\frac{\theta_n \gamma_2^2}{(2(\gamma_2 - \gamma_1 \rho)} M_3 + \frac{3M_2 (1 - \theta_n \gamma_2)^2}{2(\gamma_2 - \gamma_1 \rho)} \frac{\alpha_n}{\theta_n} \|x_n - x_{n-1}\| \\ &+ \frac{1}{(\gamma_2 - \gamma_1 \rho)} \langle \gamma_1 fp - Dp, x_{n+1} - p \rangle \Big] \\ &- \frac{(1 - \theta_n \gamma_2)^2}{(1 - \theta_n \gamma_1 \rho)} \Big[\Big(1 - \frac{\lambda_n^2 \mu^2}{\lambda_{n+1}^2} \Big) \|y_n - w_n\|^2 + \beta_n (1 - \beta_n) \|W_n z_n - z_n\|^2 \Big], \end{split}$$

where $M_3 = \sup\{||x_n - p||^2 : n \in \mathbb{N}\}$ and thus we obtain the desired conclusion.

We are now in the position to give the main theorem for Algorithm 3.3.2.

Theorem 3.3.9. Let $\{x_n\}$ be a sequence generated by Algorithm 3.3.2 under Assumption 3.3.1. Suppose that $\{W_n\}$ is the sequence defined by (3.2.1). Then, the sequence $\{x_n\}$ converges strongly to a point $x^* \in \Gamma$, where $x^* = P_{\Gamma}(I - D + \gamma_1 f)x^*$ is a solution of the variational inequality

$$\langle (D - \gamma_1 f) x^*, x^* - x \rangle \leq 0, \ \forall x \in \Gamma.$$

Proof. Since $x^* = P_{\Gamma}(I - D + \gamma_1 f)x^*$, we obtain from Lemma 3.3.8 that

$$\begin{aligned} \|x_{n+1} - x^*\|^2 &\leq (1 - \eta_n) \|x_n - x^*\|^2 \\ &+ \eta_n \Big[\frac{\theta_n \gamma_2^2}{2(\gamma_2 - \gamma_1 \rho)} M_3 + \frac{3M_2(1 - \theta_n \gamma_2)^2}{2(\gamma_2 - \gamma_1 \rho)} \frac{\alpha_n}{\theta_n} ||x_n - x_{n-1}|| \\ &+ \frac{1}{(\gamma_2 - \gamma_1 \rho)} \langle \gamma_1 f x^* - D x^*, x_{n+1} - x^* \rangle \Big]. \end{aligned}$$
(3.3.14)

Now, we claim that the sequence $\{\|x_n - x^*\|\}$ converges to zero. To show this, by Lemma 2.5.36 it suffices to show that $\limsup_{k \to \infty} \langle \gamma_1 f x^* - D x^*, x_{n_k+1} - x^* \rangle \leq 0$ for every subsequence

 $\{\|x_{n_k} - x^*\|\}$ of $\{\|x_n - x^*\|\}$ satisfying

$$\liminf_{k \to \infty} \left(\|x_{n_k+1} - x^*\| - \|x_{n_k} - x^*\| \right) \ge 0.$$
(3.3.15)

Suppose that $\{\|x_{n_k} - x^*\|\}$ is a subsequence of $\{\|x_n - x^*\|\}$ such that (3.3.15) holds. From Lemma 3.3.8, we obtain

$$\begin{aligned} \frac{(1-\theta_{n_k}\gamma_2)^2}{(1-\theta_{n_k}\gamma_1\rho)} \left(1-\frac{\lambda_{n_k}^2\mu^2}{\lambda_{n_k+1}^2}\right) \|y_{n_k} - w_{n_k}\|^2 &\leq (1-\eta_{n_k})\|x_{n_k} - x^*\|^2 - \|x_{n_k+1} - x^*\|^2 \\ &+ \eta_{n_k} \Big[\frac{\theta_{n_k}\gamma_2^2}{2(\gamma_2 - \gamma_1\rho)}M_3 \\ &+ \frac{3M_2(1-\theta_{n_k}\gamma_2)^2}{2(\gamma_2 - \gamma_1\rho)}\frac{\alpha_n}{\theta_{n_k}}\|x_{n_k} - x_{n_k-1}\| \\ &+ \frac{1}{(\gamma_2 - \gamma_1\rho)}\langle\gamma_1 f x^* - D x^*, x_{n_k+1} - x^*\rangle\Big]. \end{aligned}$$

By (3.3.15) and the fact that $\lim_{k\to\infty} \eta_{n_k} = 0$ (since $\lim_{k\to\infty} \theta_{n_k} = 0$), we obtain

$$\frac{(1-\theta_{n_k}\gamma_2)^2}{(1-\theta_{n_k}\gamma_1\rho)} \left(1-\frac{\lambda_{n_k}^2\mu^2}{\lambda_{n_k+1}^2}\right) \|y_{n_k}-w_{n_k}\|^2 \to 0, \text{ as } k \to \infty$$

Consequently, we have

$$\lim_{k \to \infty} \|y_{n_k} - w_{n_k}\| = 0.$$
(3.3.16)

Following similar argument, from Lemma 3.3.8 we have

$$\lim_{k \to \infty} \|W_{n_k} z_{n_k} - z_{n_k}\| = 0.$$
(3.3.17)

From the definition of z_{n_k} in **Step 4** and (3.3.16), we have

$$\begin{aligned} \|z_{n_{k}} - w_{n_{k}}\| &= \|y_{n_{k}} - \lambda_{n_{k}}(Ay_{n_{k}} - Aw_{n_{k}}) - w_{n_{k}}\| \\ &\leq \|y_{n_{k}} - w_{n_{k}}\| + \lambda_{n_{k}}\|Ay_{n_{k}} - Aw_{n_{k}}\| \\ &\leq \|y_{n_{k}} - w_{n_{k}}\| + \lambda_{n_{k}}\frac{\mu}{\lambda_{n_{k}+1}}\|w_{n_{k}} - y_{n_{k}}\| \\ &= \left(1 + \frac{\lambda_{n_{k}}\mu}{\lambda_{n_{k}+1}}\right)\|y_{n_{k}} - w_{n_{k}}\| \to 0, \text{ as } k \to \infty, \end{aligned}$$

which implies that

$$\lim_{k \to \infty} \|z_{n_k} - w_{n_k}\| = 0.$$
(3.3.18)

From (3.3.16) and (3.3.18) we have

$$\lim_{k \to \infty} \|y_{n_k} - z_{n_k}\| = 0.$$
(3.3.19)

Also, from (3.3.17), (3.3.18) and (3.3.19), we get

$$\lim_{k \to \infty} \|w_{n_k} - y_{n_k}\| = 0 \qquad \lim_{k \to \infty} \|W_{n_k} z_{n_k} - w_{n_k}\| = 0, \qquad \lim_{k \to \infty} \|W_{n_k} z_{n_k} - y_{n_k}\| = 0.$$
(3.3.20)

Now, from Step 2 and by Remark 3.3.3, we get

$$\lim_{k \to \infty} \|w_{n_k} - x_{n_k}\| = \lim_{k \to \infty} \alpha_{n_k} \|x_{n_k} - x_{n_k-1}\| = 0.$$
(3.3.21)

From (3.3.18), (3.3.20) and (3.3.21), we obtain

$$\lim_{k \to \infty} \|z_{n_k} - x_{n_k}\| = 0, \qquad \lim_{k \to \infty} \|W_{n_k} z_{n_k} - x_{n_k}\| = 0.$$
(3.3.22)

By applying (3.3.17), we have

$$\|T_{n_k} z_{n_k} - z_{n_k}\| = \|(1 - \beta_{n_k}) z_{n_k} + \beta_{n_k} W_{n_k} z_{n_k} - z_{n_k}\|$$

$$\leq (1 - \beta_{n_k}) \|z_{n_k} - z_{n_k}\| + \beta_{n_k} \|W_{n_k} z_{n_k} - z_{n_k}\| \to 0, \quad k \to \infty.$$
(3.3.23)

Now, by using (3.3.18), (3.3.21) and (3.3.23) we obtain

$$\lim_{k \to \infty} ||T_{n_k} z_{n_k} - w_{n_k}|| = 0, \qquad \lim_{k \to \infty} ||T_{n_k} z_{n_k} - x_{n_k}|| = 0.$$
(3.3.24)

Consequently, by applying the fact that $\lim_{k\to\infty}\theta_{n_k}=0$ we get

$$\begin{aligned} \|x_{n_{k}+1} - x_{n_{k}}\| &= \|\theta_{n_{k}}\gamma_{1}fx_{n_{k}} + (1 - \theta_{n_{k}}D)T_{n_{k}}z_{n_{k}} - x_{n_{k}}\| \\ &= \|(\theta_{n_{k}}\gamma_{1}fx_{n_{k}} - \theta_{n_{k}}Dx_{n_{k}}) + (1 - \theta_{n_{k}}D)(T_{n_{k}}z_{n_{k}} - x_{n_{k}})\| \\ &\leq \theta_{n_{k}}\|\gamma_{1}fx_{n_{k}} - Dx_{n_{k}}\| + (1 - \theta_{n_{k}}\gamma_{2})\|T_{n_{k}}z_{n_{k}} - x_{n_{k}}\| \to 0, \ k \to \infty. \end{aligned}$$

Hence

$$\lim_{k \to \infty} \|x_{n_k+1} - x_{n_k}\| = 0.$$
(3.3.25)

Now, we show that $w_{\omega}(x_n) \subset \bigcap_{i=1}^{\infty} F(S_i) = F(W)$. Let $z \in w_{\omega}(x_n)$ and suppose that $z \notin F(W)$, that is, $Wz \neq z$. From (3.3.22), we have that $w_{\omega}(x_n) = w_{\omega}(z_n)$ and by Lemma 2.1.15 we have

$$\begin{aligned} \liminf_{k \to \infty} ||z_{n_k} - z|| &< \liminf_{k \to \infty} |z_{n_k} - Wz||| \\ &\leq \liminf_{k \to \infty} \{ ||z_{n_k} - Wz_{n_k}|| + ||Wz_{n_k} - Wz|| \} \\ &\leq \liminf_{k \to \infty} \{ ||z_{n_k} - Wz_{n_k}|| + ||z_{n_k} - z|| \}. \end{aligned}$$
(3.3.26)

Since $x_{n_k} \in K$ for all $k \ge 1$ and $w_{\omega}(x_n) = w_{\omega}(z_n)$, we obtain

$$||Wz_{n_k} - z_{n_k}|| \le ||Wz_{n_k} - W_{n_k}z_{n_k}|| + ||W_{n_k}z_{n_k} - z_{n_k}||$$

$$\le \sup_{x \in K} ||Wx - W_{n_k}x|| + ||W_{n_k}z_{n_k} - z_{n_k}||.$$

By applying Lemma 3.2.5 and (3.3.17), we have $\lim_{k\to\infty} ||Wz_{n_k} - z_{n_k}|| = 0$. Combining this with (3.3.26) yields

$$\liminf_{k \to \infty} ||z_{n_k} - z|| < \liminf_{k \to \infty} ||z_{n_k} - z||,$$

which is a contradiction. Hence, we have $z \in F(W) = \bigcap_{i=1}^{\infty} F(S_i)$, i.e., $w_{\omega}(x_n) \subset F(W) = \bigcap_{i=1}^{\infty} F(S_i)$.

Next, we show that $w_{\omega}(x_n) \subset VI(\mathcal{C}, A)$. By invoking Lemma 3.3.7, it follows from (3.3.20) that $z \in VI(\mathcal{C}, A)$. Thus, $w_{\omega}(x_n) \subset \Gamma$.

From the fact that $\lim_{k\to\infty} ||x_{n_k} - z_{n_k}|| = 0$, we have that $w_{\omega}\{x_n\} = w_{\omega}\{z_n\}$. By Lemma 3.3.6, we have that $\{x_{n_k}\}$ is bounded which implies that there exists a subsequence $\{x_{n_{k_j}}\}$ of $\{x_{n_k}\}$ such that $x_{n_{k_j}} \rightarrow \bar{x}$ and

$$\lim_{j \to \infty} \langle \gamma_1 f x^* - Dx^*, x_{n_{kj}} - x^* \rangle = \limsup_{k \to \infty} \langle \gamma_1 f x^* - Dx^*, x_{n_k} - x^* \rangle$$
$$= \limsup_{k \to \infty} \langle \gamma_1 f x^* - Dx^*, z_{n_k} - x^* \rangle.$$

Since $x^* = P_{\Gamma}(I - D + \gamma_1 f)x^*$, we have that

$$\lim_{k \to \infty} \sup \langle \gamma_1 f x^* - D x^*, x_{n_k} - x^* \rangle = \lim_{j \to \infty} \langle \gamma_1 f x^* - D x^*, x_{n_{k_j}} - x^* \rangle$$
$$= \langle \gamma_1 f x^* - D x^*, \bar{x} - x^* \rangle$$
$$\leq 0. \tag{3.3.27}$$

From (3.3.25) and (3.3.27), we have

$$\limsup_{k \to \infty} \langle \gamma_1 f x^* - Dx^*, x_{n_k+1} - x^* \rangle = \limsup_{k \to \infty} \langle \gamma_1 f x^* - Dx^*, x_{n_k} - x^* \rangle$$
$$= \langle \gamma_1 f x^* - Dx^*, \bar{x} - x^* \rangle$$
$$\leq 0. \tag{3.3.28}$$

By applying Lemma 2.5.36 to (3.3.14) and using (3.3.28) together with the fact that $\lim_{n \to \infty} \theta_n = 0$ and $\lim_{n \to \infty} \frac{\alpha_n}{\theta_n} ||x_n - x_{n-1}|| = 0$, we have that $\lim_{n \to \infty} ||x_n - x^*|| = 0$. Therefore, $\{x_n\}$ converges strongly to x^* .

Next, we propose our second algorithm which is a slight modification of Algorithm (3.3.2). Algorithm 3.3.3. Inertial method with adaptive step size strategy.

Step 0: Choose sequences $\{\theta_n\}_{n=1}^{\infty}$ and $\{\tau_n\}_{n=1}^{\infty}$ such that condition (B5) holds and let $\lambda_1 > 0, \ \mu \in (0,1), \ \alpha \ge 3$ and $x_0, x_1 \in \mathcal{H}$ be given arbitrarily. Set n := 1.

Step 1: Given the iterates x_{n-1} and x_n for each $n \ge 1$, choose α_n such that $0 \le \alpha_n \le \overline{\alpha}_n$, where

$$\bar{\alpha}_{n} := \begin{cases} \min\left\{\frac{n-1}{n+\alpha-1}, \frac{\tau_{n}}{\|x_{n}-x_{n-1}\|}\right\}, & \text{if } x_{n} \neq x_{n-1} \\ \frac{n-1}{n+\alpha-1}, & \text{otherwise.} \end{cases}$$
(3.3.29)

Step 2: Compute

$$w_n = x_n + \alpha_n (x_n - x_{n-1}).$$

Step 3: Construct the halfspace

$$\mathcal{C}_n = \{ w \in \mathcal{H} \ | c(w_n) + \langle \xi_n, w - w_n \rangle \le 0 \}$$

and compute

$$y_n = P_{\mathcal{C}_n}(w_n - \lambda_n A w_n).$$

If $y_n = w_n (Aw_n = 0)$, then set $w_n = z_n$ and go to Step 5. Else go to Step 4. Step 4: compute

$$z_n = y_n - \lambda_n (Ay_n - Aw_n).$$

Step 5: Compute

$$x_{n+1} = \theta_n \gamma_1 f w_n + (I - \theta_n D) T_n z_n$$

where

$$T_n = (1 - \beta_n)I + \beta_n W_n$$
 and W_n is the mapping defined by (3.2.1)

Step 6: Compute

$$\lambda_{n+1} = \begin{cases} \min\left\{\frac{\mu||w_n - y_n||}{||Aw_n - Ay_n||}, \lambda_n\right\}, & \text{if } Aw_n \neq Ay_n\\ \lambda_n, & \text{otherwise.} \end{cases}$$
(3.3.30)

Set n := n + 1 and go back to Step 1.

Theorem 3.3.10. Let $\{x_n\}$ be a sequence generated by Algorithm 3.3.3 under Assumption 3.3.1. Suppose that $\{W_n\}$ is the sequence defined by (3.2.1). Then, the sequence $\{x_n\}$ converges strongly to a point $x^* \in \Gamma$, where $x^* = P_{\Gamma}(I - D + \gamma_1 f)x^*$ is a solution of the variational inequality

$$\langle (D - \gamma_1 f) x^*, x^* - x \rangle \leq 0, \ \forall x \in \Gamma.$$

Proof. The proof of this result follows similar argument with the proof of the result of Theorem (3.3.9).

Taking $\gamma_1 = 1$ and D = I in Theorem 3.3.9 (where I is the identity mapping), we obtain the following:

Algorithm 3.3.4. Inertial method with adaptive step size strategy.

Step 0: Choose sequences $\{\theta_n\}_{n=1}^{\infty}$ and $\{\tau_n\}_{n=1}^{\infty}$ such that condition (B5) holds and let $\lambda_1 > 0, \mu \in (0, 1), \alpha \ge 3$ and $x_0, x_1 \in \mathcal{H}$ be given arbitrarily. Set n := 1.

Step 1: Given the iterates x_{n-1} and x_n for each $n \ge 1$, choose α_n such that $0 \le \alpha_n \le \overline{\alpha}_n$, where

$$\bar{\alpha}_n := \begin{cases} \min\left\{\frac{n-1}{n+\alpha-1}, \frac{\tau_n}{\|x_n - x_{n-1}\|}\right\}, & \text{if } x_n \neq x_{n-1}\\ \frac{n-1}{n+\alpha-1}, & \text{otherwise.} \end{cases}$$

Step 2: Compute

$$w_n = x_n + \alpha_n (x_n - x_{n-1}).$$

Step 3: Construct the halfspace

$$\mathcal{C}_n = \{ w \in \mathcal{H} \ | c(w_n) + \langle \xi_n, w - w_n \rangle \le 0 \}$$

and compute

$$y_n = P_{\mathcal{C}_n}(w_n - \lambda_n A w_n).$$

If $y_n = w_n$ ($Aw_n = 0$), then set $w_n = z_n$ and go to Step 5. Else go to Step 4. Step 4: compute

$$z_n = y_n - \lambda_n (Ay_n - Aw_n).$$

Step 5: Compute

$$x_{n+1} = \theta_n f x_n + (I - \theta_n) T_n z_n,$$

where

$$T_n = (1 - \beta_n)I + \beta_n W_n$$
 and W_n is the mapping defined by (3.2.1)

Step 6: Compute

$$\lambda_{n+1} = \begin{cases} \min\left\{\frac{\mu||w_n - y_n||}{||Aw_n - Ay_n||}, \lambda_n\right\}, & \text{if } Aw_n \neq Ay_n\\ \lambda_n, & \text{otherwise.} \end{cases}$$

Set n := n + 1 and go back to Step 1.

Corollary 3.3.5. Let $\{x_n\}$ be a sequence generated by Algorithm 3.3.4 under Assumption 3.3.1. Suppose that $\{W_n\}$ is the sequence defined by (3.2.1). Then, the sequence $\{x_n\}$ converges strongly to a point $x^* \in \Gamma$, where $x^* = P_{\Gamma}(f)x^*$ is a solution of the variational inequality

$$\langle (I-f)x^*, x^*-x \rangle \le 0, \ \forall x \in \Gamma.$$

Taking $\gamma_1 = 1$, D = I (where I is the identity mapping) and $S_n = S$ for all $n \ge 1$ in Theorem 3.3.9, we obtain the following:

Algorithm 3.3.6. Inertial method with adaptive step size strategy.

Step 0: Choose sequences $\{\theta_n\}_{n=1}^{\infty}$ and $\{\tau_n\}_{n=1}^{\infty}$ such that condition (B5) holds and let $\lambda_1 > 0, \mu \in (0, 1), \alpha \ge 3$ and $x_0, x_1 \in \mathcal{H}$ be given arbitrarily. Set n := 1.

Step 1: Given the iterates x_{n-1} and x_n for each $n \ge 1$, choose α_n such that $0 \le \alpha_n \le \overline{\alpha}_n$, where

$$\bar{\alpha}_n := \begin{cases} \min\left\{\frac{n-1}{n+\alpha-1}, \frac{\tau_n}{\|x_n - x_{n-1}\|}\right\}, & \text{if } x_n \neq x_{n-1}\\ \frac{n-1}{n+\alpha-1}, & \text{otherwise.} \end{cases}$$

Step 2: Compute

$$w_n = x_n + \alpha_n (x_n - x_{n-1}).$$

Step 3: Construct the halfspace

$$\mathcal{C}_n = \{ w \in \mathcal{H} \ | c(w_n) + \langle \xi_n, w - w_n \rangle \le 0 \}$$

and compute

$$y_n = P_{\mathcal{C}_n}(w_n - \lambda_n A w_n).$$

If $y_n = w_n (Aw_n = 0)$, then set $w_n = z_n$ and go to Step 5. Else go to Step 4. Step 4: compute

$$z_n = y_n - \lambda_n (Ay_n - Aw_n).$$

Step 5: Compute

$$x_{n+1} = \theta_n f x_n + (I - \theta_n) T_n z_n$$

where

$$T_n = (1 - \beta_n)I + \beta_n S.$$

Step 6: Compute

$$\lambda_{n+1} = \begin{cases} \min\left\{\frac{\mu||w_n - y_n||}{||Aw_n - Ay_n||}, \lambda_n\right\}, & \text{if } Aw_n \neq Ay_n\\ \lambda_n, & \text{otherwise.} \end{cases}$$

Set n := n + 1 and go back to Step 1.

Corollary 3.3.7. Let $\{x_n\}$ be a sequence generated by Algorithm 3.3.6 under Assumption 3.3.1. Then the sequence $\{x_n\}$ converges strongly to a point $x^* \in \Gamma$, where $x^* = P_{\Gamma}(f)x^*$ is a solution of the variational inequality

$$\langle (I-f)x^*, x^*-x \rangle \leq 0, \ \forall x \in \Gamma.$$

3.3.3 Applications

The zero point problem

In this section, we apply our results to finding a common solution of VIP and ZPP for an infinite family of maximal monotone operators.

Let $F : \mathcal{H} \to \mathcal{H}$ be a single-valued nonlinear mapping and $B : \mathcal{H} \to 2^{\mathcal{H}}$ be a multivalued mapping. The problem of finding a zero of the sum of two monotone operators, which is formulated as the following MIP is to find a point $x \in \mathcal{H}$ such that

$$0 \in (F+B)x.$$
 (3.3.31)

This problem includes, as special cases, convex programming, VIP, SFP and MP. More precisely, some concrete problems in machine learning, image processing and linear inverse problem can be modeled mathematically as this form, see [86, 92, 99, 110]. We denote the zero point set $\{x \in \mathcal{H} : 0 \in (F+B)x\}$ of F+B by $(F+B)^{-1}0$. **Definition 3.3.11.** Let $B : \mathcal{H} \to 2^{\mathcal{H}}$ be a multivalued mapping. The effective domain of B denoted by D(B) is given as $D(B) = \{x \in \mathcal{H} : Bx \neq \emptyset\}.$

Definition 3.3.12. Let $B : \mathcal{H} \to 2^{\mathcal{H}}$ be a multi-valued operator on \mathcal{H} . Then

(1) the graph G(B) is defined by

$$G(B) := \{ (x, u) \in \mathcal{H} \times \mathcal{H} : u \in B(x) \};$$

- (2) the operator B is said to be monotone if $\langle x-y, u-v \rangle \ge 0$ for all $x, y \in D(B), u \in Bx$, and $v \in By$;
- (3) A monotone operator B on \mathcal{H} is said to be maximal if its graph is not properly contained in the graph of any other monotone operator on \mathcal{H} ;
- (4) For a maximal monotone multivalued mapping B on \mathcal{H} and r > 0, the operator

$$J_r^B := (I + rB)^{-1} : \mathcal{H} \to D(B)$$

is called the resolvent of B.

Remark 3.3.13. In [101], it was shown that $F(J_r^B) = B^{-1}0 \equiv \{x \in \mathcal{H} : 0 \in Bx\}$ for all r > 0 and J_r^B is singled-valued and firmly nonexpansive, that is,

$$||J_r^B x - J_r^B y|| \le \langle J_r^B x - J_r^B y, x - y \rangle, \text{ for all } x, y \in \mathcal{H}.$$

The following lemma will also be employed in establishing our results in this section.

Lemma 3.3.14. [22] Let C be a nonempty closed convex subset of $\mathcal{H}, G : \mathcal{H} \to \mathcal{H}$ be a mapping and $B : \mathcal{H} \to 2^{\mathcal{H}}$ be a maximal monotone operator. Then $F(J_r^B(I - rG)) = (G + B)^{-1}(0)$.

Now, we have the following results.

Theorem 3.3.15. Let \mathcal{H} be a Hilbert space and suppose that $\{W_n\}$ is the sequence defined by (3.2.1). Let $B_i : \mathcal{H} \to 2^{\mathcal{H}}$ be an infinite family of maximal monotone mappings with $D(B_i) \neq \emptyset$ and $J_{r_i}^{B_i}$ is the resolvent of B_i for each $r_i > 0$. Suppose that $\{x_n\}$ is a sequence generated by Algorithm 3.3.2 such that Assumption 3.3.1 is satisfied. Then the sequence $\{x_n\}$ converges strongly to a point $\hat{x} \in \Gamma = VI(\mathcal{C}, A) \bigcap \bigcap_{i=1}^{\infty} (B_i^{-1}0) \neq \emptyset$, where $\hat{x} = P_{\Gamma}(I - D + \gamma f)(\hat{x})$ is a solution of the variational inequality

$$\langle (D - \gamma f)\hat{x}, \hat{x} - x \rangle \leq 0 \text{ for all } x \in \Gamma.$$

Proof. Since $J_{r_i}^{B_i}$ is nonexpansive and $F(J_{r_i}^{B_i}) = B_i^{-1}0$, then the result follows from Theorem 3.3.9 by taking $J_{r_i}^{B_i} = S_i$ in Definition 3.2.3.

Theorem 3.3.16. Let \mathcal{H} be a Hilbert space and suppose that $\{W_n\}$ is the sequence defined by (3.2.1). Let $B_i : \mathcal{H} \to 2^{\mathcal{H}}$ be an infinite family of maximal monotone mappings with $D(B_i) \neq \emptyset, J_{r_i}^{B_i}$ is the resolvent of B_i for each $r_i \in (0, 2\delta_i)$ and $F_i : \mathcal{H} \to \mathcal{H}$ be an infinite family of δ_i -inverse strongly monotone mappings. Suppose that $\{x_n\}$ is a sequence generated by Algorithm 3.3.2 such that Assumption 3.3.1 is satisfied. Then the sequence $\{x_n\}$ converges strongly to a point $\hat{x} \in \Gamma = VI(\mathcal{C}, A) \bigcap \bigcap_{i=1}^{\infty} (B_i + F_i)^{-1} 0 \neq \emptyset$, where $\hat{x} = P_{\Gamma}(I - D + \gamma f)(\hat{x})$ is a solution of the variational inequality

$$\langle (D - \gamma f)\hat{x}, \hat{x} - x \rangle \leq 0 \text{ for all } x \in \Gamma.$$

Proof. Since F_i is δ_i -inverse strongly monotone, $I - r_i F_i$ is nonexpansive. By the nonexpansiveness of $J_{r_i}^{B_i}$, it follows that $J_{r_i}^{B_i}(I - r_i F_i)$ is also nonexpansive. The proof follows from Theorem 3.3.9 by applying Lemma 3.3.14 and taking $J_{r_i}^{B_i}(I - r_i G_i) = S_i$ in Definition 3.2.3.

3.3.4 Numerical experiments

In this section, using some test examples, we discuss the numerical behavior of Algorithm 3.3.2 as well as compare it with Appendix 3.3.19 proposed by Thong and Hieu [249], Appendix 3.3.20 proposed by Yang and Liu [266], the method of Shehu and Iyiola [223, Algorithm 1] (see Appendix 3.3.21) and the method of Fan and Qin [102, Algorithm 5] (see Appendix 3.3.22). We perform all implementations using Matlab 2016 (b), installed on a personal computer with Intel(R) Core(TM) i5-2600 CPU@2.30GHz and 8.00 Gb-RAM running on Windows 10 operating system. In Tables 3.3.1-3.3.2, "No. of Iter." means the number of iterations.

In our computations, we define $h(x) = ||x||^2$ and choose $\theta_n = \frac{1}{n+1}$, $\gamma_1 = 0.1$, $\beta_n = \frac{1}{2(1+\frac{1}{n})}$, $\tau_n = \frac{1}{(n+1)^3}$, $\lambda_1 = 0.8$, and $\mu = 0.7$. Furthermore, in the implementation, we define $\text{TOL}_n := ||x_{n+1} - x_n||$ and use the stopping criterion $\text{TOL}_n < 10^{-2}$ for the iterative processes.

Example 3.3.17. Let $\mathcal{H} = \mathbb{R}^m$ with standard topology, $fx = \frac{x}{6}$, Dx = x and $S_n : \mathcal{H} \to \mathcal{H}$ be a mapping defined by $S_n x := \frac{1}{5n} x \,\forall x \in \mathcal{H}$. Let $A : \mathbb{R}^m \to \mathbb{R}^m (m = 15, 30, 45, 60)$ be an operator defined as

$$Ax = Mx + q$$

and

$$M = NN^T + S^* + E$$

where N is an $m \times m$ matrix, S^* is an $m \times m$ skew-symmetric matrix, E is an $m \times m$ diagonal matrix with its diagonal entries being nonnegative and $q \in \mathbb{R}^m$. The feasible set \mathcal{C} is given by

$$\mathcal{C} = \{ x = (x_1, \cdots, x_m) \in \mathbb{R}^m : -2 \le x_i \le 5, i = 1, \cdots, m \}.$$

It is obvious that A is monotone and Lipschitz continuous with Lipschitz constant L = ||M||. The entries of N, S^{*} are generated randomly in [-2, 2] as well as the starting points



Figure 3.1: Top left: Case I; Top right: Case II; Bottom left: Case III; Bottom right: Case IV.

 x_0, x_1 and the diagonal entries of E are in (0, 2), and q is taken as the zero vector in \mathbb{R}^m . We plot the graph of $||x_{n+1} - x_n||$ against number of iterations choosing m = 15, 30, 45, 60and $\alpha = 7$. The numerical results are reported in Figure 3.1 and Table 3.3.1.

		App.	App.	App.	App.	Alg.
		3.3.19	3.3.20	3.3.21	3.3.22	3.3.2
m = 15	No. of Iter.	20	20	4	19	12
m = 30	No. of Iter.	20	20	4	19	12
m = 45	No. of Iter.	25	25	4	22	13
m = 60	No. of Iter.	13	21	4	21	13

Table 3.3.1. Numerical results for Example 3.3.17.

Example 3.3.18. Let $\mathcal{H} = L^2([0,1])$ be endowed with inner product

$$\langle x,y\rangle = \int_0^1 x(t)y(t)dt, \quad \forall \ x,y \in \mathcal{H} \text{ and norm } ||x|| := \Big(\int_0^1 |x(t)|^2 dt\Big)^{\frac{1}{2}}, \quad \forall \ x,y \in \mathcal{H}$$

Now, define $A : \mathcal{H} \to \mathcal{H}$ by

$$A(x)(t) = \max\{0, x(t)\}, \text{ for all } t \in [0, 1], x \in \mathcal{H}.$$

It is easy to see that A is monotone 1-Lipschitz continuous on \mathcal{H} . Let $f(x) = \frac{x}{2}(t)$, $Dx = \int_0^1 tx(s)dx$ and $\mathcal{C} := \{x \in \mathcal{H} : ||x|| \le 1\}$ be the unit ball. Then $VI(\mathcal{C}, A) = \{0\} \neq \emptyset$. We define the mapping $S_n : L^2([0, 1]) \to L^2([0, 1])$ by

$$S_n x(t) = \int_0^1 t^n x(s) ds, \quad t \in [0, 1].$$

Then, S_n is an infinite family of nonexpansive mappings. Indeed, we have

$$|S_n x(t) - S_n y(t)|^2 = |\int_0^1 t^n (x(s) - y(s)) ds|^2$$

$$\leq \left(\int_0^1 t^n |x(s) - y(s)| ds\right)^2$$

$$\leq \int_0^1 |x(s) - y(s)|^2 ds$$

$$= ||x - y||^2.$$

Thus, we obtain that

$$||S_n x - S_n y||^2 = \int_0^1 |S_n x(t) - S_n y(t)|^2 \le ||x - y||^2.$$

Let $\{\zeta_n\}$ be a sequence of nonnegative real numbers defined by $\zeta_n = \{\frac{n}{n+1}\}$ for all $n \in \mathbb{N}$ and W_n be a mapping generated by $\{S_n\}$ and $\{\zeta_n\}$. We consider the following cases with $\alpha = 3$ for the numerical experiments of this example. **Case 1:** Take $x_1(t) = 1 + t^2$ and $x_0(t) = t + 5$.

Case 1. Take $x_1(t) = 1 + t$ and $x_0(t) = t + 0$.

Case 2: Take $x_1(t) = t^2 + 1$ and $x_0(t) = \cos(t)$.

Case 3: Take $x_1(t) = t + 1$ and $x_0(t) = t + t^3$.

Case 4: Take $x_1(t) = t + 1$ and $x_0(t) = 2\sin(3t)$. The numerical results are reported in Figure 3.2 and Table 3.3.2.

Table 3.3.2:	Numerical	$\mathbf{results}$	for	Example	3.3.18
--------------	-----------	--------------------	-----	---------	--------

	App.	App.	App.	App.	Alg.			
	3.3.19	3.3.20	3.3.21	3.3.22	3.3.2			
No. of Iter.	14	13	9	7	5			
No. of Iter.	14	13	9	8	5			
No. of Iter.	15	13	10	8	5			
No. of Iter.	15	13	10	8	5			



Figure 3.2: Top left: Case I; Top right: Case II; Bottom left: Case III; Bottom right: Case IV.

Appendix 3.3.19. The Algorithm in [249]

Initialization: Given $\gamma > 0, l \in (0, 1), \mu \in (0, 1)$. Let $x_0 \in \mathcal{H}$ be arbitrary.

Iterative Steps: Calculate x_{n+1} as follows:

Step1. Compute

$$y_n = P_{\mathcal{C}}(x_n - \lambda_n A x_n),$$

where λ_n is chosen to be the largest $\lambda \in \{\gamma, \gamma l, \gamma l^2, \cdots\}$ satisfying

$$\lambda \|Ax_n - Ay_n\| \le \mu \|x_n - y_n\|.$$
(3.3.32)

If $x_n = y_n$, then stop and x_n is the solution of VIP (1.2.1). Otherwise,

Step 2: Compute

$$x_{n+1} = \alpha_n g(x_n) + (1 - \alpha_n) z_n,$$

where $z_n = y_n - \lambda_n (Ay_n - Ax_n)$, Set n := n + 1 and return to Step 1,

Appendix 3.3.20. The Algorithm [266]

Initialization: Given $\lambda_0 > 0, \mu \in (0, 1)$. Let $x_0 \in \mathcal{H}$ be arbitrary.

Iterative Steps: Calculate x_{n+1} as follows:

Step1. Given the current iterate x_n , compute

$$y_n = P_{\mathcal{C}}(x_n - \lambda_n A x_n),$$

If $x_n = y_n$, then stop and x_n is the solution of VIP (1.2.1). Otherwise, Step 2: Compute

$$x_{n+1} = \alpha_n g(x_n) + (1 - \alpha_n) z_n,$$

and

$$\lambda_{n+1} = \begin{cases} \min\{\frac{\mu \|x_n - y_n\|}{\|Ax_n - Ay_n\|}, \lambda_n\}, & \text{if } Ax_n - Ay_n \neq 0, \\ \lambda_n, & \text{otherwise} \end{cases}$$

where $z_n = y_n - \lambda_n (Ay_n - Ax_n)$, Set n := n + 1 and return to Step 1

Appendix 3.3.21. The Algorithm in [223].

Initialization: Given $l \in (0, 1)$, $\mu \in (0, 1)$. Let $x_0 \in \mathcal{H}$ be arbitrary.

Iterative Steps: Calculate x_{n+1} as follows:

Step1. Compute

$$y_n = P_{\mathcal{C}}(x_n - \lambda_n A x_n),$$

where $\lambda_n = l^{m_n}$, where m_n is the smallest nonnegative integer m such that

$$\lambda_n \|Ax_n - Ay_n\| \le \mu \|x_n - y_n\|.$$
(3.3.33)

If $x_n = y_n$, then stop and x_n is the solution of VIP (1.2.1). Otherwise,

Step 2: Construct

$$T_n = \{ x \in \mathcal{H} : \langle x_n - \lambda_n A x_n - y_n, x - y_n \rangle \le 0 \}$$

and compute

$$z_n = P_{T_n}(x_n - \lambda_n A y_n).$$

Step 3: Compute

$$x_{n+1} = \alpha_n f(x_n) + (1 - \alpha_n) z_n,$$

where $f: \mathcal{H} \to \mathcal{H}$ is a contraction mapping with constant $\rho \in [0, 1)$.

Set n := n + 1 and return to Step 1.

Appendix 3.3.22. The Algorithm in [102].

Initialization: Given
$$\gamma > 0, l \in (0, 1), \mu \in (0, 1)$$
. Let $x_0, x_1 \in \mathcal{H}$ be arbitrary.

Iterative Steps: Calculate x_{n+1} as follows:

Step1. Set

$$w_n = x_n + \alpha_n (x_n - x_{n-1})$$

and compute

$$y_n = P_{\mathcal{C}}(w_n - \lambda_n A x_n),$$

where λ_n is chosen to be the largest $\lambda \in \{\gamma, \gamma l, \gamma l^2, \cdots\}$ satisfying

$$\lambda \|Aw_n - Ay_n\| \le \mu \|w_n - y_n\|.$$
(3.3.34)

If $w_n = y_n$, then stop and y_n is the solution of VIP (1.2.1). Otherwise,

Step 2: Compute

$$x_{n+1} = \alpha_n f(x_n) + \gamma_n z_n + \beta_n w_n,$$

where $z_n = y_n - \lambda_n (Ay_n - Aw_n)$, $A : \mathcal{H} \to \mathcal{H}$ is monotone and Lipschitz continuous, $f : \mathcal{H} \to \mathcal{H}$ is a contraction with constant $\rho \in [0, 1)$ and $\alpha_n + \beta_n + \gamma_n = 1$.

Set n := n + 1 and return to Step 1.

3.4 On quasimonotone variational inequality problems

In this section, we present two new relaxed inertial subgradient extragradient methods for solving VIPs in a real Hilbert space. We establish the convergence of the sequence generated by these methods when the cost operator is quasimonotone and Lipschitz continuous, and when it is Lipschitz continuous without any form of monotonicity. The methods combine both the inertial and relaxation techniques in order to achieve high convergence speed. Furthermore, we present some experimental results to illustrate the profits gained from the relaxed inertial steps.

3.4.1 Proposed methods

In this section, we present our methods and discuss their features. We begin with the following assumptions under which we obtain our convergence results.

Assumption 3.4.1. Conditions on the inertial and relaxation factors:

Suppose that $\theta_n \in [0,1)$ and $\rho_n \in (0,1]$ for all $n \ge 1$ such that $\liminf_{n \to \infty} \rho_n > 0$. Assume that there exists $\epsilon \in (0,1)$ such that for n large enough,

$$\begin{aligned} & \left(2(1-\epsilon) \frac{1-\rho_n}{2\rho_n} (1-\theta_{n-1}) \ge \\ & \theta_n t_{n+1} \left(1+\theta_n + 2 \left[\frac{1-\rho_n}{2\rho_n} (1-\theta_n) - \frac{1-\rho_{n-1}}{2\rho_{n-1}} (1-\theta_{n-1}) \right]_+ \right), & \text{if } \rho_n \in (0,1), \\ & \left((1-\epsilon)(1-\theta_{n-1}) \ge \theta_n t_{n+1} \left(1+\theta_n + \left[\theta_{n-1} - \theta_n \right]_+ \right), & \text{if } \rho_n = 1. \end{aligned}$$

(3.4.1)

Assumption 3.4.2. We further make the following assumptions:

- (a) $\Gamma \neq \emptyset$,
- (b) A is Lipschitz-continuous on \mathcal{H} with constant L > 0,
- (c) A is sequentially weakly continuous on C,
- (d) A is quasimonotone on \mathcal{H} ,
- (e) The set $\{z \in \mathcal{C} : Az = 0\} \setminus \Gamma$ is a finite set.

We now present the proposed methods of this section.

When the Lipschitz constant L is known, we present the following method for solving the VIP (1.2.1).

Algorithm 3.4.3. Relaxed inertial subgradient extragradient method with fixed stepsize. Step 0: Choose sequences $\{\theta_n\}$ and $\{\rho_n\}$ such that $\theta_n \in [0,1)$ and $\rho_n \in (0,1]$ for all $n \ge 1$. Let $\lambda \in (0, \frac{1}{L})$ and $x_0, x_1 \in \mathcal{H}$ be given arbitrarily. Set n := 1. Step 1: Given the current iterates x_{n-1} and x_n $(n \ge 1)$, compute

$$w_n = x_n + \theta_n (x_n - x_{n-1})$$

and

$$y_n = P_{\mathcal{C}}(w_n - \lambda A w_n).$$

If $w_n = y_n$: STOP. Otherwise, go to **Step 2**. **Step 2**: Construct the half-space

$$T_n = \{ x \in \mathcal{H} : \langle w_n - \lambda A w_n - y_n, \ x - y_n \rangle \le 0 \}.$$

Then, compute

$$z_n = P_{T_n}(w_n - \lambda A y_n)$$

and

$$x_{n+1} = (1 - \rho_n)w_n + \rho_n z_n.$$

Set n := n + 1 and return to Step 1.

In situations where the Lipschitz constant L is not known, we present the following method with adaptive stepsize for solving the VIP (1.2.1).

Algorithm 3.4.4. Relaxed inertial subgradient extragradient method with adaptive stepsize strategy.

Step 0: Choose sequences $\{\theta_n\}$ and $\{\rho_n\}$ such that $\theta_n \in [0,1)$ and $\rho_n \in (0,1]$ for all $n \ge 1$. Let $\lambda_1 > 0$, $\mu \in (0,1)$ and $x_0, x_1 \in \mathcal{H}$ be given arbitrarily. Choose a nonnegative real sequence $\{d_n\}$ such that $\sum_{n=1}^{\infty} d_n < \infty$. Set n := 1.

Step 1: Given the current iterates x_{n-1} and x_n $(n \ge 1)$, compute

$$w_n = x_n + \theta_n (x_n - x_{n-1})$$

and

$$y_n = P_{\mathcal{C}}(w_n - \lambda_n A w_n).$$

If $w_n = y_n$: STOP. Otherwise, go to **Step 2**. **Step 2**: Construct the half-space

$$T_n = \{ x \in \mathcal{H} : \langle w_n - \lambda_n A w_n - y_n, \ x - y_n \rangle \le 0 \}.$$

Then, compute

$$z_n = P_{T_n}(w_n - \lambda_n A y_n)$$

and

$$x_{n+1} = (1 - \rho_n)w_n + \rho_n z_n,$$

where

$$\lambda_{n+1} = \begin{cases} \min\left\{\frac{\mu\left(\|w_n - y_n\|^2 + \|z_n - y_n\|^2\right)}{2\langle Aw_n - Ay_n, z_n - y_n\rangle}, \ \lambda_n + d_n\right\}, & \text{if } \langle Aw_n - Ay_n, z_n - y_n\rangle > 0, \\ \lambda_n + d_n, & \text{otherwise.} \end{cases}$$
(3.4.2)

Set n := n + 1 and return to Step 1.

Remark 3.4.1. In a special case when $\rho_n = 1$, $\forall n \geq 1$, we have that $x_{n+1} = z_n$. Thus, Algorithms 3.4.3 and 3.4.4 reduce to the inertial version of the subgradient extragradient method of Censor *et al.* [58–60]. In such case, we have the following simple criteria which guarantee Assumption 3.4.1, as well as assumption (2.5.37).

Proposition 3.4.5. Assume that $\{\theta_n\}$ is a nondecreasing sequence that satisfies $\theta_n \in [0,1) \ \forall n \geq 1 \ \text{with} \lim_{n \to \infty} \theta_n = \theta \ \text{such that the following condition holds:}$

$$1 - 3\theta > 0.$$
 (3.4.3)

Then, assumption (2.5.37) and Assumption 3.4.1 hold.

Proof. Observe that $\theta_n \leq \theta$, $\forall n \geq 1$. Thus, we have that assumption (2.5.37) is satisfied and $t_n \leq \frac{1}{1-\theta}$, $\forall n \geq 1$ (see [29]). Now, observe that $1-3\theta > 0$ implies that $(1-\theta) > \frac{\theta(1+\theta)}{1-\theta}$. This further implies that there exists $\epsilon \in (0, 1)$ such that

$$(1-\epsilon)(1-\theta) \ge \frac{\theta(1+\theta)}{1-\theta}.$$
(3.4.4)

Since $\theta_n \leq \theta$, $\forall n \geq 1$, we obtain from (3.4.4) that

$$(1-\epsilon)(1-\theta_{n-1}) \ge \frac{\theta(1+\theta)}{1-\theta} \ge \theta_n t_{n+1}(1+\theta_n), \qquad (3.4.5)$$

for some $\epsilon \in (0, 1)$. Since $\theta_{n-1} \leq \theta_n$, $\forall n \geq 1$, we obtain that

$$\theta_n t_{n+1}(1+\theta_n) = \theta_n t_{n+1}(1+\theta_n + [\theta_{n-1} - \theta_n]_+).$$

Combining this with (3.4.5), we get that Assumption 3.4.1 is satisfied.

Proposition 3.4.6. Suppose that $\theta_n \in [0,1), \forall n \ge 1$ and there exists $c \in [0,\frac{1}{2})$ such that

$$\lim_{n \to \infty} \left(\frac{1}{1 - \theta_{n+1}} - \frac{1}{1 - \theta_n} \right) = c \tag{3.4.6}$$

and

$$\liminf_{n \to \infty} (1 - \theta_n)^2 > \limsup_{n \to \infty} \frac{\theta_n (1 + \theta_n)}{1 - 2c}.$$
(3.4.7)

Then, assumption (2.5.37) and Assumption 3.4.1 hold.

Proof. It follows from Proposition 2.5.32(i) that assumption (2.5.37) holds. Now, from (3.4.7), we obtain that

$$\liminf_{n \to \infty} (1 - \theta_{n-1})^2 > \limsup_{n \to \infty} \frac{\theta_n (1 + \theta_n)}{1 - 2c}.$$
(3.4.8)

Thus, there exists $\epsilon \in (0, 1)$ sufficiently small enough such that

$$\liminf_{n \to \infty} (1 - \theta_{n-1})^2 > \limsup_{n \to \infty} \frac{\theta_n (1 + \theta_n)}{1 - 2c - \epsilon(1 - c)} > \limsup_{n \to \infty} \frac{\theta_n (1 + \theta_n)}{1 - 2c}.$$
(3.4.9)

This implies that

$$(1+o(1))\theta_n(1+\theta_n) \le [1-2c-\epsilon(1-c)+o(1)](1-\theta_{n-1})^2$$

= $[(1-\epsilon)(1-c)-(2c-c+o(1))](1-\theta_{n-1})^2$
 $\le [(1-\epsilon)(1-c)-\theta_n(c+o(1))](1-\theta_{n-1})^2,$

which implies that

$$(1-\epsilon)(1-c)(1-\theta_{n-1})^2 \ge (1+o(1))\theta_n \left(1+\theta_n+(1-\theta_{n-1})^2+o\left((1-\theta_{n-1})^2\right)\right).$$
(3.4.10)

Now, observe from (3.4.6) that

$$\theta_{n-1} - \theta_n + c(1 - \theta_{n-1})(1 - \theta_n) = o\left((1 - \theta_{n-1})(1 - \theta_n)\right),$$

which implies from Proposition 2.5.32(ii) that

$$\theta_{n-1} - \theta_n = -c(1 - \theta_{n-1})(1 - \theta_n) + o\left((1 - \theta_{n-1})(1 - \theta_n)\right)$$

= $-c(1 - \theta_{n-1})^2 + o(1 - \theta_{n-1})^2$ as $n \to \infty$.

This implies that

$$\begin{aligned} |\theta_{n-1} - \theta_n| &= |-c(1 - \theta_{n-1})^2 + o(1 - \theta_{n-1})^2| \\ &\leq c(1 - \theta_{n-1})^2 + o(1 - \theta_{n-1})^2 \text{ as } n \to \infty. \end{aligned}$$
(3.4.11)

Combining (3.4.10) and (3.4.11), we obtain that

$$(1-\epsilon)(1-c)(1-\theta_{n-1})^2 \ge (1+o(1))\,\theta_n\,(1+\theta_n+[\theta_{n-1}-\theta_n]_+)\,. \tag{3.4.12}$$

By Proposition 2.5.32, we have that $t_{n+1} \sim t_n \sim \frac{1}{(1-c)(1-\theta_{n-1})}$ as $n \to \infty$. Hence, (3.4.12) is equivalent to

$$(1-\epsilon)(1-c)(1-\theta_{n-1})^2 \ge \frac{\theta_n}{(1-c)(1-\theta_{n-1})} t_{n+1} \left(1+\theta_n + [\theta_{n-1}-\theta_n]_+\right),$$

which further implies Assumption 3.4.1.

When $\rho_n \neq 1$, then we have the following proposition which provides some criteria for ensuring Assumption 3.4.1, as well as assumption (2.5.37).

Proposition 3.4.7. (See for example, [29, Proposition 3.3]). Suppose that $\theta_n \in [0, 1)$ and $\rho_n \in (0, 1)$ for all $n \ge 1$. Assume that there exist $c \in [0, 1)$ and $\bar{c} \in [c, 1)$ such that

$$\lim_{n \to \infty} \left(\frac{1}{1 - \theta_{n+1}} - \frac{1}{1 - \theta_n} \right) = c,$$

$$\lim_{n \to \infty} \frac{\gamma_{n+1} - \gamma_n}{\gamma_n (1 - \theta_n)} = \bar{c}$$

and

$$\liminf_{n \to \infty} \gamma_n (1 - \theta_n)^2 > \limsup_{n \to \infty} \frac{\theta_n (1 + \theta_n)}{1 - \bar{c}}$$

where $\gamma_n = \frac{1-\rho_n}{2\rho_n}$. Then, assumption (2.5.37) and Assumption 3.4.1 hold.

Remark 3.4.2. It is worthy of note that many practical choices for the inertial and relaxation factors θ_n and ρ_n , respectively, satisfy Assumption 3.4.1. In fact, similar conditions as in Propositions 3.4.5-3.4.7 have already been used in the literature to ensure the convergence of inertial and relaxation methods (see [165, 246, 247] and the references therein). Thus, Assumption 3.4.1, as well as assumption (2.5.37) are much more weaker than the assumptions in those works in the literature.

Moreover, we shall give in Section 3.4.3, some typical examples of θ_n and ρ_n which satisfy the conditions in Propositions 3.4.5-3.4.7 (therefore, satisfying assumption (2.5.37) and Assumption 3.4.1). Then, we check the sensitivity of both θ_n and ρ_n in order to find numerically, the optimum choice of these parameters with respect to the convergence speed of our proposed methods.

3.4.2 Convergence analysis

Lemma 3.4.3. Let $\{x_n\}$ be a sequence generated by Algorithm 3.4.3 such that Assumption 3.4.2(a)-(b) hold. Then,

$$\Gamma_{n+1} - \Gamma_n - \theta_n (\Gamma_n - \Gamma_{n-1})$$

 $\leq \frac{1}{2} (\theta_n + \theta_n^2) \|x_n - x_{n-1}\|^2 - \gamma_n \|x_{n+1} - w_n\|^2 - \frac{\rho_n}{2} (1 - \lambda L) \left[||w_n - y_n||^2 + ||z_n - y_n||^2 \right]$

where $\gamma_n := \frac{1-\rho_n}{2\rho_n}$ and $\Gamma_n := \frac{1}{2} ||x_n - z||^2$, $\forall z \in \Gamma$.

Proof. Let $z \in \Gamma$. Since $x_{n+1} - w_n = \rho_n(z_n - w_n)$, we obtain

$$\langle x_{n+1} - w_n, w_n - z \rangle = \rho_n \langle z_n - w_n, w_n - z_n \rangle + \rho_n \langle z_n - w_n, z_n - z \rangle = -\rho_n ||z_n - w_n||^2 + \rho_n \langle z_n - w_n, z_n - z \rangle = -\rho_n^{-1} ||x_{n+1} - w_n||^2 + \rho_n \langle z_n - w_n + \lambda Ay_n, z_n - z \rangle - \lambda \rho_n \langle Ay_n, z_n - z \rangle \leq -\rho_n^{-1} ||x_{n+1} - w_n||^2 + \lambda \rho_n \langle Ay_n, z - z_n \rangle,$$
 (3.4.13)

where the last inequality follows from $z \in \Gamma \subseteq T_n$ and the characteristic property of P_{T_n} (see (2.4.1)).

Now, using (3.4.13) in Lemma 2.5.14, we obtain

$$\begin{split} &\Gamma_{n+1} - \Gamma_n - \theta_n (\Gamma_n - \Gamma_{n-1}) \\ &\leq \frac{1}{2} (\theta_n + \theta_n^2) \| x_n - x_{n-1} \|^2 - \rho_n^{-1} \| x_{n+1} - w_n \|^2 + \lambda \rho_n \langle Ay_n, \ z - z_n \rangle + \frac{1}{2} \| x_{n+1} - w_n \|^2 \\ &= \frac{1}{2} (\theta_n + \theta_n^2) \| x_n - x_{n-1} \|^2 + \frac{\rho_n - 1}{2\rho_n} \| x_{n+1} - w_n \|^2 - \frac{1}{2\rho_n} \| x_{n+1} - w_n \|^2 \\ &+ \lambda \rho_n \langle Ay_n, \ z - z_n \rangle \\ &= \frac{1}{2} (\theta_n + \theta_n^2) \| x_n - x_{n-1} \|^2 + \frac{\rho_n - 1}{2\rho_n} \| x_{n+1} - w_n \|^2 + \lambda \rho_n \langle Ay_n, \ z - z_n \rangle \\ &- \frac{1}{2\rho_n} \cdot \rho_n^2 \| w_n - z_n \|^2 \\ &= \frac{1}{2} (\theta_n + \theta_n^2) \| x_n - x_{n-1} \|^2 + \frac{\rho_n - 1}{2\rho_n} \| x_{n+1} - w_n \|^2 + \lambda \rho_n \langle Ay_n, \ z - z_n \rangle \tag{3.4.14} \\ &- \frac{\rho_n}{2} \| w_n - y_n \|^2 - \frac{\rho_n}{2} \| y_n - z_n \|^2 - \rho_n \langle w_n - y_n, \ y_n - z_n \rangle, \end{split}$$

where the last equality follows from Lemma 2.1.1.

Since $y_n \subset \mathcal{C}$ and $z \in \Gamma$, we get from (3.2.3) that $\langle Ay_n, y_n - z \rangle \geq 0, \forall n \geq 1$. That is $\langle Ay_n, y_n - z_n + z_n - z \rangle \geq 0, \forall n \geq 1$. This implies that

$$\lambda \langle Ay_n, \ z - z_n \rangle - \langle w_n - y_n, \ y_n - z_n \rangle \leq \lambda \langle Ay_n, \ y_n - z_n \rangle - \langle w_n - y_n, \ y_n - z_n \rangle$$
$$= \langle \lambda Ay_n - w_n + y_n, \ y_n - z_n \rangle. \tag{3.4.15}$$

Substituting (3.4.15) into (3.4.14), we obtain that

$$\begin{aligned}
& \Gamma_{n+1} - \Gamma_n - \theta_n (\Gamma_n - \Gamma_{n-1}) \\
& \leq \frac{1}{2} (\theta_n + \theta_n^2) \| x_n - x_{n-1} \|^2 + \frac{\rho_n - 1}{2\rho_n} \| x_{n+1} - w_n \|^2 \\
& - \frac{\rho_n}{2} \left(\| w_n - y_n \|^2 + \| y_n - z_n \|^2 \right) + \rho_n \langle \lambda A y_n - w_n + y_n, y_n - z_n \rangle \\
& = \frac{1}{2} (\theta_n + \theta_n^2) \| x_n - x_{n-1} \|^2 + \frac{\rho_n - 1}{2\rho_n} \| x_{n+1} - w_n \|^2 - \frac{\rho_n}{2} \left(\| w_n - y_n \|^2 + \| y_n - z_n \|^2 \right) \\
& + \rho_n \langle w_n - \lambda A w_n - y_n, z_n - y_n \rangle + \lambda \rho_n \langle A w_n - A y_n, z_n - y_n \rangle \\
& \leq \frac{1}{2} (\theta_n + \theta_n^2) \| x_n - x_{n-1} \|^2 + \frac{\rho_n - 1}{2\rho_n} \| x_{n+1} - w_n \|^2 \\
& - \frac{\rho_n}{2} \left(\| w_n - y_n \|^2 + \| y_n - z_n \|^2 \right) + \lambda \rho_n \langle A w_n - A y_n, z_n - y_n \rangle,
\end{aligned}$$
(3.4.16)

where the last inequality follows from the definition of T_n .

Now, from the Lipschitz continuity of A, we obtain

$$\langle Aw_n - Ay_n, z_n - y_n \rangle \leq L||w_n - y_n||||z_n - y_n|| = \frac{L}{2} \Big[||w_n - y_n||^2 + ||z_n - y_n||^2 - (||w_n - y_n|| - ||z_n - y_n||)^2 \Big] \leq \frac{L}{2} (||w_n - y_n||^2 + ||z_n - y_n||^2).$$
 (3.4.17)

Substituting (3.4.17) into (3.4.16), we obtain the desired conclusion.

Lemma 3.4.4. Let $\{x_n\}$ be a sequence generated by Algorithm 3.4.3 such that assumption (2.5.37) and Assumption 3.4.2(a)-(b) hold. Then, the following inequality holds:

$$\sum_{i=1}^{n-1} t_{i+1,n} \Big[\Big(2\gamma_i (1-\theta_i)^2 - (\theta_i + \theta_i^2) \Big) \|x_i - x_{i-1}\|^2 \\ + 2\gamma_i (1-\theta_i) \Big(\|x_{i+1} - x_i\|^2 - \|x_i - x_{i-1}\|^2 \Big) \Big] \\ \leq 2t_1 |\Gamma_1 - \Gamma_0| + 2\Gamma_0,$$

where $t_{i,n}$ is as defined in (2.5.40).

Proof. By Lemma 2.1.1, we obtain

$$\begin{aligned} \|x_{n+1} - w_n\|^2 &= \|x_{n+1} - x_n - (x_n - x_{n-1}) + (1 - \theta_n)(x_n - x_{n-1})\|^2 \\ &= \|x_{n+1} - 2x_n + x_{n-1}\|^2 + (1 - \theta_n)^2 \|x_n - x_{n-1}\|^2 \\ &+ 2(1 - \theta_n)\langle x_{n+1} - 2x_n + x_{n-1}, x_n - x_{n-1}\rangle \\ &= \|x_{n+1} - 2x_n + x_{n-1}\|^2 + (1 - \theta_n)^2 \|x_n - x_{n-1}\|^2 \\ &+ (1 - \theta_n) \left[\|x_{n+1} - x_n\|^2 - \|x_n - x_{n-1}\|^2 - \|x_{n+1} - 2x_n + x_{n-1}\|^2 \right] \\ &= \theta_n \|x_{n+1} - 2x_n + x_{n-1}\|^2 + (1 - \theta_n)^2 \|x_n - x_{n-1}\|^2 \\ &+ (1 - \theta_n) \left[\|x_{n+1} - x_n\|^2 - \|x_n - x_{n-1}\|^2 \right] \\ &\geq (1 - \theta_n)^2 \|x_n - x_{n-1}\|^2 \\ &+ (1 - \theta_n) \left[\|x_{n+1} - x_n\|^2 - \|x_n - x_{n-1}\|^2 \right]. \end{aligned}$$
(3.4.18)

Using (3.4.18) in Lemma 3.4.3, and noting that $1 - \lambda L > 0$, we obtain

$$\begin{split} \Gamma_{n+1} - \Gamma_n - \theta_n (\Gamma_n - \Gamma_{n-1}) &\leq \frac{1}{2} (\theta_n + \theta_n^2) \| x_n - x_{n-1} \|^2 - \gamma_n (1 - \theta_n)^2 \| x_n - x_{n-1} \|^2 \\ &- \gamma_n (1 - \theta_n) \Big[\| x_{n+1} - x_n \|^2 - \| x_n - x_{n-1} \|^2 \Big] \\ &= \Big[\frac{1}{2} (\theta_n + \theta_n^2) - \gamma_n (1 - \theta_n)^2 \Big] \| x_n - x_{n-1} \|^2 \\ &- \gamma_n (1 - \theta_n) \Big[\| x_{n+1} - x_n \|^2 - \| x_n - x_{n-1} \|^2 \Big]. \end{split}$$

This together with Lemma 2.5.13(a) imply that

$$\begin{split} \Gamma_n - \Gamma_0 &= \sum_{i=1}^n (\Gamma_i - \Gamma_{i-1}) \\ &\leq t_{1,n} (\Gamma_1 - \Gamma_0) \\ &+ \sum_{i=1}^{n-1} t_{i+1,n} \Big[\Big(\frac{1}{2} (\theta_i + \theta_i^2) - \gamma_i (1 - \theta_i)^2 \Big) \|x_i - x_{i-1}\|^2 \Big] \\ &- \sum_{i=1}^{n-1} t_{i+1,n} \Big[\gamma_i (1 - \theta_i) \Big(\|x_{i+1} - x_i\|^2 - \|x_i - x_{i-1}\|^2 \Big) \Big] \end{split}$$

Noting that $t_{1,n} \leq t_1$, we obtain

$$\sum_{i=1}^{n-1} t_{i+1,n} \Big[\Big(2\gamma_i (1-\theta_i)^2 - (\theta_i + \theta_i^2) \Big) \|x_i - x_{i-1}\|^2 \\ + 2\gamma_i (1-\theta_i) \Big(\|x_{i+1} - x_i\|^2 - \|x_i - x_{i-1}\|^2 \Big) \Big] \\ \leq 2t_{1,n} (\Gamma_1 - \Gamma_0) + 2(\Gamma_0 - \Gamma_n) \\ \leq 2t_1 |\Gamma_1 - \Gamma_0| + 2\Gamma_0 - 2\Gamma_n.$$

Using $\Gamma_n = \frac{1}{2} ||x_n - z||^2 \ge 0$, $\forall n \ge 1$ in the above inequality, we obtain the desired conclusion.

Lemma 3.4.5. Let $\{x_n\}$ be a sequence generated by Algorithm 3.4.3 such that assumption (2.5.37) and Assumption 3.4.2(a)-(b) hold. Then, the following inequality holds:

$$\sum_{i=1}^{n-1} 2\gamma_{i-1}(1-\theta_{i-1}) - \theta_i t_{i+1} \Big(1+\theta_i + 2\Big[\gamma_i(1-\theta_i) - \gamma_{i-1}(1-\theta_{i-1})\Big]_+ \Big) \|x_i - x_{i-1}\|^2 \\ \leq 2\Big[t_1|\Gamma_1 - \Gamma_0| + \Gamma_0 + t_1\gamma_0(1-\theta_0)\|x_1 - x_0\|^2\Big].$$

Proof. Observe that

$$\sum_{i=1}^{n-1} t_{i+1,n} \cdot 2\gamma_i (1-\theta_i) \Big[\|x_{i+1} - x_i\|^2 - \|x_i - x_{i-1}\|^2 \Big]$$

$$= 2 \sum_{i=1}^{n-1} \Big(t_{i,n} \gamma_{i-1} (1-\theta_{i-1}) - t_{i+1,n} \gamma_i (1-\theta_i) \Big) \|x_i - x_{i-1}\|^2$$

$$+ 2 t_{1,n} \gamma_{n-1} (1-\theta_{n-1}) \|x_n - x_{n-1}\|^2 - 2 t_{1,n} \gamma_0 (1-\theta_0) \|x_1 - x_0\|^2$$

$$\geq 2 \sum_{i=1}^{n-1} \Big(t_{i,n} \gamma_{i-1} (1-\theta_{i-1}) - t_{i+1,n} \gamma_i (1-\theta_i) \Big) \|x_i - x_{i-1}\|^2$$

$$- 2 t_{1,n} \gamma_0 (1-\theta_0) \|x_1 - x_0\|^2$$

$$\geq 2 \sum_{i=1}^{n-1} \Big(t_{i,n} \gamma_{i-1} (1-\theta_{i-1}) - t_{i+1,n} \gamma_i (1-\theta_i) \Big) \|x_i - x_{i-1}\|^2$$
(3.4.19)

$$- 2 t_1 \gamma_0 (1-\theta_0) \|x_1 - x_0\|^2,$$

where the last inequality follows from $t_{1,n} \leq t_1$. Now, using (3.4.19) in Lemma 3.4.4, we obtain that

$$\sum_{i=1}^{n-1} t_{i+1,n} \left(2\gamma_i (1-\theta_i)^2 - (\theta_i + \theta_i^2) \right) \|x_i - x_{i-1}\|^2$$

+2
$$\sum_{i=1}^{n-1} \left(t_{i,n} \gamma_{i-1} (1-\theta_{i-1}) - t_{i+1,n} \gamma_i (1-\theta_i) \right) \|x_i - x_{i-1}\|^2$$

$$\leq 2t_1 |\Gamma_1 - \Gamma_0| + 2\Gamma_0 + 2t_1 \gamma_0 (1-\theta_0) \|x_1 - x_0\|^2.$$

That is

$$\sum_{i=1}^{n-1} \left[t_{i+1,n} (2\gamma_i (1-\theta_i)^2 - (\theta_i + \theta_i^2) - 2\gamma_i (1-\theta_i)) + 2t_{i,n} \gamma_{i-1} (1-\theta_{i-1}) \right] \|x_i - x_{i-1}\|^2 \\ \leq 2 \left[t_1 |\Gamma_1 - \Gamma_0| + \Gamma_0 + t_1 \gamma_0 (1-\theta_0) \|x_1 - x_0\|^2 \right].$$
(3.4.20)

Now, recall from (2.5.41) that $t_{i,n} = 1 + \theta_i t_{i+1,n}$ for all $i \ge 1$ and $n \ge i+1$. Hence, we have

$$2t_{i,n}\gamma_{i-1}(1-\theta_{i-1}) = 2\Big[\gamma_{i-1}(1-\theta_{i-1}) + \theta_i t_{i+1,n}\gamma_{i-1}(1-\theta_{i-1})\Big].$$

This implies that

$$t_{i+1,n} \Big[2\gamma_i (1-\theta_i)^2 - (\theta_i + \theta_i^2) - 2\gamma_i (1-\theta_i) \Big] + 2t_{i,n}\gamma_{i-1} (1-\theta_{i-1}) \\ = t_{i+1,n} \Big[2\gamma_i (1-\theta_i)^2 - (\theta_i + \theta_i^2) - 2\gamma_i (1-\theta_i) + 2\theta_i \gamma_{i-1} (1-\theta_{i-1}) \Big] \\ + 2\gamma_{i-1} (1-\theta_{i-1}) \\ = 2\gamma_{i-1} (1-\theta_{i-1}) + t_{i+1,n} \Big(-2\gamma_i \theta_i (1-\theta_i) - (\theta_i + \theta_i^2) + 2\theta_i \gamma_{i-1} (1-\theta_{i-1}) \Big) \\ = 2\gamma_{i-1} (1-\theta_{i-1}) - \theta_i t_{i+1,n} \Big(2\gamma_i (1-\theta_i) + 1 + \theta_i - 2\gamma_{i-1} (1-\theta_{i-1}) \Big) \\ \ge 2\gamma_{i-1} (1-\theta_{i-1}) - \theta_i t_{i+1,n} \Big(1 + \theta_i + 2 \Big[\gamma_i (1-\theta_i) - \gamma_{i-1} (1-\theta_{i-1}) \Big]_+ \Big) \\ \ge 2\gamma_{i-1} (1-\theta_{i-1}) - \theta_i t_{i+1,n} \Big(1 + \theta_i + 2 \big[\gamma_i (1-\theta_i) - \gamma_{i-1} (1-\theta_{i-1}) \big]_+ \Big), \quad (3.4.21)$$

where the last inequality follows from $t_{i+1,n} \leq t_{i+1}$. Now, using (3.4.21) in (3.4.20), we obtain that the desired conclusion.

Lemma 3.4.6. Let $\{x_n\}$ be a sequence generated by Algorithm 3.4.3 such that assumption (2.5.37), Assumption 3.4.1 and Assumption 3.4.2(a)-(b) hold. Then,

$$\sum_{n=1}^{\infty} \theta_n t_{n+1} \|x_n - x_{n-1}\|^2 < \infty.$$

Proof. Without loss of generality, we may assume that inequality (3.4.1) holds true for all $n \ge 1$. That is,

$$\begin{cases} 2\epsilon\gamma_{n-1}(1-\theta_{n-1}) \leq \\ 2\gamma_{n-1}(1-\theta_{n-1}) - \theta_n t_{n+1} \Big(1+\theta_n + 2\Big[\gamma_n(1-\theta_n) - \gamma_{n-1}(1-\theta_{n-1})\Big]_+ \Big), & \text{if } \rho_n \in (0,1), \\ \epsilon(1-\theta_{n-1}) \leq \\ (1-\theta_{n-1}) - \theta_n t_{n+1} \Big(1+\theta_n + \Big[\theta_{n-1} - \theta_n\Big]_+ \Big), & \text{if } \rho_n = 1, \end{cases}$$
where $\gamma_n = \frac{1-\rho_n}{2\rho_n}$, for all $n \geq 1$.

At this point, we divide our proof into two cases.

Case 1: Suppose that $\rho_n \in (0,1), \forall n \ge 1$. Then, using (3.4.22) in Lemma 3.4.5, we obtain

$$\sum_{i=1}^{n-1} \epsilon \gamma_{i-1} (1-\theta_{i-1}) \|x_i - x_{i-1}\|^2 \le t_1 |\Gamma_1 - \Gamma_0| + \Gamma_0 + t_1 \gamma_0 (1-\theta_0) \|x_1 - x_0\|^2.$$
(3.4.23)

Now, taking limit as $n \to \infty$ in (3.4.23), we obtain that

$$\sum_{i=1}^{\infty} \gamma_{i-1} (1 - \theta_{i-1}) \| x_i - x_{i-1} \|^2 < \infty.$$
(3.4.24)

Again, from (3.4.22), we obtain that

$$\frac{1}{2}\theta_{n}t_{n+1} \leq \gamma_{n-1}(1-\theta_{n-1}) - \frac{1}{2}\theta_{n}t_{n+1}\Big(\theta_{n} + 2\Big[\gamma_{n}(1-\theta_{n}) - \gamma_{n-1}(1-\theta_{n-1})\Big]_{+}\Big) \\
-\epsilon\gamma_{n-1}(1-\theta_{n-1}) \\
\leq \gamma_{n-1}(1-\theta_{n-1}).$$
(3.4.25)

Using (3.4.25) in (3.4.24), we obtain

$$\sum_{i=1}^{\infty} \theta_i t_{i+1} \| x_i - x_{i-1} \|^2 < \infty.$$

Case 2: Suppose that $\rho_n = 1$, $\forall n \ge 1$. Then, $x_{n+1} = z_n$ and $\gamma_n = 0$, $\forall n \ge 1$. Setting $x = w_n - y_n$ and $y = x_{n+1} - y_n$ in Lemma 2.1.1, and using its result in Lemma 3.4.3, we obtain

$$\begin{split} \Gamma_{n+1} - \Gamma_n - \theta_n (\Gamma_n - \Gamma_{n-1}) &\leq \frac{1}{2} (\theta_n + \theta_n^2) \|x_n - x_{n-1}\|^2 \\ &- \frac{1}{2} \left[||w_n - y_n + x_{n+1} - y_n||^2 + ||w_n - x_{n+1}||^2 \right] \\ &\leq \frac{1}{2} (\theta_n + \theta_n^2) \|x_n - x_{n-1}\|^2 - \frac{1}{2} ||w_n - x_{n+1}||^2. \quad (3.4.26) \end{split}$$

Now, using (3.4.18) in (3.4.26), and repeating the same line of proof as in Lemma 3.4.4, we get

$$\sum_{i=1}^{n-1} t_{i+1,n} \Big[\left((1-\theta_i)^2 - (\theta_i + \theta_i^2) \right) \|x_i - x_{i-1}\|^2 + (1-\theta_i) \Big(\|x_{i+1} - x_i\|^2 - \|x_i - x_{i-1}\|^2 \Big) \Big] \\ \leq 2t_1 |\Gamma_1 - \Gamma_0| + 2\Gamma_0.$$
(3.4.27)

Similar to (3.4.19), we have

$$\sum_{i=1}^{n-1} t_{i+1,n} (1-\theta_i) \Big[\|x_{i+1} - x_i\|^2 - \|x_i - x_{i-1}\|^2 \Big]$$

$$\geq \sum_{i=1}^{n-1} \Big(t_{i,n} (1-\theta_{i-1}) - t_{i+1,n} (1-\theta_i) \Big) \|x_i - x_{i-1}\|^2 \qquad (3.4.28)$$

$$- t_1 (1-\theta_0) \|x_1 - x_0\|^2.$$

Also, using (3.4.28) in (3.4.27), and repeating the same line of proof as in Lemma 3.4.5, we get

$$\sum_{i=1}^{n-1} (1-\theta_{i-1}) - \theta_i t_{i+1} \Big(1 + \theta_i + \Big[\theta_{i-1} - \theta_i \Big]_+ \Big) \|x_i - x_{i-1}\|^2 \\ \leq 2t_1 |\Gamma_1 - \Gamma_0| + 2\Gamma_0 + t_1 (1-\theta_0) \|x_1 - x_0\|^2.$$
(3.4.29)

Using (3.4.22) in (3.4.29), and repeating the same line of proof as in Case 1, we obtain

$$\sum_{i=1}^{\infty} \theta_i t_{i+1} \| x_i - x_{i-1} \|^2 < \infty,$$

which yields the desired conclusion.

Lemma 3.4.7. Let $\{x_n\}$ be a sequence generated by Algorithm 3.4.3 such that assumption (2.5.37), Assumption 3.4.1 and Assumption 3.4.2(a)-(b) hold. Then,

(a) $\lim_{n\to\infty} ||x_n - z||$ exists for all $z \in \Gamma$, and consequently, $\{x_n\}$ is bounded.

(b)
$$\lim_{n \to \infty} \left[||w_n - y_n||^2 + ||z_n - y_n||^2 \right] = 0,$$

(c)
$$\lim_{n \to \infty} ||x_n - w_n|| = 0.$$

Proof.

(a) From Lemma 3.4.3, we obtain

$$\Gamma_{n+1} - \Gamma_n \leq \theta_n \Big(\Gamma_n - \Gamma_{n-1} \Big) + \frac{1}{2} \Big(\theta_n + \theta_n^2 \Big) \|x_n - x_{n-1}\|^2
- \frac{\rho_n}{2} (1 - \lambda L) \Big[||w_n - y_n||^2 + ||z_n - y_n||^2 \Big]
\leq \theta_n \Big(\Gamma_n - \Gamma_{n-1} \Big) + \theta_n \|x_n - x_{n-1}\|^2 - \frac{\rho_n}{2} (1 - \lambda L) \Big[||w_n - y_n||^2 + ||z_n - y_n||^2 \Big]
(3.4.30)
\leq \theta_n \Big(\Gamma_n - \Gamma_{n-1} \Big) + \theta_n \|x_n - x_{n-1}\|^2,$$
(3.4.31)

where the second inequality follows from $\theta_n^2 \leq \theta_n$ and the third inequality follows from $1 - \lambda L > 0$.

Now, applying Lemma 2.5.13(b) and Lemma 3.4.6 in (3.4.31), we obtain that $\sum_{n=1}^{\infty} \left[\Gamma_n - \Gamma_{n-1} \right]_+ < \infty$. Since $\Gamma_n = \frac{1}{2} ||x_n - z||^2$, we get that $\lim_{n \to \infty} ||x_n - z||^2$ exists. Hence, $\{x_n\}$ is bounded.

(b) By applying Lemma 2.5.13(a) in (3.4.30), we obtain that

$$\begin{split} \Gamma_n - \Gamma_0 &= \sum_{i=1}^n \left(\Gamma_i - \Gamma_{i-1} \right) \\ &\leq t_{1,n} \left(\Gamma_1 - \Gamma_0 \right) \\ &+ \sum_{i=1}^{n-1} t_{i+1,n} \left[\theta_i \| x_i - x_{i-1} \|^2 - \frac{\rho_i}{2} \left(1 - \lambda L \right) \left[||w_i - y_i||^2 + ||z_i - y_i||^2 \right] \right] \\ &\leq t_1 \left(\Gamma_1 - \Gamma_0 \right) + \sum_{i=1}^{n-1} t_{i+1} \theta_i \| x_i - x_{i-1} \|^2 \\ &- \sum_{i=1}^{n-1} t_{i+1,n} \frac{\rho_i}{2} \left(1 - \lambda L \right) \left[||w_i - y_i||^2 + ||z_i - y_i||^2 \right] \Big], \end{split}$$

which implies that

$$\sum_{i=1}^{n-1} t_{i+1,n} \frac{\rho_i}{2} (1 - \lambda L) \left[||w_i - y_i||^2 + ||z_i - y_i||^2 \right] \leq \Gamma_0 - \Gamma_n + t_1 (\Gamma_1 - \Gamma_0) + \sum_{i=1}^{n-1} t_{i+1} \theta_i ||x_i - x_{i-1}||^2 \leq \Gamma_0 + t_1 |\Gamma_1 - \Gamma_0| + \sum_{i=1}^{n-1} t_{i+1} \theta_i ||x_i - x_{i-1}||^2 < \infty,$$

where the last inequality follows from Lemma 3.4.6.

Since $t_{i+1,n} = 0$ for $i \ge n$, letting n tend to ∞ , the monotone convergence theorem then implies that

$$\sum_{i=1}^{\infty} t_{i+1} \frac{\rho_i}{2} \left(1 - \lambda L\right) \left[||w_i - y_i||^2 + ||z_i - y_i||^2 \right] < \infty.$$
(3.4.32)

Since $\liminf_{n\to\infty} \rho_n > 0$, there exists M > 0 such that $\rho_n \ge M$ for n large enough. Now, replacing i with n in (3.4.32) and noting that $t_n \ge 1, \forall n \ge 1$, we obtain from (3.4.32) that $\sum_{n=1}^{\infty} \frac{M}{2} (1 - \lambda L) [||w_n - y_n||^2 + ||z_n - y_n||^2] < \infty$, which further gives that

$$\lim_{n \to \infty} \left[||w_n - y_n||^2 + ||z_n - y_n||^2 \right] = 0.$$

(c) Since $w_n = x_n + \theta_n(x_n - x_{n-1})$, we obtain from Lemma 3.4.6 that

$$\sum_{n=1}^{\infty} t_{n+1} \|w_n - x_n\|^2 \le \sum_{n=1}^{\infty} \theta_n t_{n+1} \|x_n - x_{n-1}\|^2 < \infty.$$

Noting that $t_n \ge 1, \forall n \ge 1$, we conclude immediately that $\lim_{n \to \infty} ||w_n - x_n|| = 0$. \Box

Lemma 3.4.8. Let $\{x_n\}$ be a sequence generated by Algorithm 3.4.3 such that assumption (2.5.37), Assumption 3.4.1 and Assumption 3.4.2(a)-(d) hold. If x^* is one of the weak cluster points of $\{x_n\}$, then we have at least one of the following: $x^* \in \Gamma$ or $Ax^* = 0$.

Proof. By Lemma 3.4.7(a), we can choose a subsequence of $\{x_n\}$ denoted by $\{x_{n_k}\}$ such that $x_{n_k} \rightarrow x^* \in \mathcal{H}$. Also, from Lemma 3.4.7(b),(c), we obtain that $\lim_{n\to\infty} ||y_n - x_n|| = 0$. Hence, we can choose a subsequence $\{y_{n_k}\}$ of $\{y_n\}$ such that $y_{n_k} \rightarrow x^*$. Note that $x^* \in \mathcal{C}$ since $\{y_{n_k}\} \subset \mathcal{C}$.

We now consider two possible cases.

Case 1: Suppose that $\limsup_{k\to\infty} ||Ay_{n_k}|| = 0$. Then, $\lim_{k\to\infty} ||Ay_{n_k}|| = \liminf_{k\to\infty} ||Ay_{n_k}|| = 0$. Also, by the sequentially weakly continuity of A on C, we obtain that $Ay_{n_k} \rightharpoonup Ax^*$. Thus, by the weakly lower semicontinuity of $|| \cdot ||$, we have that

$$0 < ||Ax^*|| \le \liminf_{k \to \infty} ||Ay_{n_k}|| = 0, \qquad (3.4.33)$$

which implies that $Ax^* = 0$.

Case 2: Suppose that $\limsup_{k\to\infty} ||Ay_{n_k}|| > 0$. Then, without loss of generality, we can choose a subsequence of $\{Ay_{n_k}\}$ still denoted by $\{Ay_{n_k}\}$ such that $\lim_{k\to\infty} ||Ay_{n_k}|| = M > 0$. Also by the characteristics property of $P_{\mathcal{C}}$, we obtain for all $x \in \mathcal{C}$ that

$$\langle w_{n_k} - \lambda A w_{n_k} - y_{n_k}, x - y_{n_k} \rangle \le 0.$$

This implies that

$$\frac{1}{\lambda} \langle w_{n_k} - y_{n_k}, x - y_{n_k} \rangle + \langle A w_{n_k}, y_{n_k} - w_{n_k} \rangle \le \langle A w_{n_k}, x - w_{n_k} \rangle.$$
(3.4.34)

Thus, we obtain from Lemma 3.4.7(b) that

$$0 \le \liminf_{k \to \infty} \langle Aw_{n_k}, x - w_{n_k} \rangle \le \limsup_{k \to \infty} \langle Aw_{n_k}, x - w_{n_k} \rangle < \infty, \ \forall x \in \mathcal{C}.$$
(3.4.35)

Now, note that

$$\langle Ay_{n_k}, x - y_{n_k} \rangle = \langle Ay_{n_k} - Aw_{n_k}, x - w_{n_k} \rangle + \langle Aw_{n_k}, x - w_{n_k} \rangle + \langle Ay_{n_k}, w_{n_k} - y_{n_k} \rangle.$$
(3.4.36)

Moreover, since A is Lipschitz continuous on \mathcal{H} , we obtain from Lemma 3.4.7(b) that $\lim_{k\to\infty} ||Aw_{n_k} - Ay_{n_k}|| = 0$. Hence, we obtain from Lemma 3.4.7(b), (3.4.35) and (3.4.36) that

$$0 \le \liminf_{k \to \infty} \langle Ay_{n_k}, x - y_{n_k} \rangle \le \limsup_{k \to \infty} \langle Ay_{n_k}, x - y_{n_k} \rangle < \infty, \ \forall x \in \mathcal{C}.$$
(3.4.37)

Based on (3.4.37), we consider two cases under **Case 2**, as follows:

Case A: Suppose that $\limsup_{k\to\infty} \langle Ay_{n_k}, x - y_{n_k} \rangle > 0, \forall x \in \mathcal{C}$. Then, we can choose a subsequence of $\{y_{n_k}\}$ denoted by $\{y_{n_{k_j}}\}$ such that $\lim_{j\to\infty} \langle Ay_{n_{k_j}}, x - y_{n_{k_j}} \rangle > 0$. Thus, there exists

 $j_0 \geq 1$ such that $\langle Ay_{n_{k_j}}, x - y_{n_{k_j}} \rangle > 0, \forall j \geq j_0$, which by the quasimonotonicity of A on \mathcal{H} , implies $\langle Ax, x - y_{n_{k_j}} \rangle \geq 0, \forall x \in \mathcal{C}, j \geq j_0$. Thus, letting $j \to \infty$, we obtain that $\langle Ax, x - x^* \rangle \geq 0, \forall x \in \mathcal{C}$. Hence, $x^* \in \Gamma$.

Case B: Suppose that $\limsup_{k\to\infty} \langle Ay_{n_k}, x - y_{n_k} \rangle = 0, \forall x \in \mathcal{C}$. Then, by (3.4.37), we obtain that

$$\lim_{k \to \infty} \langle Ay_{n_k}, x - y_{n_k} \rangle = 0, \forall x \in \mathcal{C},$$
(3.4.38)

which implies that

$$\langle Ay_{n_k}, x - y_{n_k} \rangle + |\langle Ay_{n_k}, x - y_{n_k} \rangle| + \frac{1}{k+1} > 0, \forall x \in \mathcal{C}.$$

$$(3.4.39)$$

Also, since $\lim_{k\to\infty} ||Ay_{n_k}|| = M > 0$, there exists $k_0 \ge 1$ such that $||Ay_{n_k}|| > \frac{M}{2}, \forall k \ge k_0$. Therefore, we can set $q_{n_k} = \frac{Ay_{n_k}}{||Ay_{n_k}||^2}, \forall k \ge k_0$. Thus, $\langle Ay_{n_k}, q_{n_k} \rangle = 1, \forall k \ge k_0$. Hence, by (3.4.39), we obtain

$$\left\langle Ay_{n_k}, x + q_{n_k} \left[\left| \left\langle Ay_{n_k}, x - y_{n_k} \right\rangle \right| + \frac{1}{k+1} - y_{n_k} \right] \right\rangle > 0,$$

which by the quasimonotonicity of A on \mathcal{H} , implies

$$\left\langle A\left(x+q_{n_k}\left[|\langle Ay_{n_k}, x-y_{n_k}\rangle|+\frac{1}{k+1}\right]\right), x+q_{n_k}\left[|\langle Ay_{n_k}, x-y_{n_k}\rangle|+\frac{1}{k+1}\right]-y_{n_k}\right\rangle \ge 0.$$

This further implies that

$$\begin{aligned} \langle Ax, x + q_{n_{k}} \Big[|\langle Ay_{n_{k}}, x - y_{n_{k}} \rangle| + \frac{1}{k+1} \Big] - y_{n_{k}} \rangle \\ &\geq \langle Ax, A(x + q_{n_{k}} \big[|\langle Ay_{n_{k}}, x - y_{n_{k}} \rangle| + \frac{1}{k+1} \big] \rangle, x + q_{n_{k}} \big[|\langle Ay_{n_{k}}, x - y_{n_{k}} \rangle| + \frac{1}{k+1} \big] - y_{n_{k}} \rangle \\ &\geq - ||Ax - A(x + q_{n_{k}} \Big[|\langle Ay_{n_{k}}, x - y_{n_{k}} \rangle| + \frac{1}{k+1} \Big])|| \\ &\times \Big(||x + q_{n_{k}} \Big[|\langle Ay_{n_{k}}, x - y_{n_{k}} \rangle| + \frac{1}{k+1} \Big] - y_{n_{k}} || \Big) \\ &\geq - L ||q_{n_{k}} \Big[|\langle Ay_{n_{k}}, x - y_{n_{k}} \rangle| + \frac{1}{k+1} \Big] || \cdot ||x + q_{n_{k}} \Big[|\langle Ay_{n_{k}}, x - y_{n_{k}} \rangle| + \frac{1}{k+1} \Big] - y_{n_{k}} || \\ &= \frac{-L}{||Ay_{n_{k}}||} \Big(|\langle Ay_{n_{k}}, x - y_{n_{k}} \rangle| + \frac{1}{k+1} \Big) \cdot ||x + q_{n_{k}} \Big[|\langle Ay_{n_{k}}, x - y_{n_{k}} \rangle| + \frac{1}{k+1} \Big] - y_{n_{k}} || \\ &\geq \frac{-2L}{M} \Big(|\langle Ay_{n_{k}}, x - y_{n_{k}} \rangle| + \frac{1}{k+1} \Big) M_{1}, \end{aligned}$$

$$(3.4.40)$$

for some $M_1 > 0$, where the existence of M_1 follows from the boundedness of $\{x + q_{n_k} \left[|\langle Ay_{n_k}, x - y_{n_k} \rangle| + \frac{1}{k+1} \right] - y_{n_k} \}$. Note from (3.4.38) that $\lim_{k \to \infty} \left(|\langle Ay_{n_k}, x - y_{n_k} \rangle| + \frac{1}{k+1} \right) = 0$. Hence, letting $k \to \infty$ in (3.4.40), we get that $\langle Ax, x - x^* \rangle \ge 0, \forall x \in \mathcal{C}$. Therefore, $x^* \in \Gamma$. **Lemma 3.4.9.** Let $\{x_n\}$ be a sequence generated by Algorithm 3.4.3 such that assumption (2.5.37), Assumption 3.4.1 and Assumption 3.4.2(a)-(d) hold. Then $\{x_n\}$ has at most one weak cluster point in Γ .

Proof. Suppose on the contrary that $\{x_n\}$ has at least two weak cluster points in Γ . Let $x^* \in \Gamma$ and $\bar{x} \in \Gamma$ be any two weak cluster points of $\{x_n\}$ such that $x^* \neq \bar{x}$. Also, let $\{x_{n_j}\}$ be a subsequence of $\{x_n\}$ such that $x_{n_j} \rightharpoonup \bar{x}$, as $j \rightarrow \infty$. Then, by Lemma 3.4.7(a) and Lemma 2.5.15, we have

$$\lim_{n \to \infty} \|x_n - \bar{x}\| = \lim_{j \to \infty} \|x_{n_j} - \bar{x}\|$$
$$= \liminf_{j \to \infty} \|x_{n_j} - \bar{x}\|$$
$$< \liminf_{j \to \infty} \|x_{n_j} - x^*\|$$
$$= \lim_{n \to \infty} \|x_n - x^*\|$$
$$= \liminf_{k \to \infty} \|x_{n_k} - x^*\|$$
$$< \liminf_{k \to \infty} \|x_{n_k} - \bar{x}\| = \lim_{n \to \infty} \|x_n - \bar{x}\|$$

which is a contradiction. Therefore, $\{x_n\}$ has at most one weak cluster point in Γ . \Box

Theorem 3.4.10. Let $\{x_n\}$ be a sequence generated by Algorithm 3.4.3 such that assumption (2.5.37), Assumption 3.4.1 and Assumption 3.4.2(a)-(e) hold. Then $\{x_n\}$ converges weakly to an element of $VI(\mathcal{C}, A)$.

Proof. By Assumption 3.4.2(e), $\{x \in \mathcal{C} : Ax = 0\} \setminus \Gamma$ is a finite set. Hence, by Lemma 3.4.8 and Lemma 3.4.9, we have that $\{x_n\}$ has finite weak cluster points in $VI(\mathcal{C}, A)$. Let x^1, x^2, \dots, x^m be the weak cluster points of $\{x_n\}$, and let $\{x_{n_k}^i\}$ be a subsequence of $\{x_n\}$ such that $x_{n_k}^i \rightharpoonup x^i$, as $k \to \infty$. Then, we obtain

$$\lim_{k \to \infty} \left\langle x_{n_k}^i, x^i - x^j \right\rangle = \left\langle x^i, x^i - x^j \right\rangle, \forall j \neq i.$$
(3.4.41)

Now, for $j \neq i$, set $q = \beta^{-1}(x^i - x^j)$, where $\beta = ||x^i - x^j||$. Then,

$$\langle x^{i}, q \rangle = \beta^{-1} \langle x^{i}, x^{i} - x^{j} \rangle$$

$$= \beta^{-1} \Big(\|x^{i}\|^{2} - \langle x^{i}, x^{j} \rangle \Big)$$

$$= \beta^{-1} \Big[\|x^{i}\|^{2} - \frac{1}{2} \|x^{i}\|^{2} - \frac{1}{2} \|x^{j}\|^{2} + \frac{1}{2} \|x^{i} - x^{j}\|^{2} \Big]$$

$$= \frac{1}{2\beta} \Big(\|x^{i}\|^{2} - \|x^{j}\|^{2} \Big) + \frac{1}{2}\beta$$

$$> \frac{1}{2\beta} \Big(\|x^{i}\|^{2} - \|x^{j}\|^{2} \Big) + \frac{1}{4}\beta.$$

$$(3.4.42)$$

For sufficiently large k, we obtain from (3.4.41) and (3.4.42) that

$$x_{n_k}^i \in \left\{ x : \langle x, q \rangle > \frac{1}{2\beta} \left(\|x^i\|^2 - \|x^j\|^2 \right) + \frac{1}{4}\beta \right\}.$$
 (3.4.43)
Hence, there exists $N_1 > N$ $(N \in \mathbb{N})$ such that

$$x_{n_{k}}^{i} \in B^{i} := \bigcap_{j=1, j \neq i}^{m} \left\{ x : \left\langle x, \frac{x^{i} - x^{j}}{\|x^{i} - x^{j}\|} \right\rangle > \frac{1}{2\|x^{i} - x^{j}\|} \left(\|x^{i}\|^{2} - \|x^{j}\|^{2} \right) + \epsilon_{0} \right\}, \forall n \ge N_{1},$$

$$(3.4.44)$$

where

$$\epsilon_0 = \min\left\{\frac{1}{4} \|x^i - x^j\| : i, j = 1, 2, \cdots, m, i \neq j\right\}$$

Now, from Lemma 3.4.7(b), we obtain that $\lim_{n\to\infty} ||z_n - w_n|| = 0$. Since $x_{n+1} - w_n = \rho_n(z_n - w_n)$ and $\{\rho_n\}$ is bounded, we obtain that $\lim_{n\to\infty} ||x_{n+1} - w_n|| = 0$. This together with Lemma 3.4.7(c), imply that $\lim_{n\to\infty} ||x_{n+1} - x_n|| = 0$. Hence, there exists $N_2 > N_1 > N$ such that $||x_{n+1} - x_n|| < \epsilon_0, \forall n \ge N_2$.

Claim: $\{x_n\}$ has only one weak cluster point in $VI(\mathcal{C}, A)$.

Suppose on the contrary that $\{x_n\}$ has more than one weak cluster points in $VI(\mathcal{C}, A)$. Then, there exists $N_3 \ge N_2 > N_1 > N$ such that $x_{N_3} \in B^i$ and $x_{N_3+1} \in B^j$, where

$$B^{j} := \bigcap_{i=1, i \neq j}^{m} \left\{ x : \left\langle -x, \frac{x^{j} - x^{i}}{\|x^{j} - x^{i}\|} \right\rangle < \frac{1}{2\|x^{j} - x^{i}\|} \left(\|x^{i}\|^{2} - \|x^{j}\|^{2} \right) - \epsilon_{0} \right\},\$$

$$i, j \in \{1, 2, \cdots, m\} \text{ and } m \geq 2.$$
(3.4.45)

In particular, we have

$$\|x_{N_3+1} - x_{N_3}\| < \epsilon_0. \tag{3.4.46}$$

Since $x_{N_3} \in B^i$ and $x_{N_3+1} \in B^j$, we obtain that

$$\left\langle x_{N_3}, \frac{x^i - x^j}{\|x^i - x^j\|} \right\rangle > \frac{1}{2\|x^i - x^j\|} \left(\|x^i\|^2 - \|x^j\|^2 \right) + \epsilon_0$$
 (3.4.47)

and

$$\left\langle -x_{N_{3}+1}, \frac{x^{i}-x^{j}}{\|x^{j}-x^{i}\|} \right\rangle > \frac{1}{2\|x^{i}-x^{j}\|} \left(\|x^{j}\|^{2} - \|x^{i}\|^{2} \right) + \epsilon_{0}.$$
 (3.4.48)

Adding (3.4.47) and (3.4.48), and using (3.4.46), we obtain

$$2\epsilon_0 < \left\langle x_{N_3} - x_{N_3+1}, \frac{x_i - x^j}{\|x^i - x^j\|} \right\rangle \le \|x_{N_3+1} - x_{N_3}\| < \epsilon_0, \tag{3.4.49}$$

which is not possible. Hence, our claim holds. That is, $\{x_n\}$ has only one weak cluster point in $VI(\mathcal{C}, A)$. Therefore, we conclude that $\{x_n\}$ converges weakly to an element of $VI(\mathcal{C}, A)$.

Remark 3.4.11.

- (i) The conclusion of Theorem 3.4.10 still hold even if $\lambda \in (0, \frac{1}{L})$ in Algorithm 3.4.3 is replaced with a variable stepsize λ_n such that $0 < \inf_{n \ge 1} \lambda_n \le \sup_{n \ge 1} \lambda_n < \frac{1}{L}$. However, choosing this variable stepsize still requires the knowledge of the Lipschitz constant L.
- (ii) When the Lipschitz constant is not known, we refer to Algorithm 3.4.4, where the choice of λ_n does not depend on its knowledge. Note from the stepsize λ_n in (3.4.2), that $\lim_{n\to\infty} \lambda_n = \lambda$ and $\lambda \in [\min\{\frac{\mu}{L}, \lambda_1\}, \lambda_1 + d]$, where $d = \sum_{n=1}^{\infty} d_n$ (see also [162, Lemma 3.1]).
- (iii) When $d_n = 0$, then the stepsize λ_n generated by Algorithm 3.4.4 is similar to that in [267]. We recall that the stepsize in [267] is monotone non-increasing, thus, their methods may depend on the choice of the initial stepsize λ_1 . However, the stepsize given in (3.4.2) is non-monotonic and hence, the dependence on the initial stepsize λ_1 is reduced.

In the light of the above remark, we analyze the convergence of Algorithm 3.4.4 in what follows.

Lemma 3.4.12. Let $\{x_n\}$ be a sequence generated by Algorithm 3.4.4 such that Assumption 3.4.2(a)-(b) hold. Then,

$$\begin{split} &\Gamma_{n+1} - \Gamma_n - \theta_n (\Gamma_n - \Gamma_{n-1}) \\ &\leq \frac{1}{2} (\theta_n + \theta_n^2) \| x_n - x_{n-1} \|^2 - \gamma_n \| x_{n+1} - w_n \|^2 \\ &- \frac{\rho_n}{2} \left(1 - \lambda_n \frac{\mu}{\lambda_{n+1}} \right) \left[||w_n - y_n||^2 + ||z_n - y_n||^2 \right], \\ &l \; \Gamma_n := \frac{1}{2} \| x_n - z \|^2, \; \forall z \in \Gamma. \end{split}$$

where $\gamma_n := \frac{1-\rho_n}{2\rho_n}$ and $\Gamma_n := \frac{1}{2} ||x_n - z||^2, \forall z \in \Gamma.$

Proof. By following the same line of proof used in obtaining (3.4.16), we have

$$\begin{aligned} &\Gamma_{n+1} - \Gamma_n - \theta_n (\Gamma_n - \Gamma_{n-1}) \\ &\leq \frac{1}{2} (\theta_n + \theta_n^2) \|x_n - x_{n-1}\|^2 + \frac{\rho_n - 1}{2\rho_n} \|x_{n+1} - w_n\|^2 \\ &- \frac{\rho_n}{2} \left(\|w_n - y_n\|^2 + \|y_n - z_n\|^2 \right) + \lambda_n \rho_n \langle Aw_n - Ay_n, \ z_n - y_n \rangle. \end{aligned}$$
(3.4.50)

If $\langle Aw_n - Ay_n, z_n - y_n \rangle \leq 0$, then we obtain the conclusion of Lemma 3.4.12 immediately from (3.4.50).

In the case that $\langle Aw_n - Ay_n, z_n - y_n \rangle > 0$, we obtain from (3.4.2) that

$$\langle Aw_n - Ay_n, z_n - y_n \rangle \le \frac{\mu}{2\lambda_{n+1}} \Big(\|w_n - y_n\|^2 + \|z_n - y_n\|^2 \Big).$$
 (3.4.51)

Substituting (3.4.51) into (3.4.50), we obtain the desired conclusion.

Lemma 3.4.13. Let $\{x_n\}$ be a sequence generated by Algorithm 3.4.3 such that assumption (2.5.37) and Assumption 3.4.2(a)-(b) hold. Then, the following inequality holds:

$$\sum_{i=1}^{n-1} t_{i+1,n} \Big[\Big(2\gamma_i (1-\theta_i)^2 - (\theta_i + \theta_i^2) \Big) \|x_i - x_{i-1}\|^2 \\ + 2\gamma_i (1-\theta_i) \Big(\|x_{i+1} - x_i\|^2 - \|x_i - x_{i-1}\|^2 \Big) \Big] \\ \leq 2t_1 |\Gamma_1 - \Gamma_0| + 2\Gamma_0,$$

where $t_{i,n}$ is as defined in (2.5.40).

Proof. By following the same line of proof used in obtaining (3.4.18), we have

$$||x_{n+1} - w_n||^2 \ge (1 - \theta_n)^2 ||x_n - x_{n-1}||^2 + (1 - \theta_n) [||x_{n+1} - x_n||^2 - ||x_n - x_{n-1}||^2].$$
(3.4.52)

Also, by Remark 3.4.11(ii), we obtain that $\lim_{n\to\infty} \lambda_n \frac{\mu}{\lambda_{n+1}} = \mu \in (0, 1)$. Thus, there exists $n_0 \ge 1$ such that $\forall n \ge n_0, 0 < \lambda_n \frac{\mu}{\lambda_{n+1}} < 1$. Hence, we get from Lemma 3.4.12 and (3.4.52) that

$$\Gamma_{n+1} - \Gamma_n - \theta_n (\Gamma_n - \Gamma_{n-1}) \leq \frac{1}{2} (\theta_n + \theta_n^2) \|x_n - x_{n-1}\|^2 - \gamma_n \|x_{n+1} - w_n\|^2$$

$$\leq \left[\frac{1}{2} (\theta_n + \theta_n^2) - \gamma_n (1 - \theta_n)^2 \right] \|x_n - x_{n-1}\|^2$$

$$- \gamma_n (1 - \theta_n) \left[\|x_{n+1} - x_n\|^2 - \|x_n - x_{n-1}\|^2 \right], \ \forall n \geq n_0.$$

The remaining part of the proof is the same as the proof of Lemma 3.4.4.

In similar manner, we have that Lemma 3.4.5 to Lemma 3.4.9 hold for Algorithm 3.4.4. Thus, we have the following theorem whose proof is the same as the proof of Theorem 3.4.10.

Theorem 3.4.14. Let $\{x_n\}$ be a sequence generated by Algorithm 3.4.4 such that assumption (2.5.37), Assumption 3.4.1 and Assumption 3.4.2(a)-(e) hold. Then $\{x_n\}$ converges weakly to an element of $VI(\mathcal{C}, A)$.

Remark 3.4.15. From our analyses, one can see that Assumption 3.4.1 is mainly used to guarantee the summation:

$$\sum_{n=1}^{\infty} \theta_n t_{n+1} \| x_n - x_{n-1} \|^2 < \infty$$
(3.4.53)

obtained in Lemma 3.4.6. Thus, if we assume that (3.4.53) directly, then we do not need Assumption 3.4.1 for the convergence of our methods.

In the case where $\rho_n = 1$, $\forall n \ge 1$, our methods correspond to the inertial subgradient extragradient methods. Note that if $\theta_n \in [0, \theta]$ for every $n \ge 1$, where $\theta \in [0, 1)$, then $t_n \le \frac{1}{(1-\theta)}$ $\forall n \ge 1$. Under these settings, (3.4.53) is guaranteed by the condition

$$\sum_{n=1}^{\infty} \theta_n \|x_n - x_{n-1}\|^2 < \infty.$$
(3.4.54)

Thus, instead of Assumption 3.4.1, we may assume directly that $\theta_n \in [0, \theta], \forall n \geq 1$ and that condition (3.4.54) holds. Recall that this condition has been used by numerous authors to ensure convergence of inertial methods (see, for example, [18, 84, 165, 167, 184] and the references therein). **Remark 3.4.16.** Note that we did not make use of Assumption 3.4.2(c)-(e) in the proof of Lemma 3.4.3-3.4.7. Suppose that \mathcal{H} is a finite dimensional Hilbert space, then under Assumption 3.4.2(a)-(b), we get from Lemma 3.4.7(a) that there exists a subsequence $\{x_{n_k}\}$ of $\{x_n\}$ such that $\{x_{n_k}\}$ converges to some point x^* . By Lemma 3.4.7(b)(c), we get

$$\lim_{k \to \infty} \|w_{n_k} - y_{n_k}\| = 0$$

and

$$\lim_{k \to \infty} \|w_{n_k} - x_{n_k}\| = 0$$

Thus, by the definition of y_n and the continuity of A, we have

$$x^* = \lim_{k \to \infty} x_{n_k} = \lim_{k \to \infty} y_{n_k} = \lim_{k \to \infty} P_{\mathcal{C}}(w_{n_k} - \lambda A w_{n_k}) = P_{\mathcal{C}}(x^* - \lambda A x^*),$$

which implies that $x^* \in VI(\mathcal{C}, A)$.

Now, replacing z by x^* in Lemma 3.4.7(a), we obtain that $\lim_{n \to \infty} ||x_n - x^*||^2$ exists. Since x^* is a cluster point of $\{x_n\}$, we obtain that $\{x_n\}$ converges to x^* .

In summary, in a finite dimensional Hilbert space, our methods require that $\Gamma \neq \emptyset$ and the operator A only needs to be Lipschitz continuous without any form of monotonicity.

To achieve this (convergence without any form of monotonicity) in infinite dimensional Hilbert space, we replace Assumption 3.4.2(d)-(e) with the following:

(d)* If
$$x_n \to x^*$$
 and $\limsup_{n \to \infty} \langle Ax_n, x_n \rangle \leq \langle Ax^*, \bar{x} \rangle$, then $\lim_{n \to \infty} \langle Ax_n, x_n \rangle = \langle Ax^*, x^* \rangle$.

(e)* The set $VI(\mathcal{C}, A) \setminus \Gamma$ is a finite set.

In fact, we have the following theorem.

Theorem 3.4.17. Let $\{x_n\}$ be a sequence generated by Algorithm 3.4.3 (or Algorithm 3.4.4) such that assumption (2.5.37), Assumption 3.4.1, Assumption 3.4.2(a)-(c) and conditions $(d)^*$ -(e)* hold. Then $\{x_n\}$ converges weakly to an element of $VI(\mathcal{C}, A)$.

Proof. First notice that Assumption 3.4.2(d) was used only after (3.4.37) in order to establish the conclusion of Lemma 3.4.8. But from (3.4.37), we have that

$$0 \leq \liminf_{k \to \infty} \langle Ay_{n_k}, x - y_{n_k} \rangle, \ \forall x \in \mathcal{C}.$$

Now, let $\{c_k\}$ be a sequence of positive numbers such that $\lim_{k\to\infty} c_k = 0$ and $\langle Ay_{n_k}, x - y_{n_k} \rangle + c_k > 0, \ \forall k \ge 0, x \in \mathcal{C}$. Then,

$$\langle Ay_{n_k}, x \rangle + c_k > \langle Ay_{n_k}, y_{n_k} \rangle, \ \forall k \ge 0, x \in \mathcal{C}.$$
 (3.4.55)

Note that $y_{n_k} \rightharpoonup x^*$ and $x^* \in \mathcal{C}$. Thus, we have in particular,

$$\langle Ay_{n_k}, x^* \rangle + c_k > \langle Ay_{n_k}, y_{n_k} \rangle, \ \forall k \ge 0.$$
(3.4.56)

Taking limit as $k \to \infty$ in (3.4.56), and using the sequentially weakly continuity of A, we obtain that

$$\langle Ax^*, x^* \rangle \ge \limsup_{k \to \infty} \langle Ay_{n_k}, y_{n_k} \rangle,$$

which by condition $(d)^*$ and (3.4.55) implies that

$$\begin{array}{lll} \langle Ax^*, x^* \rangle &=& \lim_{k \to \infty} \langle Ay_{n_k}, y_{n_k} \rangle \\ &=& \liminf_{k \to \infty} \langle Ay_{n_k}, y_{n_k} \rangle \\ &\leq& \lim_{k \to \infty} \left(\langle Ay_{n_k}, x \rangle + c_k \right) = \langle Ax^*, x \rangle. \end{array}$$

This further implies that $x^* \in VI(\mathcal{C}, A)$.

Now, using condition (e)* and following similar line of proof as in Lemma 3.4.9 and Theorem 3.4.17, we get that $\{x_n\}$ converges weakly to x^* .

Remark 3.4.18.

- (i) If $x_n \rightharpoonup x^*$ and A is sequentially weakly-strongly continuous, then A satisfies condition (d)*.
- (ii) In the numerical experiments, we do not need to consider condition (e) (or (e)*). First note that whenever $||y_n w_n|| < \epsilon$, Algorithm 3.4.3 and Algorithm 3.4.4 terminate in a finite step of the iterations (and y_n is the solution of the VIP (1.2.1)). But from Lemma 3.4.7(b), $\lim_{n \to \infty} ||y_n w_n|| = 0$ and condition (e) (or (e)*) was not used in establishing it.

We now give some remarks regarding the contributions in this work.

Remark 3.4.19.

- (1) If we set the inertial factor $\theta_n = 0$ and relaxation factor $\rho_n = 1$, then our Algorithm 3.4.4 reduces to Algorithm 3.3 of [162]. Note that these parameters (factors) play vital role in improving the convergence rate of iterative methods. In fact, their influence with regards to the numerical performance of iterative schemes was discussed in [129]. Moreover, the benefits gained from incorporating the steps of these two parameters in our algorithms, are further verified in Section 3.4.3. Thus, bearing in mind the importance of these two parameters in iterative algorithms, we can see that our methods significantly improve the methods in [162].
- (2) In the study of inertial methods for solving VIPs (even for monotone mappings), the inertia parameter is usually restricted in $[0, \frac{1}{3})$ and/or required to be nondecreasing

with an upper bound (see, for example, [18, 165, 246, 247]). In many cases, to ensure convergence, authors usually require the inertial parameter to depend on the knowledge of the Lipschitz constant of the cost operator which sometimes is very difficult to estimate in practice (see, for instance, [97]). Another condition usually imposed on the inertial parameter in the literature, is condition (3.4.54), which rely on the sequence $\{x_n\}$. One of the novelties of this work is that we derived a general condition (Assumption 3.4.1) which is weaker than the above conditions used in the literature for ensuring the convergence of inertial methods for VIPs. As a result, we developed a different technique to ensure the convergence of our methods.

(3) In addition to (2) above, bearing in mind the Nestrov's accelerated scheme ([187]), another novelty of this work, is that the assumptions on the inertial and relaxation paramters of this work, allow the case where θ_n converges to a point very close to 1 (see Remark 2.5.12 and the choices in **Experiment 1** of Section 3.4.3), which is very crucial in the study of inertial methods. This is actually where the relaxation effect and the parameter ρ_n come into play, as crucial ingredients of our methods. Thus, the novelty of this work is not only in improving the convergence rate of the methods in [162] but to provide weaker conditions on the inertial parameter in methods for solving VIPs and also to offer a different but unified technique in proving their convergence. Furthermore, we employed condition (Assumption 3.4.1) where joint adjustments of the inertial and relaxation parameters play crucial role (for instance, see **Experiment 1**). Indeed, this is a new perspective in the study of inertial methods for VIPs. Moreover, our study offers many possibilities for future research in this direction; like how to modify Assumption 3.4.1 so that θ_n is allowed to converge to 1?

3.4.3 Numerical experiments

In this section, we give some numerical examples to show the implementation of our proposed methods (Algorithm 3.4.3 and Algorithm 3.4.4). We also compare our new methods with Algorithms 3.1 and 3.3 in [162], and Algorithm 2.1 in [269].

The codes are written in Matlab 2016 (b) and performed on a personal computer with an Intel(R) Core(TM) i5-2600 CPU at 2.30GHz and 8.00 Gb-RAM. In Tables 3.4.1-3.4.5, "Iter." means the number of iterations while "CPU" means the CPU time in seconds.

In our computations, we define $\text{TOL}_n := ||y_n - w_n||$ for Algorithm 3.4.3 and Algorithm 3.4.3. While for Algorithms 3.1 and 3.3 in [162], we define $\text{TOL}_n := ||y_n - x_n||/\min\{\lambda_n, 1\}$ and for Algorithm 2.1 in [269], we define $\text{TOL}_n := ||x_n - z_n|| = ||x_n - P_{\mathcal{C}}(x_n - Ax_n)||$ (as done in [162] and [269], respectively). Then, we use the stopping criterion $\text{TOL}_n < \varepsilon$ for the iterative processes, where ε is the predetermined error. These choices of stopping criterion for these methods are the best to be able to terminate the algorithms based on the examples we consider. As done in [162], we take $\varepsilon = 10^{-6}$ for all Algorithms.

We choose $\mu = 0.5$, $d_n = \frac{100}{(n+1)^{1.1}}$ and $\lambda_1 = 1$ for Algorithm 3.4.4 and Algorithms 3.1 and 3.3 in [162]; $\lambda = \frac{1}{2L}$ for Algorithm 3.4.3; $\gamma = 0.4$ and $\sigma = 0.99$ for Algorithm 2.1 in [269]. These choices are the same as in [162, 269] and are optimal values for these parameters.

Example 3.4.20. Let C = [-1, 1] and

$$Ax = \begin{cases} 2x - 1, & x > 1, \\ x^2, & x \in [-1, 1], \\ -2x - 1, & x < -1. \end{cases}$$

Then, A is quasimonotone and 2-Lipschitz continuous. Also, $\Gamma = \{-1\}$ and $VI(\mathcal{C}, A) = \{-1, 0\}$.

Example 3.4.21. Let $C = \{x \in \mathbb{R}^2 : x_1^2 + x_2^2 \leq 1, 0 \leq x_1\}$ and $A(x_1, x_2) = (-x_1e^{x_2}, x_2)$. Then, A may not be quasimonotone. We can also check that $(1,0) \in \Gamma$ and $VI(C, A) = \{(1,0), (0,0)\}$ (see [162, Section 4]), which by Remark 3.4.16 satisfy our assumptions. This example was also tested in [162].

Example 3.4.22. We next consider the following problem which was also considered in [162, 170, 231]. Let $C = [0, 1]^m$ and $Ax = (f_1x, f_2x, \dots, f_mx)$, where $f_ix = x_{i-1}^2 + x_i^2 + x_{i-1}x_i + x_ix_{i+1} - 2x_{i-1} + 4x_i + x_{i+1} - 1, i = 1, 2, \dots, m$,

 $x_0 = x_{m+1} = 0.$

We test these examples under the following experiments.

Experiment 1

In this first experiment, we check the behavior of our methods by fixing the inertial parameter and varying the relaxation parameter. We do this in order to check the effects of the relaxation parameter on our methods.

For Example 3.4.20: We take $\theta_n = \frac{3n+1}{10n+5}$ with $\rho_n = 1$ (which by Proposition 3.4.5, satisfies assumption (2.5.37) and Assumption 3.4.1), and $\theta_n = \frac{19}{20} - \frac{1}{(n+1)^{\frac{1}{2}}}$ with $\rho_n \in \left\{\frac{1}{20} + \frac{1}{(n+1)^2}, \frac{1}{20} + \frac{2}{(n+1)^3}, \frac{1}{20} + \frac{3}{(n+1)^4}, \frac{1}{20} + \frac{5}{(n+1)^4}\right\}$ (which also satisfies assumption (2.5.37) and Assumption 3.4.1 by Proposition 3.4.7).

Also, we take $x_1 = 1$ and $x_0 = 0.5$ for this example. Since the Lipshitz constant is known for this example, we use Algorithm 3.4.3 and Algorithm 3.4.4 for the experiment and obtain the numerical results listed in Table 3.4.1 and Figure 3.3. From the table and graph, we can see that $\rho_n = 1$ performs better than other choices made for Algorithm 3.4.3 and Algorithm 3.4.4.

For Example 3.4.21 and Example 3.4.22: We take $\theta_n = \frac{9.9}{10} - \frac{1}{n+1}$ with $\rho_n = 1$ (which by Proposition 3.4.6 (see also Remark 2.5.12), satisfies assumption (2.5.37) and Assumption 3.4.1), and $\theta_n = \frac{9}{10} - \frac{1}{(n+1)^{\frac{1}{3}}}$ with $\rho_n \in \{\frac{1}{10} + \frac{1}{n+1}, \frac{1}{11} + \frac{1}{n+1}, \frac{1}{12} + \frac{1}{n+1}, \frac{1}{13} + \frac{1}{n+1}\}$. Also, we take $x_1 = (0.1, 0.5)$ and $x_0 = (0.2, 0.1)$ for Example 3.4.21 while for Example 3.4.22, we choose x_1 and x_0 randomly with m = 50. For these examples, we use Algorithm 3.4.4 for the experiment and obtain the numerical results listed in Table 3.4.2 and Figure 3.4. From the table and graph, we can see that the $\rho_n = \frac{1}{11} + \frac{1}{n+1}$ performs better than other choices made for Example 3.4.21 while $\rho_n = \frac{1}{10} + \frac{1}{10}$ performs better for Example 3.4.22, which validates the benefits sometimes brought by the relaxation parameter.

Experiment 2

In this experiment, we compare our methods with Algorithms 3.1 and 3.3 in [162], and Algorithm 2.1 in [269]. Here, we randomly choose the θ_n and ρ_n from Experiment 1, and then, consider the following cases for the starting points in each example.

For Example 3.4.20: Case I: $x_1 = 0.2$, $x_0 = 0.1$ and Case II: $x_1 = 0.6$, $x_0 = 0.2$.

For Example 3.4.21: Case I: $x_1 = (1, 0.5)$, $x_0 = (0.2, 0.1)$ and Case II: $x_1 = (0.1, 0.2)$, $x_0 = (0.3, 0.5)$.

For Example 3.4.22: Case I: m = 100 and Case II: m = 150 (x_1 and x_0 are randomly taken).

We use Algorithm 3.4.1 and Algorithm 3.4.4 for the comparison in Example 3.4.20 while we use only Algorithm 3.4.4 for the comparison in Example 3.4.21 and Example 3.4.22. The numerical results are given in Tables 3.4.3-3.4.5 and Figures 3.5-3.7. The results show that our methods perform better than Algorithms 3.1 and 3.3 in [162], and Algorithm 2.1 in [269].

$ ho_n$		Algorithm	Algorithm
		3.4.3	3.4.4
1	CPU	0.0016	0.0015
	Iter.	7	3
$\frac{1}{20} + \frac{1}{(n+1)^2}$	CPU	0.0022	0.0027
	Iter.	13	6
$\frac{1}{20} + \frac{2}{(n+1)^3}$	CPU	0.0026	0.0020
	Iter.	8	9
$\frac{1}{20} + \frac{3}{(n+1)^4}$	CPU	0.0028	0.0028
	Iter.	11	12
$\frac{1}{20} + \frac{5}{(n+1)^4}$	CPU	0.0597	0.0146
	Iter.	21	20

Table 3.4.1. Numerical results for Example 3.4.20 (Experiment 1).

Table 3.4.2. Numerical results for Algorithm 3.4.4 (Experiment 1).

ρ_n		Example	Example
		3.4.21	3.4.22
1	CPU	0.0123	0.0232
	Iter.	20	68
$\frac{1}{10} + \frac{1}{n+1}$	CPU	0.0162	0.0107
	Iter.	23	39
$\frac{1}{11} + \frac{1}{n+1}$	CPU	0.0050	0.0304
	Iter.	16	45
$\frac{1}{12} + \frac{1}{n+1}$	CPU	0.0149	0.0346
	Iter.	30	51
$\frac{1}{13} + \frac{1}{n+1}$	CPU	0.0163	0.0257
	Iter.	45	71



Figure 3.3: The behavior of Algorithms 3.4.3 and 3.4.4 for Example 3.4.20 (Experiment 1).



Figure 3.4: The behavior of Algorithm 3.4.4 for Examples 3.4.21 and 3.4.22 (Experiment 1).



Figure 3.5: The behavior of TOL_n for Example 3.4.20 (Experiment 2): Left: Case I; Right: Case II.

Table 3.4.3. Numerical results for	• Example 3.4.20	(Experiment 2).
------------------------------------	-------------------------	-----------------

Cases		Algorithm	Algorithm	Algorithm	Algorithm	Algorithm
		3.4.3	3.4.4	3.1 in [162]	3.3 in [162]	2.1 in [269]
Ι	CPU	0.0223	0.0289	0.0612	0.0621	0.0923
	Iter.	795	922	1002	998	1848
II	CPU	0.0244	0.0272	0.0672	0.0662	0.0900
	Iter.	788	936	996	992	1828

Table 3.4.4. Numerical results for Example 3.4.21 (Experiment 2).

Cases		Algorithm	Algorithm	Algorithm	Algorithm
		3.4.4	3.1 in	3.3 in [162]	2.1 in [269]
			[162]		
Ι	CPU	0.0036	0.0200	0.0149	0.1780
	Iter.	16	55	52	927
II	CPU	0.0062	0.0185	0.0181	0.1289
	Iter.	44	80	77	1427

Table 3.4.5. Numerical results for Example 3.4.22 (Experiment 2).



Figure 3.6: The behavior of TOL_n for Example 3.4.21 (Experiment 2): Left: Case I; Right: Case II.

Cases		Algorithm	Algorithm	Algorithm	Algorithm
		3.4.4	3.1 in	3.3 in [162]	2.1 in [269]
			[162]		
Ι	CPU	0.0049	0.0931	0.0911	0.2021
	Iter.	37	65	62	952
II	CPU	0.0041	0.0916	0.0900	0.2102
	Iter.	42	56	54	2427

3.5 On minimum-norm solutions of quasimonotone variational inequalities with fixed point constraint

The class of quasimonotone mappings are known to be more general and applicable than the classes of pseudomonotone and monotone mappings. However, only very few results can be found in the literature on quasimonotone VIPs and most of these results are on weak convergent algorithms. In this section, we study the quasimonotone VIP with constraint of FPP of quasi-pseudocontractive mappings. We introduce a new inertial Tseng's extragradient method with self-adaptive step size for approximating the minimumnorm solutions of the aforementioned problem in the framework of Hilbert spaces. We prove that the sequence generated by the proposed method converges strongly to a common (minimum-norm) solution of the quasimonotone VIP and FPP of quasi-pseudocontractive mappings without the Lipschitz continuity and sequentially weakly continuity conditions often assumed by authors when solving VIPs. We provide several numerical experiments for the proposed method in comparison with existing methods in the literature. Finally,



Figure 3.7: The behavior of TOL_n for Example 3.4.22 (Experiment 2): Left: Case I; Right: Case II.

we applied our result to image restoration problem. Our result improves, extends and generalizes several of the recently announced results in this direction.

3.5.1 Proposed method

In this section, we present our proposed method. We begin by giving the following assumptions under which our strong convergence result is obtained.

Assumption 3.5.1. Suppose that the following conditions hold:

- (a) The feasible set C is a nonempty, closed and convex subsets of the real Hilbert space \mathcal{H} .
- (b) $A: \mathcal{H} \to \mathcal{H}$ is quasimonotone and uniformly continuous on \mathcal{H} .
- (c) The mapping $A : \mathcal{H} \to \mathcal{H}$ satisfies the following property: whenever $\{x_n\} \subset \mathcal{C}, \quad x_n \rightharpoonup x^*$, one has $||Ax^*|| \le \liminf_{n \to \infty} ||Ax_n||$.
- (d) $T: \mathcal{H} \to \mathcal{H}$ is a quasi-pseudocontractive mapping such that I T is demiclosed at zero.
- (e) The set $\Gamma_D = \Omega_D \cap F(T) \neq \emptyset$.
- $\begin{array}{ll} (f) \ \{\alpha_n\} \subset (0,1), \ \sum_{n=1}^{\infty} \alpha_n = \infty, & \lim_{n \to \infty} \alpha_n = 0, & \{\tau_n\} \subset [a,b] \subset (0,1), & 0 < \zeta < \eta < \\ & \frac{1}{1+\sqrt{1+L^2}}. \end{array} \end{array}$

- (g) Let $\{\epsilon_n\}$ be a positive sequence such that $\lim_{n\to\infty}\frac{\epsilon_n}{\alpha_n}=0.$
- (h) Let $\{\rho_n\}$ be a nonnegative sequence such that $\sum_{n=1}^{\infty} \rho_n < +\infty$.

Algorithm 3.5.2.

Step 1: Select initial point $x_0, x_1 \in \mathcal{H}_1$, let $\lambda_1 > 0, \mu \in (0, 1), \theta \ge 3$ and set n = 1. Given the iterates x_{n-1} and x_n for each $n \ge 1$, choose θ_n such that $0 \le \theta_n \le \overline{\theta}_n$, where

$$\bar{\theta}_n := \begin{cases} \min\left\{\frac{n-1}{n+\theta-1}, \frac{\epsilon_n}{\|x_n - x_{n-1}\|}\right\}, & \text{if } x_n \neq x_{n-1} \\ \frac{n-1}{n+\theta-1}, & \text{otherwise.} \end{cases}$$
(3.5.1)

Step 2: Compute

$$w_n = (1 - \alpha_n) \Big(x_n + \theta_n (x_n - x_{n-1}) \Big).$$

Step 3: Compute

$$y_n = P_{\mathcal{C}}(w_n - \lambda_n A w_n).$$

If $w_n = y_n$ and $Tw_n = w_n$, then stop: w_n is a solution of the Problem (1.2.3). Otherwise, go to Step 4.

Step 4: Compute

$$z_n = y_n - \lambda_n (Ay_n - Aw_n).$$

Step 5 Compute

$$x_{n+1} = (1 - \tau_n)w_n + \tau_n \Big((1 - \zeta)I + \zeta T((1 - \eta)I + \eta T) \Big) z_n,$$

where

$$\lambda_{n+1} = \begin{cases} \min\left\{\frac{\mu \|w_n - y_n\|}{\|Aw_n - Ay_n\|}, \lambda_n + \rho_n\right\} & \text{if } Aw_n - Ay_n \neq 0, \\ \lambda_n + \rho_n, & \text{otherwise.} \end{cases}$$
(3.5.2)

Set n := n + 1 and go back to **Step 1**.

Remark 3.5.1.

- Step 1 of Algorithm 3.5.2 is easily implemented since the knowledge of the value of $||x_n x_{n-1}||$ is prior known before choosing θ_n .
- The algorithm involves only one projection onto the feasible set C per iteration, which makes the implementation of the algorithm computationally less expensive.
- Step 5 of Algorithm 3.5.2 guarantees the strong convergence of the algorithm to a point in the solution set of the problem.

- The step size (3.5.2) is self adaptive, and generates a non-monotonic sequence of step sizes. This makes the implementation of Algorithm 3.5.2 possible without the prior knowledge of the Lipschitz constant.
- The condition (c) is strictly weaker than the sequentially weakly continuity condition, which is commonly used in the literature when solving pseudomonotone VIP.

Remark 3.5.2. By conditions (f) and (g), from (3.5.1) we observe that

$$\lim_{n \to \infty} \theta_n ||x_n - x_{n-1}|| = 0 \text{ and } \lim_{n \to \infty} \frac{\theta_n}{\alpha_n} ||x_n - x_{n-1}|| = 0.$$
 (3.5.3)

First, we prove some lemmas which will be employed in proving our strong convergence theorem for the proposed algorithm.

3.5.2 Convergence analysis

Lemma 3.5.3. Let $\{\lambda_n\}$ be a sequence generated by Algorithm 3.5.2. Then, we have $\lim_{n \to \infty} \lambda_n = \lambda$, where $\lambda \in \left[\min\{\frac{\mu}{\mathcal{K}}, \lambda_1\}, \lambda_1 + \Phi\right]$ and $\Phi = \sum_{n=1}^{\infty} \rho_n$.

Proof. Since A is uniformly continuous, then by (2.1.4) it follows that for any given $\epsilon > 0$, there exists $M < \infty$ such that $||Aw_n - Ay_n|| \le M ||w_n - y_n|| + \epsilon$. Thus, when $Aw_n - Ay_n \ne 0$ for all $n \ge 1$ we have

$$\frac{\mu \|w_n - y_n\|}{\|Aw_n - Ay_n\|} \ge \frac{\mu \|w_n - y_n\|}{M \|w_n - y_n\| + \epsilon} = \frac{\mu \|w_n - y_n\|}{(M + \epsilon_1) \|w_n - y_n\|} = \frac{\mu}{\mathcal{K}},$$

where $\epsilon = \epsilon_1 ||w_n - y_n||$ for some $\epsilon_1 \in (0, 1)$ and $\mathcal{K} = M + \epsilon_1$. Therefore, by the definition of λ_{n+1} , the sequence $\{\lambda_n\}$ has lower bound $\min\{\frac{\mu}{\mathcal{K}}, \lambda_1\}$ and upper bound $\lambda_1 + \Phi$. By Lemma 2.5.34, it follows that $\lim_{n \to \infty} \lambda_n$ exists and denoted by $\lambda = \lim_{n \to \infty} \lambda_n$. Clearly, we have $\lambda \in [\min\{\frac{\mu}{\mathcal{K}}, \lambda_1\}, \lambda_1 + \Phi]$.

Remark 3.5.4. The step size generated in Algorithm 3.5.2 is permitted to increase per iteration which reduces its dependence on the initial step size. The step size may not increase when n is large enough. In our convergence analysis, we assume that Algorithm 3.5.2 does not terminate in finite number of iterations.

Lemma 3.5.5. Let $\{x_n\}$ be a sequence generated by Algorithm 3.5.2 under Assumption 3.5.1. Then

$$||z_n - p||^2 \le ||w_n - p||^2 - \left(1 - \frac{\lambda_n^2 \mu^2}{\lambda_{n+1}^2}\right) ||y_n - w_n||^2.$$

Proof. From (3.5.2), we obtain

$$\lambda_{n+1} = \min\left\{\frac{\mu \|w_n - y_n\|}{\|Aw_n - Ay_n\|}, \lambda_n + \rho_n\right\} \le \frac{\mu \|w_n - y_n\|}{\|Aw_n - Ay_n\|},$$

which implies that

$$||Aw_n - Ay_n|| \le \frac{\mu}{\lambda_{n+1}} ||w_n - y_n||, \quad \forall n \ge 1.$$
 (3.5.4)

Let $p \in \Gamma_D$. From Lemma 2.1.1 and the definition of z_n in **Step 4**, we have

$$\begin{aligned} \|z_{n} - p\|^{2} &\leq \|y_{n} - \lambda_{n}(Ay_{n} - Aw_{n}) - p\|^{2} \\ &= \|y_{n} - p\|^{2} + \lambda_{n}^{2} \|Ay_{n} - Aw_{n}\|^{2} - 2\lambda_{n} \langle Ay_{n} - Aw_{n}, y_{n} - p \rangle \\ &= \|w_{n} - p\|^{2} + \|y_{n} - w_{n}\|^{2} + 2\langle y_{n} - w_{n}, w_{n} - p \rangle + \lambda_{n}^{2} \|Ay_{n} - Aw_{n}\|^{2} \\ &- 2\lambda_{n} \langle Ay_{n} - Aw_{n}, y_{n} - p \rangle \\ &= \|w_{n} - p\|^{2} + \|y_{n} - w_{n}\|^{2} - 2\langle y_{n} - w_{n}, y_{n} - w_{n} \rangle + 2\langle y_{n} - w_{n}, y_{n} - p \rangle \\ &- 2\lambda_{n} \langle Ay_{n} - Aw_{n}, y_{n} - p \rangle + \lambda_{n}^{2} \|Ay_{n} - Aw_{n}\|^{2} \\ &= \|w_{n} - p\|^{2} - \|y_{n} - w_{n}\|^{2} + 2\langle y_{n} - w_{n}, y_{n} - p \rangle \\ &- 2\lambda_{n} \langle Ay_{n} - Aw_{n}, y_{n} - p \rangle + \lambda_{n}^{2} \|Ay_{n} - Aw_{n}\|^{2}. \end{aligned}$$

$$(3.5.5)$$

Since $y_n = P_{\mathcal{C}}(w_n - \lambda_n w_n)$ and $p \in \Gamma_D \subset C$, we obtain from the characteristic property of $P_{\mathcal{C}}$ that

$$\langle y_n - w_n + \lambda_n A w_n, y_n - p \rangle \le 0,$$

which implies that

$$\langle y_n - w_n, y_n - p \rangle \le -\lambda_n \langle Aw_n, y_n - p \rangle.$$
 (3.5.6)

Also since $y_n \in \mathcal{C}$ and $p \in \Gamma_D \subset \Omega_D$, we have

$$\langle Ay_n, y_n - p \rangle \ge 0, \quad \forall \ n \ge 0.$$
 (3.5.7)

Applying (3.5.4), (3.5.6) and (3.5.7) in (3.5.5), we obtain

$$\begin{aligned} \|z_{n} - p\|^{2} &\leq \|w_{n} - p\|^{2} - \|y_{n} - w_{n}\|^{2} - 2\lambda_{n}\langle Aw_{n}, y_{n} - p \rangle - 2\lambda_{n}\langle Ay_{n} - Aw_{n}, y_{n} - p \rangle \\ &+ \lambda_{n}^{2} \|Ay_{n} - Aw_{n}\|^{2} \\ &= \|w_{n} - p\|^{2} - \|y_{n} - w_{n}\|^{2} - 2\lambda_{n}\langle Ay_{n}, y_{n} - p \rangle + \lambda_{n}^{2} \|Ay_{n} - Aw_{n}\|^{2} \\ &\leq \|w_{n} - p\|^{2} - \|y_{n} - w_{n}\|^{2} + \lambda_{n}^{2} \frac{\mu^{2}}{\lambda_{n+1}^{2}} \|y_{n} - w_{n}\|^{2} \\ &= \|w_{n} - p\|^{2} - \left(1 - \frac{\lambda_{n}^{2} \mu^{2}}{\lambda_{n+1}^{2}}\right) \|y_{n} - w_{n}\|^{2}. \end{aligned}$$
(3.5.8)

Since the limit

$$\lim_{n \to \infty} \left(1 - \frac{\lambda_n^2 \mu^2}{\lambda_{n+1}^2} \right) = 1 - \mu^2 > 0, \qquad (3.5.9)$$

there exists $n_0 > 0$ such that for all $n > n_0$, we have $\left(1 - \frac{\lambda_n^2 \mu^2}{\lambda_{n+1}^2}\right) > 0$. Hence, applying (3.5.9) to (3.5.8), we have that

$$||z_n - p|| \le ||w_n - p||. \tag{3.5.10}$$

Lemma 3.5.6. Let $\{x_n\}$ be a sequence generated by Algorithm 3.5.2 under Assumption 3.5.1. Then, $\{x_n\}$ is bounded.

Proof. Let $p \in \Gamma_D$. From the definition of w_n in Step 2, we have

$$\begin{aligned} \|w_n - p\| &= \|(1 - \alpha_n)(x_n + \theta_n(x_n - x_{n-1})) - p\| \\ &= \|(1 - \alpha_n)(x_n - p) + (1 - \alpha_n)\theta_n(x_n - x_{n-1}) - \alpha_n p\| \\ &\leq (1 - \alpha_n)\|x_n - p\| + (1 - \alpha_n)\theta_n\|x_n - x_{n-1}\| + \alpha_n\|p\| \\ &= (1 - \alpha_n)\|x_n - p\| + \alpha_n\Big[(1 - \alpha_n)\frac{\theta_n}{\alpha_n}\|x_n - x_{n-1}\| + \|p\|\Big]. \end{aligned}$$

By (3.5.3), we have

$$\lim_{n \to \infty} \left[(1 - \alpha_n) \frac{\theta_n}{\alpha_n} \| x_n - x_{n-1} \| + \| p \| \right] = \| p \|.$$

Thus, there exists a constant $M_1 > 0$ such that $(1 - \alpha_n) \frac{\theta_n}{\alpha_n} ||x_n - x_{n-1}|| + ||p|| \le M_1$ for all $n \in \mathbb{N}$. This implies that

$$||w_n - p|| \le (1 - \alpha_n) ||x_n - p|| + \alpha_n M_1.$$
(3.5.11)

Hence, from (3.5.10) and (3.5.11) we have

$$||z_n - p|| \le ||w_n - p|| \le (1 - \alpha_n) ||x_n - p|| + \alpha_n M_1.$$
(3.5.12)

Let $K = (1 - \zeta)I + \zeta T((1 - \eta)I + \eta T)$ then, from the definition of x_{n+1} in **Step 5**, Lemma 2.5.27, Lemma 3.2.8 and (3.5.8), we have

$$\begin{aligned} \|x_{n+1} - p\|^2 &= \|(1 - \tau_n)w_n + \tau_n K z_n - p\|^2 \\ &= \|(1 - \tau_n)(w_n - p) + \tau_n (K z_n - p)\|^2 \\ &= (1 - \tau_n)\|w_n - p\|^2 + \tau_n \|K z_n - p\|^2 - \tau_n (1 - \tau_n)\|K z_n - w_n\|^2 \\ &\leq (1 - \tau_n)\|w_n - p\|^2 + \tau_n \|z_n - p\|^2 - \tau_n (1 - \tau_n)\|K z_n - w_n\|^2 \\ &\leq (1 - \tau_n)\|w_n - p\|^2 + \tau_n \Big[\|w_n - p\|^2 - \Big(1 - \frac{\lambda_n^2 \mu^2}{\lambda_{n+1}^2}\Big)\|y_n - w_n\|^2\Big] \\ &- \tau_n (1 - \tau_n)\|K z_n - w_n\|^2 \\ &= \|w_n - p\|^2 - \tau_n \Big(1 - \frac{\lambda_n^2 \mu^2}{\lambda_{n+1}^2}\Big)\|y_n - w_n\|^2 - \tau_n (1 - \tau_n)\|K z_n - w_n \quad (3.5.13) \end{aligned}$$

By the condition on τ_n , (3.5.9) and (3.5.11), we have

$$||x_{n+1} - p|| \le ||w_n - p|| \le (1 - \alpha_n) ||x_n - p|| + \alpha_n M_1 \le \max \{||x_n - p||, M_1\} \le \max \{||x_{n_0} - p||, M_1\}.$$

Hence, $\{x_n\}$ is bounded. Consequently, $\{w_n\}, \{y_n\}$ and $\{z_n\}$ are also bounded.

Lemma 3.5.7. Assume that $\{w_n\}$ and $\{y_n\}$ are sequences generated by Algorithm 3.5.2 under Assumption 3.5.1 and $\lim_{k\to\infty} ||w_{n_k} - y_{n_k}|| = 0$. Suppose there exists a subsequence $\{w_{n_k}\}$ of $\{w_n\}$ such that $\{w_{n_k}\}$ converges weakly to some $x^* \in \mathcal{H}$ as $k \to \infty$, then we having one of the following: $x^* \in \Omega_D$ or $Ax^* = 0$.

Proof. Since $\{w_n\}$ is bounded, then $w_{\omega}(w_n)$ is nonempty. Let $x^* \in w_{\omega}(w_n)$ be an arbitrary element. Then, there exists a subsequence $\{w_{n_k}\}$ of $\{w_n\}$ such that $w_{n_k} \rightharpoonup x^*$ as $k \rightarrow \infty$. By the hypothesis of the lemma we have that $y_{n_k} \rightharpoonup x^* \in \mathcal{C}$ as $k \rightarrow \infty$. Now, we divide the proof into two cases.

Case 1: If $\limsup_{k\to\infty} ||Ay_{n_k}|| = 0$, then $\lim_{k\to\infty} ||Ay_{n_k}|| = \liminf_{k\to\infty} ||Ay_{n_k}|| = 0$. Since $\{y_{n_k}\}$ converges weakly to $x^* \in \mathcal{C}$ and A satisfies condition (c), we have

$$0 \le ||Ax^*|| \le \liminf_{k \to \infty} ||Ay_{n_k}|| = 0.$$

Hence, we obtain $Ax^* = 0$.

Case 2: If $\limsup_{k\to\infty} ||Ay_{n_k}|| > 0$, without loss of generality, we let $\lim_{k\to\infty} ||Ay_{n_k}|| = N > 0$. Then there exists a $K' \in \mathbb{N}$ such that $||Ay_{n_k}|| > \frac{N}{2}$, $\forall k \ge K'$. By the characteristic property of $P_{\mathcal{C}}$, we have

$$\langle w_{n_k} - \lambda_{n_k} A w_{n_k} - y_{n_k}, x - y_{n_k} \rangle \le 0, x \in \mathcal{C},$$

which implies that

$$\frac{1}{\lambda_{n_k}} \left\langle w_{n_k} - y_{n_k}, \ x - y_{n_k} \right\rangle \le \left\langle A w_{n_k}, \ x - y_{n_k} \right\rangle, \quad \forall \ x \in \mathcal{C}$$

Consequently, we have

$$\frac{1}{\lambda_{n_k}} \langle w_{n_k} - y_{n_k}, x - y_{n_k} \rangle + \langle A w_{n_k}, y_{n_k} - w_{n_k} \rangle \le \langle A w_{n_k}, x - w_{n_k} \rangle, \quad \forall x \in \mathcal{C}.$$
(3.5.14)

Applying $\lim_{k\to\infty} \lambda_{n_k} = \lambda > 0$ and the hypothesis of the lemma to (3.5.14), we have

$$0 \le \liminf_{k \to \infty} \langle Aw_{n_k}, x - w_{n_k} \rangle \le \limsup_{k \to \infty} \langle Aw_{n_k}, x - w_{n_k} \rangle < +\infty, \quad \forall x \in \mathcal{C}.$$
(3.5.15)

Observe that

$$\langle Ay_{n_k}, x - y_{n_k} \rangle = \langle Ay_{n_k} - Aw_{n_k}, x - w_{n_k} \rangle + \langle Aw_{n_k}, x - w_{n_k} \rangle + \langle Ay_{n_k}, w_{n_k} - y_{n_k} \rangle.$$

Since A is uniformly continuous on \mathcal{H} , we have from the hypothesis of the lemma that

$$\lim_{k \to \infty} \|Aw_{n_k} - Ay_{n_k}\| = 0. \tag{3.5.16}$$

From the hypothesis of the lemma, (3.5.14), (3.5.15) and (3.5.16), we have

$$0 \le \liminf_{k \to \infty} \langle Ay_{n_k}, x - y_{n_k} \rangle \le \limsup_{k \to \infty} \langle Ay_{n_k}, x - y_{n_k} \rangle < +\infty, \quad \forall x \in \mathcal{C}.$$
(3.5.17)

If $\limsup_{k\to\infty} \langle Ay_{n_k}, x - y_{n_k} \rangle > 0$, $\forall x \in \mathcal{C}$, then there exists a subsequence $\{y_{n_{k_j}}\}$ of $\{y_{n_k}\}$ such that

$$\lim_{j \to \infty} \left\langle A y_{n_{k_j}}, \ x - y_{n_{k_j}} \right\rangle > 0.$$

Thus, there exists $j_0 \in \mathbb{N}$ such that $\langle Ay_{n_{k_j}}, x - y_{n_{k_j}} \rangle > 0$, $\forall j \ge j_0$, which by the quasimonotonicity of A on \mathcal{H} implies that $\langle Ax, x - y_{n_{k_j}} \rangle \ge 0$, $\forall x \in \mathcal{C}, j \ge j_0$. Letting $j \to \infty$, we have $x^* \in \Omega_D$.

If $\limsup \langle Ay_{n_k}, x - y_{n_k} \rangle = 0$, $\forall x \in \mathcal{C}$, we obtain from (3.5.17) that

 $\lim_{k \to \infty} \langle Ay_{n_k}, x - y_{n_k} \rangle = \limsup_{k \to \infty} \langle Ay_{n_k}, x - y_{n_k} \rangle = \liminf_{k \to \infty} \langle Ay_{n_k}, x - y_{n_k} \rangle = 0.$ (3.5.18)

Let $\delta_k = \left| \langle Ay_{n_k}, x - y_{n_k} \rangle \right| + \frac{1}{k+1}$. Then, we have

$$\langle Ay_{n_k}, \ x - y_{n_k} \rangle + \delta_k > 0. \tag{3.5.19}$$

Let $u_{n_k} = \frac{Ay_{n_k}}{\|Ay_{n_k}\|^2} \quad \forall \ k \ge K'$, we have $\langle Ay_{n_k}, u_{n_k} \rangle = 1$. Then, from (3.5.19) we have that for all $k \ge K'$

$$\langle Ay_{n_k}, x + \delta_k u_{n_k} - y_{n_k} \rangle > 0.$$

By the quasimonotonicity of A, we have for all $k \ge K'$

$$\langle A(x+\delta_k u_{n_k}), \ x+\delta_k u_{n_k}-y_{n_k}\rangle \ge 0.$$
(3.5.20)

Consequently, by applying (2.1.4) and (3.5.20) we have $\forall x \in \mathcal{C}, k \geq K'$

$$\langle Ax, x + \delta_k u_{n_k} - y_{n_k} \rangle = \langle Ax - A(x + \delta_k u_{n_k}), x + \delta_k u_{n_k} - y_{n_k} \rangle + \langle A(x + \delta_k u_{n_k}), x + \delta_k u_{n_k} - y_{n_k} \rangle \geq \langle Ax - A(x + \delta_k u_{n_k}), x + \delta_k u_{n_k} - y_{n_k} \rangle \geq - \|Ax - A(x + \delta_k u_{n_k})\| \|x + \delta_k u_{n_k} - y_{n_k}\| \geq -\delta_k (M' + \epsilon_2') \|u_{n_k}\| \|x + \delta_k u_{n_k} - y_{n_k}\| (where \ \epsilon' = \epsilon_2' \|\delta_k u_{n_k}\| \ for some \ \epsilon_2' \in (0, 1), M' < +\infty) = -\delta_k \frac{(M' + \epsilon_2')}{\|Ay_{n_k}\|} \|x + \delta_k u_{n_k} - y_{n_k}\| \geq -\delta_k \frac{2(M' + \epsilon_2')}{N} \|x + \delta_k u_{n_k} - y_{n_k}\|.$$
(3.5.21)

Letting $k \to \infty$ in (3.5.21), and applying the fact that $\lim_{k\to\infty} \delta_k = 0$ together with the boundedness of $\{\|x + \delta_k u_{n_k} - y_{n_k}\|\}$, we have

$$\langle Ax, x - x^* \rangle \ge 0, \quad \forall \ x \in \mathcal{C},$$

which implies that $x^* \in \Omega_D$ as desired.

Lemma 3.5.8. Let $\{x_n\}$ be a sequence generated by Algorithm 3.5.2 under Assumption 3.5.1. Then,

$$\|x_{n+1} - p\|^2 \le (1 - \alpha_n) \|x_n - p\|^2 + \alpha_n d_n - \tau_n \left(1 - \frac{\lambda_n^2 \mu^2}{\lambda_{n+1}^2}\right) \|y_n - w_n\|^2$$
$$- \tau_n (1 - \tau_n) \|Kz_n - w_n\|^2.$$

Proof. Let $p \in \Gamma_D$. From Lemma 2.1.1 and the definition of w_n , we have

$$||w_{n} - p||^{2} = ||(1 - \alpha_{n})(x_{n} - p) + (1 - \alpha_{n})\theta_{n}(x_{n} - x_{n-1}) - \alpha_{n}p||^{2}$$

$$\leq ||(1 - \alpha_{n})(x_{n} - p) + (1 - \alpha_{n})\theta_{n}(x_{n} - x_{n-1})||^{2} + 2\alpha_{n}\langle -p, \quad w_{n} - p\rangle$$

$$\leq (1 - \alpha_{n})^{2}||x_{n} - p||^{2} + 2(1 - \alpha_{n})\theta_{n}||x_{n} - p||||x_{n} - x_{n-1}|| + \theta_{n}^{2}||x_{n} - x_{n-1}||^{2}$$

$$(3.5.22)$$

$$+ 2\alpha_{n}\langle -p, \quad w_{n} - x_{n+1}\rangle + 2\alpha_{n}\langle -p, \quad x_{n+1} - p\rangle$$

Applying (3.5.22) in (3.5.13), we have

$$\begin{aligned} \|x_{n+1} - p\|^2 &\leq (1 - \alpha_n)^2 \|x_n - p\|^2 + 2(1 - \alpha_n)\theta_n \|x_n - p\| \|x_n - x_{n-1}\| + \theta_n^2 \|x_n - x_{n-1}\|^2 \\ &+ 2\alpha_n \langle -p, \ w_n - x_{n+1} \rangle + 2\alpha_n \langle -p, \ x_{n+1} - p \rangle - \tau_n \left(1 - \frac{\lambda_n^2 \mu^2}{\lambda_{n+1}^2}\right) \|y_n - w_n\|^2 \\ &- \tau_n (1 - \tau_n) \|K z_n - w_n\|^2 \\ &\leq (1 - \alpha_n) \|x_n - p\|^2 \\ &+ \alpha_n \Big[2(1 - \alpha_n) \|x_n - p\| \frac{\theta_n}{\alpha_n} \|x_n - x_{n-1}\| + \theta_n \|x_n - x_{n-1}\| \frac{\theta_n}{\alpha_n} \|x_n - x_{n-1}\| \\ &+ 2 \|p\| \|w_n - x_{n+1}\| + 2\langle p, \ p - x_{n+1} \rangle \Big] - \tau_n \left(1 - \frac{\lambda_n^2 \mu^2}{\lambda_{n+1}^2}\right) \|y_n - w_n\|^2 \\ &- \tau_n (1 - \tau_n) \|K z_n - w_n\|^2 \\ &= (1 - \alpha_n) \|x_n - p\|^2 + \alpha_n d_n - \tau_n \left(1 - \frac{\lambda_n^2 \mu^2}{\lambda_{n+1}^2}\right) \|y_n - w_n\|^2 \\ &- \tau_n (1 - \tau_n) \|K z_n - w_n\|^2, \end{aligned}$$

where

$$d_n = \left[2(1 - \alpha_n) \| x_n - p \| \frac{\theta_n}{\alpha_n} \| x_n - x_{n-1} \| + \theta_n \| x_n - x_{n-1} \| \frac{\theta_n}{\alpha_n} \| x_n - x_{n-1} \| + 2 \| p \| \| w_n - x_{n+1} \| + 2 \langle p, p - x_{n+1} \rangle \right].$$

This completes the proof.

Theorem 3.5.9. Let $\{x_n\}$ be a sequence generated by Algorithm 3.5.2 such that Assumption 3.5.1 holds and $Ax \neq 0, \forall x \in C$. Then, the sequence $\{x_n\}$ converges strongly to a point $\bar{x} \in \Gamma_D \subset \Gamma$, where $\bar{x} = P_{\Gamma_D}(0)$.

Proof. Let $\bar{x} \in \Gamma_D \subset \Gamma$, where $\bar{x} = P_{\Gamma_D}(0)$. Then, it follows from Lemma 3.5.8 that

$$\|x_{n+1} - \bar{x}\|^2 \le (1 - \alpha_n) \|x_n - \bar{x}\|^2 + \alpha_n d_n \tag{3.5.23}$$

where

$$d_n = \left[2(1 - \alpha_n) \| x_n - \bar{x} \| \frac{\theta_n}{\alpha_n} \| x_n - x_{n-1} \| + \theta_n \| x_n - x_{n-1} \| \frac{\theta_n}{\alpha_n} \| x_n - x_{n-1} \| + 2 \| \bar{x} \| \| w_n - x_{n+1} \| + 2 \langle \bar{x}, \ \bar{x} - x_{n+1} \rangle \right].$$

Now, we claim that the sequence $\{\|x_n - \bar{x}\|\}$ converges to zero. To show this, by Lemma 2.5.36 it suffices to show that $\limsup_{k \to \infty} d_{n_k} \leq 0$ for every subsequence $\{\|x_{n_k} - \bar{x}\|\}$ of $\{\|x_n - \bar{x}\|\}$ satisfying

$$\liminf_{k \to \infty} \left(\|x_{n_{k+1}} - \bar{x}\| - \|x_{n_k} - \bar{x}\| \right) \ge 0.$$
(3.5.24)

Suppose that $\{\|x_{n_k} - \bar{x}\|\}$ is a subsequence of $\{\|x_n - \bar{x}\|\}$ such that (3.5.24) holds. Again, from Lemma 3.5.8, we obtain

$$\tau_{n_k} \left(1 - \frac{\lambda_{n_k}^2 \mu^2}{\lambda_{n_{k+1}}^2} \right) \|y_{n_k} - w_{n_k}\|^2 + \tau_{n_k} (1 - \tau_{n_k}) \|K z_{n_k} - w_{n_k}\|^2 \le (1 - \alpha_{n_k}) \|x_{n_k} - \bar{x}\|^2 - \|x_{n_{k+1}} - \bar{x}\|^2 + \alpha_{n_k} d_{n_k}.$$

By (3.5.24) and the condition on α_{n_k} it follows that

$$\lim_{k \to \infty} \left[\tau_{n_k} \left(1 - \frac{\lambda_{n_k}^2 \mu^2}{\lambda_{n_{k+1}}^2} \right) \| y_{n_k} - w_{n_k} \|^2 + \tau_{n_k} (1 - \tau_{n_k}) \| K z_{n_k} - w_{n_k} \|^2 \right] = 0.$$

By (3.5.9) and the conditions on the control parameters, we obtain

$$\lim_{k \to \infty} \|y_{n_k} - w_{n_k}\| = 0, \quad \lim_{k \to \infty} \|K z_{n_k} - w_{n_k}\| = 0.$$
(3.5.25)

From (3.5.25), we have

$$\lim_{k \to \infty} \|K z_{n_k} - y_{n_k}\| = 0.$$

Moreover, from the definition of z_n and by applying (3.5.4) and (3.5.25), we get

$$||z_{n_k} - y_{n_k}|| = \lambda_n ||Aw_{n_k} - Ay_{n_k}|| \le \frac{\mu \lambda_{n_k}}{\lambda_{n_k+1}} ||w_{n_k} - y_{n_k}|| \to 0, \quad k \to \infty.$$
(3.5.26)

From the definition of $x_{n_{k+1}}$ in Step 5 and (3.5.25), we have

$$||x_{n_{k+1}} - w_{n_k}|| = ||(1 - \tau_{n_k})w_{n_k} + \tau_{n_k}Kz_{n_k} - w_{n_k}||$$

= $\tau_{n_k}||Kz_{n_k} - w_{n_k}|| \to 0, \quad k \to \infty.$ (3.5.27)

From (3.5.25) and (3.5.27), we obtain

$$\lim_{k \to \infty} \|x_{n_{k+1}} - y_{n_k}\| = 0. \tag{3.5.28}$$

Now, from Step 2 and by Remark 3.5.3, we get

$$\begin{aligned} \|w_{n_{k}} - x_{n_{k}}\| &= \|(1 - \alpha_{n_{k}})(x_{n_{k}} + \theta_{n_{k}}(x_{n_{k}} - x_{n_{k}-1})) - x_{n_{k}}\| \\ &= \|(1 - \alpha_{n_{k}})(x_{n_{k}} - x_{n_{k}}) + (1 - \alpha_{n_{k}})\theta_{n_{k}}(x_{n_{k}} - x_{n_{k}-1}) - \alpha_{n_{k}}x_{n_{k}}\| \\ &\leq (1 - \alpha_{n_{k}})\|x_{n_{k}} - x_{n_{k}}\| + (1 - \alpha_{n_{k}})\theta_{n_{k}}\|x_{n_{k}} - x_{n_{k}-1}\| + \alpha_{n_{k}}\|x_{n_{k}}\| \\ &\to 0, \quad k \to \infty. \end{aligned}$$

$$(3.5.29)$$

From (3.5.25)-(3.5.29), we have

$$\lim_{k \to \infty} \|y_{n_k} - x_{n_k}\| = 0, \quad \lim_{k \to \infty} \|K z_{n_k} - z_{n_k}\| = 0, \quad \lim_{k \to \infty} \|z_{n_k} - x_{n_k}\| = 0.$$
(3.5.30)

Moreover, from (3.5.28) and (3.5.30) we obtain

$$\lim_{k \to \infty} \|x_{n_{k+1}} - x_{n_k}\| = 0. \tag{3.5.31}$$

Since $\{x_n\}$ is bounded, we have that $w_{\omega}(x_n) \neq \emptyset$. Let $z \in w_{\omega}(x_n)$ be arbitrary. Then, there exist a subsequence $\{x_{n_k}\}$ of $\{x_n\}$ such that $x_{n_k} \rightharpoonup z$ as $k \rightarrow \infty$. By (3.5.29) and (3.5.30) we have $w_{\omega}(x_n) = w_{\omega}(y_n) = w_{\omega}(w_n)$. Since $y_n \in C$ and C is weakly-closed, we have $z \in C$. By the assumption that $Ax \neq 0$, $\forall x \in C$, it follows that $Az \neq 0$. Thus, by Lemma 3.5.7 and (3.5.25), it follows that $z \in \Omega_D$. Consequently, we have $w_{\omega}(x_n) \subset \Omega_D$. Also, since $\lim_{n\to\infty} ||x_{n_k} - z_{n_k}|| = 0$, we have $z_{n_k} \rightharpoonup z$ as $k \rightarrow \infty$. Since I - K is demiclosed at zero, it follows from (3.5.30) and Lemma 3.2.8(a) that $z \in F(K) = F(T)$. Consequently, we have $w_{\omega}(x_n) \subset F(T)$. It therefore follows that $w_{\omega}(x_n) \subset \Gamma_D$.

Next, by the boundedness of $\{x_{n_k}\}$, it follows that there exists a subsequence $\{x_{n_{k_j}}\}$ of $\{x_{n_k}\}$ which converges weakly to some $\hat{x} \in \mathcal{H}$, and such that

$$\lim_{j \to \infty} \left\langle \bar{x}, \ \bar{x} - x_{n_{k_j}} \right\rangle = \limsup_{k \to \infty} \left\langle \bar{x}, \ \bar{x} - x_{n_k} \right\rangle.$$
(3.5.32)

From (3.5.32) and the fact that $\bar{x} = P_{\Gamma_D}(0)$ we have

$$\limsup_{k \to \infty} \langle \bar{x}, \ \bar{x} - x_{n_k} \rangle = \lim_{j \to \infty} \left\langle \bar{x}, \ \bar{x} - x_{n_{k_j}} \right\rangle = \langle \bar{x}, \ \bar{x} - \hat{x} \rangle \le 0.$$
(3.5.33)

From (3.5.31) and (3.5.33), it follows that

$$\limsup_{k \to \infty} \left\langle \bar{x}, \ \bar{x} - x_{n_{k+1}} \right\rangle = \limsup_{k \to \infty} \left\langle \bar{x}, \ \bar{x} - x_{n_k} \right\rangle = \left\langle \bar{x}, \ \bar{x} - \hat{x} \right\rangle \le 0.$$
(3.5.34)

Thus, by (3.5.34) and the fact that $\lim_{k\to\infty} ||x_{n_k+1} - w_{n_k}|| = \lim_{k\to\infty} \frac{\theta_{n_k}}{\alpha_{n_k}} ||x_{n_k} - x_{n_{k-1}}|| = 0$, we obtain $\limsup_{k\to\infty} d_{n_k} \leq 0$. Now, applying Lemma 2.5.36 to (3.5.23) we have $\lim_{n\to\infty} ||x_n - \bar{x}|| = 0$. Therefore, $\{x_n\}$ converges strongly to \bar{x} .

3.5.3 Numerical experiments

In this section, we discuss the numerical behavior of our proposed Algorithm 3.5.2 and also compare it with the methods of Salahuddin [217] (see Appendix 3.5.15), Liu et.al [162] (see Appendix 3.5.16), Ogwo *et al.* [193] (see Appendix 3.5.3), Alakoya *et al.* [12] (see Appendix 3.5.4) and Yin et.al [270] (see Appendix 3.5.17). We perform all implementations using Matlab 2016 (b), installed on a personal computer with Intel(R) Core(TM) i5-2600 CPU@2.30GHz and 8.00 Gb-RAM running on Windows 10 operating system. In Tables 3.5.1-3.5.2, "No. of Iter." means the number of iterations.

In our computation, we choose $\theta = 3.5$, $\lambda_1 = \lambda = s = 0.65$, $\mu = 0.7$, $\epsilon_n = \frac{1}{(2n+1)^3}$, $\alpha_n = \frac{1}{2n+1}$, $\tau_n = \frac{n+1}{2(n+3)}$, $\rho_n = \frac{10}{(n+1)^2}$, $f(x) = \frac{x}{2}$ and $Sx = \frac{x}{3}$. For Appendix 3.5.3, we choose $\psi_n = \frac{3n+1}{10n+5}$ and for Appendix 3.5.17, we choose $\varphi_n = \frac{n}{2n+7}$, $\chi_n = \frac{n}{2n+5}$, $\phi_n = \frac{n}{10n+2}$.

We begin by presenting some examples in the finite dimensional Hilbert space.

Example 3.5.10. Let C = [-1, 1] and define

$$Ax = \begin{cases} 2x - 1, & x > 1, \\ x^2, & x \in [-1, 1], \\ -2x - 1, & x < -1. \end{cases}$$

Then, A is quasimonotone and uniformly continuous but neither pseudomonotone nor monotone. Moreover, let Tx = x. Then, T is 1-Lipschitz quasi-pseudocontractive. Let $\zeta = 0.3, \eta = 0.4$.

We consider the following cases for the numerical experiments of this example. Case 1: Take $x_1 = 0.3$ and $x_0 = 0.5$.

Case 2: Take $x_1 = 0.29$ and $x_0 = 0.66$.

Case 3: Take $x_1 = 0.28$ and $x_0 = 0.72$.

Case 4: Take $x_1 = 0.30$ and $x_0 = 0.48$.

Table 3.5.1: Numerical results for Example 3.5.10



Figure 3.8: Top left: Case 1; Top right: Case 2; Bottom left: Case 3; Bottom right: Case 4.

Cases		App.	App	App.	App.	App.	Alg.
		3.5.15	3.5.16	3.5.3	3.5.4	3.5.17	3.5.2
Case 1	No. of Iter.	41	41	45	25	66	21
	CPU time	0.0063	0.0128	0.0238	0.0084	0.0091	0.0083
	(sec)						
Case 2	No. of Iter.	42	42	39	24	80	21
	CPU time	0.0106	0.0067	0.0081	0.0057	0.0079	0.0067
	(sec)						
Case 3	No. of Iter.	40	40	45	25	64	21
	CPU time	0.0106	0.0064	0.0087	0.0059	0.0056	0.0062
	(sec)						
Case 4	No. of Iter.	43	43	33	24	84	21
	CPU time	0.0107	0.0059	0.0083	0.0059	0.0063	0.0056
	(sec)						

Example 3.5.11. Let $A : \mathbb{R}^2 \to \mathbb{R}^2$ be defined by

$$A(x_1, x_2) = (-x_1 e^{x_2}, x_2)$$

$$\mathcal{C} = \{ x \in \mathbb{R}^2 : x_1^2 + x_2^2 \le 1, \ 0 < x_1 \}$$

Moreover, we define $Tx = -\frac{5}{3}x$. Then, T is $\frac{5}{3}$ -Lipschitz quasi-pseudocontractive. Let $\zeta = 0.28, \eta = 0.32$. It can be easily verified that all the conditions imposed in Algorithm 3.5.2 are satisfied. We choose $T(x_1, x_2) = \left(-\frac{5}{4}x_1, x_2\right)$. One can easily show that T is $-\frac{5}{4}$ -Lipschitz continuous and quasi-pseudocontractive. We consider the following cases for the numerical experiments of this example.

Case 1: Take $x_1 = (0.1, 0.2)^T$ and $x_0 = (0.2, 0.9)^T$.

Case 2: Take $x_1 = (0.1, 0.2)^T$ and $x_0 = (0.3, 0.8)^T$.

Case 3: Take $x_1 = (0.1, 0.1)^T$ and $x_0 = (0.2, 0.7)^T$.

Case 4: Take $x_1 = (0.1, 0.1)^T$ and $x_0 = (0.3, 0.6)^T$.

Cases		App.	App	App.	App.	App.	Alg.
		3.5.15	3.5.16	3.5.3	3.5.4	3.5.17	3.5.2
Case 1	No. of Iter.	15	20	16	31	89	10
	CPU time	0.0129	0.0071	0.0090	0.0080	0.0099	0.0075
	(sec)						
Case 2	No. of Iter.	14	20	14	31	85	10
	CPU time	0.0111	0.0066	0.0088	0.0077	0.0108	0.0078
	(sec)						
Case 3	No. of Iter.	14	20	19	31	80	10
	CPU time	0.0117	0.0064	0.0087	0.0079	0.0100	0.0078
	(sec)						
Case 4	No. of Iter.	14	19	18	31	75	10
	CPU time	0.0120	0.0062	0.0085	0.0079	0.0097	0.0075
	(sec)						

Table 3.5.2: Numerical results for Example 3.5.11

Now, we present an example in the infinite dimensional Hilbert space.

Example 3.5.12. Let
$$\mathcal{H} = \left\{ x = (x_1, x_2, \cdots, x_i, \cdots) : \sum_{i=1}^{\infty} |x_i|^2 < +\infty \right\}$$
. Let $u, v \in \mathbb{R}$ such that $0 < \frac{v}{2} < u < v$. Let $\mathcal{C}_u = \left\{ x \in \mathcal{H} : ||x|| \le u \right\}$ and $A_v x = \left(v - ||x|| \right) x$. It is known that A is quasimonotone and Lipschitz continuous (see [217]). In this example, we take $u = 4, v = 5$. Furthermore, let $Tx = -2x$. Then, T is 2-Lipschitz quasi-pseudocontractive. Let $\zeta = 0.22, \eta = 0.27$. We consider the following cases for the numerical experiments of this example

Case 1: Take
$$x_1 = \left(\frac{5}{7}, \frac{1}{7}, \frac{1}{35}, \cdots\right)$$
 and $x_0 = \left(\frac{1}{2}, \frac{1}{6}, \frac{1}{18}\right)$.

and



Figure 3.9: Top left: Case 1; Top right: Case 2; Bottom left: Case 3; Bottom right: Case 4.

Case 2: Take
$$x_1 = \left(\frac{4}{7}, \frac{1}{7}, \frac{1}{28}, \cdots\right)$$
 and $x_0 = \left(\frac{1}{3}, \frac{1}{9}, \frac{1}{27}, \cdots\right)$.
Case 3: Take $x_1 = \left(\frac{3}{4}, \frac{1}{4}, \frac{1}{12}, \cdots\right)$ and $x_0 = \left(\frac{2}{5}, \frac{1}{5}, \frac{1}{10}, \cdots\right)$.
Case 4: Take $x_1 = \left(\frac{3}{5}, \frac{1}{5}, \frac{1}{15}, \cdots\right)$ and $x_0 = \left(\frac{2}{3}, \frac{1}{3}, \frac{1}{6}, \cdots\right)$.

 Table 3.5.3: Numerical results for Example 3.5.12

Cases		App.	Alg.
		3.5.4	3.5.2
Case 1	No. of Iter.	23	18
	CPU time	0.0378	0.0162
	(sec)		
Case 2	No. of Iter.	23	18
	CPU time	0.0297	0.0116
	(sec)		
Case 3	No. of Iter.	23	18
	CPU time	0.0243	0.0125
	(sec)		
Case 4	No. of Iter.	23	18
	CPU time	0.0283	0.0123
	(sec)		

Remark 3.5.13. Most of the results in the literature for approximating the solution of quasimonotone (or without monotonicity) VIP in infinite dimensional Hilbert spaces are weak convergence results. Hence, we can only compare our Algorithm 3.5.2 in infinite dimensional Hilbert spaces with Appendix 3.5.4 in Example 3.5.12.

Example 3.5.14. (Application to Image Restoration Problem).

In this last experiment, we consider an application to image restoration problem. We compare the performance of our Algorithm 3.5.2 with Appendix 3.5.15, Appendix 3.5.16 and Appendix 3.5.3.

We recall that the image restoration problem can be formulated as the following linear inverse problem:

$$v = Dx + e \tag{3.5.35}$$

where $x \in \mathbb{R}^N$ is the original image, $D \in \mathbb{R}^{M \times N}$ is the blurring matrix, $v \in \mathbb{R}^M$ is the observed blurred image while e is the Gaussian noise. Solving Problem (3.5.35) is equivalent to solving the convex MPs

$$\min_{x \in \mathbb{R}^N} \left\{ \frac{1}{2} ||Dx - v||_2^2 + \lambda ||x||_1 \right\},$$
(3.5.36)



Figure 3.10: Top left: Case 1; Top right: Case 2; Bottom left: Case 3; Bottom right: Case 4.

where $\lambda > 0$ is the regularization parameter, $\|\cdot\|_2$ denotes the Euclidean norm and $\|\cdot\|_1$ is the ℓ_1 -norm. Our goal is to restore the original image x given the data of the blurred image v. We can express the MPs (3.5.36) as a VIP by taking $A := D^T(Dx - v)$. It is known in this case that the operator A is monotone and $\|D^TD\|$ -Lipschitz continuous (thus A is quasimonotone and uniformly continuous). We consider the 291 × 240 Pout, 256 × 256 Cameraman 275 × 252 Circuit, and 128 × 128 Forest images from MATLAB Image Processing Toolbox. Moreover, we use the Gaussian blur of size 7 × 7 and standard deviation $\sigma = 4$ to create the blurred and noisy image (observed image) and use the algorithms to recover the original image from the blurred image. Also, we measure the quality of the restored image using the signal-to-noise ratio defined by

SNR = 20 × log₁₀
$$\left(\frac{\|x\|_2}{\|x - x^*\|_2}\right)$$
,

where x is the original image and x^* is the restored image. Note that, the larger the SNR, the better the quality of the restored image. We choose the initial values as $x_0 = \mathbf{0} \in \mathbb{R}^N$ and $x_1 = \mathbf{1} \in \mathbb{R}^N$. The results are reported in Table 3.5.4, which shows the SNR values for each algorithm, and Figure 3.11 shows the original, blurred and restored images. The major advantages of our proposed Algorithm 3.5.2 over the algorithms in Appendix 3.5.15, Appendix 3.5.16 and Appendix 3.5.3 compared with are the higher SNR values for generating the recovered images.

 Table 3.5.4: Numerical results for image restoration Problem 3.5.36

		Pout	Cameraman	Circuit	Forest
App.	SNR	5.91	5.82	5.82	5.91
3.5.15					
App.	SNR	9.78	12.40	13.36	9.19
3.5.16					
App. 3.5.3	SNR	9.78	9.52	9.55	9.79
Alg. 3.5.2	SNR	18.93	18.63	16.07	23.79

Appendix 3.5.15. Algorithm 1 of Salahuddin [217]

Initialization: Let $x_0 \in \mathcal{C}$ and $\lambda \in [a, b]$, where $0 < a \le b < \frac{1}{L}$ be arbitrary.

Iterative Steps: Calculate x_{n+1} as follows:

Step 0. Set n = 0.

Step 1. If

$$x_n = P_{\mathcal{C}}(x_n - \lambda A x_n),$$

then stop and x_n is the solution of VIP (1.2.1). Otherwise, set

Step 2:

$$y_n = P_{\mathcal{C}}(x_n - \lambda A x_n),$$

and

$$x_{n+1} = P_{\mathcal{C}}(x_n - \lambda A y_n),$$

Set n := n + 1 and return to Step 1.



Figure 3.11: Comparison for numerical results for image restoration Problem (3.5.36) : Top: Pout; Next: Cameraman; Next:Circuit; Bottom: Forest.

Step 0: Take $\lambda_0 > 0$, $x_0 \in \mathcal{H}$, $0 < \mu < 1$. Choose a nonnegative real sequence $\{\rho_n\}$ such that $\sum_{n=0}^{\infty} \rho_n < +\infty$.

Step 1: Given the current iterate x_n , compute

$$y_n = P_{\mathcal{C}}(x_n - \lambda_n A x_n).$$

If $x_n = y_n$, (or $Ay_n = 0$), then stop and y_n is a solution of the VIP. Otherwise, go to Step 2.

Step 2: Compute

$$x_{n+1} = y_n - \lambda_n (Ay_n - Ax_n),$$

where

$$\lambda_{n+1} = \begin{cases} \min\left\{\frac{\mu \|x_n - y_n\|}{\|Ax_n - Ay_n\|}, \lambda_n + \rho_n\right\} & \text{if } Ax_n - Ay_n \neq 0, \\ \lambda_n + \rho_n, & \text{otherwise.} \end{cases}$$
(3.5.37)

Set n := n + 1 and go back to Step 1.

Algorithm 3.5.3. Algorithm 2 of Ogwo et al. [193]

Step 0: Choose sequences $\{\psi_n\}$ and $\{\tau_n\}$ such that $\psi_n \in [0, 1)$ and $\tau_n \in (0, 1]$ for all $n \ge 1$. Let $\lambda_1 > 0$, $\mu \in (0, 1)$ and $x_0, x_1 \in \mathcal{H}$ be given arbitrarily. Choose a nonnegative real sequence $\{\rho_n\}$ such that $\sum_{n=1}^{\infty} \rho_n < \infty$. Set n := 1. **Step 1**: Given the current iterates x_{n-1} and x_n $(n \ge 1)$, compute

$$w_n = x_n + \psi_n (x_n - x_{n-1})$$

and

$$y_n = P_C(w_n - \lambda_n A w_n).$$

If $w_n = y_n$: STOP. Otherwise, go to Step 2. Step 2: Construct the half-space

$$T_n = \{ x \in \mathcal{H} : \langle w_n - \lambda_n A w_n - y_n, x - y_n \rangle \le 0 \}.$$

Then, compute

$$z_n = P_{T_n}(w_n - \lambda_n A y_n)$$

and

$$x_{n+1} = (1 - \tau_n)w_n + \tau_n z_n,$$

where

$$\lambda_{n+1} = \begin{cases} \min\left\{\frac{\mu\left(\|w_n - y_n\|^2 + \|z_n - y_n\|^2\right)}{2\langle Aw_n - Ay_n, z_n - y_n\rangle}, \lambda_n + \rho_n\right\}, & \text{if } \langle Aw_n - Ay_n, z_n - y_n\rangle > 0, \\ \lambda_n + \rho_n, & \text{otherwise.} \end{cases}$$
(3.5.38)

Set n := n + 1 and return to **Step 1**.

Algorithm 3.5.4. Algorithm 1 of Alakoya *et al.* [12]

Step 1: Select initial point $x_0, x_1 \in \mathcal{H}_1$. Given the iterates x_{n-1} and x_n for each $n \geq 1$, choose ϑ_n such that $0 \leq \vartheta_n \leq \overline{\vartheta}_n$, where

$$\bar{\vartheta}_n := \begin{cases} \min\left\{\vartheta, \frac{\epsilon_n}{\|x_n - x_{n-1}\|}\right\}, & \text{if } x_n \neq x_{n-1}\\ \vartheta, & \text{otherwise.} \end{cases}$$
(3.5.39)

Step 2: Compute

$$w_n = x_n + \vartheta_n (x_n - x_{n-1}).$$

Step 3: Compute

$$y_n = P_{\mathcal{C}}(w_n - \lambda_n A w_n).$$

If $w_n = y_n$ (or $Ay_n = 0$) then stop: w_n is a solution of the Problem (1.2.3). Otherwise, go to **Step 4.**

Step 4: Compute

$$z_n = y_n + \lambda_n (Ay_n - Aw_n).$$

Step 5 Compute

$$x_{n+1} = \alpha_n f(x_n) + (1 - \alpha_n) z_n$$

where

$$\lambda_{n+1} = \begin{cases} \min\left\{\frac{\mu \|w_n - y_n\|}{\|Aw_n - Ay_n\|}, \lambda_n + \rho_n\right\} & \text{if } Aw_n - Ay_n \neq 0, \\ \lambda_n + \rho_n, & \text{otherwise.} \end{cases}$$
(3.5.40)

 mi

Set n := n + 1 and go back to **Step 1**.

Appendix 3.5.17. Algorithm 1 of Yin et. al [270] Step 0: Take $s_0 > 0$, $x_0 \in \mathcal{H}$, $0 < \mu < 1$. Step 1: Let the nth iterate x_n be given. Compute

$$\bar{v_n} = (1 - \chi_n)x_n + \chi_n S x_n$$

and

$$v_n = (1 - \varphi_n) x_n + \varphi_n S \bar{v_n}$$

Step 2: Let the nth step size s_n be known. Compute

$$y_n = P_{\mathcal{C}}(v_n - s_n A v_n)$$

and

$$x_{n+1} = (1 - \phi_n)v_n + \phi_n \Big[y_n + s_n (Av_n - Ay_n) \Big]$$

where

$$s_{n+1} = \begin{cases} \min\{\frac{\mu \|v_n - y_n\|}{\|Av_n - Ay_n\|}, s_n\} & \text{if } Av_n - Ay_n \neq 0, \\ s_n, & \text{otherwise.} \end{cases}$$
(3.5.41)

Set n := n + 1 and go back to Step 1.

CHAPTER 4

Results on Split Variational Inequality Problems

4.1 Introduction

In this chapter, we propose and study iterative algorithms for approximating solutions of SVIPs and GVIP in the framework of real Hilbert spaces. We present numerical examples of our proposed methods in comparison to other methods in literature to illustrate the applicability of our proposed methods.

4.2 On split variational inequality problems without product space formulation

Many methods have been proposed in the literature for solving the SVIP. Most of these methods either require that this problem is transformed into an equivalent VIP in a product space, or that the underlying operators are co-coercive. However, it has been discovered that such product space transformation may cause some potential difficulties during implementation and its approach may not fully exploit the attractive splitting nature of the SVIP. On the other hand, the co-coercive assumption of the underlying operators would preclude the potential applications of these methods. To avoid these setbacks, we propose two new relaxed inertial methods for solving the SVIP without any product space transformation, and for which the underlying operators are freed from the restrictive co-coercive assumption. The methods proposed, involve projections onto half-spaces only, and originate from an explicit discretization of a dynamical system, which combines both the inertial and relaxation techniques in order to achieve high convergence speed. Moreover, the sequence generated by these methods is shown to converge strongly to a minimum-norm solution of the problem in real Hilbert spaces. Furthermore, numerical implementations and comparisons are given to support our theoretical findings.

4.2.1 Proposed methods

In this section, we present our proposed methods and discuss their features. We begin with the following assumptions under which our strong convergent results are obtained.

Assumption 4.2.1. Suppose that the following conditions hold:

- (a) The feasible sets C and Q are nonempty closed and convex subsets of the real Hilbert spaces \mathcal{H}_1 and \mathcal{H}_2 , respectively.
- (b) The sets C and Q are defined as

$$\mathcal{C} = \{x \in \mathcal{H}_1 \mid c(x) \le 0\} \text{ and } Q = \{w \in \mathcal{H}_2 \mid q(w) \le 0\},\$$

where $c : \mathcal{H}_1 \to \mathbb{R}$ and $q : \mathcal{H}_2 \to \mathbb{R}$ are continuously differentiable convex functions such that $c'(\cdot)$ and $q'(\cdot)$ are Lipschitz continuous with constants K_1 and K_2 , respectively.

- (c) $A : \mathcal{H}_1 \to \mathcal{H}_1$ and $F : \mathcal{H}_2 \to \mathcal{H}_2$ are monotone and Lipschitz continuous with Lipschitz constants L_1 and L_2 , respectively.
- (d) $T : \mathcal{H}_1 \to \mathcal{H}_2$ is a bounded linear operator and the solution set $\Gamma := \{z \in VI(A, \mathcal{C}) : Tz \in VI(F, \mathcal{Q})\}$ of the SVIP (1.2.4)-(1.2.5) is nonempty, where $VI(A, \mathcal{C})$ is the solution set of the classical VIP (1.2.4).
- (e) $\{\delta_n\}_{n=1}^{\infty}, \{\theta_n\}_{n=1}^{\infty} and \{\tau_n\}_{n=1}^{\infty}$ are positive sequences satisfying the following conditions: $\delta_n \in (0,1), \quad \lim_{n \to \infty} \delta_n = 0, \quad \sum_{n=1}^{\infty} \delta_n = \infty, \quad \lim_{n \to \infty} \frac{\tau_n}{\delta_n} = 0 \text{ and } \{\theta_n\} \subset (a, 1 - \delta_n) \text{ for some } a > 0.$
- (f) $\{\phi_n\}_{n=1}^{\infty}$ is a sequence of positive numbers such that $\phi_n \in (0,1]$ and $\lim_{n \to \infty} \phi_n = \psi \in (0,1]$.

We present the following method for solving the SVIP (1.2.4)-(1.2.5) when L_1 , L_2 , K_1 and K_2 are known.

Algorithm 4.2.2. Relaxed inertial method with fixed stepsizes.

Step 0: Choose sequences $\{\delta_n\}_{n=1}^{\infty}, \{\theta_n\}_{n=1}^{\infty}$ and $\{\tau_n\}_{n=1}^{\infty}$ such that the conditions from Assumption 4.2.1 (e) hold and let $\eta \geq 0, \phi \in (0, 1], \alpha \geq 3$ and $x_0, x_1 \in \mathcal{H}_1$ be given arbitrarily. Set n := 1.

Step 1: Given the iterates x_{n-1} and x_n $(n \ge 1)$, choose α_n such that $0 \le \alpha_n \le \overline{\alpha}_n$, where

$$\bar{\alpha}_n := \begin{cases} \min\left\{\frac{n-1}{n+\alpha-1}, \frac{\tau_n}{\|x_n - x_{n-1}\|}\right\}, & \text{if } x_n \neq x_{n-1}, \\ \frac{n-1}{n+\alpha-1}, & \text{otherwise.} \end{cases}$$
(4.2.1)

Step 2: Compute

$$w_n = x_n + \alpha_n (x_n - x_{n-1})$$

and construct the half-space

$$\mathcal{Q}_n = \{ w \in \mathcal{H}_2 \mid q(Tw_n) + \langle q'(Tw_n), w - Tw_n \rangle \le 0 \}.$$

Then compute

$$y_n = P_{\mathcal{Q}_n}(Tw_n - \lambda FTw_n),$$

where

$$0 < \lambda < \frac{\sqrt{\rho_2^2 K_2^2 + 2\rho_2 K_2 (1-\phi) L_2 + L_2^2} - [\rho_2 K_2 + (1-\phi) L_2]}{\phi L_2^2}, \text{ for some } \rho_2 > 0.$$
(4.2.2)

Step 3: Compute

$$z_n = (1 - \phi)Tw_n + \phi \Big(y_n + \lambda (FTw_n - Fy_n) \Big)$$

and

$$b_n = w_n + \eta_n T^* (z_n - T w_n),$$

where the stepsize η_n is chosen such that for small enough $\epsilon > 0$,

$$\eta_n \in \left[\epsilon, \quad \frac{\|Tw_n - z_n\|^2}{\|T^*(Tw_n - z_n)\|^2} - \epsilon\right],$$

if $z_n \neq Tw_n$; otherwise, $\eta_n = \eta$. Step 4: Construct the half space

$$\mathcal{C}_n = \{ x \in \mathcal{H}_1 \mid c(b_n) + \langle c'(b_n), x - b_n \rangle \le 0 \}$$

Then compute

$$u_n = P_{\mathcal{C}_n}(b_n - \mu A b_n),$$

where

$$0 < \mu < \frac{\sqrt{\rho_1^2 K_1^2 + 2\rho_1 K_1 (1 - \phi) L_1 + L_1^2} - [\rho_1 K_1 + (1 - \phi) L_1]}{\phi L_1^2}, \text{ for some } \rho_1 > 0.$$

$$(4.2.3)$$

Step 5: Compute

$$x_{n+1} = (1 - \theta_n - \delta_n)b_n + \theta_n t_n,$$

where

$$t_n = (1 - \phi)b_n + \phi \Big(u_n + \mu (Ab_n - Au_n)\Big).$$

Set n := n + 1 and go back to Step 1.

In the case where the Lipschitz constants L_1 , L_2 , K_1 and K_2 are not known, we present the following method with adaptive stepsizes for solving the SVIP (1.2.4)-(1.2.5).

Algorithm 4.2.3. Relaxed inertial method with adaptive stepsize strategy.

Step 0: Choose sequences $\{\delta_n\}_{n=1}^{\infty}, \{\theta_n\}_{n=1}^{\infty}, \{\phi_n\}_{n=1}^{\infty} \text{ and } \{\tau_n\}_{n=1}^{\infty} \text{ such that the conditions from Assumption 4.2.1 (e)-(f) hold and let } \eta \geq 0, \ 0 < a_i < \sqrt{(1-\psi)(2-\psi) + (1-\rho_i^2)} - (1-\psi) \text{ for some } \rho_i > 0, \ i = 1, 2, \ \lambda_1 > 0, \ \mu_1 > 0, \ \alpha \geq 3 \text{ and } x_0, x_1 \in \mathcal{H}_1 \text{ be given arbitrarily. Set } n := 1.$

Step 1: Given the iterates x_{n-1} and x_n for each $n \ge 1$, choose α_n such that $0 \le \alpha_n \le \overline{\alpha}_n$, where

$$\bar{\alpha}_n := \begin{cases} \min\left\{\frac{n-1}{n+\alpha-1}, \frac{\tau_n}{\|x_n - x_{n-1}\|}\right\}, & \text{if } x_n \neq x_{n-1}, \\ \frac{n-1}{n+\alpha-1}, & \text{otherwise.} \end{cases}$$
(4.2.4)

Step 2: Compute

$$w_n = x_n + \alpha_n (x_n - x_{n-1}),$$

and construct the half-space

$$\mathcal{Q}_n = \{ w \in \mathcal{H}_2 \mid q(Tw_n) + \langle q'(Tw_n), x - Tw_n \rangle \le 0 \}.$$

Then compute

$$y_n = P_{\mathcal{Q}_n}(Tw_n - \lambda_n FTw_n),$$

where

$$\lambda_{n+1} = \begin{cases} \min\left\{\frac{a_2||Tw_n - y_n||}{\sqrt{||FTw_n - Fy_n||^2 + ||q'(Tw_n) - q'(y_n)||^2}}, \lambda_n\right\},\\ \text{if } ||FTw_n - Fy_n||^2 + ||q'(Tw_n) - q'(y_n)||^2 \neq 0,\\ \lambda_n, \quad \text{otherwise.} \end{cases}$$
(4.2.5)

Step 3: Compute

$$z_n = (1 - \phi_n)Tw_n + \phi_n \Big(y_n + \lambda_n (FTw_n - Fy_n) \Big)$$

and

$$b_n = w_n + \eta_n T^*(z_n - Tw_n),$$

where the stepsize η_n is chosen such that for small enough $\epsilon > 0$,

$$\eta_n \in \left[\epsilon, \quad \frac{\|Tw_n - z_n\|^2}{\|T^*(Tw_n - z_n)\|^2} - \epsilon\right],$$

if $z_n \neq Tw_n$; otherwise, $\eta_n = \eta$. Step 4: Construct the half-space

$$\mathcal{C}_n = \{ x \in \mathcal{H}_1 \mid c(b_n) + \langle c'(b_n), x - b_n \rangle \le 0 \}.$$

Then compute

$$u_n = P_{\mathcal{C}_n}(b_n - \mu_n A b_n),$$
where

$$\mu_{n+1} = \begin{cases} \min\left\{\frac{a_1||b_n - u_n||}{\sqrt{||Au_n - Ab_n||^2 + ||c'(u_n) - c'(b_n)||^2}}, \ \mu_n\right\},\\ \text{if } ||Au_n - Ab_n||^2 + ||c'(u_n) - c'(b_n)||^2 \neq 0,\\ \mu_n, & \text{otherwise.} \end{cases}$$
(4.2.6)

Step 5: Compute

$$x_{n+1} = (1 - \theta_n - \delta_n)b_n + \theta_n t_n$$

where

$$t_n = (1 - \phi_n)b_n + \phi_n \Big(u_n + \mu_n (Ab_n - Au_n)\Big).$$

Set n := n + 1 and go back to **Step 1**.

Remark 4.2.1.

- Observe that Algorithms 4.2.2 and 4.2.3 can be viewed as modified RITFBF (2.5.14), involving only one projection onto the half-space C_n per iteration for solving the classical VIP in \mathcal{H}_1 and another RITFBF involving one projection onto the half-space Q_n per iteration under a bounded linear operator T for solving another VIP in another space \mathcal{H}_2 , without further projections onto feasible sets, unlike other existing methods for solving SVIPs, where projections onto feasible sets are required; for instance, the method in [106, 221] for the case of classical VIP. A notable advantage of these methods (Algorithms 4.2.2 and 4.2.3) is that the co-coercive assumption on the operators A and F usually used in many works (see for example, [131, 142, 143, 146, 190]) to guarantee convergence is dispensed with, and no product space formulation of any sort is required under this setting, unlike in [63, 121].
- The stepsizes $\{\lambda_n\}$ and $\{\mu_n\}$ given by (4.2.5) and (4.2.6), respectively are generated at each iteration by some simple computations. Thus, Algorithm 4.2.3 is easily implemented without the prior knowledge of the Lipschitz constants L_1 , L_2 , K_1 and K_2 unlike in [106, Algorithm 4.1] where the knowledge of the Lipschitz constant is required and [221, Algorithm 3.3] where the stepsize is assumed to be 1.
- Step 1 of our methods is also easily implemented since the value of $||x_n x_{n-1}||$ is a prior known before choosing α_n .
- Step 5 of both algorithms guarantee the strong convergence to the minimum-norm solution of the SVIP .
- It is easy to see that $\mathcal{C}_n \supset \mathcal{C}$ and $\mathcal{Q}_n \supset \mathcal{Q}$. Furthermore, since \mathcal{C}_n and \mathcal{Q}_n are half-spaces, the projections onto them have explicit formulas. In fact, u_n and y_n in our methods can be calculated as:

$$u_{n} = \begin{cases} b_{n} - \mu A b_{n}, & \text{if } c(b_{n}) - \mu \langle c'(b_{n}), A b_{n} \rangle \leq 0, \\ b_{n} - \mu A b_{n} - \frac{c(b_{n}) - \mu \langle c'(b_{n}), A b_{n} \rangle}{||c'(b_{n})||^{2}} c'(b_{n}), & \text{otherwise} \end{cases}$$
(4.2.7)

and

$$y_n = \begin{cases} Tw_n - \lambda FTw_n, & \text{if } q(Tw_n) - \lambda \langle q'(Tw_n), FTw_n \rangle \leq 0, \\ Tw_n - \lambda FTw_n - \frac{q(Tw_n) - \lambda \langle q'(Tw_n), FTw_n \rangle}{||q'(Tw_n)||^2} q'(Tw_n), & \text{otherwise.} \end{cases}$$
(4.2.8)

Hence, our methods are computationally less expensive than other existing methods for solving the SVIP (1.2.4)-(1.2.5). This benefit (brought by the half-spaces) is further verified in our numerical analysis.

Remark 4.2.2. The choice of the stepsize η_n in Step 3 of Algorithms 4.2.2 and 4.2.3 do not require the prior knowledge of the operator norm ||T||, unlike in [63, 121, 183, 244]. Note that algorithms whose implementation depends on the operator norm require the computation of the norm of the bounded linear operator, which in general is a very difficult task to accomplish as shown in [124, Theorem 2.3].

Also, we note that the value of the constant η in our methods does not influence the algorithms, but it was introduced for the sake of clarity. In what follows, we show that η_n is well-defined, and then, present an alternative choice for η_n without involving the constant η .

Proposition 4.2.4. The stepsize η_n given in Step 3 of Algorithm 4.2.2 (or Algorithm 4.2.3) is well-defined.

Proof. Let $z \in \Gamma$. Then, by Lemma 2.1.1, we obtain

$$\|T^{*}(Tw_{n} - z_{n})\| \|w_{n} - z\| \geq \langle T^{*}(Tw_{n} - z_{n}), w_{n} - z \rangle$$

= $\langle Tw_{n} - z_{n}, Tw_{n} - Tz \rangle$
= $\frac{1}{2} \Big[\|Tw_{n} - z_{n}\|^{2} + \|Tw_{n} - Tz\|^{2} - \|z_{n} - Tz\|^{2} \Big].$ (4.2.9)

Furthermore, we claim that

$$||z_n - Tz||^2 \le ||Tw_n - Tz||^2 - \phi \left[2 - \phi - \phi \lambda^2 L_2^2 - 2\lambda(1 - \phi)L_2 - 2\rho_2 \lambda K_2\right] ||Tw_n - y_n||^2.$$
(4.2.10)

We shall prove this claim (inequality (4.2.10)) later in Lemma 4.2.3 without involving the stepsize η_n .

Now, by the condition on λ , that is, from (4.2.2), we obtain

$$\left(\phi L_2^2 \lambda + \rho_2 K_2 + (1 - \phi) L_2 \right)^2 < \rho_2^2 K_2^2 + 2\rho_2 K_2 (1 - \phi) L_2 + L_2^2$$

= $\rho_2^2 K_2^2 + 2\rho_2 K_2 (1 - \phi) L_2 + (1 - \phi)^2 L_2^2 + \phi L_2^2 (2 - \phi).$

By some simple simplifications, we get

$$\lambda^2 \phi^2 L_2^4 + 2\lambda \phi L_2^2 \rho_2 K_2 + 2\lambda \phi L_2^2 (1 - \phi_2) L_2 - \phi L_2^2 (2 - \phi) < 0.$$

Since $L_2, \phi > 0$, dividing through by $L_2^2 \phi$, we obtain

$$\lambda^2 \phi L_2^2 + 2\lambda \rho_2 K_2 + 2\lambda (1-\phi) L_2 - (2-\phi) < 0,$$

which implies that

$$2 - \phi - \lambda^2 \phi L_2^2 - 2\lambda \rho_2 K_2 - 2\lambda (1 - \phi) L_2 > 0.$$
(4.2.11)

Hence, we obtain from (4.2.10) that

$$||z_n - Tz||^2 \le ||Tw_n - Tz||^2.$$
(4.2.12)

Note that if λ is replaced with λ_n as in Algorithm 4.2.3, we will use inequality (4.2.65) (instead of (4.2.10)) and relation (4.2.67) (instead of (4.2.11)) to get (4.2.12). Therefore, substituting (4.2.12) into (4.2.9), we obtain that

$$||T^*(Tw_n - z_n)|| ||w_n - z|| \ge \frac{1}{2} ||Tw_n - z_n||^2.$$
(4.2.13)

Now, for $z_n \neq Tw_n$, we have that $||Tw_n - z_n|| > 0$. This together with (4.2.13), imply that $||T^*(Tw_n - z_n)|| ||w_n - z|| > 0$. Hence, we have that $||T^*(Tw_n - z_n)|| \neq 0$. Therefore, η_n is well-defined.

We now give an alternative way of choosing the stepsize η_n in our methods. First, observe from Step 3 of Algorithms 4.2.2 and 4.2.3, that

$$\eta_n \|T^*(Tw_n - z_n)\|^2 \le \|Tw_n - z_n\|^2 - \epsilon \|T^*(Tw_n - z_n)\|^2,$$
(4.2.14)

which implies that

$$\eta_n^2 \|T^*(Tw_n - z_n)\|^2 - \eta_n \|Tw_n - z_n\|^2 \le -\epsilon \eta_n \|T^*(Tw_n - z_n)\|^2.$$
(4.2.15)

Thus, η_n can be chosen such that for small enough $\epsilon > 0$,

$$\eta_n \in \left[\epsilon, \quad \frac{\|Tw_n - z_n\|^2}{1 + \|T^*(Tw_n - z_n)\|^2} - \epsilon\right].$$
(4.2.16)

Then, we can see that

$$\eta_n \le \frac{\|Tw_n - z_n\|^2 - \epsilon \left(1 + \|T^*(Tw_n - z_n)\|^2\right)}{1 + \|T^*(Tw_n - z_n)\|^2},$$

which implies that

$$\eta_n + \eta_n \|T^*(Tw_n - z_n)\|^2 \le \|Tw_n - z_n\|^2 - \epsilon - \epsilon \|T^*(Tw_n - z_n)\|^2.$$

This further gives (4.2.14) and consequently (4.2.15).

As we shall see in our convergence analysis, (4.2.14) and (4.2.15) play crucial role in our proofs. Therefore, one can either choose the stepsize in Step 3 or that in (4.2.16), to ensure the convergence of Algorithms 4.2.2 and 4.2.3.

4.2.2 Convergence analysis

Lemma 4.2.3. Let $\{x_n\}$ be a sequence generated by Algorithm 4.2.2 under Assumption 4.2.1. Then

$$||z_n - Tp||^2 \le ||Tw_n - Tp||^2 - \phi \left[2 - \phi - \phi \lambda^2 L_2^2 - 2\lambda(1 - \phi)L_2 - 2\rho_2 \lambda K_2\right] ||Tw_n - y_n||^2$$
(4.2.17)

and

$$||t_n - p||^2 \le ||b_n - p||^2 - \phi \left[2 - \phi - \phi \mu^2 L_1^2 - 2\mu(1 - \phi)L_1 - 2\rho_1 \mu K_1\right] ||b_n - u_n||^2,$$
(4.2.18)

for all $p \in \Gamma$, where ρ_1, ρ_2 are positive constants.

Proof. Let $p \in \Gamma$. Then $Tp \in VI(F, \mathcal{Q}) \subset \mathcal{Q} \subset \mathcal{Q}_n$. Since $y_n = P_{\mathcal{Q}_n}(Tw_n - \lambda FTw_n)$ and $Tp \in \mathcal{Q}_n$, we obtain from the characteristic property of $P_{\mathcal{Q}_n}$, that

$$\langle Tw_n - \lambda FTw_n - y_n, y_n - Tp \rangle \ge 0$$

or equivalently

$$2\langle Tw_n - y_n, y_n - Tp \rangle - 2\lambda \langle FTw_n - Fy_n, y_n - Tp \rangle - 2\lambda \langle Fy_n, y_n - Tp \rangle \ge 0.$$
 (4.2.19)

Also, from Lemma 2.1.1, we obtain that

$$2\langle Tw_n - y_n, y_n - Tp \rangle = \|Tw_n - Tp\|^2 - \|Tw_n - y_n\|^2 - \|y_n - Tp\|^2.$$
(4.2.20)

Again, since F is monotone, we obtain that

$$\langle Fy_n, y_n - Tp \rangle = \langle Fy_n - FTp, y_n - Tp \rangle + \langle FTp, y_n - Tp \rangle \geq \langle FTp, y_n - Tp \rangle.$$
 (4.2.21)

Substituting (4.2.20) and (4.2.21) into (4.2.19), we obtain

$$||y_n - Tp||^2 \le ||Tw_n - Tp||^2 - ||Tw_n - y_n||^2 - 2\lambda \langle FTw_n - Fy_n, y_n - Tp \rangle \qquad (4.2.22) + 2\lambda \langle FTp, Tp - y_n \rangle.$$

Also, since $z_n = (1 - \phi)Tw_n + \phi \Big(y_n + \lambda (FTw_n - Fy_n) \Big)$, we get

$$\begin{split} \|z_{n} - Tp\|^{2} &= \|(1 - \phi)(Tw_{n} - Tp) + \phi(y_{n} - Tp) + \phi\lambda(FTw_{n} - Fy_{n})\|^{2} \\ &= (1 - \phi)^{2}\|Tw_{n} - Tp\|^{2} + \phi^{2}\|y_{n} - Tp\|^{2} + \phi^{2}\lambda^{2}\|FTw_{n} - Fy_{n}\|^{2} \\ &+ 2\phi(1 - \phi)\langle Tw_{n} - Tp, y_{n} - Tp\rangle + 2\lambda\phi(1 - \phi)\langle Tw_{n} - Tp, FTw_{n} - Fy_{n}\rangle \\ &+ 2\lambda\phi^{2}\langle y_{n} - Tp, FTw_{n} - Fy_{n}\rangle \\ &= (1 - \phi)^{2}\|Tw_{n} - Tp\|^{2} + \phi^{2}\|y_{n} - Tp\|^{2} + \phi^{2}\lambda^{2}\|FTw_{n} - Fy_{n}\|^{2} \\ &+ \phi(1 - \phi)\left[\|Tw_{n} - Tp\|^{2} + \|y_{n} - Tp\|^{2} - \|Tw_{n} - y_{n}\|^{2}\right] \\ &+ 2\lambda\phi(1 - \phi)\langle Tw_{n} - Tp, FTw_{n} - Fy_{n}\rangle + 2\lambda\phi^{2}\langle y_{n} - Tp, FTw_{n} - Fy_{n}\rangle \\ &= (1 - \phi)\|Tw_{n} - Tp\|^{2} + \phi\|y_{n} - Tp\|^{2} - \phi(1 - \phi)\|Tw_{n} - y_{n}\|^{2} \\ &+ (1 - \phi)\|Tw_{n} - Tp\|^{2} + \phi\|y_{n} - Tp\|^{2} - \phi(1 - \phi)\|Tw_{n} - y_{n}\|^{2} \\ &+ (1 - \phi)\langle Tw_{n} - Tp, FTw_{n} - Fy_{n}\rangle + 2\lambda\phi^{2}\langle y_{n} - Tp, FTw_{n} - Fy_{n}\rangle. \end{split}$$

Substituting (4.2.22) into (4.2.23), we obtain

$$\begin{aligned} \|z_{n} - Tp\|^{2} &\leq (1 - \phi) \|Tw_{n} - Tp\|^{2} \\ &+ \phi \left[\|Tw_{n} - Tp\|^{2} - \|Tw_{n} - y_{n}\|^{2} - 2\lambda \langle FTw_{n} - Fy_{n}, y_{n} - Tp \rangle \right] \\ &+ \phi 2\lambda \langle FTp, Tp - y_{n} \rangle - \phi (1 - \phi) \|Tw_{n} - y_{n}\|^{2} + \phi^{2}\lambda^{2} \|FTw_{n} - Fy_{n}\|^{2} \\ &+ 2\lambda \phi (1 - \phi) \langle Tw_{n} - Tp, FTw_{n} - Fy_{n} \rangle + 2\lambda \phi^{2} \langle y_{n} - Tp, FTw_{n} - Fy_{n} \rangle \\ &= \|Tw_{n} - Tp\|^{2} - \phi (2 - \phi) \|Tw_{n} - y_{n}\|^{2} + \phi^{2}\lambda^{2} \|FTw_{n} - Fy_{n}\|^{2} \\ &+ 2\lambda \phi (1 - \phi) \langle Tw_{n} - y_{n}, FTw_{n} - Fy_{n} \rangle + 2\lambda \phi \langle FTp, Tp - y_{n} \rangle \\ &\leq \|Tw_{n} - Tp\|^{2} - \phi (2 - \phi) \|Tw_{n} - y_{n}\|^{2} + \phi^{2}\lambda^{2}L_{2}^{2} \|Tw_{n} - y_{n}\|^{2} \\ &+ 2\lambda \phi (1 - \phi)L_{2} \|Tw_{n} - y_{n}\|^{2} + 2\lambda \phi \langle FTp, Tp - y_{n} \rangle \\ &= \|Tw_{n} - Tp\|^{2} - \phi \left[2 - \phi - \phi\lambda^{2}L_{2}^{2} - 2\lambda (1 - \phi)L_{2}\right] \|Tw_{n} - y_{n}\|^{2} \\ &+ 2\lambda \phi \langle FTp, Tp - y_{n} \rangle. \end{aligned}$$

Now, if FTp = 0, then we obtain the desired conclusion from (4.2.25). However, if $FTp \neq 0$, then we obtain from Lemma 2.5.10 that $Tp \in \partial Q$ and there exists $\rho_2 > 0$ such that $FTp = -\rho_2 q'(Tp)$. Since $Tp \in \partial Q$, q(Tp) = 0. Also, since q is a differentiable convex function, it follows from (2.4.7) that

$$q(y_n) \ge q(Tp) + \langle q'(Tp), y_n - Tp \rangle$$

= $\frac{-1}{\rho_2} \langle FTp, y_n - Tp \rangle$,

which implies that

$$\langle Tp - y_n, FTp \rangle \le \rho_2 q(y_n).$$
 (4.2.26)

Since $y_n \in \mathcal{Q}_n$, we have

$$q(Tw_n) + \langle q'(Tw_n), y_n - Tw_n \rangle \le 0.$$
 (4.2.27)

Again, since q is a differentiable convex function, we obtain from (2.4.7) that

$$\langle q'(y_n), Tw_n - y_n \rangle + q(y_n) \le q(Tw_n).$$
 (4.2.28)

From (4.2.27), (4.2.28) and the Lipschitz continuity of $q'(\cdot)$, we obtain

$$q(y_n) \le \langle q'(y_n) - q'(Tw_n), y_n - Tw_n \rangle \le K_2 \|y_n - Tw_n\|^2.$$
(4.2.29)

By substituting (4.2.26) and (4.2.29) into (4.2.25), we obtain the desired conclusion (4.2.17). Following the same line of argument, we obtain (4.2.18).

Lemma 4.2.4. Let $\{x_n\}$ be a sequence generated by Algorithm 4.2.2 under Assumption 4.2.1. Then, $\{x_n\}$ is bounded.

Proof. Let $p \in \Gamma$. Since $w_n = x_n + \alpha_n(x_n - x_{n-1})$, we have

$$||w_n - p|| \le ||x_n - p|| + \alpha_n ||x_n - x_{n-1}||.$$

Also, from Step 1, we observe that $\alpha_n ||x_n - x_{n-1}|| \le \tau_n$, $\forall n \ge 1$, which implies that

$$\frac{\alpha_n}{\delta_n} \|x_n - x_{n-1}\| \le \frac{\tau_n}{\delta_n} \to 0, \quad \text{as} \quad n \to \infty.$$
(4.2.30)

Hence, there exists $M_1 > 0$ such that

$$\frac{\alpha_n}{\delta_n} \|x_n - x_{n-1}\| \le M_1, \quad \forall n \ge 1.$$

$$(4.2.31)$$

This implies that

$$||w_n - p|| \le ||x_n - p|| + \delta_n M_1, \ \forall n \ge 1.$$
(4.2.32)

Now, by Lemma 4.2.3 (inequality (4.2.17)) and (4.2.11), we obtain that

$$||z_n - Tp||^2 \le ||Tw_n - Tp||^2.$$
(4.2.33)

From Step 3, Lemma 2.1.1 and (4.2.33), we obtain

$$\begin{aligned} \|b_{n} - p\|^{2} &= \|w_{n} - p\|^{2} + \eta_{n}^{2} \|T^{*}(z_{n} - Tw_{n})\|^{2} + 2\eta_{n} \langle w_{n} - p, T^{*}(z_{n} - Tw_{n}) \rangle \\ &= \|w_{n} - p\|^{2} + \eta_{n}^{2} \|T^{*}(z_{n} - Tw_{n})\|^{2} + 2\eta_{n} \langle Tw_{n} - Tp, z_{n} - Tw_{n} \rangle \\ &= \|w_{n} - p\|^{2} + \eta_{n}^{2} \|T^{*}(z_{n} - Tw_{n})\|^{2} \\ &+ \eta_{n} \left[\|z_{n} - Tp\|^{2} - \|Tw_{n} - Tp\|^{2} - \|z_{n} - Tw_{n}\|^{2} \right] \\ &\leq \|w_{n} - p\|^{2} + \eta_{n}^{2} \|T^{*}(z_{n} - Tw_{n})\|^{2} - \eta_{n} \|z_{n} - Tw_{n}\|^{2} \\ &= \|w_{n} - p\|^{2} - \eta_{n} \left[\|z_{n} - Tw_{n}\|^{2} - \eta_{n} \|T^{*}(z_{n} - Tw_{n})\|^{2} \right]. \end{aligned}$$

$$(4.2.34)$$

Thus, by the condition on η_n , that is, from (4.2.14), we obtain that

$$||b_n - p||^2 \le ||w_n - p||^2,$$

which implies from (4.2.32) that

$$||b_n - p|| \le ||w_n - p|| \le ||x_n - p|| + \delta_n M_1 \ \forall n \ge 1.$$
(4.2.35)

From Step 5, we have

$$||x_{n+1} - p|| = ||(1 - \theta_n - \delta_n)(b_n - p) + \theta_n(t_n - p) - \delta_n p||$$

$$\leq ||(1 - \theta_n - \delta_n)(b_n - p) + \theta_n(t_n - p)|| + \delta_n ||p||.$$
(4.2.36)

But from (4.2.18), condition (4.2.3) and (4.2.35), we get that

$$\begin{aligned} \|(1-\theta_n-\delta_n)(b_n-p)+\theta_n(t_n-p)\|^2 &= (1-\theta_n-\delta_n)^2 \|b_n-p\|^2 + \theta_n^2 \|t_n-p\|^2 \\ &+ 2(1-\theta_n-\delta_n)\theta_n \langle b_n-p, t_n-p \rangle \\ &\leq (1-\theta_n-\delta_n)^2 \|b_n-p\|^2 + \theta_n^2 \|b_n-p\|^2 \\ &+ 2(1-\theta_n-\delta_n)\theta_n \|b_n-p\|^2 \\ &\leq (1-\delta_n)^2 \|w_n-p\|^2. \end{aligned}$$
(4.2.37)

Hence, (4.2.36) becomes

$$\begin{aligned} \|x_{n+1} - p\| &\leq (1 - \delta_n) \|w_n - p\| + \delta_n \|p\| \\ &\leq (1 - \delta_n) \left[\|x_n - p\| + \delta_n M_1 \right] + \delta_n \|p\| \\ &\leq (1 - \delta_n) \|x_n - p\| + \delta_n (M_1 + \|p\|), \end{aligned}$$

which implies from Lemma 2.5.35 that $\{x_n\}$ is bounded.

Lemma 4.2.5. Let $\{x_n\}$ be a sequence generated by Algorithm 4.2.2 under Assumption 4.2.1. Suppose that there exists a subsequence $\{x_{n_k}\}$ of $\{x_n\}$ which converges weakly to a point $z \in \mathcal{H}_1$ and $\lim_{k \to \infty} ||b_{n_k} - w_{n_k}|| = 0 = \lim_{k \to \infty} ||b_{n_k} - t_{n_k}||$. Then $z \in \Gamma$.

Proof. From Step 2 and (4.2.30), we get that

$$\lim_{n \to \infty} \|w_n - x_n\| = \lim_{n \to \infty} \delta_n \cdot \frac{\alpha_n}{\delta_n} \|x_n - x_{n-1}\| = 0.$$
(4.2.38)

Since the subsequence $\{x_{n_k}\}$ of $\{x_n\}$ is weakly convergent to a point $z \in \mathcal{H}_1$, it follows that the subsequence $\{w_{n_k}\}$ of $\{w_n\}$ is also weakly convergent to $z \in \mathcal{H}_1$. This, together with our hypothesis imply that $\{b_{n_k}\}$ converges weakly to z. Again, since T is a bounded linear operator, we obtain that $\{Tw_{n_k}\}$ converges weakly to Tz.

Now, without loss of generality, we may assume that $z_n \neq Tw_n$. Then,

$$\eta_n \in \left[\epsilon, \quad \frac{\|z_n - Tw_n\|^2}{\|T^*(z_n - Tw_n)\|^2} - \epsilon\right].$$

Thus, putting (4.2.15) into (4.2.34), we obtain that

$$||b_{n_k} - p||^2 \le ||w_{n_k} - p||^2 - \eta_{n_k} \epsilon ||T^*(z_{n_k} - Tw_{n_k})||^2 \le ||w_{n_k} - p||^2 - \epsilon^2 ||T^*(z_{n_k} - Tw_{n_k})||^2,$$
(4.2.39)

which implies that

$$\epsilon^{2} \|T^{*}(z_{n_{k}} - Tw_{n_{k}})\|^{2} \leq \|w_{n_{k}} - p\|^{2} - \|b_{n_{k}} - p\|^{2} \\ \leq \|w_{n_{k}} - b_{n_{k}}\|^{2} + 2\|w_{n_{k}} - b_{n_{k}}\|\|b_{n_{k}} - p\|.$$

Thus, by our hypothesis, we obtain

$$\lim_{k \to \infty} \|T^*(z_{n_k} - Tw_{n_k})\| = 0.$$
(4.2.40)

From (4.2.34) and (4.2.40), we obtain

$$\begin{aligned} \eta_{n_k} \|z_{n_k} - Tw_{n_k}\|^2 &\leq \|w_{n_k} - p\|^2 - \|b_{n_k} - p\|^2 + \eta_{n_k}^2 \|T^*(z_{n_k} - Tw_{n_k})\|^2 \\ &\leq \|w_{n_k} - b_{n_k}\|^2 + 2\|w_{n_k} - b_{n_k}\|\|b_{n_k} - p\| + \eta_{n_k}^2 \|T^*(z_{n_k} - Tw_{n_k})\|^2 \\ &\to 0, \quad \text{as} \quad k \to \infty. \end{aligned}$$

Since $\eta_{n_k} \ge \epsilon > 0$, we get that

$$\lim_{k \to \infty} \|z_{n_k} - Tw_{n_k}\| = 0.$$
(4.2.41)

From (4.2.17), we obtain

$$\phi \left[2 - \phi - \phi \lambda^2 L_2^2 - 2\lambda (1 - \phi) L_2 - 2\rho_2 \lambda K_2 \right] \|Tw_{n_k} - y_{n_k}\|^2 \le \|Tw_{n_k} - Tp\|^2 - \|z_{n_k} - Tp\|^2 \le \|Tw_{n_k} - z_{n_k}\|^2 + 2\|Tw_{n_k} - z_{n_k}\|\|z_{n_k} - Tp\|.$$

Thus, taking limit as $k \to \infty$ in the previous inequality, we obtain from (4.2.41) and (4.2.11) that

$$\lim_{k \to \infty} \|Tw_{n_k} - y_{n_k}\| = 0.$$
(4.2.42)

Now, by the monotonicity of F, the characteristic property of $P_{\mathcal{Q}_n}$ and $\mathcal{Q} \subset \mathcal{Q}_n$, we obtain for all $w \in \mathcal{Q}$ that

$$\begin{array}{rcl}
0 &\leq & \langle y_{n_{k}} - Tw_{n_{k}} + \lambda FTw_{n_{k}}, w - y_{n_{k}} \rangle \\
&= & \langle y_{n_{k}} - Tw_{n_{k}}, w - y_{n_{k}} \rangle + \lambda \langle FTw_{n_{k}}, Tw_{n_{k}} - y_{n_{k}} \rangle + \lambda \langle FTw_{n_{k}}, w - Tw_{n_{k}} \rangle \\
&\leq & \|y_{n_{k}} - Tw_{n_{k}}\| \|w - y_{n_{k}}\| + \lambda \|FTw_{n_{k}}\| \|Tw_{n_{k}} - y_{n_{k}}\| \\
&\quad + \lambda [\langle FTw_{m_{k}} - Fw, w - Tw_{n_{k}} \rangle + \langle Fw, w - Tw_{n_{k}} \rangle] \\
&\leq & \|y_{n_{k}} - Tw_{n_{k}}\| \|w - y_{n_{k}}\| + \lambda \|FTw_{n_{k}}\| \|Tw_{n_{k}} - y_{n_{k}}\| \\
&\quad + \lambda \langle Fw, w - Tw_{n_{k}} \rangle.
\end{array}$$
(4.2.43)

Thus, passing limit as $k \to \infty$ in the previous inequality, we obtain

$$\langle Fw, w - Tz \rangle \ge 0, \quad \forall w \in \mathcal{Q},$$

and from Lemma 3.2.2, we have that $Tz \in VI(F, \mathcal{Q})$.

Again, by our hypothesis and (4.2.18), we obtain that

$$\phi \left[2 - \phi - \phi \mu^2 L_1^2 - 2\mu (1 - \phi) L_1 - 2\rho_1 \mu K_1 \right] \|b_{n_k} - u_{n_k}\|^2 \le \|b_{n_k} - p\|^2 - \|t_{n_k} - p\|^2$$

$$\le \|b_{n_k} - t_{n_k}\|^2$$

$$+ 2\|b_{n_k} - t_{n_k}\| \|t_{n_k} - p\|$$

$$\to 0, \text{ as } k \to \infty,$$

which by condition (4.2.3), implies that

$$\lim_{k \to \infty} \|b_{n_k} - u_{n_k}\| = 0 \tag{4.2.44}$$

Similar to (4.2.43), we obtain from the monotonicity of A, the characteristic property of $P_{\mathcal{C}_n}$ and $\mathcal{C} \subset \mathcal{C}_n$, that

$$0 \le ||u_{n_k} - b_{n_k}|| ||x - u_{n_k}|| + \mu ||Ab_{n_k}|| ||b_{n_k} - u_{n_k}|| + \mu \langle Ax, x - b_{n_k} \rangle, \ \forall x \in \mathcal{C}.$$

Passing limit as $k \to \infty$ in the previous inequality, we obtain from (4.2.44) that

$$\langle Ax, x-z \rangle \ge 0, \quad \forall x \in \mathcal{C}.$$

Therefore, we obtain from Lemma 3.2.2 that $z \in VI(A, \mathcal{C})$. Hence, $z \in \Gamma$.

We now present the first theorem of this work.

Theorem 4.2.6. Let $\{x_n\}$ be a sequence generated by Algorithm 4.2.2 under Assumption 4.2.1. Then, $\{x_n\}$ converges strongly to $p \in \Gamma$, where $||p|| = \min\{||z|| : z \in \Gamma\}$.

Proof. Let $p \in \Gamma$. Then, from Step 2, we obtain

$$\begin{aligned} \|w_n - p\|^2 &= \|x_n - p\|^2 + \alpha_n^2 \|x_n - x_{n-1}\|^2 + 2\alpha_n \langle x_n - p, x_n - x_{n-1} \rangle \\ &\leq \|x_n - p\|^2 + \alpha_n^2 \|x_n - x_{n-1}\|^2 + 2\alpha_n \|x_n - p\| \|x_n - x_{n-1}\| \\ &= \|x_n - p\|^2 + \alpha_n \|x_n - x_{n-1}\| [\alpha_n \|x_n - x_{n-1}\| + 2\|x_n - p\|] \\ &\leq \|x_n - p\|^2 + 3\alpha_n \|x_n - x_{n-1}\| M_2, \end{aligned}$$
(4.2.45)

for some $M_2 > 0$. Thus, from (4.2.35), (4.2.18) and (4.2.45), we get

$$\begin{aligned} \|(1-\theta_n)b_n + \theta_n t_n - p\|^2 &= \|(1-\theta_n)(b_n - p) + \theta_n(t_n - p)\|^2 \\ &= (1-\theta_n)^2 \|b_n - p\|^2 + \theta_n^2 \|t_n - p\|^2 + 2(1-\theta_n)\theta_n \langle b_n - p, t_n - p \rangle \\ &\leq (1-\theta_n)^2 \|w_n - p\|^2 + \theta_n^2 \|w_n - p\|^2 + 2(1-\theta_n)\theta_n \langle b_n - p, t_n - p \rangle \\ &\leq (1-\theta_n)^2 \|w_n - p\|^2 + \theta_n^2 \|w_n - p\|^2 + 2(1-\theta_n)\theta_n \|w_n - p\|^2 \\ &\leq \|x_n - p\|^2 + 3\alpha_n \|x_n - x_{n-1}\| M_2. \end{aligned}$$

From Step 5, the previous inequality and (4.2.30), we get

$$\begin{aligned} \|x_{n+1} - p\|^2 &= \|(1 - \delta_n) \left[(1 - \theta_n) b_n + \theta_n t_n - p \right] - \left[\delta_n \theta_n (b_n - t_n) + \delta_n p \right] \|^2 \\ &\leq (1 - \delta_n)^2 \| (1 - \theta_n) b_n + \theta_n t_n - p \|^2 - 2 \langle \delta_n \theta_n (b_n - t_n) + \delta_n p, x_{n+1} - p \rangle \\ &\leq (1 - \delta_n) \| (1 - \theta_n) b_n + \theta_n t_n - p \|^2 + 2 \delta_n \theta_n \langle b_n - t_n, p - x_{n+1} \rangle \\ &+ 2 \delta_n \langle p, p - x_{n-1} \rangle \\ &\leq (1 - \delta_n) \| (1 - \theta_n) b_n + \theta_n t_n - p \|^2 + 2 \delta_n \theta_n \| b_n - t_n \| \| p - x_{n+1} \| \\ &+ 2 \delta_n \langle p, p - x_{n+1} \rangle \\ &\leq (1 - \delta_n) \left(\|x_n - p\|^2 + 3 \alpha_n \|x_n - x_{n-1}\| M_2 \right) + 2 \delta_n \theta_n \| b_n - t_n \| \| p - x_{n+1} \| \\ &+ 2 \delta_n \langle p, p - x_{n+1} \rangle \\ &\leq (1 - \delta_n) \| x_n - p \|^2 + 3 \alpha_n \| x_n - x_{n-1} \| M_2 + 2 \delta_n \theta_n \| b_n - t_n \| \| p - x_{n+1} \| \\ &+ 2 \delta_n \langle p, p - x_{n+1} \rangle \\ &= (1 - \delta_n) \| x_n - p \|^2 + \delta_n d_n, \ n \geq 1, \end{aligned}$$

where $d_n = 3\frac{\alpha_n}{\delta_n} \|x_n - x_{n-1}\| M_2 + 2(\theta_n \|b_n - t_n\| \|p - x_{n+1}\| + \langle p, p - x_{n+1} \rangle)$. According to Lemma 2.5.36, to conclude our proof, it suffices to show that $\limsup_{k \to \infty} d_{n_k} \leq 0$ for every subsequence $\{\|x_{n_k} - p\|\}$ of $\{\|x_n - p\|\}$ satisfying the condition;

$$\liminf_{k \to \infty} \left(\|x_{n_k+1} - p\| - \|x_{n_k} - p\| \right) \ge 0.$$
(4.2.46)

To show that $\limsup_{k\to\infty} d_{n_k} \leq 0$, suppose that $\{\|x_{n_k} - p\|\}$ is a subsequence of $\{\|x_n - p\|\}$ such that (4.2.46) holds. Then,

$$\liminf_{k \to \infty} \left(\|x_{n_k+1} - p\|^2 - \|x_{n_k} - p\|^2 \right) = \liminf_{k \to \infty} \left[\left(\|x_{n_k+1} - p\| - \|x_{n_k} - p\| \right) \\ \times \left(\|x_{n_k+1} - p\| + \|x_{n_k} - p\| \right) \right] \\ \ge 0.$$
(4.2.47)

Now, observe that

$$\begin{aligned} \|(1-\theta_{n}-\delta_{n})(b_{n}-p)+\theta_{n}(t_{n}-p)\|^{2} &= (1-\theta_{n}-\delta_{n})^{2}\|b_{n}-p\|^{2} \\ &+ 2(1-\theta_{n}-\delta_{n})\theta_{n}\langle b_{n}-p,t_{n}-p\rangle + \theta_{n}^{2}\|t_{n}-p\|^{2} \\ &\leq (1-\theta_{n}-\delta_{n})^{2}\|b_{n}-p\|^{2} \\ &+ (1-\theta_{n}-\delta_{n})\theta_{n}\|b_{n}-p\|^{2} \\ &+ (1-\theta_{n}-\delta_{n})\theta_{n}\|t_{n}-p\|^{2} + \theta_{n}^{2}\|t_{n}-p\|^{2} \\ &\leq (1-\theta_{n}-\delta_{n})(1-\delta_{n})\|b_{n}-p\|^{2} \\ &+ \theta_{n}(1-\delta_{n})\|t_{n}-p\|^{2}. \end{aligned}$$
(4.2.48)

Then, from Step 5, (4.2.48) and (4.2.18), we get

$$\begin{aligned} \|x_{n+1} - p\|^2 &= \|(1 - \theta_n - \delta_n)(b_n - p) + \theta_n(t_n - p) - \delta_n p\|^2 \\ &= \|(1 - \theta_n - \delta_n)(b_n - p) + \theta_n(t_n - p)\|^2 + \delta_n^2 \|p\|^2 \\ &- 2\delta_n \langle (1 - \theta_n - \delta_n)(b_n - p) + \theta_n(t_n - p), p \rangle \\ &\leq (1 - \theta_n - \delta_n)(1 - \delta_n) \|b_n - p\|^2 + \theta_n(1 - \delta_n) \|t_n - p\|^2 + \delta_n M_3 \\ &\leq (1 - \theta_n - \delta_n)(1 - \delta_n) \|b_n - p\|^2 \\ &+ \theta_n(1 - \delta_n) \Big(\phi \left[2 - \phi - \phi \mu^2 L_1^2 - 2\mu(1 - \phi) L_1 - 2\rho_1 \mu K_1 \right] \|b_n - u_n\|^2 \Big) \\ &\leq (1 - \theta_n - \delta_n)(1 - \delta_n) \|w_n - p\|^2 + \delta_n M_3 \\ &+ \theta_n(1 - \delta_n) \|w_n - p\|^2 \\ &- \theta_n(1 - \delta_n) \Big(\phi \left[2 - \phi - \phi \mu^2 L_1^2 - 2\mu(1 - \phi) L_1 - 2\rho_1 \mu K_1 \right] \|b_n - u_n\|^2 \Big) \\ &\leq \|x_n - p\|^2 \\ &- \theta_n(1 - \delta_n) \phi \left[2 - \phi - \phi \mu^2 L_1^2 - 2\mu(1 - \phi) L_1 - 2\rho_1 \mu K_1 \right] \|b_n - u_n\|^2 \Big) \\ &\leq \|x_n - p\|^2 \\ &+ \delta_n \left(3 \frac{\alpha_n}{\delta_n} \|x_n - x_{n-1}\| M_2 + M_3 \right), \end{aligned}$$

for some $M_3 > 0$. Let $V := 2 - \phi - \phi \mu^2 L_1^2 - 2\mu (1 - \phi) L_1 - 2\rho_1 \mu K_1$. Then, from (4.2.50), (4.2.47) and (4.2.31), we obtain

$$\limsup_{k \to \infty} \left(\theta_{n_k} (1 - \delta_{n_k}) \phi V \| b_{n_k} - u_{n_k} \|^2 \right) \leq \limsup_{k \to \infty} \left(\| x_{n_k} - p \|^2 - \| x_{n_k+1} - p \|^2 \right) \\
+ \limsup_{k \to \infty} \delta_{n_k} \left(3M_1 M_2 + M_3 \right) \\
= -\liminf_{k \to \infty} \left(\| x_{n_k+1} - p \|^2 - \| x_{n_k} - p \|^2 \right) \\
\leq 0.$$

This implies that

$$\lim_{k \to \infty} \|b_{n_k} - u_{n_k}\| = 0.$$
(4.2.51)

Since A is Lipschitz continuous on \mathcal{H}_1 , we obtain from (4.2.51) that

$$\lim_{k \to \infty} \|Ab_{n_k} - Au_{n_k}\| = 0.$$
(4.2.52)

Also, from Step 5, we have that

$$\|t_{n_k} - b_{n_k}\| = \|(1 - \phi)b_{n_k} + \phi u_{n_k} + \phi \mu(Ab_{n_k} - Au_{n_k}) - b_{n_k}\|$$

$$\leq \phi \|u_{n_k} - b_{n_k}\| + \phi \mu \|Ab_{n_k} - Au_{n_k}\|,$$

which implies from (4.2.51) and (4.2.52) that

$$\lim_{k \to \infty} \|t_{n_k} - b_{n_k}\| = 0.$$
(4.2.53)

From (4.2.49) and (4.2.39), we obtain

$$\begin{aligned} \|x_{n_k+1} - p\|^2 &\leq (1 - \theta_{n_k} - \delta_{n_k})(1 - \delta_{n_k}) \|b_{n_k} - p\|^2 + \theta_{n_k}(1 - \delta_{n_k}) \|b_{n_k} - p\|^2 + \delta_{n_k} M_3 \\ &\leq \|b_{n_k} - p\|^2 + \delta_{n_k} M_3 \\ &\leq \|w_{n_k} - p\|^2 - \epsilon^2 \|T^*(z_{n_k} - Tw_{n_k})\|^2 + \delta_{n_k} M_3, \end{aligned}$$

which implies from (4.2.45) that

$$\begin{split} \limsup_{k \to \infty} \|T^*(z_{n_k} - Tw_{n_k})\|^2 &\leq \frac{1}{\epsilon^2} \limsup_{k \to \infty} \left(\|x_{n_k} - p\|^2 - \|x_{n_k+1} - p\|^2 \right) \\ &+ \frac{1}{\epsilon^2} \limsup_{k \to \infty} \left(\delta_{n_k} \left(3\frac{\alpha_{n_k}}{\delta_{n_k}} \|x_{n_k} - x_{n_k-1}\|M_2 + M_3 \right) \right) \\ &\leq \frac{-1}{\epsilon^2} \liminf_{k \to \infty} \left(\|x_{n_k+1} - p\|^2 - \|x_{n_k} - p\|^2 \right) \leq 0. \end{split}$$

Thus,

$$\lim_{k \to \infty} \|T^*(z_{n_k} - Tw_{n_k})\| = 0.$$
(4.2.54)

Hence, by Step 3, we obtain

$$||b_{n_k} - w_{n_k}|| = \eta_{n_k} ||T^*(z_{n_k} - Tw_{n_k})|| \to 0, \quad \text{as} \quad k \to \infty.$$
(4.2.55)

From Step 5 and (4.2.53), we get

$$||x_{n_k+1} - b_{n_k}|| \le \theta_{n_k} ||b_{n_k} - t_{n_k}|| + \delta_{n_k} ||b_{n_k}|| \to 0, \text{ as } k \to \infty.$$
(4.2.56)

From (4.2.55) and (4.2.56), we get

$$\lim_{k \to \infty} \|x_{n_k+1} - w_{n_k}\| = 0.$$
(4.2.57)

From (4.2.38) and (4.2.57), we get

$$\lim_{k \to \infty} \|x_{n_k+1} - x_{n_k}\| = 0.$$
(4.2.58)

By Lemma 4.2.4, there exists a subsequence $\{x_{n_{k_j}}\}$ of $\{x_{n_k}\}$ which converges weakly to z, and such that

$$\limsup_{k \to \infty} \langle p, p - x_{n_k} \rangle = \lim_{j \to \infty} \langle p, p - x_{n_{k_j}} \rangle = \langle p, p - z \rangle.$$

Also, since we have established (4.2.53) and (4.2.55), we can apply Lemma 4.2.5 to get that $z \in \Gamma$. Note also that $p = P_{\Gamma}0$. Thus, we obtain from the previous inequality that

$$\limsup_{k \to \infty} \langle p, p - x_{n_k} \rangle = \langle p, p - z \rangle \le 0,$$

which together with (4.2.58), imply that

$$\lim_{k \to \infty} \langle p, p - x_{n_k+1} \rangle \le 0. \tag{4.2.59}$$

Now, recall that $d_{n_k} = 3 \frac{\alpha_{n_k}}{\delta_{n_k}} \|x_{n_k} - x_{n_{k-1}}\| M_2 + 2\left(\theta_{n_k} \|b_{n_k} - t_{n_k}\| \|p - x_{n_{k+1}}\| + \langle p, p - x_{n_{k+1}} \rangle\right)$. So, by (4.2.30), (4.2.53) and (4.2.59), we obtain that $\limsup_{k \to \infty} d_{n_k} \leq 0$. Hence, we get that $\lim_{n \to \infty} \|x_n - p\| = 0$. Therefore, $\{x_n\}$ converges strongly to $p = P_{\Gamma} 0$.

Remark 4.2.7.

- We observe that if we set $\mathcal{H}_1 = \mathcal{H}_2 = \mathcal{H}$, F = 0 and $T = I_{\mathcal{H}}$ (the identity operator on \mathcal{H}) in Algorithm 4.2.2, we obtain as a corollary, a relaxed inertial Tseng's forward-backward-forward with fixed stepsizes, involving only one projection onto the half-space \mathcal{C}_n per iteration, for solving the classical VIP (1.2.4) when A is monotone and Lipschitz continuous. This method for solving the classical VIP (1.2.4) is also the first in the literature.
- The conclusion of Theorem 4.2.6 still holds even if the fixed stepsizes λ and μ given by (4.2.2) and (4.2.3), respectively, are replaced with variable stepsizes μ_n and λ_n , respectively such that

$$0 < \inf_{n \ge 1} \mu_n \le \sup_{n \ge 1} \mu_n < \frac{\sqrt{\rho_2^2 K_2^2 + 2\rho_2 K_2 (1-\phi) L_2 + L_2^2} - [\rho_2 K_2 + (1-\phi) L_2]}{\phi L_2^2}$$

and

$$0 < \inf_{n \ge 1} \lambda_n \le \sup_{n \ge 1} \lambda_n < \frac{\sqrt{\rho_1^2 K_1^2 + 2\rho_1 K_1 (1 - \phi) L_1 + L_1^2 - [\rho_1 K_1 + (1 - \phi) L_1]}}{\phi L_1^2}.$$

Having concluded the convergence analysis of the first method of this work, we now study the convergence analysis of the second method (Algorithm 4.2.3), which do not depend on the availability of the Lipschitz constants. We begin with the following useful remarks and lemmas.

Remark 4.2.8. The stepsizes $\{\lambda_n\}$ and $\{\mu_n\}$ generated by Algorithm 4.2.3 are monotonically non-increasing sequences with lower bounds min $\left\{\lambda_1, \frac{a_2}{\sqrt{L_2^2 + K_2^2}}\right\}$ and min $\left\{\mu_1, \frac{a_1}{\sqrt{L_1^2 + K_1^2}}\right\}$,

respectively.

Indeed, it is obvious that $\lambda_{n+1} \leq \lambda_n$, $\forall n \in \mathbb{N}$. Also, since F is L_2 -Lipschitz continuous and q' is K₂-Lipschitz continuous, we get in the case where $||FTw_n - Fy_n||^2 + ||q'(Tw_n) - Fy_n||^2$ $q'(y_n) \parallel^2 \neq 0$ that

$$\lambda_{n+1} = \min\left\{\frac{a_2 \|Tw_n - y_n\|}{\sqrt{\|FTw_n - Fy_n\|^2 + \|q'(Tw_n) - q'(y_n)\|^2}}, \lambda_n\right\}$$

$$\geq \min\left\{\frac{a_2 \|Tw_n - y_n\|}{\sqrt{L_2^2} \|Tw_n - y_n\|^2 + K_2^2 \|Tw_n - y_n\|^2}}, \lambda_n\right\}$$

$$= \min\left\{\frac{a_2}{\sqrt{L_2^2 + K_2^2}}, \lambda_n\right\}.$$

Hence, by induction, we obtain that $\{\lambda_n\}$ is bounded below by min $\left\{\frac{a_2}{\sqrt{L_2^2+K_2^2}}, \lambda_1\right\}$. Similar argument holds for $\{\mu_n\}$.

Note that Remark 4.2.8 implies that the limits of $\{\lambda_n\}, \{\mu_n\}$ exist, and $\lim_{n\to\infty} \lambda_n \geq 1$ $\min\left\{\frac{a_2}{\sqrt{L_2^2 + K_2^2}}, \lambda_1\right\}, \lim_{n \to \infty} \mu_n \ge \min\left\{\frac{a_1}{\sqrt{L_1^2 + K_1^2}}, \mu_1\right\}.$

Remark 4.2.9. From (4.2.5), we observe that

$$\sqrt{\|FTw_n - Fy_n\|^2 + \|q'(Tw_n) - q'(y_n)\|^2} \le \frac{a_2}{\lambda_{n+1}} \|Tw_n - y_n\|, \quad \forall n \in \mathbb{N}$$

holds for both $||FTw_n - Fy_n|| + ||q'(Tw_n) - q'(y_n)|| \neq 0$ and otherwise. This implies that

$$\|FTw_n - Fy_n\|^2 + \|q'(Tw_n) - q'(y_n)\|^2 \le \frac{a_2^2}{\lambda_{n+1}^2} \|Tw_n - y_n\|^2, \ \forall n \in \mathbb{N},$$
(4.2.60)

which further implies

$$||FTw_n - Fy_n|| \le \frac{a_2}{\lambda_{n+1}} ||Tw_n - y_n||, \ \forall n \in \mathbb{N}.$$
 (4.2.61)

Lemma 4.2.10. Let $\{x_n\}$ be a sequence generated by Algorithm 4.2.3 under Assumption 4.2.1. Then, for all $p \in \Gamma$, there exists $n_0 \ge 1$ such that

$$||Tw_n - y_n||^2 \le K_n \left(||Tw_n - z_n||^2 + 2||Tw_n - z_n||||z_n - Tp|| \right), \qquad (4.2.62)$$

and

$$||b_n - u_n||^2 \le L_n \left(||b_n - t_n||^2 + 2||b_n - t_n||||t_n - p|| \right), \qquad (4.2.63)$$

where

$$K_{n} = \left(\frac{1}{\phi_{n}\left[2 - \phi_{n} - \lambda_{n}^{2}\frac{a_{2}^{2}}{\lambda_{n+1}^{2}} - \rho_{2}^{2} - 2\lambda_{n}(1 - \phi_{n})\frac{a_{2}}{\lambda_{n+1}}\right]}\right)$$

and

$$L_n = \left(\frac{1}{\phi_n \left[2 - \phi_n - \mu_n^2 \frac{a_1^2}{\lambda_{n+1}^2} - \rho_1^2 - 2\mu_n (1 - \phi_n) \frac{a_1}{\mu_{n+1}}\right]}\right),$$

with ρ_1, ρ_2 been positive constants, for all $n \ge n_0$.

Proof. By replacing λ with λ_n and ϕ with ϕ_n in (4.2.24), we have

$$||z_{n} - Tp||^{2} \leq ||Tw_{n} - Tp||^{2} - \phi_{n}(2 - \phi_{n})||Tw_{n} - y_{n}||^{2}$$

$$+ \phi_{n}^{2}\lambda_{n}^{2}||FTw_{n} - Fy_{n}||^{2} + 2\lambda_{n}\phi_{n}(1 - \phi_{n})\langle Tw_{n} - y_{n}, FTw_{n} - Fy_{n}\rangle$$

$$+ 2\lambda_{n}\phi_{n}\langle FTp, Tp - y_{n}\rangle.$$

$$(4.2.64)$$

Suppose FTp = 0. Then, we obtain from (4.2.61) and (4.2.64) that

$$\begin{aligned} \|z_{n} - Tp\|^{2} &\leq \|Tw_{n} - Tp\|^{2} - \phi_{n}(2 - \phi_{n})\|Tw_{n} - y_{n}\|^{2} + \phi_{n}\lambda_{n}^{2}\|FTw_{n} - Fy_{n}\|^{2} \\ &+ 2\lambda_{n}\phi_{n}(1 - \phi_{n})\langle Tw_{n} - y_{n}, FTw_{n} - Fy_{n}\rangle \\ &\leq \|Tw_{n} - Tp\|^{2} - \phi_{n}(2 - \phi_{n})\|Tw_{n} - y_{n}\|^{2} + \phi_{n}\lambda_{n}^{2}\frac{a_{2}^{2}}{\lambda_{n+1}^{2}}\|Tw_{n} - y_{n}\|^{2} \\ &+ 2\lambda_{n}\phi_{n}(1 - \phi_{n})\frac{a_{2}}{\lambda_{n+1}}\|Tw_{n} - y_{n}\|^{2} \\ &= \|Tw_{n} - Tp\|^{2} - \phi_{n}\left[2 - \phi_{n} - \lambda_{n}^{2}\frac{a_{2}^{2}}{\lambda_{n+1}^{2}} - 2\lambda_{n}(1 - \phi_{n})\frac{a_{2}}{\lambda_{n+1}}\right]\|Tw_{n} - y_{n}\|^{2} \\ &\leq \|Tw_{n} - Tp\|^{2} \\ &- \phi_{n}\left[2 - \phi_{n} - \lambda_{n}^{2}\frac{a_{2}^{2}}{\lambda_{n+1}^{2}} - 2\lambda_{n}(1 - \phi_{n})\frac{a_{2}}{\lambda_{n+1}} - \rho_{2}^{2}\right]\|Tw_{n} - y_{n}\|^{2}, \end{aligned}$$

for some $\rho_2 > 0$.

Now, since the limit of $\{\lambda_n\}$ exists, then $\lim_{n \to \infty} \lambda_n = \lim_{n \to \infty} \lambda_{n+1}$. Hence, by Assumption 4.2.1 (f) and the condition on a_2 , we obtain that

$$\lim_{n \to \infty} \left[2 - \phi_n - \lambda_n^2 \frac{a_2^2}{\lambda_{n+1}^2} - \rho_2^2 - 2\lambda_n (1 - \phi_n) \frac{a_2}{\lambda_{n+1}} \right] = \left(2 - \psi - a_2^2 - \rho_2^2 - 2a_2(1 - \psi) \right) > 0.$$
(4.2.66)

Thus, there exists $n_0 \ge 1$ such that

$$2 - \phi_n - \lambda_n^2 \frac{a_2^2}{\lambda_{n+1}^2} - \rho_2^2 - 2\lambda_n (1 - \phi_n) \frac{a_2}{\lambda_{n+1}} > 0, \ \forall n \ge n_0.$$
(4.2.67)

Hence, we obtain from (4.2.65) that

$$||Tw_{n} - y_{n}||^{2} \leq \left[\frac{1}{\phi_{n}\left(2 - \phi_{n} - \lambda_{n}^{2}\frac{a_{2}^{2}}{\lambda_{n+1}^{2}} - \rho_{2}^{2} - 2\lambda_{n}(1 - \phi_{n})\frac{a_{2}}{\lambda_{n+1}}\right)}\right]$$
$$\cdot \left(||Tw_{n} - Tp||^{2} - ||z_{n} - Tp||^{2}\right)$$
$$= \left[\frac{1}{\phi_{n}\left(2 - \phi_{n} - \lambda_{n}^{2}\frac{a_{2}^{2}}{\lambda_{n+1}^{2}} - \rho_{2}^{2} - 2\lambda_{n}(1 - \phi_{n})\frac{a_{2}}{\lambda_{n+1}}\right)}\right]$$
$$\cdot \left(||(Tw_{n} - z_{n}) + (z_{n} - Tp)||^{2} - ||z_{n} - Tp||^{2}\right)$$
(4.2.68)

for all $n \ge n_0$, which by some simple simplifications, gives (4.2.62).

Now, suppose $FTp \neq 0$. Then, from (4.2.26) and (4.2.29), we obtain

$$\langle FTp, Tp - y_n \rangle \le \rho_2 \langle q'(y_n) - q'(Tw_n), y_n - Tw_n \rangle.$$
(4.2.69)

Also, observe that

$$2\lambda_n \rho_2 \langle q'(y_n) - q'(Tw_n), y_n - Tw_n \rangle \le \lambda_n^2 \|q'(y_n) - q'(Tw_n)\|^2 + \rho_2^2 \|y_n - Tw_n\|^2.$$
(4.2.70)

Substituting (4.2.69) and (4.2.70) into (4.2.64), and using (4.2.61) and (4.2.60), we obtain

$$\begin{split} \|z_{n} - Tp\|^{2} &\leq \|Tw_{n} - Tp\|^{2} - \phi_{n}(2 - \phi_{n})\|Tw_{n} - y_{n}\|^{2} + \phi_{n}\lambda_{n}^{2}\|FTw_{n} - Fy_{n}\|^{2} \\ &+ 2\lambda_{n}\phi_{n}(1 - \phi_{n})\langle Tw_{n} - y_{n}, FTw_{n} - Fy_{n}\rangle + \phi_{n}\lambda_{n}^{2}\|q'(y_{n}) - q'(Tw_{n})\|^{2} \\ &+ \phi_{n}\rho_{2}^{2}\|y_{n} - Tw_{n}\|^{2} \\ &= \|Tw_{n} - Tp\|^{2} - \phi_{n}(2 - \phi_{n})\|Tw_{n} - y_{n}\|^{2} \\ &+ \phi_{n}\lambda_{n}^{2}\left[\|FTw_{n} - Fy_{n}\|^{2} + \|q'(Tw_{n}) - q'(y_{n})\|^{2}\right] \\ &+ 2\lambda_{n}\phi_{n}(1 - \phi_{n})\langle Tw_{n} - y_{n}, FTw_{n} - Fy_{n}\rangle + \phi_{n}\rho_{2}^{2}\|y_{n} - Tw_{n}\|^{2} \\ &\leq \|Tw_{n} - Tp\|^{2} - \phi_{n}(2 - \phi_{n})\|Tw_{n} - y_{n}\|^{2} + \phi_{n}\lambda_{n}^{2}\frac{a_{2}^{2}}{\lambda_{n+1}^{2}}\|Tw_{n} - y_{n}\|^{2} \\ &+ 2\lambda_{n}\phi_{n}(1 - \phi_{n})\frac{a_{2}}{\lambda_{n+1}}\|Tw_{n} - y_{n}\|^{2} + \phi_{n}\rho_{2}^{2}\|y_{n} - Tw_{n}\|^{2} \\ &= \|Tw_{n} - Tp\|^{2} \\ &- \phi_{n}\left[2 - \phi_{n} - \lambda_{n}^{2}\frac{a_{2}^{2}}{\lambda_{n+1}^{2}} - 2\lambda_{n}(1 - \phi_{n})\frac{a_{2}}{\lambda_{n+1}} - \rho_{2}^{2}\right]\|Tw_{n} - y_{n}\|^{2}. \end{split}$$

Notice that this is the same as (4.2.65). Hence, we obtain (4.2.62). Also, we can follow the same line of proof as above to establish (4.2.63).

We now state and prove the strong convergence theorem for Algorithm 4.2.3.

Theorem 4.2.11. Let $\{x_n\}$ be a sequence generated by Algorithm 4.2.3 under Assumption 4.2.1. Then, $\{x_n\}$ converges strongly to $p \in \Gamma$, where $||p|| = \min\{||z|| : z \in \Gamma\}$.

Proof. Let $p \in \Gamma$. Then, by replacing μ and λ with μ_n and λ_n , respectively in Lemma 4.2.4, and using (4.2.65) and (4.2.67), we can easily get that $\{x_n\}$ is bounded. Furthermore, we can follow the same argument as in the proof of Theorem 4.2.6 to obtain (4.2.53), (4.2.54) and (4.2.55). Also, using (4.2.54) and (4.2.55), we can get (4.2.41).

Hence, from (4.2.41), Lemma 4.2.10 (that is, (4.2.62)) and (4.2.66), we obtain that

$$\lim_{k \to \infty} \|Tw_{n_k} - y_{n_k}\| = 0.$$

In a similar way, we can show that

$$\lim_{k \to \infty} \|b_{n_k} - u_{n_k}\| = 0.$$

Following the same arguments as in Lemma 4.2.5 and Theorem 4.2.6, we obtain the rest of the proof. $\hfill \Box$

Remark 4.2.12.

- (1). Similar to Remark 4.2.7, we obtain a relaxed inertial Tseng's forward-backward-forward with adaptive stepsizes, involving only one projection onto the half-space C_n per iteration, for solving the classical VIP (1.2.4) when A is monotone and Lipschitz continuous as a corollary. This method is also new.
- (2). Our proofs are independent of the usual "Two cases Approach" which have been used widely in many works (see, for example, [106, 221, 244]) to guarantee strong convergence of iterative methods. Hence, the techniques used in obtaining our strong convergence analysis are new for solving the SVIP (1.2.4)-(1.2.5).
- (3). If Step 1 of Algorithm 4.2.2 and Algorithm 4.2.3 are replaced with the following: $0 \le \alpha_n \le \bar{\alpha}_n$, where

$$\bar{\alpha}_n := \begin{cases} \min\left\{\alpha, \frac{\tau_n}{\|x_n - x_{n-1}\|}\right\}, & \text{if } x_n \neq x_{n-1} \\ \alpha, & \text{otherwise,} \end{cases}$$

with $\alpha \in [0, 1)$, we still have valid conclusions of Theorem 4.2.6 and Theorem 4.2.11.

4.2.3 Numerical experiments

In this section, using various test examples, we discuss the numerical behavior of Algorithm 4.2.2 and Algorithm 4.2.3, and we compare them with the methods of He *et al.* [121, Algorithm 1] (see Appendix 4.2.17), Reich and Tuyen [213, Theorem 4.4] (see Appendix 4.2.18) and Tian and Jiang [244, Algorithm (4.8)] (see Appendix 4.2.19).

We perform all implementations using Matlab 2016 (b), installed on a personal computer with Intel(R) Core(TM) i5-2600 CPU@2.30GHz and 8.00 Gb-RAM running on Windows 10 operating system. In Tables 4.2.1-4.2.3, "Iter." means the number of iterations while "CPU" means the CPU time in seconds.

In our computations, we randomly choose the relaxation stepsize $\phi \in (0, 1]$ and the starting points $x_0, x_1 \in \mathcal{H}_1$ (see the cases below). We also choose randomly, the parameters $\eta \geq 0$, $\lambda_1 > 0$ and $\mu_1 > 0$ (the choices of these parameters will be discussed in Remark 4.2.16). We choose $\delta_n = \frac{1}{n+1}$, $\theta_n = \frac{1}{2} - \delta_n$ and $\alpha_n = \bar{\alpha}_n$ with $\tau_n = \frac{\delta_n}{n^{0.01}}$ and different choices of $\alpha := 3, 6, 9, 12, 15$, which we will discuss in detail in our numerical analysis. Furthermore, in the implementation, we define

$$TOL_n := \frac{1}{2} \left(||x_n - P_{\mathcal{C}_n}(x_n - \mu A x_n)||^2 + ||Tx_n - P_{\mathcal{Q}_n}(Tx_n - \lambda FTx_n)||^2 \right),$$

and use the stopping criterion $\text{TOL}_n < \varepsilon$ for the iterative processes, where ε is the predetermined error.

Firstly, we consider two examples in finite dimensional spaces. In these examples, we compare Algorithm 4.2.2 with the method of He *et al.* [121] since their method is given in finite dimension spaces.

Example 4.2.13. Consider the following separable, convex and quadratic programming problem, which has been considered by [121, Example 5.2] for their numerical experiments.

$$\min_{x,y} \{ G_1(x) + G_2(y) \mid Tx = y, \ x \in \mathcal{C}, \ y \in \mathcal{Q} \},$$
(4.2.71)

where

$$G_1(x) = \frac{1}{2}x^T M_1 x + q_1^T x$$
 and $G_2(y) = \frac{1}{2}y^T M_2 y + q_2^T y$ (x^T means the transpose of x).

Problem (4.2.71) can also be rewritten as the SVIP (1.2.4)-(1.2.5), where

$$A(x) = M_1 x + q_1$$
 and $F(y) = M_2 y + q_2$.

For i = 1, 2, the matrices M_i are formed as: $M_i = V_i \sum_i V_i^T$, where $V_i = I - \frac{2v_i v_i^T}{\|v_i\|^2}$ and $\sum_i = \text{diag}(\sigma_{i1}, \sigma_{i2}, \cdots, \sigma_{iN_i})$ are the Householder and the diagonal matrices, respectively, with $N_1 = N$ and $N_2 = m$ being the dimensional of x and y, respectively. Furthermore,

$$\sigma_{i,j} = \cos\frac{j\pi}{N_i + 1} + 1 + \frac{\cos\frac{\pi}{N_i + 1} + 1 - \widehat{C}_i(\cos\frac{N_i\pi}{N_i + 1} + 1)}{\widehat{C}_i - 1}, \quad j = 1, 2, \cdots, N_i$$

where \widehat{C}_i is the preset condition number of M_i .

As done in He *et al.* [121, Example 5.2], we choose $\widehat{C}_i = 10^4$, $q_i = 0$, i = 1, 2 and uniformly take the vector $v_i \in \mathbb{R}^{N_i}$ (i = 1, 2) in (-1, 1). In this case, A and F are monotone and Lipschitz continuous operators with $L_i = ||M_i||$, i = 1, 2 (which can be computed in Matlab). Furthermore, we generate the bounded linear operator $T \in \mathbb{R}^{M \times N}$ with independent Gaussian components distributed in the interval (0, 1), and then normalize each column of T with the unit norm.

Now, consider an ellipsoid in \mathbb{R}^N as the feasible set \mathcal{C} . That is, $\mathcal{C} = \{x \in \mathbb{R}^N : (x - d)^T P(x - d) \leq r^2\}$, where P is a positive definite matrix, $d \in \mathbb{R}^N$ and r > 0. Then, the projection onto \mathcal{C} is difficult to compute (see, for example [119]). Consider the convex function $c : \mathbb{R}^N \to \mathbb{R}$ defined by $c(x) = \frac{1}{2} [(x - d)^T P(x - d) - r^2]$, $\forall x \in \mathbb{R}^N$. Then, \mathcal{C} is a level set of c. That is, $\mathcal{C} := \{x \in \mathbb{R}^N : c(x) \leq 0\}$. Also, it is easy to see that c is differentiable on \mathbb{R}^N . In fact, $c'(x) = P(x - d), \forall x \in \mathbb{R}^N$. Hence, c' is K_1 -Lipschitz continuous with $K_1 = ||P||$.

Now, let λ_{max} and λ_{min} be maximum and minimum eigenvalues of P, respectively. Suppose d = 0, since

$$\frac{||A(x)||}{||c'(x)||} = \frac{||M_1x||}{||Px||} \le \frac{||M_1||||x||}{\lambda_{\min}||x||} = \frac{||M_1||}{\lambda_{\min}}$$

holds for arbitrary $x \in \mathbb{R}^N$, $x \neq 0$, we have that $\rho_1 = \frac{||M_1||}{\lambda_{\min}}$ (see [119]). On the other hand, suppose $d \neq 0$, since

$$\frac{||A(x)||}{||c'(x)||} = \frac{||M_1x||}{||P(x-d)||} \le \frac{||M_1|| \left(||x-d||+||d||\right)}{\lambda_{\min}||x-d||} \le \frac{||M_1||\sqrt{\lambda_{\max}}\left(r+||d||\sqrt{\lambda_{\min}}\right)}{\lambda_{\min}^{\frac{3}{2}}r}$$

holds for all $x \in \partial C$, we have that $\rho_1 = \frac{||M_1||\sqrt{\lambda_{\max}}(r+||d||\sqrt{\lambda_{\min}})}{\lambda_{\min}^{\frac{3}{2}}r}$.

Let $\mathcal{Q} = \{y \in \mathbb{R}^m : q(y) \leq 0\}$, where $q(y) = y_1^2 - 2y_1 + \sum_{j=2}^m y_j^2$. Then, the explicit expression for projection onto \mathcal{Q} may be difficult to find (see for example, [120]). Also, we note that q is a countinuously differentiable convex function, and q' is K_2 -Lipschitz continuous with $K_2 = 2$. Moreover, $\rho_2 = ||M_2||$ (see [120]). Since the Lipschitz constants L_1, L_2, K_1, K_2 are known and we know ρ_1, ρ_2 , we can choose in Algorithm 4.2.2, $\mu = \frac{\sqrt{\rho_1^2 K_1^2 + 2\rho_1 K_1 (1-\phi) L_1 + L_1^2 - [\rho_1 K_1 + (1-\phi) L_1]}}{2\phi L_1^2}$ and $\lambda = \frac{\sqrt{\rho_2^2 K_2^2 + 2\rho_2 K_2 (1-\phi) L_2 + L_2^2 - [\rho_2 K_2 + (1-\phi) L_2]}}{2\phi L_2^2}$, and take the starting point $x_1 = (1, 1, \ldots, 1)^T$ while the entries of x_0 are randomly generated in [0, 1].

For Algorithm 1 of He et al. [121], we take

 $\mu_1 = 5 \left(||T^T \mathbb{H}T|| + L_1 \right) / v, \mu_2 = 10 \left(||T^T \mathbb{H}T|| + L_2 \right) / v, v = 0.8, \mathbb{H} = \frac{2}{||T^TT||} I_N, \gamma = 1.2$ and $\rho_1 = -1.5$ (which is the optimum choice in their implementation), with starting points $x_1 = (1, 1, \ldots, 1)^T$, $y_1 = (0, 0, \ldots, 0)^T$ and $\lambda_1 = (0, 0, \ldots, 0)^T$. Since $P_{\mathcal{C}}$ and $P_{\mathcal{Q}}$ seem not to have explicit expressions, Algorithm 1 of He *et al.* [121] may be difficult to implement for this example. Therefore, we use the technique proposed in this study, that is, we replace \mathcal{C} and \mathcal{Q} in their algorithm with \mathcal{C}_n and \mathcal{Q}_n , then use the formulas similar to (4.2.7) and (4.2.8) for the implementation. Furthermore, we consider different scenarios of



Figure 4.1: The behavior of TOL_n with $\varepsilon = 10^{-6}$ for Example 4.2.13: Top Left: (N, m) = (100, 50); Top Right: (N, m) = (300, 150); Bottom Left: (N, m) = (500, 250); Bottom Right: (N, m) = (1000, 500).

the problem's dimensions. That is, N = 100, 300, 500, 1000 and m = N/2. We also take $\varepsilon = 10^{-6}$ for the stopping criterion and obtain the numerical results reported in Table 4.2.1 and Figure 4.1. We stress that these choices are the same as in He *et al.* [121, Example 5.2].

Table 4.2.1. Numerical results for Example 4.2.13 with $\varepsilon = 10^{-6}$.

(N,m)		Alg.4.2.2	Alg.4.2.2	Alg.4.2.2	Alg.4.2.2	Alg.4.2.2	He <i>et al</i> .
		$\alpha = 3$	$\alpha = 6$	$\alpha = 9$	$\alpha = 12$	$\alpha = 15$	
	CPU	0.1917	0.0805	0.1205	0.0948	0.1337	10.5882
(100, 50)	Iter.	826	759	798	735	803	2482
	CPU	0.1378	0.1455	0.1609	0.1235	0.1217	15.6402
(300, 150)	Iter.	1211	1241	1238	1233	1251	8140
	CPU	0.2162	0.1814	0.1739	0.1953	0.1944	16.9445
(500, 250)	Iter.	1458	1480	1456	1468	1459	30043
	CPU	0.4489	0.4561	0.5391	0.4120	0.4624	39.0961
(1000, 500)	Iter.	1604	1622	1617	1620	1627	73092

Example 4.2.14. Let $C = \{x \in \mathbb{R}^2 : c(x) := x_1^2 - x_2 \leq 0\}$ and $Q = \{y \in \mathbb{R}^2 : q(y) := y_1^2 + y_2^2 - 2 \leq 0\}$, where $x = (x_1, x_2)$ and $y = (y_1, y_2)$. Define $A, F : \mathbb{R}^2 \to \mathbb{R}^2$ by

$$A(x) = (3\Psi(x_1), 2x_1 + x_2), \ F(x) = (6\Psi(x_1), 4x_1 + 2x_2),$$

where

$$\Psi(s) = \begin{cases} e(s-1) + e, & \text{if } s > 1, \\ e^s, & \text{if } -1 \le s \le 1, \\ e^{-1}(s+1) + e^{-1}, & \text{if } s < -1. \end{cases}$$

Then, one can easily verify that $L_1 = \sqrt{9e^2 + 5}$ and $L_2 = 2\sqrt{9e^2 + 5}$. Also, it is easy to see that c and q are continuously differentiable convex functions with Lipschitz constants $K_1 = K_2 = 2$. Moreover, we have that $\rho_1 = 3\sqrt{e^2 + 1}$ and $\rho_2 = \frac{3}{2}\sqrt{e^2 + 1}$ (see for example, [120]). Furthermore, we choose the same parameters as in Example 4.2.13 and use the stopping criterion $\varepsilon = 10^{-3}$.

By considering the following cases for the numerical experiments, we obtain the numerical results displayed in Table 4.2.2 and Figure 4.2.

Case 1: Take $x_1 = (1, 0.5)^T$ and $x_0 = (2, 3)^T$.

Case 2: Take $x_1 = (2, 1)^T$ and $x_0 = (0.5, -1)^T$.

Case 3: Take $x_1 = (1, -1)^T$ and $x_0 = (2, -2)^T$.

Case 4: Take $x_1 = (0.5, 4)^T$ and $x_0 = (-1, -3)^T$.

Table 4.2.2. Numerical results for Example 4.2.14 with $\varepsilon = 10^{-3}$.



Figure 4.2: The behavior of TOL_n with $\varepsilon = 10^{-3}$ for Example 4.2.14: Top Left: Case 1; Top Right: Case 2; Bottom Left: Case 3; Bottom Right: Case 4.

Cases		Alg.4.2.2	Alg.4.2.2	Alg.4.2.2	Alg.4.2.2	Alg.4.2.2	He <i>et al</i> .
		$\alpha = 3$	$\alpha = 6$	$\alpha = 9$	$\alpha = 12$	$\alpha = 15$	
1	CPU	0.0203	0.0160	0.0168	0.0261	0.0202	1.1282
	Iter.	29	30	28	28	29	73
2	CPU	0.0229	0.0162	0.0168	0.0161	0.0147	1.0324
	Iter.	36	38	35	38	37	86
3	CPU	0.0240	0.0166	0.0167	0.0160	0.0153	1.1263
	Iter.	37	35	37	36	34	89
4	CPU	0.0216	0.0181	0.0169	0.0187	0.0165	1.1308
	Iter.	52	52	47	53	46	75

We present the next example in an infinite dimensional Hilbert space. In this example, we compare Algorithm 4.2.3 with the methods of Reich and Tuyen [213] and Tian and Jiang [244], since their methods are given in infinite dimensional Hilbert spaces.

Example 4.2.15. Let $\mathcal{H}_1 = \mathcal{H}_2 = \mathcal{L}_2([0,1])$ be endowed with inner product

$$\langle x, y \rangle = \int_0^1 x(t)y(t)dt, \quad \forall \ x, y \in \mathcal{L}_2([0, 1])$$

and norm

$$||x|| := \left(\int_0^1 |x(t)|^2 dt\right)^{\frac{1}{2}}, \quad \forall \ x, y \in \mathcal{L}_2([0,1]).$$

Let φ be an arbitrary fixed element in $\mathcal{C}([0,1])$ (the space of all real-valued continuous functions on [0,1]) and $\mathcal{C} = \mathcal{Q} = \{x \in \mathcal{L}_2([0,1]) : \|\varphi x\| \leq 1\}$. In particular, for all $t \in [0,1]$, we define $\varphi(t) = e^{-t}$ for \mathcal{C} and $\varphi(t) = e^{-5t}$ for \mathcal{Q} . Then, \mathcal{C} and \mathcal{Q} are nonempty closed and convex subsets of $\mathcal{L}_2([0,1])$. However, we know from [119] that the explicit formulas for projections onto \mathcal{C} and \mathcal{Q} may be difficult to find.

Now, let $c, q : \mathcal{L}_2([0,1]) \to \mathbb{R}$ be defined by $c(x) = q(x) = \frac{1}{2} (\|\varphi x\|^2 - 1)$, $\forall x \in \mathcal{L}_2([0,1])$, where $\varphi(t) = e^{-t}$ for c and $\varphi(t) = e^{-5t}$ for q; $t \in [0,1]$. Then, c, q are convex functions with \mathcal{C} and \mathcal{Q} being level sets of c and q, respectively. Also, c and q are differentiable on $\mathcal{L}_2([0,1])$. In fact, $c'(x) = \varphi^2 x$, $\forall x \in \mathcal{L}_2([0,1])$. Furthermore, c' and q' are Lipschitz continuous.

Now, define $A : \mathcal{L}_2([0,1]) \to \mathcal{L}_2([0,1])$ by $Ax(t) = \max\{0, x(t)\}, \forall x \in \mathcal{L}_2([0,1])$. Then, A is monotone and Lipschitz continuous. Also, let $F : \mathcal{L}_2([0,1]) \to \mathcal{L}_2([0,1])$ be defined by $Fx(t) = \int_0^t x(s) ds, \forall x \in \mathcal{L}_2([0,1]), t \in [0,1]$. Then F is monotone and Lipschitz continuous.

Let $m\varphi = \min_{t \in [0,1]} |\varphi(t)| = e^{-1}$. Then,

$$\frac{\|Ax\|}{\|c'(x)\|} = \frac{\|x_+\|}{\|\varphi^2 x\|} \le \frac{\|x\|}{m^2 \varphi \|x\|} = e^2$$

holds for all $x \in \mathcal{L}_2([0,1])$, $x \neq 0$ (see [119]). Thus, we have that $\rho_1 = e^2$. Similarly, $\rho_2 = \frac{2}{\pi}e^{10}$.

We consider the following cases for the numerical experiments of this example. The results of the experiments are given in Table 4.2.3 and Figure 4.3. **Case 1:** Take $x_1(t) = 1 + t^2$, $x_0(t) = t + 5$ and $\phi_n = \frac{n}{2n+5}$.



Figure 4.3: The behavior of TOL_n with $\varepsilon = 10^{-5}$ for Example 4.2.15: Top Left: **Case 1**; Top Right: **Case 2**; Bottom Left: **Case 3**; Bottom Right: **Case 4**.

Case 2: Take $x_1(t) = \sin(t)$, $x_0(t) = t + 1$ and $\phi_n = \frac{n}{n+10}$. **Case 3:** Take $x_1(t) = t + 1$, $x_0(t) = t + t^3$ and $\phi_n = \frac{n}{n+10}$. **Case 4:** Take $x_1(t) = 0.7e^{-t} + 1$, $x_0(t) = t + t^3$ and $\phi_n = \frac{2n}{18n+1}$.

Table 4.2.3. Numerical results for Example 4.2.15 with $\varepsilon = 10^{-5}$.

Cases		Alg.4.2.3	Alg.4.2.3	Alg.4.2.3	Alg.4.2.3	Alg.4.2.3	Reich	Tian
		$\alpha = 3$	$\alpha = 6$	$\alpha = 9$	$\alpha = 12$	$\alpha = 15$	Tuyen	Jiang
1	CPU	2.9827	2.3552	2.3725	2.1437	2.3514	7.0796	12.3607
	Iter.	16	16	16	16	16	20	81
2	CPU	2.4189	1.9850	1.9034	1.8323	1.8094	5.4496	9.9440
	Iter.	17	17	17	17	17	20	83
3	CPU	2.2188	1.8285	1.8093	1.6663	1.6729	5.3220	9.9360
	Iter.	16	16	16	16	16	20	82
4	CPU	2.3943	2.0577	1.9658	1.7530	1.7740	5.6309	10.0070
	Iter.	16	16	16	16	16	20	80

Remark 4.2.16. In each example, using different starting points and varying both the inertial and relaxation stepsizes α and ϕ (ϕ_n), respectively, we obtain the numerical results displayed in Tables 4.2.1-4.2.3 and Figures 4.1-4.3. We compared our proposed Algorithm 4.2.2 with the method of He *et al.* [121, Algorithm 1] in Examples 4.2.13 and 4.2.14 while our proposed Algorithm 4.2.3 is compared with the methods of Reich and Tuyen [213, Theorem 4.4] and Tian and Jiang [244, Algorithm (4.8)] in Example 4.2.15. Furthermore, the following were observed from our numerical experiments:

- We have seen from the numerical examples that the projections onto feasible sets are generally difficult to compute. Thus, the benefits brought from the half-space technique (introduced in our methods) are further verified by our numerical experiments.
- In the numerical experiments, we choose the parameters $\eta \ge 0$, $\lambda_1 > 0$ and $\mu_1 > 0$ randomly and observed that the number of iteration does not change and no significant difference in the CPU time irrespective of the choices of the parameters.
- In each of the examples, we check the sensitivity of α as we vary ϕ (ϕ_n) for each starting points, in order to know if these choices affect the efficiency of our methods. We can see from the tables and graphs that the gap between the best results and the worst results of Algorithm 4.2.2 and Algorithm 4.2.3 is acceptably small, especially in Example 4.2.14 and Example 4.2.15. We can also see from Table 4.2.1 and Figure 4.1 that for solving problem (4.2.71), the efficiency of Algorithm 4.2.2 is slightly affected by the dimensionality of the problem. These means that the proposed relaxed inertial technique has efficient and stable performance for each group of random data.
- It can also be inferred from the tables and figures that in terms of both CPU time and number of iterations, our proposed Algorithm 4.2.2 outperforms the method of He *et al.* [121, Algorithm 1] in Example 4.2.13 and Example 4.2.14, while our proposed Algorithm 4.2.3 outperforms the methods of Reich and Tuyen [213, Theorem 4.4] and Tian and Jiang [244, Algorithm (4.8)] in Example 4.2.15. Hence, our methods are more efficient and more implementable than these other methods for solving the SVIP.

Appendix 4.2.17. Algorithm 1 of He et al. [121].

Stop 0. Given a symmetric positive definite matrix $\mathbb{H} \in \mathbb{R}^{m \times m}$, $\gamma \in (0,2)$ and $\rho \in (\rho_{\min}, 3)$, where $\rho_{\min} := \max \left\{-3, \frac{2(v-1)\mu_1}{4\zeta_{\max}(T^T HT)}\right\}$ and $T : \mathbb{R}^N \to \mathbb{R}^m$ is a linear operator (where T^T means the transpose of T). Set an initial point $\mathbf{u}_1 := (x_1, y_1, \lambda_1) \in \Omega := \mathcal{C} \times \mathcal{Q} \times \mathbb{R}^m$, where \mathcal{C} and \mathcal{Q} are nonempty closed and convex subsets of \mathbb{R}^N and \mathbb{R}^m , respectively.

Step 1. Generate a predictor $\tilde{\mathbf{u}}_n := (\tilde{x}_n, \tilde{y}_n, \tilde{\lambda}_n)$ with appropriate parameters μ_1 and μ_2 :

$$\begin{cases} \bar{\lambda}_n = \lambda_n - \mathbb{H}(Tx_n - y_n), \\ \tilde{x}_n = P_{\mathcal{C}} \left[x_n - \frac{1}{\mu_1} \left(A(x_n) - T^T \bar{\lambda}_n \right) \right], \\ \hat{\lambda}_n = \lambda_n - \mathbb{H}(T\hat{x}_n - y_n), \quad where \quad \hat{x}_n := \rho x_n + (1 - \rho)\tilde{x}_n, \\ \tilde{y}_n = P_{\mathcal{Q}} \left[y_n - \frac{1}{\mu_2} (F(y_n) + \hat{\lambda}_n) \right], \\ \tilde{\lambda}_n = \lambda_n - \mathbb{H}(T\tilde{x}_n - \tilde{y}_n). \end{cases}$$
(4.2.72)

Step 2. Update the next iterative $\boldsymbol{u}_{n+1} := (x_{n+1}, y_{n+1}, \lambda_{n+1})$ via

$$\boldsymbol{u}_{n+1} := \boldsymbol{u}_n - \gamma \alpha_k d(\mathbf{u}_n, \tilde{\mathbf{u}}_n),$$

where

$$\begin{cases} \alpha_k := \frac{\varphi(\mathbf{u}_n, \tilde{\mathbf{u}}_n)}{\|d(\mathbf{u}_n, \tilde{\mathbf{u}}_n)\|^2}, \\ d(\mathbf{u}_n, \tilde{\mathbf{u}}_n) := G(\mathbf{u}_n - \tilde{\mathbf{u}}_n) - \xi_n, \\ \varphi(\mathbf{u}_n, \tilde{\mathbf{u}}_n) := \left\langle \lambda_n - \tilde{\lambda}_n, \rho T(x_n - \tilde{x}_n) - (y_n - \tilde{y}_n) \right\rangle + \left\langle \mathbf{u}_n - \tilde{\mathbf{u}}_n, d(\mathbf{u}_n, \tilde{\mathbf{u}}_n) \right\rangle, \end{cases}$$

$$\xi_n := \begin{pmatrix} \xi_{n_x} \\ \xi_{n_y} \\ 0 \end{pmatrix} := \begin{pmatrix} A(x_n) - A(\tilde{x}_n) + T^T \mathbb{H}T(x_n - \tilde{x}_n) \\ F(y_n) - F(\tilde{y}_n) + \mathbb{H}(y_n - \tilde{y}_n) \\ 0 \end{pmatrix}, A \text{ and } F \text{ are monotone and}$$

Lipschitz continuous with constants L_1 and L_2 respectively, and

$$G := \begin{pmatrix} \mu_1 I_N + \rho T^T \mathbb{H}T & 0 & 0\\ 0 & \mu_2 I_m + \mathbb{H} & 0\\ 0 & 0 & \mathbb{H}^{-1} \end{pmatrix} \text{ is the block diagonal matrix, with identity}$$

matrices I_N and I_m of size N and m, respectively. The parameters μ_1 and μ_2 are chosen such that

 $\|\xi_{n_x}\| \le v\mu_1 \|x_n - \tilde{x}_n\|$ and $\|\xi_{n_y}\| \le v\mu_2 \|y_k - \tilde{y}_n\|,$

where $v \in (0, 1)$.

Appendix 4.2.18. The Algorithm in Reich and Tuyen [213, Theorem 4.4].

For any initial guess $x_1 = x \in \mathcal{H}_1$, define the sequence $\{x_n\}$ by

$$y_n = VI(\mathcal{C}, \ \lambda_n A + I_{\mathcal{H}_1} - x_n),$$

$$z_n = VI(\mathcal{Q}, \ \mu_n F + I_{\mathcal{H}_2} - Ty_n),$$

$$C_n = \{z \in \mathcal{H}_1 : ||y_n - z|| \le ||x_n - z||\},$$

$$D_n = \{z \in \mathcal{H}_1 : ||z_n - Tz|| \le ||Ty_n - Tz||\},$$

$$W_n = \{z \in \mathcal{H}_1 : \langle z - x_n, x_1 - x_n \rangle \le 0\},$$

$$x_{n+1} = P_{C_n \cap D_n \cap W_n}(x_1), \ n \ge 1,$$

where $I_{\mathcal{H}_1}$ and $I_{\mathcal{H}_2}$ are identity operators in \mathcal{H}_1 and \mathcal{H}_2 respectively, and $\{\lambda_n\}$ and $\{\mu_n\}$ are two given sequences of positive numbers satisfying the following condition:

$$\min\left\{\inf_{n}\{\lambda_{n}\}, \quad \inf_{n}\{\mu_{n}\}\right\} \ge r > 0.$$

Appendix 4.2.19. Algorithm (4.8) of Tian and Jiang [244].

Let $\{x_n\}, \{y_n\}, \{t_n\}$ and $\{w_n\}$ be sequences generated by $x_1 = x \in \mathcal{C}$ and

$$\begin{cases} y_n = P_{\mathcal{C}}(x_n - \eta_n T^*(I - P_{\mathcal{Q}}(I - \lambda F))Tx_n), \\ t_n = P_{\mathcal{C}}(y_n - \lambda_n Ay_n), \\ T_n = \{w \in \mathcal{H}_1 : \langle y_n - \lambda_n Ay_n - t_n, w - t_n \rangle \le 0\}, \\ w_n = P_{T_n}(y_n - \lambda_n At_n), \\ x_{n+1} = \alpha_n h(x_n) + (1 - \alpha_n)w_n, \quad n \ge 1, \end{cases}$$

where $T : \mathcal{H}_1 \to \mathcal{H}_2$ is a bounded linear operator, $A : \mathcal{C} \to \mathcal{H}_1$ is a monotone and *L*-Lipschitz continuous operator with L > 0, $F : \mathcal{H}_2 \to \mathcal{H}_2$ is a *k*-inverse strongly monotone mapping and $h : \mathcal{H}_1 \to \mathcal{H}_1$ is a σ -contractive mapping with $0 \le \sigma < 1$. The sequence $\{\eta_n\}$ is in [a, b], for some $a, b \in \left(0, \frac{1}{\|T\|^2}\right), \{\lambda_n\}$ is in [c, d], for some $c, d \in \left(0, \frac{1}{L}\right), \{\alpha_n\}$ is in (0, 1) with $\lim_{n \to \infty} \alpha_n = 0$ and $\sum_{n=1}^{\infty} \alpha_n = \infty$.

4.3 On split variational inequality problem beyond monotonicity

In solving the SVIP, very few methods have been considered in the literature and most of these few methods require the underlying operators to be co-coercive. This restrictive co-coercive assumption has been dispensed with in some methods; many of which require a product space formulation of the problem. However, it has been discovered that this product space formulation may cause some potential difficulties during implementation and its approach may not fully exploit the attractive splitting structure of the SVIP. In this study, we present two new methods with inertial steps for solving the SVIP in real Hilbert spaces without any product space formulation. We prove that the sequence generated by these methods converges strongly to a minimum-norm solution of the problem when the operators are pseudomonotone and Lipschitz continuous. Also, we provide several numerical experiments of the proposed methods in comparison with other related methods in the literature.

4.3.1 Proposed methods

In this section, we present our proposed methods and discuss their features. We begin with the following assumptions under which our strong convergence results are obtained.

Assumption 4.3.1. Suppose that the following conditions hold:

- (a) The feasible sets C and Q are nonempty closed and convex subsets of the real Hilbert spaces \mathcal{H}_1 and \mathcal{H}_2 , respectively.
- (b) $A: \mathcal{H}_1 \to \mathcal{H}_1$ and $F: \mathcal{H}_2 \to \mathcal{H}_2$ are pseudomonotone, sequentially weakly continuous and Lipschitz continuous with Lipschitz constants L_1 and L_2 , respectively.
- (c) $T : \mathcal{H}_1 \to \mathcal{H}_2$ is a bounded linear operator and the solution set $\Gamma := \{z \in VI(A, \mathcal{C}) : Tz \in VI(F, \mathcal{Q})\}$ is nonempty, where $VI(A, \mathcal{C})$ is the solution set of the classical VIP (1.2.4).
- (d) $\{\delta_n\}_{n=1}^{\infty}$ and $\{\tau_n\}_{n=1}^{\infty}$ are positive sequences satisfying the following conditions: $\delta_n \in (0,1), \quad \lim_{n \to \infty} \delta_n = 0, \quad \sum_{n=1}^{\infty} \delta_n = \infty \text{ and } \lim_{n \to \infty} \frac{\tau_n}{\delta_n} = 0.$
- (e) $\{\theta_n\} \subset (a, 1 \delta_n)$ for some a > 0.

We present the following method for solving the SVIP (1.2.4)-(1.2.5) when L_1 and L_2 are known.

Algorithm 4.3.2. Modified projection and contraction method with fixed stepsize.

Step 0: Choose sequences $\{\delta_n\}_{n=1}^{\infty}, \{\theta_n\}_{n=1}^{\infty}$ and $\{\tau_n\}_{n=1}^{\infty}$ such that the conditions from Assumption 4.3.1 (d)-(e) hold and let $\eta \geq 0, \gamma_i \in (0, 2), i = 1, 2, \mu \in (0, \frac{1}{L_1}), \lambda \in (0, \frac{1}{L_2}), \alpha \geq 3$ and $x_0, x_1 \in \mathcal{H}_1$ be given arbitrarily. Set n := 1.

Step 1: Given the iterates x_{n-1} and x_n $(n \ge 1)$, choose α_n such that $0 \le \alpha_n \le \overline{\alpha}_n$, where

$$\bar{\alpha}_n := \begin{cases} \min\left\{\frac{n-1}{n+\alpha-1}, \frac{\tau_n}{\|x_n - x_{n-1}\|}\right\}, & \text{if } x_n \neq x_{n-1} \\ \frac{n-1}{n+\alpha-1}, & \text{otherwise.} \end{cases}$$
(4.3.1)

Step 2: Compute

$$w_n = x_n + \alpha_n (x_n - x_{n-1}),$$

$$y_n = P_{\mathcal{Q}}(Tw_n - \lambda FTw_n),$$

$$z_n = Tw_n - \gamma_2 \beta_n r_n,$$

where $r_n := Tw_n - y_n - \lambda(FTw_n - Fy_n)$ and $\beta_n := \frac{\langle Tw_n - y_n, r_n \rangle}{\|r_n\|^2}$, if $r_n \neq 0$; otherwise, $\beta_n = 0$. Step 3: Compute

$$b_n = w_n + \eta_n T^*(z_n - Tw_n),$$

where the stepsize η_n is chosen such that for small enough $\epsilon > 0$,

$$\eta_n \in \left[\epsilon, \quad \frac{\|Tw_n - z_n\|^2}{\|T^*(Tw_n - z_n)\|^2} - \epsilon\right],$$

if $z_n \neq Tw_n$; otherwise, $\eta_n = \eta$. Step 4: Compute

$$u_n = P_{\mathcal{C}}(b_n - \mu A b_n),$$

$$t_n = b_n - \gamma_1 \gamma_n v_n,$$

where $v_n := b_n - u_n - \mu(Ab_n - Au_n)$ and $\gamma_n := \frac{\langle b_n - u_n, v_n \rangle}{\|v_n\|^2}$, if $v_n \neq 0$; otherwise, $\gamma_n = 0$. Step 5: Compute

$$x_{n+1} = (1 - \theta_n - \delta_n)b_n + \theta_n t_n.$$

Set n := n + 1 and go back to Step 1.

In the situation where L_1 and L_2 are not available, we present the following method with adaptive stepsize for solving the SVIP (1.2.4)-(1.2.5).

Algorithm 4.3.3. Modified projection and contraction method with adaptive stepsize strategy.

Step 0: Choose sequences $\{\delta_n\}_{n=1}^{\infty}, \{\theta_n\}_{n=1}^{\infty}$ and $\{\tau_n\}_{n=1}^{\infty}$ such that the conditions from Assumption 4.3.1 (d)-(e) hold and let $\eta \geq 0, \gamma_i \in (0,2), a_i \in (0,1), i = 1, 2, \lambda_1 > 0, \mu_1 > 0, \alpha \geq 3$ and $x_0, x_1 \in \mathcal{H}_1$ be given arbitrarily. Set n := 1.

Step 1: Given the iterates x_{n-1} and x_n for each $n \ge 1$, choose α_n such that $0 \le \alpha_n \le \overline{\alpha}_n$, where

$$\bar{\alpha}_{n} := \begin{cases} \min\left\{\frac{n-1}{n+\alpha-1}, \frac{\tau_{n}}{\|x_{n}-x_{n-1}\|}\right\}, & \text{if } x_{n} \neq x_{n-1} \\ \frac{n-1}{n+\alpha-1}, & \text{otherwise.} \end{cases}$$
(4.3.2)

Step 2: Compute

$$w_n = x_n + \alpha_n (x_n - x_{n-1}),$$

$$y_n = P_{\mathcal{Q}}(Tw_n - \lambda_n FTw_n),$$

$$z_n = Tw_n - \gamma_2 \beta_n r_n,$$

$$z_n = Tw_n - \gamma_2 \beta_n r_n,$$

where $r_n := Tw_n - y_n - \lambda_n (FTw_n - Fy_n), \beta_n := \frac{\langle Tw_n - y_n, r_n \rangle}{\|r_n\|^2}$, if $r_n \neq 0$; otherwise, $\beta_n = 0$; and

$$\lambda_{n+1} = \begin{cases} \min\left\{\frac{a_2||Tw_n - y_n||}{||FTw_n - Fy_n||}, \lambda_n\right\}, & \text{if } FTw_n \neq Fy_n\\ \lambda_n, & \text{otherwise.} \end{cases}$$
(4.3.3)

Step 3: Compute

$$b_n = w_n + \eta_n T^*(z_n - Tw_n),$$

where the stepsize η_n is chosen such that for small enough $\epsilon > 0$,

$$\eta_n \in \left[\epsilon, \quad \frac{\|Tw_n - z_n\|^2}{\|T^*(Tw_n - z_n)\|^2} - \epsilon\right],$$

if $z_n \neq Tw_n$; otherwise, $\eta_n = \eta$. Step 4: Compute

$$u_n = P_{\mathcal{C}}(b_n - \mu_n A b_n),$$

 $t_n = b_n - \gamma_1 \gamma_n v_n,$ where $v_n := b_n - u_n - \mu_n (Ab_n - Au_n), \ \gamma_n = \frac{\langle b_n - u_n, v_n \rangle}{\|v_n\|^2}$, if $v_n \neq 0$; otherwise, $\gamma_n = 0$; and

$$\mu_{n+1} = \begin{cases} \min\left\{\frac{a_1||b_n - u_n||}{||Au_n - Ab_n||}, \ \mu_n \right\}, & \text{if } Ab_n \neq Au_n \\ \mu_n, & \text{otherwise.} \end{cases}$$
(4.3.4)

Step 5: Compute

$$x_{n+1} = (1 - \theta_n - \delta_n)b_n + \theta_n t_n.$$

Set n := n + 1 and go back to Step 1.

We now highlight some of the features of our proposed methods.

Remark 4.3.1.

• Observe that Algorithm 4.3.2 and Algorithm 4.3.3 can be viewed as modified projection and contraction methods involving one projection onto \mathcal{C} per iteration for solving the classical VIP in \mathcal{H}_1 and another projection and contraction methods involving one projection onto \mathcal{Q} per iteration under a bounded linear operator T for solving another VIP in another space \mathcal{H}_2 , with no extra projections onto half-spaces or feasible sets unlike the method in [206] (see Appendix 4.3.16), where extra projections onto half-spaces are required. In fact, an interesting feature of the projection and contraction methods used here is that r_n of Step 2 in Algorithms 4.3.2 and 4.3.3 can be described as weighted average of $(Tw_n - y_n \sim \lambda FTw_n)$ and a hypothetical $(T\tilde{w}_n - \tilde{y}_n \sim \lambda FT\tilde{w}_n)$ in \mathcal{H}_2 , where $T\tilde{w}_n = Tw_n - \lambda FTw_n$ and $\tilde{y}_n = y_n - \lambda Fy_n$. We also have similar description for v_n of Step 4. This looks very similar to the Heun's method or modified Euler method from numerical methods for solving ordinary differential equations (see [230, page 328] for details). Furthermore, we can see that

$$\beta_n ||r_n||^2 = \langle Tw_n - y_n, r_n \rangle, \ \forall n \ge 1$$

$$(4.3.5)$$

holds for both $r_n = 0$ and $r_n \neq 0$. Similarly, we have that

$$\gamma_n ||v_n||^2 = \langle b_n - u_n, v_n \rangle, \ \forall n \ge 1$$

$$(4.3.6)$$

holds for both $v_n = 0$ and $v_n \neq 0$.

- Another notable advantage of Algorithm 4.3.2 and Algorithm 4.3.3 for solving the SVIP (1.2.4)-(1.2.5) is that the monotonicity assumption on the operators A and F usually used in many other works (see for example, [61, 63, 121, 131, 145, 190, 213, 243, 244]) to guarantee convergence, is dispensed with and no extra projections are required under this setting unlike in [206].
- The stepsizes $\{\lambda_n\}$ and $\{\mu_n\}$ given by (4.3.3) and (4.3.4), respectively are generated at each iteration by some simple computations. Thus, Algorithm 4.3.3 is easily implemented without the prior knowledge of the Lipschitz constants L_1 and L_2 .

- Step 1 of our methods is also easily implemented since the value of $||x_n x_{n-1}||$ is a priori known before choosing α_n . In our numerical analysis (Section 4.3.3), we shall check the sensitivity of α in order to find numerically, the optimum choice of α with respect to the convergence speed of our proposed methods.
- Step 5 of both algorithms guarantee the strong convergence to a minimum-norm solution of the problem.
- Unlike in [61, 63, 121], we can see that Algorithm 4.3.2 and Algorithm 4.3.3 do not require any product space formulation, thereby avoiding any potential difficulties that might be caused by the product space.

Remark 4.3.2. The choice of the stepsize η_n in Step 3 of Algorithms 4.3.2 and 4.3.3 do not require the prior knowledge of the operator norm ||T||. Furthermore, the value of η does not influence the algorithms, but it was introduced for the sake of clarity. We show in the following lemma that η_n is well-defined (see also [275, Remark 2.3, Lemma 2.3] and [274, Lemma 3.3]).

Lemma 4.3.3. The stepsize η_n given in Step 3 of Algorithms 4.3.2 and 4.3.3 is welldefined.

Proof. Let $z \in \Gamma$. That is, $z \in VI(A, \mathcal{C})$ and $Tz \in VI(F, \mathcal{Q})$. Then, by Cauchy-Schwarz inequality and Lemma 2.1.1, we obtain

$$\|T^{*}(Tw_{n} - z_{n})\|\|w_{n} - z\| \geq \langle T^{*}(Tw_{n} - z_{n}), w_{n} - z \rangle$$

= $\langle Tw_{n} - z_{n}, Tw_{n} - Tz \rangle$
= $\frac{1}{2} \Big[\|Tw_{n} - z_{n}\|^{2} + \|Tw_{n} - Tz\|^{2} - \|z_{n} - Tz\|^{2} \Big].$ (4.3.7)

Since $z_n = Tw_n - \gamma_2 \beta_n r_n$, we obtain that

$$\begin{aligned} \|z_n - Tz\|^2 &= \|Tw_n - Tz\|^2 + \gamma_2^2 \beta_n^2 \|r_n\|^2 - 2\gamma_2 \beta_n \langle Tw_n - Tz, r_n \rangle \\ &= \|Tw_n - Tz\|^2 + \gamma_2^2 \beta_n^2 \|r_n\|^2 - 2\gamma_2 \beta_n \langle Tw_n - y_n, r_n \rangle - 2\gamma_2 \beta_n \langle y_n - Tz, r_n \rangle. \end{aligned}$$

$$(4.3.8)$$

Since $y_n = P_Q(Tw_n - \lambda FTw_n)$ and $Tz \in Q$, we obtain from the characteristics property of P_Q that

 $\langle Tw_n - \lambda FTw_n - y_n, y_n - Tz \rangle \ge 0. \tag{4.3.9}$

Also, since $Tz \in VI(F, \mathcal{Q})$ and $y_n \in \mathcal{Q}$, we have that

$$\langle FTz, y_n - Tz \rangle \ge 0$$
 (see Inequality (1.2.1)),

which by the pseudomonotonicity of F and $\lambda > 0$, implies

$$\langle \lambda F y_n, y_n - T z \rangle \ge 0. \tag{4.3.10}$$

Adding (4.3.9) and (4.3.10), we obtain

$$\langle Tw_n - y_n - \lambda (FTw_n - Fy_n), y_n - Tz \rangle \ge 0.$$

That is

 $\langle r_n, y_n - Tz \rangle \ge 0$ (which is still true if λ is replaced with λ_n as in Algorithm 4.3.3). (4.3.11)

On the other hand, we have from the Lipschitz continuity of F and $\lambda \in (0, \frac{1}{L_2})$, that

$$\langle Tw_n - y_n, r_n \rangle = \langle Tw_n - y_n, Tw_n - y_n - \lambda (FTw_n - Fy_n) \rangle$$

= $||Tw_n - y_n||^2 - \langle Tw_n - y_n, \lambda (FTw_n - Fy_n) \rangle$
 $\geq ||Tw_n - y_n||^2 - \lambda ||Tw_n - y_n|| ||FTw_n - Fy_n||$
 $\geq (1 - \lambda L_2) ||Tw_n - y_n||^2 \geq 0,$ (4.3.12)

which by the definition of β_n , implies that $\beta_n \ge 0$, $\forall n \ge 1$ (this is still true even if we replace λ with λ_n as in Algorithm 4.3.3). Thus, using (4.3.11) and (4.3.5) in (4.3.8), and nothing that $\gamma_2 \in (0, 2)$, we obtain

$$\begin{aligned} \|z_n - Tz\|^2 &\leq \|Tw_n - Tz\|^2 + \gamma_2^2 \beta_n^2 \|r_n\|^2 - 2\gamma_2 \beta_n \langle Tw_n - y_n, r_n \rangle \\ &= \|Tw_n - Tz\|^2 + \gamma_2^2 \beta_n^2 \|r_n\|^2 - 2\gamma_2 \beta_n \cdot \beta_n \|r_n\|^2 \\ &= \|Tw_n - Tz\|^2 - \gamma_2 (2 - \gamma_2) \beta_n^2 \|r_n\|^2 \\ &\leq \|Tw_n - Tz\|^2. \end{aligned}$$

$$(4.3.13)$$

Substituting (4.3.13) into (4.3.7), we obtain that

$$||T^*(Tw_n - z_n)|| ||w_n - z|| \ge \frac{1}{2} ||Tw_n - z_n||^2.$$
(4.3.14)

Now, for $z_n \neq Tw_n$, we have that $||Tw_n - z_n|| > 0$. This together with (4.3.14), imply that $||T^*(Tw_n - z_n)|| ||w_n - z|| > 0$. Hence, we have that $||T^*(Tw_n - z_n)|| \neq 0$. Therefore, η_n is well-defined.

We also make the following observation regarding η_n . Note from Step 3 of Algorithms 4.3.2 and 4.3.3, that

$$\eta_n \|T^*(Tw_n - z_n)\|^2 \le \|Tw_n - z_n\|^2 - \epsilon \|T^*(Tw_n - z_n)\|^2,$$
(4.3.15)

which implies that

$$\eta_n^2 \|T^*(Tw_n - z_n)\|^2 - \eta_n \|Tw_n - z_n\|^2 \le -\epsilon \eta_n \|T^*(Tw_n - z_n)\|^2.$$
(4.3.16)

Note also that, we can replace the choice of η_n in Step 3 of Algorithm 4.3.2 and Algorithm 4.3.3 with the following: For small enough $\epsilon > 0$,

$$\eta_n \in \left[\epsilon, \quad \frac{\|Tw_n - z_n\|^2}{1 + \|T^*(Tw_n - z_n)\|^2} - \epsilon\right].$$
(4.3.17)

Then, we can see that

$$\eta_n \le \frac{\|Tw_n - z_n\|^2 - \epsilon \left(1 + \|T^*(Tw_n - z_n)\|^2\right)}{1 + \|T^*(Tw_n - z_n)\|^2},$$

which implies that

$$\eta_n + \eta_n \|T^*(Tw_n - z_n)\|^2 \le \|Tw_n - z_n\|^2 - \epsilon - \epsilon \|T^*(Tw_n - z_n)\|^2.$$

This further gives (4.3.15) and consequently (4.3.16).

As we shall see in our convergence analysis, (4.3.15) and (4.3.16) play crucial role in our proofs. Therefore, one can either choose the stepsize in Step 3 or that in (4.3.17), to ensure the convergence of Algorithms 4.3.2 and 4.3.3.

4.3.2 Convergence analysis

Lemma 4.3.4. Let $\{x_n\}$ be a sequence generated by Algorithm 4.3.2 under Assumption 4.3.1. Then, $\{x_n\}$ is bounded.

Proof. Let $p \in \Gamma$. Then $Tp \in VI(F, \mathcal{Q}) \subset \mathcal{Q}$. From the definition of w_n , we have

$$||w_n - p|| = ||x_n + \alpha_n(x_n - x_{n-1}) - p||$$

$$\leq ||x_n - p|| + \alpha_n ||x_n - x_{n-1}||$$

Also, from Step 1, we observe that $\alpha_n ||x_n - x_{n-1}|| \le \tau_n$, $\forall n \ge 1$, which implies that

$$\frac{\alpha_n}{\delta_n} \|x_n - x_{n-1}\| \le \frac{\tau_n}{\delta_n} \to 0, \quad \text{as} \quad n \to \infty.$$
(4.3.18)

Hence, there exists $M_1 > 0$ such that

$$\frac{\alpha_n}{\delta_n} \|x_n - x_{n-1}\| \le M_1, \quad \forall n \ge 1.$$

$$(4.3.19)$$

This implies that

$$||w_n - p|| \le ||x_n - p|| + \delta_n M_1, \ \forall n \ge 1.$$
(4.3.20)

Since $y_n = P_Q(Tw_n - \lambda FTw_n)$, we obtain by the characteristic property of P_Q that

$$\langle y_n - Tw_n + \lambda FTw_n, y_n - Tp \rangle \le 0. \tag{4.3.21}$$

Again, since $Tp \in VI(F, \mathcal{Q})$ and $y_n \in \mathcal{Q}$, we obtain that

$$\langle FTp, y_n - Tp \rangle \ge 0,$$

which follows from the pseudomonotonicity of F that $\langle Fy_n, y_n - Tp \rangle \ge 0$. Since $\lambda > 0$, we get

$$\langle \lambda F y_n, y_n - T p \rangle \ge 0. \tag{4.3.22}$$

Combining (4.3.21) and (4.3.22), we have

$$\langle Tw_n - y_n - \lambda (FTw_n - Fy_n), y_n - Tp \rangle \ge 0.$$
(4.3.23)

From Step 2 and (4.3.23), we obtain

$$\begin{aligned} \langle Tw_n - Tp, r_n \rangle &= \langle Tw_n - y_n, r_n \rangle + \langle y_n - Tp, r_n \rangle \\ &= \langle Tw_n - y_n, r_n \rangle + \langle y_n - Tp, Tw_n - y_n - \lambda (FTw_n - Fy_n) \rangle \\ &\geq \langle Tw_n - y_n, r_n \rangle, \end{aligned}$$

which implies that

$$-\langle Tw_n - Tp, r_n \rangle \le -\langle Tw_n - y_n, r_n \rangle.$$
(4.3.24)

Since $z_n = Tw_n - \gamma_2 \beta_n r_n$, we have that $||\beta_n \cdot r_n||^2 = \gamma_2^{-2} ||z_n - Tw_n||^2$. Thus, from Lemma 2.1.1, (4.3.5) and (4.3.24) (nothing that β_n and γ_2 are nonnegative), we obtain

$$||z_{n} - Tp||^{2} = ||Tw_{n} - \gamma_{2}\beta_{n}r_{n} - Tp||^{2}$$

$$= ||Tw_{n} - Tp||^{2} + \gamma_{2}^{2}\beta_{n}^{2}||r_{n}||^{2} - 2\gamma_{2}\beta_{n}\langle Tw_{n} - Tp, r_{n}\rangle$$

$$\leq ||Tw_{n} - Tp||^{2} + \gamma_{2}^{2}\beta_{n}^{2}||r_{n}||^{2} - 2\gamma_{2}\beta_{n}\langle Tw_{n} - y_{n}, r_{n}\rangle$$

$$= ||Tw_{n} - Tp||^{2} + \gamma_{2}^{2}\beta_{n}^{2}||r_{n}||^{2} - 2\gamma_{2}\beta_{n} \cdot \beta_{n}||r_{n}||^{2}$$

$$= ||Tw_{n} - Tp||^{2} - \gamma_{2}(2 - \gamma_{2})||\beta_{n} \cdot r_{n}||^{2}$$

$$= ||Tw_{n} - Tp||^{2} - \gamma_{2}^{-1}(2 - \gamma_{2})||z_{n} - Tw_{n}||^{2}.$$
(4.3.25)

Also, we obtain from Step 3, Lemma 2.1.1 and (4.3.25) that

$$\begin{aligned} \|b_{n} - p\|^{2} &= \|w_{n} - p\|^{2} + \eta_{n}^{2} \|T^{*}(z_{n} - Tw_{n})\|^{2} + 2\eta_{n} \langle w_{n} - p, T^{*}(z_{n} - Tw_{n}) \rangle \\ &= \|w_{n} - p\|^{2} + \eta_{n}^{2} \|T^{*}(z_{n} - Tw_{n})\|^{2} + 2\eta_{n} \langle Tw_{n} - Tp, z_{n} - Tw_{n} \rangle \\ &= \|w_{n} - p\|^{2} + \eta_{n}^{2} \|T^{*}(z_{n} - Tw_{n})\|^{2} \\ &+ \eta_{n} \left[\|z_{n} - Tp\|^{2} - \|Tw_{n} - Tp\|^{2} - \|z_{n} - Tw_{n}\|^{2} \right] \\ &\leq \|w_{n} - p\|^{2} + \eta_{n}^{2} \|T^{*}(z_{n} - Tw_{n})\|^{2} - \eta_{n} \|z_{n} - Tw_{n}\|^{2} \\ &= \|w_{n} - p\|^{2} - \eta_{n} \left[\|z_{n} - Tw_{n}\|^{2} - \eta_{n} \|T^{*}(z_{n} - Tw_{n})\|^{2} \right]. \end{aligned}$$

$$(4.3.26)$$

Thus, by (4.3.15), we obtain that

$$||b_n - p||^2 \le ||w_n - p||^2$$
,

which implies from (4.3.20) that

$$||b_n - p|| \le ||w_n - p|| \le ||x_n - p|| + \delta_n M_1, \ \forall n \ge 1.$$
(4.3.27)

Following similar argument used in obtaining (4.3.25), we get

$$\|t_n - p\|^2 = \|b_n - \gamma_1 \gamma_n v_n - p\|^2$$

$$\leq \|b_n - p\|^2 - \gamma_1^{-1} (2 - \gamma_1) \|t_n - b_n\|^2.$$
(4.3.28)

Also, by Step 5, we get

$$||x_{n+1} - p|| = ||(1 - \theta_n - \delta_n)(b_n - p) + \theta_n(t_n - p) - \delta_n p||$$

$$\leq ||(1 - \theta_n - \delta_n))(b_n - p) + \theta_n(t_n - p)|| + \delta_n ||p||.$$
(4.3.29)

But from (4.3.27) and (4.3.28), we obtain that

$$\|(1 - \theta_n - \delta_n)(b_n - p) + \theta_n(t_n - p)\|^2 = (1 - \theta_n - \delta_n)^2 \|b_n - p\|^2 + \theta_n^2 \|t_n - p\|^2 + 2(1 - \theta_n - \delta_n)\theta_n \langle b_n - p, t_n - p \rangle \leq (1 - \theta_n - \delta_n)^2 \|w_n - p\|^2 + \theta_n^2 \|b_n - p\|^2 + 2(1 - \theta_n - \delta_n)\theta_n \|b_n - p\|^2 \leq (1 - \delta_n)^2 \|w_n - p\|^2.$$
(4.3.30)

Hence, (4.3.29) becomes

$$\begin{aligned} \|x_{n+1} - p\| &\leq (1 - \delta_n) \|w_n - p\| + \delta_n \|p\| \\ &\leq (1 - \delta_n) \left[\|x_n - p\| + \delta_n M_1 \right] + \delta_n \|p\| \\ &\leq (1 - \delta_n) \|x_n - p\| + \delta_n (M_1 + \|p\|), \end{aligned}$$

which implies from Lemma 2.5.35 that $\{x_n\}$ is bounded.

Lemma 4.3.5. Let $\{x_n\}$ be a sequence generated by Algorithm 4.3.2 under Assumption 4.3.1. Suppose that, there exists a subsequence $\{x_{n_k}\}$ of $\{x_n\}$ which converges weakly to a point $z \in \mathcal{H}_1$ and $\lim_{k \to \infty} ||b_{n_k} - w_{n_k}|| = 0 = \lim_{k \to \infty} ||b_{n_k} - t_{n_k}||$, then $z \in \Gamma$.

Proof. Without loss of generality, we may assume that $z_n \neq Tw_n$. Then,

$$\eta_n \in \left[\epsilon, \quad \frac{\|z_n - Tw_n\|^2}{\|T^*(z_n - Tw_n)\|^2} - \epsilon\right].$$

Thus, putting (4.3.16) into (4.3.26), we obtain that

$$\begin{aligned} \|b_{n_k} - p\|^2 &\leq \|w_{n_k} - p\|^2 - \eta_{n_k} \epsilon \|T^*(z_{n_k} - Tw_{n_k})\|^2 \\ &\leq \|w_{n_k} - p\|^2 - \epsilon^2 \|T^*(z_{n_k} - Tw_{n_k})\|^2, \end{aligned}$$
(4.3.31)

which implies that

$$\epsilon^{2} \|T^{*}(z_{n_{k}} - Tw_{n_{k}})\|^{2} \leq \|w_{n_{k}} - p\|^{2} - \|b_{n_{k}} - p\|^{2} \\ \leq \|w_{n_{k}} - b_{n_{k}}\|^{2} + 2\|w_{n_{k}} - b_{n_{k}}\|\|b_{n_{k}} - p\|.$$

Thus, by our hypothesis, we obtain

$$\lim_{k \to \infty} \|T^*(z_{n_k} - Tw_{n_k})\| = 0.$$
(4.3.32)

From (4.3.26) and (4.3.32), we obtain

$$\begin{aligned} \eta_{n_k} \|z_{n_k} - Tw_{n_k}\|^2 &\leq \|w_{n_k} - p\|^2 - \|b_{n_k} - p\|^2 + \eta_{n_k}^2 \|T^*(z_{n_k} - Tw_{n_k})\|^2 \\ &\leq \|w_{n_k} - b_{n_k}\|^2 + 2\|w_{n_k} - b_{n_k}\|\|b_{n_k} - p\| \\ &+ \eta_{n_k}^2 \|T^*(z_{n_k} - Tw_{n_k})\|^2 \to 0, \quad \text{as} \quad k \to \infty. \end{aligned}$$

Since $\eta_{n_k} > \epsilon > 0$, we obtain that

$$\lim_{k \to \infty} \|z_{n_k} - Tw_{n_k}\| = 0.$$
(4.3.33)

From (4.3.12), we obtain

$$\langle Tw_{n_k} - y_{n_k}, r_{n_k} \rangle \ge (1 - \lambda L_2) \|Tw_{n_k} - y_{n_k}\|^2.$$
 (4.3.34)

Since $\lambda \in (0, \frac{1}{L_2})$, we have that $1 - \lambda L_2 > 0$. Thus, we obtain from (4.3.34) and (4.3.5) that

$$\begin{split} \|Tw_{n_{k}} - y_{n_{k}}\|^{2} &\leq \left(\frac{1}{1 - \lambda L_{2}}\right) \langle Tw_{n_{k}} - y_{n_{k}}, r_{n_{k}} \rangle \\ &= \left(\frac{1}{1 - \lambda L_{2}}\right) \beta_{n_{k}} \|r_{n_{k}}\|^{2} \\ &= \left(\frac{1}{1 - \lambda L_{2}}\right) \beta_{n_{k}} \|r_{n_{k}}\| \|Tw_{n_{k}} - y_{n_{k}} - \lambda(FTw_{n_{k}} - Fy_{n_{k}})\| \\ &\leq \left(\frac{1}{1 - \lambda L_{2}}\right) \beta_{n_{k}} \|r_{n_{k}}\| \left(\|Tw_{n_{k}} - y_{n_{k}}\| + \lambda\|FTw_{n_{k}} - Fy_{n_{k}}\|\right) \\ &\leq \left(\frac{1}{1 - \lambda L_{2}}\right) \beta_{n_{k}} \|r_{n_{k}}\| \left(\|Tw_{n_{k}} - y_{n_{k}}\| + \lambda L_{2} \|Tw_{n_{k}} - y_{n_{k}}\|\right) \\ &= \left(\frac{1 + \lambda L_{2}}{1 - \lambda L_{2}}\right) \|Tw_{n_{k}} - y_{n_{k}}\|\beta_{n_{k}}\|r_{n_{k}}\| \\ &= \gamma_{2}^{-1} \left(\frac{1 + \lambda L_{2}}{1 - \lambda L_{2}}\right) \|Tw_{n_{k}} - y_{n_{k}}\| \|z_{n_{k}} - Tw_{n_{k}}\|, \end{split}$$

where the last equality follows from $z_{n_k} = Tw_{n_k} - \gamma_2 \beta_{n_k} r_{n_k}$, that is, $\beta_{n_k} ||r_{n_k}|| = \gamma_2^{-1} ||z_{n_k} - Tw_{n_k}||$.

Therefore, we obtain from (4.3.33) that

$$\|Tw_{n_k} - y_{n_k}\| \le \gamma_2^{-1} \left(\frac{1 + \lambda L_2}{1 - \lambda L_2}\right) \|Tw_{n_k} - z_{n_k}\| \to 0, \quad \text{as} \quad k \to \infty.$$
(4.3.35)

By the characteristic property of $P_{\mathcal{Q}}$, we obtain for all $x \in \mathcal{Q}$ that

$$\langle Tw_{n_k} - \lambda FTw_{n_k} - y_{n_k}, x - y_{n_k} \rangle \le 0.$$

This implies that

$$\frac{1}{\lambda}\langle Tw_{n_k} - y_{n_k}, x - y_{n_k} \rangle + \langle FTw_{n_k}, y_{n_k} - Tw_{n_k} \rangle \le \langle FTw_{n_k}, x - Tw_{n_k} \rangle.$$
(4.3.36)

Thus, we obtain from (4.3.35) that

$$0 \le \liminf_{k \to \infty} \langle FTw_{n_k}, x - Tw_{n_k} \rangle, \ \forall x \in \mathcal{Q}.$$
(4.3.37)

Now, note that

$$\langle Fy_{n_k}, x - y_{n_k} \rangle = \langle Fy_{n_k} - FTw_{n_k}, x - Tw_{n_k} \rangle + \langle FTw_{n_k}, x - Tw_{n_k} \rangle + \langle Fy_{n_k}, Tw_{n_k} - y_{n_k} \rangle.$$

$$(4.3.38)$$

Moreover, since F is Lipschitz continuous on \mathcal{H}_2 , we obtain from (4.3.35) that

$$\lim_{k \to \infty} ||FTw_{n_k} - Fy_{n_k}|| = 0.$$

Hence, we obtain from (4.3.35), (4.3.37) and (4.3.38) that

$$0 \le \liminf_{k \to \infty} \langle F y_{n_k}, x - y_{n_k} \rangle, \ \forall x \in \mathcal{Q}.$$
(4.3.39)

We now show that $Tz \in VI(F, \mathcal{Q})$. Let $\{c_k\}$ be a sequence of positive numbers such that $c_{k+1} \leq c_k$, $\forall k \geq 1$ and $c_k \to 0$ as $k \to \infty$. For each $k \geq 1$, denote by N_k , the smallest positive integer such that

$$\langle Fy_{n_j}, x - y_{n_j} \rangle + c_k \ge 0, \quad \forall j \ge N_k, \tag{4.3.40}$$

where the existence of N_k follows from (4.3.39). Note that $\{N_k\}$ is increasing since $\{c_k\}$ is decreasing. Also, since $\{y_{N_k}\} \subset \mathcal{Q}$ for all $k \geq 1$, we have that $Fy_{N_k} \neq 0$. Hence, we can set $q_{N_k} = \frac{Fy_{N_k}}{\|Fy_{N_k}\|^2}$, for each $k \geq 1$. Then, $\langle Fy_{N_k}, q_{N_k} \rangle = 1$, for each $k \geq 1$. Thus by (4.3.40), we have $\langle Fy_{N_k}, x + c_k q_{N_k} - y_{N_k} \rangle \geq 0$, which by the pseudomonotonicity of F on \mathcal{H}_2 , yields

$$\langle F(x + c_k q_{N_k}), x + c_k q_{N_k} - y_{N_k} \rangle \ge 0.$$
 (4.3.41)

This further implies that

$$\langle Fx, x - y_{N_k} \rangle \ge \langle Fx - F(x + c_k q_{N_k}), x + c_k q_{N_k} - y_{N_k} \rangle - c_k \langle Fx, q_{N_k} \rangle.$$
(4.3.42)

Now, from Step 2 and (4.3.18), we get that

$$\lim_{n \to \infty} \|w_n - x_n\| = \lim_{n \to \infty} \alpha_n \|x_n - x_{n-1}\| = \lim_{n \to \infty} \delta_n \cdot \frac{\alpha_n}{\delta_n} \|x_n - x_{n-1}\| = 0.$$
(4.3.43)

Since the subsequence $\{x_{n_k}\}$ of $\{x_n\}$ is weakly convergent to a point $z \in \mathcal{H}_1$, it follows that the subsequence $\{w_{n_k}\}$ of $\{w_n\}$ is also weakly convergent to $z \in \mathcal{H}_1$. Again, since T is a bounded linear operator, we obtain that $\{Tw_{n_k}\}$ converges weakly to Tz. Hence, by (4.3.35), we have that $\{y_{n_k}\}$ also converges weakly to Tz. Note that $Tz \in \mathcal{Q}$ since $\{y_{n_k}\} \subset \mathcal{Q}$. Thus, by the sequentially weakly continuity of F on \mathcal{Q} , we obtain that $\{Fy_{n_k}\}$ converges weakly to FTz. Now, if FTz = 0, then $Tz \in VI(F, \mathcal{Q})$. So, let $FTz \neq 0$. Then, by the weakly lower semicontinuity of $\|\cdot\|$, we obtain that

$$0 < \|FTz\| \le \liminf_{k \to \infty} \|Fy_{N_k}\|.$$

Since $\{y_{n_k}\} \subset \{y_{N_k}\}$, we obtain that

$$0 \le \limsup_{k \to \infty} \|c_k q_{N_k}\| = \limsup_{k \to \infty} \left(\frac{c_k}{\|Fy_{n_k}\|}\right) \le \frac{\limsup_{k \to \infty} c_k}{\liminf_{k \to \infty} \|Fy_{n_k}\|} \le \frac{0}{\|FTz\|} = 0.$$

Therefore, $\lim_{k\to\infty} c_k q_{N_k} = 0$. Thus, letting $k \to \infty$ in (4.3.42), we have

$$\langle Fx, x - Tz \rangle \ge 0, \quad \forall x \in \mathcal{Q},$$
 (4.3.44)
which implies by Lemma 2.5.8 that $Tz \in VI(F, \mathcal{Q})$.

Next, we show that $z \in VI(A, C)$. Now, observe that, if we follow similar argument used in getting (4.3.35), and noting (by our hypothesis) that $\lim_{k\to\infty} ||b_{n_k} - t_{n_k}|| = 0$, we can also get

$$\|b_{n_k} - u_{n_k}\| \le r_1^{-1} \left(\frac{1 + \mu L_1}{1 + \mu L_1}\right) \|b_{n_k} - t_{n_k}\| \to 0, \quad \text{as} \quad k \to \infty.$$
(4.3.45)

Thus, using (4.3.45), the characteristics property of $P_{\mathcal{C}}$ and the Lipschitz continuity of A on \mathcal{H}_1 , we obtain (by following similar argument used in getting (4.3.39)) that

$$0 \le \liminf_{k \to \infty} \langle Au_{n_k}, y - u_{n_k} \rangle, \ \forall y \in \mathcal{C}.$$
(4.3.46)

Now, since $\{w_{n_k}\}$ converges weakly to z, we obtain by our hypothesis and (4.3.45) that the subsequences $\{b_{n_k}\}$ and $\{u_{n_k}\}$ of $\{b_n\}$ and $\{u_n\}$ respectively, converge weakly to z. Again, since $\{u_{n_k}\} \subset C$, we have that $z \in C$. Thus, by the sequential weakly continuity of A on C, we obtain that $\{Au_{n_k}\}$ converges weakly to Az.

Thus, we can follow similar proof used in obtaining (4.3.44), to get

$$\langle Ay, y - z \rangle \ge 0, \quad \forall \ y \in \mathcal{C},$$
 (4.3.47)

which by Lemma 2.5.8, implies $z \in VI(A, C)$. Hence, we conclude that $z \in \Gamma$.

We are now in position to give the main theorem for Algorithm 4.3.2.

Theorem 4.3.6. Let $\{x_n\}$ be a sequence generated by Algorithm 4.3.2 under Assumption 4.3.1. Then, $\{x_n\}$ converges strongly to $p \in \Gamma$, where $||p|| = \min\{||z|| : z \in \Gamma\}$.

Proof. Let $p \in \Gamma$. Then, from Step 2 and (4.3.19), we obtain

$$\begin{aligned} \|w_{n} - p\|^{2} &= \|x_{n} - p\|^{2} + \alpha_{n}^{2} \|x_{n} - x_{n-1}\|^{2} + 2\alpha_{n} \langle x_{n} - p, x_{n} - x_{n-1} \rangle \\ &\leq \|x_{n} - p\|^{2} + \alpha_{n}^{2} \|x_{n} - x_{n-1}\|^{2} + 2\alpha_{n} \|x_{n} - p\| \|x_{n} - x_{n-1}\| \\ &= \|x_{n} - p\|^{2} + \alpha_{n} \|x_{n} - x_{n-1}\| \left[\delta_{n} \cdot \frac{\alpha_{n}}{\delta_{n}} \|x_{n} - x_{n-1}\| + 2\|x_{n} - p\| \right] \\ &\leq \|x_{n} - p\|^{2} + \alpha_{n} \|x_{n} - x_{n-1}\| \left[\delta_{n} M_{1} + 2\|x_{n} - p\| \right]. \end{aligned}$$
(4.3.48)

Since by Lemma 4.3.4, $\{x_n\}$ is bounded, then, there exists $M_2 > 0$ such that

 $(\delta_n M_1 + ||x_n - p||) \le M_2, \ \forall n \ge 1.$

Hence, we obtain from (4.3.48) that

$$||w_n - p||^2 \leq ||x_n - p||^2 + 3\alpha_n ||x_n - x_{n-1}|| M_2.$$
(4.3.49)

Thus, from (4.3.27) and (4.3.28), we get

$$\begin{aligned} \|(1-\theta_n)b_n + \theta_n t_n - p\|^2 &= \|(1-\theta_n)(b_n - p) + \theta_n(t_n - p)\|^2 \\ &= (1-\theta_n)^2 \|b_n - p\|^2 + \theta_n^2 \|t_n - p\|^2 \\ &+ 2(1-\theta_n)\theta_n \langle b_n - p, t_n - p \rangle \\ &\leq (1-\theta_n)^2 \|w_n - p\|^2 + \theta_n^2 \|w_n - p\|^2 \\ &+ 2(1-\theta_n)\theta_n \langle b_n - p, t_n - p \rangle \\ &\leq (1-\theta_n)^2 \|w_n - p\|^2 + \theta_n^2 \|w_n - p\|^2 + 2(1-\theta_n)\theta_n \|w_n - p\|^2 \\ &\leq \|x_n - p\|^2 + 3\alpha_n \|x_n - x_{n-1}\| M_2. \end{aligned}$$

From Step 5, the previous inequality and (4.3.18), we get

$$\begin{aligned} \|x_{n+1} - p\|^2 &= \|(1 - \delta_n) \left[(1 - \theta_n) b_n + \theta_n t_n - p \right] - \left[\delta_n \theta_n (b_n - t_n) + \delta_n p \right] \|^2 \\ &\leq (1 - \delta_n)^2 \| (1 - \theta_n) b_n + \theta_n t_n - p \|^2 - 2 \langle \delta_n \theta_n (b_n - t_n) + \delta_n p, x_{n+1} - p \rangle \\ &\leq (1 - \delta_n) \| (1 - \theta_n) b_n + \theta_n t_n - p \|^2 + 2 \delta_n \theta_n \langle b_n - t_n, p - x_{n+1} \rangle \\ &+ 2 \delta_n \langle p, p - x_{n-1} \rangle \\ &\leq (1 - \delta_n) \| (1 - \theta_n) b_n + \theta_n t_n - p \|^2 + 2 \delta_n \theta_n \| b_n - t_n \| \| p - x_{n+1} \| \\ &+ 2 \delta_n \langle p, p - x_{n+1} \rangle \\ &\leq (1 - \delta_n) \left(\|x_n - p\|^2 + 3 \alpha_n \|x_n - x_{n-1}\| M_2 \right) + 2 \delta_n \theta_n \| b_n - t_n \| \| p - x_{n+1} \| \\ &+ 2 \delta_n \langle p, p - x_{n+1} \rangle \\ &\leq (1 - \delta_n) \| x_n - p \|^2 + 3 \alpha_n \| x_n - x_{n-1} \| M_2 + 2 \delta_n \theta_n \| b_n - t_n \| \| p - x_{n+1} \| \\ &+ 2 \delta_n \langle p, p - x_{n+1} \rangle \\ &= (1 - \delta_n) \| x_n - p \|^2 + \delta_n d_n, \ n \geq 1, \end{aligned}$$

where $d_n = 3\frac{\alpha_n}{\delta_n} \|x_n - x_{n-1}\| M_2 + 2(\theta_n \|b_n - t_n\| \|p - x_{n+1}\| + \langle p, p - x_{n+1} \rangle)$. According to Lemma 2.5.36, to conclude the proof, it sufficies to show that $\limsup_{k \to \infty} d_{n_k} \leq 0$ for every subsequence $\{\|x_{n_k} - p\|\}$ of $\{\|x_n - p\|\}$ satisfying the condition;

$$\liminf_{k \to \infty} \left(\|x_{n_k+1} - p\| - \|x_{n_k} - p\| \right) \ge 0.$$
(4.3.50)

To show that $\limsup_{k\to\infty} d_{n_k} \leq 0$, suppose that $\{\|x_{n_k} - p\|\}$ is a subsequence of $\{\|x_n - p\|\}$ such that (4.3.50) holds. Then,

$$\liminf_{k \to \infty} \left(\|x_{n_k+1} - p\|^2 - \|x_{n_k} - p\|^2 \right) = \liminf_{k \to \infty} \left[\left(\|x_{n_k+1} - p\| - \|x_{n_k} - p\| \right) \\ \times \left(\|x_{n_k+1} - p\| + \|x_{n_k} - p\| \right) \right] \\ \ge 0.$$
(4.3.51)

Now, observe that

$$\begin{aligned} \|(1-\theta_{n}-\delta_{n})(b_{n}-p)+\theta_{n}(t_{n}-p)\|^{2} &= (1-\theta_{n}-\delta_{n})^{2}\|b_{n}-p\|^{2} \\ &+ 2(1-\theta_{n}-\delta_{n})\theta_{n}\langle b_{n}-p,t_{n}-p\rangle + \theta_{n}^{2}\|t_{n}-p\|^{2} \\ &\leq (1-\theta_{n}-\delta_{n})^{2}\|b_{n}-p\|^{2} + (1-\theta_{n}-\delta_{n})\theta_{n}\|b_{n}-p\|^{2} \\ &+ (1-\theta_{n}-\delta_{n})\theta_{n}\|t_{n}-p\|^{2} + \theta_{n}^{2}\|t_{n}-p\|^{2} \\ &\leq (1-\theta_{n}-\delta_{n})(1-\delta_{n})\|b_{n}-p\|^{2} \\ &+ \theta_{n}(1-\delta_{n})\|t_{n}-p\|^{2}. \end{aligned}$$
(4.3.52)

Also, since $\{x_n\}$ and $\{\delta_n\}$ are bounded, there exists $M_3 > 0$ such that

$$\left(\delta_{n} \|p\| + 2||x_{n+1} - (1 - \delta_{n})p||\right)||p|| \le M_{3}$$

, for all $n\geq 1.$ Hence, from Step 5, (4.3.28), (4.3.49) and (4.3.52), we get

$$\begin{split} \|x_{n+1} - p\|^2 &= \|(1 - \theta_n - \delta_n)(b_n - p) + \theta_n(t_n - p) - \delta_n p\|^2 \\ &= \|(1 - \theta_n - \delta_n)(b_n - p) + \theta_n(t_n - p)\|^2 + \delta_n^2 \|p\|^2 \\ &- 2\delta_n \langle (1 - \theta_n - \delta_n)(b_n - p) + \theta_n(t_n - p), p \rangle \\ &\leq \|(1 - \theta_n - \delta_n)(b_n - p) + \theta_n(t_n - p)\|^2 \\ &+ \delta_n \left(\delta_n \|p\|^2 + 2\|(1 - \theta_n - \delta_n)(b_n - p) + \theta_n(t_n - p)\|\| \|p\| \right) \\ &= \|(1 - \theta_n - \delta_n)(b_n - p) + \theta_n(t_n - p)\|^2 \\ &+ \delta_n \left(\delta_n \|p\| + 2\|x_{n+1} - (1 - \delta_n)p\|\| \right) \|p\| \\ &\leq (1 - \theta_n - \delta_n)(1 - \delta_n)\|b_n - p\|^2 + \theta_n(1 - \delta_n)\|t_n - p\|^2 + \delta_n M_3 \\ &\leq (1 - \theta_n - \delta_n)(1 - \delta_n)\|b_n - p\|^2 + \theta_n(1 - \delta_n)(\|b_n - p\|^2 \quad (4.3.53) \\ &- \gamma_1^{-1}(2 - \gamma_1)\|t_n - b_n\|^2) + \delta_n M_3 \\ &\leq (1 - \theta_n - \delta_n)(1 - \delta_n)\|w_n - p\|^2 + \theta_n(1 - \delta_n)(\|w_n - p\|^2 \\ &- \gamma_1^{-1}(2 - \gamma_1)\|t_n - b_n\|^2) + \delta_n M_3 \\ &\leq \|w_n - p\|^2 - \theta_n(1 - \delta_n)\gamma_1^{-1}(2 - \gamma_1)\|t_n - b_n\|^2 + \delta_n M_3 \\ &\leq \|x_n - p\|^2 - \theta_n(1 - \delta_n)\gamma_1^{-1}(2 - \gamma_1)\|t_n - b_n\|^2 \\ &+ \delta_n \left(3\frac{\alpha_n}{\delta_n}\|x_n - x_{n-1}\|M_2 + M_3 \right), \end{split}$$

Thus, from (4.3.51) and (4.3.19), we obtain

$$\limsup_{k \to \infty} \left((1 - \delta_{n_k}) \theta_{n_k} \gamma_1^{-1} (2 - \gamma_1) \| t_{n_k} - b_{n_k} \|^2 \right) \le \limsup_{k \to \infty} \left(\| x_{n_k} - p \|^2 - \| x_{n_k+1} - p \|^2 \right) + \limsup_{k \to \infty} \left(\delta_{n_k} \left(3M_1 M_2 + M_3 \right) \right) = - \liminf_{k \to \infty} \left(\| x_{n_k+1} - p \|^2 - \| x_{n_k} - p \|^2 \right) \le 0,$$

which implies that

$$\lim_{k \to \infty} \|t_{n_k} - b_{n_k}\| = 0.$$
(4.3.54)

From (4.3.53) and (4.3.31), we obtain

$$\begin{aligned} \|x_{n_{k}+1} - p\|^{2} &\leq (1 - \theta_{n_{k}} - \delta_{n_{k}})(1 - \delta_{n_{k}})\|b_{n_{k}} - p\|^{2} + \theta_{n_{k}}(1 - \delta_{n_{k}})\|b_{n_{k}} - p\|^{2} + \delta_{n_{k}}M_{3} \\ &\leq \|b_{n_{k}} - p\|^{2} - \epsilon^{2}\|T^{*}(z_{n_{k}} - Tw_{n_{k}})\|^{2} + \delta_{n_{k}}M_{3}, \end{aligned}$$

which implies from (4.3.49) that

$$\limsup_{k \to \infty} \|T^*(z_{n_k} - Tw_{n_k})\|^2 \le \frac{1}{\epsilon^2} \limsup_{k \to \infty} \left(\|x_{n_k} - p\|^2 - \|x_{n_k+1} - p\|^2 \right) \\ + \frac{1}{\epsilon^2} \limsup_{k \to \infty} \left(\delta_{n_k} \left(3 \frac{\alpha_{n_k}}{\delta_{n_k}} \|x_{n_k} - x_{n_k-1}\| M_2 + M_3 \right) \right) \\ \le \frac{-1}{\epsilon^2} \liminf_{k \to \infty} \left(\|x_{n+1} - p\|^2 - \|x_{n_k} - p\|^2 \right) \le 0.$$

Thus,

$$\lim_{k \to \infty} \|T^*(z_{n_k} - Tw_{n_k})\| = 0.$$
(4.3.55)

Hence, by Step 3, we obtain

$$\|b_{n_k} - w_{n_k}\| = \eta_{n_k} \|T^*(z_{n_k} - Tw_{n_k})\| \to 0, \quad \text{as} \quad k \to \infty.$$
(4.3.56)

From Step 5 and (4.3.54), we get

$$\|x_{n_k+1} - b_{n_k}\| \le \theta_{n_k} \|b_{n_k} - t_{n_k}\| + \delta_{n_k} \|b_{n_k}\| \to 0, \text{ as } k \to \infty.$$
(4.3.57)

From (4.3.56) and (4.3.57), we get

$$\lim_{k \to \infty} \|x_{n_k+1} - w_{n_k}\| = 0.$$
(4.3.58)

From (4.3.43) and (4.3.58), we get

$$\lim_{k \to \infty} \|x_{n_k+1} - x_{n_k}\| = 0.$$
(4.3.59)

By Lemma 4.3.4, there exists a subsequence $\{x_{n_{k_j}}\}$ of $\{x_{n_k}\}$ which converges weakly to z, such that

$$\limsup_{k \to \infty} \langle p, p - x_{n_k} \rangle = \lim_{j \to \infty} \langle p, p - x_{n_{k_j}} \rangle = \langle p, p - z \rangle.$$

Also, since we have established (4.3.54) and (4.3.56), then we can apply Lemma 4.3.5 to get that $z \in \Gamma$.

Thus, since $p = P_{\Gamma} 0$, we obtain from the previous inequality that

$$\limsup_{k \to \infty} \langle p, p - x_{n_k} \rangle = \langle p, p - z \rangle \le 0,$$

which implies by (4.3.59) that

$$\lim_{k \to \infty} \langle p, p - x_{n_k+1} \rangle \le 0. \tag{4.3.60}$$

Now, recall that $d_{n_k} = 3 \frac{\alpha_{n_k}}{\delta_{n_k}} \|x_{n_k} - x_{n_{k-1}}\| M_2 + 2\left(\theta_{n_k}\|b_{n_k} - t_{n_k}\|\|p - x_{n_{k+1}}\| + \langle p, p - x_{n_{k+1}}\rangle\right)$. Thus, by (4.3.18), (4.3.54) and (4.3.60), we obtain that $\limsup_{k \to \infty} d_{n_k} \leq 0$. Hence, get that $\lim_{n \to \infty} \|x_n - p\| = 0$. Therefore, $\{x_n\}$ converges strongly to $p = P_{\Gamma} 0$.

Remark 4.3.7.

- Observe that by setting $\mathcal{H}_1 = \mathcal{H}_2 = \mathcal{H}$, F = 0 and $T = I_{\mathcal{H}}$ (the identity operator on \mathcal{H}) in Theorem 4.3.6, we obtain as a corollary, an inertial projection and contraction method requiring only one projection onto the feasible set \mathcal{C} per iteration, with fixed stepsize for solving the classical VIP (1.2.1) when A is pseudomonotone and Lipschitz continuous.
- The conclusions of Lemma 4.3.4, Lemma 4.3.5 and Theorem 4.3.6 still hold even if $\mu \in (0, \frac{1}{L_1})$ and $\lambda \in (0, \frac{1}{L_2})$ in Algorithm 4.3.2 are replaced with variable stepsizes μ_n and λ_n , respectively such that

$$0 < \inf_{n \ge 1} \mu_n \le \sup_{n \ge 1} \mu_n < \frac{1}{L_1}$$
 and $0 < \inf_{n \ge 1} \lambda_n \le \sup_{n \ge 1} \lambda_n < \frac{1}{L_2}$.

We now turn to establish the strong convergence of Algorithm 4.3.3. We begin with the following useful results.

Remark 4.3.8.

(i) From (4.3.3) in Algorithm 4.3.3, it is clear that $\lambda_{n+1} \leq \lambda_n$, $\forall n \geq 1$. Also, since F is L_2 -Lipschitz continuous, we get in the case of $FTw_n \neq Fy_n$ in Algorithm 4.3.3, that

$$\lambda_{n+1} = \min\left\{\frac{a_2||Tw_n - y_n||}{||FTw_n - Fy_n||}, \ \lambda_n\right\} \ge \min\left\{\frac{a_2}{L_2}, \ \lambda_n\right\},$$

which by induction, implies that $\{\lambda_n\}$ is bounded below by $\min\left\{\frac{a_2}{L_2}, \lambda_1\right\}$. Since $\{\lambda_n\}$ is also monotone nonincreasing, we have that the limit exists, and $\lim_{n\to\infty} \lambda_n \geq \min\left\{\frac{a_2}{L_2}, \lambda_1\right\} > 0$.

(ii) Similar to (i), we have that the limit of the stepsize $\{\mu_n\}$ exists and $\lim_{n\to\infty} \mu_n \ge \min\{\frac{a_1}{L_1}, \mu_1\} > 0.$

Lemma 4.3.9. Let $\{x_n\}$ be a sequence generated by Algorithm 4.3.3 under Assumption 4.3.1. Then,

$$||Tw_n - y_n|| \le \gamma_2^{-1} \left(\frac{\lambda_{n+1} + \lambda_n a_2}{\lambda_{n+1} - \lambda_n a_2}\right) ||Tw_n - z_n||, \ \forall n \ge 1$$
(4.3.61)

and

$$||b_n - u_n|| \le \gamma_1^{-1} \left(\frac{\mu_{n+1} + \mu_n a_1}{\mu_{n+1} - \mu_n a_1}\right) ||b_n - t_n||, \ \forall n \ge 1.$$
(4.3.62)

Proof. From (4.3.3), we know that

$$||FTw_n - Fy_n|| \le \frac{a_2}{\lambda_{n+1}} ||Tw_n - y_n||, \ \forall n \ge 1,$$
(4.3.63)

holds for both $FTw_n = Fy_n$ and $FTw_n \neq Fy_n$. Thus, similar to (4.3.12), we obtain that

$$\langle Tw_n - y_n, r_n \rangle \ge ||Tw_n - y_n||^2 - \lambda_n ||Tw_n - y_n|| ||FTw_n - Fy_n||$$

 $\ge (1 - \lambda_n \frac{a_2}{\lambda_{n+1}}) ||Tw_n - y_n||^2,$

which implies that

$$\begin{aligned} \|Tw_{n} - y_{n}\|^{2} &\leq \frac{1}{\left(1 - \lambda_{n} \frac{a_{2}}{\lambda_{n+1}}\right)} \langle Tw_{n} - y_{n}, r_{n} \rangle \\ &\leq \frac{1}{\left(1 - \lambda_{n} \frac{a_{2}}{\lambda_{n+1}}\right)} \beta_{n} \|r_{n}\| \left(\|Tw_{n} - y_{n}\| + \lambda_{n} \frac{a_{2}}{\lambda_{n+1}}\|Tw_{n} - y_{n}\|\right) \\ &= \left(\frac{1 + \frac{\lambda_{na_{2}}}{\lambda_{n+1}}}{1 - \frac{\lambda_{na_{2}}}{\lambda_{n+1}}}\right) \beta_{n} \|r_{n}\| \|Tw_{n} - y_{n}\| \\ &= \gamma_{2}^{-1} \left(\frac{\lambda_{n+1} + \lambda_{n}a_{2}}{\lambda_{n+1} - \lambda_{n}a_{2}}\right) \|Tw_{n} - z_{n}\| \|Tw_{n} - y_{n}\|. \end{aligned}$$

This further gives (4.3.61). In a similar manner, we get (4.3.62).

In the light of Remark
$$4.3.8$$
 and Lemma $4.3.9$, we have the following result.

Theorem 4.3.10. Let $\{x_n\}$ be a sequence generated by Algorithm 4.3.3 under Assumption 4.3.1. Then, $\{x_n\}$ converges strongly to $p \in \Gamma$, where $||p|| = \min\{||z|| : z \in \Gamma\}$.

Proof. Let $p \in \Gamma$. Then, by replacing μ and λ with μ_n and λ_n , respectively in Lemma 4.3.4, we can easily get that $\{x_n\}$ is bounded. Also, from Remark 4.3.8 (i), we obtain that

$$\lim_{n \to \infty} \left(\frac{\lambda_{n+1} + \lambda_n a_2}{\lambda_{n+1} - \lambda_n a_2} \right) = \frac{1 + a_2}{1 - a_2}.$$
(4.3.64)

Similarly, from Remark 4.3.8 (ii), we obtain that

$$\lim_{n \to \infty} \left(\frac{\mu_{n+1} + \mu_n a_1}{\mu_{n+1} - \mu_n a_1} \right) = \frac{1 + a_1}{1 - a_1}.$$
(4.3.65)

Furthermore, we can follow the same argument as in the proof of Theorem 4.3.6 to obtain (4.3.54), (4.3.55) and (4.3.56). Also, using (4.3.55) and (4.3.56), we can get (4.3.33). Hence, from (4.3.33), (4.3.61) and (4.3.64), we obtain that

$$\lim_{k \to \infty} \|Tw_{n_k} - y_{n_k}\| = 0.$$

Again, from (4.3.54), (4.3.47) and (4.3.65), we obtain that

$$\lim_{k \to \infty} \|b_{n_k} - u_{n_k}\| = 0.$$

The rest of the proof follows the same arguments as in Lemma 4.3.5 and Theorem 4.3.6. \Box

Remark 4.3.11.

- Similar to Remark 4.3.7, we can obtain as corollary, an inertial projection and contraction method with adaptive stepsize, requiring only one projection onto the feasible set C per iteration, for solving the classical VIP (1.2.4) when A is pseudomonotone and Lipschitz continuous.
- As seen in our convergence analysis, we did not follow the usual "Two Cases Approach" (i.e., Case 1 and Case 2) used in many works to guarantee strong convergence of iterative methods. Thus, the techniques and ideas employed in our strong convergence analysis are new for solving the SVIP (1.2.4)-(1.2.5).
- If the operators A and F are monotone and Lipschitz continuous, then we do not need them to be sequentially weakly continuous. This is because the sequential weakly continuity assumption was only used after (4.3.36) to get the conclusion of Lemma 4.3.5. But, from (4.3.36), we obtain

$$0 \leq \langle FTw_{n_{k}}, x - Tw_{n_{k}} \rangle + \frac{1}{\lambda} \langle y_{n_{k}} - Tw_{n_{k}}, x - y_{n_{k}} \rangle + \langle FTw_{n_{k}}, Tw_{n_{k}} - y_{n_{k}} \rangle$$

$$\leq (\langle FTw_{n_{k}} - Fx, x - Tw_{n_{k}} \rangle + \langle Fx, x - Tw_{n_{k}} \rangle) + \frac{1}{\lambda} ||y_{n_{k}} - Tw_{n_{k}}|| ||x - y_{n_{k}}||$$

$$+ ||FTw_{n_{k}}|| ||Tw_{n_{k}} - y_{n_{k}}||$$

$$\leq \langle Fx, x - Tw_{n_{k}} \rangle + \frac{1}{\lambda} ||y_{n_{k}} - Tw_{n_{k}}|| ||x - y_{n_{k}}|| + ||FTw_{n_{k}}|| ||Tw_{n_{k}} - y_{n_{k}}||, \forall x \in \mathcal{Q}$$

where the last inequality follows from the monotonicty of F. Thus, by passing limit as $k \to \infty$ and noting that $\{Tw_{n_k}\}$ converges weakly to z, we obtain from (4.3.35) that

$$\langle Fx, x - Tz \rangle \ge 0, \quad \forall x \in \mathcal{Q},$$

which follows from Lemma 2.5.8 that $Tz \in VI(F, \mathcal{Q})$. Similarly, we also get that $z \in VI(A, \mathcal{C})$. Hence, the conclusion of Lemma 4.3.5 holds.

- In finite dimensional spaces, Theorem 4.3.6 and Theorem 4.3.10 are still true if the operators A and F are only required to be pseudomonotone and Lipschitz continuous. This is an improvement over the result of He *et al.* [121] since no product space formulation is required even with the relaxed pseudomonotonicity assumption.
- The conclusions of Theorem 4.3.6 and Theorem 4.3.10 are still valid if Step 1 of Algorithm 4.3.2 and Algorithm 4.3.3 is replaced with the following: $0 \le \alpha_n \le \bar{\alpha}_n$, where

$$\bar{\alpha}_n := \begin{cases} \min\left\{\alpha, \frac{\tau_n}{\|x_n - x_{n-1}\|}\right\}, & \text{if } x_n \neq x_{n-1} \\ \alpha, & \text{otherwise,} \end{cases}$$

with $\alpha \in [0, 1)$.

4.3.3 Numerical experiments

In this section, we discuss the numerical behavior of Algorithm 4.3.2 and Algorithm 4.3.3 using various test examples. In all of these examples, we give numerical comparison of our methods with the methods of He *et al.* [121, Algorithm 1] (see Appendix 4.2.17), Pham *et al.* [206, Algorithm 1] (see Appendix 4.3.16), Reich and Tuyen [213, Theorem 4.4] (see Appendix 4.2.18) and Tian and Jiang [244, Algorithm (3.1)] (see Algorithm (2.5.9)).

All codes are written in Matlab 2016 (b) and performed on a personal computer with an Intel(R) Core(TM) i5-2600 CPU at 2.30GHz and 8.00 Gb-RAM. In Tables 4.3.1-4.3.3, "Iter." means the number of iterations while "CPU" means the CPU time in seconds. In our computations, we randomly choose $x_0, x_1 \in \mathcal{H}_1$ (see the cases below), $\eta \geq 0$, $\gamma_i \in (0, 2), a_i \in (0, 1), i = 1, 2, \lambda_1 > 0$ and $\mu_1 > 0$ (the choices of these parameters will be discussed in Remark 4.3.15). We choose $\delta_n = \frac{1}{n+1}, \ \theta_n = \frac{1}{2} - \delta_n$ and $\alpha_n = \bar{\alpha}_n$ with $\tau_n = \frac{\delta_n}{n^{0.01}}$ and different choices of $\alpha := 3, 6, 9, 12, 15$, which will be discussed in detail in our numerical analysis.

Furthermore, we define

$$TOL_{n} := \frac{1}{2} \left(||x_{n} - P_{\mathcal{C}}(x_{n} - \mu A x_{n})||^{2} + ||Tx_{n} - P_{\mathcal{Q}}(Tx_{n} - \lambda FTx_{n})||^{2} \right)$$

for Algorithm 4.3.2, Algorithm 4.3.3, He *et al.* [121, Algorithm 1], Pham *et al.* [206, Algorithm 1] and Reich and Tuyen [213, Theorem 4.4]. While for Algorithm (2.5.9), we define

$$TOL_n := \frac{1}{2} \left(||x_n - P_{\mathcal{C}}(x_n - \mu A x_n)||^2 + ||Tx_n - STx_n||^2 \right),$$

and use the stopping criterion $\text{TOL}_n < \varepsilon$ for the iterative processes, where ε is the predetermined error. Note that if $\text{TOL}_n = 0$, then $x_n \in \Gamma$, that is, x_n is a solution of the SVIP considered in this work.

We first consider an example in finite dimensional spaces. In this example, we carry out our comparison with the method of He *et al.* [121] since their method is given in finite dimensional spaces.

Example 4.3.12. Following [121, Example 5.2] (see also [116]), we consider a separable, convex and quadratic programming problem

$$\min_{x,y} \{ G_1(x) + G_2(y) \mid Tx = y, \ x \in \mathcal{C}, \ y \in \mathcal{Q} \},$$
(4.3.66)

where

$$G_1(x) = \frac{1}{2}x'M_1x + q'_1x$$
 and $G_2(y) = \frac{1}{2}y'M_2y + q'_2y$ (x' means the transpose of x).

Problem (4.3.66) can also be rewritten as the SVIP (1.2.4)-(1.2.5), where

$$A(x) = M_1 x + q_1$$
 and $F(y) = M_2 y + q_2$.

The matrices M_i (i = 1, 2) are formed as: $M_i = V_i \sum_i V'_i$, where $V_i = I - \frac{2v_i v'_i}{\|v_i\|^2}$ and $\sum_i = \text{diag}(\sigma_{i1}, \sigma_{i2}, \cdots, \sigma_{iN_i})$ are the Householder and the diagonal matrix, respectively,

with $N_1 = N$ and $N_2 = m$ being the dimensional of x and y, respectively. Furthermore,

$$\sigma_{i,j} = \cos\frac{j\pi}{N_i + 1} + 1 + \frac{\cos\frac{\pi}{N_i + 1} + 1 - \widehat{C}_i(\cos\frac{N_i\pi}{N_i + 1} + 1)}{\widehat{C}_i - 1}, \quad j = 1, 2, \cdots, N_i,$$

where \hat{C}_i is the present condition number of M_i .

As in He *et al.* [121, Example 5.2], we choose $\widehat{C}_i = 10^4$, $q_i = 0$, i = 1, 2 and uniformly take the vector $v_i \in \mathbb{R}^{N_i}$ (i = 1, 2) in (-1, 1). Thus, A, F are monotone and Lipschitz continuous operators with $L_i = ||M_i||$, i = 1, 2 (which can be computed in Matlab). Furthermore, we generate the bounded linear operator $T \in \mathbb{R}^{M \times N}$ with independent Gaussian components distributed in the interval (0, 1), and then normalize each column of T with the unit norm. We set $\mathcal{C} = \{x \in \mathbb{R}^N : ||x|| \leq 1\}$ and $\mathcal{Q} = \{y \in \mathbb{R}^m : l \leq y \leq u\}$, where all the entries of $l \in \mathbb{R}^m$ and $u \in \mathbb{R}^m$ are the smallest and the largest components of $\tilde{y} = T\tilde{x}$, respectively, with \tilde{x} been a sparse vector whose components are uniformly distributed in (0, 1). The projections onto \mathcal{C} and \mathcal{Q} are effectively computed in Matlab. More so, we consider different scenarios of the problem's dimensions. That is, N = 100, 300, 500, 1000and m = N/2.

Since the Lipschitz constants L_i (i = 1, 2) can be known, we choose in Algorithm 4.3.2, $\mu = \frac{0.5}{L_1}$ and $\lambda = \frac{0.75}{L_2}$, and take the starting point $x_1 = (1, 1, \ldots, 1)'$ while the entries of x_0 are randomly generated in [0, 1]. For Algorithm 1 of He *et al.* [121], we take $\mu_1 = 5 \left(||T' \mathbb{H}T|| + L_1 \right) / v, \ \mu_2 = 10 \left(||T' \mathbb{H}T|| + L_2 \right) / v, \ v = 0.8, \ \mathbb{H} = \frac{2}{||T'T||} I_N, \ \gamma = 1.2$ and $\rho = -1.5$ (which is the optimum choice in their implementation), with starting points $x_1 = (1, 1, \ldots, 1)', \ y_1 = (0, 0, \ldots, 0)'$ and $\lambda_1 = (0, 0, \ldots, 0)'$. Furthermore, we take $\varepsilon = 10^{-8}$ for the stopping criterion and obtain the numerical results reported in Table 4.3.1 and Figure 4.4. We stress that these choices as well as the stopping criterion are the same as in He *et al.* [121, Example 5.2].

(N,m)		Alg.4.3.2	Alg.4.3.2	Alg.4.3.2	Alg.4.3.2	Alg.4.3.2	He <i>et al</i> .
		$\alpha = 3$	$\alpha = 6$	$\alpha = 9$	$\alpha = 12$	$\alpha = 15$	
(100, 50)	CPU	0.0305	0.0284	0.0195	0.0190	0.0180	0.2552
	Iter.	18	18	18	18	18	178
(300, 150)	CPU	0.0048	0.0033	0.0031	0.0030	0.0030	0.0388
	Iter.	19	19	19	19	19	216
(500, 250)	CPU	0.0055	0.0045	0.0042	0.0041	0.0041	0.0619
	Iter.	20	20	20	20	20	232
(1000, 500)	CPU	0.0081	0.0081	0.0080	0.0079	0.0074	0.1591
	Iter.	21	21	21	21	21	275

Table 4.3.1. Numerical results for Example 4.3.12 with $\varepsilon = 10^{-8}$.

The next two examples are given in infinite dimensional Hilbert spaces. In these examples, we carry out our comparisons with the methods of Pham *et al.* [206], Reich and Tuyen [213] and Tian and Jiang [244] since their methods are given in infinite dimensional Hilbert spaces.



Figure 4.4: The behavior of TOL_n with $\varepsilon = 10^{-8}$ for Example 4.3.12: Top Left: (N, m) = (100, 50); Top Right: (N, m) = (300, 150); Bottom Left: (N, m) = (500, 250); Bottom Right: (N, m) = (1000, 500).

Example 4.3.13. Let $\mathcal{H}_1 = (l_2(\mathbb{R}), ||.||_{l_2}) = \mathcal{H}_2$, where $l_2(\mathbb{R}) := \{x = (x_1, x_2, x_3, \dots), x_i \in \mathbb{R} : \sum_{i=1}^{\infty} |x_i|^2 < \infty\}$ and $||x||_{l_2} := \left(\sum_{i=1}^{\infty} |x_i|^2\right)^{\frac{1}{2}}, \quad \forall x \in l_2(\mathbb{R}).$ Now, define the operator $T : l_2(\mathbb{R}) \to l_2(\mathbb{R})$ by $Tx = \left(0, x_1, \frac{x_2}{2}, \frac{x_3}{3}, \dots\right), \quad \forall x \in l_2(\mathbb{R}).$

Then, T is a bounded linear operator on $l_2(\mathbb{R})$ with adjoint

$$T^*y = \left(y_2, \frac{y_3}{2}, \frac{y_4}{3}, \dots\right), \ \forall y \in l_2(\mathbb{R}).$$

Indeed, let $x = (x_1, x_2, x_3, ...), y = (y_1, y_2, y_3, ...)$ be arbitrary in $l_2(\mathbb{R})$ and α_1, α_2 be arbitrary in \mathbb{R} . Then,

$$T(\alpha_1 x + \alpha_2 y) = \left(0, \ \alpha_1 x_1 + \alpha_2 y_1, \ \frac{\alpha_1 x_2 + \alpha_2 y_2}{2}, \ \frac{\alpha_1 x_3 + \alpha_2 y_3}{3}, \cdots\right) \\ = \left(0, \ \alpha_1 x_1, \frac{\alpha_1 x_2}{2}, \ \frac{\alpha_1 x_3}{3}, \cdots\right) + \left(0, \ \alpha_2 y_1, \frac{\alpha_2 y_2}{2}, \ \frac{\alpha_2 y_3}{3}, \cdots\right) \\ = \alpha_1 T(x) + \alpha_2 T(y).$$

Therefore, T is linear. Furthermore, $||Tx||_{l_2} \leq ||x||_{l_2}$, $\forall x \in l_2(\mathbb{R})$. Thus, T is also bounded. The verification that T^* is the adjoint of T follows directly from definition.

Let $\mathcal{C} = \mathcal{Q} = \{x \in l_2(\mathbb{R}) : ||x - a||_{l_2} \leq r\}$, where $a = (1, \frac{1}{2}, \frac{1}{3}, \cdots)$, r = 3 for \mathcal{C} and $a = (\frac{1}{2}, \frac{1}{4}, \frac{1}{8}, \cdots)$, r = 1 for \mathcal{Q} . Then \mathcal{C}, \mathcal{Q} are nonempty closed and convex subsets of $l_2(\mathbb{R})$. Thus,

$$P_{\mathcal{C}}(x) = P_{\mathcal{Q}}(x) = \begin{cases} x, & \text{if } x \in ||x-a||_{l_2} \le r, \\ \frac{x-a}{||x-a||_{l_2}}r + a, & \text{otherwise.} \end{cases}$$

Now, define the operators $A, F : l_2(\mathbb{R}) \to l_2(\mathbb{R})$ by

$$A(x_1, x_2, x_3, \dots) = (x_1 e^{-x_1^2}, 0, 0, \dots), \quad F(x_1, x_2, x_3, \dots) = (5x_1 e^{-x_1^2}, 0, 0, \dots)$$

Then, by Example 2.1.5, A, F are pseudomonotone, Lipschitz continuous and sequentially weakly continuous but not monotone.

More so, for Algorithm (2.5.9), we define the mappings $S, h : l_2(\mathbb{R}) \to l_2(\mathbb{R})$ by $Sx = (0, x_1, x_2, ...)$ and $hx = (0, \frac{x_1}{2}, \frac{x_2}{2}, ...)$, for all $x \in l_2(\mathbb{R})$. Then, we consider the following cases for the numerical experiments.

Case 1: Take $x_1 = (1, \frac{1}{2}, \frac{1}{3}, \cdots)$ and $x_0 = (\frac{1}{2}, \frac{1}{5}, \frac{1}{10}, \cdots)$. **Case 2:** Take $x_1 = (\frac{1}{2}, \frac{1}{5}, \frac{1}{10}, \cdots)$ and $x_0 = (1, \frac{1}{2}, \frac{1}{3}, \cdots)$. **Case 3:** Take $x_1 = (1, \frac{1}{4}, \frac{1}{9}, \cdots)$ and $x_0 = (\frac{1}{2}, \frac{1}{4}, \frac{1}{8}, \cdots)$.

Case 4: Take
$$x_1 = (\frac{1}{2}, \frac{1}{4}, \frac{1}{8}, \cdots)$$
 and $x_0 = (1, \frac{1}{4}, \frac{1}{9}, \cdots)$.



Figure 4.5: The behavior of TOL_n with $\varepsilon = 10^{-9}$ for Example 4.3.13: Top Left: Case 1; Top Right: Case 2; Bottom Left: Case 3; Bottom Right: Case 4.

Cases		Alg.4.3.3	Alg.4.3.3	Alg.4.3.3	Alg.4.3.3	Alg.4.3.3	Pham	Reich	Tian
		$\alpha = 3$	$\alpha = 6$	$\alpha = 9$	$\alpha = 12$	$\alpha = 15$	et al.	Tuyen	Jiang
1	CPU	0.0040	0.0031	0.0025	0.0027	0.0027	0.0056	0.0113	0.1071
	Iter.	17	17	17	17	17	69	105	159
2	CPU	0.0039	0.0038	0.0033	0.0048	0.0051	0.0120	0.0671	0.1000
	Iter.	16	16	16	16	16	66	130	152
3	CPU	0.0029	0.0034	0.0029	0.0045	0.0051	0.0070	0.0133	0.1059
	Iter.	17	17	17	17	17	69	101	158
4	CPU	0.0030	0.0032	0.0027	0.0034	0.0037	0.0076	0.0142	0.1060
	Iter.	17	17	17	17	17	68	125	157

Table 4.3.2. Numerical results for Example 4.3.13 with $\varepsilon = 10^{-9}$

Example 4.3.14. Let $\mathcal{H}_1 = \mathcal{H}_2 = L_2([0, 1])$ be endowed with inner product

$$\langle x, y \rangle = \int_0^1 x(t)y(t)dt, \quad \forall \ x, y \in L_2([0, 1])$$

and norm

$$||x|| := \left(\int_0^1 |x(t)|^2 dt\right)^{\frac{1}{2}}, \quad \forall \ x, y \in L_2([0,1]).$$

Let $T: L_2([0,1]) \to L_2([0,1])$ be defined by

$$Tx(s) = \int_0^1 K(s,t)x(t)dt, \ \forall x \in L_2([0,1]),$$

where K is a continuous real-valued function on $[0, 1] \times [0, 1]$. Then, T is a bounded linear operator with adjoint

$$T^*x(s) = \int_0^1 K(t,s)x(t)dt, \ \forall x \in L_2([0,1]).$$

In particular, we define $K(s,t) = e^{-st}$ for all $s, t \in [0,1]$.

Let $C = \{x \in L_2([0,1]) : \langle y, x \rangle \leq b\}$, where y = t + 1 and b = 1, then C is a nonempty closed and convex subset of $L_2([0,1])$. Thus, we define the metric projection P_C as:

$$P_{\mathcal{C}}(x) = \begin{cases} \frac{b - \langle y, x \rangle}{||y||^2} y + x, & \text{if } \langle y, x \rangle > b, \\ x, & \text{if } \langle y, x \rangle \le b. \end{cases}$$

Also, let $\mathcal{Q} = \{x \in L_2([0,1]) : ||x|| \leq r\}$, where r = 2, then \mathcal{Q} is a nonempty closed and convex subset of $L_2([0,1])$. Thus, we define $P_{\mathcal{Q}}$ as:

$$P_{\mathcal{Q}}(x) = \begin{cases} x, & \text{if } x \in \mathcal{Q}, \\ \frac{x}{\|x\|^2}r, & \text{otherwise.} \end{cases}$$

Now, define $A: L_2([0,1]) \to L_2([0,1])$ by

$$A(x)(t) = e^{-\|x\|} \int_0^t x(s) ds, \quad \forall x \in L_2([0,1]), \quad t \in [0,1].$$

Then, A is pseudomonotone and Lipschitz continuous but not monotone on $L_2([0, 1])$ (see [250]).

Also, define $F : \mathcal{Q} \to L_2([0,1])$ by

$$F(x)(t) := g(x)M(x)(t), \quad \forall \ x \in \mathcal{Q}, \quad t \in [0,1],$$

where $g: \mathcal{Q} \to \mathbb{R}$ is defined by $g(x) := \frac{1}{1+\|x\|^2}$ and $M: L_2([0,1]) \to L_2([0,1])$ is defined by $M(x)(t) := \int_0^t x(s) ds$, $\forall x \in L_2([0,1]), t \in [0,1]$. As given in [222], g is $\frac{16}{25}$ -Lipschitz continuous and $\frac{1}{5} \leq g(x) \leq 1$, $\forall x \in \mathcal{C}$. Also, M is the Volterra integral mapping which is bounded and linear monotone. Hence, F is pseudomonotone and Lipschitz continuous but not monotone (see [222]).

For Algorithm (2.5.9), we define the mapping $S: L_2([0,1]) \to L_2([0,1])$ by

$$Sx(t) = \int_0^1 tx(s)ds, \quad t \in [0, 1].$$

Then, S is nonexpansive. Indeed, we have

$$|Sx(t) - Sy(t)|^{2} = |\int_{0}^{1} t(x(s) - y(s))ds|^{2} \le \left(\int_{0}^{1} t|x(s) - y(s)|ds\right)^{2}$$
$$\le \int_{0}^{1} |x(s) - y(s)|^{2}ds$$
$$= ||x - y||^{2}.$$

Thus, we obtain that

$$||Sx - Sy||^{2} = \int_{0}^{1} |Sx(t) - Sy(t)|^{2} \le ||x - y||^{2}.$$

We also define $h: L_2([0,1]) \to L_2([0,1])$ by

$$hx(t) = \int_0^1 \frac{t}{2} x(s) ds, \quad x \in [0, 1].$$

Similar to above, it can also be shown that h is a contraction mapping.

We consider the following cases for the numerical experiments of this example. Case 1: Take $x_1(t) = 1 + t^2$ and $x_0(t) = t + 5$.

Case 2: Take $x_1(t) = \sin(t)$ and $x_0(t) = t + 1$.

Case 3: Take $x_1(t) = t + 1$ and $x_0(t) = t + t^3$.

Case 4: Take $x_1(t) = 0.7e^{-t} + 1$ and $x_0(t) = t + t^3$.

Table 4.3.3. Numerical results for Example 4.3.14 with $\varepsilon = 10^{-10}$.



Figure 4.6: The behavior of TOL_n with $\varepsilon = 10^{-10}$ for Example 4.3.14: Top Left: **Case 1**; Top Right: **Case 2**; Bottom Left: **Case 3**; Bottom Right: **Case 4**.

Cases		Alg.4.3.3	Alg.4.3.3	Alg.4.3.3	Alg.4.3.3	Alg.4.3.3	Pham	Reich	Tian
		$\alpha = 3$	$\alpha = 6$	$\alpha = 9$	$\alpha = 12$	$\alpha = 15$	et al.	Tuyen	Jiang
1	CPU	3.6417	3.4282	3.4009	3.3256	3.3112	12.6238	9.7880	25.7461
	Iter.	30	30	30	30	30	77	35	178
2	CPU	3.9291	3.7850	3.6274	3.4970	3.4821	12.8552	10.3671	27.8420
	Iter.	31	31	31	31	31	80	37	181
3	CPU	3.6099	3.3950	3.3538	3.3495	3.3447	12.6815	10.1993	26.2060
	Iter.	30	30	30	30	30	78	36	179
4	CPU	3.6508	3.4405	3.3442	3.3112	3.2636	12.9326	10.0650	26.3111
	Iter.	30	30	30	30	30	79	36	178

Remark 4.3.15. By using different starting points and varying the inertial extrapolation factor α in each example (Examples 4.3.12-4.3.14), we obtain the numerical results displayed in Tables 4.3.1-4.3.3 and Figures 4.4-4.6. We compared our proposed Algorithm 4.3.2 with the method of He *et al.* [121, Algorithm 1] in Example 4.3.12 while our proposed Algorithm 4.3.3 is compared with the methods of Pham *et al.* [206, Algorithm 1], Reich and Tuyen [213, Theorem 4.4] and Tian and Jiang [244, Algorithm (3.1)] in Examples 4.3.13-4.3.14.

Furthermore, we note the following from our numerical experiments:

- In the numerical experiments, we randomly choose the parameters $\eta \geq 0$, $\gamma_i \in (0,2)$, $a_i \in (0,1)$, i = 1, 2, $\lambda_1 > 0$ and $\mu_1 > 0$, and observed that irrespective of the choices made, the number of iteration does not change and no significant difference in the CPU time.
- In all the examples, we check the sensitivity of α for each starting points in order to know if the choices of α affect the efficiency of our methods. We can see from the tables and graphs that the number of iterations for our proposed methods remain consistent (well-behaved) for α = 3, 6, 9, 12, 15. Also, there are no much significant difference in the CPU time as we vary α in {3, 6, 9, 12, 15}. However, it can be inferred from Table 4.3.1 that as α increases from 3 to 15, the convergence speed of Algorithm 4.3.2 increases. Therefore, the optimum choice of α for Example 4.3.12 is α = 15. Similarly, we can see from Table 4.3.3 that as α increases from 3 to 15, the convergence speed of Algorithm 4.3.3 also increases. Hence, the optimum choice of α for Example 4.3.14 is again α = 15. Unfortunately, we cannot follow similar pattern for Example 4.3.13, but we can see clearly that the the optimum choice of α for this example is α = 9.
- It can also be inferred from Table 4.3.1 that in Example 4.3.12, the number of iterations for our proposed Algorithm 4.3.2 is almost the same for all starting points while the method of He *et al.* [121, Algorithm 1] depends on the different starting points. Also, from Tables 4.3.2-4.3.3, we see that for all starting points in Example 4.3.13-4.3.14, the number of iterations for our proposed Algorithm 4.3.3 is more consistent than every other methods.

• From the tables and figures, we can see clearly that in terms of both CPU time and number of iterations, our proposed Algorithm 4.3.2 outperforms the method of He *et al.* [121, Algorithm 1] in Example 4.3.12 while our proposed Algorithm 4.3.3 outperforms the methods of Pham *et al.* [206, Algorithm 1], Reich and Tuyen [213, Theorem 4.4] and Tian and Jiang [244, Algorithm (3.1)] in Examples 4.3.13-4.3.14. Hence, our methods are more efficient than these other methods.

Appendix 4.3.16. Algorithm 1 of Pham et al. [206].

Step 0. Choose $\mu_0, \lambda_0 > 0, \quad \mu, \lambda \in (0, 1), \quad \{\tau_n\} \subset [\underline{\tau}, \ \overline{\tau}] \subset \left(0, \frac{1}{||T||^2 + 1}\right), \quad \{\alpha_n\} \subset (0, 1)$ such that $\lim_{n \to \infty} \alpha_n = 0$ and $\sum_{n=1}^{\infty} \alpha_n = \infty$. Step 1. Let $x_1 \in \mathcal{H}_1$. Set n = 1. Step 2. Compute

$$u_n = T x_n,$$

$$v_n = P_{\mathcal{Q}}(u_n - \mu_n f u_n),$$

$$w_n = P_{\mathcal{Q}_n}(u_n - \mu_n f v_n),$$

where

$$\mathcal{Q}_n = \{ w_2 \in \mathcal{H}_2 : \langle u_n - \mu_n f u_n - v_n, w_2 - v_n \rangle \le 0 \}$$

and

$$\mu_{n+1} = \begin{cases} \min\left\{\frac{\mu||u_n - v_n||}{||fu_n - fv_n||}, \ \mu_n\right\}, & \text{if } fu_n \neq fv_n, \\ \mu_n, & \text{otherwise.} \end{cases}$$

Step 3. Compute

$$y_n = x_n + \tau_n T^*(w_n - u_n),$$

$$z_n = P_{\mathcal{C}}(y_n - \lambda_n A y_n),$$

$$t_n = P_{\mathcal{C}_n}(y_n - \lambda_n A z_n),$$

where

$$\mathcal{C}_n = \{ w_1 \in \mathcal{H}_1 : \langle y_n - \lambda_n A y_n - z_n, w_1 - z_n \rangle \le 0 \}$$

and

$$\lambda_{n+1} = \begin{cases} \min\left\{\frac{\lambda||y_n - z_n||}{||Ay_n - Az_n||}, \lambda_n\right\}, & \text{if } Ay_n \neq Az_n, \\ \lambda_n, & \text{otherwise.} \end{cases}$$

Step 4. Compute

$$x_{n+1} = \alpha_n x_1 + (1 - \alpha_n) t_n$$

Set n := n + 1 and go back to Step 2.

4.4 Projection and contraction methods for split variational inequality problem

In the literature, several methods have been proposed for solving the SVIP and most of these methods require that the underlying operators be co-coercive while some of them requires product space formulation of the problem. These restrictive conditions affect the feasibility of these existing methods. In order to overcome these setbacks, we propose two new inertial projection and contraction methods for solving the SVIP in real Hilbert spaces without the co-coercive condition and without the product space formulation, which does not fully exploit the attractive splitting structure of the SVIP. The sequences generated by these methods converge strongly to the solution of the SVIP in real Hilbert spaces under the assumptions that the operators are pseudomonotone, Lipschitz continuous and without the sequentially weakly continuity condition. Furthermore, we present several numerical experiments for the proposed methods and compare their performance with other related methods in the literature.

4.4.1 Proposed methods

In this section, we present our proposed methods for solving the SVIP (1.2.4)-(1.2.5).

Assumption 4.4.1. Suppose that the following conditions hold:

- (a) The feasible sets C and Q are nonempty closed and convex subsets of the real Hilbert spaces \mathcal{H}_1 and \mathcal{H}_2 , respectively.
- (b) $A: \mathcal{H}_1 \to \mathcal{H}_1$ and $F: \mathcal{H}_2 \to \mathcal{H}_2$ are pseudomonotone and Lipschitz continuous with Lipschitz constants L_1 and L_2 , respectively.
- (c) $A: \mathcal{H}_1 \to \mathcal{H}_1 \text{ and } F: \mathcal{H}_2 \to \mathcal{H}_2 \text{ satisfy the following property}$ whenever $\{x_n\} \subset \mathcal{C}$ and $\{y_n\} \subset \mathcal{Q}$, and $x_n \rightharpoonup x$, $y_n \rightharpoonup y$ one has $||Ax|| \leq \liminf_{n \to \infty} ||Ax_n||$ and $||Fy|| \leq \liminf_{n \to \infty} ||Fy_n||$.
- (d) $T : \mathcal{H}_1 \to \mathcal{H}_2$ is a bounded linear operator and the solution set $\Gamma := \{z \in VI(A, \mathcal{C}) : Tz \in VI(F, \mathcal{Q})\}$ is nonempty, where $VI(A, \mathcal{C})$ is the solution set of the classical VIP (1.2.4).
- (e) $\{\alpha_n\} \subset (0,1]$ is non-increasing with $\lim_{n \to \infty} \alpha_n = 0$ and $\sum_{n=1}^{\infty} \alpha_n = \infty$.
- (f) $0 \le \theta_n \le \theta_{n+1} \le \theta < \frac{1}{3}, \sigma \in (0, \frac{1}{2}].$
- (g) $\{\phi_n\}$ and $\{\psi_n\}$ are non-negative sequences such that $\sum_{n=1}^{\infty} \phi_n < +\infty$ and $\sum_{n=1}^{\infty} \psi_n < +\infty$.

When the Lipschitz constants L_1 and L_2 are known, we present the following method for solving the SVIP (1.2.4)-(1.2.5).

Algorithm 4.4.2. Inertial projection and contraction method with fixed step size.

Step 0: Choose sequences $\{\alpha_n\}_{n=1}^{\infty}$ and $\{\theta_n\}_{n=1}^{\infty}$ such that the conditions from Assumption 4.4.1 (e)-(f) hold and let $\eta \ge 0, \gamma_i \in (0, 2), i = 1, 2, \mu \in (0, \frac{1}{L_1}), \lambda \in (0, \frac{1}{L_2}),$ and $x_0, x_1 \in \mathcal{H}_1$ be given arbitrarily. Set n := 1.

Step 1: Given the iterates x_{n-1} and x_n $(n \ge 1)$, $\alpha_n \in (0, 1)$ and $\theta_n \in [0, \frac{1}{3})$, compute

$$w_n = \alpha_n x_0 + (1 - \alpha_n) x_n + \theta_n (x_n - x_{n-1}).$$

Step 2: Compute

$$y_n = P_{\mathcal{Q}}(Tw_n - \lambda FTw_n),$$
$$z_n = Tw_n - \gamma_2 \beta_n r_n,$$

where $r_n := Tw_n - y_n - \lambda(FTw_n - Fy_n)$ and $\beta_n := \frac{\langle Tw_n - y_n, r_n \rangle}{\|r_n\|^2}$, if $r_n \neq 0$, otherwise $\beta_n = 0$. Step 3: Compute

$$b_n = w_n + \eta_n T^* (z_n - Tw_n),$$

where the step size η_n is chosen such that for some $\epsilon > 0$, $\eta_n \in \left(\epsilon, \frac{\|Tw_n - z_n\|^2}{\|T^*(Tw_n - z_n)\|^2} - \epsilon\right)$, if $z_n \neq Tw_n$; otherwise $\eta_n = \eta$. Step 4: Compute

$$u_n = P_{\mathcal{C}}(b_n - \mu A b_n),$$
$$t_n = b_n - \gamma_1 \gamma_n v_n,$$

where $v_n := b_n - u_n - \mu(Ab_n - Au_n)$ and $\gamma_n := \frac{\langle b_n - u_n, v_n \rangle}{\|v_n\|^2}$, if $v_n \neq 0$, otherwise $\gamma_n = 0$. Step 5: Compute

$$x_{n+1} = (1 - \sigma)w_n + \sigma t_n.$$

Set n := n + 1 and go back to **Step 1**.

When the Lipschitz constants L_1 and L_2 are not known, we present the following method with adaptive step size for solving the SVIP (1.2.4)-(1.2.5).

Algorithm 4.4.3. Inertial projection and contraction method with adaptive step size strategy.

Step 0: Choose the control parameters such that conditions (e)-(g) of Assumption 4.4.1 hold and let $\eta \ge 0, \gamma_i \in (0, 2)$ $a_i \in (0, 1), i = 1, 2, \lambda_1 > 0, \mu_1 > 0, \alpha \ge 3$ and $x_0, x_1 \in \mathcal{H}_1$ be given arbitrarily. Set n := 1.

Step 1: Given the iterates x_{n-1} and x_n $(n \ge 1)$, $\alpha_n \in (0, 1)$ and $\theta_n \in [0, \frac{1}{3})$, compute

$$w_n = \alpha_n x_0 + (1 - \alpha_n) x_n + \theta_n (x_n - x_{n-1}).$$

Step 2: Compute

$$y_n = P_{\mathcal{Q}}(Tw_n - \lambda_n FTw_n),$$
$$z_n = Tw_n - \gamma_2 \beta_n r_n,$$

where $r_n := Tw_n - y_n - \lambda_n (FTw_n - Fy_n), \beta_n := \frac{\langle Tw_n - y_n, r_n \rangle}{\|r_n\|^2}$, if $r_n \neq 0$, otherwise $\beta_n = 0$; and

$$\lambda_{n+1} = \begin{cases} \min\left\{\frac{a_2||Tw_n - y_n||}{||FTw_n - Fy_n||}, \ \lambda_n + \phi_n\right\}, & \text{if } FTw_n \neq Fy_n\\ \lambda_n + \phi_n, & \text{otherwise.} \end{cases}$$
(4.4.1)

Step 3: Compute

$$b_n = w_n + \eta_n T^*(z_n - Tw_n),$$

where the step size η_n is chosen such that for some $\epsilon > 0$, $\eta_n \in \left(\epsilon, \frac{\|Tw_n - z_n\|^2}{\|T^*(Tw_n - z_n)\|^2} - \epsilon\right)$, if $z_n \neq Tw_n$; otherwise $\eta_n = \eta$. **Step 4:** Compute

$$u_n = P_{\mathcal{C}}(b_n - \mu_n A b_n),$$
$$t_n = b_n - \gamma_1 \gamma_n v_n,$$

where $v_n := b_n - u_n - \mu_n (Ab_n - Au_n)$, $\gamma_n = \frac{\langle b_n - u_n, v_n \rangle}{\|v_n\|^2}$, if $v_n \neq 0$, otherwise $\gamma_n = 0$; and

$$\mu_{n+1} = \begin{cases} \min\left\{\frac{a_1||b_n - u_n||}{||Au_n - Ab_n||}, \ \mu_n + \psi_n\right\}, & \text{if } Ab_n \neq Au_n \\ \mu_n + \psi_n, & \text{otherwise.} \end{cases}$$
(4.4.2)

Step 5: Compute

$$x_{n+1} = (1 - \sigma)w_n + \sigma t_n.$$

Set n := n + 1 and go back to Step 1.

We outline and discuss some of the properties of our proposed methods.

Remark 4.4.1.

- The choice of the inertial factor $\theta_n \in [0, \frac{1}{3})$ in Algorithms 4.4.2 and 4.4.3 is new and different from the choices in literature (see for example [55] and other references therein). As far as we know, this is the first time the inertial factor θ_n is chosen such that $\theta_n \in [0, \frac{1}{3})$ and solves the SVIP when the underlying operators are pseudomonotone and Lipschitz continuous.
- Algorithm 4.4.3 uses simple step size rules in (4.4.1) and (4.4.2), which generate non-monotonic sequences of step sizes. The step sizes are constructed such that the dependence of the algorithm on the initial step sizes λ_1 and μ_1 is reduced.
- We point out that if the pseudomonotone operators A and F are sequentially weakly continuous, then A and F satisfy condition (c) but the converse is not true. Hence, condition (c) is strictly weaker than the sequentially weakly continuity condition commonly employed in the literature (e.g., see [55, 191]).

• Algorithm 4.4.2 can be viewed as a modified inertial projection and contraction method involving one projection onto \mathcal{C} per iteration for solving the classical VIP in \mathcal{H}_1 . Algorithm 4.4.3 can be viewed as a modified inertial projection and contraction method involving one projection onto \mathcal{Q} per iteration under a bounded linear operator T for solving VIP in \mathcal{H}_2 . Our methods improves other methods in literature which requires extra projections onto half-spaces or feasible sets (see [206] (see Appendix 4.3.16) and other references therein). In Step 2 of Algorithms 4.4.2 and 4.4.3, r_n can be described as weighted average of $(Tw_n - y_n \sim \lambda FTw_n)$ and a hypothetical $(T\tilde{w}_n - \tilde{y}_n \sim \lambda FT\tilde{w}_n)$ in \mathcal{H}_2 , where $T\tilde{w}_n = Tw_n - \lambda FTw_n$ and $\tilde{y}_n = y_n - \lambda Fy_n$. In Step 4 of Algorithms 4.4.3 and 4.4.3, v_n follows similar description. From Step 2 and Step 4 of Algorithms 4.4.3 and 4.4.3, we have

$$\beta_n ||r_n||^2 = \langle Tw_n - y_n, r_n \rangle, \ \forall n \ge 1$$
(4.4.3)

holds for both $r_n = 0$ and $r_n \neq 0$. Similarly, we have that

$$\gamma_n ||v_n||^2 = \langle b_n - u_n, v_n \rangle, \ \forall n \ge 1$$
(4.4.4)

holds for both $v_n = 0$ and $v_n \neq 0$.

- The step sizes $\{\lambda_n\}$ and $\{\mu_n\}$ given by (4.4.1) and (4.4.2), respectively are generated at each iteration by some simple computations which makes Algorithm 4.4.3 easier to implement since it does not require the prior knowledge of the Lipschitz constants L_1 and L_2 .
- Algorithms 4.4.2 and 4.4.3 does not require any product space formulation unlike other algorithms in literature which require that the problem be transformed into a product space (see [63] and other references therein). This makes our algorithms easier to implement since they do not encounter the difficulties that might be caused by the product space.

Remark 4.4.2. [192] The choice of the step size η_n in Step 3 of Algorithms 4.4.2 and 4.4.3 do not require the prior knowledge of the operator norm ||T||. Furthermore, the value of η does not influence the algorithms, but it was introduced for the sake of clarity.

Lemma 4.4.3. [192] The step size η_n given in Step 3 of Algorithms 4.4.2 and 4.4.3 is well-defined.

4.4.2 Convergence analysis

Lemma 4.4.4. Let $\{x_n\}$ be a sequence generated by Algorithm 4.4.2 under Assumption 4.4.1. Then, the following inequality holds:

$$||x_{n+1} - p||^2 \le ||w_n - p||^2 - ||x_{n+1} - w_n||^2, \ \forall \ p \in \Gamma.$$
(4.4.5)

Proof. Let $p \in \Gamma$. From the definition of y_n and the characteristic property of P_Q , we obtain

$$\langle y_n - Tw_n + \lambda FTw_n, y_n - Tp \rangle \le 0.$$
 (4.4.6)

Since $Tp \in VI(F,Q)$ and $y_n \in Q$, we have

$$\langle FTp, y_n - Tp \rangle \ge 0$$

and from the pseudomonotonicity of F we have,

$$\langle Fy_n, y_n - Tp \rangle \ge 0$$

Since $\lambda > 0$, we obtain

$$\langle \lambda F y_n, y_n - T p \rangle \ge 0. \tag{4.4.7}$$

Adding (4.4.6) and (4.4.7), we obtain

$$\langle Tw_n - y_n - \lambda (FTw_n - Fy_n), y_n - Tp \rangle \ge 0.$$
(4.4.8)

From (4.4.8) and the definition of r_n in Step 2, we obtain

$$\langle Tw_n - Tp, r_n \rangle = \langle Tw_n - y_n, r_n \rangle + \langle y_n - Tp, r_n \rangle = \langle Tw_n - y_n, r_n \rangle + \langle y_n - Tp, Tw_n - y_n - \lambda (FTw_n - Fy_n) \rangle \geq \langle Tw_n - y_n, r_n \rangle,$$

which implies that

$$-\langle Tw_n - Tp, r_n \rangle \le -\langle Tw_n - y_n, r_n \rangle.$$
(4.4.9)

From the definition of z_n in Step 2, we have

$$||\beta_n \cdot r_n||^2 = \gamma_2^{-2} ||z_n - Tw_n||^2.$$
(4.4.10)

Hence, from Lemma 2.1.1, (4.4.3), (4.4.9) and (4.4.10) we obtain

$$\begin{aligned} \|z_n - Tp\|^2 &= \|Tw_n - \gamma_2\beta_n r_n - Tp\|^2 \\ &= \|Tw_n - Tp\|^2 + \gamma_2^2\beta_n^2\|r_n\|^2 - 2\gamma_2\beta_n\langle Tw_n - Tp, r_n\rangle \\ &\leq \|Tw_n - Tp\|^2 + \gamma_2^2\beta_n^2\|r_n\|^2 - 2\gamma_2\beta_n\langle Tw_n - y_n, r_n\rangle \\ &= \|Tw_n - Tp\|^2 + \gamma_2^2\beta_n^2\|r_n\|^2 - 2\gamma_2\beta_n \cdot \beta_n\|r_n\|^2 \\ &= \|Tw_n - Tp\|^2 - \gamma_2(2 - \gamma_2)\|\beta_n \cdot r_n\|^2 \\ &= \|Tw_n - Tp\|^2 - \gamma_2^{-1}(2 - \gamma_2)\|z_n - Tw_n\|^2. \end{aligned}$$
(4.4.11)

Also, from Step 3, Lemma 2.1.1 and (4.4.11) we obtain

$$\begin{aligned} \|b_{n} - p\|^{2} &= \|w_{n} - p\|^{2} + \eta_{n}^{2} \|T^{*}(z_{n} - Tw_{n})\|^{2} + 2\eta_{n} \langle w_{n} - p, T^{*}(z_{n} - Tw_{n}) \rangle \\ &= \|w_{n} - p\|^{2} + \eta_{n}^{2} \|T^{*}(z_{n} - Tw_{n})\|^{2} + 2\eta_{n} \langle Tw_{n} - Tp, z_{n} - Tw_{n} \rangle \\ &= \|w_{n} - p\|^{2} + \eta_{n}^{2} \|T^{*}(z_{n} - Tw_{n})\|^{2} \\ &+ \eta_{n} \left[\|z_{n} - Tp\|^{2} - \|Tw_{n} - Tp\|^{2} - \|z_{n} - Tw_{n}\|^{2} \right] \\ &\leq \|w_{n} - p\|^{2} + \eta_{n}^{2} \|T^{*}(z_{n} - Tw_{n})\|^{2} - \eta_{n} \|z_{n} - Tw_{n}\|^{2} \\ &= \|w_{n} - p\|^{2} - \eta_{n} \left[\|z_{n} - Tw_{n}\|^{2} - \eta_{n} \|T^{*}(z_{n} - Tw_{n})\|^{2} \right]. \end{aligned}$$

$$(4.4.12)$$

Thus, by the condition on η_n , we obtain that

$$||b_n - p||^2 \le ||w_n - p||^2$$

Following similar argument used in obtaining (4.4.11), we obtain

$$\|t_n - p\|^2 = \|b_n - \gamma_1 \gamma_n v_n - p\|^2$$

$$\leq \|b_n - p\|^2 - \gamma_1^{-1} (2 - \gamma_1) \|t_n - b_n\|^2.$$
(4.4.13)

From Step 5 we have,

$$||x_{n+1} - p||^{2} = ||(1 - \sigma)w_{n} + \sigma t_{n} - p||^{2}$$

= $||(1 - \sigma)(w_{n} - p) + \sigma(t_{n} - p)||^{2}$
= $(1 - \sigma)||w_{n} - p||^{2} + \sigma ||t_{n} - p||^{2} - (1 - \sigma)\sigma ||w_{n} - t_{n}||^{2}.$ (4.4.14)

Substituting (4.4.13) into (4.4.14), we have

$$\begin{aligned} \|x_{n+1} - p\|^{2} &\leq (1 - \sigma) \|w_{n} - p\|^{2} + \sigma \left(\|b_{n} - p\|^{2} - \gamma_{1}^{-1}(2 - \gamma_{1})\|t_{n} - b_{n}\|^{2} \right) \\ &- (1 - \sigma)\sigma \|w_{n} - t_{n}\|^{2} \\ &= (1 - \sigma) \|w_{n} - p\|^{2} + \sigma \|b_{n} - p\|^{2} - \sigma\gamma_{1}^{-1}(2 - \gamma_{1})\|t_{n} - b_{n}\|^{2} \\ &- (1 - \sigma)\sigma \|w_{n} - t_{n}\|^{2} \\ &\leq (1 - \sigma) \|w_{n} - p\|^{2} + \sigma \|w_{n} - p\|^{2} - \sigma\gamma_{1}^{-1}(2 - \gamma_{1})\|t_{n} - b_{n}\|^{2} \\ &- (1 - \sigma)\sigma \|w_{n} - t_{n}\|^{2} \\ &= \|w_{n} - p\|^{2} - \sigma\gamma_{1}^{-1}(2 - \gamma_{1})\|t_{n} - b_{n}\|^{2} - (1 - \sigma)\sigma \|w_{n} - t_{n}\|^{2}. \end{aligned}$$
(4.4.15)

From Step 4, we have $t_n - w_n = \frac{1}{\sigma}(x_{n+1} - w_n)$. Substituting this into the previous equality we have,

$$\begin{aligned} \|x_{n+1} - p\|^2 &\leq \|w_n - p\|^2 - \sigma \gamma_1^{-1} (2 - \gamma_1) \|t_n - b_n\| - (1 - \sigma) \sigma \cdot \frac{1}{\sigma^2} \|x_{n+1} - w_n\|^2 \\ &= \|w_n - p\|^2 - \sigma \gamma_1^{-1} (2 - \gamma_1) \|t_n - b_n\| - \left(\frac{1}{\sigma} - 1\right) \|x_{n+1} - w_n\|^2 \\ &\leq \|w_n - p\|^2 - \left(\frac{1}{\sigma} - 1\right) \|x_{n+1} - w_n\|^2 \\ &\leq \|w_n - p\| - \zeta \|x_{n+1} - w_n\|^2, \end{aligned}$$

where $\zeta := \left(\frac{1}{\sigma} - 1\right)$.

Lemma 4.4.5. Let $\{x_n\}$ be a sequence generated by Algorithm 4.4.2 under Assumption 4.4.1. Then $\forall p \in \Gamma$, we have

$$||x_{n+1} - p||^2 - ||x_n - p||^2 \le \alpha_n ||x_n - x_0||^2 + \theta_n ||x_n - p||^2 - \theta_{n-1} ||x_{n-1} - p||^2 \qquad (4.4.16)$$

- $(1 - 3\theta_{n+1} - \alpha_n) ||x_n - x_{n+1}||^2$
- $2\alpha_n \langle x_n - p, x_n - x_0 \rangle - 2\theta_{n+1} ||x_{n+1} - x_n||^2$
+ $2\theta_n ||x_n - x_{n-1}||^2 - \alpha_{n+1} ||x_0 - x_{n+1}||^2.$

Proof. From the definition of w_n and Lemma (2.1.1), we have

$$||w_{n} - p||^{2} = ||\alpha_{n}x_{0} + (1 - \alpha_{n})x_{n} + \theta_{n}(x_{n} - x_{n-1}) - p||^{2}$$

$$= ||(x_{n} - p) + \theta_{n}(x_{n} - x_{n-1}) - \alpha_{n}(x_{n} - x_{0})||^{2}$$

$$= ||x_{n} - p||^{2} + ||\theta_{n}(x_{n} - x_{n-1}) - \alpha_{n}(x_{n} - x_{0})|^{2}$$

$$+ 2\langle x_{n} - p, \theta_{n}(x_{n} - x_{n-1}) - \alpha_{n}(x_{n} - x_{0})\rangle$$

$$= ||x_{n} - p||^{2} + ||\theta_{n}(x_{n} - x_{n-1}) - \alpha_{n}(x_{n} - x_{0})||^{2}$$

$$+ 2\theta_{n}\langle x_{n} - p, x_{n} - x_{n-1}\rangle - 2\alpha_{n}\langle x_{n} - p, x_{n} - x_{0}\rangle.$$

(4.4.17)

Now, replacing p with x_{n+1} in (4.4.17), we obtain

$$\|w_n - x_{n+1}\|^2 = \|x_n - x_{n+1}\|^2 + \|\theta_n(x_n - x_{n-1}) - \alpha_n(x_n - x_0)\|^2$$

$$+ 2\theta_n \langle x_n - x_{n+1}, x_n - x_{n-1} \rangle - 2\alpha_n \langle x_n - x_{n+1}, x_n - x_0 \rangle.$$
(4.4.18)

Substituting (4.4.17) and (4.4.18) into (4.4.5) and from the condition on σ , we have

$$\begin{aligned} \|x_{n+1} - p\|^2 &\leq \|x_n - p\|^2 + \|\theta_n(x_n - x_{n-1}) - \alpha_n(x_n - x_0)\|^2 + 2\theta_n\langle x_n - p, x_n - x_{n-1}\rangle \\ &- 2\alpha_n\langle x_n - p, x_n - x_0\rangle - \|x_n - x_{n+1}\|^2 - \|\theta_n(x_n - x_{n-1}) \\ &- \alpha_n(x_n - x_0)\|^2 - 2\theta_n\langle x_n - x_{n+1}, x_n - x_{n-1}\rangle + 2\alpha_n\langle x_n - x_{n+1}, x_n - x_0\rangle \\ &= \|x_n - p\|^2 + 2\theta_n\langle x_n - p, x_n - x_{n-1}\rangle - 2\alpha_n\langle x_n - p, x_n - x_0\rangle \\ &- 2\theta_n\langle x_n - x_{n+1}, x_n - x_{n-1}\rangle - \|x_n - x_{n+1}\|^2 + 2\alpha_n\langle x_n - x_{n+1}, x_n - x_0\rangle \\ &= \|x_n - p\|^2 + 2\theta_n\langle x_n - p, x_n - x_{n-1}\rangle - 2\alpha_n\langle x_n - p, x_n - x_0\rangle \\ &+ \theta_n\|x_n - x_{n+1}\|^2 + \theta_n\|x_n - x_{n-1}\|^2 - \theta_n\|(x_n - x_{n+1}) + (x_n - x_{n-1})\|^2 \\ &- \|x_n - x_{n+1}\|^2 + 2\alpha_n\langle x_n - x_{n+1}, x_n - x_0\rangle. \end{aligned}$$

Hence,

$$||x_{n+1} - p||^2 - ||x_n - p||^2 \le 2\theta_n \langle x_n - p, x_n - x_{n-1} \rangle - 2\alpha_n \langle x_n - p, x_n - x_0 \rangle$$

$$- (1 - \theta_n) ||x_n - x_{n+1}||^2 + \theta_n ||x_n - x_{n-1}||^2$$

$$- \theta_n ||(x_n - x_{n+1}) + (x_n - x_{n-1})||^2 + 2\alpha_n \langle x_n - x_{n+1}, x_n - x_0 \rangle.$$
(4.4.19)

Applying Lemma 2.1.1 to (4.4.19), we obtain

$$\begin{aligned} \|x_{n+1} - p\|^2 - \|x_n - p\|^2 &\leq \theta_n \|x_n - x_{n-1}\|^2 - (1 - \theta_n) \|x_n - x_{n+1}\|^2 - 2\alpha_n \langle x_n - p, x_n - x_0 \rangle \\ &+ 2\theta_n \langle x_n - p, x_n - x_{n-1} \rangle + 2\alpha_n \langle x_n - x_{n+1}, x_n - x_0 \rangle \\ &= \theta_n \|x_n - x_{n-1}\|^2 - (1 - \theta_n) \|x_n - x_{n+1}\|^2 - 2\alpha_n \langle x_n - p, x_n - x_0 \rangle \\ &- \theta_n \|x_{n-1} - p\|^2 \\ &+ \theta_n \|x_n - p\|^2 + \theta_n \|x_n - x_{n-1}\|^2 - \alpha_n \|x_0 - x_{n+1}\| \\ &+ \alpha_n \|x_{n+1} - x_n\|^2 + \alpha_n \|x_n - x_0\|^2, \end{aligned}$$

which implies that

$$\begin{aligned} \|x_{n+1} - p\|^2 - \|x_n - p\|^2 &\leq \alpha_n \Big(\|x_n - x_0\|^2 - \|x_0 - x_{n+1}\|^2 \Big) + \theta_n \Big(\|x_n - p\|^2 - \|x_{n-1} - p\|^2 \Big) \\ (4.4.20) \\ &- \Big(1 - \theta_n - 2\theta_{n+1} - \alpha_n \Big) \|x_{n+1} - x_n\|^2 - 2\alpha_n \langle x_n - p, x_n - x_0 \rangle \\ &- 2\theta_{n+1} \|x_{n+1} - x_n\|^2 + 2\theta_n \|x_n - x_{n-1}\|^2. \end{aligned}$$

Using the fact that $\{\theta_n\}$ is non-decreasing and $\{\alpha_n\}$ is non-increasing on (4.4.20), we obtain (4.4.16), which is the desired conclusion.

Lemma 4.4.6. Let $\{x_n\}$ be a sequence generated by Algorithm 4.4.2 under Assumption 4.4.1. Then $\{x_n\}$ is bounded.

Proof. Let $p \in \Gamma$, then from (4.4.16) and Lemma 2.1.1, we have

$$\begin{aligned} \|x_{n+1} - p\|^2 - \|x_n - p\|^2 &\leq \alpha_n \|x_n - x_0\|^2 + \theta_n \|x_n - p\|^2 - \theta_{n-1} \|x_{n-1} - p\|^2 \\ &- (1 - 3\theta_{n+1} - \alpha_n) \|x_n - x_{n+1}\|^2 - 2\theta_{n+1} \|x_{n+1} - x_n\|^2 \\ &+ 2\theta_n \|x_n - x_{n-1}\|^2 - \alpha_{n+1} \|x_0 - x_{n+1}\|^2 - 2\alpha_n \langle x_n - p, x_n - x_0 \rangle \\ &= \alpha_n \|x_n - x_0\|^2 + \theta_n \|x_n - p\|^2 - \theta_{n+1} \|x_{n-1} - p\|^2 \\ &- (1 - 3\theta_{n+1} - \alpha_n) \|x_n - x_{n+1}\|^2 - 2\theta_{n+1} \|x_{n+1} - x_n\|^2 \\ &+ 2\theta_n \|x_n - x_{n-1}\|^2 - \alpha_{n+1} \|x_0 - x_{n+1}\|^2 - \alpha_n \|x_n - p\|^2 \\ &- \alpha_n \|x_n - x_0\|^2 + \alpha_n \|x_0 - p\|^2 \\ &\leq \theta_n \|x_n - p\|^2 - \theta_{n-1} \|x_{n-1} - p\|^2 - (1 - 3\theta_{n+1} - \alpha_n) \|x_n - x_{n+1}\|^2 \\ &- 2\theta_{n+1} \|x_{n+1} - x_n\|^2 \\ &+ 2\theta_n \|x_n - x_{n-1}\|^2 - \alpha_n \|x_n - p\|^2 + \alpha_n \|x_0 - p\|^2. \end{aligned}$$

From this we obtain

$$\begin{aligned} \|x_{n+1} - p\|^2 - \|x_n - p\|^2 + \alpha_n \|x_n - p\|^2 \\ &\leq \theta_n \|x_n - p\|^2 - \theta_{n-1} \|x_{n-1} - p\|^2 - (1 - 3\theta_{n+1} - \alpha_n) \|x_n - x_{n+1}\|^2 \\ &- 2\theta_{n+1} \|x_{n+1} - x_n\|^2 + 2\theta_n \|x_n - x_{n-1}\|^2 + \alpha_n \|x_0 - p\|^2. \end{aligned}$$
(4.4.22)

Let
$$\rho_j := e^{\sum_{i=1}^{j} \alpha_i}, j \ge 1$$
. Since $e^x \ge x + 1$ for all $x \in \mathbb{R}$, we have

$$\frac{1}{\rho_{n+1}} \Big(\rho_{n+1} \| x_{n+1} - p \|^2 - \rho_n \| x_n - p \|^2 \Big) = \| x_{n+1} - p \|^2 - \| x_n - p \|^2 \\
+ \frac{1}{\rho_{n+1}} \Big(\rho_{n+1} - \rho_n \Big) \| x_n - p \|^2 \\
\le \| x_{n+1} - p \|^2 - \| x_n - p \|^2 + \alpha_{n+1} \| x_n - p \|^2.$$

Since $\{\alpha_n\} \subset (0, 1]$ is non-increasing, we have

$$\frac{1}{\rho_{n+1}} \Big(\rho_{n+1} \|x_{n+1} - p\|^2 - \rho_n \|x_n - p\|^2\Big) \le \|x_{n+1} - p\|^2 - \|x_n - p\|^2 + \alpha_n \|x_n - p\|^2.$$
(4.4.23)

From (4.4.22) and (4.4.23), we obtain

$$\frac{1}{\rho_{n+1}} \Big[\rho_{n+1} \| x_{n+1} - p \|^2 - \rho_n \| x_n - p \|^2 \Big] \le \theta_n \| x_n - p \|^2 - \theta_{n-1} \| x_{n-1} - p \|^2 - (1 - 3\theta_{n+1} - \alpha_n) \| x_{n+1} - x_n \|^2 - 2\theta_{n+1} \| x_{n+1} - x_n \|^2 + 2\theta_n \| x_n - x_{n-1} \|^2 + \alpha_n \| x_0 - p \|^2.$$

Since $\rho_n \leq \rho_{n+1}, \rho_{n+1} = \rho_n e^{\alpha_{n+1}}$ and $\{\alpha_n\} \subset (0, 1]$ is non-increasing, we have

$$\begin{split} \rho_{n+1} \|x_{n+1} - p\|^2 &- \rho_n \|x_n - p\|^2 \le \rho_{n+1} \theta_n \|x_n - p\|^2 - \rho_n \theta_{n-1} \|x_{n-1} - p\|^2 \\ &- \rho_{n+1} (1 - 3\theta_{n+1} - \alpha_n) \|x_{n+1} - x_n\|^2 \\ &- 2\rho_{n+1} \theta_{n+1} \|x_{n+1} - x_n\|^2 + 2\rho_n \theta_n e^{\alpha_{n+1}} \|x_n - x_{n-1}\|^2 \\ &+ \rho_{n+1} \alpha_n \|x_0 - p\|^2, \end{split}$$

which implies that

$$\begin{aligned} \rho_{n+1} \|x_{n+1} - p\|^2 &- \rho_n \|x_n - p\|^2 \le \rho_{n+1} \theta_n \|x_n - p\|^2 - \rho_n \theta_{n-1} \|x_{n-1} - p\|^2 \\ &- \rho_{n+1} \left[1 - \theta_{n+1} \left(3 + 2(e^{\alpha_{n+1}} - 1) \right) - \alpha_n \right] \|x_{n+1} - x_n\|^2 \\ &- 2\rho_{n+1} \theta_{n+1} e^{\alpha_{n+1}} \|x_{n+1} - x_n\|^2 + 2\rho_n \theta_n e^{\alpha_n} \|x_n - x_{n-1}\|^2 \\ &+ \rho_{n+1} \alpha_n \|x_0 - p\|^2. \end{aligned}$$

Since $\{\theta_n\} \subset [0, \theta]$, we have

$$1 - \theta_{n+1}(3 + 2(e^{\alpha_{n+1}} - 1)) - \alpha_n \ge 1 - \theta(3 + 2(e^{\alpha_{n+1}} - 1)) - \alpha_n, \quad \forall n \in \mathbb{N}.$$
(4.4.24)

Since $\theta \in [0, \frac{1}{3})$ and $\lim_{n \to \infty} \alpha_n = 0$, it follows that the right-hand side of (4.4.24) is bounded below by a positive number, i.e., there exists a constant $\xi > 0$ such that $1 - \theta_{n+1}(3 + 2(e^{\alpha_{n+1}} - 1)) - \alpha_n \ge \xi$, for all $n \in \mathbb{N}$ sufficiently large, say for all $n \ge n_0$. Hence, we have

$$\begin{aligned} \rho_{n+1} \|x_{n+1} - p\|^2 &- \rho_n \|x_n - p\|^2 \le \rho_{n+1} \theta_n \|x_n - p\|^2 - \rho_n \theta_{n-1} \|x_{n-1} - p\|^2 \\ &- \xi \|x_{n+1} - x_n\|^2 \\ &- 2\rho_{n+1} \theta_{n+1} e^{\alpha_{n+1}} \|x_{n+1} - x_n\|^2 + 2\rho_n \theta_n e^{\alpha_n} \|x_n - x_{n-1}\|^2 \\ &+ \rho_{n+1} \alpha_n \|x_0 - p\|^2, \end{aligned}$$

which implies that for all $n \ge n_0$,

$$||x_{0} - p||^{2} \sum_{k=n_{0}+1}^{n} \rho_{k+1}\alpha_{k} \ge \rho_{n+1}||x_{n+1} - p||^{2} + 2\rho_{n+1}\theta_{n+1}e^{\alpha_{n+1}}||x_{n+1} - x_{n}||^{2} - \rho_{n+1}\theta_{n}||x_{n} - p||^{2} - \rho_{n_{0}+1}||x_{n_{0}+1} - p||^{2} - 2\rho_{n_{0}+1}\theta_{n_{0}+1}e^{\alpha_{n_{0}+1}}||x_{n_{0}+1} - x_{n_{0}}||^{2} + \rho_{n_{0}+1}\theta_{n_{0}}||x_{n_{0}} - p||^{2}.$$

$$(4.4.25)$$

Dividing the last inequality by ρ_{n+1} and omitting non-positive terms, we have

$$\|x_{n+1} - p\|^{2} - \theta_{n} \|x_{n} - p\|^{2} \leq e^{-t_{n+1}} \Big[\rho_{n_{0}+1} \|x_{n_{0}+1} - p\|^{2} + 2\rho_{n_{0}+1} \theta_{n_{0}+1} e^{\alpha_{n_{0}+1}} \|x_{n_{0}+1} - x_{n_{0}}\|^{2} - \rho_{n_{0}+1} \theta_{n_{0}} \|x_{n_{0}} - p\|^{2} \Big] + \|x_{0} - p\|^{2} e^{-t_{n}+1} \sum_{k=n_{0}+1}^{n} \alpha_{k} e^{t_{k}+1}$$

$$(4.4.26)$$

where $t_n := \sum_{i=1}^n \alpha_i$. Since $\alpha_k \in (0, 1]$ for all $k \in \mathbb{N}$, we observe that $\alpha_k e^{t_k + 1} \leq e^2(e^{t_k} - e^{t_k - 1})$, for all $k \geq 2$, so that

$$\sum_{k=n_0+1}^n \rho_{k+1}\alpha_k = \sum_{k=n_0+1}^n \alpha_k e^{t_k+1} \le e^2(e^{t_n} - e^{t_{n_0}}) \le e^2 e^{t_n}.$$

Using (4.4.26), the fact that $\{\theta_n\} \subset [0, \theta] \subset [0, \frac{1}{3})$ and $e^{-t_n+1} \leq 1$, we have

$$||x_{n+1} - p||^2 \le \theta ||x_n - p||^2 + \rho_{n_0+1} ||x_{n_0+1} - p||^2 + 2\rho_{n_0+1}\theta_{n_0+1}e^{\alpha_{n_0+1}} ||x_{n_0+1} - x_{n_0}||^2 + e^2 ||x_0 - p||^2.$$
(4.4.27)

Applying (4.4.27), $\theta \in [0, 1)$ and the convergence of the geometric series, we obtain

$$\|x_{n+1} - p\|^{2} \leq \theta^{n-n_{0}} \|x_{n_{0}+1} - p\|^{2}$$

$$+ \frac{1}{1-\theta} \Big[\rho_{n_{0}+1} \|x_{n_{0}+1} - p\|^{2} + 2\rho_{n_{0}+1} \theta_{n_{0}+1} e^{\alpha_{n_{0}+1}} \|x_{n_{0}+1} - x_{n_{0}}\|^{2} + e^{2} \|x_{0} - p\|^{2} \Big]$$

$$(4.4.28)$$

Since $\theta < 1$, it follows that $\{x_n\}$ is bounded.

Lemma 4.4.7. Let $\{x_n\}$ be a sequence generated by Algorithm 4.4.2 under Assumption 4.4.1. Suppose

$$\lim_{n \to \infty} \|x_{n+1} - x_n\| = 0$$

and

$$\lim_{n \to \infty} \left(\|x_{n+1} - p\|^2 - \theta_n \|x_n - p\|^2 \right) = 0.$$

Then $\{x_n\}$ converges strongly to p.

Proof. By the hypothesis of the lemma we have that

$$0 = \lim_{n \to \infty} \left(\|x_{n+1} - p\|^2 - \theta_n \|x_n - p\|^2 \right) = \lim_{n \to \infty} \left[\left(\|x_{n+1} - p\| + \sqrt{\theta_n} \|x_n - p\| \right) \times \left(\|x_{n+1} - p\| - \sqrt{\theta_n} \|x_n - p\| \right) \right]. \quad (4.4.29)$$

We claim that this implies

$$\lim_{n \to \infty} \left(\|x_{n+1} - p\| + \sqrt{\theta_n} \|x_n - p\| \right) = 0,$$

and from this it follows that $\{x_n\}$ converges strongly to p. On the contrary, assume that this limit does not hold. Then there exists a subset $K \subseteq \mathbb{N}$ and a constant $\beta > 0$ such that

$$||x_{n+1} - p|| + \sqrt{\theta_n} ||x_n - p|| \ge \beta, \forall \ n \in K.$$
(4.4.30)

Using (4.4.29) and the fact that $\theta_n \leq \theta < 1$, we have

$$0 = \lim_{n \in K} \left(\|x_{n+1} - p\| - \sqrt{\theta_n} \|x_n - p\| \right)$$

=
$$\lim_{n \in K} \sup \left(\|x_{n+1} - x_n + x_n - p\| - \sqrt{\theta_n} \|x_n - p\| \right)$$

\geq
$$\lim_{n \in K} \sup \left(\|x_n - p\| - \|x_{n+1} - x_n\| - \sqrt{\theta_n} \|x_n - p\| \right)$$

\geq
$$\lim_{n \in K} \sup \left((1 - \sqrt{\theta}) \|x_n - p\| - \|x_{n+1} - x_n\| \right)$$

=
$$(1 - \sqrt{\theta}) \limsup_{n \in K} \|x_n - p\| - \lim_{n \in K} \|x_{n+1} - x_n\|$$

=
$$(1 - \sqrt{\theta}) \limsup_{n \in K} \|x_n - p\|.$$

Thus, we have $\limsup_{n \in K} \|x_n - p\| \le 0$. Since $\liminf_{n \in K} \|x_n - p\| \ge 0$ holds, it follows that $\lim_{n \in K} \|x_n - p\| = 0$.

Applying (4.4.30), we obtain

$$||x_{n+1} - x_n|| \ge ||x_{n+1} - p|| - ||x_n - p||$$

= $||x_{n+1} - p|| + \sqrt{\theta_n} ||x_n - p|| - (1 + \sqrt{\theta_n}) ||x_n - p||$
 $\ge \frac{\beta}{2}$

for all $n \in K$ sufficiently large, which contradicts the assumption that $\lim_{n \to \infty} ||x_{n+1} - x_n|| = 0$. Hence, the result follows.

Lemma 4.4.8. Let $\{x_n\}$ be a sequence generated by Algorithm 4.4.2 under Assumption 4.4.1 such that $\lim_{n\to\infty} ||x_{n+1} - x_n|| = 0$. Suppose there exists a subsequence $\{x_{n_k}\}$ of $\{x_n\}$, which converges weakly to a point $z \in \mathcal{H}_1$ and $\lim_{k\to\infty} ||b_{n_k} - w_{n_k}|| = 0 = \lim_{k\to\infty} ||b_{n_k} - t_{n_k}||$, then $z \in \Gamma$. *Proof.* From the definition of w_n in Step 1 and by the statement of the hypothesis together with the fact that $\lim_{n\to\infty} \alpha_n = 0$, we obtain

$$\|w_n - x_n\| = \|\alpha_n(x_0 - x_n) + \theta_n(x_n - x_{n-1})\|$$

$$\leq \alpha_n \|x_0 - x_n\| + \theta_n \|x_n - x_{n-1}\| \to 0, \quad \text{as } n \to \infty$$
(4.4.31)

Since the subsequence $\{x_{n_k}\}$ of $\{x_n\}$ is weakly convergent to a point $z \in \mathcal{H}_1$, it follows that the subsequence $\{w_{n_k}\}$ of $\{w_n\}$ is also weakly convergent to $z \in \mathcal{H}_1$. Again, since T is a bounded linear operator, we obtain that $\{Tw_{n_k}\}$ converges weakly to Tz.

Without loss of generality, we may assume that $z_n \neq Tw_n$, then $\eta_n \in \left(\epsilon, \frac{\|z_n - Tw_n\|^2}{\|T^*(z_n - Tw_n)\|^2} - \epsilon\right)$. Hence, we obtain from (4.4.12) that

$$||b_{n_k} - p||^2 \le ||w_{n_k} - p||^2 - \eta_{n_k} \epsilon ||T^*(z_{n_k} - Tw_{n_k})||^2 \le ||w_{n_k} - p||^2 - \epsilon^2 ||T^*(z_{n_k} - Tw_{n_k})||^2,$$
(4.4.32)

which implies that

$$\epsilon^{2} \|T^{*}(z_{n_{k}} - Tw_{n_{k}})\|^{2} \leq \|w_{n_{k}} - p\|^{2} - \|b_{n_{k}} - p\|^{2} \\ \leq \|w_{n_{k}} - b_{n_{k}}\|^{2} + 2\|w_{n_{k}} - b_{n_{k}}\|\|b_{n_{k}} - p\|.$$

From our hypothesis, we have

$$\lim_{k \to \infty} \|T^*(z_{n_k} - Tw_{n_k})\| = 0.$$
(4.4.33)

From (4.4.12) and (4.4.33), we have

$$\begin{aligned} \eta_{n_k} \|z_{n_k} - Tw_{n_k}\|^2 &\leq \|w_{n_k} - p\|^2 - \|b_{n_k} - p\|^2 + \eta_{n_k}^2 \|T^*(z_{n_k} - Tw_{n_k})\|^2 \\ &\leq \|w_{n_k} - b_{n_k}\|^2 + 2\|w_{n_k} - b_{n_k}\|\|b_{n_k} - p\| + \eta_{n_k}^2 \|T^*(z_{n_k} - Tw_{n_k})\|^2 \\ &\to 0, \quad \text{as} \quad k \to \infty. \end{aligned}$$

Since $\eta_{n_k} > \epsilon > 0$, we obtain

$$\lim_{k \to \infty} \|z_{n_k} - Tw_{n_k}\| = 0.$$
(4.4.34)

Using the definition of r_n in Step 2, we observe

$$\langle Tw_{n_{k}} - y_{n_{k}}, r_{n_{k}} \rangle = \langle Tw_{n_{k}} - y_{n_{k}}, Tw_{n_{k}} - y_{n_{k}} - \lambda (FTw_{n_{k}} - Fy_{n_{k}}) \rangle$$

$$= \|Tw_{n_{k}} - y_{n_{k}}\|^{2} - \langle Tw_{n_{k}} - y_{n_{k}}, \lambda (FTw_{n_{k}} - Fy_{n_{k}}) \rangle$$

$$\geq \|Tw_{n_{k}} - y_{n_{k}}\|^{2} - \lambda \|Tw_{n_{k}} - y_{n_{k}}\| \|FTw_{n_{k}} - Fy_{n_{k}}\|$$

$$\geq (1 - \lambda L_{2}) \|Tw_{n_{k}} - y_{n_{k}}\|^{2}.$$

$$(4.4.35)$$

Since $\lambda \in (0, \frac{1}{L_2})$, we have that $1 - \lambda L_2 > 0$. Hence, from (4.4.35), (4.4.10) and (4.4.3) we

obtain

$$\begin{split} \|Tw_{n_{k}} - y_{n_{k}}\|^{2} &\leq \left(\frac{1}{1 - \lambda L_{2}}\right) \langle Tw_{n_{k}} - y_{n_{k}}, r_{n_{k}} \rangle \\ &= \left(\frac{1}{1 - \lambda L_{2}}\right) \beta_{n_{k}} \|r_{n_{k}}\|^{2} \\ &= \left(\frac{1}{1 - \lambda L_{2}}\right) \beta_{n_{k}} \|r_{n_{k}}\| \|Tw_{n_{k}} - y_{n_{k}} - \lambda(FTw_{n_{k}} - Fy_{n_{k}})\| \\ &= \left(\frac{1}{1 - \lambda L_{2}}\right) \beta_{n_{k}} \|r_{n_{k}}\| \|(Tw_{n_{k}} - y_{n_{k}}) + \lambda(Fy_{n_{k}} - FTw_{n_{k}})\| \\ &\leq \left(\frac{1}{1 - \lambda L_{2}}\right) \beta_{n_{k}} \|r_{n_{k}}\| \left(\|Tw_{n_{k}} - y_{n_{k}}\| + \lambda\|Fy_{n_{k}} - FTw_{n_{k}}\|\right) \\ &\leq \left(\frac{1}{1 - \lambda L_{2}}\right) \beta_{n_{k}} \|r_{n_{k}}\| \left(\|Tw_{n_{k}} - y_{n_{k}}\| + \lambda L_{2}\|y_{n_{k}} - Tw_{n_{k}}\|\right) \\ &= \left(\frac{1 + \lambda L_{2}}{1 - \lambda L_{2}}\right) \|Tw_{n_{k}} - y_{n_{k}}\|\beta_{n_{k}}\|r_{n_{k}}\| \\ &= \gamma_{2}^{-1} \left(\frac{1 + \lambda L_{2}}{1 - \lambda L_{2}}\right) \|Tw_{n_{k}} - y_{n_{k}}\|\|z_{n_{k}} - Tw_{n_{k}}\|, \end{split}$$

which implies from (4.4.34) that

$$\|Tw_{n_k} - y_{n_k}\| \le \gamma_2^{-1} \left(\frac{1 + \lambda L_2}{1 - \lambda L_2}\right) \|Tw_{n_k} - z_{n_k}\| \to 0, \quad \text{as} \quad k \to \infty.$$
(4.4.36)

Since $\{Tw_{n_k}\}$ converges weakly to Tz, then it follows from (4.4.36) that $\{y_{n_k}\}$ also converges weakly to Tz. Also, since $\{y_{n_k}\} \subset \mathcal{Q}$, we have that $Tz \in \mathcal{Q}$.

By the characteristic property of $P_{\mathcal{Q}}$, we obtain $\forall x \in \mathcal{Q}$ that

$$\langle Tw_{n_k} - \lambda FTw_{n_k} - y_{n_k}, x - y_{n_k} \rangle \le 0,$$

which implies

$$\frac{1}{\lambda}\langle Tw_{n_k} - y_{n_k}, x - y_{n_k} \rangle + \langle FTw_{n_k}, y_{n_k} - Tw_{n_k} \rangle \le \langle FTw_{n_k}, x - Tw_{n_k} \rangle.$$
(4.4.37)

Hence, applying (4.4.36) in (4.4.37), we obtain that

$$0 \le \liminf_{k \to \infty} \langle FTw_{n_k}, x - Tw_{n_k} \rangle, \ \forall x \in \mathcal{Q}.$$
(4.4.38)

Observe that

$$\langle Fy_{n_k}, x - y_{n_k} \rangle = \langle Fy_{n_k} - FTw_{n_k}, x - Tw_{n_k} \rangle + \langle FTw_{n_k}, x - Tw_{n_k} \rangle$$

$$+ \langle Fy_{n_k}, Tw_{n_k} - y_{n_k} \rangle.$$

$$(4.4.39)$$

Since F is Lipschitz continuous on \mathcal{H}_2 , we obtain from (4.4.36) that

$$\lim_{k \to \infty} ||FTw_{n_k} - Fy_{n_k}|| = 0.$$

Hence, from (4.4.36), (4.4.38) and (4.4.39), we obtain that

$$0 \le \liminf_{k \to \infty} \langle Fy_{n_k}, x - y_{n_k} \rangle, \ \forall x \in \mathcal{Q}.$$
(4.4.40)

Next, we show that $Tz \in VI(F, Q)$. Now, we choose a sequence $\{\delta_k\}$ of positive numbers such that $\delta_{k+1} \leq \delta_k$, $\forall k \geq 1$ and $\delta_k \to 0$ as $k \to \infty$. From (4.4.40), we denote by N_k (for each $k \geq 1$), the smallest positive integer such that

$$\langle Fy_{n_j}, x - y_{n_j} \rangle + \delta_k \ge 0, \quad \forall j \ge N_k.$$
 (4.4.41)

Since $\{\delta_k\}$ is decreasing, we have that $\{N_k\}$ is increasing. Also, since $\{y_{N_k}\} \subset \mathcal{Q}$ for all $k \geq 1$, we can suppose $Fy_{N_k} \neq 0$ (otherwise, y_{N_k} is a solution). Hence, we can set $q_{N_k} = \frac{Fy_{N_k}}{\|Fy_{N_k}\|^2}$ for each $k \geq 1$. Then, $\langle Fy_{N_k}, q_{N_k} \rangle = 1$ for each $k \geq 1$.

Therefore, from (4.4.41) we have

$$\langle Fy_{N_k}, x + \delta_k q_{N_k} - y_{N_k} \rangle \ge 0,$$

which implies from the pseudomonotonicity of F on \mathcal{H}_2 that

$$\langle F(x+\delta_k q_{N_k}), x+\delta_k q_{N_k}-y_{N_k}\rangle \ge 0.$$
(4.4.42)

This implies that

$$\langle Fx, x - y_{N_k} \rangle \ge \langle Fx - F(x + \delta_k q_{N_k}), x + \delta_k q_{N_k} - y_{N_k} \rangle - \delta_k \langle Fx, q_{N_k} \rangle.$$
(4.4.43)

Now, if FTz = 0, then $Tz \in VI(F, \mathcal{Q})$. So, we may suppose that $FTz \neq 0$. Since $\{y_{n_k}\}$ converges weakly to Tz, then by Condition (c) we obtain

$$0 < \|FTz\| \le \liminf_{k \to \infty} \|Fy_{N_k}\|.$$

Since $\{y_{n_k}\} \subset \{y_{N_k}\}$, we obtain that

$$0 \leq \limsup_{k \to \infty} \|\delta_k q_{N_k}\| = \limsup_{k \to \infty} \left(\frac{\delta_k}{\|Fy_{n_k}\|}\right) \leq \frac{\limsup_{k \to \infty} \delta_k}{\liminf_{k \to \infty} \|Fy_{n_k}\|} \leq \frac{0}{\|FTz\|} = 0$$

Therefore, $\lim_{k\to\infty} \delta_k q_{N_k} = 0$. Thus, letting $k \to \infty$ in (4.4.43), we have

$$\langle Fx, x - Tz \rangle \ge 0, \quad \forall x \in Q,$$

$$(4.4.44)$$

which implies by Lemma 2.5.8 that $Tz \in VI(F, Q)$.

Next, we show that $z \in VI(A, C)$. Following similar method of proof used in obtaining (4.4.36) and noting our hypothesis $\lim_{k\to\infty} ||b_{n_k} - t_{n_k}|| = 0$, we obtain

$$\|b_{n_k} - u_{n_k}\| \le r_1^{-1} \left(\frac{1 + \mu L_1}{1 + \mu L_1}\right) \|b_{n_k} - t_{n_k}\| \to 0, \quad \text{as} \quad k \to \infty.$$
(4.4.45)

Following similar method of proof used in obtaining (4.4.40), we obtain from (4.4.45), the characteristic property of $P_{\mathcal{C}}$ and the Lipschitz continuity of A on \mathcal{H}_1 that

$$0 \le \liminf_{k \to \infty} \langle Au_{n_k}, y - u_{n_k} \rangle, \ \forall y \in C.$$
(4.4.46)

From our hypothesis, (4.4.45) and the fact that $\{w_{n_k}\}$ converges weakly to z, we obtain that the subsequences $\{b_{n_k}\}$ and $\{u_{n_k}\}$ of $\{b_n\}$ and $\{u_n\}$ respectively, converge weakly to z. Also, since $\{u_{n_k}\} \subset C$, we have that $z \in C$. Following similar method of proof used in obtaining (4.4.44), we obtain

$$\langle Ay, y - z \rangle \ge 0, \quad \forall \ y \in C,$$
 (4.4.47)

which implies by Lemma 2.5.8, that $z \in VI(A, C)$. Hence, we conclude that $z \in \Gamma$. \Box

Lemma 4.4.9. Let $\{x_n\}$ be a sequence generated by Algorithm 4.4.2 under Assumption 4.4.1. Then, for each $n \ge 1$

$$v_n := \|x_n - p\|^2 + \alpha_n \|x_n - x_0\|^2 - \theta_{n-1} \|x_{n-1} - p\|^2 + 2\theta_n \|x_{n-1} - x_n\|^2 \ge 0.$$

Proof. Since $\{\theta_n\} \in [0, \frac{1}{3})$ is non-decreasing, we have from Lemma 2.1.1 that

$$\begin{split} v_n &= \|x_n - p\|^2 + \alpha_n \|x_n - x_0\|^2 - \theta_{n-1} \|x_{n-1} - x_n + x_n - p\|^2 + 2\theta_n \|x_{n-1} - x_n\|^2 \\ &= \|x_n - p\|^2 + \alpha_n \|x_n - x_0\|^2 - \theta_{n-1} \|x_{n-1} - x_n\|^2 - \theta_{n-1} \|x_n - p\|^2 \\ &- 2\theta_{n-1} \langle x_{n-1} - x_n, x_n - p \rangle + 2\theta_n \|x_{n-1} - x_n\|^2 - \theta_{n-1} \|x_n - p\|^2 \\ &= \|x_n - p\|^2 + \alpha_n \|x_n - x_0\|^2 - \theta_{n-1} \|x_{n-1} - x_n\|^2 - \theta_{n-1} \|x_n - p\|^2 \\ &- \theta_{n-1} \Big[\|x_{n-1} - x_n\|^2 + \|x_n - p\|^2 - \|x_{n-1} - 2x_n + p\|^2 \Big] + 2\theta_n \|x_{n-1} - x_n\|^2 \\ &= \|x_n - p\|^2 + \alpha_n \|x_n - x_0\|^2 - 2\theta_{n-1} \|x_{n-1} - x_n\|^2 - 2\theta_{n-1} \|x_n - p\|^2 \\ &+ \theta_{n-1} \|x_{n-1} - 2x_n + p\|^2 + 2\theta_n \|x_{n-1} - x_n\|^2 \\ &\geq \|x_n - p\|^2 + \alpha_n \|x_n - x_0\|^2 - 2\theta_n \|x_{n-1} - x_n\|^2 - \frac{2}{3} \|x_n - p\|^2 \\ &+ \theta_{n-1} \|x_{n-1} - 2x_n + p\|^2 + 2\theta_n \|x_{n-1} - x_n\|^2 \\ &= \frac{1}{3} \|x_n - p\|^2 + \alpha_n \|x_n - x_0\|^2 + \theta_{n-1} \|x_{n-1} - 2x_n + p\|^2 \\ &\geq \frac{1}{3} \|x_n - p\|^2 + \alpha_n \|x_n - x_0\|^2 \\ &\geq 0, \end{split}$$

which is the desired conclusion.

We are now in a position to prove the main theorem for Algorithm 4.4.2.

Theorem 4.4.10. Let $\{x_n\}$ be a sequence generated by Algorithm 4.4.2 under Assumption 4.4.1. Then, $\{x_n\}$ converges strongly to $p \in \Gamma$, where $p = P_{\Gamma}x_0$.

Proof. From Lemma 4.4.9 and (4.4.16), we obtain

$$v_{n+1} - v_n + (1 - 3\theta_{n-1} - \alpha_n) \|x_n - x_{n+1}\|^2 \le -2\alpha_n \langle x_n - p, x_n - x_0 \rangle.$$
(4.4.48)

We consider two cases for our proof.

CASE 1: Let $z \in \Gamma$. Suppose for some $n_0 \in \mathbb{N}$ large enough, we have $v_{n+1} \leq v_n$ for all $n \geq n_0$. Then by Lemma 4.4.9 we have $v_n \geq 0, \forall n \geq 1$ and $\lim_{n \to \infty} v_n = \lim_{n \to \infty} v_{n+1}$ exists. Since $\{x_n\}$ is bounded, there exists a constant M > 0 such that $2|\langle x_n - p, x_n - x_0 \rangle| \leq M$. Hence, there exists $N \in \mathbb{N}$ and $\xi_1 > 0$ such that $(1 - 3\theta_{n+1} - \alpha_n) \geq \xi_1, \forall n \geq N$. Hence, from (4.4.48) we have that for all $n \geq N$

$$\xi_1 \|x_n - x_{n+1}\|^2 \le v_n - v_{n+1} + \alpha_n M \to 0, \quad n \to \infty.$$

Thus,

$$\lim_{n \to \infty} \|x_{n+1} - x_n\| = 0. \tag{4.4.49}$$

From the definition of w_n in Step 1 and by applying (4.4.49) together with the fact that $\lim_{n\to\infty} \alpha_n = 0$, we have

$$||w_n - x_n|| = ||\alpha_n(x_0 - x_n) + \theta_n(x_n - x_{n-1})||$$

$$\leq \alpha_n ||x_0 - x_n|| + \theta_n ||x_n - x_{n-1}|| \to 0, \quad n \to \infty$$
(4.4.50)

Consequently, we have

$$||w_n - x_{n+1}|| \to 0, \quad n \to \infty.$$
 (4.4.51)

From (4.4.15), we have

$$\sigma \gamma_1^{-1} (2 - \gamma_1) \| t_n - b_n \|^2 \le \| w_n - p \|^2 - \| x_{n+1} - p \|^2$$

= $(\| w_n - p \| - \| x_{n+1} - p \|) (\| w_n - p \| + \| x_{n+1} - p)$
 $\le \| w_n - x_{n+1} \| (\| w_n - p \| + \| x_{n+1} - p)$
 $\le \| w_n - x_{n+1} \| M_1$

where $M_1 := \sup_{n \ge 1} \{ \|w_n - p\| + \|x_{n+1} - p\| \}$. Hence

$$\|t_n - b_n\| \to 0, \ n \to \infty. \tag{4.4.52}$$

Similarly, we obtain from (4.4.15) that

$$||w_n - t_n|| \to 0, \ n \to \infty.$$

Consequently, we have

$$||b_n - w_n|| \to 0, \ n \to \infty.$$
 (4.4.53)

Using the fact that $\{x_n\}$ is bounded, $\{v_n\}$ is convergent and $\lim_{n\to\infty} \alpha_n = 0$, we obtain from Lemma 4.4.9 that

$$\lambda := \lim_{n \to \infty} \left(\|x_{n+1} - p\|^2 - \theta_n \|x_n - p\|^2 \right) < \infty,$$
(4.4.54)

which is the limit of $\lim_{n\to\infty} v_{n+1}$. Consequently, from Lemma 4.4.9, we have $\lambda \ge 0$. We show that $\lambda = 0$ holds. So that it follows from Lemma 4.4.7 that the sequence $\{x_n\}$ converges strongly to the solution p.

Suppose on the contrary $\lambda > 0$. Since $\{x_n\}$ is bounded by Lemma 4.4.6, there exists a subsequence $\{x_{n_k}\}$ of $\{x_n\}$ which converges weakly to z, such that

$$\liminf_{n \to \infty} \langle x_n - p, p - x_0 \rangle = \lim_{k \to \infty} \langle x_{n_k} - p, p - x_0 \rangle = \langle z - p, p - x_0 \rangle.$$
(4.4.55)

By applying (4.4.52) and (4.4.53), it follows from Lemma 4.4.8 that $z \in \Gamma$. Since $p = P_{\Gamma} x_0$, we obtain from (4.4.55)

$$\liminf_{n \to \infty} \langle x_n - p, p - x_0 \rangle = \langle z - p, p - x_0 \rangle \ge 0, \qquad (4.4.56)$$

which follows from (4.4.55) that

$$\lim_{k \to \infty} \langle x_{n_k} - p, p - x_0 \rangle \ge 0.$$

From (4.4.54), we have

$$\liminf_{n \to \infty} \|x_{n+1} - p\|^2 \ge \lim_{n \to \infty} \left(\|x_{n+1} - p\|^2 - \theta_n \|x_n - p\|^2 \right) = \lambda,$$

and since $\lambda > 0$, we have

$$||x_{n+1} - p||^2 \ge \frac{1}{2}\lambda, \quad \forall n \ge n_1$$

for some sufficiently large $n_1 \in \mathbb{N}$. Observe that

$$\langle x_n - p, x_n - x_0 \rangle = ||x_n - p||^2 + \langle x_n - p, p - x_0 \rangle.$$

Then, by applying (4.4.56) we have

$$\liminf_{n \to \infty} \langle x_n - p, x_n - x_0 \rangle = \liminf_{n \to \infty} \left(\|x_n - p\|^2 + \langle x_n - p, p - x_0 \rangle \right)$$
$$\geq \liminf_{n \to \infty} \left(\frac{1}{2} \lambda + \langle x_n - p, p - x_0 \rangle \right)$$
$$= \frac{1}{2} \lambda + \liminf_{n \to \infty} \langle x_n - p, p - x_0 \rangle$$
$$\geq \frac{1}{2} \lambda.$$

Again, using the assumption that $\lambda > 0$, we have

$$\langle x_n - p, x_n - x_0 \rangle \ge \frac{1}{4}\lambda, \ \forall n \ge n_2,$$

for some sufficiently large $n_2 \in \mathbb{N}$ such that $n_2 \geq n_1$. From (4.4.48), we have

$$v_{n+1} - v_n \le -\frac{1}{2}\alpha\lambda, \ \forall n \ge n_2.$$

Applying Lemma 4.4.9, it follows from the last inequality that

$$\frac{1}{2}\lambda \sum_{k=n_2}^n \alpha_k \le v_{n_2} - v_n \le v_{n_2}, \ \forall n \ge n_2.$$

Since $\lambda > 0$, this gives the summability of the sequence $\{\alpha_n\}$ which contradicts $\sum_{n=1}^{\infty} \alpha_n = \infty$. Therefore, we must have $\lambda = 0$, and it follows that the sequence $\{x_n\}$ converges strongly to $p = P_{\Gamma} x_0$ as required.

CASE 2: Suppose that $\{v_n\}$ is not monotonically decreasing. Let $\tau : \mathbb{N} \to \mathbb{N}$ be defined for all $n \ge n_0$ for some $n_0 \in \mathbb{N}$ large enough by

$$\tau_{(n)} := \max\{k \in \mathbb{N} : k \le n, v_k \le v_{k+1}\}.$$

Observe that $\tau_{(n)}$ is a non-decreasing sequence such that $\tau_{(n)} \to \infty$ as $n \to \infty$ and $v_{\tau_{(n)}} \le v_{\tau_{(n)}+1}$ for all $n \ge n_0$. Similar to **CASE 1**, for some constant M > 0, we obtain from (4.4.48) that

$$\xi_1 \| x_{\tau_{(n)+1}} - x_{\tau_{(n)}} \| \le \alpha_{\tau(n)} M \to 0.$$
(4.4.57)

Consequently, we get

$$\|x_{\tau_{(n)+1}} - x_{\tau_{(n)}}\| \to 0, \ n \to \infty.$$
(4.4.58)

Also, following similar procedure as in CASE 1, we obtain

$$\begin{aligned} \|x_{\tau_{(n)}} - w_{\tau_{(n)}}\| &\to 0, \ n \to \infty. \\ \|t_{\tau_{(n)}} - b_{\tau_{(n)}}\| &\to 0, \ n \to \infty. \\ \|w_{\tau_{(n)}} - t_{\tau_{(n)}}\| &\to 0, \ n \to \infty. \\ \|b_{\tau_{(n)}} - w_{\tau_{(n)}}\| &\to 0, \ n \to \infty. \end{aligned}$$

$$(4.4.59)$$

Observe from (4.4.48) that for $j \ge 0$, we have $v_{j+1} < v_j$ when $x_j \notin \Omega := \{x \in \mathcal{H} : \langle x - x_0, x - p \rangle \le 0\}$. Since $v_{\tau_{(n)}} \le v_{\tau_{(n)}+1}$, we have that $x_{\tau_{(n)}} \in \Omega \forall n \ge n_0$. We have from

Lemma 4.4.6 that $\{x_{\tau_{(n)}}\}$ is bounded, hence there exists a subsequence, again say $\{x_{\tau_{(n)}}\}\$ which converges weakly to some $z \in \mathcal{H}_1$. Since Ω is a closed and convex set, then it is weakly closed and it follows that $z \in \Omega$. By (4.4.59), it follows from Lemma 4.4.8 that $z \in \Gamma$. Hence, we have $z \in \Omega \cap \Gamma$. In view of Lemma 2.5.18, we know that $\Omega \cap \Gamma$ contains only p as its element. Consequently, we have z = p. Moreover, since $x_{\tau_n} \in \Omega$ we have

$$||x_{\tau_{(n)}} - p||^2 = \langle x_{\tau_{(n)}} - x_0, x_{\tau_{(n)}} - p \rangle - \langle p - x_0, x_{\tau_{(n)}} - p \rangle$$

$$\leq -\langle p - x_0, x_{\tau_{(n)}} - p \rangle.$$

Taking the lim sup of the above inequality, we get

$$\limsup_{n \to \infty} \|x_{\tau_{(n)}} - p\| \le 0.$$

Thus,

$$||x_{\tau_{(n)}} - p|| \to 0, \ n \to \infty.$$
 (4.4.60)

We claim that this implies $\lim_{n\to\infty} v_{\tau_{(n)}+1} = 0$. From the definition of $v_{\tau_{(n)+1}}$, we have

$$\begin{aligned} v_{\tau_{(n)+1}} &= \|x_{\tau_{(n)+1}} - p\|^2 + \alpha_{\tau_{(n)+1}} \|x_{\tau_{(n)+1}} - x_0\|^2 - \theta_{\tau_{(n)}} \|x_{\tau_{(n)}} - p\|^2 \\ &+ 2\theta_{\tau_{(n)+1}} \|x_{\tau_{(n)+1}} - x_{\tau_{(n)}}\|^2 \\ &= \|x_{\tau_{(n)+1}} - x_{\tau_{(n)}} + x_{\tau_{(n)}} - p\|^2 + \alpha_{\tau_{(n)+1}} \|x_{\tau_{(n)+1}} - x_0\|^2 - \theta_{\tau_{(n)}} \|x_{\tau_{(n)}} - p\|^2 \\ &+ 2\theta_{\tau_{(n)+1}} \|x_{\tau_{(n)+1}} - x_{\tau_{(n)}}\|^2. \end{aligned}$$

Using (4.4.58), (4.4.60), the boundedness of $\{\theta_n\}$ and $\{x_n\}$ and the fact that $\lim_{n \to \infty} \alpha_n = 0$, we obtain that $\lim_{n \to \infty} v_{\tau_{(n)+1}} = 0$.

Next, we show that $\lim_{n\to\infty} v_n = 0$. Observe that for all $n \ge n_0$, we have $v_{\tau_{(n)}} \le v_{\tau_{(n)+1}}$ if $n \ne \tau(n)$ since $v_j > v_{j+1}$ for $\tau(n) + 1 \le j \le n - 1$. It follows that $\forall n \ge n_0$, we have

$$v_n \le \max\{v_{\tau_{(n)}}, v_{\tau_{(n)+1}}\} = v_{\tau_{(n)+1}} \to 0.$$

Hence,

$$\limsup_{n \to \infty} v_n \le 0.$$

From Lemma 4.4.9, we have that

$$\liminf_{n \to \infty} v_n \ge 0.$$

Thus,

 $\lim_{n \to \infty} v_n = 0.$
Using the fact that $\{x_n\}$ is bounded, $\lim_{n\to\infty} \alpha_n = 0$ and by (4.4.48), we have

$$||x_n - x_{n+1}|| \to 0, \ n \to \infty,$$

which implies from the definition of v_n that

$$\lim_{n \to \infty} \left(\|x_{n+1} - p\|^2 - \theta_n \|x_n - p\|^2 \right) = 0.$$

Thus, by Lemma 4.4.7 we obtain that $\{x_n\}$ converges strongly to $p = P_{\Gamma} x_0$, which completes the proof.

Remark 4.4.11. [192]

- Setting $\mathcal{H}_1 = \mathcal{H}_2 = \mathcal{H}$, F = 0 and $T = I_{\mathcal{H}}$ (the identity operator on \mathcal{H}) in Theorem 4.4.10, we obtain an inertial projection and contraction method requiring only one projection onto the feasible set \mathcal{C} per iteration with fixed step size for solving the classical VIP (1.2.4) when A is pseudomonotone and Lipschitz continuous as a corollary.
- The conclusions of Lemma 4.4.4, Lemma 4.4.8 and Theorem 4.4.10 still hold even if $\mu \in (0, \frac{1}{L_1})$ and $\lambda \in (0, \frac{1}{L_2})$ in Algorithm 4.4.2 are replaced with variable step sizes μ_n and λ_n , respectively such that

$$0 < \inf_{n \ge 1} \mu_n \le \sup_{n \ge 1} \mu_n < \frac{1}{L_1}$$
 and $0 < \inf_{n \ge 1} \lambda_n \le \sup_{n \ge 1} \lambda_n < \frac{1}{L_2}$.

For the convergence analysis of Algorithm 4.4.3, which does not require the Lipschitz constants of the underlying cost operators to be known, we first state the following lemma on the step size rules derived from [162]. The proof of the lemma is similar to the method of proof in [162]. Hence, we omit the proof here.

Lemma 4.4.12. Let $\{\lambda_n\}$ and $\{\mu_n\}$ be the sequences generated by (4.4.1) and (4.4.2), respectively. Then the sequences $\{\lambda_n\}$ and $\{\mu_n\}$ are well defined, and $\lim_{n\to\infty} \lambda_n = \lambda$, $\lim_{n\to\infty} \mu_n = \mu$, where $\lambda \in \left[\min\left\{\frac{a_2}{L_2}, \lambda_1\right\}, \lambda_1 + \Phi\right], \mu \in \left[\min\left\{\frac{a_1}{L_1}, \mu_1\right\}, \mu_1 + \Psi\right], \text{ and } \Phi = \sum_{n=1}^{\infty} \phi_n, \Psi = \sum_{n=1}^{\infty} \psi_n.$

Lemma 4.4.13. Let $\{x_n\}$ be a sequence generated by Algorithm 4.4.3 under Assumption 4.4.1. Then,

$$||Tw_n - y_n|| \le \gamma_2^{-1} \left(\frac{\lambda_{n+1} + \lambda_n a_2}{\lambda_{n+1} - \lambda_n a_2}\right) ||Tw_n - z_n||, \ \forall n \ge 1$$
(4.4.61)

and

$$||b_n - u_n|| \le \gamma_1^{-1} \left(\frac{\mu_{n+1} + \mu_n a_1}{\mu_{n+1} - \mu_n a_1}\right) ||b_n - t_n||, \ \forall n \ge 1.$$
(4.4.62)

Proof. From (4.4.1), we have that

$$||FTw_n - Fy_n|| \le \frac{a_2}{\lambda_{n+1}} ||Tw_n - y_n||, \ \forall n \ge 1,$$
(4.4.63)

holds for both $FTw_n = Fy_n$ and $FTw_n \neq Fy_n$. Similar to (4.4.35), we obtain

$$\langle Tw_n - y_n, r_n \rangle \ge ||Tw_n - y_n||^2 - \lambda_n ||Tw_n - y_n|| ||FTw_n - Fy_n||$$

 $\ge (1 - \lambda_n \frac{a_2}{\lambda_{n+1}}) ||Tw_n - y_n||^2,$

which implies that

$$\begin{aligned} \|Tw_{n} - y_{n}\|^{2} &\leq \frac{1}{\left(1 - \lambda_{n} \frac{a_{2}}{\lambda_{n+1}}\right)} \langle Tw_{n} - y_{n}, r_{n} \rangle \\ &\leq \frac{1}{\left(1 - \lambda_{n} \frac{a_{2}}{\lambda_{n+1}}\right)} \beta_{n} \|r_{n}\| \left(\|Tw_{n} - y_{n}\| + \lambda_{n} \frac{a_{2}}{\lambda_{n+1}}\|Tw_{n} - y_{n}\|\right) \\ &= \left(\frac{1 + \frac{\lambda_{na_{2}}}{\lambda_{n+1}}}{1 - \frac{\lambda_{na_{2}}}{\lambda_{n+1}}}\right) \beta_{n} \|r_{n}\| \|Tw_{n} - y_{n}\| \\ &= \gamma_{2}^{-1} \left(\frac{\lambda_{n+1} + \lambda_{n}a_{2}}{\lambda_{n+1} - \lambda_{n}a_{2}}\right) \|Tw_{n} - z_{n}\| \|Tw_{n} - y_{n}\|, \end{aligned}$$

which reduces to (4.4.61) when simplified further. In a similar manner, we get (4.4.62).

Remark 4.4.14. Replacing μ and λ with μ_n and λ_n , respectively in Lemma 4.4.4, we obtain that $\{x_n\}$ is bounded.

By Lemma 4.4.12, it follows that

$$\lim_{n \to \infty} \left(\frac{\lambda_{n+1} + \lambda_n a_2}{\lambda_{n+1} - \lambda_n a_2} \right) = \frac{1 + a_2}{1 - a_2}$$
(4.4.64)

and

$$\lim_{n \to \infty} \left(\frac{\mu_{n+1} + \mu_n a_1}{\mu_{n+1} - \mu_n a_1} \right) = \frac{1 + a_1}{1 - a_1}.$$
(4.4.65)

Following the similar procedure used in Theorem 4.4.10, we obtain the following strong convergence theorem for Algorithm 4.4.3.

Theorem 4.4.15. Let $\{x_n\}$ be a sequence generated by Algorithm 4.4.3 under Assumption 4.4.1. Then, $\{x_n\}$ converges strongly to $p \in \Gamma$, where $p = P_{\Gamma}x_0$.

Remark 4.4.16.

- The method of proof in this work is different from the method of proof used in obtaining strong convergence for SVIPs.
- Similar to Remark 4.4.11, we obtain an inertial projection and contraction method with adaptive step size, requiring only one projection onto the feasible set C per iteration, for solving the classical VIP (1.2.4) when A is pseudomonotone and Lipschitz continuous as a corollary.
- When the operators A and F are monotone and Lipschitz continuous, we do not need them to satisfy Condition (c). This is because Condition (c) was only used after (4.4.37) to get the conclusion of Lemma 4.4.8. But, from (4.4.37) and the monotonicity of F, we obtain

$$0 \leq \langle FTw_{n_{k}}, x - Tw_{n_{k}} \rangle + \frac{1}{\lambda} \langle y_{n_{k}} - Tw_{n_{k}}, x - y_{n_{k}} \rangle + \langle FTw_{n_{k}}, Tw_{n_{k}} - y_{n_{k}} \rangle$$

$$\leq (\langle FTw_{n_{k}} - Fx, x - Tw_{n_{k}} \rangle + \langle Fx, x - Tw_{n_{k}} \rangle) + \frac{1}{\lambda} ||y_{n_{k}} - Tw_{n_{k}}|| ||x - y_{n_{k}}||$$

$$+ ||FTw_{n_{k}}|| ||Tw_{n_{k}} - y_{n_{k}}||$$

$$\leq \langle Fx, x - Tw_{n_{k}} \rangle + \frac{1}{\lambda} ||y_{n_{k}} - Tw_{n_{k}}|| ||x - y_{n_{k}}|| + ||FTw_{n_{k}}|| ||Tw_{n_{k}} - y_{n_{k}}||, \forall x \in \mathcal{Q}$$

Passing limit as $k \to \infty$, noting that $\{Tw_{n_k}\}$ converges weakly to Tz and applying (4.4.36), it follows from the last inequality that

$$\langle Fx, x - Tz \rangle \ge 0, \quad \forall x \in \mathcal{Q}.$$

Consequently, by Lemma 2.5.8 we have that $Tz \in VI(F, Q)$. Similarly, we obtain that $z \in VI(A, C)$. Hence, we conclude that Lemma 4.4.8 holds.

• Theorem 4.4.10 and Theorem 4.4.15 are still true if the operators A and F in finite dimensional spaces are only required to be pseudomonotone and Lipschitz continuous which is an improvement over the results in literature since no product space formulation is required even with the relaxed pseudomonotonicity assumption.

4.4.3 Numerical experiments

In this section, using some test examples, we discuss the numerical behavior of our methods, Algorithm 4.4.2 and Algorithm 4.4.3, as well as compare them with the methods of Tian and Jiang [243] (Algorithm (2.5.11)), Tian and Jiang [244] (Algorithm (2.5.13)), Pham et. al [206] (see Appendix 4.3.16) and Ogwo et. al [192] (see Algorithm 4.3.3).

In our computations, we randomly choose $x_0, x_1 \in \mathcal{H}_1, \gamma_1 = 1.8, \gamma_2 = 1.1, a_1 = 0.6, a_2 = 0.4, \lambda_1 = 0.85$ and $\mu_1 = 0.9$. We choose $\alpha_n = \frac{1}{n+1}, \theta_n = 0.29, \sigma = 0.45, \phi_n = \frac{1}{(n+1)^2}, \psi_n = \frac{1}{(n+2)^2}$ in Algorithm 4.4.2 and Algorithm 4.4.3. Also, we choose $\delta_n = \frac{1}{n+1}, \theta_n = \frac{1}{2} - \delta_n, \alpha_n = \bar{\alpha}_n, \tau_n = \frac{\delta_n}{n^{0.01}}$ and $\alpha = 3$ in the method of Ogwo *et al.* [192, Algorithm 3.3]. Using

MATLAB 2019(b) and the stopping criterion $||x_{n+1}-x_n|| < 10^{-3}$, we plot the graphs of errors against the number of iterations in each case. The numerical results are reported in Figure 4.7, Figure 4.8, Table 4.4.1 and Table 4.4.2.

Example 4.4.17. Let $\mathcal{H}_1 = \mathcal{H}_2 = L_2([0, 2\pi])$ be equipped with inner product

$$\langle x, y \rangle = \int_0^{2\pi} x(t)y(t)dt, \quad \forall \ x, y \in L_2([0, 2\pi])$$

and norm

$$||x|| := \left(\int_0^{2\pi} |x(t)|^2 dt\right)^{\frac{1}{2}}, \quad \forall \ x, y \in L_2([0, 2\pi]).$$

Then we define $A: L_2([0, 2\pi]) \to L_2([0, 2\pi])$ by

$$A(x)(t) = e^{-\|x\|} \int_0^t x(s) ds, \quad \forall x \in L_2([0, 2\pi]), \quad t \in [0, 2\pi].$$

From [250], we have that A is pseudomonotone and Lipschitz continuous but not monotone on $L_2([0, 1])$.

Let $C = \{x \in L_2([0, 2\pi]) : \langle y, x \rangle \leq v\}$, where $y = t + e^t$ and v = 1, then C is a nonempty closed and convex subset of $L_2([0, 2\pi])$. We define the metric projection P_C as:

$$P_{\mathcal{C}}(x) = \begin{cases} x - \frac{\langle y, x \rangle - v}{||y||^2} y, & \text{if } \langle y, x \rangle > v, \\ x, & \text{if } \langle y, x \rangle \le v. \end{cases}$$

Also, let $\mathcal{Q} = \{x \in L_2([0, 2\pi]) : ||x - a||_{l_2} \leq d\}$, where a = t + 3 and d = 2, then \mathcal{Q} is a nonempty closed and convex subset of $L_2([0, 2\pi])$. We define $P_{\mathcal{Q}}$ as:

$$P_{\mathcal{Q}}(x) = \begin{cases} x, & \text{if } x \in \mathcal{Q}, \\ \frac{x-a}{\|x-a\|^2}d + a, & \text{otherwise.} \end{cases}$$

We define the operator $F: \mathcal{Q} \to L_2([0, 2\pi])$ by

$$F(x)(t) := \mathcal{G}(x)\mathcal{M}(x)(t), \quad \forall \ x \in \mathcal{Q}, \quad t \in [0, 2\pi],$$

where $\mathcal{G}: \mathcal{Q} \to \mathbb{R}$ is defined by $g(x) := \frac{1}{1+\|x\|^2}$ and $\mathcal{M}: L_2([0,2\pi]) \to L_2([0,2\pi])$ is defined by $\mathcal{M}(x)(t) := \int_0^t x(s) ds$, $\forall x \in L_2([0,2\pi])$, $t \in [0,2\pi]$. We have that \mathcal{G} is $\frac{16}{25}$ -Lipschitz continuous and $\frac{1}{5} \leq \mathcal{G}(x) \leq 1$, $\forall x \in \mathcal{C}$ (see [222]). Hence, from [222], we have that F is pseudomonotone and Lipschitz continuous but not monotone since \mathcal{M} is a Volterra intergral mapping which is bounded and linear monotone.

Let $T: L_2([0, 2\pi]) \to L_2([0, 2\pi])$ be defined by

$$Tx(s) = \int_0^{2\pi} \mathcal{K}(s,t)x(t)dt, \ \forall \ x \in L_2([0,2\pi]),$$

where \mathcal{K} is a continuous real-valued function on $[0, 2\pi] \times [0, 2\pi]$. Then, T is a bounded linear operator with adjoint

$$T^*x(s) = \int_0^{2\pi} \mathcal{K}(t, s) x(t) dt, \ \forall \ x \in L_2([0, 2\pi]).$$

In particular, we define $\mathcal{K}(s,t) = e^{-st}$ for all $s,t \in [0,2\pi]$. For Algorithms (2.5.11) and (2.5.13), we define the mapping $S: L_2([0,2\pi]) \to L_2([0,2\pi])$ by

$$Sx(t) = \int_0^{2\pi} tx(s)ds, \quad t \in [0, 1].$$

Then, S is nonexpansive. For Algorithm (2.5.13), we define $h: L_2([0, 2\pi]) \to L_2([0, 2\pi])$ by

$$hx(t) = \int_0^{2\pi} \frac{t}{2} x(s) ds, \quad x \in [0, 1].$$

Then, h is a contraction mapping.

We consider the following cases for the numerical experiments of this example. Case 1: Take $x_0(t) = t + 2$ and $x_1(t) = 0.7e^{-t}$.

Case 2: Take $x_0(t) = 2t + 1$ and $x_1(t) = e^{-3t}$.

Case 3: Take
$$x_0(t) = 2t + 1$$
 and $x_1(t) = e^{-t}$.

Case 4: Take $x_0(t) = t^2 + 2t + 1$ and $x_1(t) = e^{-3t}$.

		Alg.	Alg.	App.	App.	Alg.	Alg.
		(2.5.11)	(2.5.13)	4.3.16	4.3.3	4.4.2	4.4.3
Case 1	No. of Iter.	2	13	19	6	4	4
	CPU time	0.3973	1.2733	3.4674	8.6063	6.1012	4.0227
	(sec)						
Case 2	No. of Iter.	2	12	19	6	4	4
	CPU time	0.3821	1.3016	3.3446	8.5436	4.9278	3.7949
	(sec)						
Case 3	No. of Iter.	2	12	19	6	4	4
	CPU time	0.4129	1.2087	3.5745	8.7931	8.9403	7.7985
	(sec)						
Case 4	No. of Iter.	2	13	19	6	4	4
	CPU time	0.3821	1.3016	3.6063	8.7372	11.3910	11.9466
	(sec)						

Example 4.4.18. Let $\mathcal{H}_1 = (l_2(\mathbb{R}), ||.||_{l_2}) = \mathcal{H}_2$, where $l_2(\mathbb{R}) := \{x = (x_1, x_2, x_3, \dots), x_i \in \mathbb{R} : \sum_{i=1}^{\infty} |x_i|^2 < \infty\}$ and $||x||_{l_2} := \left(\sum_{i=1}^{\infty} |x_i|^2\right)^{\frac{1}{2}}, \quad \forall x \in l_2(\mathbb{R}).$

Define the operators $A, F : l_2(\mathbb{R}) \to l_2(\mathbb{R})$ by $A(x_1, x_2, x_3, \dots) = (3x_1e^{-x_1^2}, 0, 0, \dots)$ and $F(x_1, x_2, x_3, \dots) = (7x_1e^{-x_1^2}, 0, 0, \dots)$ respectively. Then, A, F are pseudo-monotone, Lipschitz continuous and sequentially weakly continuous but not monotone. Let $T : l_2(\mathbb{R}) \to l_2(\mathbb{R})$ be defined by $Tx = (0, x_1, \frac{x_2}{2}, \frac{x_3}{3}, \dots)$, for all $x \in l_2(\mathbb{R})$. Then, T is a bounded linear operator on $\ell_2(\mathbb{R})$ with adjoint $T^*y = (y_2, \frac{y_3}{2}, \frac{y_4}{3}, \dots)$ for all $y \in l_2(\mathbb{R})$. Now, define $\mathcal{C} = \mathcal{Q} = \{x \in l_2(\mathbb{R}) : ||x - a||_{l_2} \leq b\}$, where $a = (1, \frac{1}{2}, \frac{1}{3}, \cdots)$ and b = 3 for \mathcal{C}



Figure 4.7: Top left: Case 1; Top right: Case 2; Bottom left: Case 3; Bottom right: Case 4.

and $a = (\frac{1}{2}, \frac{1}{4}, \frac{1}{8}, \cdots), b = 1$ for \mathcal{Q} . Then \mathcal{C}, \mathcal{Q} are nonempty closed and convex subsets of $l_2(\mathbb{R})$. Thus,

$$P_{\mathcal{C}}(x) = \begin{cases} x, & \text{if } ||x - y||_{\ell_2} \le b, \\ \frac{x - y}{||x - y||_{\ell_2}}b + a, & \text{otherwise.} \end{cases}$$

Furthermore, we define the mappings $S, h : l_2(\mathbb{R}) \to l_2(\mathbb{R})$ by $Sx = -4(x_1, x_2, x_3, ...)$ and $hx = \left(\frac{x_1}{2}, \frac{x_2}{2}, \frac{x_3}{2}, \cdots\right)$ for all $x \in l_2(\mathbb{R})$, and consider the following cases for the starting point:

Case 1: Take $x_0 = (\frac{1}{5}, \frac{1}{15}, \frac{1}{45}, \cdots)$ and $x_1 = (1, \frac{1}{2}, \frac{1}{4}, \cdots)$. **Case 2:** Take $x_0 = (\frac{2}{5}, \frac{2}{15}, \frac{2}{45}, \cdots)$ and $x_1 = (2, 1, \frac{1}{2}, \cdots)$. **Case 3:** Take $x_0 = (\frac{1}{5}, \frac{1}{15}, \frac{1}{45}, \cdots)$ and $x_1 = (\frac{1}{2}, \frac{1}{4}, \frac{1}{8}, \cdots)$. **Case 4:** Take $x_0 = (-3, \frac{3}{2}, -\frac{3}{4}, \cdots)$ and $x_1 = (\frac{1}{3}, \frac{2}{9}, \frac{4}{27}, \cdots)$.

		Alg.	Alg.	App.	App.	Alg.	Alg.
		(2.5.11)	(2.5.13)	4.3.16	4.3.3	4.4.2	4.4.3
Case 1	No. of Iter.	2	6	14	7	4	4
	CPU time	0.0350	1.1215	0.0438	0.0512	0.1102	0.00186
	(sec)						
Case 2	No. of Iter.	2	8	14	7	4	4
	CPU time	0.0093	0.7225	0.0331	0.0393	0.1011	0.0104
	(sec)						
Case 3	No. of Iter.	2	6	14	6	4	4
	CPU time	0.0150	0.7402	0.0332	0.0378	0.1037	0.0119
	(sec)						
Case 4	No. of Iter.	2	8	14	7	4	4
	CPU time	0.0091	0.7182	0.0333	0.0347	0.1051	0.0169
	(sec)						

Table 4.4.2 Numerical results for Example 4.4.18

4.5 On a class of generalized variational inequality problem

The purpose of this study is to propose an inertial extrapolation method for solving a certain class of VIP more general than the classical VIP in real Hilbert spaces. Our proposed method is of viscosity-type and converges strongly to a solution of the aforementioned problem when the underlying/cost operator is pseudo-monotone and uniformly continuous; this makes our method to be potentially more applicable than most existing methods in the literature. To support our results numerically, we considered some examples in both finite and infinite dimensional Hilbert spaces and compared our results with other existing results in the literature.



Figure 4.8: Top left: Case 1; Top right: Case 2; Bottom left: Case 3; Bottom right: Case 4.

4.5.1 Main results

In this section, we present and study the proposed method for solving Problem (1.2.6). We begin with the following conditions: \mathcal{H}_1 and \mathcal{H}_2 are two real Hilbert spaces, $T : \mathcal{H}_1 \to \mathcal{H}_2$ is a bounded linear operator, $S : \mathcal{H}_2 \to \mathcal{H}_2$ is a κ -strictly pseudocontractive mapping with $\kappa \in [0, 1)$, and $g : \mathcal{H}_1 \to \mathcal{H}_1$ is a contraction mapping with constant $\rho \in (0, 1)$. Furthermore, we make the following assumptions:

Assumption 4.5.1. Suppose that the following conditions holds:

- (1) a. The set C is a nonempty, closed and convex subset of \mathcal{H}_1 .
 - b. $A : \mathcal{H}_1 \to \mathcal{H}_1$ is pseudo-monotone, uniformly continuous and sequentially weakly continuous on bounded subsets of C.
 - c. The solution set $\Gamma := \{z \in VI(\mathcal{C}, A) : Tz \in F(S)\}$ is nonempty.
- (2). $\{\gamma_n\}_{n=1}^{\infty}$ and $\{\epsilon_n\}_{n=1}^{\infty}$ are positive sequences satisfying the following conditions:
 - a. $\gamma_n \in (0, 1), \quad \lim_{n \to \infty} \gamma_n = 0, \quad \sum_{n=1}^{\infty} \gamma_n = \infty.$ b. $\lim_{n \to \infty} \frac{\epsilon_n}{\gamma_n} = 0.$

Algorithm 4.5.2.

Initialization: Choose sequences $\{\gamma_n\}_{n=1}^{\infty}, \{\delta_n\}_{n=1}^{\infty}$ and $\{\varepsilon_n\}_{n=1}^{\infty}$ such that the condition from Assumption 3.1(2) hold and let $\gamma, \sigma \in (0, 1), \delta \in [0, 1)$ and $x_0, x_1 \in \mathcal{H}_1$ be given arbitrarily. Set n := 1.

Iterative Steps: Step 1. Given the iterates x_{n-1} and x_n $(n \ge 1)$, choose δ_n such that $0 \le \delta_n \le \overline{\delta}_n$, where

$$\bar{\delta}_n = \begin{cases} \min\{\delta, \frac{\varepsilon_n}{\|x_n - x_{n-1}\|}\}, & if \ x_n \neq x_{n-1} \\ \delta, \ otherwise \end{cases}$$
(4.5.1)

Step 2. Set $w_n := x_n + \delta_n(x_n - x_{n-1})$ and compute

$$u_n = w_n - \tau_n T^* (I - S_\beta) T w_n, \tag{4.5.2}$$

where S_{β} is as defined in Lemma 2.1.8.

Compute $z_n := P_{\mathcal{C}}(u_n - A(u_n))$. If $u_n - z_n = 0$, then stop. Otherwise go to **Step 3**. **Step 3.** Compute $y_n = u_n - \gamma^{m_n}(u_n - z_n)$, where m_n is the smallest nonnegative integer satisfying

$$\langle A(y_n), u_n - z_n \rangle \ge \frac{\sigma}{2} ||u_n - z_n||^2.$$
 (4.5.3)

Set $\eta_n := \gamma^{m_n}$. Step 4. Compute

$$x_{n+1} = \gamma_n g(x_n) + (1 - \gamma_n) P_{\mathcal{C}_n}(u_n), \qquad (4.5.4)$$

where $C_n := \{x \in H_1 : f_n(x) \le 0\}$ and

$$f_n(x) := \langle A(y_n), x - y_n \rangle. \tag{4.5.5}$$

Set n := n + 1 and go back to Step 1.

Remark 4.5.1. If $\{y_n\}$ is bounded, then f_n is Lipschitz continuous. Indeed, if $\{y_n\}$ is bounded, then there exists $\overline{M} > 0$ such that $||A(y_n)|| \leq \overline{M}$ for all $n \geq 1$. Thus, for each $x, y \in \mathcal{C}$, we obtain

$$||f_n(x) - f_n(y)|| = ||\langle A(y_n), x - y_n \rangle - \langle A(y_n), y - y_n \rangle ||$$

= ||\langle A(y_n), x - y \rangle ||
$$\leq ||A(y_n)|| ||x - y||$$

$$\leq \bar{M} ||x - y||.$$

Therefore, f_n is Lipschitz continuous.

Lemma 4.5.2. Suppose that Assumption 4.5.1 holds. Then, the linesearch (4.5.3) is well defined.

Proof. Suppose $u_n = z_n$, then the method terminates at the solution of the GVIP (1.2.6). Let $u_n \neq z_n$, we prove by contradiction by assuming that the contrast of (4.5.3) holds, that is, for every nonnegative integer n

$$\langle A(y_n), u_n - z_n \rangle < \frac{\sigma}{2} \|u_n - z_n\|^2,$$

where $y_n = u_n - \eta_n(u_n - z_n)$. Since $y_n \to u_n$ as $n \to \infty$ and A is continuous, it follows from taking the limits as $n \to \infty$ that

$$\langle A(u_n), u_n - z_n \rangle \le \frac{\sigma}{2} ||u_n - z_n||^2.$$
 (4.5.6)

Since $z_n = P_{\mathcal{C}}(u_n - A(u_n))$, we have

$$\langle u_n - A(u_n) - z_n, u - z_n \rangle \le 0, \ \forall \ u \in \mathcal{C}.$$

Choose $u = u_n$ then, we have

$$||u_n - z_n||^2 \le \langle A(u_n), u_n - z_n \rangle.$$
 (4.5.7)

Combining (4.5.6) and (4.5.7), we obtain

$$||u_n - z_n||^2 \le \frac{\sigma}{2} ||u_n - z_n||^2$$

which contradicts the fact that $u_n \neq z_n$. Thus, the linesearch (4.5.3) is well defined. \Box

Lemma 4.5.3. Let $\{x_n\}$ be a sequence generated by Algorithm 4.5.2 such that Assumption 4.5.1 hold, then we have that $\{x_n\}$ is bounded.

Proof. Let $x^* \in \Gamma$, then from Lemma 2.1.10 (i),(ii), Lemma 2.1.8 and Lemma 2.1.9, we obtain that $I - \tau_n T^*(I - S_\beta)T$ is $\tau_n ||T||^2$ -average. That is, $I - \tau_n T^*(I - S_\beta)T = (1 - \beta_n)I + \beta_n S_n$, $\forall n \ge 1$, where $\beta_n = \tau_n ||T||^2$ and S_n is nonexpansive for all $n \ge 1$. Clearly, $\beta_n \in (0, 1)$. Therefore, we can rewrite u_n from (4.5.2) as

$$u_n = (1 - \beta_n)w_n + \beta_n S_n w_n, \ n \ge 1.$$
(4.5.8)

Hence, for any $x^* \in \Gamma$, we obtain that

$$\begin{aligned} \|u_n - x^*\|^2 &\leq (1 - \beta_n) \|w_n - x^*\|^2 + \beta_n \|S_n w_n - x^*\|^2 - \beta_n (1 - \beta_n) \|w_n - S_n w_n\|^2 \\ &\leq \|w_n - x^*\|^2 - \beta_n (1 - \beta_n) \|w_n - S_n w_n\|^2 \\ &\leq \|w_n - x^*\|^2. \end{aligned}$$
(4.5.9)

Thus, from (4.5.4) and (4.5.9), we obtain that

$$\begin{aligned} \|x_{n+1} - x^*\| &\leq \gamma_n \|g(x_n) - x^*\| + (1 - \gamma_n) \|P_{C_n}(u_n) - x^*\| \\ &\leq \gamma_n \|g(x_n) - x^*\| + (1 - \gamma_n) \|u_n - x^*\| \\ &\leq \gamma_n \|g(x_n) - x^*\| + (1 - \gamma_n) \|w_n - x^*\| \\ &\leq \gamma_n \|g(x_n) - x^*\| + (1 - \gamma_n) (\|x_n - x^*\| + \delta_n \|x_n - x_{n-1}\|) \\ &\leq (1 - \gamma_n) \|x_n - x^*\| + \gamma_n \rho \|x_n - x^*\| + \gamma_n \|g(x^*) - x^*\| \\ &+ (1 - \gamma_n) \delta_n \|x_n - x_{n-1}\| \\ &\leq (1 - \gamma_n (1 - \rho)) \|x_n - x^*\| + \gamma_n \|g(x^*) - x^*\| + \delta_n \|x_n - x_{n-1}\|. \end{aligned}$$
(4.5.10)

Now observe from (4.5.1) that

$$\delta_n \|x_n - x_{n-1}\| \le \epsilon_n \quad \forall n \ge 1,$$

which implies that

$$\frac{\delta_n}{\gamma_n} \|x_n - x_{n-1}\| \le \frac{\epsilon_n}{\gamma_n} \to 0, \quad \text{as} \quad n \to \infty.$$
(4.5.11)

Hence, there exists M > 0 such that

$$\frac{\delta_n}{\gamma_n} \|x_n - x_{n-1}\| \le M, \quad \forall n \ge 1.$$

Therefore, (4.5.10) becomes

$$\begin{aligned} \|x_{n+1} - x^*\| &\leq (1 - \gamma_n (1 - \rho)) \|x_n - x^*\| + \gamma_n (\|g(x^*) - x^*\| + M) \\ &\leq \max \left\{ \|x_n - x^*\|, \frac{\|g(x^*) - x\| + M}{1 - \rho} \right\} \\ &\vdots \\ &\leq \max \left\{ \|x_1 - x^*\|, \frac{\|g(x^*) - x\| + M}{1 - \rho} \right\}. \end{aligned}$$

Therefore, $\{x_n\}$ is bounded.

Lemma 4.5.4. Let $\{x_n\}$ be a sequence generated by Algorithm 4.5.2 such that Assumption **4.5.1** holds. If $\lim_{n \to \infty} ||P_{\mathcal{C}_n}(u_n) - u_n|| = 0$, then

- (i) $\lim_{n \to \infty} \eta_n ||u_n z_n||^2 = 0.$
- (*ii*) $\lim_{n \to \infty} ||u_n z_n|| = 0.$

Proof. Let $x^* \in \Gamma$. Since A is uniformly continuous on bounded subsets of \mathcal{H}_1 , then $\{A(x_n)\}, \{A(y_n)\}, \{w_n\}$ and $\{u_n\}$ are bounded. Also, using (4.5.3) we have

$$f_n(u_n) = \langle A(y_n), u_n - y_n \rangle = \eta_n \langle A(y_n), u_n - z_n \rangle \ge \eta_n \frac{\sigma}{2} ||u_n - z_n||^2.$$
(4.5.12)

Thus, applying Lemma 2.5.16 and Remark 4.5.1, we obtain

$$||P_{\mathcal{C}_{n}}(u_{n}) - x^{*}||^{2} \leq ||u_{n} - x^{*}||^{2} - ||P_{\mathcal{C}_{n}}(u_{n}) - u_{n}||^{2}$$

$$= ||u_{n} - x^{*}||^{2} - \operatorname{dist}^{2}(u_{n}, \mathcal{C}_{n})$$

$$\leq ||u_{n} - x^{*}||^{2} - \left(\frac{1}{M}f_{n}(u_{n})\right)^{2}$$

$$\leq ||u_{n} - x^{*}||^{2} - \left(\frac{1}{2M}\sigma\eta_{n}||u_{n} - z_{n}||^{2}\right)^{2}. \quad (4.5.13)$$

Since $\{x_n\}$ is bounded, we obtain from (4.5.13), (4.5.9) and (4.5.4) that

$$\left(\frac{1}{2M}\sigma\eta_{n}\|u_{n}-z_{n}\|^{2}\right)^{2} \leq \|u_{n}-x^{*}\|^{2}-\|x_{n+1}-x^{*}\|^{2}+\gamma_{n}\|g(x_{n})-x^{*}\|^{2}$$

$$\leq \|u_{n}-x_{n+1}\|^{2}+\|x_{n+1}-x^{*}\|^{2}+2\|u_{n}-x_{n+1}\|\|x_{n+1}-x^{*}\|$$

$$-\|x_{n+1}-x^{*}\|^{2}+\gamma_{n}\|g(x_{n})-x^{*}\|^{2}$$

$$= \|u_{n}-x_{n+1}\|^{2}+2\|u_{n}-x_{n+1}\|\|x_{n+1}-x^{*}\|+\gamma_{n}\|g(x_{n})-x^{*}\|^{2}$$

$$= \|u_{n}-x_{n+1}\|\left[\|u_{n}-x_{n+1}\|+2\|x_{n+1}-x^{*}\|\right]+\gamma_{n}\|g(x_{n})-x^{*}\|^{2}$$

$$\leq (\|u_{n}-P_{\mathcal{C}_{n}}u_{n}\|+\|P_{\mathcal{C}_{n}}u_{n}-x_{n+1}\|)M_{1}+\gamma_{n}\|g(x_{n})-x^{*}\|^{2}$$

$$\leq \|u_{n}-P_{\mathcal{C}_{n}}u_{n}\|M_{1}+\gamma_{n}M_{1}\|g(x_{n})-P_{\mathcal{C}_{n}}u_{n}\|+\gamma_{n}\|g(x_{n})-x^{*}\|^{2},$$

$$(4.5.14)$$

where $M_1 := \sup_{n \ge 1} \{ \|u_n - x_{n+1}\| + 2\|x_{n+1} - x^*\| \}$. Hence $\lim_{n \to \infty} \eta_n \|u_n - z_n\|^2 = 0$. We now show that $\lim_{n \to \infty} \|u_n - z_n\| = 0$. From Algorithm 4.5.2, we obtain that $\liminf_{n \to \infty} \eta_n \ge 0$. **Case 1:** Suppose that $\liminf_{n \to \infty} \eta_n > 0$. Then, we obtain that

$$\begin{split} \limsup_{n \to \infty} \|u_n - z_n\|^2 &\leq \limsup_{n \to \infty} \left(\eta_n \|u_n - z_n\|^2\right) \left(\limsup_{n \to \infty} \frac{1}{\eta_n}\right) \\ &\leq \left(\limsup_{n \to \infty} \eta_n \|u_n - z_n\|^2\right) \left(\frac{1}{\liminf_{n \to \infty} \eta_n}\right) \\ &= 0. \end{split}$$

Hence, $\limsup_{n \to \infty} ||u_n - z_n|| = 0$, which implies that

$$\lim_{n \to \infty} \|u_n - z_n\| = 0.$$

Case 2: Suppose that $\liminf_{n\to\infty} \eta_n = 0$. Without loss of generality, we may assume that there exists a subsequence of $\{\eta_n\}$ still denoted by $\{\eta_n\}$ such that $\lim_{n\to\infty} \eta_n = 0$ and $\lim_{n\to\infty} ||u_n - z_n|| = t \ge 0$.

Setting $v_n = u_n + \frac{1}{\gamma}\eta_n(z_n - u_n)$, we obtain that

$$\lim_{n \to \infty} \|v_n - u_n\| = 0. \tag{4.5.15}$$

Also, by the definition of v_n and (4.5.3), we obtain

$$\langle A(v_n), u_n - z_n \rangle < \frac{\sigma}{2} \|u_n - z_n\|^2,$$

which implies that

$$2\langle A(v_n) - A(u_n), u_n - z_n \rangle + 2\langle A(u_n), u_n - z_n \rangle < \sigma ||u_n - z_n||^2, \quad \forall n \ge 1.$$
 (4.5.16)

Set $s_n := u_n - A(u_n)$, then (4.5.16) becomes

$$2\langle A(v_n) - A(u_n), u_n - z_n \rangle + 2\langle u_n - s_n, u_n - z_n \rangle < \sigma ||u_n - z_n||^2,$$

which implies

$$||s_n - u_n||^2 - ||s_n - z_n||^2 < (\sigma - 1)||u_n - z_n||^2$$

$$-2\langle A(v_n) - A(u_n), u_n - z_n \rangle, \quad \forall n \ge 1.$$
(4.5.17)

Since A is uniformly continuous on bounded subsets of \mathcal{H}_1 , if t > 0 then we obtain from (4.5.15) and (4.5.17) that

$$\limsup_{n \to \infty} \left(\|s_n - u_n\|^2 - \|s_n - z_n\|^2 \right) \le (\sigma - 1)t < 0.$$

For $\varepsilon = \frac{-(\sigma-1)}{2}t > 0$, there exists $N \in \mathbb{N}$ such that

1

$$||s_n - u_n||^2 - ||s_n - z_n||^2 \le (\sigma - 1)t + \varepsilon = \frac{(\sigma - 1)t}{2} < 0 \quad \forall n \in \mathbb{N}, \quad n \ge N.$$

Thus, we obtain that

$$\|s_n - u_n\| < \|s_n - z_n\| \quad \forall n \ge 1.$$

That is, $||(u_n - A(u_n)) - u_n|| < ||(u_n - A(u_n)) - P_{\mathcal{C}}(u_n - A(u_n))||$ which is a contradiction. Therefore, t = 0. Hence, $\lim_{n \to \infty} ||u_n - z_n|| = 0$.

Lemma 4.5.5. Let $\{x_n\}$ be a sequence generated by Algorithm 4.5.2 such that Assumption 4.5.1 holds. If there exists a subsequence $\{x_{n_k}\}$ of $\{x_n\}$ which converges weakly to a point $z \in \mathcal{H}_1$ and $\lim_{k \to \infty} ||u_{n_k} - w_{n_k}|| = 0 = \lim_{k \to \infty} ||P_{\mathcal{C}_{n_k}}(u_{n_k}) - u_{n_k}||$, for some subsequences $\{u_{n_k}\}$ and $\{w_{n_k}\}$ of $\{u_n\}$ and $\{w_n\}$ respectively. Then, $z \in \Gamma$. *Proof.* By Step 2 and (4.5.11), we obtain that

$$\lim_{n \to \infty} \|w_n - x_n\| = \lim_{n \to \infty} \delta_n \|x_n - x_{n-1}\| = \lim_{n \to \infty} \gamma_n \cdot \frac{\delta_n}{\gamma_n} \|x_n - x_{n-1}\| = 0.$$
(4.5.18)

Now, let $\{x_{n_k}\}$ be a subsequence of $\{x_n\}$ that converges weakly to some $z \in \mathcal{H}_1$. Thus, $\{w_{n_k}\}$ also converges weakly to $z \in \mathcal{H}_1$. Hence, by our hypothesis, there exists a subsequence $\{u_{n_k}\}$ of $\{u_n\}$ that converges weakly to $z \in \mathcal{H}_1$. We may also assume without loss of generality that, the subsequence $\{\tau_{n_k}\}$ of $\{\tau_n\}$ converges to a point say $\overline{\tau} \in \left(0, \frac{1}{\|T\|^2}\right)$. Also, by Lemma 2.1.9, $T^*(I - S_\beta)A$ is an inverse strongly monotone operator. Therefore, $\{T^*(I - S_\beta)Tw_{n_k}\}$ is bounded. Hence, we obtain that

$$\|(I - \tau_{n_k} T^* (I - S_\beta) T) w_{n_k} - (I - \overline{\tau} T^* (I - S_\beta) T) w_{n_k}\| = |\tau_{n_k} - \overline{\tau}| \|T^* (I - S_\beta) T w_{n_k}\| \to 0, \text{ as } k \to \infty.$$

That is,

$$\lim_{k \to \infty} \|u_{n_k} - (I - \overline{\tau} T^* (I - S_\beta) T) w_{n_k}\| = 0,$$

which implies from our hypothesis that

$$\lim_{k \to \infty} \|w_{n_k} - (I - \overline{\tau} T^* (I - S_\beta) T) w_{n_k}\| = 0.$$
(4.5.19)

Thus, by Lemma 2.5.17, we obtain that $z \in F(I - \overline{\tau}T^*(I - S_\beta)T)$, which together with Lemma 2.1.8 implies that

$$Tz \in F(S_{\beta}) = F(S). \tag{4.5.20}$$

Now, from the characteristics property of metric projection $P_{\mathcal{C}}$, we obtain for all $x \in \mathcal{C}$ that

$$\langle u_{n_k} - A(u_{n_k}) - z_{n_k}, x - z_{n_k} \rangle \le 0, \quad \forall x \in \mathcal{C},$$

which implies that

$$0 \leq \langle z_{n_k} - u_{n_k} + A(u_{n_k}), x - z_{n_k} \rangle$$

= $\langle z_{n_k} - u_{n_k}, x - z_{n_k} \rangle + \langle A(u_{n_k}), u_{n_k} - z_{n_k} \rangle + \langle A(u_{n_k}), x - u_{n_k} \rangle$
 $\leq ||z_{n_k} - u_{n_k}|| ||x - z_{n_k}|| + ||A(u_{n_k})|| ||u_{n_k} - z_{n_k}|| + \langle A(u_{n_k}), x - u_{n_k} \rangle.$ (4.5.21)

Fix $x \in C$ and let $k \to \infty$, since $\lim_{k\to\infty} ||u_{n_k} - z_{n_k}|| = 0$ (by Lemma 4.5.4), we have from (4.5.21) that

$$0 \le \liminf_{k \to \infty} \langle A(u_{n_k}), x - u_{n_k} \rangle \quad \forall x \in \mathcal{C}.$$
(4.5.22)

Now, choose a sequence $\{\delta_k\}$ of positive numbers such that $\delta_{k+1} \leq \delta_k$, $\forall k \geq 1$ and $\delta_k \rightarrow 0$ as $k \rightarrow \infty$. Then, for each δ_k , we denote by N_k (which exists as a result of (4.5.22)) the smallest positive integer such that

$$\langle Au_{n_j}, x - u_{n_j} \rangle + \delta_k \ge 0 \quad \forall j \ge N_k.$$
 (4.5.23)

Since $\{\delta_k\}$ is decreasing, we have that $\{N_k\}$ is increasing. Furthermore, we set for each $k \geq 1$, $m_{N_k} = \frac{Au_{N_k}}{\|Au_{N_k}\|^2}$, provided $Au_{N_k} \neq 0$. Then it is easy to see that $\langle Au_{N_k}, m_{N_k} \rangle = 1$ for each $k \geq 1$. Thus, by (4.5.23), we have that

$$\langle Au_{N_k}, x + \delta_k m_{N_k} - u_{N_k} \rangle \ge 0,$$

which implies by the pseudo-monotonicity of A that

$$\langle A(x+\delta_k m_{N_k}), x+\delta_k m_{N_k}-u_{N_k}\rangle \ge 0.$$
(4.5.24)

Now, by the sequentially weakly continuity of A, we have that $\{Au_{n_k}\}$ converges weakly to Az. If Az = 0, then $z \in VI(\mathcal{C}, A)$. On the other hand, if we suppose that $Az \neq 0$, then by the weakly lower semicontinuity of $\|\cdot\|$, we obtain that

$$0 < ||Az|| \le \liminf_{k \to \infty} ||Au_{n_k}||.$$

Since $\{u_{N_k}\} \subset \{u_{n_k}\}$, we obtain that

$$0 \le \limsup_{k \to \infty} \|\delta_k m_{N_k}\| = \limsup_{k \to \infty} \left(\frac{\delta_k}{\|Au_{n_k}\|}\right)$$

$$\leq \frac{\limsup_{k \to \infty} \delta_k}{\liminf_{k \to \infty} \|Au_{n_k}\|} \\ \leq \frac{0}{\|Az\|} = 0.$$

Therefore, $\lim_{k\to\infty} \|\delta_k m_{N_k}\| = 0$. Thus, letting $k\to\infty$ in (4.5.24) yields

$$\langle Ax, x-z \rangle \ge 0 \quad \forall x \in \mathcal{C},$$
 (4.5.25)

which implies by Lemma 2.5.2 that $z \in VI(C, A)$. This together with (4.5.20) gives that $z \in \Gamma$.

We give our main result in the next theorem.

Theorem 4.5.6. Let Assumption 4.5.1 hold. Then the sequence $\{x_n\}$ generated by Algorithm 4.5.2 converges strongly to $z^* := P_{\Gamma}g(z^*)$.

Proof. Let $z^* = P_{\Gamma}g(z^*)$, then from (4.5.9) and Lemma 2.1.1, we obtain

$$\begin{aligned} \|u_n - z^*\|^2 &\leq \|w_n - z^*\|^2 \\ &= \|x_n - z^* + \delta n(x_n - x_{n-1})\|^2 \\ &= \|x_n - z^*\|^2 + 2\delta_n \langle x_n - x_{n-1}, x_n - z^* \rangle + \delta_n \|x_n - x_{n-1}\|^2. \\ &= \|x_n - z^*\|^2 + \delta_n \left(-\|x_{n-1} - z^*\|^2 + \|x_n - z^*\|^2 + \|x_n - x_{n-1}\|^2 \right) \\ &+ \delta_n \|x_n - x_{n-1}\|^2 \\ &= \|x_n - z^*\|^2 + \delta_n \left(\|x_n - z^*\|^2 - \|x_{n-1} - z^*\|^2 \right) + 2\delta_n \|x_n - x_{n-1}\|^2. \end{aligned}$$
(4.5.26)

By (4.5.4), (4.5.26) and Lemma 2.1.1, we obtain that

$$\begin{aligned} \|x_{n+1} - z^*\|^2 &= \|\gamma_n(g(x_n) - z^*) + (1 - \gamma_n)(P_{\mathcal{C}_n}(u_n) - z^*)\|^2 \\ &\leq (1 - \gamma_n) \|u_n - z^*\|^2 - (1 - \gamma_n) \|P_{\mathcal{C}_n}(u_n) - u_n\|^2 \\ &+ 2\gamma_n \langle g(x_n) - z^*, x_{n+1} - z^* \rangle \\ &= (1 - \gamma_n) \|x_n - z^*\|^2 - (1 - \gamma_n) \|P_{\mathcal{C}_n}(u_n) - u_n\|^2 \\ &+ \delta_n (1 - \gamma_n) \left(\|x_n - z^*\|^2 - \|x_{n-1} - z^*\|^2 \right) \\ &+ 2\delta_n (1 - \gamma_n) \|x_n - x_{n-1}\|^2 + 2\gamma_n \langle g(x_n) - z^*, x_{n+1} - z^* \rangle. \\ &\leq (1 - \gamma_n) \|x_n - z^*\|^2 + \gamma_n \left[2\langle g(x_n) - z^*, x_{n+1} - z^* \rangle \right] \\ &+ \gamma_n \left[\frac{\delta_n}{\gamma_n} (1 - \gamma_n) \left[\|x_n - z^*\|^2 - \|x_{n-1} - z^*\|^2 + 2\|x_n - x_{n-1}\|^2 \right] \right] \\ &= (1 - \gamma_n) \|x_n - z^*\|^2 + \gamma_n \bar{v}_n \end{aligned}$$
(4.5.28)

where $\bar{v}_n := 2\langle g(x_n) - z^*, x_{n+1} - z^* \rangle + \frac{\delta_n}{\gamma_n} (1 - \gamma_n) \left[\|x_n - z^*\|^2 - \|x_{n-1} - z^*\|^2 + \|x_n - x_{n+1}\|^2 \right]$. To show that $\{x_n\}$ converges to z^* , we shall apply Lemma 2.5.36. That is, we show that $\limsup_{k \to \infty} \bar{v}_{n_k} \leq 0$ for every subsequence $\{ \|x_{n_k} - z^*\| \}$ of $\{ \|x_n - z^*\| \}$ satisfying the condition

$$\liminf_{k \to \infty} \left(\|x_{n+1} - z^*\| - \|x_{n_k} - z^*\| \right) \ge 0.$$
(4.5.29)

Now, suppose that $\{||x_{n_k} - z^*||\}$ is a subsequence of $\{||x_n - z^*||\}$ such that (4.5.29) holds. Then,

$$\lim_{k \to \infty} \inf \left(\|x_{n_k+1} - z^*\|^2 - \|x_{n_k} - z^*\|^2 \right) \\
= \lim_{k \to \infty} \inf \left[\left(\|x_{n_k+1} - z^*\| - \|x_{n_k} - z^*\| \right) \left(\|x_{n_k+1} - z^*\| + \|x_{n_k} - z^*\| \right) \right] \\
\geq 0.$$
(4.5.30)

On the other hand, we obtain from (4.5.27) that

$$(1 - \gamma_{n_k}) \| P_{\mathcal{C}_{n_k}} u_{n_k} - u_{n_k} \|^2 \leq \| x_{n_k} - z^* \| - \| x_{n_k+1} - z^* \|^2 + \delta_{n_k} (1 - \gamma_{n_k}) \left[\| x_{n_k} - z^* \|^2 - \| x_{n_k-1} - z^* \|^2 \right] + 2\delta_{n_k} \| x_{n_k} - x_{n_{k-1}} \|^2 + 2\gamma_{n_k} \langle g(x_{n_k}) - z^*, x_{n_k+1} - z^* \rangle = \| x_{n_k} - z^* \| - \| x_{n_k+1} - z^* \|^2 + \delta_{n_k} (1 - \gamma_{n_k}) \left[(\| x_{n_k} - x_{n_k-1} \| + \| x_{n_k-1} - z^* \|)^2 \right] - \delta_{n_k} (1 - \gamma_{n_k}) \left[\| x_{n_k-1} - z^* \|^2 \right] + 2\delta_{n_k} \| x_{n_k} - x_{n_k-1} \|^2 + 2\gamma_{n_k} \langle g(x_{n_k}) - z^*, x_{n_k+1} - z^* \rangle \leq \| x_{n_k} - z^* \| - \| x_{n_k+1} - z^* \|^2 + \delta_{n_k} \left[\| x_{n_k} - x_{n_k-1} \|^2 + 2 \| x_{n_k} - x_{n_k-1} \| \| x_{n_k-1} - z^* \| \right] + 2\delta_{n_k} \| x_{n_k} - x_{n_k-1} \|^2 + 2\gamma_{n_k} \langle g(x_{n_k}) - z^*, x_{n_k+1} - z^* \rangle \rightarrow 0 \quad \text{as} \quad k \to \infty.$$

From (4.5.30) and Assumption 3.1, we have

$$\begin{split} \limsup_{k \to \infty} (1 - \gamma_{n_k}) \| P_{\mathcal{C}_{n_k}} u_{n_k} - u_{n_k} \|^2 &\leq \limsup_{k \to \infty} \left[\| x_{n_k} - z^* \| - \| x_{n_{k+1}} - z^* \|^2 \right] \\ &+ \limsup_{k \to \infty} \left[\gamma_{n_k} \cdot \frac{\delta_{n_k}}{\gamma_{n_k}} \left[\| x_{n_k} - x_{n_{k-1}} \|^2 \right] \right] \\ &+ \limsup_{k \to \infty} \left[\gamma_{n_k} \cdot \frac{\delta_{n_k}}{\gamma_{n_k}} \left[2 \| x_{n_k} - x_{n_{k-1}} \| \| x_{n_{k-1}} - z^* \| \right] \right] \\ &+ \limsup_{n \to \infty} \left[2 \gamma_{n_k} \cdot \frac{\delta_{n_k}}{\gamma_{n_k}} \| x_{n_k} - x_{n_{k-1}} \|^2 \right] \\ &+ \limsup_{n \to \infty} \left[2 \gamma_{n_k} \langle g(x_{n_k}) - z^*, x_{n_k+1} - z^* \rangle \right] \\ &= -\liminf_{k \to \infty} \left[\| x_{n_k+1} - z^* \|^2 - \| x_{n_k} - z^* \|^2 \right] \\ &\leq 0. \end{split}$$

Hence,

$$\lim_{k \to \infty} \|P_{\mathcal{C}_{n_k}} u_{n_k} - u_{n_k}\| = 0.$$
(4.5.31)

From (4.5.9) and (4.5.26), we obtain that

$$\beta_{n_{k}}(1-\beta_{n_{k}})\|w_{n_{k}}-S_{n_{k}}w_{n_{k}}\| \leq \|w_{n_{k}}-z^{*}\|^{2} - \|u_{n_{k}}-z^{*}\|^{2} \\ \leq \|x_{n_{k}}-z^{*}\|^{2} - \|u_{n_{k}}-z^{*}\|^{2} \\ + \delta_{n_{k}}(\|x_{n_{k}}-z^{*}\|^{2} - \|x_{n_{k}-1}-z^{*}\|) \\ + 2\delta_{n_{k}}\|x_{n_{k}}-x_{n_{k}-1}\|^{2}.$$

$$(4.5.32)$$

Again from (4.5.27), we obtain

$$-\|u_{n_k} - z^*\|^2 \le -\|x_{n_k+1} - z^*\|^2 + 2\gamma_{n_k} \langle g(x_{n_k}) - z^*, x_{n_k+1} - z^* \rangle,$$

which implies from (4.5.32) that

$$\beta_{n_{k}}(1-\beta_{n_{k}})\|w_{n_{k}}-S_{n_{k}}w_{n_{k}}\| \leq \|x_{n_{k}}-z^{*}\|^{2}-\|x_{n_{k}+1}-z^{*}\|^{2} + 2\gamma_{n_{k}}\langle g(x_{n_{k}})-z^{*},x_{n_{k}+1}-z^{*}\rangle + \delta_{n_{k}}\left(\|x_{n_{k}}-z^{*}\|^{2}-\|x_{n_{k}-1}-z^{*}\|^{2}\right) + 2\delta_{n_{k}}\|x_{n_{k}}-x_{n_{k}-1}\|.$$

$$(4.5.33)$$

By taking the lim sup as $k \to \infty$ in (4.5.33), we obtain that

$$\limsup_{k \to \infty} \beta_{n_k} (1 - \beta_{n_k}) \| w_{n_k} - S_{n_k} w_{n_k} \| = 0.$$

Hence,

$$\lim_{k \to \infty} \|w_{n_k} - S_{n_k} w_{n_k}\| = 0.$$

This implies that

$$\lim_{k \to \infty} \|w_{n_k} - u_{n_k}\| = \beta_{n_k} \lim_{n \to \infty} \|w_{n_k} - S_{n_k} w_{n_k}\| = 0.$$
(4.5.34)

By similar method as in (4.5.18), we obtain

$$\lim_{k \to \infty} \|w_{n_k} - x_{n_k}\| = 0. \tag{4.5.35}$$

From (4.5.4), we obtain that

$$\lim_{k \to \infty} \|x_{n_k+1} - P_{C_{n_k}}(u_{n_k})\| = 0.$$
(4.5.36)

Using (4.5.36), (4.5.31), (4.5.34) and (4.5.35), we obtain that

$$\lim_{k \to \infty} \|x_{n_k+1} - x_{n_k}\| = 0.$$
(4.5.37)

Since $\{x_{n_k}\}$ is bounded, we take a subsequence $\{x_{n_{k_j}}\}$ of $\{x_{n_k}\}$ such that $\{x_{n_{k_j}}\}$ converges weakly to $z \in \mathcal{H}_1$ and

$$\begin{split} \limsup_{k \to \infty} \langle g(z^*) - z^*, x_{n_k} - z^* \rangle &= \lim_{j \to \infty} \langle g(z^*) - z^*, x_{n_{k_j}} - z^* \rangle \\ &= \langle g(z^*) - z^*, z - z^* \rangle. \end{split}$$

Also, we obtain from (4.5.31), (4.5.34) and Lemma 4.5.5 that $z \in \Gamma$. Since $z^* = P_{\Gamma}g(z^*)$, we obtain from the previous equality that

$$\limsup_{k \to \infty} \langle g(z^*) - z^*, x_{n_k} - z^* \rangle \le 0, \qquad (4.5.38)$$

which implies by (4.5.37) that

$$\limsup_{k \to \infty} \langle g(z^*) - z^*, x_{n_k+1} - z^* \rangle \le 0.$$
(4.5.39)

Thus, $\limsup_{k\to\infty} \bar{v}_{n_k} \leq 0$. Hence from Lemma 2.5.36, we obtain that $\{x_n\}$ converges strongly to z^* .

Remark 4.5.7. Observe that by setting $\mathcal{H}_1 = \mathcal{H}_2 = \mathcal{H}$, S = I and $T = I_{\mathcal{H}}$ (the identity operator on \mathcal{H}) in Theorem 4.5.6, we obtain as a corollary, an inertial extrapolation method for solving the classical VIP (1.2.1) when the cost operator is pseudo-monotone and Lipschitz continuous.

4.5.2 Numerical experiments

We give in this section, some numerical examples of Algorithm 4.5.2 in comparison with Algorithm 4.5.11 and related methods in the literature (Algorithm (4.5.41), Algorithm (4.5.42) and Algorithm (4.5.43)) for solving the class of split VIP (Problem (1.2.6)) considered in this work. All codes are written in Matlab 2016 (b) and performed on a personal computer with an Intel(R) Core(TM) i5-2600 CPU at 2.30GHz and 8.00 Gb-RAM. In Tables 4.5.1-4.5.3, "Iter." means the number of iterations while "CPU" means the CPU time in seconds. In our computations, we choose $\gamma = \frac{1}{4} = \sigma$, $\gamma_n = \frac{1}{5n+1}$, $\delta_n = \bar{\delta}_n$, $\epsilon_n = \frac{\gamma_n}{n^{0.01}}$ and $\delta = 0.3$. Also, we define $g(x) = \frac{2x}{7}$ and S(x) = -4x. Then, g is a contraction with

coefficient $\rho = \frac{2}{7}$ and S is $\frac{3}{5}$ -strictly pseudocontractive. Thus, we can choose $\beta = \frac{3}{5}$, so that $S_{\beta}(x) = -x$.

Furthermore, we define $\text{TOL}_n := \frac{1}{2} (||x_n - P_C(x_n - \lambda A x_n)||^2 + ||Sx_n - STx_n||^2)$, and use the stopping criterion $\text{TOL}_n < \varepsilon$ for the iterative processes, where ε is the predetermined error. Note that if $\text{TOL}_n = 0$, then x_n is a solution of problem (1.2.6).

Example 4.5.8. Let $\mathcal{H}_1 = \mathbb{R}^N$ and $\mathcal{H}_2 = \mathbb{R}^m$. Define $A : \mathbb{R}^N \to \mathbb{R}^N$ by A(x) = Mx + q, where the matrix M is formed as: $M = V \sum V'$, with $V = I - \frac{2vv'}{\|v\|^2}$ and $\sum = \text{diag}(\sigma_{11}, \sigma_{12}, \cdots, \sigma_{1N})$ been the Householder and the diagonal matrix, respectively, and

$$\sigma_{1j} = \cos\frac{j\pi}{N+1} + 1 + \frac{\cos\frac{\pi}{N+1} + 1 - \widehat{C}(\cos\frac{N\pi}{N+1} + 1)}{\widehat{C} - 1}, \quad j = 1, 2, \cdots, N,$$

where \widehat{C} is the present condition number of M ([121, Example 5.2]). In the numerical computation, we choose $\widehat{C} = 10^4$, q = 0 and uniformly take the vector $v \in \mathbb{R}^N$ in (-1, 1). Thus, A is pseudo-monotone and uniformly continuous (see [121]). Furthermore, we generate the bounded linear operator $T \in \mathbb{R}^{M \times N}$ with independent Gaussian components distributed in the interval (0, 1), and then normalize each column of T with the unit norm. We set $\mathcal{C} = \{x \in \mathbb{R}^N : ||x|| \leq 1\}$. The projection onto \mathcal{C} is effectively computed in Matlab. We consider different scenarios of the problem's dimensions. That is, N = 100, 300, 500, 1000and m = N/2, with starting points $x_1 = (1, 1, \ldots, 1)'$ and $x_0 = (0, 0, \ldots, 0)'$. For this example, we take $\varepsilon = 10^{-6}$ as the stopping criterion and obtain the numerical results reported in Table 4.5.1 and Figure 4.9.

(N,m)		Algorithm	Algorithm	Algorithm	Algorithm	Algorithm
		4.5.2	4.5.11	(4.5.41)	(4.5.42)	(4.5.43)
(100, 50)	CPU	0.0545	0.1525	0.1435	0.1708	0.1762
	Iter.	412	785	633	1465	979
	CPU	0.0368	0.1510	0.1285	0.1523	0.1619
(300, 150)	Iter.	277	635	448	1465	865
	CPU	0.0314	0.1365	0.1280	0.1498	0.1505
(500, 250)	Iter.	283	823	501	1465	838
	CPU	0.0569	0.1414	0.1345	0.1747	0.1704
(1000, 500)	Iter.	473	795	578	1465	790

Table 4.5.1. Numerical results for Example 4.5.8 with $\varepsilon = 10^{-6}$.

Example 4.5.9. Let $\mathcal{H}_1 = (l_2(\mathbb{R}), ||.||_{l_2}) = \mathcal{H}_2$, where $l_2(\mathbb{R}) := \{x = (x_1, x_2, x_3, \dots), x_i \in \mathbb{R} : \sum_{i=1}^{\infty} |x_i|^2 < \infty\}$ and $||x||_{l_2} = \left(\sum_{i=1}^{\infty} |x_i|^2\right)^{\frac{1}{2}}, \quad \forall x \in l_2(\mathbb{R})$. Now, define the operator $T : l_2(\mathbb{R}) \to l_2(\mathbb{R})$ by $Tx = (0, x_1, \frac{x_2}{2}, \frac{x_3}{3}, \dots), \quad \forall x \in l_2(\mathbb{R})$. Then, T is a bounded linear operator on $l_2(\mathbb{R})$ with adjoint $T^*y = (y_2, \frac{y_3}{2}, \frac{y_4}{3}, \dots), \quad \forall y \in l_2(\mathbb{R})$.



Figure 4.9: The behavior of TOL_n with $\varepsilon = 10^{-6}$ for Example 4.5.8: Top Left: (N, m) = (100, 50); Top Right: (N, m) = (300, 150); Bottom Left: (N, m) = (500, 250); Bottom Right: (N, m) = (1000, 500).

Let $\mathcal{C} = \{x \in l_2(\mathbb{R}) : |x_i| \leq \frac{1}{i}, i = 1, 2, 3, ...\}$. Then, we have explicit formula for $P_{\mathcal{C}}$. Now, define the operator $A: l_2(\mathbb{R}) \to l_2(\mathbb{R})$ by

$$Ax = \left(||x|| + \frac{1}{||x|| + \alpha}\right)\alpha,$$

for some $\alpha > 0$. Then, A is pseudomonotone on $l_2(\mathbb{R})$, uniformly continuous and sequentially weakly continuous on C but not Lipschitz continuous (see [248]). For this example, we take $\varepsilon = 10^{-8}$ as the stopping criterion and choose the starting points as follows: **Case 1:** Take $x_1 = (1, \frac{1}{2}, \frac{1}{3}, \cdots)$ and $x_0 = (\frac{1}{2}, \frac{1}{5}, \frac{1}{10}, \cdots)$.

Case 2: Take $x_1 = (\frac{1}{2}, \frac{1}{5}, \frac{1}{10}, \cdots)$ and $x_0 = (1, \frac{1}{2}, \frac{1}{3}, \cdots)$.

Case 3: Take $x_1 = (1, \frac{1}{4}, \frac{1}{9}, \cdots)$ and $x_0 = (\frac{1}{2}, \frac{1}{4}, \frac{1}{8}, \cdots)$.

Case 4: Take $x_1 = (\frac{1}{2}, \frac{1}{4}, \frac{1}{8}, \cdots)$ and $x_0 = (1, \frac{1}{4}, \frac{1}{9}, \cdots)$. The numerical results reported in Table 4.5.2 and Figure 4.10.

Table 4.5.2. Numerical results for Example 4.5.9 with $\varepsilon = 10^{-8}$.

Cases		Algorithm	Algorithm	Algorithm	Algorithm	Algorithm
		4.5.2	4.5.11	(4.5.41)	(4.5.42)	(4.5.43)
1	CPU	0.0136	0.1101	0.1110	0.1229	0.1334
	Iter.	37	44	58	145	118
2	CPU	0.0146	0.1103	0.1100	0.1238	0.1314
	Iter.	40	57	60	199	123
3	CPU	0.0190	0.1127	0.1212	0.1185	0.1283
	Iter.	38	50	58	148	118
4	CPU	0.0136	0.1183	0.1183	0.1253	0.1435
	Iter.	34	65	53	89	107

Example 4.5.10. Let $\mathcal{H}_1 = \mathcal{H}_2 = L_2([0, 2\pi])$ be endowed with inner product

$$\langle x, y \rangle = \int_0^{2\pi} x(t)y(t)dt \quad \forall \ x, y \in L_2([0, 2\pi])$$

and norm

$$||x|| := \left(\int_0^{2\pi} |x(t)|^2 dt\right)^{\frac{1}{2}} \quad \forall \ x, y \in L_2([0, 2\pi]).$$

Let $C = \{x \in L_2([0, 2\pi]) : ||x - e||_{L_2} \le b\}$, where $e = 2 + \sin 2t$ and b = 1. Then, the metric projection $P_{\mathcal{C}}$ is defined as:

$$P_{\mathcal{C}}(x) = \begin{cases} x, & \text{if } x \in \mathcal{C}, \\ \frac{x-e}{||x-e||_{L_2}}b + e, & \text{otherwise.} \end{cases}$$

Now, define the operator $A: L_2([0, 2\pi]) \to L_2([0, 2\pi])$ by

$$Ax(t) = L(x)D(x)(t), \ x \in L_2([0, 2\pi]), \ t \in [0, 2\pi],$$



Figure 4.10: The behavior of TOL_n with $\varepsilon = 10^{-8}$ for Example 4.5.9: Top Left: **Case 1**; Top Right: **Case 2**; Bottom Left: **Case 3**; Bottom Right: **Case 4**.

where $L(x) := \frac{1}{1+||x||_{L_2}^2}$ and D is the Volterra integral mapping (which is monotone, bounded and linear, see [39]), defined by

$$Dx(t) = \int_0^t x(s)ds, \ x \in L_2([0, 2\pi]), \ t \in [0, 2\pi].$$

Then A is pseudo-monotone on $L_2([0, 2\pi])$. Let $T: L_2([0, 2\pi]) \to L_2([0, 2\pi])$ be defined by

$$Tx(s) = \int_0^{2\pi} K(s,t)x(t)dt \ \forall x \in L_2([0,2\pi]),$$

where K is a continuous real-valued function defined on $[0, 2\pi] \times [0, 2\pi]$. Then T is a bounded linear operator with adjoint

$$T^*x(s) \int_0^{2\pi} K(t,s)x(t)dt \ \forall x \in L_2([0,2\pi]).$$

We take $\varepsilon = 10^{-10}$ as the stopping criterion and choose the starting points as follows: **Case 1:** Take $x_1(t) = 1 + t^2$ and $x_0(t) = t + 5$.

Case 2: Take $x_1(t) = 2t + t^2$ and $x_0(t) = t + 1$.

Case 3: Take $x_1(t) = 0.7e^{-t} + 1$ and $x_0(t) = t + t^3$.

Case 4: Take $x_1(t) = \sin(t) + 1$ and $x_0(t) = e^t$. The numerical results shown in Table 4.5.3 and Figure 4.11.

Cases		Algorithm	Algorithm	Algorithm	Algorithm	Algorithm
		4.5.2	4.5.11	(4.5.41)	(4.5.42)	(4.5.43)
1	CPU	5.2616	12.2498	12.6245	13.4485	150.5827
	Iter.	56	95	76	68	158
2	CPU	5.8211	8.9063	8.5748	9.4896	124.5990
	Iter.	53	93	73	65	151
3	CPU	5.1099	11.1716	11.8042	14.2559	173.8901
	Iter.	55	95	76	68	157
4	CPU	6.5188	24.9020	29.0189	46.8546	167.9730
	Iter.	55	67	102	89	157

Table 4.5.3. Numerical results for Example 4.5.10 with $\varepsilon = 10^{-10}$.

Appendix 4.5.11. The Algorithm of [249]

Initialization: Given $\gamma > 0, l \in (0, 1), \mu \in (0, 1)$. Let $x_0 \in \mathcal{H}$ be arbitrary.

Iterative Steps: Calculate x_{n+1} as follows:

Step1. Compute

$$y_n = P_{\mathcal{C}}(x_n - \lambda_n A x_n),$$

where λ_n is chosen to be the largest $\lambda \in \{\gamma, \gamma l, \gamma l^2, \cdots\}$ satisfying

$$\lambda_n \|Ax_n - Ay_n\| \le \mu \|x_n - y_n\|.$$
(4.5.40)



Figure 4.11: The behavior of TOL_n with $\varepsilon = 10^{-10}$ for Example 4.5.10: Top Left: Case 1; Top Right: Case 2; Bottom Left: Case 3; Bottom Right: Case 4.

If $x_n = y_n$, then stop and x_n is the solution of VIP. Otherwise, Step 2: Compute

$$x_{n+1} = \alpha_n g(x_n) + (1 - \alpha_n) z_n,$$

where $z_n = y_n - \lambda_n (Ay_n - Ax_n)$ and g is a contraction mapping on \mathcal{H} .

Appendix 4.5.12. The Algorithm in [243].

Let $x_1 \in \mathcal{C}$, define the sequences $\{x_n\}$, $\{y_n\}$ and $\{t_n\}$ by

$$\begin{cases} y_n = P_{\mathcal{C}}(x_n - \tau_n T^*(I - S)Tx_n), \\ t_n = P_{\mathcal{C}}(y_n - \lambda_n A(y_n)), \\ x_{n+1} = P_{\mathcal{C}}(y_n - \lambda_n A(t_n)), \quad n \ge 1, \end{cases}$$
(4.5.41)

where $\{\tau_n\} \subset [a, b]$ for some $a, b \in \left(0, \frac{1}{\|T\|^2}\right)$ and $\{\lambda_n\} \subset [c, d]$ for some $c, d \in \left(0, \frac{1}{L}\right)$, $S : \mathcal{H}_2 \to \mathcal{H}_2$ is a nonexpansive mapping, and $A : \mathcal{C} \to \mathcal{H}_1$ is a monotone and L-Lipschitz continuous operator.

Appendix 4.5.13. The Algorithm in Tian and Jiang [244]. Let $x_1 \in C$, define the sequences $\{x_n\}$, $\{y_n\}$, $\{w_n\}$ and $\{t_n\}$ by

$$\begin{cases} y_n = P_{\mathcal{C}}(x_n - \tau_n T^* (I - S) T x_n), \\ t_n = P_{\mathcal{C}}(y_n - \lambda_n A(y_n)), \\ w_n = P_{\mathcal{C}}(y_n - \lambda_n A(t_n)), \\ x_{n+1} = \alpha_n g(x_n) + (1 - \alpha_n) w_n, \ n \ge 1, \end{cases}$$
(4.5.42)

where $\{\alpha_n\} \subset (0,1)$ with $\lim_{n\to\infty} \alpha_n = 0$ and $\sum_{n=1}^{\infty} \alpha_n = \infty$, $\{\tau_n\} \subset [a,b]$ for some $a,b \in \left(0,\frac{1}{\|T\|^2}\right)$ and $\{\lambda_n\} \subset [c,d]$ for some $c,d \in \left(0,\frac{1}{L}\right)$, $S: \mathcal{H}_2 \to \mathcal{H}_2$ is a nonexpansive mapping, $A: \mathcal{C} \to \mathcal{H}_1$ is a monotone and L-Lipschitz continuous operator and g is a contraction on \mathcal{C} .

Appendix 4.5.14. The Algorithm in Chidume and Nnakwe [76]. For $x_1 = x \in \mathcal{H}_1, \mathcal{C}_1 = \mathcal{H}_1$ and $W_1 = \mathcal{H}_1$, define the sequence $\{x_n\}$ by

$$\begin{cases} y_{n}^{i} = P_{\mathcal{C}_{i}} \left(x_{n} - \tau T^{*} (I - S_{i}) T x_{n} \right), i = 1, \cdots, N \\ u_{n}^{i} = P_{\mathcal{C}_{i}} \left(y_{n}^{i} - \lambda_{n} A_{i} (y_{n}^{i}) \right), i = 1, \cdots, N \\ t_{n}^{i} = P_{\mathcal{C}_{i}} \left(y_{n}^{i} - \lambda_{n} A_{i} (u_{n}^{i}) \right), i = 1, \cdots, N \\ \mathcal{C}_{n}^{i} = \left\{ z \in \mathcal{H} : \| t_{n}^{i} - z \| \leq \| x_{n} - z \| \right\} \\ W_{n} = \left\{ z \in \mathcal{H} : \langle z - x_{n}, x - x_{n} \rangle \leq 0 \right\} \\ W_{n+1} = P_{\mathcal{C}_{n}} \cap W_{n} x, \ \forall n \geq 1, \end{cases}$$

$$(4.5.43)$$

where $C_n = \bigcap_{i=1}^N C_n^i, \tau \in \left(0, \frac{1}{\|A\|^2}\right), \lambda_n \in \left(0, \frac{1}{L}\right), C_i, i = 1, \cdots, N$ are nonempty closed and convex subsets of \mathcal{H}_1 such that $\mathcal{C} = \bigcap_{i=1}^N C_i \neq \emptyset, T : \mathcal{H}_1 \to \mathcal{H}_2$ is a bounded linear operator such that $T \neq 0$ and T^* is the adjoint of T. $A_i : C_i \to \mathcal{H}_1, i = 1, \cdots, N$ is a finite family of monotone and L-Lipschitz mappings and $S_i : \mathcal{H}_2 \to \mathcal{H}_2, i = 1, \cdots, N$ is a finite family of nonexpansive mappings.

CHAPTER 5

Results on Split Equalities Variational Inclusion and Split Equilibrium Problems

5.1 Introduction

In this chapter, we propose an inertial Tseng's extragradient algorithm for approximating a common solution of split equalities of VIP, EP and FPP of nonexpansive semigroups mapping. Furthermore, we propose and study an iterative algorithm for approximating the common solution of V_q IP and SEP in the framework of real Hilbert spaces and apply our result to study certain optimization problems. Finally, we present some numerical examples of our proposed methods in comparison with other methods in the literature to show the applicability of our proposed methods.

5.2 On split equalities of some nonlinear optimization problems

In this work, we introduce a new inertial Tseng's extragradient method with self-adaptive step sizes for approximating a common solution of split equalities of equilibrium problem (EP), non-Lipschitz pseudomonotone variational inequality problem (VIP) and fixed point problem (FPP) of nonexpansive semigroups in real Hilbert spaces. We prove that the sequence generated by our proposed method converges strongly to a common solution of the EP, pseudomonotone VIP and FPP of nonexpansive semigroups mapping without any linesearch procedure nor the sequential weak continuity condition often assumed by authors when solving non-Lipschitz VIPs. Finally, we provide some numerical experiments for the proposed method in comparison with related method in the literature. Our result improves, extends and generalizes several of the existing results in this direction. The following assumptions are required in solving equilibrium problems.

Assumption 5.2.1. [43] Let $F : C \times C \to \mathbb{R}$ be a bifunction satisfying the following assumptions:

- 1) $F(x,x) = 0, \quad \forall x \in \mathcal{C};$
- 2) F is monotone, i.e., $F(x,y) + F(y,x) \le 0$, $\forall x \in C$;
- 3) for each $x, y, z \in \mathcal{C}$, $\limsup_{t \to 0} F(tz + (1-t)x, y) \le F(x, y);$
- 4) for each $x \in \mathcal{C}, y \to F(x, y)$ is convex and lower semi continuous.

Lemma 5.2.1. [87] Let $F : \mathcal{C} \times \mathcal{C} \to \mathbb{R}$ be a bifunction satisfying Assumption 5.2.1. For any r > 0 and $x \in \mathcal{H}$, define a mapping $T_r^F : \mathcal{H} \to \mathcal{C}$ as follows

$$T_r^F(x) = \left\{ z \in \mathcal{C} : F(z, y) + \frac{1}{r} \left\langle y - z, z - x \right\rangle \ge 0, \ \forall y \in \mathcal{C} \right\}.$$

Then, we have the following

- (1) T_r^F is nonempty and single valued;
- (2) T_r^F is firmly nonexpansive, that is

$$\left\langle T_r^F x - T_r^F y, \ x - y \right\rangle \ge \|T_r^F x - T_r^F y\|^2;$$

equivalently

$$||T_r^F x - T_r^F y||^2 \le ||x - y||^2 - ||(I - T_r^F) x - (I - T_r^F) y||^2, \quad \forall \ x, y \in \mathcal{H};$$

(3) $F(T_r^F) = EP(F)$ is closed and convex.

5.2.1 Proposed method

In this section, we present our proposed method and discuss its features. We begin with the following assumptions under which our strong convergence result is obtained.

Assumption 5.2.2. Suppose that the following conditions hold:

- (a) The feasible sets C and Q are nonempty, closed and convex subsets of the real Hilbert spaces \mathcal{H}_1 and \mathcal{H}_2 , respectively.
- (b) $A: \mathcal{H}_1 \to \mathcal{H}_1$ and $B: \mathcal{H}_2 \to \mathcal{H}_2$ are pseudomonotone and uniformly continuous.

- (c) The mapping $A : \mathcal{H}_1 \to \mathcal{H}_1$ and $B : \mathcal{H}_2 \to \mathcal{H}_2$ satisfies the following property: whenever $\{x_n\} \subset \mathcal{C}, \quad x_n \rightharpoonup x^*$, one has $||Ax^*|| \leq \liminf_{n \to \infty} ||Ax_n||$ and whenever $\{x_n\} \subset \mathcal{Q}, \quad x_n \rightharpoonup x^*$, one has $||Bx^*|| \leq \liminf_{n \to \infty} ||Bx_n||$, respectively.
- (d) $\mathcal{F}_1: \mathcal{H}_1 \to \mathcal{H}_3$ and $\mathcal{F}_2: \mathcal{H}_2 \to \mathcal{H}_3$ are bounded linear operators.
- (e) $\Phi_1 : \mathcal{C} \times \mathcal{C} \to \mathbb{R}, \ \Phi_2 : \mathcal{Q} \times \mathcal{Q} \to \mathbb{R}$ are bifunctions satisfying Assumption (5.2.1) and Φ_2 is upper semi continuous in the first argument.
- (f) $\mathcal{T}_a = \{T_1(s) : 0 \leq s < \infty\}$ and $\mathcal{T}_b = \{T_2(u) : 0 \leq u < \infty\}$ are one-parameter nonexpansive semigroups on \mathcal{H}_1 and \mathcal{H}_2 , respectively.
- (g) The solution set $\Gamma = \left\{ x \in EP(\Phi_1) \cap VI(\mathcal{C}, A) \cap F(\mathcal{T}_a), y \in EP(\Phi_2) \cap VI(\mathcal{Q}, B) \cap F(\mathcal{T}_b) : \mathcal{F}_1 x = \mathcal{F}_2 y \right\} \neq \emptyset.$
- $\begin{array}{ll} (h) \ \{\alpha_n\} \subset (0,1), \ \sum_{n=1}^{\infty} \alpha_n = +\infty, & \lim_{n \to \infty} \alpha_n = 0, & 0 < \liminf_{n \to \infty} \beta_n < \limsup_{n \to \infty} \beta_n < 1, & 0 < \lim_{n \to \infty} \inf_{n \to \infty} \gamma_n < 1. \end{array}$
- (i) Let $\{\epsilon_n\}$ and $\{\zeta_n\}$ be positive sequences such that $\lim_{n\to\infty}\frac{\epsilon_n}{\alpha_n} = 0$ and $\lim_{n\to\infty}\frac{\zeta_n}{\alpha_n} = 0$, respectively.
- (j) Let $\{\sigma_n\}$ and $\{\mu_n\}$ be a nonnegative sequences such that $\sum_{n=1}^{\infty} \sigma_n < +\infty$ and $\sum_{n=1}^{\infty} \mu_n < +\infty$, respectively, $\{t_{n,1}\}, \{t_{n,2}\} \subset (0, +\infty)$, $\liminf r_{n,1} > 0$, $\liminf r_{n,2} > 0$

Algorithm 5.2.3.

Step 0: Choose sequences $\{\beta_n\}_{n=1}^{\infty}, \{\gamma_n\}_{n=1}^{\infty}, \{\theta_n\}_{n=1}^{\infty}$ and $\{\tau_n\}_{n=1}^{\infty}$ such that the conditions from Assumption 5.2.2 (h)-(i) hold. Select initial point $(x_0, y_0) \in \mathcal{H}_1 \times \mathcal{H}_2$, let $\eta \ge 0$, $a_i \in (0, 1), i = 1, 2, \lambda_1 > 0, \rho_1 > 0, \theta > 0, \tau > 0$ and set n := 1.

Step 1: Given the iterates x_{n-1}, y_{n-1} and x_n, y_n for each $n \ge 1$, choose θ_n such that $0 \le \theta_n \le \overline{\theta}_n$ and τ_n such that $0 \le \tau_n \le \overline{\tau}_n$ where

$$\bar{\theta}_n := \begin{cases} \min\left\{\theta, \frac{\epsilon_n}{\|x_n - x_{n-1}\|}\right\}, & \text{if } x_n \neq x_{n-1} \\ \theta, & \text{otherwise.} \end{cases}$$
(5.2.1)

Step 2: Compute

$$w_n = (1 - \alpha_n) \Big(x_n + \theta_n (x_n - x_{n-1}) \Big).$$

Step 3: Compute

$$z_n = w_n - \eta_n \mathcal{F}_1^* (\mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n),$$

$$\phi_n = U_{r_{n,1}}^{\Phi_1} z_n,$$

$$u_n = P_{\mathcal{C}}(\phi_n - \lambda_n A \phi_n),$$

$$v_n = u_n - \lambda_n (Au_n - A\phi_n),$$
$$x_{n+1} = (1 - \beta_n)v_n + \beta_n \frac{1}{t_{n,1}} \int_0^{t_{n,1}} T_1(s)v_n \ ds$$

and

 $\lambda_{n+1} = \begin{cases} \min\left\{\frac{a_1||u_n - \phi_n||}{||Au_n - A\phi_n||}, \ \lambda_n + \sigma_n\right\}, & \text{if } Au_n \neq A\phi_n\\ \lambda_n + \sigma_n, & \text{otherwise.} \end{cases}$ (5.2.2)

Step 4: Compute

$$\bar{\tau}_n := \begin{cases} \min\left\{\tau, \frac{\zeta_n}{\|y_n - y_{n-1}\|}\right\}, & \text{if } y_n \neq y_{n-1} \\ \tau, & \text{otherwise.} \end{cases}$$
(5.2.3)

Step 5 Compute

$$\varphi_n = (1 - \alpha_n) \Big(y_n + \tau_n (y_n - y_{n-1}) \Big),$$

$$k_n = \varphi_n + \eta_n \mathcal{F}_2^* (\mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n),$$

- /-

Step 6: Compute

$$\psi_{n} = U_{r_{n,2}}^{\Phi_{2}} k_{n},$$

$$s_{n} = P_{\mathcal{Q}}(\psi_{n} - \rho_{n} B \psi_{n}),$$

$$b_{n} = s_{n} - \rho_{n} (Bs_{n} - B\psi_{n}),$$

$$y_{n+1} = (1 - \gamma_{n})b_{n} + \gamma_{n} \frac{1}{t_{n,2}} \int_{0}^{t_{n,2}} T_{2}(u)b_{n} du$$

and

$$\rho_{n+1} = \begin{cases} \min\left\{\frac{a_2||s_n - \psi_n||}{||Bs_n - B\psi_n||}, \ \rho_n + \mu_n\right\}, & \text{if } Bs_n \neq B\psi_n\\ \rho_n + \mu_n, & \text{otherwise,} \end{cases}$$
(5.2.4)

where the step size η_n is chosen such that for small enough $\epsilon > 0$,

$$\eta_n \in \left[\epsilon, \quad \frac{2\|\mathcal{F}_1w_n - \mathcal{F}_2\varphi_n\|^2}{\|\mathcal{F}_2^*(\mathcal{F}_1w_n - \mathcal{F}_2\varphi_n)\|^2 + \|\mathcal{F}_1^*(\mathcal{F}_1w_n - \mathcal{F}_2\varphi_n)\|^2} - \epsilon\right],$$

if $\mathcal{F}_1 w_n \neq \mathcal{F}_2 \varphi_n$; otherwise, $\eta_n = \eta$. Set n := n + 1 and go back to Step 1.

Remark 5.2.2. The step sizes generated in (5.2.2) and (5.2.4) are allowed to increase per iteration. This reduces their dependence on the initial step sizes. When n is large enough the step size may not increase. We assume that the Algorithm 5.2.3 does not terminate in finite number of iterations.

Remark 5.2.3. By conditions (h) and (i), from (5.2.1) we observe that

$$\lim_{n \to \infty} \theta_n ||x_n - x_{n-1}|| = 0 \text{ and } \lim_{n \to \infty} \frac{\theta_n}{\alpha_n} ||x_n - x_{n-1}|| = 0.$$
 (5.2.5)

Similarly, from (5.2.3) we have

$$\lim_{n \to \infty} \tau_n ||y_n - y_{n-1}|| = 0 \text{ and } \lim_{n \to \infty} \frac{\tau_n}{\alpha_n} ||y_n - y_{n-1}|| = 0.$$
 (5.2.6)

Remark 5.2.4. Since the limit

$$\lim_{n \to \infty} \left(1 - \frac{\lambda_n^2 a_1^2}{\lambda_{n+1}^2} \right) = 1 - a_1^2 > 0, \tag{5.2.7}$$

there exists $n_{0_1} > 0$ such that for all $n > n_{0_1}$, we have $\left(1 - \frac{\lambda_n^2 a_1^2}{\lambda_{n+1}^2}\right) > 0$. Similarly, we have that

$$\lim_{n \to \infty} \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2} \right) = 1 - a_2^2 > 0, \tag{5.2.8}$$

and there exists $n_{0_2} > 0$ such that for all $n > n_{0_2}$, we have $\left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) > 0$. Now, we set $n_0 = \max\{n_{0_1}, n_{0_2}\}$.

Remark 5.2.5. From the definition of η_n , that is,

$$\eta_n \in \left[\epsilon, \quad \frac{2\|\mathcal{F}_1w_n - \mathcal{F}_2\varphi_n\|^2}{\|\mathcal{F}_2^*(\mathcal{F}_1w_n - \mathcal{F}_2\varphi_n)\|^2 + \|\mathcal{F}_1^*(\mathcal{F}_1w_n - \mathcal{F}_2\varphi_n)\|^2} - \epsilon\right]$$

we have

$$(\eta_n + \epsilon) \Big[\|\mathcal{F}_2^*(\mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n)\|^2 + \|\mathcal{F}_1^*(\mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n)\|^2 \Big] \le 2 \|\mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n\|^2.$$

Expanding the last inequality, we have

$$\eta_{n} \cdot \epsilon \Big[\|\mathcal{F}_{2}^{*}(\mathcal{F}_{1}w_{n} - \mathcal{F}_{2}\varphi_{n})\|^{2} + \|\mathcal{F}_{1}^{*}(\mathcal{F}_{1}w_{n} - \mathcal{F}_{2}\varphi_{n}\|)^{2} \Big] \leq \eta_{n} \Big(2\|\mathcal{F}_{1}w_{n} - \mathcal{F}_{2}\varphi_{n}\|^{2} - \eta_{n} \Big[\|\mathcal{F}_{2}^{*}(\mathcal{F}_{1}w_{n} - \mathcal{F}_{2}\varphi_{n})\|^{2} + \\\|\mathcal{F}_{1}^{*}(\mathcal{F}_{1}w_{n} - \mathcal{F}_{2}\varphi_{n})\|^{2} \Big] \Big).$$
(5.2.9)

5.2.2 Convergence analysis

Lemma 5.2.6. Let $\{\lambda_n\}$ and $\{\rho_n\}$ be sequences generated by Algorithm 5.2.3. Then, we have $\lim_{n\to\infty} \lambda_n = \lambda$, where $\lambda \in \left[\min\{\frac{a_1}{K_1}, \lambda_1\}, \lambda_1 + b_1\right], b_1 = \sum_{n=1}^{\infty} \sigma_n$, for some $K_1 > 0$ and $\lim_{n\to\infty} \rho_n = \rho$, where $\rho \in \left[\min\{\frac{a_2}{K_2}, \rho_1\}, \rho_1 + b_2\right], b_2 = \sum_{n=1}^{\infty} \mu_n$, for some $K_2 > 0$.

Proof. Since A is uniformly continuous, we obtain from (2.1.4) that for any given $\epsilon > 0$, there exists a constant $M < +\infty$ such that $||Au_n - A\phi_n|| \le M ||u_n - \phi_n|| + \epsilon$. Thus, when $Au_n - A\phi_n \ne 0$ for all $n \ge 1$ we have

$$\frac{a_1 \|u_n - \phi_n\|}{\|Au_n - A\phi_n\|} \ge \frac{a_1 \|u_n - \phi_n\|}{M \|u_n - \phi_n\| + \epsilon} = \frac{a_1 \|u_n - \phi_n\|}{(M + \epsilon_1) \|u_n - \phi_n\|} = \frac{a_1}{K_1},$$

where $\epsilon = \epsilon_1 \|u_n - \phi_n\|$ for some $\epsilon_1 \in (0, 1)$ and $K_1 = M + \epsilon_1$. Hence, from the definition of λ_{n+1} , the sequence $\{\lambda_n\}$ is bounded below by $\min\{\frac{a_1}{K_1}, \lambda_1\}$ and above by $\lambda_1 + b_1$. By Lemma 2.5.34, it follows that $\lim_{n \to \infty} \lambda_n$ denoted by $\lambda = \lim_{n \to \infty} \lambda_n$ exists. Clearly, we have $\lambda \in [\min\{\frac{a_1}{K_1}, \lambda_1\}, \lambda_1 + b_1].$

Similarly, we have $\lim_{n \to \infty} \rho_n = \rho$, and $\rho \in \left[\min\{\frac{a_2}{K_2}, \rho_1\}, \rho_1 + b_2\right]$.

Lemma 5.2.7. Let $\{(x_n, y_n)\}$ be a sequence generated by Algorithm 5.2.3 under Assumption 5.2.2. Then

$$||z_n - x^*||^2 + ||k_n - y^*||^2 \le ||w_n - x^*||^2 + ||\varphi_n - y^*||^2.$$

Proof. Let $(x^*, y^*) \in \Gamma$. Then, by applying Lemma 2.1.1, we have

$$\begin{aligned} \|z_n - x^*\|^2 &= \|w_n - \eta_n \mathcal{F}_1^* (\mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n) - x^*\|^2 \\ &= \|w_n - x^*\|^2 + \eta_n^2 \|\mathcal{F}_1^* (\mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n)\|^2 - 2\eta_n \left\langle w_n - x^*, \ \mathcal{F}_1^* (\mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n) \right\rangle \\ &= \|w_n - x^*\|^2 + \eta_n^2 \|\mathcal{F}_1^* (\mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n)\|^2 - 2\eta_n \left\langle \mathcal{F}_1 w_n - \mathcal{F}_1 x^*, \ \mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n \right\rangle \\ &= \|w_n - x^*\|^2 + \eta_n^2 \|\mathcal{F}_1^* (\mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n)\|^2 - \eta_n \|\mathcal{F}_1 w_n - \mathcal{F}_1 x^*\|^2 - (5.2.10) \\ &\eta_n \|\mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n\|^2 + \eta_n \|\mathcal{F}_2 \varphi_n - \mathcal{F}_1 x^*\|^2. \end{aligned}$$

Similarly, we have

$$||k_n - y^*||^2 = ||\varphi_n + \eta_n \mathcal{F}_2^* (\mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n) - y^*||^2$$

= $||\varphi_n - y^*||^2 + \eta_n^2 ||\mathcal{F}_2^* (\mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n)||^2 - \eta_n ||\mathcal{F}_2 \varphi_n - \mathcal{F}_2 y^*||^2$ (5.2.11)
 $- \eta_n ||\mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n||^2 + \eta_n ||\mathcal{F}_1 w_n - \mathcal{F}_2 y^*||^2$

Adding (5.2.10) and (5.2.11), we have

$$\begin{aligned} \|z_n - x^*\|^2 + \|k_n - y^*\|^2 &= \|w_n - x^*\|^2 + \|\varphi_n - y^*\|^2 \\ &+ \eta_n^2 \Big[\|\mathcal{F}_1^*(\mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n)\|^2 + \|\mathcal{F}_2^*(\mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n)\|^2 \Big] \\ &- \eta_n \Big[\|\mathcal{F}_1 w_n - \mathcal{F}_1 x^*\|^2 + \|\mathcal{F}_2 \varphi_n - \mathcal{F}_2 y^*\|^2 \Big] - 2\eta_n \|\mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n\|^2 \\ &+ \eta_n \Big[\|\mathcal{F}_1 w_n - \mathcal{F}_2 y^*\|^2 + \|\mathcal{F}_2 \varphi_n - \mathcal{F}_1 x^*\|^2 \Big]. \end{aligned}$$

By (5.2.9) and the fact that $\mathcal{F}_1 x^* = \mathcal{F}_2 y^*$, we have

$$||z_{n} - x^{*}||^{2} + ||k_{n} - y^{*}||^{2} = ||w_{n} - x^{*}||^{2} + ||\varphi_{n} - y^{*}||^{2} - \eta_{n} \Big[2||\mathcal{F}_{1}w_{n} - \mathcal{F}_{2}\varphi_{n}||^{2} - \eta_{n} \Big(||\mathcal{F}_{1}^{*}(\mathcal{F}_{1}w_{n} - \mathcal{F}_{2}\varphi_{n})||^{2} + ||\mathcal{F}_{2}^{*}(\mathcal{F}_{1}w_{n} - \mathcal{F}_{2}\varphi_{n})||^{2} \Big) \Big] \leq ||w_{n} - x^{*}||^{2} + ||\varphi_{n} - y^{*}||^{2} - \eta_{n} \cdot \epsilon \Big[||\mathcal{F}_{2}^{*}(\mathcal{F}_{1}w_{n} - \mathcal{F}_{2}\varphi_{n})||^{2} + ||\mathcal{F}_{1}^{*}(\mathcal{F}_{1}w_{n} - \mathcal{F}_{2}\varphi_{n})||^{2} \Big] \leq ||w_{n} - x^{*}||^{2} + ||\varphi_{n} - y^{*}||^{2},$$
(5.2.12)

which is the desired result.

Lemma 5.2.8. Let $\{(x_n, y_n)\}$ be a sequence generated by Algorithm 5.2.3 under Assumption 5.2.2. Then

$$\|v_n - x^*\|^2 \le \|z_n - x^*\|^2 - \|z_n - \phi_n\|^2 - \left(1 - \frac{\lambda_n^2 a_1^2}{\lambda_{n+1}^2}\right)\|u_n - \phi_n\|^2$$

and

$$\|b_n - y^*\|^2 \le \|k_n - y^*\|^2 - \|k_n - \psi_n\|^2 - \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right)\|s_n - \psi_n\|^2.$$

Proof. Let $(x^*, y^*) \in \Gamma$. Since $U_{r_{n,1}}^{\Phi_1}$ is firmly nonexpansive, it follows from Lemma 5.2.1 that

$$\|\phi_n - x^*\|^2 = \|U_{r_{n,1}}^{\Phi_1} z_n - x^*\| \le \|z_n - x^*\|^2 - \|z_n - \phi_n\|^2.$$
 (5.2.13)

Similarly, we have

$$\|\psi_n - y^*\|^2 = \|U_{r_{n,2}}^{\Phi_2} k_n - y^*\| \le \|k_n - y^*\|^2 - \|k_n - \psi_n\|^2.$$
 (5.2.14)

From (5.2.2), we obtain

$$\lambda_{n+1} = \min\left\{\frac{a_1 \|u_n - \phi_n\|}{\|Au_n - A\phi_n\|}, \lambda_n + \sigma_n\right\} \le \frac{a_1 \|u_n - \phi_n\|}{\|Au_n - A\phi_n\|},$$

which implies that

$$||Au_n - A\phi_n|| \le \frac{a_1}{\lambda_{n+1}} ||u_n - \phi_n||, \quad \forall n \ge 1.$$
 (5.2.15)

Similarly, we have

$$||Bs_n - B\psi_n|| \le \frac{a_2}{\rho_{n+1}} ||s_n - \psi_n||, \quad \forall n \ge 1.$$
(5.2.16)

From the definition of v_n in **Step 3** and Lemma 2.1.1, we have

$$\begin{aligned} \|v_n - x^*\|^2 &\leq \|u_n - \lambda_n (Au_n - A\phi_n) - x^*\|^2 \\ &= \|u_n - x^*\|^2 + \lambda_n^2 \|Au_n - A\phi_n\|^2 - 2\lambda_n \langle Au_n - A\phi_n, u_n - x^* \rangle \\ &= \|\phi_n - x^*\|^2 + \|u_n - \phi_n\|^2 + 2\langle u_n - \phi_n, \phi_n - x^* \rangle + \lambda_n^2 \|Au_n - A\phi_n\|^2 \\ &- 2\lambda_n \langle Au_n - A\phi_n, u_n - x^* \rangle \\ &= \|\phi_n - x^*\|^2 + \|u_n - \phi_n\|^2 - 2\langle u_n - \phi_n, u_n - \phi_n \rangle + 2\langle u_n - \phi_n, u_n - x^* \rangle \\ &- 2\lambda_n \langle Au_n - A\phi_n, u_n - x^* \rangle + \lambda_n^2 \|Au_n - A\phi_n\|^2 \\ &= \|\phi_n - x^*\|^2 - \|u_n - \phi_n\|^2 + 2\langle u_n - \phi_n, u_n - x^* \rangle \\ &- 2\lambda_n \langle Au_n - A\phi_n, u_n - x^* \rangle + \lambda_n^2 \|Au_n - A\phi_n\|^2. \end{aligned}$$
(5.2.17)

Since $u_n = P_{\mathcal{C}}(\phi_n - \lambda_n A \phi_n)$ and $x^* \in \mathcal{C}$, we obtain from the characteristic property of $P_{\mathcal{C}}$ that

$$\langle u_n - \phi_n + \lambda_n A \phi_n, u_n - x^* \rangle \le 0.$$

This implies that

$$\langle u_n - \phi_n, u_n - x^* \rangle \le -\lambda_n \langle A\phi_n, u_n - x^* \rangle.$$
 (5.2.18)

Also since $u_n \in \mathcal{C}$ and $x^* \in \Gamma$ we have

$$\langle Au_n, u_n - x^* \rangle \ge 0, \quad \forall \ n \ge 0.$$
 (5.2.19)

Applying (5.2.13), (5.2.15), (5.2.18) and (5.2.19) in (5.2.17), we obtain

$$\begin{aligned} \|v_n - x^*\|^2 &\leq \|\phi_n - x^*\|^2 - \|u_n - \phi_n\|^2 - 2\lambda_n \langle A\phi_n, u_n - x^* \rangle - 2\lambda_n \langle Au_n - A\phi_n, u_n - x^* \rangle \\ &+ \lambda_n^2 \|Au_n - A\phi_n\|^2 \\ &= \|\phi_n - x^*\|^2 - \|u_n - \phi_n\|^2 - 2\lambda_n \langle Au_n, u_n - x^* \rangle + \lambda_n^2 \|Au_n - A\phi_n\|^2 \\ &\leq \|\phi_n - x^*\|^2 - \|u_n - \phi_n\|^2 + \lambda_n^2 \frac{a_1^2}{\lambda_{n+1}^2} \|u_n - \phi_n\|^2 \\ &= \|z_n - x^*\|^2 - \|z_n - \phi_n\|^2 - \left(1 - \frac{\lambda_n^2 a_1^2}{\lambda_{n+1}^2}\right) \|u_n - \phi_n\|^2. \end{aligned}$$
(5.2.20)

Following the same line of argument, we have

$$\|b_n - y^*\|^2 \le \|k_n - y^*\|^2 - \|k_n - \psi_n\|^2 - \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2,$$
(5.2.21)

which completes the proof.

Lemma 5.2.9. Let $\{(x_n, y_n)\}$ be a sequence generated by Algorithm 5.2.3 satisfying Assumption 5.2.2. Then $\{(x_n, y_n)\}$ is bounded.

Proof. Let $x^* \in \Gamma$. From the definition of w_n and Lemma 2.1.1, we have

$$\|w_{n} - x^{*}\| = \|(1 - \alpha_{n})(x_{n} + \theta_{n}(x_{n} - x_{n-1})) - x^{*}\|$$

$$= \|(1 - \alpha_{n})(x_{n} - x^{*}) + (1 - \alpha_{n})\theta_{n}(x_{n} - x_{n-1}) - \alpha_{n}x^{*}\|$$

$$\leq (1 - \alpha_{n})\|x_{n} - x^{*}\| + (1 - \alpha_{n})\theta_{n}\|x_{n} - x_{n-1}\| + \alpha_{n}\|x^{*}\|$$

$$= (1 - \alpha_{n})\|x_{n} - x^{*}\| + \alpha_{n}\Big[(1 - \alpha_{n})\frac{\theta_{n}}{\alpha_{n}}\|x_{n} - x_{n-1}\| + \|x^{*}\|\Big].$$
(5.2.22)

By (5.2.5), we have

$$\lim_{n \to \infty} \left[(1 - \alpha_n) \frac{\theta_n}{\alpha_n} \| x_n - x_{n-1} \| + \| x^* \| \right] = \| x^* \|.$$

Thus, there exists a constant $M_1 > 0$ such that $(1 - \alpha_n) \frac{\theta_n}{\alpha_n} ||x_n - x_{n-1}|| + ||x^*|| \le M_1$ for all $n \in \mathbb{N}$. Thus, from (5.2.22) it follows that

$$||w_n - x^*|| \le (1 - \alpha_n) ||x_n - x^*|| + \alpha_n M_1.$$

Consequently, we have

$$\|w_n - x^*\|^2 \le (1 - \alpha_n)^2 \|x_n - x^*\|^2 + 2\alpha_n (1 - \alpha_n) M_1 \|x_n - x^*\| + \alpha_n^2 M_1^2.$$
 (5.2.23)

Following similar procedure, we have

$$\|\varphi_n - y^*\|^2 \le (1 - \alpha_n)^2 \|y_n - y^*\|^2 + 2\alpha_n (1 - \alpha_n) M_2 \|y_n - y^*\| + \alpha_n^2 M_2^2.$$
(5.2.24)

Adding (5.2.23) and (5.2.24), we obtain

$$\|w_{n} - x^{*}\|^{2} + \|\varphi_{n} - y^{*}\|^{2} \leq (1 - \alpha_{n})^{2} \Big[\|x_{n} - x^{*}\|^{2} + \|y_{n} - y^{*}\|^{2} \Big] + 2\alpha_{n}(1 - \alpha_{n}) \Big(M_{1} \|x_{n} - x^{*}\| + M_{2} \|y_{n} - y^{*}\| \Big) + \alpha_{n}^{2} (M_{1}^{2} + M_{2}^{2}) \leq (1 - \alpha_{n}) \Big[\|x_{n} - x^{*}\|^{2} + \|y_{n} - y^{*}\|^{2} \Big] + 2\alpha_{n} \Big(M_{1} \|x_{n} - x^{*}\| + M_{2} \|y_{n} - y^{*}\| \Big) + \alpha_{n} (M_{1}^{2} + M_{2}^{2}) = (1 - \alpha_{n}) \Big[\|x_{n} - x^{*}\|^{2} + \|y_{n} - y^{*}\|^{2} \Big] + \alpha_{n} c_{n},$$
(5.2.25)

where $c_n = 2\left(M_1 \|x_n - x^*\| + M_2 \|y_n - y^*\|\right) + M_1^2 + M_2^2$. From **STEP 3**, and by applying Lemma 2.1.1, (5.2.20) together with Remark 5.2.7, we have

$$\begin{aligned} \|x_{n+1} - x^*\|^2 &= \left\| (1 - \beta_n) v_n + \beta_n \frac{1}{t_{n,1}} \int_0^{t_{n,1}} T_1(s) v_n ds - x^* \right\|^2 \\ &= \left\| (1 - \beta_n) (v_n - x^*) + \beta_n \left(\frac{1}{t_{n,1}} \int_0^{t_{n,1}} T_1(s) v_n ds - x^* \right) \right\|^2 \\ &= (1 - \beta_n) \|v_n - x^*\|^2 + \beta_n \| \frac{1}{t_{n,1}} \int_0^{t_{n,1}} T_1(s) v_n ds - x^* \|^2 \\ &- \beta_n (1 - \beta_n) \left\| \frac{1}{t_{n,1}} \int_0^{t_{n,1}} T_1(s) v_n ds - v_n \right\|^2 \\ &= (1 - \beta_n) \|v_n - x^*\|^2 + \beta_n \| \frac{1}{t_{n,1}} \int_0^{t_{n,1}} T_1(s) v_n ds - \frac{1}{t_{n,1}} \int_0^{t_{n,1}} T_1(s) x^* ds \|^2 \\ &- \beta_n (1 - \beta_n) \left\| \frac{1}{t_{n,1}} \int_0^{t_{n,1}} T_1(s) v_n ds - v_n \right\|^2 \\ &\leq (1 - \beta_n) \|v_n - x^*\|^2 + \beta_n \|v_n - x^*\|^2 \\ &- \beta_n (1 - \beta_n) \left\| \frac{1}{t_{n,1}} \int_0^{t_{n,1}} T_1(s) v_n ds - v_n \right\|^2 \\ &\leq \|v_n - x^*\|^2 - \beta_n (1 - \beta_n) \right\| \frac{1}{t_{n,1}} \int_0^{t_{n,1}} T_1(s) v_n ds - v_n \|^2 \\ &\leq \|z_n - x^*\|^2 - \|z_n - \phi_n\|^2 - \left(1 - \frac{\lambda_n^2 a_1^2}{\lambda_{n+1}^2}\right) \|u_n - \phi_n\|^2 \\ &- \beta_n (1 - \beta_n) \left\| \frac{1}{t_{n,1}} \int_0^{t_{n,1}} T_1(s) v_n ds - v_n \right\|^2 \\ &\leq \|z_n - x^*\|^2. \end{aligned}$$
(5.2.27)

Similarly, from **STEP 5**, and by applying Lemma 2.1.1, (5.2.21) together with Remark 5.2.7, we have

$$\|y_{n+1} - y^*\|^2 \le \|k_n - y^*\|^2 - \|k_n - \psi_n\|^2 - \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2$$

$$- \gamma_n (1 - \gamma_n) \left\| \frac{1}{t_{n,2}} \int_0^{t_{n,2}} T_2(u) b_n du - b_n \right\|^2$$

$$\le \|k_n - y^*\|^2.$$
(5.2.29)

From (5.2.12), (5.2.25), (5.2.27) and (5.2.29), we have

$$\begin{aligned} \|x_{n+1} - x^*\|^2 + \|y_{n+1} - y^*\|^2 &\leq \|z_n - x^*\|^2 + \|k_n - y^*\|^2 \\ &\leq \|w_n - x^*\|^2 + \|\varphi_n - y^*\|^2 \\ &\leq (1 - \alpha_n) \Big[\|x_n - x^*\|^2 + \|y_n - y^*\|^2 \Big] + \alpha_n c_n \\ &\leq \max \ \{ \|x_n - x^*\|^2 + \|y_n - y^*\|^2, \ c_n \} \\ &\vdots \\ &\leq \max \ \{ \|x_{n_0} - x^*\|^2 + \|y_{n_0} - y^*\|^2, \ c_{n_0} \} \end{aligned}$$
Thus, $\{(x_n, y_n)\}$ is bounded. Consequently, $\{z_n\}, \{v_n\}, \{k_n\}$ and $\{b_n\}$ are also bounded.

Lemma 5.2.10. Let $\{(x_n, y_n)\}$ be a sequence generated by Algorithm 5.2.3 under Assumption 5.2.2. Then,

$$\begin{aligned} \|x_{n+1} - x^*\|^2 + \|y_{n+1} - y^*\|^2 &\leq (1 - \alpha_n) \left[\|x_n - x^*\|^2 + \|y_n - y^*\|^2 \right] + \alpha_n d_n \\ &- \eta_n \cdot \epsilon \left[\|\mathcal{F}_2^* (\mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n)\|^2 + \|\mathcal{F}_1^* (\mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n)\|^2 \right] \\ &- \|z_n - \phi_n\|^2 - \|k_n - \psi_n\|^2 - \left(1 - \frac{\lambda_n^2 a_1^2}{\lambda_{n+1}^2}\right) \|u_n - \phi_n\|^2 \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \beta_n (1 - \beta_n) \|\frac{1}{t_{n,1}} \int_0^{t_{n,1}} T_1(s) v_n ds - v_n\|^2 \\ &- \gamma_n (1 - \gamma_n) \|\frac{1}{t_{n,2}} \int_0^{t_{n,2}} T_2(u) b_n du - b_n\|^2. \end{aligned}$$

Proof. Let $(x^*, y^*) \in \Gamma$. From Lemma 2.1.1 and the definition of w_n , we have

$$\begin{split} \|w_{n} - x^{*}\|^{2} &= \|(1 - \alpha_{n})(x_{n} - x^{*}) + (1 - \alpha_{n})\theta_{n}(x_{n} - x_{n-1}) - \alpha_{n}x^{*}\|^{2} \\ &\leq \|(1 - \alpha_{n})(x_{n} - x^{*}) + (1 - \alpha_{n})\theta_{n}(x_{n} - x_{n-1})\|^{2} + 2\alpha_{n}\langle -x^{*}, w_{n} - x^{*}\rangle \\ &\leq (1 - \alpha_{n})^{2}\|x_{n} - x^{*}\|^{2} + 2(1 - \alpha_{n})\theta_{n}\|x_{n} - x^{*}\|\|x_{n} - x_{n-1}\| + \theta_{n}^{2}\|x_{n} - x_{n-1}\|^{2} \\ &+ 2\alpha_{n}\langle -x^{*}, w_{n} - x_{n+1}\rangle + 2\alpha_{n}\langle -x^{*}, x_{n+1} - x^{*}\rangle \\ &\leq (1 - \alpha_{n})\|x_{n} - x^{*}\|^{2} \\ &+ \alpha_{n}\Big[2(1 - \alpha_{n})\|x_{n} - x^{*}\|\frac{\theta_{n}}{\alpha_{n}}\|x_{n} - x_{n-1}\| + \theta_{n}\|x_{n} - x_{n-1}\| \cdot \frac{\theta_{n}}{\alpha_{n}}\|x_{n} - x_{n-1}\| \\ &+ 2\|x^{*}\|\|w_{n} - x_{n+1}\| + 2\langle x^{*}, x^{*} - x_{n+1}\rangle\Big]. \end{split}$$

Following the same line of argument, we have

$$\begin{aligned} \|\varphi_{n} - y^{*}\|^{2} &\leq (1 - \alpha_{n}) \|y_{n} - y^{*}\|^{2} \\ &+ \alpha_{n} \Big[2(1 - \alpha_{n}) \|y_{n} - y^{*}\| \frac{\tau_{n}}{\alpha_{n}} \|y_{n} - y_{n-1}\| + \tau_{n} \|y_{n} - y_{n-1}\| \cdot \frac{\tau_{n}}{\alpha_{n}} \|y_{n} - y_{n-1}\| \\ &+ 2 \|y^{*}\| \|\varphi_{n} - y_{n+1}\| + 2\langle y^{*}, y^{*} - y_{n+1}\rangle \Big]. \end{aligned}$$

$$(5.2.31)$$

Adding (5.2.30) and (5.2.31) we have

$$\|w_n - x^*\|^2 + \|\varphi_n - y^*\|^2 \le (1 - \alpha_n) \Big[\|x_n - x^*\|^2 + \|y_n - y^*\|^2 \Big] + \alpha_n d_n$$
 (5.2.32)

where $d_n = \left[2(1-\alpha_n)\|x_n - x^*\|\frac{\theta_n}{\alpha_n}\|x_n - x_{n-1}\| + \theta_n\|x_n - x_{n-1}\| \cdot \frac{\theta_n}{\alpha_n}\|x_n - x_{n-1}\| + 2\|x^*\|\|w_n - x_{n+1}\| + 2\langle x^*, x^* - x_{n+1}\rangle\right] + \left[2(1-\alpha_n)\|y_n - y^*\|\frac{\tau_n}{\alpha_n}\|y_n - y_{n-1}\| + \tau_n\|y_n - y_{n-1}\| \cdot \frac{\tau_n}{\alpha_n}\|y_n - y_{n-1}\| + 2\|y^*\|\|\varphi_n - y_{n+1}\| + 2\langle y^*, y^* - y_{n+1}\rangle\right].$

From (5.2.12), (5.2.26), (5.2.28) and (5.2.32), we have

$$\begin{split} \|x_{n+1} - x^*\|^2 + \|y_{n+1} - y^*\|^2 &\leq \|z_n - x^*\|^2 + \|k_n - y^*\|^2 - \|z_n - \phi_n\|^2 - \|k_n - \psi_n\|^2 \\ &- \left(1 - \frac{\lambda_n^2 a_1^2}{\lambda_{n+1}^2}\right) \|u_n - \phi_n\|^2 - \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \beta_n (1 - \beta_n) \|\frac{1}{t_{n,1}} \int_0^{t_{n,1}} T_1(s) v_n ds - v_n\|^2 \\ &- \gamma_n (1 - \gamma_n) \|\frac{1}{t_{n,2}} \int_0^{t_{n,2}} T_2(u) b_n du - b_n\|^2 \\ &\leq \|w_n - x^*\|^2 + \|\varphi_n - y^*\|^2 \\ &- \eta_n \cdot \epsilon \left[\|\mathcal{F}_2^*(\mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n)\|^2 + \|\mathcal{F}_1^*(\mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n)\|^2 \right] \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \beta_n (1 - \beta_n) \|\frac{1}{t_{n,2}} \int_0^{t_{n,2}} T_2(u) b_n du - b_n\|^2 \\ &\leq (1 - \alpha_n) \left[\|x_n - x^*\|^2 + \|y_n - y^*\|^2 \right] + \alpha_n d_n \\ &- \eta_n \cdot \epsilon \left[\|\mathcal{F}_2^*(\mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n)\|^2 + \|\mathcal{F}_1^*(\mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n)\|^2 \right] \\ &- \left\|z_n - \phi_n\|^2 - \|k_n - \psi_n\|^2 - \left(1 - \frac{\lambda_n^2 a_1^2}{\lambda_{n+1}^2}\right) \|u_n - \phi_n\|^2 \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_{n+1}^2}\right) \|s_n - \psi_n\|^2 \\ &- \left(1 - \frac{\rho_n^2 a_2^2}{\rho_$$

which is the required result.

Theorem 5.2.11. Let $\{(x_n, y_n)\}$ be a sequence generated by Algorithm 5.2.3 such that Assumption 5.2.2 holds. Then, the sequence $\{(x_n, y_n)\}$ converges strongly to $(\hat{x}, \hat{y}) \in \Gamma$, where $\hat{x} = P_D(0)$ and $\hat{y} = P_E(0), D := EP(\Phi_1) \cap VI(\mathcal{C}, A) \cap F(\mathcal{T}_a)$ and $E := EP(\Phi_2) \cap VI(\mathcal{Q}, B) \cap F(\mathcal{T}_b)$.

Proof. Let $(\hat{x}, \hat{y}) \in \Gamma$, where $\hat{x} = P_D(0)$ and $\hat{y} = P_E(0)$. Then, it follows from Lemma

5.2.10 that

$$\|x_{n+1} - \hat{x}\|^2 + \|y_{n+1} - \hat{y}\|^2 \le (1 - \alpha_n) \Big[\|x_n - \hat{x}\|^2 + \|y_n - \hat{y}\|^2 \Big] + \alpha_n \hat{d}_n$$
(5.2.33)

where $\hat{d}_n = \left[2(1-\alpha_n)\|x_n - \hat{x}\|\frac{\theta_n}{\alpha_n}\|x_n - x_{n-1}\| + \theta_n\|x_n - x_{n-1}\| \cdot \frac{\theta_n}{\alpha_n}\|x_n - x_{n-1}\| + 2\|\hat{x}\|\|w_n - x_{n+1}\| + 2\langle \hat{x}, \hat{x} - x_{n+1}\rangle\right] + \left[2(1-\alpha_n)\|y_n - \hat{y}\|\frac{\tau_n}{\alpha_n}\|y_n - y_{n-1}\| + \tau_n\|y_n - y_{n-1}\| \cdot \frac{\tau_n}{\alpha_n}\|y_n - y_{n-1}\| + 2\|\hat{y}\|\|\varphi_n - y_{n+1}\| + 2\langle \hat{y}, \hat{y} - y_{n+1}\rangle\right].$ Now, we claim that the sequence $\{\|x_n - \hat{x}\| + \|y_n - \hat{y}\|\}$ converges to zero. To show this, by Lemma 2.5.36 it suffices to show that $\limsup_{k \to \infty} \hat{d}_{n_k} \leq 0$ for every subsequence $\{\|x_{n_k} - \hat{x}\| + \|y_{n_k} - \hat{y}\|\}$ of $\{\|x_n - \hat{x}\| + \|y_n - \hat{y}\|\}$ satisfying

$$\liminf_{k \to \infty} \left(\left(\|x_{n_{k+1}} - \hat{x}\| + \|y_{n_{k+1}} - \hat{y}\| \right) - \left(\|x_{n_k} - \hat{x}\| + \|y_{n_k} - \hat{y}\| \right) \right) \ge 0.$$
 (5.2.34)

Suppose that $\{\|x_{n_k} - \hat{x}\| + \|y_{n_k} - \hat{y}\|\}$ is a subsequence of $\{\|x_n - \hat{x}\| + \|y_n - \hat{y}\|\}$ such that (5.2.34) holds. Again, from Lemma 5.2.10, we obtain

$$\begin{aligned} \eta_{n_{k}} \cdot \epsilon \left[\|\mathcal{F}_{2}^{*}(\mathcal{F}_{1}w_{n_{k}} - \mathcal{F}_{2}\varphi_{n_{k}})\|^{2} + \|\mathcal{F}_{1}^{*}(\mathcal{F}_{1}w_{n_{k}} - \mathcal{F}_{2}\varphi_{n_{k}})\|^{2} \right] + \|z_{n_{k}} - \phi_{n_{k}}\|^{2} + \|k_{n_{k}} - \psi_{n_{k}}\|^{2} \\ + \left(1 - \frac{\lambda_{n_{k}}^{2}a_{1}^{2}}{\lambda_{n_{k+1}}^{2}}\right)\|u_{n_{k}} - \phi_{n_{k}}\|^{2} + \left(1 - \frac{\rho_{n_{k}}^{2}a_{2}^{2}}{\rho_{n_{k+1}}^{2}}\right)\|s_{n_{k}} - \psi_{n_{k}}\|^{2} \\ + \beta_{n_{k}}(1 - \beta_{n_{k}})\|\frac{1}{t_{n_{k,1}}}\int_{0}^{t_{n_{k,1}}} T_{1}(s)v_{n_{k}}ds - v_{n_{k}}\|^{2} + \gamma_{n_{k}}(1 - \gamma_{n_{k}})\|\frac{1}{t_{n_{k,2}}}\int_{0}^{t_{n_{k,2}}} T_{2}(u)b_{n_{k}}du - b_{n_{k}}\|^{2} \\ \leq (1 - \alpha_{n_{k}})\left[\|x_{n_{k}} - \hat{x}\|^{2} + \|y_{n_{k}} - \hat{y}\|^{2}\right] - \left[\|x_{n_{k+1}} - \hat{x}\|^{2} + \|y_{n_{k+1}} - \hat{y}\|^{2}\right] + \alpha_{n_{k}}\hat{d}_{n_{k}}. \end{aligned}$$

From (5.2.34) and the condition on α_{n_k} we have

$$\begin{split} \lim_{k \to \infty} \left(\eta_{n_k} \cdot \epsilon \Big[\|\mathcal{F}_2^*(\mathcal{F}_1 w_{n_k} - \mathcal{F}_2 \varphi_{n_k})\|^2 + \|\mathcal{F}_1^*(\mathcal{F}_1 w_{n_k} - \mathcal{F}_2 \varphi_{n_k})\|^2 \Big] + \|z_{n_k} - \phi_{n_k}\|^2 + \|k_{n_k} - \psi_{n_k}\|^2 \\ &+ \Big(1 - \frac{\lambda_{n_k}^2 a_1^2}{\lambda_{n_{k+1}}^2} \Big) \|u_{n_k} - \phi_{n_k}\|^2 + \Big(1 - \frac{\rho_{n_k}^2 a_2^2}{\rho_{n_{k+1}}^2} \Big) \|s_{n_k} - \psi_{n_k}\|^2 \\ &+ \beta_{n_k} (1 - \beta_{n_k}) \|\frac{1}{t_{n_{k,1}}} \int_0^{t_{n_{k,1}}} T_1(s) v_{n_k} ds - v_{n_k}\|^2 \\ &+ \gamma_{n_k} (1 - \gamma_{n_k}) \|\frac{1}{t_{n_{k,2}}} \int_0^{t_{n_{k,2}}} T_2(u) b_{n_k} du - b_{n_k}\|^2 \Big) = 0. \end{split}$$

From (5.2.7), (5.2.8) and the conditions on the control parameters, we have

$$\lim_{k \to \infty} \|z_{n_k} - \phi_{n_k}\| = 0, \quad \lim_{k \to \infty} \|k_{n_k} - \psi_{n_k}\| = 0, \quad \lim_{k \to \infty} \|u_{n_k} - \phi_{n_k}\| = 0 \quad \lim_{k \to \infty} \|s_{n_k} - \psi_{n_k}\| = 0 \quad (5.2.35)$$

$$\lim_{k \to \infty} \left\| \frac{1}{t_{n_{k,2}}} \int_0^{t_{n_{k,2}}} T_2(u) b_{n_k} du - b_{n_k} \right\| = 0, \quad \lim_{k \to \infty} \left\| \frac{1}{t_{n_{k,1}}} \int_0^{t_{n_{k,1}}} T_1(s) v_{n_k} ds - v_{n_k} \right\| = 0.$$
(5.2.36)

Also, we have

$$\lim_{k \to \infty} \left[\|\mathcal{F}_{2}^{*}(\mathcal{F}_{1}w_{n_{k}} - \mathcal{F}_{2}\varphi_{n_{k}})\|^{2} + \|\mathcal{F}_{1}^{*}(\mathcal{F}_{1}w_{n_{k}} - \mathcal{F}_{2}\varphi_{n_{k}})\|^{2} \right] = 0$$

which implies that

$$\lim_{k \to \infty} \left\| \mathcal{F}_2^* (\mathcal{F}_1 w_{n_k} - \mathcal{F}_2 \varphi_{n_k}) \right\| = 0, \lim_{k \to \infty} \left\| \mathcal{F}_1^* (\mathcal{F}_1 w_{n_k} - \mathcal{F}_2 \varphi_{n_k}) \right\| = 0, \lim_{k \to \infty} \left\| \mathcal{F}_1 w_{n_k} - \mathcal{F}_2 \varphi_{n_k} \right\| = 0.$$

From the definition of z_{n_k}, k_{n_k} and the previous inequality we have

$$\begin{aligned} \|z_{n_k} - w_{n_k}\| &= \eta_{n_k} \|\mathcal{F}_1^*(\mathcal{F}_1 w_{n_k} - \mathcal{F}_2 \varphi_{n_k})\| \to 0, \text{ as } k \to \infty. \\ \|k_{n_k} - \varphi_{n_k}\| &= \eta_{n_k} \|\mathcal{F}_2^*(\mathcal{F}_1 w_{n_k} - \mathcal{F}_2 \varphi_{n_k})\| \to 0, \text{ as } k \to \infty. \end{aligned}$$

Also, from the definition of v_{n_k}, b_{n_k} and (5.2.35), we have

$$\|v_{n_{k}} - u_{n_{k}}\| = \lambda_{n_{k}} \|Au_{n_{k}} - A\phi_{n_{k}}\| \le \frac{\lambda_{n_{k}}a_{1}}{\lambda_{n_{k+1}}} \|u_{n_{k}} - \phi_{n_{k}}\| \to 0, \text{ as } k \to \infty.$$
$$\|b_{n_{k}} - s_{n_{k}}\| = \rho_{n_{k}} \|Bs_{n_{k}} - B\psi_{n_{k}}\| \le \frac{\rho_{n_{k}}a_{2}}{\rho_{n_{k+1}}} \|s_{n_{k}} - \psi_{n_{k}}\| \to 0, \text{ as } k \to \infty.$$

From (5.2.36) and Lemma 2.1.12 we have

$$\|v_{n_{k}} - T_{1}(v)v_{n_{k}}\| \leq \left\|v_{n_{k}} - \frac{1}{t_{n_{k,1}}} \int_{0}^{t_{n_{k,1}}} T_{1}(s)v_{n_{k}}ds\right\| + \left\|\frac{1}{t_{n_{k,1}}} \int_{0}^{t_{n_{k,1}}} T_{1}(s)v_{n_{k}}ds - T_{1}(v)\frac{1}{t_{n_{k,1}}} \int_{0}^{t_{n_{k,1}}} T_{1}(s)v_{n_{k}}ds\right\| + \left\|T_{1}(v)\frac{1}{t_{n_{k,1}}} \int_{0}^{t_{n_{k,1}}} T_{1}(s)v_{n_{k}}ds - T_{1}(v)v_{n_{k}}\right\| \to 0, \text{ as } k \to \infty.$$
(5.2.37)

Similarly, we have

$$\lim_{k \to \infty} \|b_{n_k} - T_2(b)b_{n_k}\| = 0.$$
(5.2.38)

From the definition of $x_{n_{k+1}}$ and (5.2.35), we have

$$\|x_{n_{k+1}} - v_{n_k}\| = \left\| (1 - \beta_{n_k}) v_{n_k} + \beta_{n_k} \frac{1}{t_{n_{k,1}}} \int_0^{t_{n_{k,1}}} T_1(s) v_{n_k} - v_{n_k} \right\|$$
$$= \beta_{n_k} \left\| \frac{1}{t_{n_{k,1}}} \int_0^{t_{n_{k,1}}} T_1(s) v_{n_k} - v_{n_k} \right\| \to 0, \text{ as } k \to \infty.$$
(5.2.39)

Similarly, we have

$$\lim_{k \to \infty} \|y_{n_{k+1}} - b_{n_k}\| = 0.$$
(5.2.40)

Now, from Step 2 and by Remark 5.2.3, we get

$$\begin{aligned} \|w_{n_{k}} - x_{n_{k}}\| &= \|(1 - \alpha_{n_{k}})(x_{n_{k}} + \theta_{n_{k}}(x_{n_{k}} - x_{n_{k}-1})) - x_{n_{k}}\| \\ &= \|(1 - \alpha_{n_{k}})(x_{n_{k}} - x_{n_{k}}) + (1 - \alpha_{n_{k}})\theta_{n_{k}}(x_{n_{k}} - x_{n_{k}-1}) - \alpha_{n_{k}}x_{n_{k}}\| \\ &\leq (1 - \alpha_{n_{k}})\|x_{n_{k}} - x_{n_{k}}\| + (1 - \alpha_{n_{k}})\theta_{n_{k}}\|x_{n_{k}} - x_{n_{k}-1}\| + \alpha_{n_{k}}\|x_{n_{k}}\| \to 0, \\ &\qquad (5.2.41) \end{aligned}$$

as $k \to \infty$.

Similarly, we have

$$\lim_{k \to \infty} \|\varphi_{n_k} - y_{n_k}\| = 0.$$
 (5.2.42)

From (5.2.35)-(5.2.42) we have

$$\lim_{k \to \infty} \|x_{n_k} - \phi_{n_k}\| = 0, \lim_{k \to \infty} \|x_{n_{k+1}} - w_{n_k}\| = 0, \lim_{k \to \infty} \|y_{n_{k+1}} - \varphi_{n_k}\| = 0..$$
(5.2.43)

From (5.2.41) and (5.2.43) we have

$$\lim_{k \to \infty} \|x_{n_{k+1}} - x_{n_k}\| = 0.$$
(5.2.44)

Similarly, from (5.2.42) and (5.2.43)

$$\lim_{k \to \infty} \|y_{n_{k+1}} - y_{n_k}\| = 0.$$

To complete the proof, we show that $(w_{\omega}(x_n), w_{\omega}(y_n)) \subset \Gamma$. Since $\{x_n\}$ and $\{y_n\}$ are bounded we have that $w_{\omega}(x_n)$ and $w_{\omega}(y_n)$ are nonempty. Let $\hat{x} \in w_{\omega}(x_n)$ and $\hat{y} \in w_{\omega}(y_n)$ be arbitrary elements. Then there exists a subsequence $\{x_{n_k}\}$ of $\{x_n\}$ such that $x_{n_k} \rightharpoonup \hat{x}$ as $k \to \infty$. By (5.2.41), we have $w_{\omega}(x_n) = w_{\omega}(w_n)$. Since $\lim_{k \to \infty} ||x_{n_k} - \phi_{n_k}|| = 0$, we have that $\phi_{n_k} \rightharpoonup \hat{x} \in \mathcal{C}$ as $k \to \infty$. From the characteristic property of $P_{\mathcal{C}}$, we have

$$\langle x - u_{n_k}, \phi_{n_k} - \lambda_{n_k} A \phi_{n_k} - u_{n_k}, \rangle \le 0, \quad x \in \mathcal{C},$$

which implies that

$$\frac{1}{\lambda_{n_k}} \left\langle \phi_{n_k} - u_{n_k}, \ x - u_{n_k} \right\rangle \le \left\langle A \phi_{n_k}, \ x - u_{n_k} \right\rangle, \quad \forall \ x \in \mathcal{C}$$

Consequently, we have

$$\frac{1}{\lambda_{n_k}} \langle \phi_{n_k} - u_{n_k}, x - u_{n_k} \rangle + \langle A \phi_{n_k}, u_{n_k} - \phi_{n_k} \rangle \le \langle A \phi_{n_k}, x - \phi_{n_k} \rangle, \quad \forall x \in \mathcal{C}.$$
(5.2.45)

Applying the fact that $\lim_{k\to\infty} \|\phi_{n_k} - u_{n_k}\| = 0$ and $\lim_{k\to\infty} \lambda_{n_k} = \lambda > 0$ to (5.2.45), we have

$$0 \le \liminf_{k \to \infty} \langle A\phi_{n_k}, \ x - \phi_{n_k} \rangle, \ \forall \ x \in \mathcal{C}.$$
(5.2.46)

Also, we have that

$$\langle Au_{n_k}, x - u_{n_k} \rangle = \langle Au_{n_k} - A\phi_{n_k}, x - \phi_{n_k} \rangle + \langle A\phi_{n_k}, x - \phi_{n_k} \rangle + \langle Au_{n_k}, \phi_{n_k} - u_{n_k} \rangle.$$
(5.2.47)

Since A is uniformly continuous on \mathcal{H} and $\lim_{k\to\infty} \|\phi_{n_k} - u_{n_k}\|$, we have

$$\lim_{k \to \infty} \|A\phi_{n_k} - Au_{n_k}\| = 0.$$
 (5.2.48)

From (5.2.46)-(5.2.48), we have

$$0 \le \liminf_{k \to \infty} \langle Au_{n_k}, \ x - u_{n_k} \rangle \ \forall \ x \in \mathcal{C}.$$
(5.2.49)

Let $\{\delta_k\}$ be a sequence of positive numbers such that $\delta_{k+1} \leq \delta_k$, $\forall k \geq 1$ and $\delta_k \rightarrow 0$ as $k \rightarrow \infty$. Then, for each $k \geq 1$, we denote by N_k the smallest positive integer such that

$$\langle Au_{n_i}, x - u_{n_i} \rangle + \delta_k \ge 0 \quad \forall j \ge N_k, \tag{5.2.50}$$

where the existence of N_k follows from (5.2.49). We have that $\{N_k\}$ is increasing since $\{\delta_k\}$ is decreasing. Furthermore, since $\{u_{n_k}\} \subset \mathcal{C}$ we can suppose $Au_{N_k} \neq 0$ (otherwise, u_{N_k} is a solution) and we set for each $k \geq 1$, $h_{N_k} = \frac{Au_{N_k}}{\|Au_{N_k}\|^2}$. Then we have that $\langle Au_{N_k}, h_{N_k} \rangle = 1$ for each $k \geq 1$. Thus, by (5.2.50), we have that

$$\langle Au_{N_k}, x + \delta_k h_{N_k} - u_{N_k} \rangle \ge 0,$$

which implies by the pseudo-monotonicity of A that

$$\langle A(x+\delta_k h_{N_k}), x+\delta_k h_{N_k} - u_{N_k} \rangle \ge 0.$$
(5.2.51)

Since $u_{n_k} \subset C$, then $\{u_{n_k}\}$ converges weakly to $\hat{x} \in C$. If $A\hat{x} = 0$, then $\hat{x} \in VI(\mathcal{C}, A)$. On the contrary, we suppose $A\hat{x} \neq 0$. Since A satisfies condition (c), we have

$$0 < \|A\hat{x}\| \le \liminf_{k \to \infty} \|Au_{n_k}\|$$

Since $\{u_{N_k}\} \subset \{u_{n_k}\}$, we obtain that

$$0 \leq \limsup_{k \to \infty} \|\delta_k h_{N_k}\| = \limsup_{k \to \infty} \left(\frac{\delta_k}{\|Au_{n_k}\|}\right) \leq \frac{\limsup_{k \to \infty} \delta_k}{\liminf_{k \to \infty} \|Au_{n_k}\|} = 0.$$

Therefore, $\lim_{k\to\infty} \|\delta_k h_{N_k}\| = 0$. Letting $k\to\infty$ in (5.2.51) gives

$$\langle Ax, x - \hat{x} \rangle \ge 0, \quad \forall x \in \mathcal{C},$$
 (5.2.52)

which implies by Lemma 2.5.2 that $\hat{x} \in VI(\mathcal{C}, A)$. By similar argument, we have that $\hat{y} \in VI(\mathcal{Q}, B)$.

Now, to show that $\hat{x} \in F(\mathcal{T}_a)$ and $\hat{y} \in F(\mathcal{T}_b)$. On the contrary, we suppose that $T_1(v)\hat{x} \neq \hat{x}$ and $T_2(b)\hat{x} \neq \hat{y}$, for all $v \ge 0$ and $b \ge 0$. Then, it follows from the Opial condition of Hilbert space and from (5.2.37) that

$$\begin{split} \liminf_{k \to \infty} \|v_{n_k} - \hat{x}\| &< \liminf_{k \to \infty} \|v_{n_k} - T_1(v) \hat{x}\| \\ &\leq \liminf_{k \to \infty} \left\{ \|v_{n_k} - T_1(v) v_{n_k}\| + \|T_1(v) v_{n_k} - T_1(v) \hat{x}\| \right\} \\ &\leq \liminf_{k \to \infty} \left\{ \|v_{n_k} - T_1(v) v_{n_k}\| + \|v_{n_k} - \hat{x}\| \right\} \\ &= \liminf_{k \to \infty} \|v_{n_k} - \hat{x}\|, \end{split}$$

which is a contradiction. Thus, it follows that $T_1(v)\hat{x} = \hat{x}$ for all $v \ge 0$ which implies that $\hat{x} \in F(\mathcal{T}_a)$. Similarly, $\hat{y} \in F(\mathcal{T}_b)$.

Next, from (5.2.35) we have that $\lim_{k\to\infty} \|\phi_{n_k} - z_{n_k}\| = \lim_{k\to\infty} \|U_{r_{n_k,1}}^{\Phi_1} z_{n_k} - z_{n_k}\| = 0$, and since $z_{n_k} \rightarrow \hat{x}$ it follows from the demiclosed property of nonexpansive mappings that $\hat{x} \in EP(\Phi_1)$. Similarly, we have that $\hat{y} \in EP(\Phi_2)$. Since $\mathcal{F}_1 \hat{x} - \mathcal{F}_2 \hat{y} \in w_{\omega}(\mathcal{F}_1 w_n - \mathcal{F}_2 \varphi_n)$, it follows from the weakly lower semi-continuity of the norm that

$$\|\mathcal{F}_1\hat{x} - \mathcal{F}_2\hat{y}\| \le \liminf_{n \to \infty} \|\mathcal{F}_1w_n - \mathcal{F}_2\varphi_n\| = 0.$$

Hence, we have that $(\hat{x}, \hat{y}) \in \Gamma$. Since $\hat{x} \in w_{\omega}(x_n)$ and $\hat{y} \in w_{\omega}(y_n)$ are arbitrary elements, it follows that $(w_{\omega}(x_n), w_{\omega}(y_n)) \subset \Gamma$.

To conclude, we show that

$$\limsup_{k \to \infty} \left(\left\langle \hat{x}, \ \hat{x} - x_{n_{k+1}} \right\rangle + \left\langle \hat{y}, \ \hat{y} - y_{n_{k+1}} \right\rangle \right) \le 0.$$

By the boundedness of $\{x_{n_k}\}$, it follows that there exists a subsequence $\{x_{n_{k_j}}\}$ of $\{x_{n_k}\}$ which converges weakly to some $\bar{x} \in \mathcal{H}$, and such that

$$\lim_{j \to \infty} \left\langle \hat{x}, \ \hat{x} - x_{n_{k_j}} \right\rangle = \limsup_{k \to \infty} \left\langle \hat{x}, \ \hat{x} - x_{n_k} \right\rangle.$$
(5.2.53)

From (5.2.53) and the fact that $\hat{x} = P_D(0)$ we have

$$\lim_{k \to \infty} \sup_{x \to \infty} \langle \hat{x}, \ \hat{x} - x_{n_k} \rangle = \lim_{j \to \infty} \left\langle \hat{x}, \ \hat{x} - x_{n_{k_j}} \right\rangle = \langle \hat{x}, \ \hat{x} - \bar{x} \rangle \le 0.$$
(5.2.54)

From (5.2.44) and (5.2.54), it follows that

$$\limsup_{k \to \infty} \left\langle \hat{x}, \ \hat{x} - x_{n_{k+1}} \right\rangle = \limsup_{k \to \infty} \left\langle \hat{x}, \ \hat{x} - x_{n_k} \right\rangle = \left\langle \hat{x}, \ \hat{x} - \bar{x} \right\rangle \le 0.$$
(5.2.55)

Following the same line of argument and the fact that $\hat{y} = P_E(0)$ we have

$$\limsup_{k \to \infty} \left\langle \hat{y}, \ \hat{y} - y_{n_{k+1}} \right\rangle = \limsup_{k \to \infty} \left\langle \hat{y}, \ \hat{y} - y_{n_k} \right\rangle = \left\langle \hat{y}, \ \hat{y} - \bar{y} \right\rangle \le 0.$$
(5.2.56)

Adding (5.2.55) and (5.2.56), we obtain

$$\limsup_{k \to \infty} \left(\left\langle \hat{x}, \ \hat{x} - x_{n_{k+1}} \right\rangle + \left\langle \hat{y}, \ \hat{y} - y_{n_{k+1}} \right\rangle \right) \le 0.$$
 (5.2.57)

Thus, by (5.2.43) and (5.2.57) we have $\limsup_{k\to\infty} \hat{d}_{n_k} \leq 0$. Now, applying Lemma 2.5.36 to (5.2.33) we have $\{\|x_n - \hat{x}\| + \|y_n - \hat{y}\|\}$ converges to zero, which implies that $\lim_{n\to\infty} \|x_n - \hat{x}\| = 0$ and $\lim_{n\to\infty} \|y_n - \hat{y}\| = 0$. Therefore, $(\{x_n\}, \{y_n\})$ converges strongly to (\hat{x}, \hat{y}) .

5.2.3 Numerical experiments

In this section, we discuss the numerical behavior of our method, (Proposed Alg.) Algorithm 5.2.3 in comparison with the method in Appendix 5.2.2 proposed by Latif *et al.* (Latif *et al.* Alg.), which is the only related result we could find in the literature. We plot the graph of errors against the number of iterations in each case of both examples using $|x_{n+1} - x_n| < 10^{-4}$ and $||x_{n+1} - x_n|| < 10^{-4}$ in Example 5.2.12 and Example 5.2.13 respectively as the stopping criterion. The numerical computations are reported in Figures 5.1 - 5.8 and Tables 5.2.1-5.2.2 with all implementations performed using Matlab 2021 (b).

In our computation, we choose $\theta = 3.5$, $\tau = 2.44$, $\lambda_1 = 1.5$, $\rho_1 = 1.8$, $a_1 = 0.8$, $a_2 = 0.9$, $\epsilon_n = \zeta_n = \frac{1}{(2n+1)^3}$, $\alpha_n = \frac{3}{2n+1}$, $\beta_n = \frac{1}{4}$, $\gamma_n = \frac{1}{4}$, $\rho_n = \sigma_n = \frac{100}{(n+1)^2}$, $\eta = 0.5$, $r_{n,1} = 2.8$, $r_{n,2} = 3.5$, $t_{n,1} = 4.5$, $t_{n,2} = 5.5$, s = u = 1.5. For Appendix 5.2.14, we choose $\alpha = 0.85$, $\zeta_n = \kappa_n = \frac{1}{6}$, $\xi_n = \delta_n = \frac{1-\alpha_n}{2}$.

Example 5.2.12. Let $\mathcal{H}_1 = \mathcal{H}_2 = \mathcal{H}_3 = \mathbb{R}$ the set of all real numbers with the inner product $\langle x, y \rangle = xy$, $\forall x, y \in \mathbb{R}$ and induced norm $|\cdot|$. For $r_i > 0$, i = 1, 2, consider $\mathcal{C} = [-10, 10]$ and $\mathcal{Q} = [0, 20]$. We define the bifunction $\Phi_1 : \mathcal{C} \times \mathcal{C} \to \mathbb{R}$ and $\Phi_2 : \mathcal{Q} \times \mathcal{Q} \to \mathbb{R}$ as follows:

$$U_{r_1}^{\Phi_1}(u) = \frac{u}{3r_1 + 1}, \quad \forall \ x \in \mathcal{C}$$

and

$$U_{r_2}^{\Phi_2}(v) = \frac{v}{r_2 + 1}, \quad \forall \ y \in \mathcal{Q}.$$

Let $\mathcal{F}_1 x = 2x$ and $\mathcal{F}_2 x = 5x$ which implies that $\mathcal{F}_1^* x = 2x$ and $\mathcal{F}_2^* x = 5x$. Next we define $A : \mathcal{H}_1 \to \mathcal{H}_1$ as Ax = 2x and $B : \mathcal{H}_2 \to \mathcal{H}_2$ as Bx = 3x. We define the mappings $T_1(s) : \mathbb{R} \to \mathbb{R}$ and $T_2(u) : \mathbb{R} \to \mathbb{R}$ as follows; $T_1(s)x = 10^{-s}x$ and $T_2(u)y = 10^{-2u}y$. Clearly, we observe that $T_1(s)$ and $T_2(u)$ are nonexpansive semigroups.

We choose $\mathcal{V}_1 = x_0$, $\mathcal{V}_2 = y_0$ and consider the following cases for the numerical experiments of this example.

Case 1: Take $(x_0, y_0) = (-13.5, 8.0)$ and $(x_1, y_1) = (5.7, -9.1)$.

Case 2: Take $(x_0, y_0) = (15.1, 7.9)$ and $(x_1, y_1) = (6.4, 81.3)$.

Case 3: Take $(x_0, y_0) = (10.9, -11.8)$ and $(x_1, y_1) = (-37.2, 26.8)$.

Case 4: Take $(x_0, y_0) = (-14.9, -9.8)$ and $(x_1, y_1) = (-25.2, -17.7)$.

Table 5.2.1: Numerical Results for Example 5.2.12								
	Case 1		Case 2		Case 3		Case 4	
	Iter.	CPU Time						
Latif <i>et al.</i> Alg.	86	0.0085	86	0.0062	89	0.0084	88	0.0064
Proposed Alg. 5.2.3	64	0.0132	64	0.0078	64	0.0093	64	0.0018



Figure 5.1: Example 5.2.12: Case 1

Figure 5.2: Example 5.2.12: Case 2

Latif & Eslamian Alg. (x_n)

Latif & Eslamian Alg. (y_n)

10²

Proposed Alg. (x_n)

Proposed Alg. (y_n)



Figure 5.3: Example 5.2.12: Case 3

Figure 5.4: Example 5.2.12: Case 4

Example 5.2.13. Let $\mathcal{H}_1 = \mathcal{H}_2 = \mathcal{H}_3 = (l_2(\mathbb{R}), \|\cdot\|_2)$, where

$$l_2(\mathbb{R}) := \{ x = (x_1, x_2, \dots, x_n, \dots), x_i \in \mathbb{R} : \sum_{i=1}^{\infty} |x_i|^2 < +\infty \},\$$

 $||x||_2 = \sqrt{(\sum_{i=1}^{\infty} |x_i|^2)}$ and $\langle x, y \rangle = \sum_{i=1}^{\infty} x_i y_i$ for all $x \in \ell_2(\mathbb{R})$. For $r_i > 0$, i = 1, 2, we define the sets $\mathcal{C} := \{x \in \ell_2 : ||x|| \leq 1\}$ and $\mathcal{Q} := \{y \in \ell_2 : ||y|| \leq 1\}$. Let $\mathcal{F}_1 : \mathcal{H}_1 \to \mathcal{H}_2$, $\mathcal{F}_2 : \mathcal{H}_2 \to \mathcal{H}_3$ be defined by $\mathcal{F}_1 x = \frac{x}{3}$ and $\mathcal{F}_2 x = \frac{2x}{5}$ respectively which implies that $\mathcal{F}_1^* y = \frac{y}{3}$ and $\mathcal{F}_2^* y = \frac{2y}{5}$. Clearly, \mathcal{F}_1 and \mathcal{F}_2 are bounded linear operators. We define $\Phi_1 : \mathcal{C} \times \mathcal{C} \to \mathbb{R}$ and $\Phi_2 : \mathcal{Q} \times \mathcal{Q} \to \mathbb{R}$ by $\Phi_1(x, y) = \langle L_1 x, y - x \rangle$ and $\Phi_2(x, y) = \langle L_2 x, y - x \rangle$, where $L_1 x = \frac{x}{3}$ and $L_2 x = \frac{x}{2}$. Observe that Φ_1 and Φ_2 satisfy Assumption 5.2.2. After simple calculation and applying Lemma 5.2.1, we obtain

$$U_{r_1}^{\Phi_1}(u) = \frac{3u}{r_1 + 3}, \quad \forall \ x \in \mathcal{C},$$

and

$$U_{r_2}^{\Phi_2}(v) = \frac{2v}{r_2 + 2}, \quad \forall \ y \in \mathcal{Q}.$$

Let $A: \mathcal{H}_1 \to \mathcal{H}_1$ be defined by $A(x_1, x_2, x_3, \cdots) = (x_1 e^{-x_1^2}, 0, 0, \cdots)$ and $B: \mathcal{H}_2 \to \mathcal{H}_2$ as $B(x_1, x_2, x_3, \cdots) = (5x_1 e^{-x_1^2}, 0, 0, \cdots)$. Clearly, we see that A and B are pseudomonotone mappings. We define the mappings $T_1(s) : \mathbb{R} \to \mathbb{R}$ and $T_2(u) : \mathbb{R} \to \mathbb{R}$ as follows; $T_1(s)x = 10^{-5s}x$ and $T_2(u)y = 10^{-3u}y$. Clearly, we observe that $T_1(s)$ and $T_2(u)$ are nonexpansive semigroups.

We choose $\mathcal{V}_1 = x_0, \mathcal{V}_2 = y_0$ and consider different initial values as follows: **Case 1:** $x_0 = (\frac{1}{2}, \frac{1}{4}, \frac{1}{8}, \cdots), y_0 = (\frac{1}{2}, \frac{1}{4}, \frac{1}{8}, \cdots);$

$$x_1 = \left(\frac{1}{3}, \frac{1}{9}, \frac{1}{27}, \cdots\right), y_1 = \left(\frac{1}{3}, \frac{1}{9}, \frac{1}{27}, \cdots\right);$$

Case 2: $x_0 = \left(\frac{1}{2}, \frac{1}{6}, \frac{1}{18}, \cdots\right), y_0 = \left(\frac{1}{2}, \frac{1}{4}, \frac{1}{8}, \cdots\right);$
 $x_1 = \left(-\frac{1}{3}, \frac{1}{6}, -\frac{1}{18}, \cdots\right), y_1 = \left(-\frac{1}{3}, \frac{1}{6}, -\frac{1}{18}, \cdots\right);$

Case 3:
$$x_0 = (\frac{3}{8}, \frac{3}{16}, \frac{3}{32}, \cdots), y_0 = (\frac{5}{9}, \frac{5}{18}, -\frac{5}{36}, \cdots);$$

 $x_1 = (-\frac{1}{3}, \frac{1}{9}, -\frac{1}{27}, \cdots), y_1 = (\frac{1}{2}, \frac{1}{6}, \frac{1}{12}, \cdots);$

Case 4: $x_0 = (\frac{3}{8}, \frac{3}{16}, \frac{3}{32}, \cdots), y_0 = (\frac{5}{9}, \frac{5}{18}, \frac{5}{36}, \cdots).$ $x_1 = (\frac{1}{9}, \frac{1}{18}, \frac{1}{36}, \cdots), y_1 = (-\frac{7}{12}, \frac{7}{24}, -\frac{7}{36}).$

	Case 1		Case 2		Case 3		Case 4	
	Iter.	CPU Time						
Latif <i>et al.</i> Alg.	72	0.0134	72	0.0192	72	0.0147	72	0.0073
Proposed Alg. 5.2.3	58	0.0211	58	0.0171	58	0.0263	58	0.0100





Figure 5.5: Example 5.2.13: Case 1

Figure 5.6: Example 5.2.13: Case 2



Figure 5.7: Example 5.2.13: Case 3

Figure 5.8: Example 5.2.13: Case 4

Appendix 5.2.14. Algorithm 1 of Latif et al. [159]

Choose sequences $\{\beta_n\}_{n=1}^{\infty}, \{\alpha_n\}_{n=1}^{\infty}, \{\delta_n\}_{n=1}^{\infty}$ such that $\beta_n + \alpha_n + \delta_n = 1$. Select initial point $x_0 \in \mathcal{H}_1, y_0 \in \mathcal{H}_2$, let $\vartheta \ge 0$. Set n := 1.

$$\begin{cases} z_{n} = x_{n} - \vartheta_{n} \mathcal{F}_{1}^{*} (\mathcal{F}_{1} x_{n} - \mathcal{F}_{2} y_{n}), \\ \phi_{n} = U_{r_{n,1}}^{\Phi_{1}} z_{n}, \\ u_{n} = P_{\mathcal{C}} (\phi_{n} - \varsigma_{n} A \phi_{n}), \\ p_{n} = P_{\mathcal{C}} (\phi_{n} - \varsigma_{n} A u_{n}), \\ x_{n+1} = \alpha_{n} \mathcal{V}_{1} + \xi_{n} p_{n} + \delta_{n} T_{1}(s) p_{n} \\ k_{n} = y_{n} + \vartheta_{n} \mathcal{F}_{2}^{*} (\mathcal{F}_{1} x_{n} - \mathcal{F}_{2} y_{n}), \\ \psi_{n} = U_{r_{n,2}}^{\Phi_{2}} k_{n}, \\ s_{n} = P_{\mathcal{Q}} (\psi_{n} - \kappa_{n} B \psi_{n}), \\ l_{n} = P_{\mathcal{Q}} (\psi_{n} - \kappa_{n} B s_{n}), \\ y_{n+1} = \alpha_{n} \mathcal{V}_{2} + \xi_{n} l_{n} + \delta_{n} T_{2}(u) l_{n} \end{cases}$$
(5.2.58)

where the step size ϑ_n is chosen such that for small enough $\epsilon > 0$,

$$\vartheta_n \in \left[\epsilon, \quad \frac{2\|\mathcal{F}_1 x_n - \mathcal{F}_2 y_n\|^2}{\|\mathcal{F}_2^*(\mathcal{F}_1 x_n - \mathcal{F}_2 y_n)\|^2 + \|\mathcal{F}_1^*(\mathcal{F}_1 x_n - \mathcal{F}_2 y_n)\|^2} - \epsilon\right],$$

if $\mathcal{F}_1 x_n \neq \mathcal{F}_2 y_n$; otherwise, $\vartheta_n = \eta$. Set n := n + 1 and go back to Step 1.

5.3 On minimum-norm solutions of inclusion and split equilibrium problems

In this work, we study the problem of finding the common solutions of V_q IP and the SEP. We propose a modified inertial forward-backward splitting algorithm with self-adaptive step size for approximating the solution of the problem in Hilbert spaces. Unlike several of the existing results in this direction, our proposed method only requires that the underlying single-valued operator be monotone and Lipschitz continuous. This relaxes the strong assumption of co-coerciveness (inverse strongly monotonicity) imposed in several of the existing results. Moreover, our method combines the inertial technique with the forward-backward splitting method and self-adaptive step sizes for solving the common solution problem. Unlike several of the existing methods, the self-adaptive step sizes ensure that knowledge of the norm of the bounded linear operator nor of the Lipschitz constant are not required to implement our algorithm. We prove that the sequence generated by our proposed method converges strongly to a minimum-norm solution of the aforementioned problem. Furthermore, we apply our result to study certain optimization problems. Finally, we provide some numerical experiments of our proposed method in comparison with other existing methods in the literature.

5.3.1 Proposed method

In this section, we present our proposed method and discuss its features. We begin by defining the following functions;

$$g(x) = \frac{1}{2} \| (I - T_r^{F_2}) A x \|^2, \quad h(x) = \frac{1}{2} \| (I - T_r^{F_1}) x \|^2$$

and

$$G(x) = A^*(I - T_r^{F_2})Ax, \quad H(x) = (I - T_r^{F_1})x.$$

It can be easily verified from Aubin [30] that g and h are weakly lower semi-continuous, convex and differentiable functions. We now give the following assumptions under which our strong convergence result is obtained.

Assumption 5.3.1. Suppose that the following conditions hold:

- (a) The feasible sets C and Q are nonempty, closed and convex subsets of the real Hilbert spaces \mathcal{H}_1 and \mathcal{H}_2 , respectively.
- (b) $F_1 : \mathcal{C} \times \mathcal{C} \to \mathbb{R}, F_2 : \mathcal{Q} \times \mathcal{Q} \to \mathbb{R}$ are bifunctions satisfying Assumption (5.2.1) and F_2 is upper semi continuous in the first argument.
- (c) $A: \mathcal{H}_1 \to \mathcal{H}_2$ is a bounded linear operator, $B: \mathcal{H}_1 \to \mathcal{H}_1$ is a Lipschitz continuous and monotone operator, and $D: \mathcal{H}_1 \to 2^{\mathcal{H}_1}$ is a maximal monotone operator such that $\Gamma = (B+D)^{-1}(0) \cap \Omega \neq \emptyset$, where $\Omega = \{z \in \mathcal{C} : z \in EP(F_1) \text{ and } Az \in EP(F_2)\}.$

- (d) $\{\alpha_n\}, \{\beta_n\}$ and $\{\gamma_n\}$ are sequences in (0,1) satisfying $\beta_n + \gamma_n \leq 1, \lim_{n \to \infty} \gamma_n = 0, \sum_{n=1}^{\infty} \gamma_n = \infty, \inf(1 \beta_n \gamma_n)\beta_n > 0, 0 < a \leq \alpha_n, \beta_n \leq b < 1, 0 < c \leq \tau_n \leq d < 4.$
- (e) Let $\{\epsilon_n\}$ be a positive sequence such that $\lim_{n\to\infty}\frac{\epsilon_n}{\gamma_n} = 0$ and $\{r_n\} \subset (0,\infty)$ such that $\liminf_{n\to\infty} r_n > 0$.

Algorithm 5.3.2.

Step 1: Select initial point $x_0, x_1 \in \mathcal{H}_1$, let $s_1 > 0, \mu \in (0, 1), \theta \ge 3$ and set n = 1. Given the iterates x_{n-1} and x_n for each $n \ge 1$, choose θ_n such that $0 \le \theta_n \le \overline{\theta}_n$, where

$$\bar{\theta}_n := \begin{cases} \min\left\{\frac{n-1}{n+\theta-1}, \frac{\epsilon_n}{\|x_n - x_{n-1}\|}\right\}, & \text{if } x_n \neq x_{n-1} \\ \frac{n-1}{n+\theta-1}, & \text{otherwise.} \end{cases}$$
(5.3.1)

Step 2: Compute

$$w_n = x_n + \theta_n (x_n - x_{n-1}).$$

Step 3: Compute

$$z_n = T_{r_n}^{F_1} (I - \lambda_n A^* (I - T_{r_n}^{F_2}) A) w_n$$

and

$$y_n = \alpha_n w_n + (1 - \alpha_n) z_n.$$

Step 4: Compute

$$u_n = (I + s_n D)^{-1} (I - s_n B) y_n = J_{s_n}^D (I - s_n B) y_n$$

and

$$v_n = u_n - s_n (Bu_n - By_n).$$

Step 5 Compute

$$x_{n+1} = (1 - \beta_n - \gamma_n)w_n + \beta_n v_n,$$

where

$$\lambda_n := \begin{cases} \tau_n \frac{g(w_n)}{\|G(w_n)\|^2 + \|H(w_n)\|^2}, & \text{if } \|G(w_n)\|^2 + \|H(w_n)\|^2 \neq 0, \\ 0, & \text{otherwise} \end{cases}$$
(5.3.2)

and

$$s_{n+1} = \begin{cases} \min\left\{\frac{\mu \|y_n - u_n\|}{\|By_n - Bu_n\|}, s_n\right\} & \text{if } By_n - Bu_n \neq 0, \\ s_n, & \text{otherwise.} \end{cases}$$
(5.3.3)

Set n := n + 1 and go back to Step 1.

Remark 5.3.1. • Different from all existing methods for finding the common solutions of the VIP (1.2.9) and SEP (1.2.10)-(1.2.11), our proposed method requires the underlying operator B to be monotone and Lipschitz continuous. This relaxes the strong assumption of co-coerciveness (inverse strongly monotonicity) on the operator.

- Contrast to the methods in the literature, the implementation of our method does not depend on the norm of the bounded linear operator nor the Lipschitz constant of the monotone operator. This makes our method easier to implement than several other methods in the literature, which depends on the norm of the bounded linear operator.
- Our method employs the inertial technique to improve the speed of convergence of the proposed algorithm. Unlike several other inertial methods in the literature, our method does not require the summability condition $\left(\sum_{n=1}^{\infty} \alpha_n ||x_n x_{n-1}|| < \infty\right)$, which makes our method easily implementable.
- The sequence generated by our proposed method was proved to converge strongly to a minimum-norm solution of the investigated problem under mild conditions. It is known that minimum-norm solutions of problems find applications in several practical problems. Moreover, our convergence analysis is not dependent on the usual "Two cases approach", which is widely used in many papers to guarantee strong convergence.

Remark 5.3.2. By conditions (d) and (e), one can easily verify from (5.3.1) that

$$\lim_{n \to \infty} \theta_n ||x_n - x_{n-1}|| = 0 \text{ and } \lim_{n \to \infty} \frac{\theta_n}{\gamma_n} ||x_n - x_{n-1}|| = 0.$$
 (5.3.4)

Remark 5.3.3. From (5.3.3) in Algorithm 5.3.2, we have that $s_{n+1} \leq s_n$ for all $n \geq 1$. Also, from our conditions we have that B is L-Lipschitz continuous, therefore when $By_n \neq Bu_n$ in Algorithm 5.3.2, we have

$$s_{n+1} = \min\left\{\frac{\mu||y_n - u_n||}{||By_n - Bu_n||}, \ s_n\right\} \ge \min\left\{\frac{\mu}{L}, \ s_n\right\}.$$

By induction, we obtain that $\{s_n\}$ is bounded below by min $\{\frac{\mu}{L}, s_1\}$. Also, since $\{s_n\}$ is monotone nonincreasing, we have that the limit exists, and $\lim_{n\to\infty} s_n \ge \min\{\frac{\mu}{L}, s_1\} > 0$.

Remark 5.3.4. From the definition of $\{s_n\}$ in (5.3.3) we have

$$||By_n - Bu_n|| \le \frac{\mu}{s_{n+1}} ||y_n - u_n||, \quad \forall \ n.$$
(5.3.5)

Inequality (5.3.5) holds if $By_n = Bu_n$. In the case where $By_n \neq Bu_n$, we have

$$s_{n+1} = \min\left\{\frac{\mu \|y_n - u_n\|}{\|By_n - Bu_n\|}, s_n\right\} \le \frac{\mu \|y_n - u_n\|}{\|By_n - Bu_n\|},$$

which implies that

$$||By_n - Bu_n|| \le \frac{\mu}{s_{n+1}} ||y_n - u_n||.$$

Hence, (5.3.5) holds for both cases when $By_n \neq Bu_n$ and $By_n = Bu_n$.

5.3.2 Convergence analysis

Lemma 5.3.5. Let $\{x_n\}$ be a sequence generated by Algorithm 5.3.2 under Assumption 5.3.1. Then,

$$||v_n - p||^2 \le ||w_n - p||^2 - \tau_n (4 - \tau_n) (1 - \alpha_n) \frac{g^2(w_n)}{||G(w_n)||^2 + ||H(w_n)||^2} - \alpha_n (1 - \alpha_n) ||w_n - z_n||^2 - \left(1 - s_n^2 \cdot \frac{\mu^2}{s_{n+1}^2}\right) ||y_n - u_n||^2.$$

Proof. Let $p \in \Gamma$. Since $G(w_n) = A^*(I - T_{r_n}^{F_2})Aw_n$, $p = T_{r_n}^{F_2}p$ and $I - T_{r_n}^{F_2}$ is firmly nonexpansive, we have

$$\langle G(w_n), w_n - p \rangle = \langle A^*(I - T_{r_n}^{F_2})Aw_n, w_n - p \rangle = \langle (I - T_{r_n}^{F_2})Aw_n, Aw_n - Ap \rangle = \langle (I - T_{r_n}^{F_2})Aw_n - (I - T_{r_n}^{F_2})Ap, Aw_n - Ap \rangle \ge \| (I - T_{r_n}^{F_2})Aw_n \|^2 = 2g(w_n).$$
 (5.3.6)

From the definition of z_n in Step 3, the nonexpansivity of $T_{r_n}^{F_1}$ and (5.3.6) we have

$$\begin{aligned} \|z_n - p\|^2 &= \|T_{r_n}^{F_1}(I - \lambda_n A^* (I - T_{r_n}^{F_2}) A) w_n - p\|^2 \\ &\leq \|(I - \lambda_n A^* (I - T_{r_n}^{F_2}) A) w_n - p\|^2 \\ &= \|w_n - p - \lambda_n G(w_n)\|^2 \\ &= \|w_n - p\|^2 + \lambda_n^2 \|G(w_n)\|^2 - 2\lambda_n \langle G(w_n), w_n - p \rangle \\ &\leq \|w_n - p\|^2 + \lambda_n^2 \|G(w_n)\|^2 - 4\lambda_n g(w_n) \\ &\leq \|w_n - p\|^2 - \tau_n (4 - \tau_n) \frac{g^2(w_n)}{\|G(w_n)\|^2 + \|H(w_n)\|^2}. \end{aligned}$$
(5.3.7)

From the condition on τ_n we obtain

$$||z_n - p|| \le ||w_n - p|| \tag{5.3.8}$$

From the definition of y_n and (5.3.7) we have,

$$\begin{aligned} \|y_n - p\|^2 &= \|\alpha_n w_n + (1 - \alpha_n) z_n - p\|^2 \\ &= \alpha_n \|w_n - p\|^2 + (1 - \alpha_n) \|z_n - p\|^2 - \alpha_n (1 - \alpha_n) \|w_n - z_n\|^2 \\ &\leq \alpha_n \|w_n - p\|^2 + (1 - \alpha_n) \left[\|w_n - p\|^2 - \tau_n (4 - \tau_n) \frac{g^2(w_n)}{\|G(w_n)\|^2 + \|H(w_n)\|^2} \right] \\ &- \alpha_n (1 - \alpha_n) \|w_n - z_n\|^2 \\ &= \|w_n - p\|^2 - \tau_n (4 - \tau_n) (1 - \alpha_n) \frac{g^2(w_n)}{\|G(w_n)\|^2 + \|H(w_n)\|^2} \\ &- \alpha_n (1 - \alpha_n) \|w_n - z_n\|^2. \end{aligned}$$
(5.3.9)

From the condition on τ_n and α_n , we have

$$||y_n - p|| \le ||w_n - p||. \tag{5.3.10}$$

Applying (5.3.5) and (5.3.9), we have

$$\begin{split} \|v_{n} - p\|^{2} &= \|u_{n} - s_{n}(Bu_{n} - By_{n}) - p\|^{2} \\ &= \|u_{n} - p\|^{2} + s_{n}^{2}\|Bu_{n} - By_{n}\|^{2} - 2s_{n}\langle u_{n} - p, Bu_{n} - By_{n}\rangle \\ &= \|y_{n} - p\|^{2} + \|u_{n} - y_{n}\|^{2} + 2\langle u_{n} - y_{n}, y_{n} - p\rangle + s_{n}^{2}\|Bu_{n} - By_{n}\|^{2} \\ &- 2s_{n}\langle u_{n} - p, Bu_{n} - By_{n}\rangle \\ &= \|y_{n} - p\|^{2} + \|u_{n} - y_{n}\|^{2} + 2\langle u_{n} - y_{n}, u_{n} - p\rangle - 2\langle u_{n} - y_{n} - u_{n} - y_{n}\rangle \\ &+ s_{n}^{2}\|Bu_{n} - By_{n}\|^{2} - 2s_{n}\langle u_{n} - p, Bu_{n} - By_{n}\rangle \\ &= \|y_{n} - p\|^{2} + \|u_{n} - y_{n}\|^{2} + 2\langle u_{n} - y_{n}, u_{n} - p\rangle - 2\|u_{n} - y_{n}\|^{2} \\ &+ s_{n}^{2}\|Bu_{n} - By_{n}\|^{2} - 2s_{n}\langle u_{n} - p, Bu_{n} - By_{n}\rangle \\ &= \|y_{n} - p\|^{2} - \|u_{n} - y_{n}\|^{2} + 2\langle u_{n} - y_{n}, u_{n} - p\rangle - 2\|u_{n} - y_{n}\|^{2} \\ &+ s_{n}^{2}\|Bu_{n} - By_{n}\|^{2} - 2s_{n}\langle u_{n} - p, Bu_{n} - By_{n}\rangle \\ &= \|y_{n} - p\|^{2} - \|u_{n} - y_{n}\|^{2} + 2\langle u_{n} - y_{n}, u_{n} - p\rangle + s_{n}^{2}\|Bu_{n} - By_{n}\|^{2} \\ &- 2s_{n}\langle u_{n} - p, Bu_{n} - By_{n}\rangle \\ &= \|y_{n} - p\|^{2} - \|u_{n} - y_{n}\|^{2} + s_{n}^{2}\|Bu_{n} - By_{n}\|^{2} \\ &- 2\langle u_{n} - p, y_{n} - u_{n} - s_{n}(By_{n} - Bu_{n})\rangle \\ &\leq \|w_{n} - p\|^{2} - \tau_{n}(4 - \tau_{n})(1 - \alpha_{n})\frac{g^{2}(w_{n})}{\|G(w_{n})\|^{2} + \|H(w_{n})\|^{2}} \\ &- \alpha_{n}(1 - \alpha_{n})\|w_{n} - z_{n}\|^{2} - \|y_{n} - u_{n}\|^{2} + s_{n}^{2} \cdot \frac{\mu^{2}}{s_{n+1}^{2}}\|y_{n} - u_{n}\|^{2} \\ &- \alpha_{n}(1 - \alpha_{n})\|w_{n} - z_{n}\|^{2} - \left(1 - s_{n}^{2} \cdot \frac{\mu^{2}}{s_{n+1}^{2}}\right)\|y_{n} - u_{n}\|^{2} \\ &- 2\langle u_{n} - p, y_{n} - u_{n} - s_{n}(By_{n} - Bu_{n})\rangle. \end{split}$$

$$(5.3.11)$$

From **Step 4** we have that $u_n = (I + s_n D)^{-1} (I - s_n B) y_n$, hence $(I - s_n B) y_n \in (I + s_n D) u_n$. Since D is maximal monotone, there exists $k_n \in Du_n$ such that

$$(I - s_n B)y_n = u_n + s_n k_n$$

which implies that

$$k_n = \frac{1}{s_n} (y_n - s_n B y_n - u_n).$$
(5.3.12)

We have that $0 \in (B+D)p$ and $Bu_n + k_n \in (B+D)u_n$. Since B+D is maximal monotone we obtain

$$\langle Bu_n + k_n, \ u_n - p \rangle \ge 0. \tag{5.3.13}$$

Substituting (5.3.12) into (5.3.13) we have

$$\frac{1}{s_n}\langle s_n B u_n + y_n - s_n B y_n - u_n, \quad u_n - p \rangle \ge 0$$

which implies that

$$\langle y_n - u_n - s_n (By_n - Bu_n), \quad u_n - p \rangle \ge 0.$$
(5.3.14)

Applying (5.3.14) in (5.3.11) we have

$$\|v_n - p\|^2 \le \|w_n - p\|^2 - \tau_n (4 - \tau_n) (1 - \alpha_n) \frac{g^2(w_n)}{\|G(w_n)\|^2 + \|H(w_n)\|^2}$$

$$- \alpha_n (1 - \alpha_n) \|w_n - z_n\|^2 - \left(1 - s_n^2 \cdot \frac{\mu^2}{s_{n+1}^2}\right) \|y_n - u_n\|^2,$$
(5.3.15)

which is the desired result.

Lemma 5.3.6. Let $\{x_n\}$ be a sequence generated by Algorithm 5.3.2 under Assumption 5.3.1. Then, $\{x_n\}$ is bounded.

Proof. Let $p \in \Gamma$. By applying the triangular inequality, from **Step 2** we have

$$|w_n - p|| = ||x_n + \theta_n(x_n - x_{n-1}) - p||$$

$$\leq ||x_n - p|| + \theta_n ||x_n - x_{n-1}||.$$

From Remark 5.3.2, $\lim_{n\to\infty} \frac{\theta_n}{\gamma_n} ||x_n - x_{n-1}|| = 0$. Hence, there exists a constant $M_1 > 0$ such that $\frac{\theta_n}{\gamma_n} ||x_n - x_{n-1}|| \le M_1$ for all $n \ge 1$. This implies that

$$||w_n - p|| \le ||x_n - p|| + \gamma_n M_1.$$
(5.3.16)

Also, since $\lim_{n\to\infty} \left(1 - \mu^2 \frac{s_n^2}{s_{n+1}^2}\right) = 1 - \mu^2 > 0$ there exists $n_0 \in \mathbb{N}$ such that

$$1 - \mu^2 \frac{s_n^2}{s_{n+1}^2} > 0, \quad \forall \ n \ge n_0.$$
(5.3.17)

Applying (5.3.17) and the conditions on τ_n and α_n in (5.3.15), we have

$$||v_n - p|| \le ||w_n - p||, \quad \forall \ n \ge n_0.$$
 (5.3.18)

From the definition of x_{n+1} in **Step 4** and (5.3.18), we have for all $n \ge n_0$

$$\|x_{n+1} - p\| = \|(1 - \beta_n - \gamma_n)(w_n - p) + \beta_n(v_n - p) - \gamma_n p\|$$

$$\leq \|(1 - \beta_n - \gamma_n)(w_n - p) + \beta_n(v_n - p)\| + \gamma_n \|p\|$$
(5.3.19)

and

$$\begin{aligned} \|(1 - \beta_n - \gamma_n)(w_n - p) + \beta_n(v_n - p)\|^2 &= (1 - \beta_n - \gamma_n)^2 \|w_n - p\|^2 + \beta_n^2 \|v_n - p\|^2 \\ &+ 2(1 - \beta_n - \gamma_n)\beta_n \langle w_n - p, v_n - p \rangle \\ &\leq (1 - \beta_n - \gamma_n)^2 \|w_n - p\|^2 + \beta_n^2 \|w_n - p\|^2 \\ &+ 2(1 - \beta_n - \gamma_n)\beta_n \|w_n - p\|^2 \\ &= (1 - \gamma_n)^2 \|w_n - p\|^2. \end{aligned}$$

Substituting the last inequality into (5.3.19) and applying (5.3.16), we have for all $n \ge n_0$

$$\begin{aligned} \|x_{n+1} - p\| &\leq (1 - \gamma_n) \|w_n - p\| + \gamma_n \|p\| \\ &\leq (1 - \gamma_n) \left[\|x_n - p\| + \gamma_n M_1 \right] + \gamma_n \|p\| \\ &\leq (1 - \gamma_n) \|x_n - p\| + \gamma_n (M_1 + \|p\|) \\ &\leq \max \left\{ \|x_n - p\|, \quad M_1 + \|p\| \right\} \\ &\vdots \\ &\leq \max \left\{ \|x_{n_0} - p\|, \quad M_1 + \|p\| \right\}, \end{aligned}$$

which implies that $\{x_n\}$ is bounded. Consequently, $\{w_n\}, \{z_n\}, \{y_n\}, \{u_n\}$ and $\{v_n\}$ are also bounded.

Lemma 5.3.7. Assume that $\{u_n\}$ and $\{y_n\}$ are sequences generated by Algorithm 5.3.2 under Assumption 5.3.1 such that $\lim_{j\to\infty} ||u_{n_j} - y_{n_j}|| = 0$ for some subsequences $\{u_{n_j}\}$ and $\{y_{n_j}\}$ of $\{u_n\}$ and $\{y_n\}$, respectively. If $\{y_{n_j}\}$ converges weakly to some $x^* \in \mathcal{H}_1$ as $j \to \infty$, then $x^* \in (B+D)^{-1}(0)$.

Proof. Let $(u, v) \in G(B + D)$, that is $v - Bu \in Du$. Since

$$u_{n_j} = (I + s_{n_j}D)^{-1}(I - s_{n_j}B)y_{n_j},$$

then we have

$$(I - s_{n_j}B)y_{n_j} \in (I + s_{n_j}D)u_{n_j}.$$

This implies that

$$\frac{1}{s_{n_j}}\Big(y_{n_j}-u_{n_j}-s_{n_j}By_{n_j}\Big)\in Du_{n_j}.$$

Since D is maximal monotone, we have

$$\left\langle u - u_{n_j}, \quad v - Bu - \frac{1}{s_{n_j}} \left(y_{n_j} - u_{n_j} - s_{n_j} B y_{n_j} \right) \right\rangle \ge 0,$$

which implies that

$$\langle u - u_{n_j}, v \rangle - \langle u - u_{n_j}, Bu + \frac{1}{s_{n_j}} (y_{n_j} - u_{n_j} - s_{n_j} By_{n_j}) \rangle \ge 0.$$

Hence, we have

$$\left\langle u - u_{n_j}, v \right\rangle \geq \left\langle u - u_{n_j}, Bu + \frac{1}{s_{n_j}} \left(y_{n_j} - u_{n_j} - s_{n_j} By_{n_j} \right) \right\rangle$$

$$= \left\langle u - u_{n_j}, Bu - By_{n_j} \right\rangle + \left\langle u - u_{n_j}, \frac{1}{s_{n_j}} (y_{n_j} - u_{n_j}) \right\rangle$$

$$= \left\langle u - u_{n_j}, Bu - Bu_{n_j} \right\rangle + \left\langle u - u_{n_j}, Bu_{n_j} - By_{n_j} \right\rangle$$

$$+ \left\langle u - u_{n_j}, \frac{1}{s_{n_j}} (y_{n_j} - u_{n_j}) \right\rangle$$

$$\ge \left\langle u - u_{n_j}, Bu_{n_j} - By_{n_j} \right\rangle + \left\langle u - u_{n_j}, \frac{1}{s_{n_j}} (y_{n_j} - u_{n_j}) \right\rangle.$$

Now, since $\lim_{j\to\infty} ||u_{n_j} - y_{n_j}|| = 0$ and B is Lipschitz continuous, we have $\lim_{j\to\infty} ||Bu_{n_j} - By_{n_j}|| = 0$. Moreover, since $\lim_{n\to\infty} s_n = s \ge \min\{s_0, \frac{\mu}{L}\}$, we get

$$\langle u - x^*, v \rangle = \lim_{j \to \infty} \langle u - u_{n_j}, v \rangle \ge 0.$$

By the maximal monotonicity of B + D, it follows that $x^* \in (B + D)^{-1}(0)$. \Box Lemma 5.3.8. The following inequality holds for all $p \in \Gamma$ and $n \in \mathbb{N}$:

$$\begin{aligned} \|x_{n+1} - p\|^2 &\leq (1 - \gamma_n) \|x_n - p\|^2 + \gamma_n \Big(3\frac{\theta_n}{\gamma_n} \|x_n - x_{n-1}\| M_2 + \|p\|^2 \Big) \\ &- \beta_n \tau_n (4 - \tau_n) (1 - \alpha_n) \frac{g^2(w_n)}{\|G(w_n)\|^2 + \|H(w_n)\|^2} - \beta_n \alpha_n (1 - \alpha_n) \|w_n - z_n\|^2 \\ &- \beta_n \Big(1 - s_n^2 \cdot \frac{\mu^2}{s_{n+1}^2} \Big) \|y_n - u_n\|^2 - (1 - \beta_n - \gamma_n) \beta_n \|w_n - v_n\|^2 \end{aligned}$$

Proof. Let $p \in \Gamma$. Then, by applying Lemma 2.1.1 and the Cauchy-Schwartz inequality, from **Step 2** we obtain

$$\begin{aligned} \|w_{n} - p\|^{2} &= \|x_{n} + \theta_{n}(x_{n} - x_{n-1}) - p\|^{2} \\ &= \|x_{n} - p\|^{2} + \theta_{n}^{2}\|x_{n} - x_{n-1}\|^{2} + 2\theta_{n}\langle x_{n} - p, x_{n} - x_{n-1}\rangle \\ &\leq \|x_{n} - p\|^{2} + \theta_{n}^{2}\|x_{n} - x_{n-1}\|^{2} + 2\theta_{n}\|x_{n} - p\|\|x_{n} - x_{n-1}\| \\ &= \|x_{n} - p\|^{2} + \theta_{n}\|x_{n} - x_{n-1}\| \left[\theta_{n}\|x_{n} - x_{n-1}\| + 2\|x_{n} - p\|\right] \\ &\leq \|x_{n} - p\|^{2} + 3\theta_{n}\|x_{n} - x_{n-1}\|M_{2} \\ &= \|x_{n} - p\|^{2} + 3\gamma_{n}\frac{\theta_{n}}{\gamma_{n}}\|x_{n} - x_{n-1}\|M_{2}, \end{aligned}$$
(5.3.20)

for some $M_2 > 0$. From **Step 5** of Algorithm 5.3.2 and by applying Lemma 2.1.1, (5.3.15) and (5.3.20) we have

$$\begin{split} \|x_{n+1} - p\|^2 &= \|(1 - \beta_n - \gamma_n)w_n + \beta v_n - p\|^2 \\ &= \|(1 - \beta_n - \gamma_n)\|w_n - p\|^2 + \beta_n \|v_n - p\|^2 + \gamma_n \|p\|^2 \\ &\leq (1 - \beta_n - \gamma_n)\|w_n - p\|^2 + \beta_n \|v_n - p\|^2 + \gamma_n \|p\|^2 \\ &\leq (1 - \beta_n - \gamma_n)\|w_n - p\|^2 \\ &+ \beta_n \Big[\|w_n - p\|^2 - \tau_n (4 - \tau_n)(1 - \alpha_n) \frac{g^2(w_n)}{\|G(w_n)\|^2 + \|H(w_n)\|^2} \\ &- \alpha_n (1 - \alpha_n)\|w_n - z_n\|^2 - \Big(1 - s_n^2 \cdot \frac{\mu^2}{s_{n+1}^2}\Big)\|y_n - u_n\|^2\Big] + \gamma_n \|p\|^2 \\ &- (1 - \beta_n - \gamma_n)\beta_n\|w_n - v_n\|^2 \\ &= (1 - \gamma_n)\|w_n - p\|^2 - \beta_n \tau_n (4 - \tau_n)(1 - \alpha_n) \frac{g^2(w_n)}{\|G(w_n)\|^2 + \|H(w_n)\|^2} \\ &- \beta_n \alpha_n (1 - \alpha_n)\|w_n - z_n\|^2 - \beta_n \Big(1 - s_n^2 \cdot \frac{\mu^2}{s_{n+1}^2}\Big)\|y_n - u_n\|^2 + \gamma_n \|p\|^2 \\ &- (1 - \beta_n - \gamma_n)\beta_n\|w_n - v_n\|^2 \\ &\leq (1 - \gamma_n) \left[\|x_n - p\|^2 + 3\gamma_n \frac{\theta_n}{\gamma_n}\|x_n - x_{n-1}\|M_2\right] \\ &- \beta_n \alpha_n (1 - \alpha_n)\|w_n - z_n\|^2 - \beta_n \Big(1 - s_n^2 \cdot \frac{\mu^2}{s_{n+1}^2}\Big)\|y_n - u_n\|^2 + \gamma_n \|p\|^2 \\ &- \beta_n \alpha_n (1 - \alpha_n)\|w_n - z_n\|^2 - \beta_n \Big(1 - s_n^2 \cdot \frac{\mu^2}{s_{n+1}^2}\Big)\|y_n - u_n\|^2 + \gamma_n \|p\|^2 \\ &- (1 - \beta_n - \gamma_n)\beta_n\|w_n - v_n\|^2 \\ &\leq (1 - \gamma_n)\|x_n - p\|^2 + \gamma_n \Big(3\frac{\theta_n}{\gamma_n}\|x_n - x_{n-1}\|M_2 + \|p\|^2\Big) \\ &- \beta_n \tau_n (4 - \tau_n)(1 - \alpha_n)\frac{g^2(w_n)}{\|G(w_n)\|^2 + \|H(w_n)\|^2} - \beta_n \alpha_n (1 - \alpha_n)\|w_n - z_n\|^2 \\ &- \beta_n (1 - s_n^2 \cdot \frac{\mu^2}{s_{n+1}^2}\Big)\|y_n - u_n\|^2 - (1 - \beta_n - \gamma_n)\beta_n\|w_n - z_n\|^2 \end{split}$$

which is the desired result.

Lemma 5.3.9. The following inequality holds for all $p \in \Gamma$, $M_2 > 0$ and $n \ge n_0$:

$$||x_{n+1} - p||^{2} \leq (1 - \gamma_{n})||x_{n} - p||^{2} + \gamma_{n} \Big[3\frac{\theta_{n}}{\gamma_{n}} ||x_{n} - x_{n-1}||M_{2} + 2\beta_{n} ||v_{n} - w_{n}|| ||p - x_{n+1}|| + 2\langle p, p - x_{n+1} \rangle \Big].$$
(5.3.21)

Proof. Let $p \in \Gamma$ and let $d_n = (1 - \beta_n)w_n + \beta_n v_n$ then we have

$$||d_n - w_n|| = ||(1 - \beta_n)w_n + \beta_n v_n - w_n|| = \beta_n ||v_n - w_n||.$$
(5.3.22)

Applying Lemma 2.1.1 and (5.3.18), we have

$$\begin{aligned} \|d_n - p\| &= \|(1 - \beta_n)w_n + \beta_n v_n - p\| \\ &\leq (1 - \beta_n)\|w_n - p\| + \beta_n\|v_n - p\| \\ &\leq (1 - \beta_n)\|w_n - p\| + \beta_n\|w_n - p\| \\ &= \|w_n - p\|, \quad \forall \quad n \ge n_0. \end{aligned}$$
(5.3.23)

Also, from the definition of x_{n+1} in Step 5, (5.3.22) and (5.3.23), we have

$$\begin{split} \|x_{n+1} - p\|^2 &= \|(1 - \beta_n - \gamma_n)w_n + \beta_n v_n - p\|^2 \\ &= \|(1 - \beta_n)w_n + \beta_n v_n - \gamma_n w_n - p\|^2 \\ &= \|(1 - \gamma_n)w_n - p\|^2 \\ &= \|(1 - \gamma_n)(d_n - p) - \gamma_n(w_n - d_n) - \gamma_n p\|^2 \\ &\leq (1 - \gamma_n)^2 \|d_n - p\|^2 - 2\langle \gamma_n(w_n - d_n) + \gamma_n p, \quad x_{n+1} - p \rangle \\ &= (1 - \gamma_n)^2 \|d_n - p\|^2 + 2\langle \gamma_n(w_n - d_n) + \gamma_n p, \quad p - x_{n+1} \rangle \\ &= (1 - \gamma_n) \|d_n - p\|^2 + 2\langle \gamma_n(w_n - d_n), \quad p - x_{n+1} \rangle + 2\langle \gamma_n p, \quad p - x_{n+1} \rangle \\ &\leq (1 - \gamma_n) \|d_n - p\|^2 + 2\gamma_n \|w_n - d_n\| \|p - x_{n+1}\| + 2\langle \gamma_n p, \quad p - x_{n+1} \rangle \\ &\leq (1 - \gamma_n) \left[\|x_n - p\|^2 + 3\gamma_n \frac{\theta_n}{\gamma_n} \|x_n - x_{n-1}\| M_2 \right] \\ &+ 2\gamma_n \|w_n - d_n\| \|p - x_{n+1}\| + 2\gamma_n \langle p, \quad p - x_{n+1} \rangle \\ &\leq (1 - \gamma_n) \|x_n - p\|^2 \\ &+ \gamma_n \left[3\frac{\theta_n}{\gamma_n} \|x_n - x_{n+1}\| M_2 + 2\beta_n \|v_n - w_n\| \|p - x_{n+1}\| + 2\langle p, \quad p - x_{n+1} \rangle \right], \end{split}$$
which is the desired result.

which is the desired result.

We now prove the strong convergence theorem for Algorithm 5.3.2.

Theorem 5.3.10. Let \mathcal{H}_1 and \mathcal{H}_2 be two real Hilbert spaces and $A : \mathcal{H}_1 \to \mathcal{H}_2$ be a bounded linear operator. Let $\{x_n\}$ be generated by Algorithm 5.3.2 and suppose Assumption 5.3.1 is satisfied. Then, the sequence $\{x_n\}$ converges strongly to a point $q \in \Gamma$, where $||q|| = \min\{||z|| : z \in \Gamma\}.$

Proof. Since $||q|| = \min\{||z|| : z \in \Gamma\}$, then $q = P_{\Gamma}(0)$. It follows that $q \in \Gamma$. Now, from Lemma 5.3.9 we obtain

$$\begin{aligned} \|x_{n+1} - q\|^2 &\leq (1 - \gamma_n) \|x_n - q\|^2 \\ &+ \gamma_n \Big[3\frac{\theta_n}{\gamma_n} \|x_n - x_{n-1}\| M_2 + 2\beta_n \|v_n - w_n\| \|q - x_{n+1}\| + 2\langle q, q - x_{n+1}\rangle \Big] \\ &= (1 - \gamma_n) \|x_n - q\|^2 + \gamma_n a_n \end{aligned}$$
(5.3.24)

where $a_n = 3\frac{\theta_n}{\gamma_n} \|x_n - x_{n-1}\| M_2 + 2\beta_n \|v_n - w_n\| \|q - x_{n+1}\| + 2\langle q, q - x_{n+1} \rangle.$

Now, we claim that $\{\|x_n - q\|\}$ converges to zero. To establish this, by Lemma 2.5.36 it suffices to show that $\limsup_{k \to \infty} a_{n_k} \leq 0$ for every subsequence $\{\|x_{n_k} - q\|\}$ of $\{\|x_n - q\|\}$ satisfying the condition

$$\liminf_{k \to \infty} \left(\|x_{n_{k+1}} - q\| - \|x_{n_k} - q\| \right) \ge 0.$$
(5.3.25)

Suppose that $\{||x_{n_k} - q||\}$ is a subsequence of $\{||x_n - q||\}$ such that (5.3.25) holds. We obtain from Lemma 5.3.8 that

$$\beta_{n_{k}}\tau_{n_{k}}(4-\tau_{n_{k}})(1-\alpha_{n_{k}})\frac{g^{2}(w_{n_{k}})}{\|G(w_{n_{k}})\|^{2}+\|H(w_{n_{k}})\|^{2}}+\beta_{n_{k}}\alpha_{n_{k}}(1-\alpha_{n_{k}})\|w_{n_{k}}-z_{n_{k}}\|^{2}$$
$$+\beta_{n_{k}}\left(1-s_{n_{k}}^{2}\cdot\frac{\mu^{2}}{s_{n_{k}+1}^{2}}\right)\|y_{n_{k}}-u_{n_{k}}\|^{2}+(1-\beta_{n_{k}}-\gamma_{n_{k}})\beta_{n_{k}}\|w_{n_{k}}-v_{n_{k}}\|^{2}$$
$$\leq (1-\gamma_{n_{k}})\|x_{n_{k}}-q\|^{2}-\|x_{n_{k}+1}-q\|^{2}$$
$$+\gamma_{n_{k}}\left(3\frac{\theta_{n_{k}}}{\gamma_{n_{k}}}\|x_{n_{k}}-x_{n_{k}-1}\|M_{2}+\|q\|^{2}\right).$$

By applying (5.3.25) and the fact that $\lim_{k\to\infty} \gamma_{n_k} = 0$, we have that

$$\lim_{k \to \infty} \beta_{n_k} \tau_{n_k} (4 - \tau_{n_k}) (1 - \alpha_{n_k}) \frac{g^2(w_{n_k})}{\|G(w_{n_k})\|^2 + \|H(w_{n_k})\|^2} = 0;$$

$$\lim_{k \to \infty} \beta_{n_k} \alpha_{n_k} (1 - \alpha_{n_k}) \|w_{n_k} - z_{n_k}\|^2 = 0;$$

$$\lim_{k \to \infty} \beta_{n_k} \left(1 - s_{n_k}^2 \cdot \frac{\mu^2}{s_{n_k+1}^2} \right) \|y_{n_k} - u_{n_k}\|^2 = 0;$$

$$\lim_{k \to \infty} (1 - \beta_{n_k} - \gamma_{n_k}) \beta_{n_k} \|w_{n_k} - v_{n_k}\|^2 = 0.$$

By the conditions on the control parameters, it follows that

$$\lim_{k \to \infty} \frac{g^2(w_{n_k})}{\|G(w_{n_k})\|^2 + \|H(w_{n_k})\|^2} = \lim_{k \to \infty} \|w_{n_k} - z_{n_k}\| = \lim_{k \to \infty} \|y_{n_k} - u_{n_k}\| = \lim_{k \to \infty} \|w_{n_k} - v_{n_k}\| = 0.$$
(5.3.26)

Since G and H are Lipschitz continuous, we have that

$$\lim_{k \to \infty} g(w_{n_k}) = 0.$$
 (5.3.27)

From the definition of $g(w_{n_k})$ we have

$$\lim_{k \to \infty} g(w_{n_k}) = \lim_{k \to \infty} \frac{1}{2} \| (I - T_{r_{n_k}}^{F_2}) A w_{n_k} \|^2 = 0,$$

which implies that

$$\lim_{k \to \infty} \| (I - T_{r_{n_k}}^{F_2}) A w_{n_k} \| = 0.$$
(5.3.28)

Consequently, we get

$$\lim_{k \to \infty} \|A^*(I - T_{r_{n_k}}^{F_2})Aw_{n_k}\| \le \|A^*\| \|(I - T_{r_{n_k}}^{F_2})Aw_{n_k}\| = \|A\| \|(I - T_{r_{n_k}}^{F_2})Aw_{n_k}\| = 0.$$
(5.3.29)

From the definition of y_{n_k} and by applying (5.3.26) we have

$$\|y_{n_k} - z_{n_k}\| \le \alpha_{n_k} \|w_{n_k} - z_{n_k}\| + (1 - \alpha_{n_k}) \|z_{n_k} - z_{n_k}\| \to 0, \quad k \to \infty.$$
(5.3.30)

From (5.3.26) and (5.3.30) we have

$$\lim_{k \to \infty} \|w_{n_k} - y_{n_k}\| = 0; \quad \lim_{k \to \infty} \|w_{n_k} - u_{n_k}\| = 0; \quad \lim_{k \to \infty} \|z_{n_k} - u_{n_k}\| = 0.$$
(5.3.31)

From **Step 5** and by applying (5.3.26) together with the fact that $\lim_{k\to\infty} \gamma_{n_k} = 0$, we have

$$\|x_{n_{k+1}} - w_{n_k}\| = \|(1 - \beta_{n_k} - \gamma_{n_k})w_{n_k} + \beta_{n_k}v_{n_k} - w_{n_k}\| = \|\beta_{n_k}(v_{n_k} - w_{n_k}) - \gamma_{n_k}w_{n_k}\| \leq \beta_{n_k}\|v_{n_k} - w_{n_k}\| + \gamma_{n_k}\|w_{n_k}\| \to 0, \quad k \to \infty.$$
(5.3.32)

Also, by Remark 5.3.2 we have

$$||w_{n_k} - x_{n_k}|| = \theta_{n_k} ||x_{n_k} - x_{n_{k-1}}|| \to 0, \quad k \to \infty.$$
(5.3.33)

Also, from (5.3.32) and (5.3.33) we have

$$\lim_{k \to \infty} \|x_{n_k+1} - x_{n_k}\| = 0 \tag{5.3.34}$$

From (5.3.26)-(5.3.33) we obtain

$$\lim_{k \to \infty} \|x_{n_k} - z_{n_k}\| = 0, \ \lim_{k \to \infty} \|x_{n_k} - v_{n_k}\| = 0, \ \lim_{k \to \infty} \|x_{n_k} - u_{n_k}\| = 0, \ \lim_{k \to \infty} \|x_{n_k} - y_{n_k}\| = 0.$$
(5.3.35)

Since $\{x_n\}$ is bounded, $w_{\omega}(x_n)$ is nonempty. Let $x^* \in w_{\omega}(x_n)$ be an arbitrary element. Then, there exists a subsequence $\{x_{n_k}\} \subset \{x_n\}$ such that $x_{n_k} \rightharpoonup x^*$ as $k \rightarrow \infty$. From (5.3.35) we obtain $y_{n_k} \rightharpoonup x^*$ as $k \rightarrow \infty$. By invoking Lemma 5.3.7, it follows from (5.3.26) that $x^* \in (B+D)^{-1}(0)$. Since x^* is an arbitrary element in $w_{\omega}(x_n)$, it follows that $w_{\omega}(x_n) \subset (B+D)^{-1}(0)$. Next, we show that $w_{\omega}(x_n) \subset \Omega$. First, we establish that $w_{\omega}(x_n) \subset EP(F_1)$. From $z_{n_k} = T_{r_{n_k}}^{F_1}(I - \lambda_{n_k}A^*(I - T_{r_{n_k}}^{F_2})A)w_{n_k}$, we have

$$F_1(z_{n_k}, y) + \frac{1}{r_{n_k}} \Big\langle y - z_{n_k}, \ z_{n_k} - w_{n_k} + \lambda_{n_k} G(w_{n_k}) \Big\rangle \ge 0$$

for all $y \in \mathcal{C}$ and this implies that

$$F_1(z_{n_k}, y) + \frac{1}{r_{n_k}} \Big\langle y - z_{n_k}, \ z_{n_k} - w_{n_k} \Big\rangle + \frac{1}{r_{n_k}} \Big\langle y - z_{n_k}, \ \lambda_{n_k} G(w_{n_k}) \Big\rangle \ge 0, \ \forall \ y \in \mathcal{C}.$$

We have from Assumption 5.2.1(2) that

$$\frac{1}{r_{n_k}} \Big\langle y - z_{n_k}, \ z_{n_k} - w_{n_k} \Big\rangle + \frac{1}{r_{n_k}} \Big\langle y - z_{n_k}, \ \lambda_{n_k} G(w_{n_k}) \Big\rangle \ge F_1(y, \ z_{n_k}), \ \forall \ y \in \mathcal{C}.$$

Since $z_{n_k} \rightharpoonup x^*$, then by applying (5.3.26), (5.3.29), Assumption 5.2.1 (4) and the fact that $\liminf_{k \to \infty} r_{n_k} > 0$, we obtain

$$F_1(y, x^*) \le 0, \ \forall \ y \in \mathcal{C}.$$
 (5.3.36)

Let $y_t = ty + (1 - t)x^*$, $\forall t \in (0, 1]$ and $y \in C$. This implies that $y_t \in C$. Hence, from (5.3.36) we get $F_1(y_t, x^*) \leq 0$. Applying Assumption 5.2.1 (1)-(4), we have

$$0 = F_1(y_t, y_t) \leq tF_1(y_t, y) + (1 - t)F_1(y_t, x^*) \leq tF_1(y_t, y).$$

Hence, we have $F_1(y_t, y) \ge 0$, $\forall y \in \mathcal{C}$. Letting $t \to 0$ and applying Assumption 5.2.1 (3), we have

$$F_1(x^*, y) \ge 0, \ \forall y \in \mathcal{C}, \tag{5.3.37}$$

which implies that $x^* \in EP(F_1)$.

Next, we show that $Ax^* \in EP(F_2)$. Since A is a bounded linear operator and $w_{n_k} \rightharpoonup x^*$, we have $Aw_{n_k} \rightharpoonup Ax^*$. Consequently, it follows from (5.3.28) that

$$T_{r_{n_k}}^{F_2} A w_{n_k} \rightharpoonup A x^* \tag{5.3.38}$$

as $k \to \infty$. From the definition of $T_{r_{n_k}}^{F_2} A w_{n_k}$, we have

$$F_2(T_{r_{n_k}}^{F_2}Aw_{n_k}, y) + \frac{1}{r_{n_k}} \Big\langle y - T_{r_{n_k}}^{F_2}Aw_{n_k}, \ T_{r_{n_k}}^{F_2}Aw_{n_k} - Aw_{n_k} \Big\rangle \ge 0, \ \forall \ y \in \mathcal{Q}.$$

Since F_2 is upper semicontinuous in the first argument, it follows from (5.3.28), (5.3.38) and the fact that $\liminf_{k\to\infty} r_{n_k} > 0$, that

$$F_2(Ax^*, y) \ge 0, \ \forall \ y \in \mathcal{Q}, \tag{5.3.39}$$

which implies that $Ax^* \in EP(F_2)$. Thus, $w_{\omega}(x_n) \subset \Gamma$.

Since $\{x_{n_k}\}$ is bounded, there exists a subsequence $\{x_{n_{k_j}}\}$ of $\{x_{n_k}\}$ which converges weakly to \hat{x} such that

$$\limsup_{k \to \infty} \left\langle q, \ q - x_{n_k} \right\rangle = \lim_{j \to \infty} \left\langle q, \ q - x_{n_{k_j}} \right\rangle.$$

Since $q = P_{\Gamma}(0)$, we have

$$\limsup_{k \to \infty} \left\langle q, \ q - x_{n_k} \right\rangle = \lim_{j \to \infty} \left\langle q, \ q - x_{n_{k_j}} \right\rangle = \left\langle q, \ q - \hat{x} \right\rangle \le 0,$$

which together with (5.3.34) implies that

$$\lim_{k \to \infty} \left\langle q, \ q - x_{n_{k+1}} \right\rangle \le 0. \tag{5.3.40}$$

Using (5.3.26), (5.3.40) together with the fact that $\lim_{n\to\infty} \frac{\theta_n}{\gamma_n} ||x_n - x_{n-1}|| = 0$, we have $\limsup_{k\to\infty} a_{n_k} \leq 0$. Now, applying Lemma 2.5.36 to (5.3.24) we have that $\lim_{n\to\infty} ||x_n - q|| = 0$. Therefore, $\{x_n\}$ converges strongly to q.

	_	_	
F			
L			
L			
L	_	_	

5.3.3 Applications

In this section, we apply our result to study certain optimization problems.

Variational inclusion and split variational inequality problems

In this subsection, we apply our result to approximate the common solution of V_q IP and SVIP.

Here, we apply our result to study the following SVIP with constraint of V_q IP:

Find
$$x^* \in (B+D)^{-1}(0)$$
 such that $\langle x - x^*, M_1 x^* \rangle \ge 0, \quad \forall x \in \mathcal{C}.$ (5.3.41)

and such that

$$y^* = Ax^* \in \mathcal{Q} \text{ solves } \langle y - y^*, M_2 y^* \rangle \ge 0, \quad \forall \ y \in \mathcal{Q}.$$
 (5.3.42)

where \mathcal{C} and \mathcal{Q} are nonempty, closed and convex subsets of real Hilbert spaces \mathcal{H}_1 and \mathcal{H}_2 , respectively, $M : \mathcal{H} \to \mathcal{H}$ be a single-valued mapping, $A : \mathcal{H}_1 \to \mathcal{H}_2$ is a bounded linear

operator, $B : \mathcal{H}_1 \to \mathcal{H}_1$ is a Lipschitz continuous and monotone operator and $D : \mathcal{H}_1 \to 2^{\mathcal{H}_1}$ is a maximal monotone operator. We denote the solution set of problem (5.3.41)-(5.3.42) by Ω_1 and assume that $(B+D)^{-1}(0) \cap \Omega_1 \neq \emptyset$. Taking $F_i(x,y) := \langle y-x, M_i x \rangle, i = 1, 2$, then the problem (5.3.41)-(5.3.42) becomes the problem of finding a solution of the SEP (1.2.10)-(1.2.11) which is also a solution of the V_q IP (1.2.9). Furthermore, all the conditions of Theorem 5.3.10 hold. Thus, Theorem 5.3.10 provides a strong convergence theorem for approximating a common solution of V_q IP (1.2.9) and the problem (5.3.41)-(5.3.42).

Convex minimization and split equilibrium problems

Let \mathcal{H} be a real Hilbert space, $f : \mathcal{H} \to \mathbb{R}$ be a convex function and $F : \mathcal{H} \to \mathbb{R}$ be a proper convex, lower semicontinuous function. We apply our result to approximate the solution of the following *convex MP*:

$$\min_{x \in H} f(x) + F(x), \tag{5.3.43}$$

which is equivalent to finding $x \in \mathcal{H}$ such that

$$0 \in \nabla f(x) + \partial F(x), \tag{5.3.44}$$

where ∇f is the gradient of f and ∂F is the subdifferential of F. It is known that if ∇f is L-Lipschitz continuous, then it is $\frac{1}{L}$ -inverse strongly monotone (co-coercive), and hence it is L-Lipschitz continuous and monotone. Also, it is known that ∂F is maximal monotone (see [216]). The solution set of (5.3.43) we denote by Ω_2 .

So, by setting $B = \nabla f$ and $D = \partial F$ in Theorem 5.3.10, we obtain the following result for approximating a common solution of convex MP (5.3.43) and split equilibrium problem (1.2.10)-(1.2.11) in Hilbert spaces.

Theorem 5.3.11. Let C and Q be nonempty, closed and convex subsets of real Hilbert spaces \mathcal{H}_1 and \mathcal{H}_2 , respectively. Suppose $F_1 : C \times C \to \mathbb{R}$, $F_2 : Q \times Q \to \mathbb{R}$ are bifunctions satisfying Assumption (5.2.1) and such that F_2 is upper semi continuous in the first argument. Let $f : \mathcal{H} \to \mathbb{R}$ be a convex and differentiable function such that ∇f is L-Lipschitz continuous and suppose $G : \mathcal{H} \to \mathbb{R}$ is a proper convex, lower semicontinuous function. Let $\{x_n\}$ be a sequence generated as follows:

Algorithm 5.3.3.

Step 1: Select initial point $x_0, x_1 \in \mathcal{H}_1$, let $s_1 > 0, \mu \in (0, 1), \theta \ge 3$ and set n = 1. Given the iterates x_{n-1} and x_n for each $n \ge 1$, choose θ_n such that $0 \le \theta_n \le \overline{\theta}_n$, where

$$\bar{\theta}_n := \begin{cases} \min\left\{\frac{n-1}{n+\theta-1}, \frac{\epsilon_n}{\|x_n - x_{n-1}\|}\right\}, & \text{if } x_n \neq x_{n-1} \\ \frac{n-1}{n+\theta-1}, & \text{otherwise.} \end{cases}$$
(5.3.45)

Step 2: Compute

$$w_n = x_n + \theta_n (x_n - x_{n-1}).$$

Step 3: Compute

$$z_n = T_{r_n}^{F_1} (I - \lambda_n A^* (I - T_{r_n}^{F_2}) A) w_n$$

and

$$y_n = \alpha_n w_n + (1 - \alpha_n) z_n.$$

Step 4: Compute

$$u_n = J_{s_n}^{\partial F} (I - s_n \nabla f) y_n$$

and

$$v_n = u_n - s_n (\nabla f u_n - \nabla f y_n).$$

Step 5 Compute

$$x_{n+1} = (1 - \beta_n - \gamma_n)w_n + \beta_n v_n,$$

where

$$\lambda_n := \begin{cases} \tau_n \frac{g(w_n)}{\|G(w_n)\|^2 + \|H(w_n)\|^2}, & \text{if } \|G(w_n)\|^2 + \|H(w_n)\|^2 \neq 0, \\ 0, & \text{otherwise} \end{cases}$$
(5.3.46)

and

$$s_{n+1} = \begin{cases} \min\left\{\frac{\mu \|y_n - u_n\|}{\|\nabla f y_n - \nabla f u_n\|}, s_n\right\} & \text{if } \nabla f y_n - \nabla f u_n \neq 0, \\ s_n, & \text{otherwise.} \end{cases}$$
(5.3.47)

Set n := n + 1 and go back to **Step 1**.

Then, the sequence $\{x_n\}$ converges strongly to a point $q \in \Omega_2 \cap \Omega$, where $||q|| = \min\{||z|| : z \in \Omega_2 \cap \Omega\}$.

5.3.4 Numerical experiments

In this section, using some test examples, we discuss the numerical behavior of Algorithm 5.3.2 as well as compare it with the standard forward-backward method (i.e when $\theta_n = 0$), Algorithm 2.5.23 proposed by Cholamjiak *et al.* [81], the shrinking projection method of Cholamjiak *et al.* [81] (see Appendix 5.3.14), Appendix 5.3.15 and Appendix 5.3.16. We perform all implementations using Matlab 2016 (b), installed on a personal computer with Intel(R) Core(TM) i5-2600 CPU@2.30GHz and 8.00 Gb-RAM running on Windows 10 operating system. In Tables 5.3.1-5.3.2, "No. of Iter." means the number of iterations.

In our computations, we choose $\alpha_n = \frac{n+1}{2n+1}$, $\gamma_n = \frac{1}{n+2}$, $\beta_n = \frac{1}{2} - \gamma_n$, $\epsilon_n = \frac{1}{(n+2)^2}$, $\theta = 7$, $s_1 = 0.65$, $\mu = 0.8$, $\tau_n = \frac{6n+2}{2n+1}$, $r_n = \frac{3n+1}{2n+1}$, and we take $\delta_n = \frac{n+1}{2n+3}$, $t_n = 0.1$ in Algorithm 2.5.23 and Appendix 5.3.14. Also, for Appendix 5.3.16 we choose $f(x) = \frac{x}{2}$. Furthermore, in the implementation, we define $\text{TOL}_n := ||x_{n+1} - x_n||$ and use the stopping criterion $\text{TOL}_n < 10^{-2}$ for the iterative processes.

Example 5.3.12. Let $\mathcal{H}_1 = \mathcal{H}_2 = \mathbb{R}$, the set of all real numbers with the inner product defined by $\langle x, y \rangle = xy$, $\forall x, y \in \mathbb{R}$ and induced norm $|\cdot|$. For r > 0, Consider $\mathcal{C} = [-10, 10]$ and $\mathcal{Q} = [0, 20]$, we define the bifunctions $F_1 : \mathcal{C} \times \mathcal{C} \to \mathbb{R}$ and $F_2 : \mathcal{Q} \times \mathcal{Q} \to \mathbb{R}$ as follows;

 $F_1 = -2x^2 + xy + y^2$, $F_2 = -x^2 + xy$. After simple calculation and applying Lemma 5.2.1, we get

$$T_r^{F_1}(u) = \frac{u}{3r+1}, \quad \forall \ u \in \mathcal{C}$$

and

$$T_r^{F_2}(v) = \frac{v}{r+1}, \quad \forall \ v \in \mathcal{Q}.$$

Let $A : \mathcal{H}_1 \to \mathcal{H}_2$ be defined by Ax = 2x, $B : \mathcal{H}_1 \to \mathcal{H}_1$ be defined by $Bx = x + \sin x$ and $D : \mathcal{H}_1 \to \mathcal{H}_1$ be defined by Dx = 5x, where $x \in \mathcal{H}_1$. Clearly, we see that B is $\frac{1}{3}$ -inverse strongly monotone and D is maximal monotone.

We consider the following cases for the numerical experiments of this example and choose $\gamma = \frac{3}{20}$

Case I: Take $x_0 = -\frac{17}{33}$ and $x_1 = \frac{2}{15}$. **Case II:** Take $x_0 = -\frac{19}{35}$ and $x_1 = \frac{1}{6}$. **Case III:** Take $x_0 = \frac{11}{20}$ and $x_1 = \frac{1}{7}$. **Case IV:** Take $x_0 = -\frac{10}{19}$ and $x_1 = -\frac{5}{34}$.

We compare the performance of our Algorithm 5.3.2 with Algorithm 2.5.23, Appendix 5.3.14, Appendix 5.3.15 and Appendix 5.3.16. We plot the graphs of errors against the number of iterations in each case. The numerical results are reported in Table 5.3.1 and Figure 5.9.

		Alg.	Арр	App.	App.	Alg.
		2.5.23	5.3.14	5.3.15	5.3.16	5.3.2
Case I	No. of Iter.	8	3	11	17	5
	CPU time	0.0099	0.0071	0.0090	0.0063	0.0114
	(sec)					
Case II	No. of Iter.	8	3	11	17	5
	CPU time	0.0095	0.0075	0.0080	0.0072	0.0155
	(sec)					
Case III	No. of Iter.	8	3	11	16	5
	CPU time	0.0103	0.0081	0.0093	0.0084	0.0125
	(sec)					
Case IV	No. of Iter.	8	3	11	17	5
	CPU time	0.0098	0.0071	0.0082	0.0070	0.0111
	(sec)					

Table 5.3.1: Numerical results for Example 5.3.12

Example 5.3.13. Let $\mathcal{H}_1 = \mathcal{H}_2 = (l_2(\mathbb{R}), \|\cdot\|_2)$, where $l_2(\mathbb{R}) := \{x = (x_1, x_2, \dots, x_n, \dots), x_j \in \mathbb{R} : \sum_{j=1}^{\infty} |x_j|^2 < \infty\}, \||x||_2 = (\sum_{j=1}^{\infty} |x_j|^2)^{\frac{1}{2}}$ and $\langle x, y \rangle = \sum_{j=1}^{\infty} x_j y_j$ for all $x \in \ell_2(\mathbb{R})$. We define $F_1 : \mathcal{C} \times \mathcal{C} \to \mathbb{R}$ and $F_2 : \mathcal{Q} \times \mathcal{Q} \to \mathbb{R}$ by $F_1(x, y) = \langle L_1 x, y - x \rangle$ and $F_2(x, y) = \langle L_2 x, y - x \rangle$, where $L_1 x = \frac{x}{3}$ and $L_2 x = \frac{x}{2}$. One can easily verify that F_1 and F_2 satisfy Assumption 5.2.1. Let $A : \mathcal{H}_1 \to \mathcal{H}_2$ be defined by $Ax = \frac{x}{3}$ and $A^*y = \frac{y}{3}$. Then,



Figure 5.9: Top left: Case I; Top right: Case II; Bottom left: Case III; Bottom right: Case IV.

 ${\cal A}$ is a bounded linear operator. After simple calculation and applying Lemma 5.2.1, we obtain

$$T_r^{F_1}(u) = \frac{3u}{r+3}, \quad \forall \ u \in \mathcal{C},$$

and

$$T_s^{F_2}(v) = \frac{2v}{s+2}, \quad \forall \ v \in \mathcal{Q}.$$

Let $B : \mathcal{H}_1 \to \mathcal{H}_1$ be defined by $Bx = \frac{1}{3}x$ and $D : \mathcal{H}_1 \to \mathcal{H}_1$ be defined by Dx = 3x, where $x \in \mathcal{H}_1$. Clearly, we see that B is $\frac{1}{2}$ -inverse strongly monotone and D is maximal monotone.

We consider different initial values as follows: Case I: $x_0 = (-23, 1, -\frac{1}{23}, \cdots), x_1 = (\frac{1}{2}, \frac{1}{4}, \frac{1}{8}, \cdots);$ Case II: $x_0 = (37, 1, \frac{1}{37}, \cdots), x_1 = (\frac{5}{18}, \frac{5}{36}, \frac{5}{72}, \cdots);$ Case III: $x_0 = (-29, 1, -\frac{1}{29}, \cdots), x_1 = (-\frac{3}{8}, \frac{3}{16}, -\frac{1}{32}, \cdots);$ Case IV: $x_0 = (25, 1, \frac{1}{25}, \cdots), x_1 = (\frac{1}{2}, \frac{1}{6}, \frac{1}{18}).$

We compare the performance of our Algorithm 5.3.2 with Appendix 5.3.14, Appendix 5.3.15 and Appendix 5.3.16. We plot the graphs of errors against the number of iterations in each case. The numerical results are reported in Table 5.3.2 and Figure 5.10.

		Alg.	Арр	App.	App.	Alg.
		2.5.23	5.3.14	5.3.15	5.3.16	5.3.2
Case I	No. of Iter.	13	3	14	11	6
	CPU time	0.0137	0.0148	0.0145	0.0092	0.0175
	(sec)					
Case II	No. of Iter.	13	3	14	11	6
	CPU time	0.0114	0.0123	0.0120	0.0097	0.0170
	(sec)					
Case III	No. of Iter.	13	3	14	11	6
	CPU time	0.0113	0.0135	0.0154	0.0082	0.0150
	(sec)					
Case IV	No. of Iter.	13	3	14	11	6
	CPU time	0.0126	0.0162	0.0139	0.0122	0.0245
	(sec)					

 Table 5.3.2: Numerical results for Example 5.3.13

Appendix 5.3.14. The Algorithm in [81]. Initialization: Given $\gamma \in (0, \frac{1}{L})$. Let $x_0, x_1 \in \mathcal{H}_1$ be arbitrary. Iterative Steps: Calculate x_{n+1} as follows:



Figure 5.10: Top left: Case I; Top right: Case II; Bottom left: Case III; Bottom right: Case IV.

$$\begin{cases} y_n = x_n + \delta_n (x_n - x_{n-1}) \\ z_n = \alpha_n y_n + (1 - \alpha_n) T_{r_n}^{F_1} (I - \gamma A^* (I - T_{r_n}^{F_2}) A) y_n, \\ w_n = \beta_n z_n + (1 - \beta_n) J_{t_n}^D (I - t_n B) z_n, \\ C_{n+1} = \{ z \in C_n : \| w_n - z \|^2 \le \| x_n - z \|^2 + 2{\delta_n}^2 \| x_n - x_{n-1} \|^2 - 2\delta \langle x_n - z, x_{n-1} - x_n \rangle \}, \\ x_{n+1} = P_{C_{n+1}} x_1, \ n \ge 1, \end{cases}$$

$$(5.3.48)$$

where $J_{t_n}^D = (I + t_n D)^{-1}$, $\{t_n\} \subset (0, 2\alpha)$, $\{\delta_n\} \subset [0, \delta]$, $\delta \in [0, 1)$, $\{r_n\} \subset (0, \infty)$ with $\gamma \in (0, \frac{1}{L})$ such that L is the spectral radius of A^*A and $\{\alpha_n\}, \{\beta_n\}$ are sequences in [0, 1]. Set n := n + 1 and return to Step 1.

Appendix 5.3.15. The Algorithm in [108].

Initialization: Given $s_1 > 0, \mu \in (0, 1)$. Let $x_1 \in \mathcal{H}$ be arbitrary.

Iterative Steps: Given the current iterates x_n , calculate the next iterate as follows: Step 1.

$$y_n = (I + s_n D)^{-1} (I - s_n B) x_n.$$

If $x_n = y_n$ then stop and y_n is a solution of (1.2.9). Otherwise, Step 2. Compute

$$z_n = y_n - s_n (By_n - Bx_n)$$

and

$$x_{n+1} = (1 - \beta_n - \gamma_n)x_n + \gamma_n z_n.$$

Update

$$s_{n+1} = \begin{cases} \min\{\frac{\mu \|x_n - y_n\|}{\|Bx_n - By_n\|}, s_n\} & \text{if } Bx_n - By_n \neq 0, \\ s_n, & \text{otherwise.} \end{cases}$$
(5.3.49)

Set n := n + 1 and return to Step 1.

Appendix 5.3.16. The Algorithm in [108].

Initialization: Given $s_1 > 0, \mu \in (0, 1)$. Let $x_1 \in \mathcal{H}$ be arbitrary.

Iterative Steps: Given the current iterates x_n , calculate the next iterate as follows: Step 1.

$$y_n = (I + s_n D)^{-1} (I - s_n B) x_n.$$

If $x_n = y_n$ then stop and y_n is a solution of (1.2.9). Otherwise, Step 2. Compute

$$z_n = y_n - s_n (By_n - Bx_n)$$

and

$$x_{n+1} = \gamma_n f(x_n) + (1 - \gamma_n) z_n$$

Update

$$s_{n+1} = \begin{cases} \min\{\frac{\mu \|x_n - y_n\|}{\|Bx_n - By_n\|}, s_n\} & \text{if } Bx_n - By_n \neq 0, \\ s_n, & \text{otherwise.} \end{cases}$$
(5.3.50)

Set n := n + 1 and return to Step 1.

CHAPTER 6

Results on Variational Inequality, Split Monotone Variational Inclusion and Fixed Point Problems in Banach Spaces

6.1 Introduction

The extension of known basis concepts from Hilbert spaces to Banach spaces has been of great interest to many researchers due to the fact that many real life problems can be posed in general Banach spaces than in Hilbert spaces. In this chapter, we propose and study an iterative algorithm for approximating the solution of VIPs in a real Banach space. Furthermore, between a Banach and Hilbert space, we propose and study an effective iterative algorithm for approximating a common solution of the SMV_qIP and FPP. We apply our result to study some optimization problems and present some numerical experiments of our methods in comparison to other methods in literature.

6.2 On pseudomonotone variational inequalities with non-Lipschitz operators

In this section, we study the pseudomonotone VIP with non-Lipschitz operators in Banach spaces. We propose an inertial subgradient extragradient method with Halpern technique and Armijo type step size for approximating the solution of the problem in the framework of 2-uniformly convex real Banach spaces. We prove that the sequence generated by our proposed method converges strongly to the solution of the problem under some mild conditions and without the weakly sequential continuity condition often assumed by authors in solving pseudomonotone VIP. Finally, we provide some numerical experiments for the proposed method in comparison with other existing methods in the literature. Our result extends and improves several of the existing results in the current literature in this direction.

6.2.1 Proposed method

In this section, we present the proposed method and highlight some of its features. We start by giving the following assumptions under which our strong convergence result is obtained.

Assumption 6.2.1. Suppose that the following conditions hold:

- (a) \mathcal{X} is a real 2-uniformly convex and uniformly smooth Banach space with 2-uniform convexity constant μ and \mathcal{C} is a nonempty, closed and convex subset of \mathcal{X} .
- (b) $A: \mathcal{X} \to \mathcal{X}^*$ is pseudomonotone and uniformly continuous on \mathcal{X} .
- (c) The solution set $VI(\mathcal{C}, A)$ is nonempty.
- (d) Let $\{\tau_n\}$ be a positive sequence such that $\lim_{n \to \infty} \frac{\tau_n}{\alpha_n} = 0$ where α_n is a sequence in (0, 1)satisfying the following conditions: $\lim_{n \to \infty} \alpha_n = 0$, $\sum_{n=1}^{\infty} \alpha_n = \infty$.

Now, our proposed method for solving the VIP (1.2.1) is presented as follows:

Algorithm 6.2.2. Inertial method with new Armijo-type step size.

Initialization: Given $\lambda_1 > 0$, $\theta > 0$, $\delta > 0$, $\ell \in (0, 1)$ and $\eta \in (0, \frac{1}{\mu})$. Let $u, x_0, x_1 \in \mathcal{X}$ be arbitrary. Set n := 1.

Step 1: Given the iterates x_{n-1} and x_n for each $n \ge 1$, choose θ_n such that $0 \le \theta_n \le \overline{\theta}_n$, where

$$\bar{\theta}_n := \begin{cases} \min\left\{\theta, \frac{\tau_n}{\|x_n - x_{n-1}\|}\right\}, & \text{if } x_n \neq x_{n-1} \\ \theta, & \text{otherwise.} \end{cases}$$
(6.2.1)

Step 2: Compute

$$w_n = J^{-1}(Jx_n + \theta_n(Jx_{n-1} - Jx_n)).$$

Step 3: Compute

$$y_n = \Pi_{\mathcal{C}} J^{-1} (Jw_n - \lambda_n Aw_n).$$

If $w_n = y_n$, or $Ay_n = 0$ then stop and y_n is a solution of the VIP. Otherwise, go to **Step** 4.

Step 4: Compute

$$z_n = \prod_{T_n} J^{-1} (Jw_n - \lambda_n Ay_n)$$

where

$$T_n := \{ x \in \mathcal{X} \mid | \langle Jw_n - \lambda_n Aw_n - Jy_n, x - y_n \rangle \le 0 \},\$$
$\lambda_n := \delta \ell^{m_n}$ and m_n is the smallest non-negative integer m satisfying

$$\delta \ell^m \langle Ay_n - Aw_n, \quad y_n - z_n \rangle \le \frac{\eta}{2} \Big[\|w_n - y_n\|^2 + \|y_n - z_n\|^2 \Big]. \tag{6.2.2}$$

Step 5: Calculate

$$x_{n+1} = J^{-1}(\alpha_n Ju + (1 - \alpha_n) Jz_n).$$

Set n := n + 1 and go to **Step 1**.

Remark 6.2.1. If $w_n = y_n$ or $Ay_n = 0$ then $y_n \in VI(\mathcal{C}, A)$. From Lemma 2.5.23 and the fact that $0 < \lambda_n \leq \delta$, we have

$$0 = \frac{\|w_n - y_n\|}{\lambda_n} = \frac{\|w_n - \Pi_{\mathcal{C}}J^{-1}(Jw_n - \lambda_n Aw_n)\|}{\lambda_n} \ge \frac{\|w_n - \Pi_{\mathcal{C}}J^{-1}(Jw_n - \delta Aw_n)\|}{\delta}$$

which shows that w_n is a solution of the VIP. Thus, $y_n \in VI(\mathcal{C}, A)$. Also, since $y_n \in \mathcal{C}$, we observe that if $Ay_n = 0$ then $y_n = \prod_{\mathcal{C}} J^{-1}(Jy_n - \delta Ay_n)$, which implies that $y_n \in VI(\mathcal{C}, A)$.

6.2.2 Convergence analysis

Lemma 6.2.2. Suppose that Assumption 6.2.1 holds. The Armijo-like criteria (6.2.2) is well defined. Furthermore, we have that $\lambda_n \leq \delta$.

Proof. If $w_n \in VI(\mathcal{C}, A)$ then $w_n = \prod_{\mathcal{C}} J^{-1}(Jw_n - \delta Aw_n)$, which implies that $w_n = y_n$. Thus $m_n = 0$. If $w_n \notin VI(\mathcal{C}, A)$, we prove by contradiction by assuming that the contrast of (6.2.2) holds, that is,

$$\delta \ell^{m} \left\langle A \Pi_{\mathcal{C}} J^{-1} (J w_{n} - \delta \ell^{m} A w_{n}) - A w_{n}, \quad \Pi_{\mathcal{C}} J^{-1} (J w_{n} - \delta \ell^{m} A w_{n}) - \Pi_{T_{n}} J^{-1} (J w_{n} - \delta \ell^{m} A y_{n}) \right\rangle$$

$$> \frac{\eta}{2} \left[\| w_{n} - \Pi_{\mathcal{C}} J^{-1} (J w_{n} - \delta \ell^{m} A w_{n}) \|^{2} + \| \Pi_{\mathcal{C}} J^{-1} (J w_{n} - \delta \ell^{m} A w_{n}) - \Pi_{T_{n}} J^{-1} (J w_{n} - \delta \ell^{m} A y_{n}) \|^{2} \right]$$

This implies that

$$\delta\ell^{m} \|A\Pi_{\mathcal{C}}J^{-1}(Jw_{n} - \delta\ell^{m}Aw_{n}) - Aw_{n}\| \cdot \|\Pi_{\mathcal{C}}J^{-1}(Jw_{n} - \delta\ell^{m}Aw_{n}) - \Pi_{T_{n}}J^{-1}(Jw_{n} - \delta\ell^{m}Ay_{n})\| \\ > \eta \|w_{n} - \Pi_{\mathcal{C}}J^{-1}(Jw_{n} - \delta\ell^{m}Aw_{n})\| \cdot \|\Pi_{\mathcal{C}}J^{-1}(Jw_{n} - \delta\ell^{m}Aw_{n}) - \Pi_{T_{n}}J^{-1}(Jw_{n} - \delta\ell^{m}Ay_{n})\|.$$

Thus, we obtain

$$\left\|A\Pi_{\mathcal{C}}J^{-1}(Jw_n - \delta\ell^m Aw_n) - Aw_n\right\| > \frac{\eta}{\delta\ell^m} \left\|w_n - \Pi_{\mathcal{C}}J^{-1}(Jw_n - \delta\ell^m Aw_n)\right\|.$$
(6.2.3)

Next, we consider the following cases: when $w_n \in \mathcal{C}$ and $w_n \notin \mathcal{C}$.

First, suppose that $w_n \in \mathcal{C}$. By the continuity of A and $\Pi_{\mathcal{C}}$ we have

$$\lim_{m \to \infty} \left\| w_n - \Pi_{\mathcal{C}} J^{-1} (Jw_n - \delta \ell^m A w_n) \right\| = 0$$

Consequently, by the continuity of A we get

$$\lim_{m \to \infty} \left\| Aw_n - A\Pi_{\mathcal{C}} J^{-1} (Jw_n - \delta \ell^m Aw_n) \right\| = 0.$$
(6.2.4)

From (6.2.3) and (6.2.4), we obtain

$$\lim_{m \to \infty} \frac{\|w_n - \Pi_{\mathcal{C}} J^{-1} (Jw_n - \delta \ell^m A w_n)\|}{\delta \ell^m} = 0.$$

By the continuity of J, we have

$$\lim_{m \to \infty} \frac{\|Jw_n - J\Pi_{\mathcal{C}} J^{-1} (Jw_n - \delta \ell^m A w_n)\|}{\delta \ell^m} = 0.$$
 (6.2.5)

Let $q_m = \prod_{\mathcal{C}} J^{-1} (Jw_n - \delta \ell^m A w_n)$. Then from (2.4.2), we have

$$\langle Jq_m - Jw_n + \delta \ell^m Aw_n, \ y - q_m \rangle \ge 0, \quad \forall y \in \mathcal{C},$$

which implies that

$$\langle Aw_n, y - q_m \rangle \ge \frac{\langle Jw_n - Jq_m, y - q_m \rangle}{\delta \ell^m}.$$

Letting $m \to \infty$ and using (6.2.5), we have

$$\langle Aw_n, y - w_n \rangle \ge 0, \quad \forall y \in \mathcal{C}.$$

This implies that $w_n \in VI(\mathcal{C}, A)$, which contradicts the hypothesis.

Next suppose $w_n \notin \mathcal{C}$, then we obtain

$$\lim_{m \to \infty} \left\| w_n - \Pi_{\mathcal{C}} J^{-1} (J w_n - \delta \ell^m A w_n) \right\| = \| w_n - \Pi_{\mathcal{C}} w_n \| > 0$$
(6.2.6)

and

$$\lim_{m \to \infty} \delta \ell^m \left\| A w_n - A \Pi_{\mathcal{C}} J^{-1} (J w_n - \delta \ell^m A w_n) \right\| = 0.$$
(6.2.7)

Combining (6.2.3), (6.2.6) and (6.2.7), we obtain a contradiction. Hence, the linesearch (6.2.2) is well defined.

Lemma 6.2.3. Let $\{x_n\}$ be a sequence generated by Algorithm 6.2.2 under Assumption 6.2.1 and $p \in VI(\mathcal{C}, A)$. Then, we have

$$\lim_{n \to \infty} \frac{\theta_n}{\alpha_n} \Big(\phi(p, x_{n-1}) - \phi(p, x_n) \Big) = 0.$$
(6.2.8)

Proof. Let $p \in VI(\mathcal{C}, A)$. From (6.2.1), we obtain

$$\theta_n \|x_n - x_{n-1}\| \le \tau_n \quad \text{for each} \quad n \ge 1.$$
(6.2.9)

Since $\lim_{n\to\infty} \frac{\tau_n}{\alpha_n} = 0$ and $\lim_{n\to\infty} \alpha_n = 0$, it follows that $\lim_{n\to\infty} \tau_n = 0$. Thus, we obtain

$$\lim_{n \to \infty} \theta_n \|x_n - x_{n-1}\| \le \lim_{n \to \infty} \tau_n = 0.$$
 (6.2.10)

Since J is norm-to-norm continuous on subsets of \mathcal{X} , it follows that

$$\lim_{n \to \infty} \theta_n \|Jx_n - Jx_{n-1}\| = 0.$$
 (6.2.11)

Again, from (6.2.9) we have

$$\lim_{n \to \infty} \frac{\theta_n}{\alpha_n} \|x_n - x_{n-1}\| \le \lim_{n \to \infty} \frac{\tau_n}{\alpha_n} = 0.$$
(6.2.12)

Since J is norm-to-norm continuous on \mathcal{X} , we have

$$\lim_{n \to \infty} \frac{\theta_n}{\alpha_n} \|Jx_n - Jx_{n-1}\| = 0.$$
 (6.2.13)

Next, observe that

$$\phi(p, x_{n-1}) - \phi(p, x_n) = \|p\|^2 - 2\langle p, Jx_{n-1} \rangle + \|x_{n-1}\|^2 - (\|p\|^2 - 2\langle p, Jx_n \rangle + \|x_n\|^2)$$

= $\|x_{n-1}\|^2 - \|x_n\|^2 + 2\langle p, Jx_n - Jx_{n-1} \rangle$
 $\leq \|x_{n-1} - x_n\| (\|x_{n-1}\| + \|x_n\|) + 2\|p\| \|Jx_n - Jx_{n-1}\|.$ (6.2.14)

Now, by applying (6.2.12) and (6.2.13), it follows from (6.2.14) that

$$\lim_{n \to \infty} \frac{\theta_n}{\alpha_n} \Big(\phi(p, x_{n-1}) - \phi(p, x_n) \Big) \le \lim_{n \to \infty} \Big(\frac{\theta_n}{\alpha_n} \|x_{n-1} - x_n\| \big(\|x_{n-1}\| + \|x_n\| \big) \Big) \\ + \lim_{n \to \infty} \Big(2 \|p\| \frac{\theta_n}{\alpha_n} \|Jx_n - Jx_{n-1}\| \Big) \\ = 0,$$

which is the desired result.

Lemma 6.2.4. Let $\{x_n\}$ be a sequence generated by Algorithm 6.2.2 under Assumption 6.2.1. Then,

$$\phi(p, z_n) \le \phi(p, w_n) - (1 - \mu\eta) \Big[\phi(y_n, w_n) + \phi(z_n, y_n) \Big].$$
 (6.2.15)

Proof. Let $p \in VI(\mathcal{C}, A)$. and $Jd_n := Jw_n - \lambda_n Ay_n$, by (2.4.3) we have

$$\begin{split} \phi(p, \ z_n) &\leq \phi(p, \ d_n) - \phi(z_n, \ d_n) \\ &= \|p\|^2 - 2\langle p, \ Jw_n - \lambda_n Ay_n \rangle + \|d_n\|^2 - \left(\|z_n\|^2 - 2\langle z_n, \ Jw_n - \lambda_n Ay_n \rangle + \|d_n\|^2\right) \\ &= \|p\|^2 + \|d_n\|^2 - \|z_n\|^2 - \|d_n\|^2 - 2\langle p, \ Jw_n - \lambda_n Ay_n \rangle + 2\langle z_n, \ Jw_n - \lambda_n Ay_n \rangle \\ &= \|p\|^2 - \|z_n\|^2 - 2\langle p, \ Jw_n - \lambda_n Ay_n \rangle + 2\langle z_n, \ Jw_n - \lambda_n Ay_n \rangle \\ &= \|p\|^2 - \|z_n\|^2 - 2\langle p, \ Jw_n \rangle + 2\langle p, \ \lambda_n Ay_n \rangle + 2\langle z_n, \ Jw_n \rangle - 2\langle z_n, \ \lambda_n Ay_n \rangle \\ &= \|p\|^2 - \|z_n\|^2 + \|w_n\|^2 - \|w_n\|^2 - 2\langle p, \ Jw_n \rangle + 2\langle p, \ \lambda_n Ay_n \rangle + 2\langle z_n, \ Jw_n \rangle + 2\langle z_n, \ Jw_n \rangle \\ &= 2\langle z_n, \ \lambda_n Ay_n \rangle \\ &= (\|p\|^2 - 2\langle p, \ Jw_n \rangle + \|w_n\|^2) - (\|w_n\|^2 - 2\langle z_n, \ Jw_n \rangle + \|z_n\|^2) - 2\langle z_n, \ \lambda_n Ay_n \rangle \\ &+ 2\langle p, \ \lambda_n Ay_n \rangle \\ &= \phi(p, \ w_n) - \phi(z_n, w_n) - 2\langle z_n, \ \lambda_n Ay_n \rangle + 2\langle p, \ \lambda_n Ay_n \rangle \\ &= \phi(p, \ w_n) - \phi(z_n, w_n) + 2\lambda_n \langle p - y_n, \ Ay_n \rangle + 2\lambda_n \langle y_n - z_n, \ Ay_n \rangle. \end{split}$$
(6.2.16)

Since $p \in VI(\mathcal{C}, A)$ and $y_n \in \mathcal{C}$, we have

$$\langle Ap, y_n - p \rangle \ge 0.$$

Also, since A is pseudomonotone, it follows that

$$\langle Ay_n, y_n - p \rangle \ge 0.$$

Applying the last inequality in (6.2.16), we have

$$\phi(p, z_n) \le \phi(p, w_n) - \phi(z_n, w_n) + 2\lambda_n \langle y_n - z_n, Ay_n \rangle.$$
(6.2.17)

Next, by applying Lemma 2.2.13, we get

$$\phi(z_n, w_n) - 2\lambda_n \langle y_n - z_n, Ay_n \rangle = \phi(z_n, y_n) + \phi(y_n, w_n) + 2\langle z_n - y_n, Jy_n - Jw_n \rangle$$

$$- 2\lambda_n \langle y_n - z_n, Ay_n \rangle$$

$$= \phi(z_n, y_n) + \phi(y_n, w_n) - 2\langle z_n - y_n, Jw_n - Jy_n - \lambda_n Ay_n \rangle$$

$$= \phi(z_n, y_n) + \phi(y_n, w_n) - 2\langle z_n - y_n, Jw_n - \lambda_n Aw_n - Jy_n \rangle$$

(6.2.18)

$$- 2\lambda_n \langle y_n - z_n, Ay_n - Aw_n \rangle.$$

Now, since $z_n \in T_n$ we obtain from the definition of T_n that

$$\langle z_n - y_n, Jw_n - \lambda_n Aw_n - Jy_n \rangle \le 0.$$

Applying the last inequality in (6.2.18), we have

$$\phi(z_n, w_n) - 2\lambda_n \langle y_n - z_n, Ay_n \rangle \ge \phi(y_n, w_n) + \phi(z_n, y_n) - 2\lambda_n \langle y_n - z_n, Ay_n - Aw_n \rangle \ge \phi(y_n, w_n) + \phi(z_n, y_n) - \eta \left[\|w_n - y_n\|^2 + \|y_n - z_n\|^2 \right].$$
(6.2.19)

Applying (6.2.19) in (6.2.17) and using Lemma 2.5.22, we obtain

$$\begin{aligned} \phi(p, \ z_n) &\leq \phi(p, \ w_n) - \phi(y_n, \ w_n) - \phi(z_n, \ y_n) + \eta \left[\|w_n - y_n\|^2 + \|y_n - z_n\|^2 \right] \\ &= \phi(p, \ w_n) - \phi(y_n, \ w_n) - \phi(z_n, \ y_n) + \eta \left[\|y_n - w_n\|^2 + \|z_n - y_n\|^2 \right] \\ &\leq \phi(p, \ w_n) - \phi(y_n, \ w_n) - \phi(z_n, \ y_n) + \eta \left[\mu \phi(y_n, \ w_n) + \mu \phi(z_n, \ y_n) \right] \\ &= \phi(p, w_n) - (1 - \mu \eta) \left[\phi(y_n, w_n) + \phi(z_n, \ y_n) \right], \end{aligned}$$

which is the desired result.

Lemma 6.2.5. Let $\{x_n\}$ be a sequence generated by Algorithm 6.2.2 under Assumption 6.2.1. Then, $\{x_n\}$ is bounded.

Proof. Let $p \in VI(\mathcal{C}, A)$. Applying the conditions on η , we obtain from Lemma 6.2.4 that

$$\phi(p, z_n) \le \phi(p, w_n). \tag{6.2.20}$$

From the definition of w_n , we have

$$\phi(p, w_n) = \phi(p, J^{-1}((1 - \theta_n)Jx_n + \theta_n Jx_{n-1}))$$

$$\leq (1 - \theta_n)\phi(p, x_n) + \theta_n\phi(p, x_{n-1}).$$
(6.2.21)

By the definition of x_{n+1} , and by applying (6.2.20) and (6.2.21), we have

$$\begin{aligned} \phi(p, \ x_{n+1}) &= \phi(p, \ J^{-1}(\alpha_n Ju + (1 - \alpha_n) Jz_n)) \\ &\leq \alpha_n \phi(p, \ u) + (1 - \alpha_n) \phi(p, \ z_n) \\ &\leq \alpha_n \phi(p, \ u) + (1 - \alpha_n) \phi(p, \ w_n) \\ &\leq \alpha_n \phi(p, \ u) + (1 - \alpha_n) \Big[(1 - \theta_n) \phi(p, \ x_n) + \theta_n \phi(p, x_{n-1}) \Big] \\ &\leq \max \Big\{ \phi(p, \ u), \ \max \{ \phi(p, \ x_n), \ \phi(p, \ x_{n-1}) \} \Big\} \\ &\vdots \\ &\leq \max \Big\{ \phi(p, \ u), \ \max \{ \phi(p, \ x_1), \ \phi(p, \ x_0) \} \Big\}. \end{aligned}$$

Hence, $\phi(p, x_n)$ is bounded, and it follows that $\{x_n\}$ is bounded. Consequently, $\{w_n\}$, $\{y_n\}$, $\{z_n\}$ are bounded.

Lemma 6.2.6. Let $\{x_n\}$ be a sequence generated by Algorithm 6.2.2 under Assumption 6.2.1. Then, the following inequality holds for all $p \in VI(\mathcal{C}, A)$:

$$\begin{split} \phi(p, \ x_{n+1}) &\leq (1 - \alpha_n)\phi(p, \ x_n) \\ &+ \alpha_n \left[(1 - \alpha_n) \frac{\theta_n}{\alpha_n} \left[\phi(p, x_{n-1}) - \phi(p, x_n) \right] + 2\langle Ju - Jp, \ x_{n+1} - p \rangle \right] \\ &- (1 - \alpha_n)(1 - \mu\eta) \left[\phi(y_n, w_n) + \phi(z_n, y_n) \right] \end{split}$$

Proof. Let $p \in VI(\mathcal{C}, A)$. From the definition of x_{n+1} and by applying (2.2.2), (6.2.15) and (6.2.21), we have

$$\begin{split} \phi(p, \ x_{n+1}) &= \phi \Big(p, J^{-1} \Big(\alpha_n J u + (1 - \alpha_n) J z_n \Big) \Big) \\ &= V(p, \ \alpha_n J u + (1 - \alpha_n) J z_n - \alpha_n (J u - J p) \Big) + 2\alpha_n \langle J u - J p, \ x_{n+1} - p \rangle \\ &\leq \alpha_n V(p, \ J p) + (1 - \alpha_n) V(p, \ J z_n) + 2\alpha_n \langle J u - J p, \ x_{n+1} - p \rangle \\ &= \alpha_n (p, \ p) + (1 - \alpha_n) \phi(p, \ z_n) + 2\alpha_n \langle J u - J p, \ x_{n+1} - p \rangle \\ &\leq (1 - \alpha_n) \Big[\phi(p, \ w_n) - (1 - \mu \eta) \Big[\phi(y_n, w_n) + \phi(z_n, y_n) \Big] \Big] \\ &+ 2\alpha_n \langle J u - J p, \ x_{n+1} - p \rangle \\ &\leq (1 - \alpha_n) \left[(1 - \theta_n) \phi(p, \ x_n) + \theta_n \phi(p, \ x_{n-1}) \right] + 2\alpha_n \langle J u - J p, \ x_{n+1} - p \rangle \\ &- (1 - \alpha_n) (1 - \mu \eta) \Big[\phi(y_n, w_n) + \phi(z_n, y_n) \Big] \\ &= (1 - \alpha_n) \phi(p, \ x_n) \\ &+ \alpha_n \left[(1 - \alpha_n) \frac{\theta_n}{\alpha_n} \Big[\phi(p, x_{n-1}) - \phi(p, x_n) \Big] + 2 \langle J u - J p, \ x_{n+1} - p \rangle \right] \\ &- (1 - \alpha_n) (1 - \mu \eta) \Big[\phi(y_n, w_n) + \phi(z_n, y_n) \Big], \end{split}$$

which is the required inequality.

Lemma 6.2.7. Let $\{w_n\}$ and $\{y_n\}$ be sequences generated by Algorithm 6.2.2 under Assumption 6.2.1. If there exists a subsequence $\{w_{n_k}\}$ of $\{w_n\}$ such that $\{w_{n_k}\}$ converges weakly to $p \in C$ and $\lim_{k \to \infty} ||w_{n_k} - y_{n_k}|| = 0$, then $p \in VI(C, A)$.

Proof. For all $x \in \mathcal{C}$, we obtain from (2.4.2) and the definition of $\{y_{n_k}\}$ that

$$0 \leq \langle Jy_{n_k} - Jw_{n_k} + \lambda_{n_k}Aw_{n_k}, x - y_{n_k} \rangle$$

= $\langle Jy_{n_k} - Jw_{n_k}, x - y_{n_k} \rangle + \lambda_{n_k} \langle Aw_{n_k}, x - y_{n_k} \rangle$
= $\langle Jy_{n_k} - Jw_{n_k}, x - y_{n_k} \rangle + \lambda_{n_k} \langle Aw_{n_k}, w_{n_k} - y_{n_k} \rangle + \lambda_{n_k} \langle Aw_{n_k}, x - w_{n_k} \rangle$ (6.2.23)

Next, we show that $\liminf_{k\to\infty} \langle Aw_{n_k}, x - w_{n_k} \rangle \ge 0$, $\forall x \in \mathcal{C}$. We consider the following two cases for λ_{n_k} :

Case 1: Suppose that $\liminf_{k\to\infty} \lambda_{n_k} > 0$. Since $\{w_{n_k}\}$ is bounded and A is uniformly continuous, we obtain from Lemma 2.5.21 that $\{Aw_{n_k}\}$ is bounded. Since $\lim_{k\to\infty} ||w_{n_k} - y_{n_k}|| = 0$, $\{y_{n_k}\}$ is bounded. From the boundedness of $\{y_{n_k}\}$, (6.2.23), the facts that J is norm-to-norm uniformly continuous and $\liminf_{k\to\infty} \lambda_{n_k} > 0$, we have

$$\liminf_{k \to \infty} \langle Aw_{n_k}, \ x - w_{n_k} \rangle \ge 0, \quad \forall \ x \in \mathcal{C}.$$
(6.2.24)

Case 2: Suppose that $\liminf_{k\to\infty} \lambda_{n_k} = 0$. Let $t_{n_k} = \prod_{\mathcal{C}} J^{-1} (Jw_{n_k} - \lambda_{n_k} \ell^{-1} A w_{n_k})$. From Lemma 2.5.23 and the fact that $\lambda_{n_k} \ell^{-1} > \lambda_{n_k}$ we have,

$$\ell ||w_{n_k} - t_{n_k}|| \le ||w_{n_k} - y_{n_k}|| \to 0$$
, as $k \to \infty$.

Thus, we have that $t_{n_k} \rightharpoonup p \in \mathcal{C}$, which implies that $\{t_{n_k}\}$ is bounded. Since $\lim_{k\to\infty} ||w_{n_k} - t_{n_k}|| = 0$ and A is uniformly continuous, we obtain

$$\lim_{k \to \infty} \|Aw_{n_k} - At_{n_k}\| = 0.$$
(6.2.25)

Also, since J is norm-to-norm continuous we have

$$\|Jw_{n_k} - Jt_{n_k}\| \to 0, \ k \to \infty.$$

From (6.2.2) we have,

$$\lambda_{n_{k}}\ell^{-1}\langle A\Pi_{\mathcal{C}}J^{-1}(Jw_{n_{k}}-\lambda_{n_{k}}\ell^{-1}Aw_{n_{k}})-Aw_{n_{k}}, \quad \Pi_{\mathcal{C}}J^{-1}(Jw_{n_{k}}-\lambda_{n_{k}}\ell^{-1}Aw_{n_{k}})-z_{n_{k}}\rangle \\ > \frac{\eta}{2}\Big[\|w_{n_{k}}-\Pi_{\mathcal{C}}J^{-1}(Jw_{n_{k}}-\lambda_{n_{k}}\ell^{-1}Aw_{n_{k}})\|^{2}+\|\Pi_{\mathcal{C}}J^{-1}(Jw_{n_{k}}-\lambda_{n_{k}}\ell^{-1}Aw_{n_{k}})-z_{n_{k}}\|^{2}\Big].$$

Applying the Cauchy-Schwartz inequality to the last inequality, we have

$$\lambda_{n_{k}}\ell^{-1} \|A\Pi_{\mathcal{C}}J^{-1}(Jw_{n_{k}}-\lambda_{n_{k}}\ell^{-1}Aw_{n_{k}}) - Aw_{n_{k}}\| \cdot \|\Pi_{\mathcal{C}}J^{-1}(Jw_{n_{k}}-\lambda_{n_{k}}\ell^{-1}Aw_{n_{k}}) - z_{n_{k}}\| \\> \eta \Big[\|w_{n_{k}}-\Pi_{\mathcal{C}}J^{-1}(Jw_{n_{k}}-\lambda_{n_{k}}\ell^{-1}Aw_{n_{k}})\| \cdot \|\Pi_{\mathcal{C}}J^{-1}(Jw_{n_{k}}-\lambda_{n_{k}}\ell^{-1}Aw_{n_{k}}) - z_{n_{k}}\| \Big],$$

which implies that

$$\lambda_{n_k}\ell^{-1} \|Aw_{n_k} - A\Pi_{\mathcal{C}}J^{-1}(Jw_{n_k} - \lambda_{n_k}\ell^{-1}Aw_{n_k})\| > \eta \|w_{n_k} - \Pi_{\mathcal{C}}J^{-1}(Jw_{n_k} - \lambda_{n_k}\ell^{-1}Aw_{n_k})\|$$

From this, we have

$$\frac{1}{\eta} \|Aw_{n_k} - At_{n_k}\| > \frac{1}{\lambda_{n_k}\ell^{-1}} \|w_{n_k} - t_{n_k}\|.$$
(6.2.26)

Applying (6.2.25) to (6.2.26) we have

$$\lim_{k \to \infty} \frac{1}{\lambda_{n_k} \ell^{-1}} \| w_{n_k} - t_{n_k} \| = 0.$$

From the definition of t_{n_k} and Lemma 2.4.3, we have

$$0 \le \left\langle Jt_{n_k} - Jw_{n_k} + \lambda_{n_k} \ell^{-1} A w_{n_k}, \ x - t_{n_k} \right\rangle.$$

It follows that

$$\frac{1}{\lambda_{n_k}\ell^{-1}} \left\langle Jw_{n_k} - Jt_{n_k}, \ x - t_{n_k} \right\rangle + \left\langle Aw_{n_k}, \ t_{n_k} - w_{n_k} \right\rangle \le \left\langle Aw_{n_k}, \ x - w_{n_k} \right\rangle, \quad \forall \ x \in \mathcal{C}.$$

Taking the limit of the last inequality as $k \to \infty$ and noting the fact that J is norm-to-norm continuous, we have

$$\liminf_{k \to \infty} \langle Aw_{n_k}, \ x - w_{n_k} \rangle \ge 0, \quad \forall x \in \mathcal{C},$$

which is the desired result.

Now, to complete the proof we show that $p \in VI(\mathcal{C}, A)$. Observe that

$$\langle Ay_{n_k}, x - y_{n_k} \rangle = \langle Ay_{n_k} - Aw_{n_k}, x - w_{n_k} \rangle + \langle Aw_{n_k}, x - w_{n_k} \rangle + \langle Ay_{n_k}, w_{n_k} - y_{n_k} \rangle.$$
(6.2.27)

Since $\lim_{k\to\infty} ||w_{n_k} - y_{n_k}|| = 0$ and A is uniformly continuous, we obtain

$$\lim_{k \to \infty} \|Aw_{n_k} - Ay_{n_k}\| = 0.$$
(6.2.28)

Then, it follows from (6.2.24) and (6.2.27) that

$$\liminf_{k \to \infty} \langle Ay_{n_k}, \ x - y_{n_k} \rangle \ge 0, \quad \forall \ x \in \mathcal{C}.$$

Now, we show that $p \in VI(\mathcal{C}, A)$. We choose a decreasing sequence of positive numbers $\{\zeta_k\}$ such that $\{\zeta_k\} \to 0$ as $k \to \infty$. For each k, we denote the smallest positive integer by N_k such that

$$\langle Ay_{n_j}, \ x - y_{n_j} \rangle + \zeta_n \ge 0, \ \forall \ j \ge N_k.$$
(6.2.29)

We observe that $\{N_k\}$ is increasing since $\{\zeta_n\}$ is decreasing. However, since $\{y_{n_k}\} \subset C$ for each k, we assume that $Ay_{N_k} \neq 0$ (otherwise, y_{N_k} is a solution). Moreover, let $d_{N_k} \in E$ such that $\lim_{k\to\infty} d_{N_k} = d \in E$ and $\langle Ay_{N_k}, d_{N_k} \rangle = 1$ for each k. Then, from (6.2.29), we have for each k that

 $\langle Ay_{N_k}, x + \zeta_k d_{N_k} - y_{N_k} \rangle \ge 0, \quad \forall x \in \mathcal{C}.$

Since A is pseudomonotone on \mathcal{X} , we have

 $\langle A(x+\zeta_k d_{N_k}), x+\zeta_k d_{N_k}-y_{N_k}\rangle \ge 0, \quad \forall x \in \mathcal{C}.$

which implies that

$$\langle Ax, \ x - y_{N_k} \rangle \ge \langle Ax - A(x + \zeta_k d_{N_k}), \ x + \zeta_k d_{N_k} - y_{N_k} \rangle - \zeta_k \langle Ax, \ d_{N_k} \rangle, \quad \forall \ x \in \mathcal{C}.$$
(6.2.30)

Since $\{d_{N_k}\}$ is bounded and $\lim_{k\to\infty} \zeta_k = 0$, we have $\lim_{k\to\infty} \zeta_k d_{N_k} = 0$. Applying the fact that A is uniformly continuous, $\lim_{k\to\infty} \zeta_k d_{N_k} = 0$, $\{y_{N_k}\}$ and $\{d_{N_k}\}$ are bounded on (6.2.30), we have

$$\liminf_{k \to \infty} \langle Ax, \ x - y_{N_k} \rangle \ge 0.$$

Therefore,

$$\langle Ax, x-p \rangle = \lim_{k \to \infty} \langle Ax, x-y_{N_k} \rangle = \liminf_{k \to \infty} \langle Ax, x-y_{N_k} \rangle \ge 0, \quad \forall x \in \mathcal{C}$$

From Lemma 2.5.2, we obtain that $p \in VI(\mathcal{C}, A)$. This completes the proof.

Remark 6.2.8. Note that Lemma 6.2.7 holds if the mapping A in Assumption 6.2.1 (b) is Lipschitz continuous instead of uniformly continuous. Moreover, the result of Lemma 6.2.7 also holds if the step-size of our proposed Algorithm 6.2.2 is a sequence of positive numbers.

We are now in the position to present our main result.

Theorem 6.2.9. Let $\{x_n\}$ be a sequence generated by Algorithm 6.2.2 under Assumption 6.2.1. Then, $\{x_n\}$ converges strongly to $p \in VI(\mathcal{C}, A)$, where $p = \prod_{VI(\mathcal{C}, A)}(u)$.

Proof. Let $p = \prod_{VI(\mathcal{C},A)} u$. From Lemma 6.2.6, we have

$$\phi(p, x_{n+1}) \leq (1 - \alpha_n)\phi(p, x_n) + \alpha_n \left[(1 - \alpha_n)\frac{\theta_n}{\alpha_n} [\phi(p, x_{n-1}) - \phi(p, x_n)] + 2\langle Ju - Jp, x_{n+1} - p \rangle \right] = (1 - \alpha_n)\phi(p, x_n) + \alpha_n v_n,$$
(6.2.31)

where $v_n = (1 - \alpha_n) \frac{\theta_n}{\alpha_n} [\phi(p, x_{n-1}) - \phi(p, x_n)] + 2\langle Ju - Jp, x_{n+1} - p \rangle.$

Now, we claim that $\{\phi(p, x_n)\}$ converges to zero. To establish this, by Lemma 2.5.36 it suffices to show that $\limsup_{k\to\infty} v_{n_k} \leq 0$ for every subsequence $\{\phi(p, x_{n_k})\}$ of $\{\phi(p, x_n)\}$ satisfying the condition

$$\liminf_{k \to \infty} \left[\phi(p, x_{n_{k+1}}) - \phi(p, x_{n_k}) \right] \ge 0.$$
(6.2.32)

Suppose that $\{\phi(p, x_{n_k})\}$ is a subsequence of $\{\phi(p, x_n)\}$ such that (6.2.32) holds. Again, we obtain from Lemma 6.2.6 that

$$(1 - \alpha_{n_k})(1 - \mu\eta) \left[\phi(y_{n_k}, w_{n_k}) + \phi(z_{n_k}, y_{n_k}) \right] \le (1 - \alpha_{n_k}) \phi(p, x_{n_k}) - \phi(p, x_{n_k+1}) + \alpha_{n_k} \left[(1 - \alpha_{n_k}) \frac{\theta_{n_k}}{\alpha_{n_k}} \left[\phi(p, x_{n_k-1}) - \phi(p, x_{n_k}) \right] + 2 \langle Ju - Jp, x_{n_k+1} - p \rangle \right].$$

By (6.2.8), (6.2.32) and the fact that $\lim_{k\to\infty} \alpha_{n_k} = 0$, we obtain

$$\lim_{k \to \infty} (1 - \alpha_{n_k}) (1 - \mu \eta) \big[\phi(y_{n_k}, w_{n_k}) + \phi(z_{n_k}, y_{n_k}) \big] = 0.$$

By the conditions on the control sequences, it follows that

$$\lim_{k \to \infty} \phi(y_{n_k}, w_{n_k}) = \lim_{k \to \infty} \phi(z_{n_k}, y_{n_k}) = 0.$$

From Lemma 2.5.24, we have

$$\|w_{n_k} - y_{n_k}\| \to 0, \text{ as } k \to \infty \tag{6.2.33}$$

and

$$||y_{n_k} - z_{n_k}|| \to 0$$
, as $k \to \infty$. (6.2.34)

From (6.2.33) and (6.2.34) we have

$$||w_{n_k} - z_{n_k}|| \to 0$$
, as $k \to \infty$. (6.2.35)

From the definition of $x_{n_{k+1}}$ in Algorithm 6.2.2, we have

$$||Jx_{n_{k+1}} - Jz_{n_k}|| \le \alpha_{n_k} ||Ju - Jz_{n_k}|| + (1 - \alpha_{n_k}) ||Jz_{n_k} - Jz_{n_k}|| \to 0, \text{ as } k \to \infty.$$
(6.2.36)

Since J^{-1} is norm-to-norm uniformly continuous on bounded subsets of \mathcal{X}^* , we obtain

$$||x_{n_{k+1}} - z_{n_k}|| \to 0, \text{ as } k \to \infty.$$
 (6.2.37)

Also, by applying (6.2.34) and (6.2.37) we obtain

$$\|x_{n_{k+1}} - y_{n_k}\| \le \|x_{n_{k+1}} - z_{n_k}\| + \|z_{n_k} - y_{n_k}\| \to 0, \text{ as } k \to \infty.$$
(6.2.38)

From the definition of w_{n_k} and by applying (6.2.11), we have

$$||Jw_{n_k} - Jx_{n_k}|| = \theta_{n_k} ||Jx_{n_{k-1}} - Jx_{n_k}|| \to 0$$
, as $k \to \infty$.

Since J^{-1} is norm-to-norm uniformly continuous on bounded subsets of \mathcal{X}^* , we have

$$||w_{n_k} - x_{n_k}|| \to 0$$
, as $k \to \infty$. (6.2.39)

From (6.2.33)-(6.2.39), we have

$$\lim_{k \to \infty} \|y_{n_k} - x_{n_k}\| = 0, \quad \lim_{k \to \infty} \|x_{n_k} - z_{n_k}\| = 0, \quad \lim_{k \to \infty} \|w_{n_k} - y_{n_k}\| = 0.$$
(6.2.40)

Also, from (6.2.38) and (6.2.40), we have

$$\|x_{n_{k+1}} - x_{n_k}\| \le \|x_{n_{k+1}} - y_{n_k}\| + \|y_{n_k} - x_{n_k}\| \to 0, \text{ as } k \to \infty.$$
(6.2.41)

Since $\{x_n\}$ is bounded, then $w_{\omega}(x_n)$ is nonempty. Let $x^* \in w_{\omega}(x_n)$ be an arbitrary element. Then, there exists a subsequence $\{x_{n_k}\}$ of $\{x_n\}$ such that $x_{n_k} \rightharpoonup x^*$ as $k \rightarrow \infty$. Then, from (6.2.39), we obtain $w_{n_k} \rightharpoonup x^*$ as $k \rightarrow \infty$. Now, by Lemma 6.2.7, it follows from (6.2.40) that $x^* \in VI(\mathcal{C}, A)$. Since $x^* \in w_{\omega}(x_n)$ is an arbitrary element, we have $w_{\omega}(x_n) \subset VI(\mathcal{C}, A)$. Moreover, since the sequence $\{x_{n_k}\}$ is bounded, there exists a subsequence $\{x_{n_{k_j}}\}$ of $\{x_{n_k}\}$ which converges weakly to $z^* \in \mathcal{X}$. Since $p = \prod_{VI(\mathcal{C},A)}(u)$, we have that

$$\lim_{k \to \infty} \sup \langle Ju - Jp, x_{n_k} - p \rangle = \lim_{j \to \infty} \langle Ju - Jp, x_{n_{k_j}} - p \rangle$$
$$= \langle Ju - Jp, z^* - p \rangle$$
$$\leq 0. \tag{6.2.42}$$

From (6.2.41) and (6.2.42), we obtain

$$\limsup_{k \to \infty} \langle Ju - Jp, \ x_{n_{k+1}} - p \rangle \le 0.$$
(6.2.43)

From (6.2.8) and (6.2.43), we obtain $\limsup_{k\to\infty} v_{n_k} \leq 0$. Hence, by applying Lemma 2.5.36 to (6.2.31) we obtain that $\lim_{n\to\infty} \phi(p, x_n) = 0$. It follows from Lemma 2.5.24 that $\lim_{n\to\infty} ||x_n - p|| = 0$. Therefore, $\{x_n\}$ converges strongly to $p \in VI(\mathcal{C}, A)$, where $p = \prod_{VI(\mathcal{C}, A)} (u)$. \Box

Remark 6.2.10. In real Hilbert spaces, we have that $\phi(x, y) = ||x - y||^2$, $\forall x, y \in \mathcal{C}$ and $\Pi_{\mathcal{C}} = P_{\mathcal{C}}$ where $P_{\mathcal{C}} : \mathcal{H} \to \mathcal{C}$ is the metric projection of \mathcal{H} onto \mathcal{C} .

A mapping $P_{\mathcal{C}}$ is said to be the metric projection of a real Hilbert space \mathcal{H} onto \mathcal{C} if for all $x \in \mathcal{H}$ there exists a unique nearest point in \mathcal{C} , denoted by $P_{\mathcal{C}}x$, such that

$$||x - P_{\mathcal{C}}x|| \le ||x - y||, \quad \forall \ y \in \mathcal{C}.$$

If \mathcal{X} is a Hilbert space, then taking J = I and $\phi(x, y) = ||x - y||^2$, $\forall x, y \in \mathcal{C}$ in Theorem 6.2.9 (where I is the identity mapping on \mathcal{X}), we have the following consequent result:

Corollary 6.2.3. Let $\{x_n\}$ be a sequence generated by Algorithm 6.2.4 below under Assumption 6.2.1. Then, $\{x_n\}$ converges strongly to $p \in VI(\mathcal{C}, A)$, where $p = P_{VI(\mathcal{C}, A)}(u)$.

Algorithm 6.2.4. Inertial method with new Armijo-type stepsize.

Initialization: Given $\lambda_1 > 0$, $\theta > 0$, $\delta > 0, \ell \in (0, 1)$ and $\eta \in (0, \frac{1}{\mu})$. Let $u, x_0, x_1 \in \mathcal{X}$ be arbitrary. Set n := 1.

Step 1: Given the iterates x_{n-1} and x_n for each $n \ge 1$, choose θ_n such that $0 \le \theta_n \le \theta_n$, where

$$\bar{\theta}_n := \begin{cases} \min\left\{\theta, \frac{\tau_n}{\|x_n - x_{n-1}\|}\right\}, & \text{if } x_n \neq x_{n-1}\\ \theta, & \text{otherwise.} \end{cases}$$

Step 2: Compute

$$w_n = x_n + \theta_n (x_{n-1} - x_n))$$

Step 3: Compute

$$y_n = P_{\mathcal{C}}(w_n - \lambda_n A w_n).$$

If $w_n = y_n$, or $Ay_n = 0$ then stop and y_n is a solution of the VIP. Otherwise, go to **Step** 4.

Step 4: Compute

$$z_n = P_{T_n}(w_n - \lambda_n A y_n)$$

where

$$T_n := \{ x \in \mathcal{X} \mid \langle w_n - \lambda_n A w_n - y_n, x - y_n \rangle \le 0 \},\$$

 $\lambda_n := \delta \ell^{m_n}$ and m_n is the smallest non-negative integer m satisfying

$$\delta \ell^m \langle Ay_n - Aw_n, \quad y_n - z_n \rangle \le \frac{\eta}{2} \Big[\|w_n - y_n\|^2 + \|y_n - z_n\|^2 \Big].$$

Step 5: Calculate

$$x_{n+1} = \alpha_n u + (1 - \alpha_n) z_n)$$

Set n := n + 1 and go to **Step 1**.

6.2.3 Numerical experiments

In this section, using some test examples, we discuss the numerical behavior of Algorithm 6.2.2 and compare it with Algorithm 2.5.6 proposed by Cai et. al [53], Algorithm 2.5.7 proposed by Tan et. al [240], Appendix 6.2.13 proposed by Thong et.al [251] and Appendix 6.2.14 proposed by Reich et.al [214]. We perform all implementations using Matlab 2016 (b), installed on a personal computer with Intel(R) Core(TM) i5-2600 CPU@2.30GHz and 8.00 Gb-RAM running on Windows 10 operating system. In Tables 6.2.1-6.2.2, "No. of Iter." means the number of iterations.

In our computations, we choose $\tau_n = \frac{10}{(n+2)^2}$, $\alpha_n = \frac{1}{n+2}$, $\lambda_1 = \lambda = 0.9$, $\theta = 0.7$, $\delta = 2.7$, $\ell = 0.5$ and $\eta = 0.6$. Furthermore, in the implementation, we define $\text{TOL}_n := ||x_{n+1} - x_n||$ and use the stopping criterion $\text{TOL}_n < 10^{-3}$ for the iterative processes.

Example 6.2.11. Consider the linear operator $A : \mathbb{R}^m \to \mathbb{R}^m$ (m = 15, 30, 45, 60) defined by

$$A = Mx + q,$$

where

$$M = BB^T + P + Q$$

with $P, Q \in \mathbb{R}^{m \times m}$ randomly generated matrices such that P is skew-symmetric, Q is a diagonal matrix of nonnegative entries and q is a vector in \mathbb{R}^m . The feasible set $\mathcal{C} \subset \mathcal{R}^m$ is a closed and convex subset defined by

$$\mathcal{C} := \{ x \in \mathbb{R}^m : Gx \le b \}.$$

where G is an $k \times m$ matrix and b is a nonnegative vector. We easily observe that A is lipschitz continuous and monotone with L = ||M||. Let q = 0 then, the solution set $VI(\mathbb{C}, A) := \{0\}$. The entries of B, P are generated randomly in [-2, 2] as well as the

starting points x_0, x_1 and the diagonal entries of Q are in (0, 2), and q is taken as the zero vector in \mathbb{R}^m . Moreover, for Algorithm 2.5.7, we define the contraction mapping fx = 0.1x for all $x \in \mathbb{R}^m$. We plot the graph of $||x_{n+1} - x_n||$ against number of iterations choosing m = 15, 30, 45, 60. The numerical results are reported in Figure 6.1 and Table 6.3.1.

Cases		Alg.	Alg.	App.	App.	Alg.
		2.5.6	2.5.7	6.2.13	6.2.14	6.2.2
m = 15	No. of Iter.	19	31	24	33	14
	CPU time	8.2375	5.3746	3.7576	5.3048	5.8052
	(sec)					
m = 30	No. of Iter.	23	33	26	44	18
	CPU time	11.6430	6.3741	4.7479	7.8673	6.9292
	(sec)					
m = 45	No. of Iter.	25	40	29	46	20
	CPU time	9.6496	8.3209	5.6280	9.5082	6.3000
	(sec)					
m = 60	No. of Iter.	26	43	31	51	24
	CPU time	9.5374	10.1644	6.5321	11.4672	6.5011
	(sec)					

Table 6.3.1: Numerical results for Example 6.2.11

Example 6.2.12. Let $\mathcal{H}_1 = (l_2(\mathbb{R}), ||.||_{l_2}) = \mathcal{H}_2$, where $l_2(\mathbb{R}) := \{x = (x_1, x_2, x_3, ...), x_i \in \mathbb{R} : \sum_{i=1}^{\infty} |x_i|^2 < \infty\}$ and $||x||_{l_2} := \left(\sum_{i=1}^{\infty} |x_i|^2\right)^{\frac{1}{2}}, \forall x \in l_2(\mathbb{R})$. Let $\mathcal{C} = \{x \in l_2(\mathbb{R}) : ||x-a||_{l_2} \leq r\}$, where $a = (1, \frac{1}{2}, \frac{1}{3}, \cdots), r = 3$. Then \mathcal{C} is a nonempty, closed and convex subset of $l_2(\mathbb{R})$. Thus,

$$P_{\mathcal{C}}(x) = \begin{cases} x, & \text{if } x \in ||x-a||_{l_2} \le r\\ \frac{x-a}{||x-a||_{l_2}}r+a, & \text{otherwise.} \end{cases}$$

Now, define the operator $A_1 \colon l_2(\mathbb{R}) \to l_2(\mathbb{R})$ by

$$Ax = \left(\|x - a\| + \frac{1}{\|x - a\| + 0.7} \right) x.$$

It is easy to see that the solution set $VI(\mathcal{C}, A) = \{0\}$, A is pseudomonotone on $l_2(\mathbb{R})$, uniformly continuous and sequentially weakly continuous on C but not Lipschitz on $l_2(\mathbb{R})$. More so, for Algorithm 2.5.7, we define the mapping $fx = \frac{x}{2}$, for all $x \in l_2(\mathbb{R})$. Then, we take $u = (1, \frac{1}{5}, \frac{1}{25}, \cdots)$ and consider the following cases for the numerical experiments: **Case 1:** Take $x_1 = (-4, 1, -\frac{1}{4}, \cdots)$ and $x_0 = (2, 1, \frac{1}{2}, \cdots)$.

Case 2: Take
$$x_1 = (-2, 1, -\frac{1}{2}, \cdots)$$
 and $x_0 = (-3, 1, -\frac{1}{3}, \cdots)$.
Case 3: Take $x_1 = (-3, 1, -\frac{1}{3}, \cdots)$ and $x_0 = (-2, 1, -\frac{1}{2}, \cdots)$.
Case 4: Take $x_1 = (-4, 1, -\frac{1}{4}, \cdots)$ and $x_0 = (3, 1, \frac{1}{3}, \cdots)$.



Figure 6.1: Top left: m = 15; Top right: m = 30; Bottom left: m = 45; Bottom right: m = 60.

We plot the graph of $||x_{n+1} - x_n||$ against number of iterations using the above four cases. The numerical results are reported in Figure 6.2 and Table 6.3.2.

		Alg.	Alg.	App.	App.	Alg.
		2.5.6	2.5.7	6.2.13	6.2.14	6.2.2
Case I	No. of Iter.	30	20	39	26	9
	CPU time	0.0169	0.0104	0.0102	0.0102	0.0107
	(sec)					
Case II	No. of Iter.	19	18	26	33	8
	CPU time	0.0164	0.0099	0.0091	0.0096	0.0111
	(sec)					
Case III	No. of Iter.	16	19	33	26	9
	CPU time	0.0181	0.0106	0.0124	0.0132	0.0144
	(sec)					
Case IV	No. of Iter.	35	20	39	32	9
	CPU time	0.0171	0.0093	0.0094	0.0098	0.0104
	(sec)					

 Table 6.3.2: Numerical results for Example 6.2.12

Appendix 6.2.13. (Algorithm 3.3 in [251])

Initialization: Given $\{\alpha_n\} \subset (0, 1), \ell \in (0, 1), \beta > 0, \lambda \in (0, \frac{1}{\beta})$. Let $x_1 \in \mathcal{C}$ be arbitrary **Iterative Steps:** Given the current iterate x_n , calculate x_{n+1} as follows: **Step 1:** Compute

$$z_n = P_{\mathcal{C}}(x_n - \lambda A x_n)$$

and $r_{\lambda}(x_n) := x_n - z_n$. If $r_{\lambda}(x_n) = 0$ then stop and x_n is a solution of $VI(\mathcal{C}, A)$. Otherwise Step 2: Compute

$$y_n = x_n - \tau_n r_\lambda(x_n),$$

where $\tau_n := \ell^{j_n}$ and j_n is the smallest non-negative integer j satisfying

$$\langle Ax_n - Ay_n, r_{\lambda}(x_n) \rangle \le \mu \| r_{\lambda}(x_n) \|^2.$$

Step 2: Compute

$$x_{n+1} = \alpha_n f(x_n) + (1 - \alpha_n) P_{C_n}(x_n)$$

where

$$C_n := \{ x \in \mathcal{C} : h_n(x) \le 0 \}$$



Figure 6.2: Top left: Case I; Top right: Case II; Bottom left: Case III; Bottom right: Case IV.

and

$$h_n(x) = \langle Ay_n, x - y_n \rangle$$

Set n := n + 1 and go to **Step 1**.

where $A : \mathcal{H} \to \mathcal{H}$ is pseudomonotone, uniformly continuous and sequentially weakly continuous on bounded subsets of C.

Appendix 6.2.14. (Algorithm 4 in [214])

Initialization: Given $\{\alpha_n\} \subset (0, 1), \ell \in (0, 1), \beta > 0, \lambda \in (0, \frac{1}{\beta})$. Let $x_1 \in \mathcal{C}$ be arbitrary **Iterative Steps:** Given the current iterate x_n , calculate x_{n+1} as follows: **Step 1:** Compute

$$z_n = P_{\mathcal{C}}(x_n - \lambda A x_n)$$

and $r_{\lambda}(x_n) := x_n - z_n$. If $r_{\lambda}(x_n) = 0$ then stop and x_n is a solution of the VIP. Otherwise Step 2: Compute

$$y_n = x_n - \tau_n r_\lambda(x_n),$$

where $\tau_n := \ell^{j_n}$ and j_n is the smallest non-negative integer j satisfying

$$\langle Ax_n - Ay_n, r_\lambda(x_n) \rangle \le \frac{\mu}{2} ||r_\lambda(x_n)||^2.$$

Step 3: Compute

$$x_{n+1} = \alpha_n f(x_n) + (1 - \alpha_n) P_{C_n}(x_n),$$

where

$$C_n := \{ x \in \mathcal{C} : h_n(x) \le 0 \}$$

and

$$h_n(x) = \langle Ay_n, x - x_n \rangle + \frac{\tau_n}{2} \| r_\lambda(x_n) \|^2.$$

Set n := n + 1 and go to **Step 1**.

where $A: \mathcal{H} \to \mathcal{H}$ is pseudomonotone and uniformly continuous.

6.3 On finite family of split monotone variational inclusion and fixed point problems

In this section, we propose and study a new inertial iterative algorithm with self-adaptive step size for approximating a common solution of finite family of SMV_qIPs and FPP of a nonexpansive mapping between a Banach space and Hilbert space. This method combines the inertial technique with viscosity method and self-adaptive step size for solving the common solution problem. We prove a strong convergence result for the proposed method under some mild conditions. Moreover, we apply our result to study the SFP and split minimization problem (SMP). Finally, we provide some numerical experiments to demonstrate the efficiency of our method in comparison with some well-known methods in the literature. Our method does not require prior knowledge or estimate of the operator norm, which makes it easily implementable unlike so many other methods in the literature which require prior knowledge of the operator norm for their implementation.

6.3.1 Proposed method

Let \mathcal{H} be a real Hilbert space and let \mathcal{X} be a uniformly convex and smooth Banach space. Let $J_{\mathcal{X}}$ be the duality mapping on \mathcal{X} . Let $T : \mathcal{H} \to \mathcal{X}$ be a bounded linear operator such that $T \neq 0$ and T^* is the adjoint operator of T. Let $f : \mathcal{H} \to \mathcal{H}$ be a contraction mapping with constant $\rho \in (0,1)$ and $0 < \gamma < \frac{\gamma}{\rho}$, and let $S : \mathcal{H} \to \mathcal{H}$ be a nonexpansive mapping. For each $i = 1, 2, \cdots, m$, let $A_i : \mathcal{H} \to \mathcal{H}$ be a finite family of α_i -inverse strongly monotone operators, $B_i : \mathcal{H} \to 2^{\mathcal{H}}$ and $C_i : \mathcal{X} \to 2^{\mathcal{X}^*}$ be finite families of maximal monotone operators, and let $J_{r_i}^{B_i}$ be the resolvent of B_i for $r_i > 0$ and $Q_{\mu_i}^{C_i}$ be the metric resolvent of C_i for $\mu_i > 0$. Suppose that the solution set $\Gamma \neq \emptyset$. We establish the convergence of the algorithm under the following assumptions on the control parameters:

(A1)
$$\{\alpha_n\}, \{\delta_n\}, \{\gamma_n\} \subset (0, 1)$$
 such that $\alpha_n + \delta_n + \gamma_n = 1$, $\lim_{n \to \infty} \alpha_n = 0$, $\sum_{n=0}^{\infty} \alpha_n = \infty$;

(A2)
$$\liminf_{n \to \infty} \delta_n > 0, \liminf_{n \to \infty} \gamma_n > 0, \{\beta_{n,i}\} \subset (0,1) \text{ and } \sum_{i=0}^m \beta_{n,i} = 1, \liminf_{n \to \infty} \beta_{n,0}\beta_{n,i} > 0;$$

(A3) $0 < a \le \tau_n \le b < 2, 0 < c \le \mu_{n,i}, 0 < d \le r_{n,i} \le e < 2\alpha_i$ for each $i = 1, 2, \dots, m$;

(A4) Let $\theta > 0$ and $\{\epsilon_n\}$ be a nonnegative sequence such that $0 < d \le \epsilon_n$ and $\epsilon_n = \circ(\alpha_n)$, i.e., $\lim_{n \to \infty} \frac{\epsilon_n}{\alpha_n} = 0$.

Algorithm 6.3.1.

Step 0: Select initial points $x_0, x_1 \in \mathcal{H}$ and set n = 1.

Step 1: Given the iterates x_{n-1} and x_n for each $n \ge 1$, choose θ_n such that $0 \le \theta_n \le \overline{\theta}_n$, where

$$\bar{\theta}_n := \begin{cases} \min\left\{\theta, \frac{\epsilon_n}{\|x_n - x_{n-1}\|}\right\}, & \text{if } x_n \neq x_{n-1} \\ \theta, & \text{otherwise.} \end{cases}$$
(6.3.1)

Step 2: Compute

$$w_n = x_n + \theta_n (x_n - x_{n-1}).$$

Step 3: Compute

$$z_{n,i} = w_n - \lambda_{n,i} T^* J_{\mathcal{X}} (I - Q_{\mu_{n,i}}^{C_i}) T w_n,$$

where

$$\lambda_{n,i} = \frac{\tau_n \|J_{\mathcal{X}}(I - Q_{\mu_{n,i}}^{C_i})Tw_n\|^2}{\|T^* J_{\mathcal{X}}(I - Q_{\mu_{n,i}}^{C_i})Tw_n\|^2}.$$

Step 4: Compute

$$u_n = \beta_{n,0} w_n + \sum_{i=1}^m \beta_{n,i} J_{r_{n,i}}^{B_i} (I - r_{n,i} A_i) z_{n,i}.$$

Step 5: Compute

$$x_{n+1} = \alpha_n f(x_n) + \delta_n x_n + \gamma_n S u_n.$$

Set n := n + 1 and return to **Step 1**.

Remark 6.3.1. By conditions (A1) and (A4), one can easily verify from (6.3.1) that

$$\lim_{n \to \infty} \theta_n ||x_n - x_{n-1}|| = 0 \text{ and } \lim_{n \to \infty} \frac{\theta_n}{\alpha_n} ||x_n - x_{n-1}|| = 0.$$
 (6.3.2)

6.3.2 Convergence analysis

Lemma 6.3.2. Let $\{x_n\}$ be a sequence generated by Algorithm 6.3.1 under Assumption (A1)-(A4). Then, $\{x_n\}$ is bounded.

Let $p \in \Gamma$. Then, it follows that

$$p = J_{r_{n,i}}^{B_i}(I - r_{n,i}A_i)p, Tp = Q_{\mu_{n,i}}^{C_i}Tp, \text{ and } p = Sp$$

for all $n \in \mathbb{N}$ and i = 1, 2, ..., m. From the definition of w_n in **Step 2** and by applying the triangle inequality, we have

$$||w_{n} - p|| = ||x_{n} + \theta_{n}(x_{n} - x_{n-1}) - p||$$

$$\leq ||x_{n} - p|| + \theta_{n}||x_{n} - x_{n-1}||$$

$$= ||x_{n} - p|| + \alpha_{n} \frac{\theta_{n}}{\alpha_{n}}||x_{n} - x_{n-1}||.$$
(6.3.3)

Since, by Remark 6.3.1, $\lim_{n\to\infty} \frac{\theta_n}{\alpha_n} ||x_n - x_{n-1}|| = 0$, it follows that there exists a constant $M_1 > 0$ such that $\frac{\theta_n}{\alpha_n} ||x_n - x_{n-1}|| \le M_1$ for all $n \ge 1$. Hence, it follows from (6.3.3) that

$$||w_n - p|| \le ||x_n - p|| + \alpha_n M_1.$$
(6.3.4)

From Step 3 and the property of the resolvent (2.4.10), we have

$$\begin{aligned} \|z_{n,i} - p\|^{2} &= \|w_{n} - \lambda_{n,i}T^{*}J_{\mathcal{X}}(I - Q_{\mu_{n,i}}^{C_{i}})Tw_{n} - p\|^{2} \\ &= \|w_{n} - p\|^{2} - 2\lambda_{n,i}\langle w_{n} - p, T^{*}J_{\mathcal{X}}(I - Q_{\mu_{n,i}}^{C_{i}})Tw_{n}\rangle \\ &+ \lambda_{n,i}^{2}\|T^{*}J_{\mathcal{X}}(I - Q_{\mu_{n,i}}^{C_{i}})Tw_{n}\|^{2} \\ &= \|w_{n} - p\|^{2} - 2\lambda_{n,i}\langle Tw_{n} - Tp, J_{\mathcal{X}}(I - Q_{\mu_{n,i}}^{C_{i}})Tw_{n}\rangle \\ &+ \lambda_{n,i}^{2}\|T^{*}J_{\mathcal{X}}(I - Q_{\mu_{n,i}}^{C_{i}})Tw_{n}\|^{2} \\ &= \|w_{n} - p\|^{2} - 2\lambda_{n,i}\langle Tw_{n} - Q_{\mu_{n,i}}^{C_{i}}Tw_{n}, J_{\mathcal{X}}(I - Q_{\mu_{n,i}}^{C_{i}})Tw_{n}\rangle \\ &- 2\lambda_{n,i}\langle Q_{\mu_{n,i}}^{C_{i}}Tw_{n} - Tp, J_{\mathcal{X}}(I - Q_{\mu_{n,i}}^{C_{i}})Tw_{n}\rangle + \lambda_{n,i}^{2}\|T^{*}J_{\mathcal{X}}(I - Q_{\mu_{n,i}}^{C_{i}})Tw_{n}\|^{2} \\ &= \|w_{n} - p\|^{2} - 2\lambda_{n,i}\|J_{\mathcal{X}}(I - Q_{\mu_{n,i}}^{C_{i}})Tw_{n}\|^{2} \\ &- 2\lambda_{n,i}\langle Q_{\mu_{n,i}}^{C_{i}}Tw_{n} - Tp, J_{\mathcal{X}}(I - Q_{\mu_{n,i}}^{C_{i}})Tw_{n}\rangle + \lambda_{n,i}^{2}\|T^{*}J_{\mathcal{X}}(I - Q_{\mu_{n,i}}^{C_{i}})Tw_{n}\|^{2} \\ &\leq \|w_{n} - p\|^{2} - 2\lambda_{n,i}\|J_{\mathcal{X}}(I - Q_{\mu_{n,i}}^{C_{i}})Tw_{n}\|^{2} + \lambda_{n,i}^{2}\|T^{*}J_{\mathcal{X}}(I - Q_{\mu_{n,i}}^{C_{i}})Tw_{n}\|^{2} \\ &= \|w_{n} - p\|^{2} - 2\lambda_{n,i}\|J_{\mathcal{X}}(I - Q_{\mu_{n,i}}^{C_{i}})Tw_{n}\|^{2} - \lambda_{n,i}\|T^{*}J_{\mathcal{X}}(I - Q_{\mu_{n,i}}^{C_{i}})Tw_{n}\|^{2} \\ &= \|w_{n} - p\|^{2} - \lambda_{n,i}\left[2\|J_{\mathcal{X}}(I - Q_{\mu_{n,i}}^{C_{i}})Tw_{n}\|^{2} - \lambda_{n,i}\|T^{*}J_{\mathcal{X}}(I - Q_{\mu_{n,i}}^{C_{i}})Tw_{n}\|^{2}\right] \\ &= (6.3.5) \end{aligned}$$

From the definition of $\lambda_{n,i}$ and (6.3.5), we have

$$||z_{n,i} - p||^2 \le ||w_n - p||^2 - \frac{\tau_n (2 - \tau_n) ||J_{\mathcal{X}} (I - Q_{Q_{\mu_{n,i}}}^{C_i}) T w_n||^4}{||T^* J_{\mathcal{X}} (I - Q_{\mu_{n,i}}^{C_i}) T w_n||^2}.$$
(6.3.6)

Thus, by the assumption on τ_n we have that

$$||z_{n,i} - p||^2 \le ||w_n - p||^2.$$
(6.3.7)

From the definition of u_n in **Step 4**, the nonexpansivity of $J_{r_{n,i}}^{B_i}$ and Lemma 2.5.7 we get

$$\|u_{n} - p\|^{2} = \|\beta_{n,0}w_{n} + \sum_{i=1}^{m} \beta_{n,i}J_{r_{n,i}}^{B_{i}}(I - r_{n,i}A_{i})z_{n,i} - J_{r_{n,i}}^{B_{i}}(I - r_{n,i}A_{i})p\|^{2}$$

$$= \beta_{n,0}\|w_{n} - J_{r_{n,i}}^{B_{i}}(I - r_{n,i}A_{i})p\|^{2} + \sum_{i=1}^{m} \beta_{n,i}\|J_{r_{n,i}}^{B_{i}}(I - r_{n,i}A_{i})z_{n,i}$$

$$- J_{r_{n,i}}^{B_{i}}(I - r_{n,i}A_{i})p\|^{2} - \beta_{n,0}\sum_{i=1}^{m} \beta_{n,i}\|J_{r_{n,i}}^{B_{i}}(I - r_{n,i}A_{i})z_{n,i} - w_{n}\|^{2}$$

$$\leq \beta_{n,0}\|w_{n} - p\|^{2} + \sum_{i=1}^{m} \beta_{n,i}\|z_{n,i} - p - r_{n,i}(A_{i}z_{n,i} - A_{i}p)\|^{2}$$

$$- \beta_{n,0}\sum_{i=1}^{m} \beta_{n,i}\|J_{r_{n,i}}^{B_{i}}(I - r_{n,i}A_{i})z_{n,i} - w_{n}\|^{2}.$$
(6.3.8)

Next, by applying the inversely strongly monotonicity of the A_i 's, Lemma 2.1.1 and (6.3.7), for each i = 1, 2, ..., m, we observe that

$$||z_{n,i} - p - r_{n,i}(A_i z_{n,i} - A_i p)||^2 = ||z_{n,i} - p||^2 - 2r_{n,i} \langle A_i z_{n,i} - A_i p, z_{n,i} - p \rangle + r_{n,i}^2 ||A_i z_{n,i} - A_i p||^2 \leq ||z_{n,i} - p||^2 - 2r_{n,i} \alpha_i ||A_i z_{n,i} - A_i p||^2 + r_{n,i}^2 ||A_i z_{n,i} - A_i p||^2 = ||z_{n,i} - p||^2 - (2\alpha_i - r_{n,i})r_{n,i} ||A_i z_{n,i} - A_i p||^2.$$
(6.3.9)

Now, by applying (6.3.6), it follows from (6.3.8) and (6.3.9) that

$$\begin{split} \|u_{n} - p\|^{2} &\leq \beta_{n,0} \|w_{n} - p\|^{2} + \sum_{i=1}^{m} \beta_{n,i} \|z_{n,i} - p\|^{2} - \sum_{i=1}^{m} \beta_{n,i} (2\alpha_{i} - r_{n,i})r_{n,i} \|A_{i}z_{n,i} - A_{i}p\|^{2} \\ &- \beta_{n,0} \sum_{i=1}^{m} \beta_{n,i} \|J_{r_{n,i}}^{B_{i}}(I - r_{n,i}A_{i})z_{n,i} - w_{n}\|^{2} \\ &\leq \beta_{n,0} \|w_{n} - p\|^{2} + \sum_{i=1}^{m} \beta_{n,i} \|w_{n} - p\|^{2} - \sum_{i=1}^{m} \beta_{n,i} \frac{\tau_{n}(2 - \tau_{n})}{\|T^{*}J_{\mathcal{X}}(I - Q_{Q_{\mu_{n,i}}}^{C_{i}})Tw_{n}\|^{2}} \\ &- \sum_{i=1}^{m} \beta_{n,i} (2\alpha_{i} - r_{n,i})r_{n,i} \|A_{i}z_{n,i} - A_{i}p\|^{2} - \beta_{n,0} \sum_{i=1}^{m} \beta_{n,i} \|J_{r_{n,i}}^{B_{i}}(I - r_{n,i}A_{i})z_{n,i} - w_{n}\|^{2} \\ &= \|w_{n} - p\|^{2} - \sum_{i=1}^{m} \beta_{n,i} \frac{\tau_{n}(2 - \tau_{n})}{\|T^{*}J_{\mathcal{X}}(I - Q_{Q_{\mu_{n,i}}}^{C_{i}})Tw_{n}\|^{4}} \\ &- \sum_{i=1}^{m} \beta_{n,i} (2\alpha_{i} - r_{n,i})r_{n,i} \|A_{i}z_{n,i} - A_{i}p\|^{2} \\ &- \sum_{i=1}^{m} \beta_{n,i} (2\alpha_{i} - r_{n,i})r_{n,i} \|A_{i}z_{n,i} - A_{i}p\|^{2} \\ &- \beta_{n,0} \sum_{i=1}^{m} \beta_{n,i} \|J_{r_{n,i}}^{B_{i}}(I - r_{n,i}A_{i})z_{n,i} - w_{n}\|^{2} \\ &\leq \|w_{n} - p\|^{2} \end{split}$$

$$(6.3.10)$$

Consequently, we have that

$$||u_n - p|| \le ||w_n - p|| \le ||x_n - p|| + \alpha_n M_1.$$
(6.3.12)

From Step 5, (6.3.12) and the nonexpansivity of S, we have

$$\begin{aligned} \|x_{n+1} - p\| &= \|\alpha_n f(x_n) + \delta_n x_n + \gamma_n S u_n - p\| \\ &\leq \alpha_n \|f(x_n) - p\| + \delta_n \|x_n - p\| + \gamma_n \|S u_n - p\| \\ &\leq \alpha_n \|f(x_n) - f(p)\| + \alpha_n \|f(p) - p\| + \delta_n \|x_n - p\| + \gamma_n \left(\|x_n - p\| + \alpha_n M_1\right) \\ &\leq \alpha_n \rho \|x_n - p\| + \alpha_n \|f(p) - p\| + (1 - \alpha_n) \|x_n - p\| + \gamma_n \alpha_n M_1 \\ &\leq \left(1 - \alpha_n (1 - \rho)\right) \|x_n - p\| + \alpha_n (1 - \rho) \left[\frac{\|f(p) - p\|}{1 - \rho} + \frac{M_1}{1 - \rho}\right] \\ &\leq \max \left\{\|x_n - p\|, \frac{\|f(p) - p\| + M_1}{1 - \rho}\right\} \end{aligned}$$

Hence, the sequence $\{x_n\}$ is bounded. Consequently, $\{w_n\}, \{z_{n,i}\}$ and $\{u_n\}$ are bounded.

Lemma 6.3.3. Let $\{x_n\}$ be a sequence generated by Algorithm 6.3.1 under Assumption (A1)-(A4). Then, for all $p \in \Gamma$ and $n \in \mathbb{N}$ we have:

$$\begin{split} \|x_{n+1} - p\|^{2} &\leq (1 - \eta_{n}) \|x_{n} - p\|^{2} + \eta_{n} \Big[\frac{\alpha_{n}}{2(1 - \rho)} M_{3} + 3M_{2} \frac{\gamma_{n}(1 - \alpha_{n})}{2(1 - \rho)} \frac{\theta_{n}}{\alpha_{n}} \|x_{n} - x_{n-1}\| \\ &+ \frac{1}{(1 - \rho)} \langle f(p) - p, x_{n+1} - p \rangle \Big] \\ &- \sigma_{n} \Big[\sum_{i=1}^{m} \beta_{n,i} \frac{\tau_{n}(2 - \tau_{n}) \|J_{\mathcal{X}}(I - Q_{Q\mu_{n,i}}^{C_{i}}) T w_{n}\|^{4}}{\|T^{*} J_{\mathcal{X}}(I - Q_{\mu_{n,i}}^{C_{i}}) T w_{n}\|^{2}} \\ &+ \sum_{i=1}^{m} \beta_{n,i} (2\alpha_{i} - r_{n,i}) r_{n,i} \|A_{i} z_{n,i} - A_{i} p\|^{2} + \beta_{n,0} \sum_{i=1}^{m} \beta_{n,i} \|J_{r_{n,i}}^{B_{i}}(I - r_{n,i}A_{i}) z_{n,i} - w_{n}\|^{2} \Big], \end{split}$$

where $\eta_n = \frac{2\alpha_n(1-\rho)}{(1-\alpha_n\rho)}$ and $\sigma_n = \frac{\gamma_n(1-\alpha_n)}{(1-\alpha_n\rho)}$.

Proof. Let $p \in \Gamma$. Then from **Step 2**, by applying the Cauchy-Schwartz inequality and Lemma 2.1.1, we obtain

$$\begin{aligned} ||w_{n} - p||^{2} &= ||x_{n} + \theta_{n}(x_{n} - x_{n-1}) - p||^{2} \\ &= ||x_{n} - p||^{2} + \theta_{n}^{2}||x_{n} - x_{n-1}||^{2} + 2\theta_{n}\langle x_{n} - p, x_{n} - x_{n-1}\rangle \\ &\leq ||x_{n} - p||^{2} + \theta_{n}^{2}||x_{n} - x_{n-1}||^{2} + 2\theta_{n}||x_{n} - x_{n-1}|||x_{n} - p|| \\ &= ||x_{n} - p||^{2} + \theta_{n}||x_{n} - x_{n-1}||(\theta_{n}||x_{n} - x_{n-1}|| + 2||x_{n} - p||) \\ &\leq ||x_{n} - p||^{2} + 3M_{2}\theta_{n}||x_{n} - x_{n-1}|| \\ &= ||x_{n} - p||^{2} + 3M_{2}\alpha_{n}\frac{\theta_{n}}{\alpha_{n}}||x_{n} - x_{n-1}||, \end{aligned}$$
(6.3.13)

where $M_2 := \sup_{n \in \mathbb{N}} \{ ||x_n - p||, \theta_n ||x_n - x_{n-1}|| \} > 0.$ By applying Lemma 2.1.1, (6.3.10) and (6.3.13) we have

$$\begin{split} \|x_{n+1} - p\|^2 &= \|\alpha_n f(x_n) + \delta_n x_n + \gamma_n S u_n - p\|^2 \\ &= \|\alpha_n (f(x_n) - p) + \delta_n (x_n - p) + \gamma_n (S u_n - p)\|^2 + 2\alpha_n (f(x_n) - p, x_{n+1} - p) \\ &= \delta_n^2 \|x_n - p\|^2 + \gamma_n^2 \|S u_n - p\|^2 + 2\delta_n \gamma_n (x_n - p, S u_n - p) \\ &+ 2\alpha_n (f(x_n) - p, x_{n+1} - p) \\ &\leq \delta_n^2 \|x_n - p\|^2 + \gamma_n^2 \|S u_n - p\|^2 + 2\delta_n \gamma_n \|x_n - p\| \|S u_n - p\| \\ &+ 2\alpha_n (f(x_n) - p, x_{n+1} - p) \\ &\leq \delta_n^2 \|x_n - p\|^2 + \gamma_n^2 \|S u_n - p\|^2 + \delta_n \gamma_n \left(\|x_n - p\|^2 + \|S u_n - p\|^2 \right) \\ &+ 2\alpha_n (f(x_n) - p, x_{n+1} - p) \\ &\leq \delta_n (\delta_n + \gamma_n) \|x_n - p\|^2 + \gamma_n (\gamma_n + \delta_n) \|u_n - p\|^2 + 2\alpha_n (f(x_n) - p, x_{n+1} - p) \\ &= \delta_n (1 - \alpha_n) \|x_n - p\|^2 + \gamma_n (1 - \alpha_n) \|u_n - p\|^2 + 2\alpha_n (f(x_n) - f(p), x_{n+1} - p) \\ &\leq \delta_n (1 - \alpha_n) \|x_n - p\|^2 \\ &+ \gamma_n (1 - \alpha_n) \left[\|w_n - p\|^2 - \sum_{i=1}^m \beta_{n,i} \frac{\tau_n (2 - \tau_n) \|J_X (I - Q_{\mu_{n,i}}^{C_i}) T w_n \|^4}{\|T^* J_X (I - Q_{\mu_{n,i}}^{C_i}) T w_n \|^2} \\ &- \sum_{i=1}^m \beta_{n,i} (2\alpha_i - r_{n,i}) r_{n,i} \|A_i z_{n,i} - A_i p\|^2 \\ &- \beta_n 0 \sum_{i=1}^m \beta_{n,i} \|J_{r_{n,i}}^R (I - r_{n,i} A_i) z_{n,i} - w_n \|^2 \right] \\ &+ \gamma_n (1 - \alpha_n) \left[\|x_n - p\|^2 + 3M_2 \alpha_n \frac{\theta_n}{\alpha_n} \|x_n - x_{n-1}\| \\ &- \sum_{i=1}^m \beta_{n,i} (2\alpha_i - r_{n,i}) T_{n,i} \|A_i z_{n,i} - A_i p\|^2 \\ &- \beta_n 0 \sum_{i=1}^m \beta_{n,i} (2\alpha_i - r_{n,i}) T_{n,i} \|A_i z_{n,i} - A_i p\|^2 \\ &- \sum_{i=1}^m \beta_{n,i} (2\alpha_i - r_{n,i}) T_{n,i} \|A_i z_{n,i} - w_n\|^2 \right] \\ &+ \gamma_n (1 - \alpha_n) \left[\|x_n - p\|^2 + 3M_2 \alpha_n \frac{\theta_n}{\alpha_n} \|x_n - x_{n-1}\| \\ &- \sum_{i=1}^m \beta_{n,i} (2\alpha_i - r_{n,i}) T_{n,i} \|A_i z_{n,i} - A_i p\|^2 \\ &- \beta_{n,0} \sum_{i=1}^m \beta_{n,i} \|J_{r_{n,i}}^{B_i} (I - r_{n,i} A_i) z_{n,i} - w_n\|^2 \right] \\ &+ \gamma_n (1 - \alpha_n) \left[\|x_n - p\|^2 + 3M_2 \alpha_n \frac{\theta_n}{\alpha_n} \|x_n - x_{n-1}\| \\ &- \sum_{i=1}^m \beta_{n,i} (2\alpha_i - r_{n,i}) T_{n,i} \|A_i z_{n,i} - A_i p\|^2 \\ &- \beta_{n,0} \sum_{i=1}^m \beta_{n,i} \|J_{r_{n,i}}^{B_i} (I - r_{n,i} A_i) z_{n,i} - w_n\|^2 \right] \\ &+ \alpha_n \rho (\|x_n - p\|^2 + \|x_{n+1} - p\|^2) + 2\alpha_n (f(p) - p, x_{n+1} - p) \end{aligned}$$

$$= ((1 - \alpha_{n})^{2} + \alpha_{n}\rho) \|x_{n} - p\|^{2} + \alpha_{n}\rho \|x_{n+1} - p\|^{2}$$

+ $3M_{2}\gamma_{n}(1 - \alpha_{n})\alpha_{n}\frac{\theta_{n}}{\alpha_{n}}\|x_{n} - x_{n-1}\| + 2\alpha_{n}\langle f(p) - p, x_{n+1} - p \rangle$
- $\gamma_{n}(1 - \alpha_{n}) \Big[\sum_{i=1}^{m} \beta_{n,i} \frac{\tau_{n}(2 - \tau_{n}) \|J_{\mathcal{X}}(I - Q_{Q\mu_{n,i}}^{C_{i}})Tw_{n}\|^{4}}{\|T^{*}J_{\mathcal{X}}(I - Q_{\mu_{n,i}}^{C_{i}})Tw_{n}\|^{2}}$
+ $\sum_{i=1}^{m} \beta_{n,i}(2\alpha_{i} - r_{n,i})r_{n,i}\|A_{i}z_{n,i} - A_{i}p\|^{2}$
+ $\beta_{n,0} \sum_{i=1}^{m} \beta_{n,i}\|J_{r_{n,i}}^{B_{i}}(I - r_{n,i}A_{i})z_{n,i} - w_{n}\|^{2} \Big].$

From this we obtain

$$\begin{split} \|x_{n+1} - p\|^2 &\leq \frac{\left(1 - 2\alpha_n + \alpha_n^2 + \alpha_n \rho\right)}{(1 - \alpha_n \rho)} \|x_n - p\|^2 + \frac{3M_2\gamma_n(1 - \alpha_n)}{(1 - \alpha_n \rho)} \alpha_n \frac{\theta_n}{\alpha_n} \|x_n - x_{n-1}\| \\ &+ \frac{2\alpha_n}{(1 - \alpha_n \rho)} \langle f(p) - p, x_{n+1} - p \rangle \\ &- \frac{\gamma_n(1 - \alpha_n)}{(1 - \alpha_n \rho)} \Big[\sum_{i=1}^m \beta_{n,i} \frac{\tau_n(2 - \tau_n) \|J_X(I - Q_{Q_{\mu_n,i}}^{C_i})Tw_n\|^4}{\|T^*J_X(I - Q_{\mu_{n,i}}^{C_i})Tw_n\|^2} \\ &+ \sum_{i=1}^m \beta_{n,i}(2\alpha_i - r_{n,i})r_{n,i} \|A_i z_{n,i} - A_i p\|^2 \\ &+ \beta_{n,0} \sum_{i=1}^m \beta_{n,i} \|J_{r_{n,i}}^{B_i}(I - r_{n,i}A_i)z_{n,i} - w_n\|^2 \Big] \\ &= \frac{\left(1 - 2\alpha_n + \alpha_n \rho\right)}{(1 - \alpha_n \rho)} \|x_n - p\|^2 + \frac{\alpha_n^2}{(1 - \alpha_n \rho)} \|x_n - p\|^2 \\ &+ \frac{3M_2\gamma_n(1 - \alpha_n)}{(1 - \alpha_n \rho)} \alpha_n \frac{\theta_n}{\alpha_n} \|x_n - x_{n-1}\| + \frac{2\alpha_n}{(1 - \alpha_n \rho)} \langle f(p) - p, x_{n+1} - p \rangle \\ &- \frac{\gamma_n(1 - \alpha_n)}{(1 - \alpha_n \rho)} \Big[\sum_{i=1}^m \beta_{n,i} \frac{\tau_n(2 - \tau_n) \|J_X(I - Q_{Q_{\mu_n,i}}^{C_i})Tw_n\|^2}{\|T^*J_X(I - Q_{\mu_{n,i}}^{C_i})Tw_n\|^2} \\ &+ \sum_{i=1}^m \beta_{n,i}(2\alpha_i - r_{n,i})r_{n,i} \|A_i z_{n,i} - A_i p\|^2 \\ &+ \beta_{n,0} \sum_{i=1}^m \beta_{n,i} \|J_{r_{n,i}}^{B_i}(I - r_{n,i}A_i)z_{n,i} - w_n\|^2 \Big] \\ &\leq \Big(1 - \frac{2\alpha_n(1 - \rho)}{(1 - \alpha_n \rho)}\Big) \Big\|x_n - p\|^2 \\ &+ \frac{2\alpha_n(1 - \rho)}{(1 - \alpha_n \rho)} \Big[\frac{\alpha_n}{2(1 - \rho)} M_3 + 3M_2 \frac{\gamma_n(1 - \alpha_n)}{2(1 - \rho)} \frac{\theta_n}{\alpha_n} \|x_n - x_{n-1}\| \\ &+ \frac{1}{(1 - \rho)} \langle f(p) - p, x_{n+1} - p \rangle \Big] \\ &- \frac{\gamma_n(1 - \alpha_n)}{(1 - \alpha_n \rho)} \Big[\sum_{i=1}^m \beta_{n,i} \frac{\tau_n(2 - \tau_n) \|J_X(I - Q_{Q_{\mu_{n,i}}}^{C_i})Tw_n\|^4}{2(1 - \rho)} \\ &+ \sum_{i=1}^m \beta_{n,i}(2\alpha_i - r_{n,i})r_{n,i} \|A_i z_{n,i} - A_i p\|^2 \\ &+ \sum_{i=1}^m \beta_{n,i}(2\alpha_i - r_{n,i})r_{n,i} \|A_i z_{n,i} - A_i p\|^2 \\ &+ \sum_{i=1}^m \beta_{n,i}(2\alpha_i - r_{n,i})r_{n,i} \|A_i z_{n,i} - A_i p\|^2 \\ &+ \sum_{i=1}^m \beta_{n,i}(2\alpha_i - r_{n,i})r_{n,i} \|A_i z_{n,i} - A_i p\|^2 \\ &+ \sum_{i=1}^m \beta_{n,i}(2\alpha_i - r_{n,i})r_{n,i} \|A_i z_{n,i} - A_i p\|^2 \\ &+ \sum_{i=1}^m \beta_{n,i}(2\alpha_i - r_{n,i})r_{n,i} \|A_i z_{n,i} - A_i p\|^2 \\ &+ \sum_{i=1}^m \beta_{n,i}(2\alpha_i - r_{n,i})r_{n,i} \|A_i z_{n,i} - A_i p\|^2 \\ &+ \beta_{n,0} \sum_{i=1}^m \beta_{n,i} \|J_{r_{n,i}}^{B_i}(I - r_{n,i}A_i)z_{n,i} - w_n\|^2 \Big], \end{aligned}$$

where $M_3 = \sup\{||x_n - p||^2 : n \in \mathbb{N}\}$. By taking $\eta_n = \frac{2\alpha_n(1-\rho)}{(1-\alpha_n\rho)}$ and $\sigma_n = \frac{\gamma_n(1-\alpha_n)}{(1-\alpha_n\rho)}$, we obtain the desired result.

Lemma 6.3.4. Let $p \in \Gamma$ and suppose $\{x_n\}$ is a sequence generated by Algorithm 6.3.1. Under Assumption (A1)-(A4), the following inequality holds for all $n \in \mathbb{N}$:

 $||x_{n+1}-p||^2 \le \alpha_n ||f(x_n)-p||^2 + (1-\alpha_n) ||x_n-p||^2 + 3M_2\gamma_n\alpha_n \frac{\theta_n}{\alpha_n} ||x_n-x_{n-1}|| - \delta_n\gamma_n ||x_n-Su_n||^2.$

Proof. Let $p \in \Gamma$. By applying Lemma 2.5.7, (6.3.11) and (6.3.13), from Step 5 we have

$$\begin{aligned} \|x_{n+1} - p\|^2 &= \|\alpha_n f(x_n) + \delta_n x_n + \gamma_n S u_n - p\|^2 \\ &\leq \alpha_n \|f(x_n) - p\|^2 + \delta_n \|x_n - p\|^2 + \gamma_n \|S u_n - p\|^2 - \delta_n \gamma_n \|x_n - S u_n\|^2 \\ &\leq \alpha_n \|f(x_n) - p\|^2 + \delta_n \|x_n - p\|^2 + \gamma_n (\|x_n - p\|^2 - \delta_n \gamma_n \|x_n - S u_n\|^2 \\ &\leq \alpha_n \|f(x_n) - p\|^2 + \delta_n \|x_n - p\|^2 + \gamma_n (\|x_n - p\|^2 + 3M_2 \alpha_n \frac{\theta_n}{\alpha_n} \|x_n - x_{n-1}\|) \\ &- \delta_n \gamma_n \|x_n - S u_n\|^2 \\ &= \alpha_n \|f(x_n) - p\|^2 + (1 - \alpha_n) \|x_n - p\|^2 + 3M_2 \gamma_n \alpha_n \frac{\theta_n}{\alpha_n} \|x_n - x_{n-1}\|) \\ &- \delta_n \gamma_n \|x_n - S u_n\|^2, \end{aligned}$$

which is the desired inequality.

We are now in the position to give the strong convergence theorem for Algorithm 6.3.1.

Theorem 6.3.5. Let \mathcal{H} be a Hilbert space, \mathcal{X} a uniformly convex and smooth Banach space and $J_{\mathcal{X}}$ the duality mapping on \mathcal{X} . Let $\{x_n\}$ be generated by Algorithm 6.3.1 and suppose Assumption (A1-A4) are satisfied. Then, the sequence $\{x_n\}$ converges strongly to a point $\bar{x} \in \Gamma$, where $\bar{x} = P_{\Gamma} \circ f(\bar{x})$.

Proof. Let $\bar{x} = P_{\Gamma} \circ f(\bar{x})$. From Lemma 6.3.3 we have

$$\|x_{n+1} - \bar{x}\|^{2} \leq (1 - \eta_{n}) \|x_{n} - \bar{x}\|^{2} + \eta_{n} \Big[\frac{\alpha_{n}}{2(1 - \rho)} M_{3} + 3M_{2} \frac{\gamma_{n}(1 - \gamma_{n})}{2(1 - \rho)} \frac{\theta_{n}}{\alpha_{n}} \|x_{n} - x_{n-1}\| + \frac{1}{(1 - \rho)} \langle f(\bar{x}) - \bar{x}, x_{n+1} - \bar{x} \rangle \Big].$$

$$(6.3.14)$$

Now, we claim that the sequence $\{\|x_n - \bar{x}\|\}$ converges to zero. To establish this, by Lemma 2.5.36 it suffices to show that $\limsup_{k\to\infty} \langle f(\bar{x}) - \bar{x}, x_{n_k+1} - \bar{x} \rangle \leq 0$ for every subsequence $\{\|x_{n_k} - \bar{x}\|\}$ of $\{\|x_n - \bar{x}\|\}$ satisfying

$$\liminf_{k \to \infty} \left(\|x_{n_k+1} - \bar{x}\| - \|x_{n_k} - \bar{x}\| \right) \ge 0.$$

Suppose that $\{||x_{n_k} - \bar{x}||\}$ is a subsequence of $\{||x_n - \bar{x}||\}$ such that

$$\liminf_{k \to \infty} \left(\|x_{n_k+1} - \bar{x}\| - \|x_{n_k} - \bar{x}\| \right) \ge 0.$$
(6.3.15)

From Lemma 6.3.3, we have

$$\sigma_{n_k} \sum_{i=1}^m \beta_{n_k,i} \frac{\tau_{n_k} (2 - \tau_{n_k}) \| J_{\mathcal{X}} \left(I - Q_{Q_{\mu_{n_k,i}}}^{C_i} \right) T w_{n_k} \|^4}{\| T^* J_{\mathcal{X}} \left(I - Q_{\mu_{n_k,i}}^{C_i} \right) T w_{n_k} \|^2} \le (1 - \eta_{n_k}) \| x_{n_k} - \bar{x} \|^2 - \| x_{n_k+1} - \bar{x} \|^2$$

$$+\eta_{n_{k}}\left[\frac{\alpha_{n_{k}}}{2(1-\rho)}M_{3}+3M_{2}\frac{\gamma_{n_{k}}(1-\alpha_{n_{k}})}{2(1-\rho)}\frac{\theta_{n_{k}}}{\alpha_{n_{k}}}\|x_{n_{k}}-x_{n_{k}-1}\|+\frac{1}{\alpha_{n_{k}}(1-\rho)}\langle f(\bar{x})-\bar{x},x_{n_{k}+1}-\bar{x}\rangle\right]$$

Since $\lim_{k\to\infty} \alpha_{n_k} = 0$, then $\lim_{k\to\infty} \eta_{n_k} = 0$. Hence, by applying (6.3.15) we obtain

$$\sigma_{n_k} \sum_{i=1}^m \beta_{n_k,i} \frac{\tau_{n_k} (2 - \tau_{n_k}) \| J_{\mathcal{X}} \left(I - Q_{Q_{\mu_{n_k,i}}}^{C_i} \right) T w_{n_k} \|^4}{\| T^* J_{\mathcal{X}} \left(I - Q_{\mu_{n_k,i}}^{C_i} \right) T w_{n_k} \|^2} \to 0, \text{ as } k \to \infty.$$

By the conditions on the control parameters, it follows that

$$\beta_{n_k,i} \frac{\tau_{n_k}(2-\tau_{n_k}) \|J_{\mathcal{X}}\left(I-Q_{Q_{\mu_{n_k,i}}}^{C_i}\right) T w_{n_k}\|^4}{\|T^* J_{\mathcal{X}}\left(I-Q_{\mu_{n_k,i}}^{C_i}\right) T w_{n_k}\|^2} \to 0, \text{ as } k \to \infty \quad \forall i = 1, 2, \dots, m.$$

Since $0 < a \leq \tau_{n_k} \leq b < 2$, $\liminf_{k \to \infty} \beta_{n_k,0} \beta_{n_k,i} > 0$ and $\|T^* J_{\mathcal{X}} \left(I - Q_{\mu_{n_k,i}}^{C_i}\right) T w_{n_k}\|$ is bounded for all $i = 1, 2, \cdots, m$, we have that

$$\lim_{k \to \infty} \| J_{\mathcal{X}} \left(I - Q_{\mu_{n_k,i}}^{C_i} \right) T w_{n_k} \| = 0 \quad \forall i = 1, 2, \cdots, m.$$
(6.3.16)

Consequently, for $i = 1, 2, \cdots, m$ we get

$$\|T^* J_{\mathcal{X}} \left(I - Q_{\mu_{n_k,i}}^{C_i} \right) T w_{n_k} \| \leq \|T^*\| \|J_{\mathcal{X}} \left(I - Q_{\mu_{n_k,i}}^{C_i} \right) T w_{n_k} \|$$

$$= \|T\| \|J_{\mathcal{X}} \left(I - Q_{\mu_{n_k,i}}^{C_i} \right) T w_{n_k} \| \to 0, \quad k \to \infty.$$
(6.3.17)

Following similar argument, from Lemma 6.3.3 we obtain

$$\|A_i z_{n_k,i} - A_i \bar{x}\| \to 0 \quad k \to \infty \quad \forall i = 1, 2, \dots, m,$$

$$(6.3.18)$$

and

$$\|J_{r_{n_k},i}^{B_i}(I - r_{n_k,i}A_i) z_{n_k,i} - w_{n_k}\| \to 0 \quad k \to \infty \quad \forall i = 1, 2, \dots, m.$$
(6.3.19)

Also, from 6.3.4 we obtain

$$\delta_{n_k} \gamma_{n_k} \|x_{n_k} - Su_{n_k}\|^2 \le \alpha_{n_k} \|f(x_{n_k}) - \bar{x}\|^2 + (1 - \alpha_{n_k}) \|x_{n_k} - \bar{x}\|^2 - \|x_{n_{k+1}} - \bar{x}\|^2 + 3M_2 \gamma_{n_k} \alpha_{n_k} \frac{\theta_{n_k}}{\alpha_{n_k}} \|x_{n_k} - x_{n_{k-1}}\|.$$

By Remark 6.3.1, (6.3.15) and the conditions on the control parameters, we obtain

$$\|x_{n_k} - Su_{n_k}\| \to \quad k \to \infty. \tag{6.3.20}$$

From Step 3 and (6.3.17), we have

$$\|z_{n_{k},i} - w_{n_{k}}\| = \|w_{n_{k}} - \lambda_{n_{k},i}T^{*}J_{\mathcal{X}}\left(I - Q_{\mu_{n_{k},i}}^{C_{i}}\right)Tw_{n_{k}} - w_{n_{k}}\|$$

$$= \lambda_{n_{k},i}\|T^{*}J_{\mathcal{X}}\left(I - Q_{\mu_{n_{k},i}}^{C_{i}}\right)Tw_{n_{k}}\| \to 0, \qquad k \to \infty \quad \forall i = 1, 2, \dots, m.$$

(6.3.21)

Also, from Step 4 and (6.3.19) we get

$$\|u_{n_k} - w_{n_k}\| \le \beta_{n_k,0} \|w_{n_k} - w_{n_k}\| + \sum_{i=1}^m \beta_{n_k,i} \|J_{r_{n_k},i}^{B_i} (I - r_{n_k,i}A_i) z_{n_k,i} - w_{n_k}\| \to 0, \text{ as } k \to \infty.$$
(6.3.22)

From (6.3.21) and (6.3.22), we obtain

$$||z_{n_k,i} - u_{n_k}|| \to 0$$
, as $k \to \infty$.

Also from Remark 6.3.1, we have

$$||w_{n_k} - x_{n_k}|| = \theta_{n_k} ||x_{n_k} - x_{n_k-1}|| \to 0, \text{ as } k \to \infty.$$
(6.3.23)

From (6.3.21), (6.3.22) and (6.3.23) we have

$$||z_{n_k,i} - x_{n_k}|| \to 0, \quad ||u_{n_k} - x_{n_k}|| \to 0, \quad \text{as } k \to \infty.$$
 (6.3.24)

Also, from (6.3.20) and (6.3.24) we obtain

$$\|u_{n_k} - Su_{n_k}\| \to 0, \quad k \to \infty.$$

$$(6.3.25)$$

By applying (6.3.20) and the condition on α_n , from Step 5 we obtain

$$\begin{aligned} \|x_{n_k+1} - x_{n_k}\| &= \|\alpha_{n_k} f(x_{n_k}) + \delta_{n_k} x_{n_k} + \gamma_{n_k} S u_{n_k} - x_{n_k}\| \\ &\leq \alpha_{n_k} \|f(x_{n_k}) - x_{n_k}\| + \delta_{n_k} \|x_{n_k} - x_{n_k}\| + \gamma_{n_k} \|S u_{n_k} - x_{n_k}\| \to 0, \text{ as } k \to \infty \end{aligned}$$

$$(6.3.26)$$

To complete the proof, we need to establish that $w_{\omega}(x_n) \subset \Gamma$. Let $x^* \in w_{\omega}(x_n)$ be an arbitrary element. Then, there exists a subsequence $\{x_{n_k}\}$ of $\{x_n\}$ such that $x_{n_k} \rightharpoonup x^*$ as $k \rightarrow \infty$. From (6.3.23), we have that $w_{n_k} \rightharpoonup x^*$ as $k \rightarrow \infty$. Since T is bounded and linear we have

$$Tw_{n_k} \rightharpoonup Tx^*. \tag{6.3.27}$$

From (6.3.16), we have

$$\lim_{k \to \infty} \| \left(I - Q_{\mu_{n_k},i}^{C_i} \right) T w_{n_k} \| = \lim_{k \to \infty} \| J_{\mathcal{X}} \left(I - Q_{\mu_{n_k},i}^{C_i} \right) T w_{n_k} \| = 0 \quad \forall i = 1, 2, \dots, m.$$
(6.3.28)

This together with (6.3.27) implies that $Q_{\mu_{n_k},i}^{C_i}Tw_{n_k} \rightharpoonup Tx^*$, as $k \rightarrow \infty$ for each $i = 1, 2, \ldots, m$. Since $Q_{\mu_{n_k},i}^{C_i}$ is the metric of C_i for $\mu_{n_k,i} > 0$, we have that

$$\frac{1}{\mu_{n_k,i}} J_{\mathcal{X}} \left(I - Q_{\mu_{n_k},i}^{C_i} \right) T w_{n_k} \in C_i Q_{\mu_{n_k},i}^{C_i} T w_{n_k}$$

for all $i = 1, 2, \dots, m$. Let $v \in X$. By the monotonicity of each C_i , it follows that

$$0 \le \left\langle v - Q_{\mu_{n_k,i}}^{C_i} T w_{n_k}, v^* - \frac{1}{\mu_{n_k,i}} J_{\mathcal{X}} \left(I - Q_{\mu_{n_k,i}}^{C_i} \right) T w_{n_k} \right\rangle,$$

for all $v^* \in C_i(v)$. Passing limit as $k \to \infty$, and by applying (6.3.28) together with the fact that $0 < c \leq \mu_{n,i}$, we obtain $0 \leq \langle v - Tx^*, v^* - 0 \rangle$ for all $v^* \in C_i(v)$, i = 1, 2, ..., m. Since each C_i is maximal monotone, we have that $Tx^* \in C_i^{-1}(0)$, for all i = 1, 2, ..., m. This implies that

$$x^* \in \bigcap_{i=1}^m T^{-1}(C_i^{-1}(0)).$$
(6.3.29)

Next, we show that $x^* \in \bigcap_{i=1}^m (A_i + B_i)^{-1}(0)$. Let $T_{r_{n,i}} = J_{r_{n,i}}^{B_i}(I - r_{n,i}A_i)$, then by applying (6.3.19) and (6.3.21) we have

$$||T_{r_{n_k,i}}z_{n_k,i} - z_{n_k,i}|| \le ||T_{r_{n_k,i}}z_{n_k,i} - w_{n_k}|| + ||w_{n_k} - z_{n_k,i}|| \to 0, \quad k \to \infty \quad \forall i = 1, 2, \dots, m.$$
(6.3.30)

for all i = 1, 2, ..., m.

By the condition on $r_{n,i}$, there exists $r_i > 0$ such that $r_{n_k,i} \ge r_i$ for all $k \ge 1$ and i = 1, 2..., m. Applying Lemma 2.5.20(ii), we have

$$\lim_{k \to \infty} \|T_{r_{n_k,i}} z_{n_k,i} - z_{n_k,i}\| \le 2 \lim_{k \to \infty} \|T_{r_{n_k,i}} z_{n_k,i} - z_{n_k,i}\| = 0 \quad \forall i = 1, 2, \dots$$

We know that T_{r_i} is nonexpansive and from (6.3.24) $z_{n_k,i} \rightharpoonup x^*$ for all $i = 1, 2, \ldots, m$. By the demiclosedness of $I - T_{r_i}$, we have that $x^* \in F(T_{r_i})$ for all $i = 1, 2, \ldots, m$. By Lemma 2.5.20(i) we have that $x^* \in (A_i + B_i)^{-1}(0)$ for all $i = 1, 2, \ldots, m$. This implies that

$$x^* \in \bigcap_{i=1}^m (A_i + B_i)^{-1}(0).$$
 (6.3.31)

Next, we show that $x^* \in F(S)$. From (6.3.24) we have that $u_{n_k} \rightharpoonup x^*$. By (6.3.25) and the demiclosedness property of I - S at zero, we have that $x^* \in F(S)$. Hence, by (6.3.29) and (6.3.31) we have that $w_{\omega}(x_n) \subset \Gamma$.

From (6.3.24) we have $w_{\omega}\{z_{n,i}\} = w_{\omega}\{x_n\}$. By the boundedness of $\{x_{n_k}\}$ there exists a subsequence $\{x_{n_{k_j}}\}$ of $\{x_{n_k}\}$ such that $x_{n_{k_j}} \rightharpoonup x^{\dagger}$ and

$$\lim_{j \to \infty} \langle f(\bar{x}) - \bar{x}, x_{n_{k_j}} - \bar{x} \rangle = \limsup_{k \to \infty} \langle f(\bar{x}) - \bar{x}, x_{n_k} - \bar{x} \rangle$$
$$= \limsup_{k \to \infty} \langle f(\bar{x}) - \bar{x}, z_{n_k, i} - \bar{x} \rangle.$$
(6.3.32)

Since $\bar{x} = P_{\Gamma} \circ f(\bar{x})$, then it follows that

$$\begin{split} \limsup_{k \to \infty} \langle f(\bar{x}) - \bar{x}, x_{n_k} - \bar{x} \rangle &= \lim_{j \to \infty} \langle f(\bar{x}) - \bar{x}, x_{n_{k_j}} - \bar{x} \rangle \\ &= f(\bar{x}) - \bar{x}, x^{\dagger} - \bar{x} \rangle \\ &\leq 0. \end{split}$$
(6.3.33)

Thus, from (6.3.26) and (6.3.33) we have

$$\limsup_{k \to \infty} \langle f(\bar{x}) - \bar{x}, x_{n_k+1} - \bar{x} \rangle = \limsup_{k \to \infty} \langle f(\bar{x}) - \bar{x}, x_{n_k+1} - x_{n_k} \rangle + \limsup_{k \to \infty} \langle f(\bar{x}) - \bar{x}, x_{n_k} - \bar{x} \rangle$$
$$= f(\bar{x}) - \bar{x}, x^* - \bar{x} \rangle$$
$$\leq 0. \tag{6.3.34}$$

Applying Lemma 2.5.36 to (6.3.14), and by Remark 6.3.1, (6.3.34) together with the fact that $\lim_{n\to\infty} \alpha_n = 0$ we have $\lim_{n\to\infty} ||x_n - \bar{x}|| = 0$ as required. This completes the proof. \Box

Taking $A_i = 0$ for all i = 1, 2, ..., m, we obtain the following consequent result.

Algorithm 6.3.2.

Step 0: Select initial points $x_0, x_1 \in \mathcal{H}$ and set n = 1. **Step 1:** Given the iterates x_{n-1} and x_n for each $n \ge 1$, choose θ_n such that $0 \le \theta_n \le \overline{\theta}_n$, where

$$\bar{\theta}_n := \begin{cases} \min\left\{\theta, \frac{\epsilon_n}{\|x_n - x_{n-1}\|}\right\}, & \text{if } x_n \neq x_{n-1} \\ \theta, & \text{otherwise.} \end{cases}$$
(6.3.35)

Step 2: Compute

$$w_n = x_n + \theta_n (x_n - x_{n-1}).$$

Step 3: Compute

$$z_{n,i} = w_n - \lambda_{n,i} T^* J_{\mathcal{X}} (I - Q^{C_i}_{\mu_{n,i}}) T w_n,$$

where

$$\lambda_{n,i} = \frac{\tau_n \|J_{\mathcal{X}}(I - Q_{\mu_{n,i}}^{C_i})Tw_n\|^2}{\|T^* J_{\mathcal{X}}(I - Q_{\mu_{n,i}}^{C_i})Tw_n\|^2}.$$

Step 4: Compute

$$u_n = \beta_{n,0} w_n + \sum_{i=1}^m \beta_{n,i} J_{r_{n,i}}^{B_i} z_{n,i}.$$

Step 5: Compute

$$x_{n+1} = \alpha_n f(x_n) + \delta_n x_n + \gamma_n S u_n.$$

Set n := n + 1 and return to Step 1.

Corollary 6.3.3. Let \mathcal{H} be a Hilbert space, \mathcal{X} a uniformly convex and smooth Banach space and $J_{\mathcal{X}}$ the duality mapping on \mathcal{X} . Let $\{x_n\}$ be generated by Algorithm 6.3.2 and suppose Assumption (A1-A4) are satisfied. Then, the sequence $\{x_n\}$ converges strongly to a point $\bar{x} \in \Omega$, where $\bar{x} = P_{\Omega} \circ f(\bar{x})$ and $\Omega = \bigcap_{i=1}^{m} B_i^{-1}(0) \cap \bigcap_{i=1}^{m} T^{-1}(C_i^{-1}0) \cap F(S)$

6.3.3 Application

In this section, we apply our result to SFP and SMP. **Split feasibility problem**

Let \mathcal{H}_1 and \mathcal{H}_2 be two real Hilbert spaces and let \mathcal{D} and \mathcal{Q} be nonempty closed convex subsets of \mathcal{H}_1 and \mathcal{H}_2 , respectively. The SFP is defined as follows:

find
$$x^* \in \mathcal{D}$$
 such that $Tx^* \in \mathcal{Q}$, (6.3.36)

where $T: \mathcal{H}_1 \to \mathcal{H}_2$ is a bounded linear operator.

Let Q be a nonempty closed convex subset of a real Hilbert space \mathcal{H} and i_Q be the indicator function on Q, that is

$$i_Q(x) = \begin{cases} 0 & \text{if } x \in Q; \\ \infty & \text{if } x \notin Q. \end{cases}$$

Moreover, we define the normal cone $N_Q u$ of Q at $u \in Q$ as follows:

$$N_Q u = \{ z \in \mathcal{H} : \langle z, v - u \rangle \le 0, \forall v \in Q \}.$$

It is known that i_Q is a proper, lower semicontinuous and convex function on \mathcal{H} . Hence, the subdifferential ∂i_Q of i_Q is a maximal monotone operator. Therefore, we define the resolvent $J_r^{\partial i_Q}$ of ∂i_Q , $\forall r > 0$ as follows:

$$J_r^{\partial i_Q} x = (I + r \partial i_Q)^{-1} x, \forall x \in \mathcal{H}.$$

Moreover, for each $x \in Q$, we have

$$\partial i_Q x = \{ z \in \mathcal{H} : i_Q x + \langle z, u - x \rangle \le i_Q u, \forall u \in \mathcal{H} \}$$

= $\{ z \in \mathcal{H} : \langle z, u - x \rangle \le 0, \forall u \in Q \}$
= $N_Q x$.

Hence, for all $\alpha > 0$, we derive

$$\begin{split} u &= J_r^{\partial i_Q} x \Leftrightarrow x \in u + r \partial i_Q u \\ \Leftrightarrow x - u \in r \partial i_Q u \\ \Leftrightarrow \langle x - u, z - u \rangle \leq 0 \quad \forall z \in Q \\ \Leftrightarrow u &= P_O x. \end{split}$$

Now, by applying Corollary 6.3.3 using the case for which the sequences $\{\mu_{n,i}\}$ and $\{r_{n,i}\}$ are taken as constant sequences $\{\mu_i\}$ and $\{r_i\}$, respectively for each i = 1, 2, ..., m. We obtain the following result for approximating a common solution of finite family of SFPs and FPP for a nonexpansive mapping between Hilbert and Banach spaces.

Theorem 6.3.6. Let D_i and Q_i be finite families of nonempty, closed and convex subsets of a Hilbert space \mathcal{H} and a uniformly convex and smooth Banach space \mathcal{X} , respectively for i = 1, 2, ..., m. Let $J_{\mathcal{X}}$ be the duality mapping on \mathcal{X} . Let $T : \mathcal{H} \to \mathcal{X}$ be a bounded linear operator such that $T \neq 0$ and T^* is the adjoint operator of T. Let $f : \mathcal{H} \to \mathcal{H}$ be a contraction mapping with $\rho \in (0, 1)$ and $0 < \gamma < \frac{\gamma}{\rho}$. Let $S : \mathcal{H} \to \mathcal{H}$ be a nonexpansive mapping and suppose that the solution set $\Gamma := \mathcal{F} \cap F(S) \neq \emptyset$, where $\mathcal{F} = \{x^* \in \bigcap_{i=1}^m D_i :$ $Tx^* \in \bigcap_{i=1}^m Q_i\}$. Let $\{x_n\}$ be a sequence generated as follows:

Algorithm 6.3.4.

Step 0: Select initial points $x_0, x_1 \in \mathcal{H}$ and set n = 1.

Step 1: Given the iterates x_{n-1} and x_n for each $n \ge 1$, choose θ_n such that $0 \le \theta_n \le \overline{\theta}_n$, where

$$\bar{\theta}_n := \begin{cases} \min\left\{\theta, \frac{\epsilon_n}{\|x_n - x_{n-1}\|}\right\}, & \text{if } x_n \neq x_{n-1} \\ \theta, & \text{otherwise.} \end{cases}$$
(6.3.37)

Step 2: Compute

$$w_n = x_n + \theta_n (x_n - x_{n-1}).$$

Step 3: Compute

$$z_{n,i} = w_n - \lambda_{n,i} T^* J_{\mathcal{X}} (I - P_{Q_i}) T w_n$$

where

$$\lambda_{n,i} = \frac{\tau_n \|J_{\mathcal{X}}(I - P_{Q_i})Tw_n\|^2}{\|T^* J_{\mathcal{X}}(I - P_{Q_i})Tw_n\|^2}.$$

Step 4: Compute

$$u_n = \beta_{n,0} w_n + \sum_{i=1}^m \beta_{n,i} P_{D_i} z_{n,i}.$$

Step 5: Compute

$$x_{n+1} = \alpha_n f(x_n) + \delta_n x_n + \gamma_n S u_n.$$

Set n := n + 1 and return to **Step 1**.

Suppose Assumption (A1-A4) are satisfied. Then the sequence $\{x_n\}$ generated by Algorithm 6.3.4 converges strongly to a point $\bar{x} \in \Gamma$, where $\bar{x} = P_{\Gamma} \circ f(\bar{x})$.

Split minimization problem

Let \mathcal{H}_1 and \mathcal{H}_2 be real Hilbert spaces, $T : \mathcal{H}_1 \to \mathcal{H}_2$ be a bounded linear operator.

Given some proper, lower semicontinuous and convex functions $f_1 : \mathcal{H}_1 \to \mathbb{R} \cup \{+\infty\}$ and $f_2 : \mathcal{H}_2 \to \mathbb{R} \cup \{+\infty\}$, the SMP is defined as: find $x^* \in \mathcal{H}_1$ such that

$$x^* \in \operatorname{argmin}_{x \in \mathcal{H}_1} f_1(x)$$
 and $Tx^* \in \operatorname{argmin}_{y \in \mathcal{H}_2} f_2(y).$ (6.3.38)

Moudafi and Thakur [182] first introduced the SMP which has attracted a lot of attention from researchers in recent years (see [1, 182, 268] and the references therein). The SMP has been applied to the study of many problems in applied science which includes; Fourier regularization, multi-resolution sparse regularization, alternating projection signal synthesis problems, amongst others.

In a real Hilbert space \mathcal{H} , the proximal operator of f is defined by

$$prox_{\lambda,f}(x) := \operatorname{argmin}_{z \in \mathcal{H}} \left\{ f(z) + \frac{1}{2\lambda} \|x - z\|^2 \right\} \quad \forall \ x \in \mathcal{H}, \ \lambda > 0.$$

It is well known that

$$prox_{\lambda,f}(x) = (I + \lambda \partial f)^{-1}(x) = J_{\lambda}^{\partial f}(x), \qquad (6.3.39)$$

where ∂f is the subdifferential of f defined by

$$\partial f(x) = \{ z \in \mathcal{H} : f(x) - f(y) \le \langle z, x - y \rangle, \forall y \in \mathcal{H} \},\$$

for each $x \in \mathcal{H}$. From [50], ∂f is a maximal monotone operator and $prox_{\lambda,f}$ is firmly nonexpansive.

By setting $B_i = \partial f_i$ and $C_i = \partial g_i$ in Corollary 6.3.3 for each i = 1, 2, ..., m, we obtain the following result for approximating a common solution of finite family of SMP and FPP for nonexpansive mapping between Hilbert and Banach spaces.

Theorem 6.3.7. Let \mathcal{H} be a real Hilbert space and let \mathcal{X} be a uniformly convex and smooth Banach space. Let $J_{\mathcal{X}}$ be the duality mapping on \mathcal{X} . Let $T : \mathcal{H} \to \mathcal{X}$ be a bounded linear operator such that $T \neq 0$ and T^* is the adjoint operator of T. Let $f : \mathcal{H} \to \mathcal{H}$ be a contraction mapping with $\rho \in (0,1)$ and $0 < \gamma < \frac{\gamma}{\rho}$, and Let $S : \mathcal{H} \to \mathcal{H}$ be a nonexpansive mapping. Let $f_i : \mathcal{H} \to \mathbb{R} \cup \{+\infty\}$ and $g_i : \mathcal{X} \to \mathbb{R} \cup \{+\infty\}$ be finite families of proper convex lower semicontinuous functions for i = 2, ..., m. Suppose that the solution set $\Gamma := \mathcal{F} \bigcap F(S) \neq \emptyset$, where $\mathcal{F} = \{x^* \in \bigcap_{i=1}^m \operatorname{argmin} f_i : Tx^* \in \bigcap_{i=1}^m \operatorname{argmin} g_i\}$. Let $\{x_n\}$ be a sequence generated as follows:

Algorithm 6.3.5.

Step 0: Select initial points $x_0, x_1 \in \mathcal{H}$ and set n = 1.

Step 1: Given the iterates x_{n-1} and x_n for each $n \ge 1$, choose θ_n such that $0 \le \theta_n \le \overline{\theta}_n$, where

$$\bar{\theta}_n := \begin{cases} \min\left\{\theta, \frac{\epsilon_n}{\|x_n - x_{n-1}\|}\right\}, & \text{if } x_n \neq x_{n-1} \\ \theta, & \text{otherwise.} \end{cases}$$
(6.3.40)

Step 2: Compute

$$w_n = x_n + \theta_n (x_n - x_{n-1}).$$

Step 3: Compute

$$z_{n,i} = w_n - \lambda_{n,i} T^* J_{\mathcal{X}} (I - prox_{\mu_{n,i},g_i}) T w_n$$

where

$$\lambda_{n,i} = \frac{\tau_n \|J_{\mathcal{X}}(I - prox_{\mu_{n,i},g_i})Tw_n\|^2}{\|T^*J_{\mathcal{X}}(I - prox_{\mu_{n,i},g_i})Tw_n\|^2}.$$

Step 4: Compute

$$u_n = \beta_{n,0}w_n + \sum_{i=1}^m \beta_{n,i} prox_{r_{n,i},f_i} z_{n,i}$$

Step 5: Compute

$$x_{n+1} = \alpha_n f(x_n) + \delta_n x_n + \gamma_n S u_n$$

Set n := n + 1 and return to **Step 1**.

Suppose Assumption (A1-A4) are satisfied. Then the sequence $\{x_n\}$ generated by Algorithm 6.3.5 converges strongly to a point $\bar{x} \in \Gamma$, where $\bar{x} = P_{\Gamma} \circ f(\bar{x})$.

6.3.4 Numerical experiments

In this section, we provide a numerical example in infinite dimensional spaces to demonstrate the performance of Algorithm 6.3.1 and then compare it with the method of Kazmi and Riziv [141, Theorem 3.1] (see Appendix 6.3.9), the method of Sitthithakerngkiet [218, Theorem 3.4] (see Appendix 6.3.10), the method of Long *et al.* [163, Algorithm 5] (see Appendix 6.3.11) and the method of Byrne *et al.* [51, Algorithm 4.4] (see Appendix 6.3.12).

We perform all implementations using Matlab 2016 (b), installed on a personal computer with Intel(R) Core(TM) i5-2600 CPU@2.30GHz and 8.00 Gb-RAM running on Windows 10 operating system. In Tables 6.3.1, "Iter." means the number of iterations while "CPU" means the CPU time in seconds.

Example 6.3.8. Let $H = (l_2(\mathbb{R}, \|\cdot\|_2)) = \mathcal{X}$, where

$$l_2(\mathbb{R}) := \Big\{ x = (x_1, x_2, \cdots, x_n, \cdots,), x_j \in \mathbb{R} : \sum_{j=1}^{\infty} |x_j|^2 < \infty \Big\},\$$

 $\|x\|_{2} = \left(\sum_{j=1}^{\infty} |x_{j}|^{2}\right)^{\frac{1}{2}} \text{ for all } x \in l_{2}(\mathbb{R}). \text{ Let } T : \mathcal{H} \to \mathcal{X} \text{ be defined by } Tx = \frac{3}{2}x. \text{ For } i = 1, 2, \cdots, 5, \text{ define } A_{i} : \mathcal{H} \to \mathcal{H} \text{ as } A_{i}x = \frac{x}{2i}, B_{i} : \mathcal{H} \to \mathcal{H} \text{ by } B_{i}x = \frac{3}{2i}x, C_{i} : \mathcal{X} \to \mathcal{X} \text{ by } C_{i}x = \frac{5}{2i}x. \text{ Then, } A_{i} \text{ is a finite family of inverse strongly monotone operators, } B_{i} \text{ and } C_{i} \text{ are maximal monotone operators for each } i = 1, 2, \cdots, 5. \text{ Let } S : \mathcal{H} \to \mathcal{H} \text{ be defined by } Sx = \frac{1}{2}x. \text{ Observe that } S \text{ is nonexpansive. We set } f(x) = \frac{x}{3}, \beta_{n,0} = \frac{5n}{2(3n+2)}, \beta_{n,i} = \frac{n+4}{10(3n+2)}, \alpha_{n} = \frac{1}{3n+1}, \delta_{n} = \frac{n}{3n+1}, \gamma_{n} = \frac{2n}{3n+1}, \epsilon_{n} = \frac{1}{(3n+1)^{3}}, \tau_{n} = \frac{n}{2n+1}, \mu_{n,i} = r_{n,i} = r = 0.01$

and $\theta = 0.6$ in Algorithm 6.3.1 for each $n \in \mathbb{N}$. We take $S_n(x) = \frac{1}{2^n}x$, D = I and $\mathcal{E} = 1$ in Appendix 6.3.9, $\theta_n = \frac{1}{(3n+1)^2}$ in Appendix 6.3.11 and $\lambda = \lambda_n = 0.05$.

We consider the following cases for the numerical experiments of this example. **Case I:** Take $x_0 = (97, 74, -68, \cdots)$ and $x_1 = (-\frac{1}{3}, -\frac{1}{6}, \frac{1}{19} \cdots)$. **Case II:** Take $x_0 = (-66, 584, 99, \cdots)$ and $x_1 = (0.1, -0.03, 0.57, \cdots)$. **Case III:** Take $x_0 = (79, 731, 49, \cdots)$ and $x_1 = (0.7, \frac{1}{43}, -\frac{7}{39}, \cdots)$. **Case IV:** Take $x_0 = (513, -27, 88, \cdots)$ and $x_1 = (\frac{3}{190}, \frac{23}{77}, -\frac{1}{2}, \cdots)$. We compare the performance of the algorithms using the stopping criterion $||x_{n+1} - x_n|| < 10^{-2}$. We plot the graphs of errors against the number of iterations in each case. The

		App.	App.	App.	App.	Alg.
		6.3.9	6.3.10	6.3.11	6.3.12	6.3.1
Case I	CPU time	0.0151	0.0062	0.0089	0.0091	0.0116
	(sec)					
	No. of Iter.	14	6	9	13	6
Case II	CPU time	0.0102	0.0091	0.0079	0.0055	0.0102
	(sec)					
	No. of Iter.	16	7	11	13	7
Case III	CPU time	0.0103	0.0055	0.0085	0.0061	0.0103
	(sec)					
	No. of Iter.	16	7	12	13	7
Case IV	CPU time	0.0101	0.0056	0.0079	0.0058	0.0106
	(sec)					
	No. of Iter.	15	7	11	13	7

Appendix 6.3.9. The Algorithm in [141].

Initialization: Given $\lambda \in (0, \frac{1}{L}), r > 0$. Let $x_0 \in \mathcal{H}_1$ be arbitrary.

numerical results are reported in Table 6.3.1 and Figure 6.3.

Iterative Steps: Calculate x_{n+1} as follows:

Step1. Compute

$$u_n = J_r^B \Big(x_n + \lambda T^* (J_r^C - I) T x_n \Big).$$

Step 2: Compute

$$x_{n+1} = \alpha_n f(x_n) + (1 - \alpha_n) S u_n,$$

where $f : \mathcal{H} \to \mathcal{H}$ is a contraction mapping with constant $\rho \in (0,1), S$ is a nonexpansive mapping such that $F(S) \cap \Gamma \neq \emptyset$.

Set n := n + 1 and return to Step 1.



Figure 6.3: Top left: Case I; Top right: Case II; Bottom left: Case III; Bottom right: Case IV.
Appendix 6.3.10. The Algorithm in [218].

Initialization: Given $\lambda \in (0, \frac{1}{L}), r > 0$. Let $x_0 \in \mathcal{H}_1$ be arbitrary.

Iterative Steps: Calculate x_{n+1} as follows:

Step1. Compute

$$y_n = J_r^B \Big(x_n + \lambda T^* (J_r^C - I) T x_n \Big).$$

Step 2: Compute

$$x_{n+1} = \alpha_n \mathcal{E}f(x_n) + (I - \alpha_n D)S_n y_n,$$

where $f : \mathcal{H} \to \mathcal{H}$ is a contraction mapping with constant $\alpha \in (0,1), \{S_n\}$ is a sequence of nonexpansive mappings such that $F(S_n) \cap \Gamma \neq \emptyset$, D is strongly positive bounded linear operator with coefficient $\bar{\gamma} > 0$ and $0 < \mathcal{E} < \frac{\bar{\gamma}}{\beta^*}$.

Set n := n + 1 and return to Step 1.

Appendix 6.3.11. The Algorithm in [163].

Initialization: Let r > 0.

Iterative Steps: Calculate x_{n+1} as follows: Let $\{x_n\}$ be a sequence in \mathcal{H}_1 defined by

$$\begin{cases} x_0, x_1 \in \mathcal{H}, \\ w_n = x_n + \theta_n (x_n - x_{n-1}), \\ y_n = J_r^B (I - \lambda_n T^* (I - J_r^C) T) w_n, \\ x_{n+1} = \alpha_n f(x_n) + (1 - \alpha_n) y_n. \end{cases}$$
(6.3.41)

where $f : \mathcal{H} \to \mathcal{H}$ is a contraction mapping with constant $\alpha \in (0, 1)$. Set n := n + 1 and return to Step 1.

Appendix 6.3.12. The Algorithm in [51].

$$\begin{cases} v \in H_1, \\ x_{n+1} = \alpha_n v + (1 - \alpha_n) J_r^B(x_n - \lambda T^*(I - J_r^C) T x_n), \ n \in \mathbb{N}, \end{cases}$$
(6.3.42)

CHAPTER 7

Results on Minimization Problems in Hadamard Spaces

7.1 Introduction

The extension of known concepts from Hilbert and Banach spaces to Hadamard spaces has been of great research interest to several researchers. In this chapter, we extend our results from the frameworks of the Hilbert and Banach spaces to the framework of an Hadamard space. We present our results on modified proximal point methods and viscosity implicit rule involving quasi-pseudocontractive mappings in Hadamard spaces. Furthermore, we present some numerical examples of our methods in Hadamard spaces which are not Hilbert spaces and compare them with other results in literature to show the applicability of our methods.

7.2 Preliminaries

Lemma 7.2.1. Let X be an Hadamard space and $T: X \to X$ be an L-Lipschitz mapping with $L \ge 1$ such that for all $x \in X$,

$$Kx := (1 - \epsilon)x \oplus \epsilon T((1 - \gamma)x \oplus \gamma Tx).$$
(7.2.1)

If $0 < \epsilon < \gamma < \frac{1}{1+\sqrt{1+L^2}}$, then the following conditions hold:

- (a) $x = Tx \iff x = T((1 \gamma)x \oplus \gamma Tx) \iff x = Kx;$
- (b) if T is \triangle -demiclosed, then K is also \triangle -demiclosed;
- (c) if T is a quasi-pseudocontractive mapping, then K is a quasi-nonexpansive mapping.

Proof. (a) Let x = Tx. Then, one can easily see that $x = T((1 - \gamma)x \oplus \gamma Tx)$. Conversely, let $x = T((1 - \gamma)x \oplus \gamma Tx)$ and define $S_{\gamma}x := (1 - \gamma)x \oplus \gamma Tx$. Then $TS_{\gamma}x = x$. Thus, we obtain from (2.3.1) that

$$d(x, S_{\gamma}x) = \gamma d(x, Tx) \le \gamma L d(S_{\gamma}x, x).$$

Since $0 < L\gamma < 1$, we obtain that $x = S_{\gamma}x$. This implies that $x = TS_{\gamma}x = Tx$. By similar argument, we obtain that $x = T((1 - \gamma)x \oplus \gamma Tx) \iff x = Kx$.

(b) Let T be \triangle -demiclosed and $\{x_n\}$ be a sequence in X such that $\triangle \lim_{n \to \infty} x_n = x$ and $\lim_{n \to \infty} d(x_n, Kx_n) = 0$, then we have that

$$d(x_n, Tx_n) \leq d(x_n, Kx_n) + d((1 - \epsilon)x_n \oplus \epsilon T((1 - \gamma)x_n \oplus \gamma Tx_n), Tx_n)$$

$$\leq d(x_n, Kx_n) + (1 - \epsilon)d(x_n, Tx_n) + \epsilon d(T((1 - \gamma)x_n \oplus \gamma Tx_n), Tx_n)$$

$$\leq d(x_n, Kx_n) + (1 - \epsilon)d(x_n, Tx_n) + \epsilon Ld((1 - \gamma)x_n \oplus \gamma Tx_n, x_n)$$

$$= d(x_n, Kx_n) + [(1 - \epsilon) + \epsilon L\gamma] d(Tx_n, x_n),$$

which implies that

$$d(x_n, Tx_n) \le \frac{1}{\epsilon(1 - L\gamma)} d(x_n, Kx_n) \to 0.$$

Since T is demiclosed, we get that x = Tx. By (a), we obtain the desired conclusion. (c) Let $y \in F(K)$ and $x \in X$. Then $y \in F(T)$. Hence, we obtain from (2.3.4) that $d^{2}(T((1-\gamma)x \oplus \gamma Tx), y) \leq d^{2}((1-\gamma)x \oplus \gamma Tx, y) + d^{2}((1-\gamma)x \oplus \gamma Tx, T((1-\gamma)x \oplus \gamma Tx)))$ $\leq (1-\gamma)d^{2}(x, y) + \gamma d^{2}(Tx, y) - \gamma(1-\gamma)d^{2}(x, Tx) + d^{2}((1-\gamma)x \oplus \gamma Tx, T((1-\gamma)x \oplus \gamma Tx)))$ $\leq (1-\gamma)d^{2}(x, y) + \gamma d^{2}(x, y) + \gamma d^{2}(x, Tx) - \gamma(1-\gamma)d^{2}(x, Tx) + d^{2}((1-\gamma)x \oplus \gamma Tx, T((1-\gamma)x \oplus \gamma Tx)))$ $= d^{2}(x, y) + \gamma^{2}d^{2}(x, Tx) + d^{2}((1-\gamma)x \oplus \gamma Tx)).$ (7.2.2)

But,

$$d^{2}((1-\gamma)x \oplus \gamma Tx, T((1-\gamma)x \oplus \gamma Tx)) \leq (1-\gamma)d^{2}(x, T((1-\gamma)x \oplus \gamma Tx)) + \gamma d^{2}(Tx, T((1-\gamma)x \oplus \gamma Tx)) - \gamma (1-\gamma)d^{2}(x, Tx) \leq (1-\gamma)d^{2}(x, T((1-\gamma)x \oplus \gamma Tx)) + \gamma L^{2}d^{2}(x, (1-\gamma)x \oplus \gamma Tx) - \gamma (1-\gamma)d^{2}(x, Tx) \leq (1-\gamma)d^{2}(x, T((1-\gamma)x \oplus \gamma Tx)) - \gamma (1-\gamma)d^{2}(x, Tx) \leq (1-\gamma)d^{2}(x, T((1-\gamma)x \oplus \gamma Tx))$$
(7.2.3)
$$- (\gamma - \gamma^{2} - \gamma^{3}L^{3})d^{2}(x, Tx).$$

Now, from the condition on γ and L, we obtain that $1 - 2\gamma - \gamma^2 L^2 > 0$. Hence, we get from (7.2.2) and (7.2.3) that

$$d^{2}(T((1-\gamma)x \oplus \gamma Tx), y) \leq d^{2}(x, y) + (1-\gamma)d^{2}(x, T((1-\gamma)x \oplus \gamma Tx)) \quad (7.2.4)$$

- $\gamma(1-2\gamma-\gamma^{2}L^{2})d^{2}(x, Tx).$
 $\leq d^{2}(x, y) + (1-\gamma)d^{2}(x, T((1-\gamma)x \oplus \gamma Tx)). \quad (7.2.5)$

Hence, we obtain from (7.2.4) that

$$d^{2}(Kx,y) \leq (1-\epsilon)d^{2}(x,y) + \epsilon d^{2}(T((1-\gamma)x \oplus \gamma Tx),y) -\epsilon(1-\epsilon)d^{2}(x,T((1-\gamma)x \oplus \gamma Tx)) \leq d^{2}(x,y) - \epsilon(\gamma-\epsilon)d^{2}(x,T((1-\gamma)x \oplus \gamma Tx)) \leq d^{2}(x,y).$$

Lemma 7.2.2. Let X be an Hadamard space and $f: X \to (-\infty, \infty]$ be a proper convex and lower semi-continuous function. Then, $d^2(J^f_{\lambda}x, x) \leq d^2(J^f_{\mu}x, x)$ for $0 < \lambda < \mu$ and $x \in X$.

Proof. Let $x, y \in X$, then we obtain from the definition of the resolvent of f that

$$f(J^f_{\mu}x) + \frac{1}{2\mu}d^2(J^f_{\mu}x, x) \le f(J^f_{\lambda}x) + \frac{1}{2\mu}d^2(J^f_{\lambda}x, x),$$
(7.2.6)

and

$$f(J_{\lambda}^{f}x) + \frac{1}{2\lambda}d^{2}(J_{\lambda}^{f}x, x) \leq f(J_{\mu}^{f}x) + \frac{1}{2\lambda}d^{2}(J_{\mu}^{f}x, x).$$
(7.2.7)

Adding both sides of relations (7.2.6) and (7.2.7), we obtain that

$$\left(1 - \frac{\lambda}{\mu}\right) d^2(J_{\lambda}^f x, x) \le \left(1 - \frac{\lambda}{\mu}\right) d^2(J_{\mu}^f x, x)$$

Since, $0 < \lambda < \mu$, we obtain that

$$d^2(J^f_{\lambda}x,x) \le d^2(J^f_{\mu}x,x).$$

Lemma 7.2.3. Let X be an Hadamard space and $f_j : X \to (-\infty, \infty], \quad j = 1, 2, \cdots, m$ be a finite family of proper, convex and lower semicontinuous functions. If $0 < \lambda < \mu$ and $\left(\bigcap_{j=1}^m F\left(J_{\mu}^{(j)}\right)\right) \neq \emptyset$. Then,

$$F\left(\Pi_{j=1}^{m}J_{\mu}^{(j)}\right)\subseteq\left(\cap_{j=1}^{m}F\left(J_{\lambda}^{(j)}\right)\right),$$

where, $\Pi_{j=1}^m J_{\mu}^{(j)} = J_{\mu}^{(1)} \circ J_{\mu}^{(2)} \circ \cdots \circ J_{\mu}^{(m)}$.

Proof. Let $x \in F\left(\prod_{j=1}^{m} J_{\mu}^{(j)}\right)$ and $y \in \left(\bigcap_{j=1}^{m} F\left(J_{\mu}^{(j)}\right)\right)$, then by the nonexpansivity of $J_{\mu}^{(1)}$, we have that

$$d^{2}(x,y) = d^{2} \left(\prod_{j=1}^{m} J_{\mu}^{(j)} x, y \right) \leq d^{2} \left(\prod_{j=2}^{m} J_{\mu}^{(j)} x, y \right).$$
(7.2.8)

More so, we obtain from Lemma 2.5.27 that

$$\frac{1}{2\mu}d^2\left(\Pi_{j=1}^m J^{(j)}x,y\right) - \frac{1}{2\mu}d^2\left(\Pi_{j=2}^m J^{(j)}_\mu x,y\right) + \frac{1}{2\mu}d^2\left(\Pi_{j=2}^m J^{(j)}_\mu x,\Pi_{j=1}^m J^{(j)}_\mu x\right) + f\left(\Pi_{j=1}^m J^{(j)}_\mu x\right) \le f(y).$$

Since $f(y) \leq f\left(\prod_{j=1}^{m} J^{(j)} x\right)$, we obtain from (7.2.8) that

$$d^{2} \left(\Pi_{j=2}^{m} J_{\mu}^{(j)} x, \Pi_{j=1}^{m} J_{\mu}^{(j)} x \right) \leq d^{2} \left(\Pi_{j=2}^{m} J_{\mu}^{(j)} x, y \right) - d^{2} \left(\Pi_{j=1}^{m} J_{\mu}^{(j)} x, y \right)$$
$$\leq d^{2} (x, y) - d^{2} \left(\Pi_{j=1}^{m} J_{\mu}^{(j)} x, y \right)$$
$$= d^{2} \left(\Pi_{j=1}^{m} J_{\mu}^{(j)} x, y \right) - d^{2} \left(\Pi_{j=1}^{m} J_{\mu}^{(j)} x, y \right),$$

which implies that

$$\Pi_{j=1}^{m} J_{\mu}^{(j)} x = \Pi_{j=2}^{m} J_{\mu}^{(j)} x.$$
(7.2.9)

Similarly, we obtain from Lemma 2.5.27 and (7.2.8) that

$$\Pi_{j=2}^{m} J_{\mu}^{(j)} x = \Pi_{j=3}^{m} J_{\mu}^{(j)} x = \Pi_{j=4}^{m} J_{\mu}^{(j)} x = \dots = \Pi_{j=m-1}^{m} J_{\mu}^{(j)} x = J_{\mu}^{(m)} x = x.$$
(7.2.10)

From (7.2.10), we obtain that

$$x = J_{\mu}^{(m)} x = \prod_{j=m-1}^{m} J_{\mu}^{(j)} x = J_{\mu}^{(m-1)} J_{\mu}^{(m)} x = J_{\mu}^{(m-1)} x.$$
(7.2.11)

By repeating the process, we obtain

$$J_{\mu}^{(1)}x = J_{\mu}^{(2)}x = \dots = J_{\mu}^{(m-1)}x = J_{\mu}^{(m)}x = x.$$
(7.2.12)

Thus, by Lemma 7.2.2, we obtain

$$d^{2}(x, J_{\lambda}^{(j)}x) \leq d^{2}(x, J_{\mu}^{(j)}x) = 0, \quad j = 1, 2, \cdots, m,$$

which implies that $x \in F(J_{\lambda}^{(j)}), \ j = 1, 2, \cdots, m$. Hence, we get the desired conclusion. \Box

7.3 Modified proximal point methods

In this section, we propose two new proximal point methods involving quasi- pseudocontractive mappings in Hadamard spaces. We prove that the first method converges strongly to a common solution of a finite family of MPs and FPP for a finite family of quasi-pseudocontractive mappings in an Hadamard space. We extend this method to a more general method involving multivalued monotone operators to approximate the solution of MIP, which is an important optimization problem. We establish that this method converges strongly to a common zero of a finite family of multivalued monotone operators which is also a common fixed point of a finite family of quasi-pseudocontractive mappings in an Hadamard space. Furthermore, we provide various nontrivial numerical implementations of our method in Hadamard spaces (which are non-Hilbert) and compare them with some other recent methods in the literature.

7.3.1 Main results

Lemma 7.3.1. Let X be an Hadamard space and $f_j: X \to (-\infty, +\infty], \quad j = 1, 2, \cdots, m$ be a finite family of proper, convex and lower semicontinuous functions. Let $T_i: X \to X$, $i = 1, 2, \cdots, N$ be a finite family of L-Lipschitzian and quasi-pseudocontractive mappings with $L \ge 1$ such that T_i is demiclosed. Suppose that $\Omega := \{\bigcap_{i=1}^N F(T_i) \cap (\bigcap_{j=1}^m \operatorname{argmin}_{y \in X} f_j(y))\} \neq \emptyset$ and for arbitrary $x_1, u \in X$, the sequence $\{x_n\}$ is generated by

$$\begin{cases} w_n = \prod_{j=1}^m J_{\lambda_n}^{(j)} x_n = J_{\lambda_n}^{(1)} \circ J_{\lambda_n}^{(2)} \circ \cdots \circ J_{\lambda_n}^{(m)} x_n, \\ y_n = \beta_{n,0} w_n \oplus (1 - \beta_{n,0}) \sum_{i=1}^N \oplus_{\overline{(1 - \beta_{n,0})}} ((1 - \epsilon_{n,i}) w_n \oplus \epsilon_{n,i} T_i ((1 - \gamma_{n,i}) w_n \oplus \gamma_{n,i} T_i w_n)) (7.3.1) \\ x_{n+1} = (1 - \alpha_n) y_n \oplus \alpha_n u, \quad n \ge 1, \end{cases}$$

where $\lambda_n > \lambda > 0$, $\{\alpha_n\}$ and $\{\beta_{n,i}\}$ are sequences in (0,1) satisfying $\sum_{i=0}^{N} \beta_{n,i} = 1$, and $0 < \epsilon_{n,i} < \gamma_{n,i} < \frac{1}{1+\sqrt{1+L^2}}$, $i = 1, 2, \cdots, N$. Then $\{x_n\}$ is bounded.

Proof. First observe that by Remark 2.5.29, Algorithm (7.3.1) is well defined. Now, let $p \in \Omega$ and $K_i w_n := ((1 - \epsilon_{n,i})w_n \oplus \epsilon_{n,i}T_i((1 - \gamma_{n,i})w_n \oplus \gamma_{n,i}T_iw_n))$, then we have from (7.3.1) and Lemma 2.5.28 that

$$d(y_n, p) = d\left(\beta_{n,0}w_n \oplus (1 - \beta_{n,0})\sum_{i=1}^N \oplus \frac{\beta_{n,i}}{(1 - \beta_{n,0})}K_iw_n, p\right)$$

$$\leq \beta_{n,0}d(w_n, p) + \sum_{i=1}^N \beta_{n,i}d(K_iw_n, p)$$

$$\leq d(w_n, p)$$

$$\vdots$$

$$\leq d(x_n, p).$$
(7.3.2)

Now, using (7.3.1) and (7.3.3), we obtain that

$$d(x_{n+1}, p) = d((1 - \alpha_n)y_n + \alpha_n u, p)$$

$$\leq (1 - \alpha_n)d(y_n, p) + \alpha_n d(u, p)$$

$$\leq (1 - \alpha_n)d(x_n, p) + \alpha_n d(u, p)$$

$$< \max\{d(x_n, p), d(u, p)\}.$$

By induction, we obtain that $\{x_n\}$ is bounded. Consequently $\{w_n\}$ and $\{y_n\}$ are bounded.

Theorem 7.3.2. Let X be an Hadamard space and $f_j: X \to (-\infty, +\infty], \quad j = 1, 2, \cdots, m$ be a finite family of proper, convex and lower semicontinuous functions. Let $T_i: X \to X$, $i = 1, 2, \cdots, N$ be a finite family of L-Lipschitzian and quasi-pseudocontractive mappings with $L \ge 1$ such that T_i is demiclosed. Suppose that $\Omega := \{\bigcap_{i=1}^N F(T_i) \cap (\bigcap_{j=1}^m \operatorname{argmin}_{y \in X} f_j(y))\} \neq \emptyset$ and the sequence $\{x_n\}$ is generated by (7.3.1), where $\lambda_n > \lambda > 0$, $\{\alpha_n\}$ and $\{\beta_{n,i}\}$ are sequences in (0, 1) satisfying the following conditions:

(i)
$$\lim_{n \to \infty} \alpha_n = 0 \text{ and } \sum_{n=1}^{\infty} \alpha_n = \infty;$$

(ii) $0 < a < \epsilon_{n,i} < \gamma_{n,i} < b < \frac{1}{1+\sqrt{1+L^2}}, \quad i = 1, 2, \cdots, N;$
(iii) $\sum_{i=0}^{N} \beta_{n,i} = 1 \text{ and } 0 < c \le \beta_{n,i} \le d < 1, \quad for \quad all \quad n \ge 1$

Then, $\{x_n\}$ converges strongly to a point $z = P_{\Omega}u$, where P_{Ω} is the metric projection of X onto Ω .

Proof. Let $p \in \Omega$, then we have from Lemma 2.5.30 and (7.3.3) that

$$d^{2}(y_{n},p) = d^{2} \left(\beta_{n,0}w_{n} \oplus (1-\beta_{n,0}) \sum_{i=1}^{N} \frac{\beta_{n,i}}{(1-\beta_{n,0})} K_{i}w_{n}, p \right)$$

$$\leq \beta_{n,0}d^{2}(w_{n},p) + \sum_{i=1}^{N} \beta_{n,i}d^{2}(K_{i}w_{n},p) - \beta_{n,0} \sum_{i=1}^{N} \beta_{n,i}d^{2}(w_{n},K_{i}w_{n})$$

$$\leq d^{2}(x_{n},p) - \beta_{n,0} \sum_{i=1}^{N} \beta_{n,i}d^{2}(w_{n},K_{i}w_{n}).$$
(7.3.4)

Now, using (7.3.1) and (7.3.4), we have that

$$d^{2}(x_{n+1}, p) = d^{2}(\alpha_{n}u \oplus (1 - \alpha_{n})y_{n}, p)$$

$$\leq \alpha_{n}^{2}d^{2}(u, p) + (1 - \alpha_{n})^{2}d^{2}(y_{n}, p) + 2\alpha_{n}(1 - \alpha_{n})\langle \overrightarrow{up}, \overrightarrow{y_{n}p} \rangle$$

$$\leq \alpha_{n}^{2}d^{2}(u, p) + (1 - \alpha_{n})d^{2}(x_{n}, p) - (1 - \alpha_{n})\beta_{n,0}\sum_{i=1}^{N} \beta_{n,i}d^{2}(w_{n}, K_{i}w_{n}) \quad (7.3.5)$$

$$+ 2\alpha_{n}(1 - \alpha_{n})\langle \overrightarrow{up}, \overrightarrow{y_{n}p} \rangle$$

$$\leq \alpha_{n}^{2}d^{2}(u, p) + (1 - \alpha_{n})d^{2}(x_{n}, p) + 2\alpha_{n}(1 - \alpha_{n})\langle \overrightarrow{up}, \overrightarrow{y_{n}p} \rangle$$

$$= (1 - \alpha_{n})d^{2}(x_{n}, p) + \alpha_{n}c_{n}, \quad (7.3.6)$$

where $c_n = [\alpha_n d^2(u, p) + 2(1 - \alpha_n) \langle \overrightarrow{up}, \overrightarrow{y_n p} \rangle]$. According to Lemma 2.5.36, to conclude our proof, it suffices to show that $\limsup_{k \to \infty} c_{n_k} \leq 0$ for every subsequence $\{d(x_{n_k}, p)\}$ of $\{d(x_n, p)\}$ satisfying the condition

$$\liminf_{k \to \infty} \left(d(x_{n_k+1}, p) - d(x_{n_k}, p) \right) \ge 0.$$
(7.3.7)

To show this, suppose that $\{d(x_{n_k}, p)\}$ is a subsequence of $\{d(x_n, p)\}$ such that (7.3.7) holds. Then,

$$\liminf_{k \to \infty} \left(d^2(x_{n_k+1}, p) - d^2(x_{n_k}, p) \right) = \liminf_{k \to \infty} \left(d(x_{n_k+1}, p) - d(x_{n_k}, p) \right) \left(d(x_{n_k+1}, p) + d(x_{n_k}, p) \right) \\ \ge 0.$$
(7.3.8)

Now, by (7.3.5), we obtain that

$$\limsup_{k \to \infty} \left((1 - \alpha_{n_k}) \beta_{n_k,0} \sum_{i=1}^N \beta_{n_k,i} d^2(w_{n_k}, K_i w_{n_k}) \right) \leq \limsup_{k \to \infty} \left(\alpha_{n_k}^2 d^2(u, p) + (1 - \alpha_{n_k}) d^2(x_{n_k}, p) \right) \\
- \limsup_{k \to \infty} \left(d^2(x_{n_k+1}, p) \right) \\
+ \limsup_{k \to \infty} \left(2\alpha_{n_k} (1 - \alpha_{n_k}) \langle \overrightarrow{up}, \overrightarrow{y_{n_k}p} \rangle \right) \\
\leq \limsup_{k \to \infty} \left(d^2(x_{n_k}, p) - d^2(x_{n_k+1}, p) \right) \\
+ \limsup_{k \to \infty} \left(\alpha_{n_k}^2 d^2(u, p) \right) \\
+ \limsup_{k \to \infty} \left(2\alpha_{n_k} (1 - \alpha_{n_k}) \langle \overrightarrow{up}, \overrightarrow{y_{n_k}p} \rangle \right) \\
= -\lim_{k \to \infty} \left[d^2(x_{n_k+1}, p) - d^2(x_{n_k}, p) \right] \\
\leq 0, \qquad (7.3.9)$$

which implies that

$$\lim_{k \to \infty} d(w_{n_k}, K_i w_{n_k}) = 0, \ i = 1, 2, \dots, N.$$
(7.3.10)

Furthermore, we obtain from (7.3.1) that

$$d(x_{n_k+1}, y_{n_k}) = \alpha_{n_k} d(u, y_{n_k}) \to 0, \quad k \to \infty.$$
(7.3.11)

Again, from (7.3.1) and (7.3.10), we obtain

$$d(y_{n_k}, w_{n_k}) = d\left(\beta_{n_k, 0} w_{n_k} \oplus (1 - \beta_{n_k, 0}) \sum_{i=1}^N \oplus \frac{\beta_{n_k, i}}{(1 - \beta_{n_k, 0})} K_i w_{n_k}, w_{n_k}\right)$$

$$\leq \sum_{i=1}^N \beta_{n_k, i} d(K_i w_{n_k}, w_{n_k}) \to 0, \text{ as } k \to \infty.$$
(7.3.12)

From Lemma 2.5.27 and (7.3.3), we obtain that

$$d^{2}(w_{n_{k}}, \Pi_{j=2}^{m} J_{\lambda_{n_{k}}}^{(j)} x_{n_{k}}) = d^{2}(\Pi_{j=1}^{m} J_{\lambda_{n_{k}}}^{(j)} x_{n_{k}}, \Pi_{j=2}^{m} J_{\lambda_{n_{k}}}^{(j)} x_{n_{k}})$$

$$\leq d^{2}(\Pi_{j=2}^{m} J_{\lambda_{n_{k}}}^{(j)} x_{n_{k}}, p) - d^{2}(w_{n_{k}}, p)$$

$$\vdots$$

$$\leq d^{2}(x_{n_{k}}, p) - d^{2}(w_{n_{k}}, p)$$

$$\leq d^{2}(x_{n_{k}}, p) - d^{2}(y_{n_{k}}, p)$$

$$= d^{2}(x_{n_{k}}, p) - d^{2}(x_{n_{k}+1}, p) + d^{2}(x_{n_{k}+1}, p) - d^{2}(y_{n_{k}}, p)$$

$$\leq d^{2}(x_{n_{k}}, p) - d^{2}(x_{n_{k}+1}, p) + \alpha_{n_{k}} d^{2}(u, p)$$

$$+ (1 - \alpha_{n_{k}})d^{2}(y_{n_{k}}, p) - d^{2}(y_{n_{k}}, p).$$
(7.3.13)

By taking limsup as $k \to \infty$ on both sides of (7.3.13), and following similar argument as in (7.3.9), we obtain that

$$\lim_{k \to \infty} d\left(w_{n_k}, \Pi_{j=2}^m J_{\lambda_{n_k}}^{(j)} x_{n_k}\right) = d\left(\Pi_{j=1}^m J_{\lambda_{n_k}}^{(j)} x_{n_k}, \Pi_{j=2}^m J_{\lambda_{n_k}}^{(j)} x_{n_k}\right) = 0.$$
(7.3.14)

Following the same argument as in (7.3.14), we can show that

$$\lim_{k \to \infty} d\left(\prod_{j=2}^{m} J_{\lambda_{n_k}}^{(j)} x_{n_k}, \prod_{j=3}^{m} J_{\lambda_{n_k}}^{(j)} x_{n_k} \right) = \dots = \lim_{k \to \infty} d\left(J_{\lambda_{n_k}}^{(m)} x_{n_k}, x_{n_k} \right) = 0.$$
(7.3.15)

Hence, we obtain that

$$d(w_{n_{k}}, x_{n_{k}}) \leq d\left(w_{n_{k}}, \prod_{j=2}^{m} J_{\lambda_{n_{k}}}^{(j)} x_{n_{k}}\right) + d\left(\prod_{j=2}^{m} J_{\lambda_{n_{k}}}^{(j)} x_{n_{k}}, \prod_{j=3}^{m} J_{\lambda_{n_{k}}}^{(j)} x_{n_{k}}\right) + d\left(\prod_{j=3}^{m} J_{\lambda_{n_{k}}}^{(j)} x_{n_{k}}, \prod_{j=4}^{m} J_{\lambda_{n_{k}}}^{(j)} x_{n_{k}}\right) + \dots + d\left(J_{\lambda_{n_{k}}}^{(m)} x_{n_{k}}, x_{n_{k}}\right) \to 0, \quad \text{as} \quad k \to \infty.$$

$$(7.3.16)$$

From (7.3.12) and (7.3.16), we obtain that

$$\lim_{k \to \infty} d(y_{n_k}, x_{n_k}) = 0.$$
(7.3.17)

Since $\{x_{n_k}\}$ is bounded in an Hadamard space X, it follows from Lemma 2.5.31 that there exists a subsequence $\{x_{n_{k_j}}\}$ of $\{x_{n_k}\}$ such that $\triangle -\lim_{j\to\infty} x_{n_{k_j}} = v$. From (7.3.16) and (7.3.17), we have that there exist subsequences $\{w_{n_{k_j}}\}$ and $\{y_{n_{k_j}}\}$ of $\{w_{n_k}\}$ and $\{y_{n_k}\}$ respectively such that $\triangle -\lim_{j\to\infty} w_{n_{k_j}} = v$ and $\triangle -\lim_{j\to\infty} y_{n_{k_j}} = v$. Since T_i , $i = 1, 2, \cdots, N$ is demiclosed, we obtain from Lemma 7.2.1 and (7.3.10) that $v \in \bigcap_{i=1}^N F(T_i) = \bigcap_{i=1}^N F(K_i)$. More so, $J_{\lambda_{n_k}}^{(j)}$ is a nonexpansive mapping for each $j = 1, 2, \cdots, m$. Thus, we obtain from (7.3.16) and Lemma 2.5.33 that $v \in F\left(\prod_{j=1}^m J_{\lambda_{n_k}}^{(j)}\right)$. Thus, by Lemma 7.2.3, we have that $v \in \bigcap_{j=1}^m F\left(J_{\lambda}^{(j)}\right) = \bigcap_{j=1}^m \arg\min_{y\in X} f_j(y)$. Hence $v \in \Omega$. Let $z = P_{\Omega}u$, since $\{y_{n_k}\}$ is bounded. We can choose a subsequence $\{y_{n_{k_j}}\}$ of $\{y_{n_k}\}$ that \triangle -converges to v, and such that

$$\limsup_{k \to \infty} \langle \overrightarrow{uz}, \overrightarrow{y_{n_k} z} \rangle = \lim_{j \to \infty} \langle \overrightarrow{uz}, \overrightarrow{y_{n_{k_j}}} z \rangle \le \langle \overrightarrow{uz}, \overrightarrow{vz} \rangle.$$
(7.3.18)

Since $v \in \Omega$, we obtain from (7.3.18) and Lemma 2.5.32 that

$$\limsup_{k \to \infty} \langle \overrightarrow{uz}, \overrightarrow{y_{n_k} z} \rangle \leq \langle \overrightarrow{uz}, \overrightarrow{vz} \rangle \leq 0.$$
(7.3.19)

Replacing p with z in (7.3.6), we obtain from (7.3.19) that $\limsup_{k\to\infty} c_{n_k} \leq 0$ (where $c_{n_k} = [\alpha_{n_k}d^2(u,z) + 2(1-\alpha_{n_k})\langle \overrightarrow{uz}, \overrightarrow{y_{n_k}z} \rangle]$). Thus, applying Lemma 2.5.36 to (7.3.6), we conclude that $d(x_n, z) \to 0$, as $n \to \infty$. Therefore, $\{x_n\}$ converges strongly to $z = P_{\Omega}u$.

Remark 7.3.3. Note that by setting N = 2 and letting T_i , i = 1, ..., N to be a demicontractive mapping, we recover the main result of Chang *et al.* [70] as corollary of our main result. Also, by setting N = 1 = m and letting T to be a nonexpansive mapping, we recover the main result of Suparatulatorn *et al.* [232] as corollary of our main result. Furthermore, we considered a finite family of MPs and FPP for a finite family of quasipseudocontractive mappings, while in [72], the authors considered one MP and FPP for a single nonexpansive mapping. Moreover, if m = 1, N = 3 and T is nonexpansive, we recover the main result of Thounthong *et al.* [252]. In a similar way, we can derive the results in [34, 79, 80, 91, 254] as corollaries of our main result.

7.3.2 Extension to monotone inclusion problems

In this section, we extend our study so far to the following MIP: Find $x \in X$ such that

$$0 \in Ax,\tag{7.3.20}$$

where the multivalued operator $A: X \to 2^{X^*}$ (X^{*} is the dual space of the Hadamard space X recently introduced in [138]) is monotone and may not necessarily be the subdifferential ∂f of a proper convex and lower semicontinuous function f. Note that if $A = \partial f$, then problem (7.3.20) is just equivalent to problem (1.2.16).

The MIPs are known to be one of the most important problems in optimization, nonlinear and convex analysis. The problem was first introduced in Hadamard spaces by Khatibzadeh and Ranjbar [150]. Since then many authors have studied this problem in Banach and Hadamard spaces (see for example [67, 100, 125, 133, 191, 233, 255] and the references therein). One striking shortcoming in all these works is that they fail to provide typical example(s) of multivlaued monotone operators which are not necessarily the subdifferential ∂f of a proper convex and lower semicontinuous function. It is worthy to note that despite that such examples are not easy to find, they are necessary to further motivate the study of problem (7.3.20) in Hadamard spaces (from application point of view) since the main results of all these works are about the multivalued monotone operator A. Thus, examples of A in the form of subdifferential operator would simply not give the full essence of studying problem (7.3.20). For this, we present in what follows, a typical example of a monotone operator defined on an Hadamard space and valued in the dual space. We also present a typical example of the associated resolvent operator. First, we present the definition of these concepts.

Definition 7.3.4. (see [150]) Let X be an Hadamard space and X^* be its dual space. A multivalued operator $A : X \to 2^{X^*}$ with domain $D(A) := \{x \in X : Ax \neq \emptyset\}$ is monotone if and only if for all $x, y \in D(A), x \neq y, x^* \in Ax, y^* \in Ay$,

$$\langle x^* - y^*, \overrightarrow{yx} \rangle \ge 0.$$

Definition 7.3.5. [150] Let X be an Hadamard space and X^* be its dual. Let $A : X \to 2^{X^*}$ be a multivalued operator, then the resolvent of A of order $\lambda > 0$ is the multivalued operator $J_{\lambda}^A : X \to 2^X$ defined by

$$J_{\lambda}^{A}(x) := \{ z \in X \mid [\frac{1}{\lambda} \overrightarrow{zx}] \in Az \}.$$

$$(7.3.21)$$

Definition 7.3.6. Let X be an Hadamard space and X^* be its dual. The subdifferential of a function $f: (-\infty, +\infty] \to \mathbb{R}$ is a multivalued function $\partial f: X \to 2^{X^*}$ defined by

$$\partial f(x) = \begin{cases} \{x^* \in X^* : f(z) - f(x) \ge \langle x^*, \overline{xz} \rangle, \ \forall z \in X\}, & \text{if } x \in D(f), \\ \emptyset, & \text{otherwise.} \end{cases}$$

Example 7.3.7. Let $Y = \mathbb{R}^2$ be an \mathbb{R} -tree with the radial metric d_r , where $d_r(x, y) = d(x, y)$ if x and y are situated on a Euclidean straight line passing through the origin and

 $d_r(x, y) = d(x, 0) + d(y, 0) := ||x|| + ||y||$ otherwise (see [93, 148] and [202, page 65]). We put p = (1, 0) and $X = B \cup C$, where

$$B = \{(h,0): h \in [0,1]\} \text{ and } C = \{(h,k): h+k = 1, h \in [0,1)\}$$

Note that X is closed and convex and so, (X, d_r) is an Hadamard space. To present an example of the dual space, first note that the dual space (X^*, \mathcal{D}) is defined as

$$X^* = \{ [t\overrightarrow{ab}] : (t, a, b) \in \mathbb{R} \times X \times X \}$$

together with the associated metric \mathcal{D} which defines an equivalence relation on $\mathbb{R} \times X \times X$, where the equivalence class of $\overrightarrow{tab} := (t, a, b)$ is given by

$$[\overrightarrow{tab}] = \{\overrightarrow{scd} : t\langle \overrightarrow{ab}, \overrightarrow{xy} \rangle = s\langle \overrightarrow{cd}, \overrightarrow{xy} \rangle \ \forall x, y \in X\}.$$

Note that $\mathcal{D}((t, a, b), (s, c, d)) = 0$ if and only if $t\langle \overrightarrow{ab}, \overrightarrow{xy} \rangle = s \langle \overrightarrow{cd}, \overrightarrow{xy} \rangle$ for all $x, y \in X$ (see [138, 150] for more details and formulation of (X^*, \mathcal{D})).

Thus, we can define the dual space X^* as the space of elements $[t\vec{ab}]$ such that

$$[t\vec{ab}] = \begin{cases} \{s\vec{cd}: c, d \in B, s \in \mathbb{R}, t(\|b\| - \|a\|) = s(\|d\| - \|c\|)\} & a, b \in B, \\ \{s\vec{cd}: c, d \in C \cup \{0\}, s \in \mathbb{R}, t(\|b\| - \|a\|) = s(\|d\| - \|c\|)\} & a, b \in C \cup \{0\}, \\ \{t\vec{ab}\} & a \in B, b \in C. \end{cases}$$

Indeed, for each $[t\overrightarrow{ab}] \in X^*$, we calculate its equivalence class as follows: If $[t\overrightarrow{ab}] = [\overrightarrow{scd}] \neq [\overrightarrow{pp}]$, we must have that $t\langle \overrightarrow{ab}, \overrightarrow{xy} \rangle = s\langle \overrightarrow{cd}, \overrightarrow{xy} \rangle$ for all $x, y \in X$.

Cases:

- (1): If $\{a, b, c, d\} \subset C \cup \{\mathbf{0}\}$, we have that $d_r(e, z) = ||e|| + ||z||$ for all $e \in A$ and $z \in X$. Thus, the equality $t\langle \overrightarrow{ab}, \overrightarrow{xy} \rangle = s\langle \overrightarrow{cd}, \overrightarrow{xy} \rangle$ is equivalent to t(||b|| - ||a||) = s(||d|| - ||c||).
- (2): If $\{a, b, c, d\} \subset B$, we obtain by similar argument as in Case 1 that the equality $t\langle \overrightarrow{ab}, \overrightarrow{xy} \rangle = s\langle \overrightarrow{cd}, \overrightarrow{xy} \rangle$ is equivalent to $t(\|b\| \|a\|) = s(\|d\| \|c\|)$.
- (3): In Case 1 and Case 2, the equation does not depend on x and y. But in other cases, the equation depends on x and y, i.e., the equality for $x, y \in B$ is different from the equality for $x, y \in C$.

Thus, we conclude that X^* is the dual space of X.

Now that we have both the Hadamard space and its dual space, we can now define the monotone operator and corresponding resolvent operator. Let $A: X \to 2^{X^*}$ be defined by

$$Ax = \begin{cases} \{ [\overrightarrow{\mathbf{0}p}] \} & x \in B, \\ \\ \\ \{ [\overrightarrow{\mathbf{0}p}], [\overrightarrow{\mathbf{0}x}] \} & x \in C. \end{cases}$$

Then A is a multivalued monotone operator. Indeed, we consider the following cases. Cases: (1): If $x, y \in B$, then $Ax = Ay = \{[\overrightarrow{\mathbf{0}p}]\}$ and $x^* = y^* = [\overrightarrow{\mathbf{0}p}]$. So, $\langle x^* - y^*, \overrightarrow{yx} \rangle = 0 \ge 0$. (2): If $x, y \in C$, then $Ax = \{[\overrightarrow{\mathbf{0}p}], [\overrightarrow{\mathbf{0}x}]\}$ and $Ay = \{[\overrightarrow{\mathbf{0}p}], [\overrightarrow{\mathbf{0}y}]\}$.

- (i) If $x^* = y^* = [\overrightarrow{\mathbf{0}p}]$, then $\langle x^* y^*, \overrightarrow{yx} \rangle = 0 \ge 0$. (ii) If $x^* = [\overrightarrow{\mathbf{0}x}]$ and $y^* = [\overrightarrow{\mathbf{0}y}]$, then $\langle x^* - y^*, \overrightarrow{yx} \rangle = d_r^2(x, y) \ge 0$.
- (iii) If $x^* = [\overrightarrow{\mathbf{0}p}]$ and $y^* = [\overrightarrow{\mathbf{0}y}]$, then

$$\begin{split} \langle x^* - y^*, \overrightarrow{yx} \rangle &= \langle \overrightarrow{yp}, \overrightarrow{yx} \rangle \\ &= \frac{1}{2} (d_r^2(y, x) + d_r^2(p, y) - d_r^2(p, x)) \\ &= \frac{1}{2} ((\|y\| + \|x\|)^2 + (1 + \|y\|)^2 - (1 + \|x\|)^2) \\ &\geq 0 \text{ (since } 1/\sqrt{2} \le \|x\|, \|y\| \le 1). \end{split}$$

(iv) If $x^* = [\overrightarrow{\mathbf{0}x}]$ and $y^* = [\overrightarrow{\mathbf{0}p}]$, then $\langle x^* - y^*, \overrightarrow{yx} \rangle = \langle \overrightarrow{px}, \overrightarrow{yx} \rangle$, which is similar to (iii).

(3): If
$$x \in B, y \in C$$
, then $Ax = \{[\overrightarrow{\mathbf{0p}}]\}, Ay = \{[\overrightarrow{\mathbf{0p}}], [\overrightarrow{\mathbf{0y}}]\}.$
(i) If $x^* = y^* = [\overrightarrow{\mathbf{0p}}]$, then $\langle x^* - y^*, \overrightarrow{yx} \rangle = 0 \ge 0.$
(ii) If $x^* = [\overrightarrow{\mathbf{0p}}]$ and $y^* = [\overrightarrow{\mathbf{0y}}]$, then
 $\langle x^* - y^*, \overrightarrow{yx} \rangle = \langle \overrightarrow{yp}, \overrightarrow{yx} \rangle$
 $= \frac{1}{2}(d_r^2(y, x) + d_r^2(p, y) - d_r^2(p, x))$
 > 0 (since $d(p, x) < 1 < d(p, y)$).

Thus, A is monotone. We now compute the resolvent of A as follows: Cases:

- (I) Let $x = (h, 0) \in B$.
 - (i) If $z = (k, 0) \in B$ and $z \in J^A_{\lambda}(x)$, then $Az = \{[\overrightarrow{\mathbf{0}p}]\}$ and $[\frac{1}{\lambda}\overrightarrow{zx}] = [\overrightarrow{\mathbf{0}p}]$. It follows from (7.3.22) that $\frac{1}{\lambda}(k-h) = 1$ or $k = h \lambda$.
 - (ii) If $z = (h', k') \in C$ and $z \in J_{\lambda}^{A}(x)$, then $Az = \{[\overrightarrow{\mathbf{0}p}], [\overrightarrow{\mathbf{0}z}]\}$ and $[\frac{1}{\lambda}\overrightarrow{zx}] = [\overrightarrow{\mathbf{0}p}]$ or $[\frac{1}{\lambda}\overrightarrow{zx}] = [\overrightarrow{\mathbf{0}z}]$. Using (7.3.22) we see that both of these two cases are impossible.
- (II) Let $x = (h, k) \in C$.
 - (i) If $z = (h', 0) \in B$ and $z \in J_{\lambda}^{A}(x)$, then $Az = \{[\overrightarrow{\mathbf{0}p}]\}$ and $[\frac{1}{\lambda}\overrightarrow{zx}] = [\overrightarrow{\mathbf{0}p}]$ which is impossible by (7.3.22).
 - (ii) If $z = (h', k') \in C$ and $z \in J_{\lambda}^{A}(x)$, then $Az = \{[\overrightarrow{\mathbf{0}p}], [\overrightarrow{\mathbf{0}z}]\}$ and $[\frac{1}{\lambda} \overrightarrow{zx}] \in Az$. The case $[\frac{1}{\lambda} \overrightarrow{zx}] = [\overrightarrow{\mathbf{0}p}]$ is impossible. For the case $[\frac{1}{\lambda} \overrightarrow{zx}] = [\overrightarrow{\mathbf{0}z}]$. Using (7.3.22), we see that $\frac{1}{\lambda}(||x|| ||z||) = ||z||$ or $||z|| = \frac{1}{1+\lambda}||x||$. Note that there are at most two solutions for z.

Therefore, we derive the resolvent as:

$$J_{\lambda}^{A}(x) = \begin{cases} \{z = (h - \lambda, 0)\} & x = (h, 0) \in B, \\ \{z = (h', k') \in C : (1 + \lambda)^{2}(h'^{2} + k'^{2}) = h^{2} + k^{2}\} & x = (h, k) \in C. \end{cases}$$

Now that we have a typical example of a multivalued monotone operator (which is not the subdifferential of a proper convex and lower semicontinuous function) in a typical Hadamard space (which is non-Hilbert), we proceed to propose a new PPA similar to (7.3.1) and establish that it converges strongly to a common solution of a finite family of MIPs (7.3.20), which is also a common fixed point of a finite family of quusipseudocontractive mappings in an Hadamard space X. We begin with the following crucial lemmas.

Lemma 7.3.8. [150] Let X be an Hadamard space and J_{λ}^{A} be the resolvent of a multivalued operator A of order λ . Then,

- (i) for any $\lambda > 0$, $R(J_{\lambda}^{A}) \subset D(A)$ and $F(J_{\lambda}^{A}) = A^{-1}(0)$, where $R(J_{\lambda}^{A})$ is the range of J_{λ}^{A} (we say that the operator A satisfies the range condition if for every $\lambda > 0$, $D(J_{\lambda}^{A}) = X$).
- (ii) if A is monotone then J^A_{λ} is a single-valued and nonexpansive mapping,
- (iii) if A is monotone and $0 < \lambda_1 \leq \lambda_2$, then $d(x, J_{\lambda_1}^A x) \leq 2d(x, J_{\lambda_2}^A x)$.

Lemma 7.3.9. [255] Let X be an Hadamard space and $A : X \to 2^{X^*}$ be a monotone operator. Then,

$$d^{2}(u, J_{\lambda}^{A}x) + d^{2}(J_{\lambda}^{A}x, x) \leq d^{2}(u, x)$$

for all $u \in A^{-1}(0)$, $x \in X$ and $\lambda > 0$.

Lemma 7.3.10. Let X be an Hadamard space and X^* be its dual space. Let $A_j : X \to 2^{X^*}$, j = 1, 2, ..., m be a finite family of multivalued monotone operators. If $0 < \lambda < \mu$ and $\left(\bigcap_{j=1}^m F\left(J_{\mu}^{(j)}\right) \right) \neq \emptyset$. Then,

$$F\left(\Pi_{j=1}^{m}J_{\mu}^{(j)}\right)\subseteq\left(\cap_{j=1}^{m}F\left(J_{\lambda}^{(j)}\right)\right),$$

where, $\Pi_{j=1}^m J_{\mu}^{(j)} = J_{\mu}^{(1)} \circ J_{\mu}^{(2)} \circ \cdots \circ J_{\mu}^{(m)}$.

Proof. By careful observation, we can see that Lemma 7.3.8 (iii) and Lemma 7.3.9 can be used in place of Lemma 7.2.2 and Lemma 2.5.27 respectively as the monotone version of J_{λ} . Thus, the proof follows similar argument as in the proof of Lemma 7.2.3.

Using Lemma 7.3.8 and Lemma 7.3.10, we can prove the following results by following similar line of arguments as in the proof of Lemma 7.3.1 and Theorem 7.3.2.

Lemma 7.3.11. Let X be an Hadamard space and X^* be its dual space. Let $A_j : X \to 2^{X^*}$, i = 1, 2, ..., m be a finite family of multivalued monotone operators that satisfy the range condition. Let $T_i : X \to X$, i = 1, 2, ..., N be a finite family of L-Lipschitzian and quasi-pseudocontractive mappings with $L \ge 1$ such that T_i is demiclosed. Suppose that $\Omega := \{\bigcap_{i=1}^N F(T_i) \cap (\bigcap_{j=1}^m A_j^{-1}(0))\} \neq \emptyset$ and for arbitrary $x_1, u \in X$, the sequence $\{x_n\}$ is generated by

$$\begin{cases} w_{n} = \prod_{j=1}^{m} J_{\lambda_{n}}^{(j)} x_{n} = J_{\lambda_{n}}^{(1)} \circ J_{\lambda_{n}}^{(2)} \circ \cdots \circ J_{\lambda_{n}}^{(m)} x_{n}, \\ y_{n} = \beta_{n,0} w_{n} \oplus (1 - \beta_{n,0}) \sum_{i=1}^{N} \oplus \frac{\beta_{n,i}}{(1 - \beta_{n,0})} \left((1 - \epsilon_{n,i}) w_{n} \oplus \epsilon_{n,i} T_{i} ((1 - \gamma_{n,i}) w_{n} \oplus \gamma_{n,i} T_{i} w_{n}) \right) \right), 3.22) \\ x_{n+1} = (1 - \alpha_{n}) y_{n} \oplus \alpha_{n} u, \quad n \ge 1, \end{cases}$$

where $\lambda_n > \lambda > 0$, $\{\alpha_n\}$ and $\{\beta_{n,i}\}$ are sequences in (0,1) satisfying $\sum_{i=0}^N \beta_{n,i} = 1$, and $0 < \epsilon_{n,i} < \gamma_{n,i} < \frac{1}{1+\sqrt{1+L^2}}$, $i = 1, 2, \cdots, N$. Then $\{x_n\}$ is bounded.

Theorem 7.3.12. Let X be an Hadamard space and X^{*} be its dual space. Let $A_j : X \to 2^{X^*}$, i = 1, 2, ..., m be a finite family of multivalued monotone operators that satisfy the range condition. Let $T_i : X \to X$, i = 1, 2, ..., N be a finite family of L-Lipschitzian and quasi-pseudocontractive mappings with $L \ge 1$ such that T_i is demiclosed. Suppose that $\Omega := \{ \bigcap_{i=1}^N F(T_i) \cap (\bigcap_{j=1}^m A_j^{-1}(0)) \} \neq \emptyset$ and the sequence $\{x_n\}$ is generated by (7.3.22), where $\lambda_n > \lambda > 0$, $\{\alpha_n\}$ and $\{\beta_{n,i}\}$ are sequences in (0,1) satisfying the following conditions:

(i)
$$\lim_{n \to \infty} \alpha_n = 0$$
 and $\sum_{n=1}^{\infty} \alpha_n = \infty;$

(*ii*)
$$0 < a < \epsilon_{n,i} < \gamma_{n,i} < b < \frac{1}{1+\sqrt{1+L^2}}, \quad i = 1, 2, \cdots, N;$$

(iii)
$$\sum_{i=0}^{N} \beta_{n,i} = 1$$
 and $0 < c \le \beta_{n,i} \le d < 1$, for all $n \ge 1$

Then, $\{x_n\}$ converges strongly to a point $z = P_{\Omega}u$, where P_{Ω} is the metric projection of X onto Ω .

We now give the following remark regarding Theorems 7.3.2 and 7.3.12.

Remark 7.3.13.

- (i) In Theorems 7.3.2 and 7.3.12, we assumed that the solution set Ω is nonempty which is a strict assumption. Thus, it will be interesting to study some sufficient conditions which guarantee this assumption. This we will investigate in our future project.
- (ii) In Theorems 7.3.2 and 7.3.12, the prior knowledge of the Lipschitz constant L is needed which is often not known in many applications. Part of our future research is to modify Algorithms (7.3.1) and (7.3.22) such that the prior knowledge of the constant L is not required.

7.3.3 Numerical experiments

In this section, we discuss the numerical behavior of Algorithm (7.3.1) using various non-trivial test examples. We compare our method with Algorithms (2.5.33)-(2.5.35).

The codes are implemented in Matlab 2016 (b). We perform all computations on a personal computer with an Intel(R) Core(TM) i5-2600 CPU at 2.30GHz and 8.00 Gb-RAM. Throughout this section, we shall take m = 4 = N, $\gamma_{n,i} = \frac{n}{100in+1} \epsilon_{n,i} = \frac{n}{100in+11}$ and $\beta_{n,i} = \frac{(i+1)n+1}{5(3n+1)}$ $\forall n \ge 1$, i = 0, 1, 2, 3, 4. Also, we choose $\alpha_n = \frac{3}{5n+7}$ for Algorithms (7.3.1) and (2.5.34). While we choose $\eta_n = \frac{3n}{7n+1}$, $\beta_n = \frac{n}{2n+3}$ and $\frac{8n}{10n+7}$ for Algorithms (2.5.33) and (2.5.35).

Algorithm (7.3.1) becomes:

$$\begin{cases} z_n = \arg\min_{y \in X} \left(f_4(y) + \frac{1}{2\lambda_n} d^2(y, x_n) \right), \\ v_n = \arg\min_{y \in X} \left(f_3(y) + \frac{1}{2\lambda_n} d^2(y, z_n) \right), \\ u_n = \arg\min_{y \in X} \left(f_2(y) + \frac{1}{2\lambda_n} d^2(y, v_n) \right), \\ w_n = \arg\min_{y \in X} \left(f_1(y) + \frac{1}{2\lambda_n} d^2(y, u_n) \right), \\ y_n = \beta_{n,0} w_n \oplus (1 - \beta_{n,0}) \sum_{i=1}^4 \oplus \frac{\beta_{n,i}}{(1 - \beta_{n,0})} \left((1 - \epsilon_{n,i}) w_n \oplus \epsilon_{n,i} T_i((1 - \gamma_{n,i}) w_n \oplus \gamma_{n,i} T_i w_n) \right) \\ x_{n+1} = (1 - \alpha_n) y_n \oplus \alpha_n u, \quad n \ge 1. \end{cases}$$

We now consider the following concrete examples of f_i and T_i .

Example 7.3.14. Let $X = \mathbb{R}^2$ be endowed with a metric $d : \mathbb{R}^2 \times \mathbb{R}^2 \to [0, \infty)$ defined by

$$d(x,y) = \sqrt{(x_1 - y_1)^2 + (x_1^2 - x_2 - y_1^2 + y_2)^2} \quad \forall x, y \in \mathbb{R}^2.$$

Then, (\mathbb{R}^2, d) is an Hadamard space (see [100]) where the geodosic joining x to y is given by

$$(1-t)x \oplus ty = \left((1-t)x_1 + ty_1, ((1-t)x_1 + ty_1)^2 - (1-t)(x_1^2 - x_2) - t(y_1^2 - y_2)\right).$$

Now define $T_1 : \mathbb{R}^2 \to \mathbb{R}^2$ by $T_1(x_1, x_2) = (x_1, 2x_1^2 - x_2)$. We see that T_1 is quasipseudocontractive in (\mathbb{R}^2, d) but T_1 is not quasi-pseudocontractive in the classical sense. Indeed, for all $x \in \mathbb{R}^2$ and $y \in F(T_1)$, we have

$$d^{2}(T_{1}x, y) = d^{2} \left((x_{1}, 2x_{1}^{2} - x_{2}), (y_{1}, y_{2}) \right)$$

= $(x_{1} - y_{1})^{2} + (x_{1}^{2} - (2x_{1}^{2} - x_{2}) - y_{1}^{2} + 2y_{1}^{2} - y_{2})$
= $(x_{1} - y_{1})^{2} + (-x_{1}^{2} + x_{2} + y_{1}^{2} - y_{2})^{2}$
= $(x_{1} - y_{1})^{2} + (x_{1}^{2} - x_{2} - y_{1}^{2} + y_{2})^{2}$
 $\leq d^{2}(x, y) + d^{2}(x, T_{1}x).$

Also, define $T_i : \mathbb{R}^2 \to \mathbb{R}^2$ by

$$T_i(x_1, x_2) = -\left(\frac{2i+1}{2}\right)(x_1, x_2), \ i = 2, 3, 4.$$

Then, T_i is an *L*-Lipschitzian and quasi-pseudocontractive mapping with $L = (\frac{2i+1}{2})^2 > 1$, i = 2, 3, 4. Define $f_1 : \mathbb{R}^2 \to \mathbb{R}$ by $f_1(x_1, x_2) = 100 ((x_2 + 1) - (x_1 + 1))^2 + x_1^2$. Then f_1 is a proper convex and lower semicontinuous function in (\mathbb{R}^2, d) but not convex in the classical sense (see [100]). We also define $f_j : \mathbb{R}^2 \to \mathbb{R}$ by $f_j(x_1, x_2) = 50jx_1^2$, j = 2, 3, 4. Then f_j is a proper convex and lower semicontinuous function for each j = 2, 3, 4.

We now consider the following 4 cases for our numerical experiments for Example 7.3.14 given in FIGURE 7.1 below.

Case 1: $x_1 = (0.5, 1)^T$, $u = (1, -0.5)^T$ and $\lambda_n = \frac{2n}{n+1}$. **Case 2:** $x_1 = (1.5, 2)^T$, $u = (1.5, 2)^T$ and $\lambda_n = \frac{2n}{n+1}$. **Case 3:** $x_1 = (0.5, e^2)^T$, $u = (-2, 2)^T$ and $\lambda_n = \frac{n}{3n+2}$.

Case 4: $x_1 = (-1, e^{-1})^T$, $u = (-2, e^{-1})^T$ and $\lambda_n = \frac{n}{3n+2}$.

Example 7.3.15. Let $Y := \{(x, e^x) : x \in \mathbb{R}\}$ and $X_n := \{(n, y) : y \ge e^n\}$ for each $n \in \mathbb{Z}$. Set $X := Y \cup \bigcup_{n \in \mathbb{Z}} X_n$ and equip it with a metric $d : X \times X \to [0, \infty)$, defined by (see [67])

$$d(x,y) = \begin{cases} \int_{x_1}^{y_1} ||\dot{\gamma}(t)||_2 dt + |x_2 - e^{x_1}| + |y_2 - e^{y_1}|, & \text{if } x_1 \neq y_1, \\ |x_2 - y_2|, & \text{if } x_1 = y_1, \end{cases}$$
(7.3.23)

where $\dot{\gamma}$ is the derivative of the curve $\gamma : \mathbb{R} \to X$, define by $\gamma(t) := (t, e^t)$ for each $t \in \mathbb{R}$. Then (X, d) is an Hadamard space. Let $f_j := j ||.||_2^2 : X \to \mathbb{R}, \ j = 1, 2, 3, 4$. Then, f_j is proper, convex and lower semicontinuous in (X, d) (see [67, Example 7.1]).

Now, define $T_i: X \to X$ by $T_1(x_1, x_2) = T_2(x_1, x_2) = (x_1, e^{x_1})$ and $T_3(x_1, x_2) = T_4(x_1, x_2) = (-x_1, e^{-x_1})$ for all $x = (x_1, x_2) \in X$. Then, we check that T_i is quasi-pseudocontractive for each i = 1, 2, 3, 4.

Indeed, for each $x, y \in X$, we have that

$$\begin{split} d^2(T_1x, T_1y) &= d^2\big((x_1, e^{x_1}), (y_1, e^{y_1})\big) \\ &= \begin{cases} \left(\int_{x_1}^{y_1} ||\dot{\gamma}(t)||_2 dt + |e^{x_1} - e^{x_1}| + |e^{y_1} - e^{y_1}|\right)^2 & \text{if } x_1 \neq y_1, \\ |e^{x_1} - e^{y_1}|^2 & \text{if } x_1 = y_1, \end{cases} \\ &= \begin{cases} \left(\int_{x_1}^{y_1} ||\dot{\gamma}(t)||_2 dt\right)^2 & \text{if } x_1 \neq y_1, \\ |e^{x_1} - e^{y_1}|^2 & \text{if } x_1 = y_1, \end{cases} \\ &\leq \begin{cases} \left(\int_{x_1}^{y_1} ||\dot{\gamma}(t)||_2 dt + |x_2 - e^{x_1}| + |y_2 - e^{y_1}|\right)^2 & \text{if } x_1 \neq y_1, \\ |x_2 - y_2|^2 & \text{if } x_1 = y_1, \end{cases} \\ &\leq d^2(x, y) + d^2(x, Tx). \end{split}$$

Therefore, T_1 is quasi-pseudocontractive. The proof that T_i , i = 2, 3, 4 are also quasipseudocontractive is very similar to that of T_1 above.

We now consider the following 4 cases for our numerical experiments for Example 7.3.15 given in FIGURE 7.2 below.

Case 1:
$$x_1 = (2, e^2)^T$$
, $u = (2, e^2)^T$ and $\lambda_n = \frac{n}{3n+2}$.
Case 2: $x_1 = (1, 3)^T$, $u = (0.5, e^{0.5})^T$ and $\lambda_n = \frac{n}{3n+2}$.
Case 3: $x_1 = (0.5, e^{0.5})^T$, $u = (3, 21)^T$ and $\lambda_n = \frac{n}{n+100}$.
Case 4: $x_1 = (-1, e^{-1})^T$, $u = (-2, e^{-1})^T$ and $\lambda_n = \frac{n}{n+100}$.

Example 7.3.16. Let $\mathbf{P}(n)$ be the space of $n \times n$ Hermitian positive definite matrices. The geodesic distance between A and B in $\mathbf{P}(n)$ (also called the Riemannian (trace) distance) $d_2: \mathbf{P}(n) \times \mathbf{P}(n) \to [0, \infty)$ is defined by (see [88] [188, Chapter 2] [78, Example 5.2])

$$d_{2}(A, B) = \inf\{L(c)|c : [0, 1] \to \mathbf{P}(n) \text{ is a solution curve with } c(0) = A \text{ and } c(1) = B\}$$

= $\|\log(A^{-\frac{1}{2}}BA^{-\frac{1}{2}})\|_{2}$
= $\left(\sum_{i=1}^{m}\log^{2}\mu_{i}(A^{-1}B)\right)^{\frac{1}{2}}$,

where $\mu_i(A^{-1}B)$ are the eignevalues of $A^{-1}B, L(c) := \int_0^1 \|c(t)^{-\frac{1}{2}}c'(t)c(t)^{-\frac{1}{2}}\|_2 dt, \|A\|_2 := (tr|A|^2)^{\frac{1}{2}}, tr$ is the usual trace functional and $|A| = (A^H A)^{\frac{1}{2}}$ (where A^H is the conjugate

transpose of A). The pair $(\mathbf{P}(n), d_2)$ is an Hadamard space with geodesic joining A to B in $\mathbf{P}(n)$ given by (see [73, 88, 188])

$$(1-t)x \oplus ty = A^{\frac{1}{2}} \left(A^{-\frac{1}{2}} B A^{-\frac{1}{2}} \right) A^{\frac{1}{2}}, \quad 0 \le t \le 1.$$

Now, define $T_i: \mathbf{P}(n) \to \mathbf{P}(n)$ by $T_i A = D^H A D$, i = 1, 2, 3, 4 where $D \in GL(n)$ (the set of $n \times n$ invertible matrices). Then T_i is nonexpansive (see [188, Chapter 2]), and hence, quasipseudocontractive. Also, define $f_1: \mathbf{P}(n) \to \mathbb{R}$ by $f_1 A = \left(\sum_{i=1}^m \log^p \mu_i(A^{-1}e^A)\right)^{\frac{1}{p}}$, where $\mu_i(A^{-1}e^A)$ are the eigenvalues of $A^{-1}e^A$. Then f_1 is convex and lower semicontinuous (see [19]). Again, define $f_2, f_3, f_4: \mathbf{P}(n) \to \mathbb{R}$ by $f_2 A = -\log \det A, f_3 A = \operatorname{tr}(A)$ and $f_4 A = \operatorname{tr}(e^A)$ respectively, then f_j is convex and lower semicontinuous for each j = 2, 3, 4

We now consider the following 4 cases for our numerical experiments for Example 7.3.16 given in FIGURE 7.3 below.

Case I:
$$x_1 = \begin{bmatrix} 2 & i \\ -i & 2 \end{bmatrix}$$
, $u = \begin{bmatrix} 5 & 1+i \\ 1-i & 4 \end{bmatrix}$ and $\lambda_n = \frac{n}{n+100}$.
Case II: $x_1 = \begin{bmatrix} 2 & 2-i \\ 2+i & 4 \end{bmatrix}$, $u = \begin{bmatrix} 5 & 1+i \\ 1-i & 4 \end{bmatrix}$ and $\lambda_n = \frac{n}{n+100}$.
Case III: $x_1 = \begin{bmatrix} 2 & 2-i \\ 2+i & 4 \end{bmatrix}$, $u = \begin{bmatrix} 1 & 4+i \\ 4-i & 3 \end{bmatrix}$ and $\lambda_n = \frac{10n}{2n+5}$.
Case IV: $x_1 = \begin{bmatrix} 3 & -3-i \\ -3+i & 4 \end{bmatrix}$, $u = \begin{bmatrix} 2 & -i \\ i & 2 \end{bmatrix}$ and $\lambda_n = \frac{10n}{2n+5}$.

(see [19, 227]).

Remark 7.3.17. The numerical results for Examples 7.3.14, 7.3.15 and 7.3.16 are displayed in Figures 7.1, 7.2 and 7.3, respectively.

In Figures 7.1, 7.2 and 7.3, $\text{Error} = d(x_{n+1}, x_n)$. We can see from these figures that our algorithm requires the least number of iterations (in all the examples) when compared with Algorithms (2.5.33)-(2.5.35). Thus, our method is more efficient than these other methods.



Figure 7.1: Errors vs Iteration numbers for **Example 7.3.14**: Case 1 (top left); Case 2 (top right); Case 3 (bottom left); Case 4 (bottom right).

7.4 On generalized viscosity implicit rule for quasipseudocontractive mappings

In this section, we propose a proximal point approach of a viscosity type iterative method for generalized viscosity implicit rule involving quasi-pseudocontractive mappings in Hadamard spaces. We prove that the sequence generated by our proposed algorithm converges strongly to a common solution of a finite family of MPs and FFP for a finite family of quasi-pseudocontractive mappings in an Hadamard space.



Figure 7.2: Errors vs Iteration numbers for **Example 7.3.15**: Case 1 (top left); Case 2 (top right); Case 3 (bottom left); Case 4 (bottom right).



Figure 7.3: Errors vs Iteration numbers for **Example 7.3.16**: **Case I** (top left); **Case II** (top right); **Case III** (bottom left); **Case IV** (bottom right).

7.4.1 Main result

Lemma 7.4.1. Let X be an Hadamard space and $f_j: X \to (-\infty, +\infty], j = 1, 2, \cdots, m$ be a finite family of proper, convex and lower semicontinuous functions. Let $g: X \to X$ be a contraction with coefficient $\sigma \in [0, 1)$. Let $J_{\lambda_n}^{f_m}: X \to X$ be the resolvents of a finite family of proper, convex and lower semicontinuous functions. Let $T_i: X \to X$, $i = 1, 2, \cdots, N$ be a finite family of L-Lipschitzian and quasi-pseudocontractive mappings with $L \ge 1$ such that T_i is demiclosed. Suppose that $\Gamma := \{\bigcap_{i=1}^N F(T_i) \cap (\bigcap_{j=1}^m \operatorname{argmin}_{y \in X} f_j(y))\} \neq \emptyset$ and for arbitrary $x_1, u \in X$, the sequence $\{x_n\}$ is generated by

$$\begin{pmatrix}
w_n = \theta_n x_n \oplus (1 - \theta_n) x_{n+1}, \\
y_n = \beta_0 w_n \oplus \beta_1 J_{\lambda_n}^{f_1} w_n \oplus \beta_2 J_{\lambda_n}^{f_2} w_n \oplus \dots \oplus \beta_n J_{\lambda_n}^{f_m} w_n, \\
u_n = \beta_{n,0} y_n \oplus (1 - \beta_{n,0}) \sum_{i=1}^N \oplus \frac{\beta_{n,i}}{(1 - \beta_{n,0})} \left((1 - \epsilon_{n,i}) y_n \oplus \epsilon_{n,i} T_i ((1 - \gamma_{n,i}) y_n \oplus \gamma_{n,i} T_i y_n) \right)^{(7.4.1)} \\
x_{n+1} = \alpha_n g(x_n) \oplus \psi_n w_n \oplus \tau_n u_n,
\end{cases}$$

where $\lambda_n > \lambda > 0$, $\{\alpha_n\}, \{\psi_n\}, \{\tau_n\}, \{\theta_n\}$ and $\{\beta_{n,i}\}$ are sequences in (0,1) satisfying;

(i)
$$\alpha_n + \psi_n + \tau_n = 1$$

(*ii*)
$$\sum_{i=0}^{N} \beta_{n,i} = 1$$
,
(*iii*) $\sum_{i=0}^{N} \beta_i = 1$,
(*iv*) $0 < \epsilon_{n,i} < \gamma_{n,i} < \frac{1}{1+\sqrt{1+L^2}}, \quad i = 1, 2, \cdots$

Then $\{x_n\}$ is bounded.

Proof. First observe that by Remark 2.5.29, Algorithm 7.4.1 is well defined. let $p \in \Gamma$, then from the definition of y_n in Algorithm 7.4.1 and Lemma 2.3.23 we have

, N.

$$d(y_n, p) = d(\beta_0 w_n \oplus \beta_1 J_{\lambda_n}^{f_1} \oplus \beta_2 J_{\lambda_n}^{f_2} \oplus \dots \oplus \beta_n J_{\lambda_n}^{f_m} w_n, p)$$

$$\leq \beta_0 d(w_n, p) + \beta_1 d(w_n, p) + \beta_2 d(w_n, p) \dots + \beta_n d(w_n, p)$$

$$\leq d(w_n, p).$$
(7.4.2)

Let $K_i y_n := \left((1 - \epsilon_{n,i}) y_n \oplus \epsilon_{n,i} T_i ((1 - \gamma_{n,i}) y_n \oplus \gamma_{n,i} T_i y_n) \right)$, then we have from Algorithm

7.4.1, Lemma 2.5.28, Remark 2.5.29 and (7.4.2) that

$$d(u_{n}, p) = d(\beta_{n,0}y_{n} \oplus (1 - \beta_{n,0})\sum_{i=1}^{N} \oplus \frac{\beta_{n,i}}{(1 - \beta_{n,0})}K_{i}y_{n}, p)$$

$$\leq \beta_{n,0}d(y_{n}, p) + \sum_{i=1}^{N}\beta_{n,i}d(K_{i}y_{n}, p)$$

$$\leq \beta_{n,0}d(y_{n}, p) + \sum_{i=1}^{N}\beta_{n,i}d(y_{n}, p)$$

$$= d(y_{n}, p)$$

$$\leq d(w_{n}, p).$$
(7.4.3)

We obtain from the definition of w_n in Algorithm 7.4.1 and Lemma 2.3.23 that

$$d(w_n, p) \le \theta_n d(x_n, p) + (1 - \theta_n) d(x_{n+1}, p).$$
(7.4.4)

From the definition of x_{n+1} in Algorithm (7.4.1), Condition (i), (7.4.3) and (7.4.4), we have

$$d(x_{n+1}, p) = d(\alpha_n g(x_n) \oplus \psi_n w_n \oplus \tau_n u_n, p)$$

$$\leq \alpha_n d(g(x_n), p) + \psi_n d(w_n, p) + \tau_n d(u_n, p)$$

$$\leq \alpha_n \sigma d(x_n, p) + \alpha_n d(g(p), p) + \psi_n d(w_n, p) + \tau_n d(u_n, p)$$

$$\leq \alpha_n \sigma d(x_n, p) + \alpha_n d(g(p), p) + \psi_n d(w_n, p) + \tau_n d(w_n, p)$$

$$\leq \alpha_n \sigma d(x_n, p) + \alpha_n d(g(p), p) + (1 - \alpha_n) \theta_n d(x_n, p) + (1 - \alpha_n) (1 - \theta_n) d(x_{n+1}, p)$$

$$= (\alpha_n \sigma + \theta_n (1 - \alpha_n)) d(x_n, p) + (1 - \alpha_n) (1 - \theta_n) d(x_{n+1}, p) + \alpha_n d(g(p), p),$$

which implies that

$$(1 - ((1 - \alpha_n)(1 - \theta_n)))d(x_{n+1}, p) \le (\alpha_n \sigma + \theta_n(1 - \alpha_n))d(x_n, p) + \alpha_n d(g(p), p).$$
(7.4.5)

Since $\alpha_n \in (0, 1)$ and $\sigma \in [0, 1)$, we have from (7.4.5) that

$$d(x_{n+1}, p) \leq \frac{\alpha_n \sigma + \theta_n (1 - \alpha_n)}{1 - ((1 - \alpha_n)(1 - \theta_n))} d(x_n, p) + \frac{\alpha_n}{1 - ((1 - \alpha_n)(1 - \theta_n))} d(g(p), p)$$

$$\leq 1 - \frac{\alpha_n (1 - \sigma)}{1 - ((1 - \alpha_n)(1 - \theta_n))} d(x_n, p)$$

$$+ \frac{\alpha_n (1 - \sigma)}{1 - ((1 - \alpha_n)(1 - \theta_n))} \left(\frac{1}{1 - \sigma} d(g(p), p)\right).$$

Thus, we have

$$d(x_{n+1}, p) \le \max\left\{d(x_n, p), \frac{1}{1 - \sigma}d(g(p), p)\right\}$$

$$\vdots$$

$$\le \max\left\{d(x_1, p), \frac{1}{1 - \sigma}d(g(p), p)\right\},\$$

which implies that $\{x_n\}$ is bounded. Consequently, $\{g(x_n)\}, \{w_n\}, \{y_n\}$ and $\{u_n\}$ are all bounded.

Theorem 7.4.2. Let X be an Hadamard space and $f_j: X \to (-\infty, +\infty], \quad j = 1, 2, \cdots, m$ be a finite family of proper, convex and lower semicontinuous functions. Let $g: X \to X$ be a contraction with coefficient $\sigma \in [0, 1)$. Let $J_{\lambda_n}^{f_m}: X \to X$ be the resolvents of a finite family of proper, convex and lower semicontinuous functions. Let $T_i: X \to X$, $i = 1, 2, \cdots, N$ be a finite family of L-Lipschitzian and quasi-pseudocontractive mappings with $L \ge 1$ such that T_i is demiclosed. Suppose that $\Gamma := \{\bigcap_{i=1}^N F(T_i) \cap (\bigcap_{j=1}^m \operatorname{argmin}_{y \in X} f_j(y))\} \neq \emptyset$ and the sequence $\{x_n\}$ is generated by Algorithm (7.4.1), where $\lambda_n > \lambda > 0$, $\{\alpha_n\}, \{\psi_n\}, \{\tau_n\}, \{\theta_n\}$ and $\{\beta_{n,i}\}$ are sequences in (0, 1) satisfying the following conditions:

(i)
$$\lim_{n \to \infty} \alpha_n = 0$$
 and $\sum_{n=1}^{\infty} \alpha_n = \infty$;

(*ii*)
$$0 < a < \epsilon_{n,i} < \gamma_{n,i} < b < \frac{1}{1+\sqrt{1+L^2}}, \quad i = 1, 2, \cdots, N;$$

(*iii*)
$$\sum_{i=0}^{N} \beta_{n,i} = 1, \sum_{i=0}^{N} \beta_i = 1 \text{ and } 0 < c \le \beta_{n,i} \le d < 1, \text{ for all } n \ge 1,$$

(*iv*) $\alpha_n + \psi_n + \tau_n = 1.$

Then, $\{x_n\}$ converges strongly to a point $p = P_{\Gamma}g(p)$, where P_{Γ} is the metric projection of X onto Γ .

Proof. We can rewrite Algorithm 7.4.1 as follows;

$$\begin{cases} w_n = \theta_n x_n \oplus (1 - \theta_n) x_{n+1}, \\ y_n = \beta_0 w_n \oplus \beta_1 J_{\lambda_n}^{f_1} w_n \oplus \beta_2 J_{\lambda_n}^{f_2} w_n \oplus \dots \oplus \beta_n J_{\lambda_n}^{f_m} w_n, \\ u_n = \beta_{n,0} y_n \oplus (1 - \beta_{n,0}) \sum_{i=1}^N \oplus \frac{\beta_{n,i}}{(1 - \beta_{n,0})} K_i y_n, \\ z_n = \left(\frac{\psi_n}{1 - \alpha_n} w_n \oplus \frac{\tau_n}{1 - \alpha_n} u_n\right), \\ x_{n+1} = \alpha_n g(x_n) \oplus (1 - \alpha_n) z_n. \end{cases}$$
(7.4.6)

Applying Lemma 2.5.30 to the definition of u_n in Algorithm 7.4.6 we obtain

$$d^{2}(u_{n},p) \leq \beta_{n,0}d^{2}(y_{n},p) + \sum_{i=1}^{N} \beta_{n,i}d^{2}(K_{i}y_{n},p) - \beta_{n,0}\sum_{i=1}^{N} \beta_{n,i}d^{2}(K_{i}y_{n},y_{n})$$
$$\leq d^{2}(y_{n},p) - \beta_{n,0}\sum_{i=1}^{N} \beta_{n,i}d^{2}(K_{i}y_{n},y_{n}).$$
(7.4.7)

We obtain from the definition of w_n in Algorithm 7.4.1 and Lemma 2.3.23 that

$$d^{2}(w_{n}, p) \leq \theta_{n} d^{2}(x_{n}, p) + (1 - \theta_{n}) d^{2}(x_{n+1}, p).$$
(7.4.8)

We obtain from the definition of x_{n+1} and z_n in Algorithm 7.4.6, (7.4.2), (7.4.7) and (7.4.8) that

$$d^{2}(x_{n+1}, p) = d^{2}(\alpha_{n}g(x_{n}) \oplus (1 - \alpha_{n})z_{n}, p)$$

$$\leq \alpha_{n}^{2}d^{2}(g(x_{n}), p) + (1 - \alpha_{n})^{2}d^{2}(z_{n}, p) + 2\alpha_{n}(1 - \alpha_{n})\langle \overrightarrow{g(x_{n})p}, \overrightarrow{z_{n}p} \rangle$$

$$\leq \alpha_{n}^{2}d^{2}(g(x_{n}), p) + \beta_{n}d^{2}(w_{n}, p) + \tau_{n}d^{2}(u_{n}, p) + 2\alpha_{n}(1 - \alpha_{n})\langle \overrightarrow{g(x_{n})p}, \overrightarrow{z_{n}p} \rangle$$

$$\leq \alpha_{n}^{2}d^{2}(g(x_{n}), p) + \beta_{n}d^{2}(w_{n}, p) + \tau_{n}d^{2}(y_{n}, p) - \beta_{n,0}\sum_{i=1}^{N}\tau_{n}\beta_{n,i}d^{2}(y_{n}, K_{i}y_{n})$$

$$+ 2\alpha_{n}(1 - \alpha_{n})\langle \overrightarrow{g(x_{n})p}, \overrightarrow{z_{n}p} \rangle$$

$$\leq \alpha_{n}^{2}d^{2}(g(x_{n}), p) + (1 - \alpha_{n})d^{2}(w_{n}, p) - \beta_{n,0}\sum_{i=1}^{N}\tau_{n}\beta_{n,i}d^{2}(y_{n}, K_{i}y_{n})$$

$$+ 2\alpha_{n}(1 - \alpha_{n})\langle \overrightarrow{g(x_{n})p}, \overrightarrow{z_{n}p} \rangle$$

$$(7.4.10)$$

$$\leq \alpha_{n}^{2}d^{2}(g(x_{n}), p) + (1 - \alpha_{n})\left(\theta_{n}d^{2}(x_{n}, p) + (1 - \theta_{n})d^{2}(x_{n+1}, p)\right)$$

$$- \beta_{n,0}\sum_{i=1}^{N}\tau_{n}\beta_{n,i}d^{2}(y_{n}, K_{i}y_{n}) + 2\alpha_{n}(1 - \alpha_{n})\langle \overrightarrow{g(x_{n})p}, \overrightarrow{z_{n}p} \rangle$$

$$(7.4.11)$$

which implies that

$$\left(1 - \left((1 - \alpha_n)(1 - \theta_n)\right)\right) d^2(x_{n+1}, p) \le \left((1 - \alpha_n)\theta_n\right) d^2(x_n, p) + \alpha_n \left[\alpha_n d^2(g(x_n), p) + 2(1 - \alpha_n)\langle \overrightarrow{g(x_n)p}, \overrightarrow{z_np} \rangle\right].$$

Hence, from the previous inequality we obtain

$$d^{2}(x_{n+1}, p) \leq \frac{(1 - \alpha_{n})\theta_{n}}{\theta_{n} + \alpha_{n}(1 - \theta_{n})} d^{2}(x_{n}, p) + \frac{\alpha_{n}}{\theta_{n} + \alpha_{n}(1 - \theta_{n})} \left[\alpha_{n} d^{2}(g(x_{n}), p) + 2(1 - \alpha_{n}) \langle \overrightarrow{g(x_{n})p}, \overrightarrow{z_{n}p} \rangle \right] = \left[1 - \frac{\alpha_{n}}{\theta_{n} + \alpha_{n}(1 - \theta_{n})} \right] d^{2}(x_{n}, p) + \frac{\alpha_{n}}{\theta_{n} + \alpha_{n}(1 - \theta_{n})} d_{n},$$
(7.4.12)

where

$$d_n = \alpha_n d^2(g(x_n), p) + 2(1 - \alpha_n) \langle \overrightarrow{g(x_n)p}, \overrightarrow{z_np} \rangle$$

Using Lemma 2.5.36, we need to show that $\limsup_{k\to\infty} d_{n_k} \leq 0$ for every subsequence $\{d(x_{n_k}, p)\}$ of $\{d(x_n, p)\}$ satisfying the condition

$$\liminf_{k \to \infty} (d(x_{n_k+1}, p) - d(x_{n_k}, p)) \ge 0.$$
(7.4.13)

Now suppose that $\{d(x_{n_k}, p)\}$ is a subsequence of $\{d(x_n, p)\}$ such that (7.4.13) holds then $\liminf_{k \to \infty} (d^2(x_{n_k+1}, p) - d^2(x_{n_k}, p)) = \liminf_{k \to \infty} (d(x_{n_k+1}, p) - d(x_{n_k}, p))(d(x_{n_k+1}, p) + d(x_{n_k}, p)).$

From (7.4.9), we have

$$\begin{split} \limsup_{k \to \infty} \left(\beta_{n_k,0} \sum_{i=1}^N \tau_{n_k} \beta_{n_k,i} d^2(y_{n_k}, K_i y_{n_k}) \right) &\leq \limsup_{k \to \infty} \left(\alpha_{n_k}^2 d^2(g(x_{n_k}), p) \right) \\ &+ \limsup_{k \to \infty} \left(1 - \alpha_{n_k} \right) \left[\theta_{n_k} d^2(x_{n_k}, p) \right] \\ &+ \limsup_{k \to \infty} \left(1 - \alpha_{n_k} \right) \left[(1 - \theta_{n_k}) d^2(x_{n_k+1}, p) \right] \\ &+ \limsup_{k \to \infty} \left(2\alpha_{n_k} (1 - \alpha_{n_k}) \langle \overline{g(x_{n_k})} p, \overline{z_{n_k}} p \rangle \right) \\ &- \limsup_{k \to \infty} \left(d^2(x_{n_k+1}, p) \right) \\ &\leq \limsup_{k \to \infty} \left(\alpha_{n_k}^2 d^2(g(x_{n_k}), p) \right) \\ &+ \limsup_{k \to \infty} \left(\theta_{n_k} \left[d^2(x_{n_k}, p) - d^2(x_{n_k+1}, p) \right] \right) \\ &+ \limsup_{k \to \infty} \left(2\alpha_{n_k} (1 - \alpha_{n_k}) \langle \overline{g(x_{n_k})} p, \overline{z_{n_k}} p \rangle \right) \end{split}$$

which implies from the conditions on the control parameter that

$$\lim_{k \to \infty} d(y_{n_k}, K_i y_{n_k}) = 0, \quad i = 1, 2, \cdots, N.$$
(7.4.14)

We also have from Algorithm 7.4.6 and (7.4.14) that

$$d(u_{n_k}, y_{n_k}) = d\left(\beta_{n_k, 0} y_{n_k} \oplus (1 - \beta_{n_k, 0}) \sum_{i=1}^N \oplus \frac{\beta_{n_k, i}}{(1 - \beta_{n_k, 0})} K_i y_{n_k}, y_{n_k}\right)$$

$$\leq \sum_{i=1}^N \beta_{n_k, i} d(K_i y_{n_k}, y_{n_k}) \to 0 \quad \text{as} \ k \to \infty.$$

Hence,

$$\lim_{k \to \infty} d(u_{n_k}, y_{n_k}) = 0.$$
(7.4.15)

Also,

$$d(y_{n_k}, w_{n_k}) = d(\beta_0 w_{n_k} \oplus \beta_1 J_{\lambda_{n_k}}^{f_1} w_{n_k} \oplus \beta_2 J_{\lambda_{n_k}}^{f_2} w_{n_k} \oplus \dots \oplus \beta_{n_k} J_{\lambda_{n_k}}^{f_m} w_{n_k}, w_{n_k})$$

$$\leq d(w_{n_k}, w_{n_k}) \to 0, \quad \text{as} \quad k \to \infty.$$

Thus,

$$\lim_{k \to \infty} d(y_{n_k}, w_{n_k}) = 0.$$
(7.4.16)

From (7.4.15) and (7.4.16), we obtain that

$$\lim_{k \to \infty} d(u_{n_k}, w_{n_k}) = 0 \tag{7.4.17}$$

From Algorithm 7.4.6 and (7.4.17), we obtain

$$d(u_{n_k}, z_{n_k}) \le \frac{\beta_{n_k}}{1 - \alpha_{n_k}} d(u_{n_k}, w_{n_k}) + \frac{\tau_{n_k}}{1 - \alpha_{n_k}} d(u_{n_k}, u_{n_k}) \to 0, \quad \text{as} \quad k \to \infty.$$
(7.4.18)

Hence, from (7.4.15) and (7.4.18) we have

$$\lim_{k \to \infty} d(z_{n_k}, y_{n_k}) = 0 \tag{7.4.19}$$

From Algorithm 7.4.6 and the condition on θ_{n_k} we have

$$\lim_{k \to \infty} d(w_{n_k}, x_{n_k}) = 0.$$
(7.4.20)

From (7.4.16) and (7.4.20), we have

$$\lim_{k \to \infty} d(y_{n_k}, x_{n_k}) = 0.$$
 (7.4.21)

Using (7.4.19) and (7.4.21), we have

$$\lim_{k \to \infty} d(x_{n_k}, z_{n_k}) = 0.$$
(7.4.22)

We obtain from Algorithm 7.4.6 and the condition on α_n that

$$d(x_{n_k+1}, z_{n_k}) = 0. (7.4.23)$$

Hence, we obtain from (7.4.22) and (7.4.23) that

$$\lim_{k \to \infty} d(x_{n_k+1}, x_{n_k}) = 0$$

Since $\{x_n\}$ is bounded it follows from Lemma 2.5.31 that there exists a subsequence $\{x_{n_{k_j}}\}$ of $\{x_{n_k}\}$ which \triangle - converges to a point v. From (7.4.22), we have that there exists a subsequence $\{z_{n_{k_j}}\}$ which also \triangle - converges to v. Since T_i , $i = 1, 2, \dots, N$ is demiclosed, we obtain from Lemma 7.2.1 and (7.4.14) that $v \in \bigcap_{i=1}^N F(T_i) = \bigcap_{i=1}^N F(K_i)$. Also, $J_{\lambda_{n_k}}$ is a nonexpansive mapping for each $j = 1, 2, \dots, m$. We obtain from (7.4.21), Lemma 2.5.33 and Lemma 7.2.3 that $v \in \bigcap_{j=1}^N F(J_\lambda^{(j)}) = \bigcap_{j=1}^m \arg\min_{y \in X} f_j(y)$. Hence, $v \in \Gamma$.

Let $p = P_{\Gamma}g(p)$. Since $\{x_{n_k}\}$ is bounded, we choose a subsequence $\{x_{n_kj}\}$ which \triangle - converges to a point v. From (7.4.22), we have that $\{z_{n_k}\}$ is bounded and we chose subsequence $\{z_{n_{k_i}}\}$ which \triangle -converges to a point v such that

$$\limsup_{k \to \infty} \left\langle \overrightarrow{g(p)p}, \overrightarrow{z_{n_k}p} \right\rangle = \lim_{j \to \infty} \left\langle \overrightarrow{g(p)p}, \overrightarrow{z_{n_kj}p} \right\rangle$$
$$= \left\langle \overrightarrow{g(p)p}, \overrightarrow{vp} \right\rangle.$$

From Lemma 2.5.32, we obtain that

$$\limsup_{k \to \infty} \left\langle \overrightarrow{g(p)p}, \overrightarrow{z_{n_k}p} \right\rangle = \left\langle \overrightarrow{g(p)p}, \overrightarrow{vp} \right\rangle \le 0.$$

Hence, from (7.4.12) and the previous inequality, we obtain that $\limsup d_{n_k} \leq 0$, where

$$d_{n_k} = \left[\alpha_{n_k} d^2(g(x_{n_k}), u) + 2(1 - \alpha_n) \left\langle \overrightarrow{g(p)p}, \overrightarrow{z_{n_k}p} \right\rangle \right].$$

Applying Lemma 2.5.36 to (7.4.12), we have that $d(x_n, p) \to 0$, as $n \to \infty$. Therefore, $\{x_n\}$ converges strongly to $p = P_{\Gamma}g(p)$.

If T is a nonexpansive mapping in Theorem 7.4.2, then we obtain the following corollary.

Corollary 7.4.1. Let X be an Hadamard space and $f_j: X \to (-\infty, +\infty], \quad j = 1, 2, \cdots, m$ be a finite family of proper, convex and lower semicontinuous functions. Let $g: X \to X$ be a contraction with coefficient $\sigma \in [0, 1)$. Let $J_{\lambda_n}^{f_m}: X \to X$ be the resolvents of a finite family of proper, convex and lower semicontinuous functions. Let $T_i: X \to X$, $i = 1, 2, \cdots, N$ be a finite family of L-Lipschitzian and nonexpansive mappings with $L \ge 1$ such that T_i is demiclosed. Suppose that $\Gamma := \{\bigcap_{i=1}^N F(T_i) \cap (\bigcap_{j=1}^m \operatorname{argmin}_{y \in X} f_j(y))\} \neq \emptyset$ and for arbitrary $x_1, u \in X$, the sequence $\{x_n\}$ is generated by

$$\begin{pmatrix}
w_n = \theta_n x_n \oplus (1 - \theta_n) x_{n+1}, \\
y_n = \beta_0 w_n \oplus \beta_1 J_{\lambda_n}^{f_1} w_n \oplus \beta_2 J_{\lambda_n}^{f_2} w_n \oplus \dots \oplus \beta_n J_{\lambda_n}^{f_m} w_n, \\
u_n = \beta_{n,0} y_n \oplus (1 - \beta_{n,0}) \sum_{i=1}^N \oplus \frac{\beta_{n,i}}{(1 - \beta_{n,0})} \left((1 - \epsilon_{n,i}) y_n \oplus \epsilon_{n,i} T_i ((1 - \gamma_{n,i}) y_n \oplus \gamma_{n,i} T_i y_n) \right), \\
\chi_{n+1} = \alpha_n g(x_n) \oplus \psi_n w_n \oplus \tau_n u_n,
\end{cases}$$
(7.4.24)

where $\lambda_n > \lambda > 0$, $\{\alpha_n\}, \{\psi_n\}, \{\tau_n\}, \{\theta_n\}$ and $\{\beta_{n,i}\}$ are sequences in (0,1) satisfying conditions (i)-(iv) then, $\{x_n\}$ converges strongly to a point $p = P_{\Gamma}g(p)$, where P_{Γ} is the metric projection of X onto Γ .

By setting $\theta_n = \frac{1}{2}$ for $n \ge 1$ in Theorem 7.4.2, we obtain the following corollary.

Corollary 7.4.2. Let X be an Hadamard space and $f_j: X \to (-\infty, +\infty], \quad j = 1, 2, \cdots, m$ be a finite family of proper, convex and lower semicontinuous functions. Let $g: X \to X$ be a contraction with coefficient $\sigma \in [0, 1)$. Let $J_{\lambda_n}^{f_m}: X \to X$ be the resolvents of a finite family of proper, convex and lower semicontinuous functions. Let $T_i: X \to X$, $i = 1, 2, \cdots, N$ be a finite family of L-Lipschitzian and quasi-pseudocontractive mappings with $L \ge 1$ such that T_i is demiclosed. Suppose that $\Gamma := \{\bigcap_{i=1}^N F(T_i) \cap (\bigcap_{j=1}^m \operatorname{argmin}_{y \in X} f_j(y))\} \neq \emptyset$ and for arbitrary $x_1, u \in X$, the sequence $\{x_n\}$ is generated by

$$\begin{cases} w_n = \left(\frac{x_n \oplus x_{n+1}}{2}\right), \\ y_n = \beta_0 w_n \oplus \beta_1 J_{\lambda_n}^{f_1} w_n \oplus \beta_2 J_{\lambda_n}^{f_2} w_n \oplus \dots \oplus \beta_n J_{\lambda_n}^{f_m} w_n, \\ u_n = \beta_{n,0} y_n \oplus (1 - \beta_{n,0}) \sum_{i=1}^N \oplus \frac{\beta_{n,i}}{(1 - \beta_{n,0})} \left((1 - \epsilon_{n,i}) y_n \oplus \epsilon_{n,i} T_i((1 - \gamma_{n,i}) y_n \oplus \gamma_{n,i} T_i y_n)\right)^{7.4.25}, \\ x_{n+1} = \alpha_n g(x_n) \oplus \psi_n w_n \oplus \tau_n u_n, \end{cases}$$

where $\lambda_n > \lambda > 0$, $\{\alpha_n\}, \{\psi_n\}, \{\tau_n\}, \{\theta_n\}$ and $\{\beta_{n,i}\}$ are sequences in (0,1) satisfying conditions (i)-(iv), then, $\{x_n\}$ converges strongly to a point $p = P_{\Gamma}g(p)$, where P_{Γ} is the metric projection of X onto Γ .

By setting $g(x_n) = u$ for all $n \ge 1, u \in X$ fixed in Theorem 7.4.2, we obtain the following corollary with the Halpern-type algorithm

Corollary 7.4.3. Let X be an Hadamard space and $f_j: X \to (-\infty, +\infty], \quad j = 1, 2, \cdots, m$ be a finite family of proper, convex and lower semicontinuous functions. Let $J_{\lambda_n}^{f_m}: X \to X$ be the resolvents of a finite family of proper, convex and lower semicontinuous functions. Let $T_i: X \to X, \quad i = 1, 2, \cdots, N$ be a finite family of L-Lipschitzian and quasi-pseudocontractive mappings with $L \ge 1$ such that T_i is demiclosed. Suppose that $\Gamma := \{\bigcap_{i=1}^N F(T_i) \cap (\bigcap_{j=1}^m \operatorname{argmin}_{y \in X} f_j(y))\} \neq \emptyset$ and for arbitrary $x_1, u \in X$, the sequence $\{x_n\}$ is generated by

$$\begin{pmatrix}
w_n = \theta_n x_n \oplus (1 - \theta_n) x_{n+1}, \\
y_n = \beta_0 w_n \oplus \beta_1 J_{\lambda_n}^{f_1} w_n \oplus \beta_2 J_{\lambda_n}^{f_2} w_n \oplus \dots \oplus \beta_n J_{\lambda_n}^{f_m} w_n, \\
u_n = \beta_{n,0} y_n \oplus (1 - \beta_{n,0}) \sum_{i=1}^N \oplus \frac{\beta_{n,i}}{(1 - \beta_{n,0})} \left((1 - \epsilon_{n,i}) y_n \oplus \epsilon_{n,i} T_i ((1 - \gamma_{n,i}) y_n \oplus \gamma_{n,i} T_i y_n) \right)^{7,4.26}, \\
\chi_{n+1} = \alpha_n u \oplus \psi_n w_n \oplus \tau_n u_n,$$

where $\lambda_n > \lambda > 0$, $\{\alpha_n\}, \{\psi_n\}, \{\tau_n\}, \{\theta_n\}$ and $\{\beta_{n,i}\}$ are sequences in (0,1) satisfying conditions (i)-(iv) then, $\{x_n\}$ converges strongly to a point $p = P_{\Gamma}g(p)$, where P_{Γ} is the metric projection of X onto Γ .

CHAPTER 8

Conclusion, Contributions to Knowledge and Future research

8.1 Conclusion

In this thesis, we studied optimization and FPP in the frameworks of Hilbert, Banach and Hadamard spaces. We developed several effective algorithms for solving these problems. Our results complement several existing results in the literature. In Chapter 1 of this thesis, we gave a comprehensive background of our study which highlighted some of the importance of the optimization problems and FPP. We discussed some iterative schemes for solving these aforementioned problems. Furthermore, we discussed our research problems and motivation. Finally, we discussed the objectives of our study and presented the organization of our study. In Chapter 2 of this thesis, we discussed the geometry of the Hilbert, Banach and Hadamard spaces and presented some basic definitions, terms and concepts used in this study. We gave a detailed literature review of past works which motivated our study and presented some results that were important to our study. In Chapter 3, we studied the VIPs and FPP in the framework of a real Hilbert space. We studied the SVIPs and GVIP in the framework of real Hilbert spaces in Chapter 4. In Chapter 5 of this thesis, we presented our results on SE_qP , V_qIP and SEP in the framework of Hilbert spaces. Furthermore, we extended our results from the framework of Hilbert spaces to the framework work of Banach spaces. In Chapter 6 of this thesis, we presented our results on VIP, V_a IP and FPP in the framework of Banach space. Finally, we extended our results from the framework of Banach space to the framework of the Hadamard space. In Chapter 7 of this thesis, we presented our results on MPs in Hadamard spaces. In each of these chapters, we presented numerical examples of our algorithms in comparison with other existing algorithms in the literature to show the applicability of our methods. We also presented where necessary some applications of our results.

8.2 Contributions to Knowledge

Generally, the results obtained in this thesis extend and improve several other existing results in the literature in the frameworks of Hilbert, Banach and Hadamard spaces. In Chapter 3 of this thesis, we studied and improved several existing works on VIP and FPP in the framework of Hilbert spaces. In Section 3.3, we generalized the result of Nadeshkina and Takahashi [185] from approximating the common solution of the VIP and FPP of a nonexpansive mapping to approximating the common solution of VIP and FPP of an infinite family of strict pseudo-contractive mappings. We obtained a strong convergence result of our proposed method. Furthermore, we applied our result to find a common solution of VIP and ZPP for an infinite family of maximal monotone operators. In Section 3.4, we proposed two new relaxed inertial subgradient extragradient methods, and proved that they converge weakly to a solution of VIP when the operator A is quasimonotone and Lipschitz continuous, and when it is Lipschitz continuous without any form of monotonicity. The assumptions on the inertial and relaxation factors in this work, are weaker than those in many works in literature for solving VIPs. Our work generalizes the results of Liu *et al.* [162] and Ye *et al.* [269]. In Section 3.5, we generalized the result of Yin *et al.* [270] from approximating a common solution of VIP and FPP of a pseudocontractive mapping when the cost operator is quasimonotone, Lipschitz continuous and sequentially weakly continuous to approximating the common solution of VIP and FPP of a quasi-pseudocontractive mapping when the cost operator is quasimonotone, uniformly continuous (which is a weaker condition that the Lipschitz continuity) and without the sequentially weakly continuous condition. We obtained a strong convergence result of our proposed method to the minimum-norm solution of the aforementioned problem. Furthermore, we applied our result to image recovery.

In Chapter 4 of this thesis, we studied the SVIPs and GVIP in the framework of real Hilbert spaces. In Section 4.2, we introduced two new relaxed inertial Tseng's forward-backwardforward for solving the SVIP in real Hilbert spaces without any product space formulation when the underlying operator is monotone and Lipschitz continuous. We proved that the proposed methods converges strongly to the minimum-norm solution of the SVIP. In Section 4.3, our results extend and improve the works of Pham *et al.* [206] and Reich *et* al. [213] by introducing two new inertial projection and contraction methods for solving the SVIP when both underlying operators are pseudomonotone, Lipschitz continuous and without any product space reformulation of the original problem with minimized number of projections per iteration. Our results in Section 4.4 generalized the works of Tian and Jiang [244] and Ogwo *et al.* [192] by introducing two inertial projection and contraction methods for solving the SVIP when the underlying operators are pseudomonotone and Lipschitz continuous without the sequentially weakly continuity condition. In Section 4.5, we proposed a viscosity-type iterative method for solving the GVIP in real Hilbert spaces. We obtained a strong convergence result of our proposed method to a solution of the GVIP.

In Chapter 5, we studied the SE_qP , V_qIP and SEP in the framework of Hilbert spaces. In Section 5.2, we generalized the result of Latif and Eslamian [159] from approximating the common solution of the split equalities of EP, Lipschitz monotone VIP and FPP of nonexpansive semigroups satisfying the u.a.r condition to approximating the common solution of the split equalities of EP, non-Lipschitz pseudomonotone VIP and FPP of nonexpansive semigroups without the u.a.r condition. We obtained a strong convergence result of our proposed method to a common solution of the aforementioned problem. In Section 5.3, we improved the result of Cholamjiak *et al.* [70] from approximating the common solution of SEP and V_q IP in Hilbert spaces using the summability condition to approximating the common solution of SEP and V_q IP without the summability condition. Furthermore, we proved that the sequence generated by our proposed method converges strongly to a minimum solution (which finds application in several practical problems) of the aforementioned problem.

Next, we extended some of our results in the previous chapters to the framework of Banach spaces. In Section 6.2, we extended the result of Tan *et al.* [240] from approximating the solution of the VIP in a real Hilbert space to a 2-uniformly convex real Banach space. In Section 6.3, we extended the result of Long *et al.* [163] from approximating the SV_qIP in Hilbert spaces to approximating a common solution of finite family of SMV_qIP and FPP for nonexpansive mappings between a Banach and Hilbert space without the prior knowledge of the operator norm. Furthermore, we applied our results to study SFP and SMP.

Finally, we extended some of our earlier results to the framework of Hadamard spaces. We studied the MPs and FPP in the framework of Hadamard spaces. In Section 7.2, we generalized the result of Chang *et al.* [70] from approximating a common solution of a finite family of MPs and FPP for two demicontractive mappings in Hadamard spaces to approximating a common solution of a finite family of MPs and FPP of a finite family of quasi-pseudocontractive mapping in Hadamard spaces. Furthermore, we extended this method to a more general method involving the resolvents of multivalued monotone operators and established that it converges strongly to a common zero of a finite family of quasi-pseudocontractive mappings in Hadamard spaces. In Section 7.3, we generalized the result of Ahmad and Ahmad [4] for approximating a common solution of a MP and FPP of a nonexpansive mapping using the viscosity implicit midpoint rule in Hadamard spaces to approximating a common solution of MP and FPP of a quasi-pseudocontractive mapping using the proximating a common solution of MP and FPP of a quasi-pseudocontractive mapping using the viscosity type iterative method for a generalized implicit rule in Hadamard spaces.

8.3 Future research

Some of the possible research problems we will like to consider for future research are highlighted below.

In Chapter 3 of this thesis, the VIPs and FPP were studied in the framework of a real Hilbert space. In future research, we desire to extend our results from the framework of Hilbert spaces to more general Banach spaces and possibly, Hadamard spaces. Also, in Chapter 4, the SVIPs were studied when the cost operators are either pseudomonotone or monotone mappings in the framework of Hilbert spaces. It is known that the quasimonotone operators are more general and applicable than the pseudomonotone and monotone

mappings. In future, we intend to extend our results on SVIPs to when the cost operators are quasimonotone mappings. Furthermore, we hope to extend our results on SVIPs to a more general Banach space.

Note that the split equalities of EP, VIP and FPP considered in this thesis were studied in the framework of Hilbert spaces when the cost operator is pseudomonotone and uniformly continuous. Also, the V_q IP and SEP considered in this thesis were studied in the framework of Hilbert spaces when the cost operators were monotone and Lipschitz continuous. In our future research, we desire to extend our results on SE_qP , V_q IP and SEP to the framework of more general Banach spaces when the cost operators are quasimonotone and uniformly continuous.

In Chapter 6, we studied the VIP in the framework of Banach space when the operator is pseudomonotone and non-Lipschitz. We also studied the SMV_qIPs and FPP of a nonexpansive mapping between a Banach space and a Hilbert space. In future, we intend to extend these results to a more general mapping when the cost operator is quasimonotone and uniformly continuous in the framework of Hadamard space. Furthermore, we note that the MPs considered in this thesis were studied in the framework of the Hadamard spaces. In future, we desire to study these results in the framework of *p*-uniformly convex metric spaces.

Bibliography

- M. Abbas, M. Al Sharani, Q. H. Ansari, G. S. Iyiola, Y. Shehu, Iterative methods for solving proximal split minimization problem. *Numer. Algorithms*, **78** (1) (2018), 193-215.
- [2] R. P. Agarwal, D. O'Regan, D. R. Sahu, Iterative construction of fixed points of nearly asymptotically nonexpansive mappings, *J. Nonlinear Convex Anal.*, 8 (2007), 61-79.
- [3] R. P. Agarwal, D. O'Regan, D. R. Sahu, Fixed point theory for Lipschitzian-type mappings with applications, Springer, 2009.
- [4] I. Ahmad, M. Ahmad, An implicit viscosity technique of nonexpansive mapping in Cat (0) spaces, Open J. Math. Anal, 1 (2017), 1-12.
- [5] T.O. Alakoya, A. Taiwo, O.T. Mewomo, Y.J. Cho, An iterative algorithm for solving variational inequality, generalized mixed equilibrium, convex minimization and zeros problems for a class of nonexpansive-type mappings, Ann. Univ. Ferrara Sez. VII Sci. Mat., 67 (1) (2021), 1-31.
- [6] T.O. Alakoya, A. Taiwo, O.T. Mewomo, On system of split generalised mixed equilibrium and fixed point problems for multivalued mappings with no prior knowledge of operator norm, *Fixed Point Theory*, 23 (1), 45-74, (2022).
- [7] T.O. Alakoya, A.O.-E Owolabi, O.T. Mewomo, Inertial algorithm for solving split mixed equilibrium and fixed point problems for hybrid-type multivalued mappings with no prior knowledge of operator norm, J. Nonlinear Convex Anal., 23 (11) (2022), 2479-2510.
- [8] T.O. Alakoya, L.O. Jolaoso, O.T. Mewomo, A self adaptive inertial algorithm for solving split variational inclusion and fixed point problems with applications, J. Ind. Manag. Optim., 18 (1) (2022), p 239.

- [9] T.O. Alakoya, L.O. Jolaoso, O.T. Mewomo, Modified inertial subgradient extragradient method with self adaptive step size for solving monotone variational inequality and fixed point problems, *Optim.*, **70** (2) (2021), 545-574.
- [10] T.O. Alakoya, O.T. Mewomo, Viscosity S-iteration method with inertial technique and self-adaptive step size for split variational inclusion, equilibrium and fixed point problems, *Comput. Appl. Math.*, **41** (1) (2022), Paper 39, 31 pp.
- [11] T.O. Alakoya, L.O. Jolaoso, O.T. Mewomo, Strong convergence and bounded perturbation resilience of a modified forward-backward and splitting algorithm and its application, J. Nonlinear Convex Anal., 23 (4) (2022), 653-682.
- [12] T. O. Alakoya, O. T. Mewomo, Y. Shehu, Strong convergence results for quasimonotone variational inequalities, *Mathematical Methods of Operations Research*, 95 (2022), 249-279.
- [13] T.O. Alakoya, V.A. Uzor, O.T. Mewomo, J.-C. Yao, On a system of monotone variational inclusion problems with fixed-point constraint, *J. Inequ. Appl.* **2022** (2022), Paper No. 47, 33 pp.
- [14] T.O. Alakoya, V.A. Uzor, O.T. Mewomo, A new projection and contraction method for solving split monotone variational inclusion, pseudomonotone variational inequality, and common fixed point problems, *Comput. Appl. Math.*, (2022), DOI :10.1007/s40314-022-02138-0.
- [15] Y. Alber, J. L. Li, The connection between the metric and generalized projection operators in Banach spaces, *Acta Math. Sin.* **23** (2007), 110-1120.
- [16] Y. I. Alber, Metric and generalized projection operators in Banach spaces:Properties and applications, in:A.G. Kartsatos (Ed.), Theory and Applications of Nonlinear Operators of Accretive and Monotone Type, in:Lecture Notes *Pure Appl.Math.*, vol.178, Dekker, NewYork, (1996), 15–50.
- [17] A. D. Aleksandrov, A theorem on triangles in a metric space and some of its applications, Trudy Mat. Inst. Steklov., Izdat. Akad. Nauk SSSR, Moscow, 38 (1951), 5-23.
- [18] F. Alvarez, H. Attouch, An inertial proximal method for maximal monotone operators via discretization of a nonlinear oscillator with damping, *Set-Valued Anal.*, 9, (2001), 3-11.
- [19] K. Alyani, M. Congedo, M. Moakher, Diagonality measures of Hermitian positivedefinite matrices with application to the approximate joint diagonalization problem, (528) (2017), 290-320.
- [20] K. Aoyama, F. Kohsaka, Strongly relatively nonexpansive sequences generated by firmly nonexpansive-like mappings. *Fixed Point Theory Appl.*, **95** (1) (2014), 1-13.

- [21] K. Aoyama, F. Kohsaka, W. Takahashi, Three generalizations of firmly nonexpansive mappings: their relations and continuity properties, J. Nonlinear Convex Anal., 10 (1) (2009), 131-147.
- [22] K. Aoyama, Y. Kimura, W. Takahashi, M. Toyoda, On a strongly nonexpansive sequence in Hilbert spaces, J. Nonlinear Convex Anal., 8 (2007), 471-489.
- [23] K. O. Aremu, H. A. Abass, C. Izuchukwu, O. T. Mewomo, A viscosity-type algorithm for an infinitely countable family of (f, g)-generalized k-strictly pseudononspreading mappings in CAT(0) spaces, *Analysis*, **40** (1), (2020), 19-37.
- [24] K. O. Aremu, C. Izuchukwu, G. N. Ogwo, O. T. Mewomo, Multi-step Iterative algorithm for minimization and fixed point problems in p-uniformly convex metric spaces, *J. Ind. Manag. Optim.*, **17**(4) (2021), 2161.
- [25] H. Attouch, A. Cabot, Convergence of a relaxed inertial forward-backward algorithm for structured monotone inclusions, *Appl. Math. Optim.*, 80 (3) (2019), 547-598.
- [26] H. Attouch, A. Cabot, Convergence rate of a relaxed inertial proximal algorithm for convex minimization, *Optimization*, **69** (6) (2020), 1281-1312.
- [27] H. Attouch, J. Peypouquet, P. Redont, A dynamical approach to an inertial forwardbackward algorithm for convex minimization, SIAM J. Optim., 24 (1) (2014), 232-256.
- [28] H. Attouch, J. Bolte, P. Redont, A. Soubeyran, A new class of alternating proximal minimization algorithms with costs to move, SIAM J. Optim., 18 (3) (2007), 1061-1081.
- [29] H. Attouch, A. Cabot, Convergence of a relaxed inertial proximal algorithm for maximally monotone operators, *Math. Program.*, 184 (1) (2020), 243-287.
- [30] J. P. Aubin, Optima and Equilibria: an Introduction to Nonlinear Analysis. Springer, Berlin (1993).
- [31] J-P, Aubin, I. Ekeland, Applied nonlinear analysis. New York: Wiley; 1984.
- [32] W. Auzinger, R. Frank, Asymptotic error expansions for stiff equations: an analysis for the implicit midpoint and trapezoidal rule in the strongly stiff case, *Numer. Math.*, 56 (1989), 469-499.
- [33] M. Bačák, *Convex analysis and optimization in Hadamard spaces*, De Gruyter Series in Nonlinear Analysis and Applications, De Gruyter, Berlin, **22** (2014).
- [34] M. Bačák, The proximal point algorithm in metric spaces, Israel J. Math., 194 (2), (2013), 689-701.
- [35] M. Bačák, Computing medians and means in Hadamard spaces, SIAM J. Optim., 24 (2014), 1542-1566.
- [36] C. Baiocchi, A. Capelo, Variational and quasivariational inequalities; applications to free boundary problems, New York: Wiley; 1984.
- [37] K. Ball, E. A. Carlen, E. H. Lieb, Sharp uniform convexity and smoothness inequalities for trace norms, *Invent. Math.*, **115** (1994), 463-482.
- [38] S. Banach, Théorie des operations lineires, Warsaw, 1932.
- [39] H. H. Bauschke, P. L. Combettes, Convex Analysis and Monotone Operator Theory in Hilbert Spaces, CMS Books in Mathematics, Springer, New York (2011), Vol (408).
- [40] A. Beck, M. Teboulle, A fast iterative shrinkage-thresholding algorithm for linear inverse problems, SIAM J. Imaging Sci., 2 (1) (2009), 183-202.
- [41] I. D. Berg, I. G. Nikolaev, Quasilinearization and curvature of Aleksandrov spaces, Geom. Dedicata, 133, (2008), 195-218.
- [42] D. G. Birkho, O. D. Kellog, Invariant points in function spaces, Trans AMS, 3 (1922), 96-115.
- [43] E. Blum, W. Oettli, From optimization and variational inequalities to equilibrium problems, *Math. Stud.*, 63 (1994), 123-145.
- [44] H. Brézis, Operateurs maximaux monotones–North Holland., Math. Stud., 5 (1973), 1-183.
- [45] M. R. Bridson, A. Haefliger, Metric Spaces of Non-Positive Curvature, Fundamental Principle of Mathematical Sciences, Springer, Berlin, Germany, 319 (1999).
- [46] L. E. J. Brouwer, Uber abbildung von mannigfaltigkiten, mathematische annalen, 71 (1912), 97-115.
- [47] F. E. Browder, Convergence of approximants to fixed points of nonexpansive nonlinear mappings in Banach spaces, Archive for Rational Mechanics and Analysis, vol. 24, (1967), no. 1, pp. 82–89.
- [48] F. Bruhat, J. Tits, Groupes Réductifs sur un Corp Local, I. Donneés Radicielles Valuées, Institut des Hautes Études Scientiques, 41 (1972).
- [49] C.L. Byrne, A Moudafi, Extensions of the CQ algorithm for the split feasibility and split equality problems, J. Nonlinear Convex Anal., 18 (2017), no. 8, 1485-1496.
- [50] D. Butnariu, A.N. Iusem, Totally convex functions for fixed points computation and infinite dimensional optimization, London: Kluwer Academic Publishers, (2000).
- [51] C. Byrne, Y. Censor, A. Gibali, S. Reich, Weak and strong convergence of algorithms for the split common null point problem, J. Nonlinear Convex Anal., 13 (2012), 759–775.
- [52] C. Byrne, A unified treatment for some iterative algorithms in signal processing and image reconstruction, *Inverse Probl.*, **20** (2004), 103-120.
- [53] G. Cai, A. Gibali, O. S. Iyiola, Y. Shehu, A new double-projection method for solving variational inequalities in Banach spaces. J. Optim. Theory Appl., 178 (2018), 219-239.

- [54] L. C. Ceng, Q. H. Ansari, J. C. Yao, Relaxed extragradient methods for finding minimum-norm solutions of the split feasibility problem, *Nonlinear Anal.*, **75** (2012), 2116-2125.
- [55] L.C. Ceng, A. Petrusel, X. Qin, J. C. Yao, A modified inertial subgradient extragradient method for solving pseudomonotone variational inequalities and common fixed point problems, *Fixed Point Theory*, **21** (1) (2020), 93-108.
- [56] L. C. Ceng, J. C. Yao, A hybrid iterative scheme for mixed equilibrium problems and fixed point problems, J. Comput. Appl. Math., 214 (2008), 186-201.
- [57] Y. Censor and T. Elfving, A multiprojection algorithm using Bregman projections in product space, Numer. Algorithms, 8 (1994), 221-239.
- [58] Y. Censor, A. Gibali, S. Reich, Extensions of Korpelevich's extragradient method for the variational inequality problem in Euclidean space, *Optim.*, **61** (9) (2012), 1119-1132.
- [59] Y. Censor, A. Gibali, S. Reich, The subgradient extragradient method for solving variational inequalities in Hilbert space, J. Optim. Theory Appl., 148 (2011), 318-335.
- [60] Y. Censor, A. Gibali, S. Reich, Strong convergence of subgradient extragradient methods for the variational inequality problem in Hilbert space, *Optim. Meth Softw.*, 26 (2011), 827-845.
- [61] Y. Censor, A. Gibali, S. Reich, Algorithms for the split variational inequality problem, Numer. Algorithms, 59 (2012), 301-323.
- [62] Y. Censor, T. Bortfeld, B. Martin, A. Trofimov, A unified approach for inversion problems in intensity modulated radiation therapy, *Phys. Med. Biol.*, **51** (2006), 2353-2365.
- [63] Y. Censor, A. Gibali, S. Reich, The split variational inequality problem. The Technion-Israel Institute of Technology, Haifa, 59 (2012), 301-323.
- [64] Y. Censor, T. Elfving, A multiprojection algorithm using Bregman projections in a product space, Numer. Algorithms, 8 (1994), 221–239.
- [65] Y. Censor, S. Petra, C. Schnorr, Superiorization vs. accelerated convex optimization, The superiorized regularized least-squares case, J. Appl. Numer. Optim., 2 (2020), 15-62.
- [66] Y. Censor, A. Segal, The split common fixed point problem for directed operators, J. Convex Anal., 16 (2009), 587–600.
- [67] P. Chaipunya, P. Kumam, On the proximal point method in Hadamard spaces, Optimization, 66, (2017), 1647-1665.
- [68] A. Chambolle, Ch. Dossal, On the convergence of the iterates of the "fast iterative shrinkage/thresholding algorithm", J. Optim. Theory Appl., 166 (2015), 968-982.

- [69] S. S. Chang, H. W. J. Lee, C. K. Chan, A new method for solving equilibrium problem, fixed point problem and variational inequality problem with application to optimization, *Nonlinear Anal.*, **70** (2009), 3307-3319.
- [70] S. S. Chang, L. Wang, X. R. Wang, L. C. Zhao, Common solution for a finite family of minimization problem and fixed point problem for a pair of demicontractive mappings in Hadamard spaces, *Rev. R. Acad. Cienc. Exactas Fis. Nat. Ser. A Mat. RACSAM*, 114 (2), (2020), 1-12.
- [71] S.-S. Chang, L. Wang, L.J. Qin, Split equality fixed point problem for quasi-pseudocontractive mappings with applications, *Fixed Point Theory Appl.*, **2015** (2015), Art. No. 208, 12 pp.
- [72] S. S. Chang, L. Wang, C-F. Wen, J. Q. Zhang, The modified proximal point algorithm in Hadamard spaces, *J. Inequal. Appl.*, volume 2018, Article number: 124 (2018).
- [73] W. H. Chen, A note on geometric mean of positive matrices, Recent advances in mathematical methods, mathematical models and simulation in science and engineering, 32 (2014).
- [74] C. E. Chidume, Geometric properties of Banach spaces and nonlinear iterations, Springer Verlag Series, Lecture Notes in Mathematics, ISBN 978-1-84882-189-7, (2009).
- [75] C. E. Chidume, M. O. Nnakwe, Convergence theorems of subgradient extragradient algorithm for solving variational inequalities and a convex feasibility problem, *Fixed Point Theory Appl.*, (1) (2018), 1-14.
- [76] C. E. Chidume, M. O. Nnakwe, Iterative algorithms for split variational inequalities and generalized split feasibility problems with applications, J. Nonlinear Var. Anal. 3 (2019), 127-140.
- [77] B. J. Choi, U. C. Ji, The proximal point algorithm in uniformly convex metric spaces, Commun. Korean Math. Soc., 31 (2016), 845-855.
- [78] B. J. Choi, U. C. Ji, Y. Lim, Convex feasibility problems on uniformly convex metric spaces, Optim. Methods Softw., 35 (1) (2020), 21-36.
- [79] P. Cholamjiak, The modified proximal point algorithm in CAT(0) spaces, Optim. Lett., 9, (2015), 1401-1410.
- [80] P. Cholamjiak, A. A. Abdou, Y. J. Cho, Proximal point algorithms involving fixed points of nonexpansive mappings in CAT(0) spaces, *Fixed Point Theory Appl.*, (2015), 1-13.
- [81] W. Cholamjiak, S. A. Khan, S. Suantai, A Modified inertial shrinking projection method for solving inclusion problems and split equilibrium problems in Hilbert Spaces, *Comm. Math. Appl.*, **10** (2) (2019), 191-213.
- [82] P. Cholamjiak, A generalized forward-backward splitting method for solving quasi inclusion problems in Banach spaces, *Numer. Algor.*, 8 (1994), 221–239.

- [83] W. Cholamjiak, P. Cholamjiak, S. Suantai, An inertial forward-backward splitting method for solving inclusion problems in Hilbert spaces, J. Fixed Point Theory Appl., 20 (1) (2018), 1-17.
- [84] C. S. Chuang, Hybrid inertial proximal algorithm for the split variational inclusion problem in Hilbert spaces with applications, *Optimization*, **66** (5) (2017), 777-792.
- [85] I. Cioranescu, Geometry of Banach spaces, Duality Mappings and Nonlinear problems, (Kluwer, Dordrecht, 1990).
- [86] P. L. Combettes, V. R. Wajs, Signal recovery by proximal forward-backward splitting, *Multiscale Model. Simul.*, 4 (2005), 1168-1200.
- [87] P. L. Combettes, S. A. Hirstoaga, Equilibrium programming in Hilbert spaces, J. Nonlinear Convex Anal., 6 (2005), 117–136.
- [88] C. Conde, Geometric interpolation in p-Schatten class, J. Math. Anal. Appl., 340 (2) (2008), 920-931.
- [89] R. W. Cottle, J. C. Yao, Pseudo-monotone complementarity problems in Hilbert space, J. Optim. Theory Appl., 75 (1992), 281-295.
- [90] F. Cui, Y. Tang, Y. Yang, An inertial three-operator splitting algorithm with applications to image inpainting, Appl. Set-Valued Anal. Optim., 1 (2019), 113-134.
- [91] A. Cuntavepanit, W. Phuengrattana, On solving the minimization problem and the fixed-point problem for a finite family of nonexpansive mappings in CAT(0) spaces, *Optim. Methods Softw.* **33** (2) (2018), 311–321.
- [92] I. Daubechies, M. Defrise, C. De Mol, An iterative thresholding algorithm for linear inverse problems with a sparsity constraint, *Commun. Pure Appl. Math.*, 57 (2004), 1413-1457.
- [93] H. Dehghan, C. Izuchukwu, O. T. Mewomo, D.A. Taba, G. C. Ugwunnadi, Iterative algorithm for a family of monotone inclusion problems in CAT(0) spaces, *Quaest. Math.*, 43 (7) (2020), 975-998.
- [94] H. Dehghan, J. Rooin, Metric projection and convergence theorems for nonexpansive mappings in Hadamard spaces, *arXiv preprint arXiv*, **1410** (1137), (2014).
- [95] S. V. Denisov, V. V. Semenov, L. M. Chabak, Convergence of the modified extragradient method for variational inequalities with non-Lipschitz operators, *Cybern. Syst. Anal.*, **51** (2015), 757–765.
- [96] Q. Dong, Y. Cho, L. Zhong, T.M. Rassias, Inertial projection and contraction algorithms for variational inequalities, J. Glob. Optim., 70 (2018), 687-704.
- [97] Q. L. Dong, Y. Y. Lu, J. Yang, The extragradient algorithm with inertial effects for solving the variational inequality, *Optimization*, **65** (2016), 2217-2226.

- [98] J. Douglas, H. H. Rachford, On the numerical solution of the heat conduction problem in 2 and 3 space variables, *Trans. Amer. Math. Soc.*, 82 (1956), 421–439.
- [99] J. Duchi, Y. Singer, Efficient online and batch learning using forward backward splitting, J. Mach. Learn. Res., 10 (2009), 2899-2934.
- [100] G. Z. Eskandani, M. Raeisi, On the zero point problem of monotone operators in Hadamard spaces, *Numer. Algorithms*, **80** (2019), 1155-1179.
- [101] J. Eckstein, D. P. Bertsekas, On the Douglas-Rachford splitting method and the proximal point algorithm for maximal monotone operators, *Math. Program.*, 55 (1992), 293-318.
- [102] J. Fan, X. Qin, Weak and strong convergence of inertial Tseng's extragradient algorithms for solving variational inequality problems, *Optim.*, (2020), 1-22.
- [103] A. Feragen, S. Hauberg, M. Nielsen, F. Lauze, Means in spaces of tree-like shapes, in Proceedings of the IEEE International Conference on Computer Vision (ICCV), 2011, IEEE, Piscataway, NJ, (2011), 736-746.
- [104] G. Fichera, Sul problema elastostatico di Signorini con ambigue condizioni al contorno, Atti Accad. Naz. Lincei VIII. Ser. Rend. Cl. Sci. Fis. Mat. Nat. (1963), 34, 138-142.
- [105] A. Gibali, L. O. Jolaoso, O. T. Mewomo, A. Taiwo, Fast and simple Bregman projection methods for solving variational inequalities and related problems in Banach spaces, *Results Math.*, **75** (2020), 1-36.
- [106] A. Gibali, Y. Shehu, An efficient iterative method for finding common fixed point and variational inequalities in Hilbert spaces, *Optimization*, **68** (1) (2019), 13-32.
- [107] A. Gibali, A new non-Lipschitzian projection method for solving variational inequalities in Euclidean spaces, J. Nonlinear Anal. Optim., 6 (2015), 41-51.
- [108] A. Gibali, D. V. Thong, Tseng type methods for solving inclusion problems and its applications, *Calcolo*, 55 (4) (2018), 1-22.
- [109] E. C. Godwin, C. Izuchukwu, O. T. Mewomo, An inertial extrapolation method for solving generalized split feasibility problems in real Hilbert spaces, *Boll. Unione Mat. Ital.*, 14 (2) (2021), 379-401.
- [110] E.C. Godwin, T.O. Alakoya, O.T. Mewomo, J.-C. Yao, Relaxed inertial Tseng extragradient method for variational inequality and fixed point problems, *Appl. Anal.*, (2022), DOI:10.1080/00036811.2022.2107913.
- [111] E.C. Godwin, C. Izuchukwu, O.T. Mewomo, Image restoration using a modified relaxed inertial method for generalized split feasibility problems *Math. Methods Appl. Sci.*, (2022), DOI:10.1002/mma.8849.
- [112] K. Goebel, W. A. Kirk, Topics in metric fixed point theory (No. 28), Cambridge University Press, (1990).

- [113] O. Güler, On the convergence of the proximal point algorithm for convex minimization, SIAM J. Control Optim., 29 (1991), 403-419.
- [114] M. Gromov, *Hyperbolic groups. Essays in group theory*, Math. Sci. Res. Inst. Publ. 8. New York: Springer. 75-253.
- [115] G. López, V. Martin-Márquez, F.-H. Wang, H.-K. Xu, Solving the split feasibility problem without prior knowledge of matrix norms, Inverse Problems, 27 (2012), 18 pages.
- [116] D. R. Han, H. J. He, H. Yang, X. M. Yuan, A customized Douglas-Rachford splitting algorithm for separable convex minimization with linear constraints, *Numer. Math.*, **127** (2014), 167-200.
- [117] B. Halpern, Fixed points of nonexpanding maps, Bull. Amer. Math. Soc., 73 (6), (1967) 957-961.
- [118] P. Hartman, G. Stampacchia, On some non-linear elliptic differential-functional equations, Acta Math., 115 (1966), 271-310.
- [119] S. He, Q. L. Dong, H. Tian, Relaxed projection and contraction methods for solving Lipschitz continuous monotone variational inequalities, *Rev. R. Acad. Cienc. Exactas Fis. Nat. Ser. A Mat. (RACSAM)*, **113** (2019), 2763-2781.
- [120] S. He, T. Wu, A. Gibali, Q. L. Dong, Totally relaxed, self-adaptive algorithm for solving variational inequalities over the intersection of sub-level sets, *Optimization*, 67 (9) (2018), 1487-1504.
- [121] H. He, C. Ling, H. K. Xu, A relaxed projection method for split variational inequalities, J. Optim. Theory Appl., 166, (2015) 213-233.
- [122] Y. R. He, A new double projection algorithm for variational inequalities, J. Comput. Appl. Math., 185 (2006), 166-173.
- [123] B-S. He, Z-H. Yang, X-M. Yuan, An approximate proximal-extragradient type method for monotone variational inequalities, J. Math. Anal. Appl., 300 (2004), 362-374.
- [124] J. M. Hendrickx, A. Olshevsky, Matrix *P*-norms are NP-hard to approximate if $P \neq 1, 2, \infty$, SIAM J. Matrix Anal. Appl., **31** (2010), 2802-2812.
- [125] M. T. Heydari, A. Khadem, S. Ranjbar, Approximating a common zero of finite family of monotone operators in Hadamard spaces, *Optimization*, 66 (12) (2017), 2233–2244.
- [126] H. Iiduka, Acceleration method for convex optimization over the fixed point set of a nonexpansive mappings, *Math. Prog. Series A.*, **149** (1-2) (2015), 131-165.
- [127] S. Ishikawa, Fixed points by a new iteration method, *Proc. Amer. Math. Soc.*, 44 (1) (1974), 147-150.

- [128] A. N. Iusem, M. Nasri, Korpelevich's method for variational inequality problems in Banach spaces, J. Global Optim., 50 (2011), 59-76.
- [129] F. Iutzeler, J. M. Hendrickx, A generic online acceleration scheme for optimization algorithms via relaxation and inertia, *Optim. Methods Softw.*, **34** (2), (2019), 383-405.
- [130] C. Izuchukwu, A. A. Mebawondu, O. T. Mewomo, A New Method for Solving Split Variational Inequality Problems without Co-coerciveness, J. Fixed Point Theory Appl., 22 (4), (2020), Art. No. 98, 23 pp.
- [131] C. Izuchukwu, C. C. Okeke, O. T. Mewomo, Systems of variational inequality problem and multiple-sets split equality fixed point problem for infinite families of multivalued type-one demicontractive-type mappings, *Ukrainian Math. J.*, **71** (2019), 1480-1501.
- [132] C. Izuchukwu, G. N. Ogwo, O. T. Mewomo, An Inertial Method for solving Generalized Split Feasibility Problems over the solution set of Monotone Variational Inclusions, *Optimization*, (2020), DOI 10.1080/02331934.2020.1808648.
- [133] C. Izuchukwu, A. A. Mebawondu, K. O. Aremu, H. A. Abass, O. T. Mewomo, Viscosity iterative techniques for approximating a common zero of monotone operators in a Hadamard space, *Rend. Circ. Mat. Palermo.*, 69 (2) (2020), 475-495.
- [134] L.O. Jolaoso, A. Taiwo, T.O. Alakoya, O.T. Mewomo, A self adaptive inertial subgradient extragradient algorithm for variational inequality and common fixed point of multivalued mappings in Hilbert spaces, *Demonstr. Math.*, **52** (2019), 183-203.
- [135] L.O. Jolaoso, A. Taiwo, T.O Alakoya, O.T. Mewomo, Strong convergence theorem for solving pseudo-monotone variational inequality problem using projection method in a reflexive Banach space, J. Optim. Theory Appl., 185 (3) (2020), 744–766.
- [136] L.O. Jolaoso, T.O. Alakoya, A. Taiwo, O.T. Mewomo, Inertial extragradient method via viscosity approximation approach for solving Equilibrium problem in Hilbert space, *Optimization*, **70** (2), (2021), 387-412.
- [137] J. S. Jung, Iterative approaches to common fixed points of nonexpansive mappings in Banach spaces, J. Math. Anal. Appl., 302 (2) (2005),509-520.
- [138] B. A. Kakavandi, M. Amini, B. A. Kakavandi, M. Amini, Duality and subdifferential for convex functions on complete CAT(0) metric spaces, *Nonlinear Anal.*, **73** (2010), 3450-3455.
- [139] S. Kamimura, W. Takahashi, Strong convergence of a proximal-type algorithm in a Banach space, SIAM J. Optim., 13 (3) (2002), 938–945.
- [140] S. Kamimura, W. Takahashi, Approximating solutions of maximal monotone operators in Hilbert spaces, J. Approx. Theory, 106 (2000), 226-240.
- [141] K. R. Kazmi, S. H. Rizvi, An iterative method for split variational inclusion problem and fixed point problem for a nonexpansive mapping, *Optim. lett.*, 8 (3) (2014), 1113-1124.

- [142] K.R. Kazmi, Split nonconvex variational inequality problem, Math. Sci., 7 (1) (2013),1-5.
- [143] K.R. Kazmi, Split general quasi-variational inequality problem, Georgian Math. J., 22 (3) (2015), 1-8.
- [144] K. R. Kazmi, S. H. Rizvi, Iterative approximation of a common solution of a split equilibrium problem, a variational inequality problem and a fixed point problem, J. Egyptian Math. Soc., 21 (2013), 44–51.
- [145] G.E. Kim, Weak and strong convergence theorems of quasi-nonexpansive mappings in a Hilbert spaces, J. Optim. Theory Appl., 152 (3) 2012, 727-738.
- [146] J.K. Kim, S. Salahuddin, W.H. Lim, General nonconvex split variational inequality problems, *Korean J. Math.*, 25 (2017), 469-481.
- [147] D. Kinderlehrer, G. Stampacchia, An introduction to variational inequalities and their applications. New York: Academic Press; 1980.
- [148] W.A. Kirk, Some recent results in metric fixed point theory, J. Fixed Point Theory Appl., 2017 (2007), 195-207.
- [149] S. H. Khan, T.O. Alakoya, O.T. Mewomo, Relaxed projection methods with selfadaptive step size for solving variational inequality and fixed point problems for an infinite family of multivalued relatively nonexpansive mappings in Banach spaces, *Math. Comput. Appl.*, 25, (2020), Art. 54, 31 pp.
- [150] H. Khatibzadeh, S. Ranjbar, Monotone operators and the proximal point algorithm in complete CAT(0) metric spaces, J. Aust. Math Soc., 103 (1), (2017), 70-90.
- [151] E. N. Khobotov, Modification of the extragradient method for solving variational inequalities and certain optimization problems, USSR Comput Math Math Phys., 27 (1989), 120-127.
- [152] F. Kohsaka, W. Takahashi, Fixed point theorems for a class of nonlinear mappings related to maximal monotone operators in Banach spaces, Arch Math., 91 (2) (2018),166–177.
- [153] F. Kohsaka, W. Takahashi, Existence and approximation of fixed points of firmly nonexpansive-type mappings in Banach spaces, SIAM J Optim. 19 (2) (2018), 824–835.
- [154] H. Komiya, W. Takahashi, Strong convergence theorem for an infinite family of demimetric mappings in a Hilbert space, J. Convex Anal., 24(4), (2017).
- [155] G. M. Korpelevich, An extragradient method for finding sadlle points and for other problems, *Ekon. Mat. Metody*, **12** (1976), 747-756.
- [156] R. Kraikaew, S. Saejung, Strong convergence of the Halpern subgradient extragradient method for solving variational inequalities in Hilbert spaces, J. Optim. Theory Appl., 163 (2014), 399-412.

- [157] M.A. Krasnoselskij, Two observations about the method of successive approximations, Uspehi Math. Nauk., 10 (1955),123-127.
- [158] W. Laowang, B. Panyanak, Strong and △- convergence theorems for multivalued mappings in CAT(0) spaces, J. Ineq. Appl., 200 (2009), Art. ID 730132, 16 pp.
- [159] A. Latif, M. Eslamian, Split equality problem with equilibrium problem, variational inequality problem, and fixed point problem of nonexpansive semigroups, J. Nonlinear Sci. Appl., 10 (2017), 3217-3230.
- [160] L. J. Lin, M. F. Yang, Q. H. Ansari, G. Kassay, Existence results for Stampacchia and Minty type implicit variational inequalities with multivalued maps, *Nonlinear Anal. Theory Methods Appl.*, **61** (2005), 1–19.
- [161] P. L. Lions, B. Mercier, Splitting algorithms for the sum of two nonlinear operators, SIAM J. Numer. Anal., 16 (1979), 964–979.
- [162] H. Liu, J. Yang, Weak convergence of iterative methods for solving quasimonotone variational inequalities, *Comput. Optim. Appl.*, 77 (2), (2020) 491-508.
- [163] L. V. Long, D. V. Thong, V. T. Dung, New algorithms for the split variational inclusion problems and application to split feasibility problems, *Optim.*, 68 (12) (2019), 2339-2367.
- [164] G. López, M.V. Márquez, F. Wang, H.K. Xu, Forward-backward splitting methods for accretive operators in Banach spaces, *Abstr. Appl. Anal.*, **2012** (2012), Article ID 109236, doi.org/10.1155/2012/109236.
- [165] D.A. Lorenz, T. Pock, An inertial forward-backward algorithm for monotone inclusions, J. Math. Imaging Vis., 51 (2015), 311-325.
- [166] P. E. Maingé, Approximation methods for common fixed points of nonexpansive mappings in Hilbert spaces, J. Math. Anal. Appl., 325 (1) (2007), 469-479.
- [167] P. E. Maingé, Convergence theorems for inertial KM-type algorithms, J. Comput. Appl. Math., 219 (1) (2008), 223-236.
- [168] P. E. Maingé, A hybrid extragradient-viscosity method for monotone operators and fixed point problems, SIAM J. Control Optim., 47 (2008), 1499-1515.
- [169] P. E. Maingé, A viscosity method with no spectral radius requirements for the split common fixed point problem, *Eur. J. Oper. Res.*, 235 (2014), 17-27.
- [170] Y. Malitsky, Projected reflected gradient methods for monotone variational inequalities, SIAM J. Optim., 25 (1) (2015), 502-520.
- [171] W. R. Mann, Mean value methods in iteration, Proc. Amer. Math. Soc., 4 (3) (1953) 506-510.
- [172] G. Marino, H. K. Xu, A general iterative method for nonexpansive mappings in Hilbert spaces, J. Math. Anal. Appl., 318 (2006), 43-52.

- [173] G. Marino, H. K. Xu, Weak and strong convergence theorems for strict pseudocontractions in Hilbert space, J. Math. Anal. Appl., 329 (2007), 336-346.
- [174] B. Martinet, Regularisation d'inequations variationelles par approximations, Rev Francaise d'inform et de Rech Re, 3 (1970), 154-158.
- [175] J. Mashregi, M. Nasri, Forcing strong convergence of Korpelevich's method in Banach spaces with applications in game theory, *Nonlinear Anal.*, **72** (2010), 2086–2099.
- [176] K. Menger, Selecta mathematica, vol. 1 and 2, ed. by B. Schweizer, A. Sklar, K. Sigmund, P. Gruber, E. Hlawka, L. Reich and L. Schmetterer, Springer-Verlag, Wien 2002 and 2003.
- [177] O.T. Mewomo, F.U. Ogbuisi, Convergence analysis of an iterative method for solving multiple-set split feasibility problems in certain Banach spaces, *Quaest. Math.*, **41** (1) (2018), 129–148.
- [178] C. Mongkolkeha, Y. J. Cho, P. Kumam, Convergence theorems for k-demicontractive mappings in Hilbert spaces, *Math. Inequal. Appl.*, 16 (4) (2013), 1065-1082.
- [179] J. J. Moreau, Proximité et dualité dans un espace hilbertien. Bull, Soc. Math., 93 (1965), 273-299.
- [180] A. Moudafi, E. Al-Shemas, Simultaneous iterative methods for split equality problem, Trans. Math. Program. Appl., 1 (2013), 1–11.
- [181] A. Moudafi, Viscosity approximation method for fixed points problems, J. Math. Anal. Appl., 241 (2000), 46-55.
- [182] A. Moudafi, B. S. Thakur, Solving proximal split feasibility problems without prior knowledge of operator norms, *Optim. Lett.*, 8 (7) (2014), 2099-2110.
- [183] A. Moudafi, Split monotone variational inclusions, J. Optim. Theory Appl., 150 (2011), 275-283.
- [184] A. Moudafi, M. Oliny, Convergence of a slitting inertial proximal method for monotone operators, J. Comput. Appl. Math., 155 (2003), 447–454.
- [185] N. Nadezhkina, W. Takahashi, Weak convergence theorem by an extragradient method for nonexpansive mappings and monotone mappings, J. Optim. Theory Appl., 128 (2006), 191-201.
- [186] S.A. Naimpally, K.L. Singh, Extensions of some fixed point theorems of Rhoades, J. math. anal. appl., 96 (2) 1983, 437-446.
- [187] Y. Nesterov, A method of solving a convex programming problem with convergence rate $O(1/k^2)$, Soviet Math. Doklady, 27 (1983), 372-376.
- [188] F. Nielsen, R. Bhatia, Matrix information geometry, Springer-verlag Berlin Heidelberg (2013), 199-207.

- [189] M.A. Noor, New approximation schemes for general variational inequalities, J. Math. Anal. Appl., 251 (2000), 217-229.
- [190] F. U. Ogbuisi, O. T. Mewomo, Convergence analysis of an inertial accelerated iterative algorithm for solving split variational inequality problems, Adv. Pure Appl. Math., 10 (4) (2019), 339-353.
- [191] G. N. Ogwo, C. Izuchukwu, K.O. Aremu, O.T. Mewomo, On θ-generalized demimetric mappings and monotone operators in Hadamard spaces, *Demonstr. Math.*, 53 (1) (2020), 95–111.
- [192] G.N. Ogwo, C. Izuchukwu, O.T. Mewomo, Inertial methods for finding minimumnorm solutions of the split variational inequality problem beyond monotonicity, *Numer. Algorithms*, 88 (3) (2021), 1419-1456.
- [193] G. N. Ogwo, C. Izuchukwu, Y. Shehu, O. T. Mewomo, Convergence of relaxed inertial subgradient extragradient methods for quasimonotone variational inequality problems, J. Sci. Comput., 90 (1), Paper No. 10 (2021), 1-35.
- [194] G. N. Ogwo, C. Izuchukwu, K.O. Aremu, O.T. Mewomo, A viscosity iterative algorithm for a family of monotone inclusion problems in an Hadamard space, *Bull. Belg. Math. Soc. Simon Stevin*, **27** (1) (2020), 127-152.
- [195] G.N. Ogwo, T.O. Alakoya, O.T. Mewomo, Inertial iterative method with selfadaptive step size for finite family of split monotone variational inclusion and fixed point problems in Banach spaces, *Demonstr. Math.*, 55 (1) (2021), 193-216
- [196] G.N. Ogwo, H.A. Abass, C. Izuchukwu, O. T. Mewomo, Modified proximal point methods involving quasi-pseudocontractive mappings in Hadamard spaces, *Acta Math. Vietnam.*, 47 (4) (2022), 847–873.
- [197] G. N. Ogwo, T. O. Alakoya, O. T. Mewomo, Iterative algorithm with self-adaptive step size for approximating the common solution of variational inequality and fixed point problems, *Optimization* (2021), DOI:10.1080/02331934.2021.1981897.
- [198] M.A. Olona, T.O. Alakoya, A. O.-E. Owolabi, O.T. Mewomo, Inertial shrinking projection algorithm with self-adaptive step size for split generalized equilibrium and fixed point problems for a countable family of nonexpansive multivalued mappings, *Demonstr. Math.*, 54 (1) (2021), 47-67.
- [199] Z. Opial, Weak convergence of successive approximations for nonexpansive mappings, Bull. Amer. Math. Soc., 73 (1967), 591-597.
- [200] M. O. Osilike, E. E. Chima, Mixed equilibrium and multiple-set split feasibility problems for asymptotically k-strictly pseudononspreading mappings, *Commun. Optim. Theory*, (2020), Article ID 14.
- [201] A. O.-E. Owolabi, T.O. Alakoya, A. Taiwo, O.T. Mewomo, A new inertial-projection algorithm for approximating common solution of variational inequality and fixed point problems of multivalued mappings, *Numer. Algebra Control Optim.*, **12** (2) 255, (2021).

- [202] A. Papadopoulos, Metric spaces, convexity and nonpositive curvature. IRMA Lectures in Mathematics and Theoretical Physics, 6. European Mathematical Society (EMS), Zürich, 2005.
- [203] G. B. Passty, Ergodic convergence to a zero of the sum of monotone operators in Hilbert space, J. Math. Anal. Appl., 72 (1979), 383-390.
- [204] D. H. Peaceman, H. H. Rachford, The numerical solution of parabolic and elliptic differentials, J. Soc. Indust. Appl. Math., 3 (1955), 28–41.
- [205] J. W. Peng, Y. C. Liou, J. C. Yao, An iterative algorithm combining viscosity method with parallel method for a generalized equilibrium problem and strict pseudocontractions, *Fixed Point Theory Appl.*, (2009), 1-21.
- [206] V. H. Pham, D. H. Nguyen, T. V. Anh, A strongly convergent modified Halpern subgradient extragradient method for solving the split variational inequality problem, *Vietnam J. Math.*, 48, 187-204 (2020).
- [207] H. Poincare, Surless courbes de ne barles equations differentiate less, J. de Math., 2 (1886), 54-65
- [208] B.T. Polyak, Some methods of speeding up the convergence of iterates methods, U.S.S.R Comput. Math. Phys., 4 (5) (1964), 1-17.
- [209] X. Qin, Y. J. Cho, S. M. Kang, Convergence theorems of common elements for equilibrium problems and fixed point problems in Banach spaces, J. Comput. Appl. Math., 225 (2009), 20-30.
- [210] H. Raguet, J. Fadili, G. Peyré, A generalized forward-backward splitting, SIAM J. Imaging Sci., 6 (2013), 1199-1226.
- [211] S. Ranjbar, H. Khatibzadeh, Strong and Δ -convergence to a zero of a monotone operator in CAT(0) spaces, *Mediterr. J. Math.*, **14** (2), (2017), 15pp.
- [212] S. Reich, Weak convergence theorems for nonexpansive mappings in Banach spaces, J. Math. Anal. Appl., 67 (1979), 274-276.
- [213] S. Reich, T. M. Tuyen, A new algorithm for solving the split common null point problem in Hilbert spaces, *Numer. Algorithms*, 83 (2020), 789-805.
- [214] S. Reich, D. V. Thong, Q. L. Dong, X. H. Li, V. T. Dung, New algorithms and convergence theorems for solving variational inequalities with non-Lipschitz mappings, *Numer. Algorithms*, 87 (2021), 527–549.
- [215] R. T. Rockafellar, Monotone operators and the proximal point algorithm, SIAM J. Control Optim., 14, 877-898, (1976).
- [216] R. T. Rockafellar, On the maximality of sums of nonlinear monotone operators, Trans. Amer. Math. Soc. 149 (1970), 75-288.

- [217] Salahuddin, The extragradient method for quasi-monotone variational inequalities, *Optim.*, (2020), 1-10.
- [218] K. Sitthithakerngkiet, J. Deepho, P. Kumam, A hybrid viscosity algorithm via modify the hybrid steepest descent method for solving the split variational inclusion in image reconstruction and fixed point problems, *Appl. Math.Comput.*, **250** (2015) 986-1001.
- [219] Y. Shehu, O. T. Mewomo, F. U. Ogbuisi, Further investigation into approximation of a common solution of fixed point problems and split feasibility problems, Acta. Math. Sci. Ser. B, Engl. Ed., 36 (3) (2016), 913-930.
- [220] Y. Shehu, P. Cholamjiak, Iterative method with inertial for variational inequalities in Hilbert spaces, *Calcolo*, 56 (1) (2019), 1-21.
- [221] Y. Shehu, X. H. Li, Q. L. Dong, An efficient projection-type method for monotone variational inequalities in Hilbert spaces, *Numer. Algorithms*, 84 (2020), 365-388.
- [222] Y. Shehu, Q.L. Dong, D. Jiang, Single projection method for pseudo-monotone variational inequality in Hilbert spaces Optimization, 68 (1) (2019), 385-409.
- [223] Y. Shehu, O. S. Iyiola, Strong convergence result for monotone variational inequalities, Numer. Algor., 76 (1) (2017), 259-282.
- [224] L. Shi, R. Chen, Y. Wu, Strong convergence of iterative algorithms for the split equality problem, J. Inequal. Appl., 2014, 2014, Article ID 478.
- [225] K. Shimoji, W. Takahashi, Strong convergence to common fixed points of infinite nonexpansive mappings and applications, *Taiwanese J. Math.*, 5 (2) (2001), 387-404.
- [226] S. Somalia, Implicit midpoint rule to the nonlinear degenerate boundary value problems, Internat. J. Comput. Math., 79 (2002), 327-332.
- [227] S. Sra, R. Hosseini, Conic geometric optimization on the manifold of positive definite matrices, SIAM J. Optim., 25(1), (2014), 713-739.
- [228] G. Stampacchia, "variational Inequalities", in: Theory and Applications of Monotone Operators, Proceedings of the NATO Advanced Study Institute, Venice, Italy (Edizioni Odersi, Gubbio, Italy, 1968), 102-192.
- [229] S. Suantai, K. Srisap, N. Naprang, M. Mamat, V. Yundon, P. Cholamjiak, Convergence theorems for finding the split common null point in Banach spaces, *Appl. Gen. Topol.*, 18 (2) (2017), 345-360.
- [230] E. Süli, D.F. Mayers, An introduction to numerical analysis, Cambridge University Press, Cambridge, 2003, x+433 pp.
- [231] D. Sun, A new step-size skill for solving a class of nonlinear projection equations, J. Comput. Math., 13 (1995), 357-368.

- [232] R. Suparatulatorn, P. Cholamjiak, S. Suantai, On solving the minimization problem and the fixed-point problem for nonexpansive mappings in CAT(0) spaces, *Optim. Methods Softw.*, **32** (1) (2016), 182-192.
- [233] A. Taiwo, T.O. Alakoya, O.T. Mewomo, Halpern-type iterative process for solving split common fixed point and monotone variational inclusion problem between Banach spaces, *Numer. Algorithms*, 86 (1) (2021), 1359–1389.
- [234] A. Taiwo, L.O. Jolaoso, O.T. Mewomo, Viscosity approximation method for solving the multiple-set split equality common fixed point problems for quasi-pseudocontractive mappings in Hilbert Spaces, J. Ind. Manag. Optim., bf 17 (5) (2021), 2733.
- [235] A. Taiwo, L.O. Jolaoso, O.T. Mewomo, Inertial-type algorithm for solving split common fixed-point problem in Banach spaces, J. Sci. Comput., 86 (1) (2021), 1-30.
- [236] W. A. Takahashi, K. Shimoji, Convergence Theorems for Nonexpansive Mappings and Feasibility Problems, *Math. Comput. Modelling*, **32** (2000), 1463-1471.
- [237] W. Takahashi, *Convex Analysis and Approximation of Fixed points*, Yokohama Publishers, Yokohama, (2000). (Japanese).
- [238] W. Takahashi, H. K. Xu, J. C. Yao, Iterative methods for generalized split feasibility problems in Hilbert spaces, *Set-valued Var. Anal.*, 23 (2) (2015) 205-221.
- [239] W.Takahashi, Nonlinear functional analysis-Fixed Point Theory and its Applications, Yokohama Publishers, Yokohama, (2000).
- [240] B. Tan, S. Li, X. Qin, Self-adaptive inertial single projection methods for variational inequalities involving non-Lipschitz and Lipschitz operators with their applications to optimal control problems, *Appl. Numer. Math.*, **170** (2021), 219-241.
- [241] D. Tian, L. Jiang, Two-step methods and relaxed two-step methods for solving the split equality problem, *Comput. Appl. Math.*, 40 (3) (2021), Article ID 83.
- [242] K. K. Tan, H. K. Xu, Approximating fixed points of nonexpansive mappings by the Ishikawa iteration process, J. Math. Anal. Appl., 178 (1993), 301-308.
- [243] M. Tian, B-N. Jiang, Weak convergence theorem for a class of split variational inequality problems and applications in Hilbert space, J. Ineq. Appl., (1) (2017), 1-17.
- [244] M. Tian, B-N. Jiang, Viscosity approximation methods for a class of generalized split feasibility problems with variational inequalities in Hilbert space, *Numer. Funct. Anal. Optim.*, 40 (2019), 902-923.
- [245] M. Tian, G. Xu, Inertial modified Tseng's extragradient algorithms for solving monotone variational inequalities and fixed point problems, *Nonlinear Funct. Anal*, 35, (2020).
- [246] D. V. Thong, D. V. Hieu, Inertial subgradient extragradient algorithms with linesearch process for solving variational inequality problems and fixed point problems, *Numer. Algorithms*, 80 (2019), 1283-1307.

- [247] D. V. Thong, D. V. Hieu, An inertial method for solving split common fixed point problems, J. Fixed Point Theory Appl., 19 (4) (2017), 3029-3051.
- [248] D. V. Thong, P. T. Vuong, Modified Tseng's extragradient methods for solving pseudo-monotone variational inequalities, *Optimization*, (2019), https://doi.org/10.1080/02331934.2019.1616191.
- [249] D. V. Thong, D. V. Hieu, Weak and strong convergence theorems for variational inequality problems, *Numer. Algor.*, 78 (4) (2018), 1045-1060.
- [250] D.V. Thong, Y. Shehu, O.S. Iyiola, Weak and strong convergence theorems for solving pseudo-monotone variational inequalities with non-Lipschitz mappings, *Numer. Algor.*, 84, (2019), 795-823.
- [251] D.V. Thong, Y. Shehu, O.S. Iyiola, A new iterative method for solving pseudomonotone variational inequalities with non-Lipschitz operators, *Comp. Applied Math.*, **39** (2020), Art. No. 108.
- [252] P. Thounthong, N. Pakkaranang, Y.J. Cho, W. Kumam, P. Kumam, The numerical reckoning of modified proximal point methods for minimization problems in non-positive curvature metric spaces, *Int. J. Comp. Math.*, 97, (2020), 1-24.
- [253] P. Tseng, A modified forward-backward splitting method for maximal monotone mappings, SIAM J. Control Optim., 38 (2000), 431-446.
- [254] G. C. Ugwunnadi, A. R. Khan, M. Abbas, A hybrid proximal point algorithm for finding minimizers and fixed points in CAT(0) spaces, J. Fixed Point Theory Appl., 20 (2) (2018), Art. No. 82, 19 pp.
- [255] G. C. Ugwunnadi, C. Izuchukwu, O. T. Mewomo, Strong convergence theorem for monotone inclusion problem in CAT(0) spaces, Afr. Mat., 30 (1-2) (2019), 151-169.
- [256] V.A. Uzor, T.O. Alakoya, O.T. Mewomo, On Split Monotone Variational Inclusion Problem with Multiple Output Sets with fixed point constraints, *Comput. Methods Appl. Math.*, (2022), DOI: 10.1515/cmam-2022-0199.
- [257] V.A. Uzor, T.O. Alakoya, O.T. Mewomo, Strong convergence of a self-adaptive inertial Tseng's extragradient method for pseudomonotone variational inequalities and fixed point problems, *Open Math.* 20, (2022), 234–257.
- [258] A. Wald, Begründung einer koordinatenlosen Differentialgeometrie der Flächen. Erg. Math. Kolloqu., 7 (1936), 24-46.
- [259] S. Wang, A general iterative method for obtaining an infinite family of strictly pseudo-contractive mappings in Hilbert spaces, Applied Math. Letters, 24 (2011), 901-907.
- [260] H. K. Xu, Iterative algorithms for nonlinear operators, J. London. Math. Soc., 68 (2002), 240-256.

- [261] H. K. Xu, Iterative methods for the split feasibility problem in infinite dimensional Hilbert spaces, *Inverse Problem*, 26 105018 (2010).
- [262] H. K. Xu, Inequalities in Banach spaces with applications. Nonlinear Anal., 16,(1991), 1127–1138.
- [263] H. K. Xu, M. A. Alghamdi, N. Shahzad, The viscosity technique for the implicit midpoint rule of nonexpansive mappings in Hilbert spaces, *Fixed Point Theory Appl.*, 2015 (2015), Art. 41, 12 pp.
- [264] H. K. Xu, Viscosity approximation methods for nonexpansive mappings, J. Math. Anal. Appl., 298 (1) (2004), 279-291.
- [265] R. Wangkeeree, P. Preechasilp, Viscosity approximation methods for nonexpansive mappings in CAT(0) spaces, J. Inequal. Appl., 2013, (2013), 15 pp.
- [266] J. Yang, H. Liu, Strong convergence result for solving monotone variational inequalities in Hilbert space, Numer. Algor., 80 (3) (2019), 741-752.
- [267] J. Yang, H. Liu, Z. Liu, Modified subgradient extragradient algorithms for solving monotone variational inequalities, *Optimization*, 67 (2018), 2247-2258.
- [268] Y. Yao, M. Postolache, X. Qin, J.-C. Yao, Iterative algorithm for proximal split feasibility problem. U.P.B. Sci. Bull., Series A 80 (3) (2018), 37-44.
- [269] M. Ye, Y. He, A double projection method for solving variational inequalities without monotonicity, *Comput. Optim. Appl.*, **60** (1) (2015), 141-150.
- [270] T. C. Yin, Y. K. Wu, C. F. Wen, An Iterative Algorithm for Solving Fixed Point Problems and Quasimonotone Variational Inequalities, J. Math., 2022.
- [271] J. Zhao, Solving split equality fixed-point problem of quasi-nonexpansive mappings without prior knowledge of operators norms, *Optimization*, 64 (2014), 2619–2630.
- [272] J. Zhao, S. He, Strong convergence of the viscosity approximation process for the split common fixed-point problem of quasi-nonexpansive mappings, J. Appl. Math., (2012), Art. ID 438023, 12 pages.
- [273] J. Zhao, S. Wang, Viscosity approximation methods for the split equality common fixed point problem of quasi-nonexpansive operators, Acta Math. Sci. Ser. B (Engl. Ed.), 36 (2016), no. 5, 1474-1486.
- [274] J. Zhao, D. Hou, A self-adaptive iterative algorithm for the split common fixed point problems, *Numer. Algorithms*, 82 (2019), 1047-1063.
- [275] J. Zhao, Solving split equality fixed-point problem of quasi-nonexpansive mappings without prior knowledge of operators norms, *Optimization*, 64 (2015), 2619-2630.
- [276] H. Zhou, Convergence theorems of fixed points for κ -strict pseudo-contractions in Hilbert spaces, Nonlinear Anal., **69**(2) (2008) 456-462.